The Use of Timber in a Temporary Multi-storey Car Park

Proof of Concept for a Structural System for a Temporary Multi-storey Car Park

P. Brink





AF 45 822

* Figure front page: (JAJA Architects + Open Platform, 2019)

The Use of Timber in a Temporary Multi-storey Car Park

Proof of Concept for a Structural System for a Temporary Multi-storey Car Park

by

Paul Brink

4375750

Delft University of Technology Faculty of Civil Engineering and Geosciences

March 2023

Thesis committee:Dr. M.A. (Mariana) Popescu (Chair)Delft University of TechnologyIr. S. (Sander) PasterkampDelft University of TechnologyDr. ir. G.A. (Sander) Van NederveenDelft University of TechnologyIr. E.A.M. (Erwin) Jacobs (Daily supervisor)Sweco Nederland B.V.

Preface

This report presents my research on the applicability of timber in a temporary multi-storey car park. It is a master thesis report, which for me concludes my time as a student at the TU Delft. During my bachelor in civil engineering and master building engineering, I have become more observant for the role of the building industry in global warming. Furthermore, timber structures have always intrigued me with their beautiful and warm appearance and in recent years, building in timber has really taken off. This has triggered me to search for a topic for my thesis, related to building with timber, which has become the use of timber in a temporary multi-storey car park.

Apart for my interest in timber, in my master, I have developed an interest in using digital tools in the field of structural engineering. I'm motivated to automate repeating tasks and search for new insights using for example parametric engineering in Grasshopper. Although this is not the topic for my master thesis, it was an important aspect for me to use such digital tools during my thesis and develop new skills in the digital field of structural engineering. This has worked out and as a result, I have gained new insights in the applicability of digital tools in structural engineering as well as its current limitations, motivating me to help further develop this field in my future career.

Conducting this research would not have been possible without the members of my committee and various colleagues from Sweco. First of all, I would like to thank Erwin Jacobs for his close commitment to the project and his personal guidance. Our weakly Fika's have not only helped me with my thesis, but were also always a source of inspiration for what is possible in the field of digital engineering. Second, I'd like to thank Steven Janssens for his guidance and expertise in structural engineering, helping me further during our Fika's. Also a big thanks to Mariana Popescu for her help in narrowing down the research questions and sharing her view on the application of digital tools in engineering. I would like to thank Sander Pasterkamp for his critical eye on the structural calculations, keeping me sharp. Finally, also thank you to Sander van Nederveen for his help on the different views of the design and research process of a proof of concept.

At this point I would also like to say a big thank you to all my friends. Writing a master thesis is a long process during which you come across many challenges. It is great to always feel the support of the people close around you and all the great times together are extremely motivating to continue the work. Finally, I would also like to thank my parents, brother and sister for their support during not just one, but two master thesis projects and of course also the rest of my study time.

Concluding this preface, this report marks the end of my time as a student in Delft. Studying has been an amazing journey, which started with my bachelor at the TU Delft. Joining DSB has given me opportunities to develop me further on personal skills and hobbies with as a highlight a full-time year without studying as the board of the fraternity. It has always felt as a second home to me, where I have made friends for the rest of my life. Studying during the COVID-19 pandemic has been weird, but also has a positive outcome as with this thesis I'm finishing my second master. For now the time has come to move on to the next phase and say goodbye to Delft. Enjoy reading this thesis!

P. Brink

Delft, March 2023

iv

Table of Contents

Pre	reface iii					
Ab	\bstract xi					
PA	RTI	Introductio	on	0		
1	Intro	oduction		1		
	1.1	Background a	nd Relevance			
	1.2	Research Def	nition	2		
		1.2.1 Proble	m Statement	2		
		1.2.2 Object	ive	2		
		1.2.3 Resea	rch Questions	2		
		1.2.4 Scope		3		
		1.2.5 Metho	dology			
	1.3	Reading Guid	Э	7		

10

PART II: Literature Study

2	Car Park Design			
	2.1	User Requirements		
		2.1.1	Car Park Category	
		2.1.2	Dimensions of Lanes and Ramps	
		2.1.3	Dimensions of Bays and Aisles	
		2.1.4	Circulation	
		2.1.5	Miscellaneous User Experiences	
	2.2	Owner	r Requirements	
		2.2.1	General	
		2.2.2	Cost-efficiency	
		2.2.3	Design Flexibility	
		2.2.4	Sustainability	
		2.2.5	Construction Process	
		2.2.6	Durability and Maintenance	
	2.3	Struct	ural Rules and Norms	15
		2.3.1	Preconditions and Assumptions	
		2.3.2	Loads and Limit States	
		2.3.3	Vibrations	
		2.3.4	Fire Safety	
	2.4	Car Pa	ark Design Rules and Norms	20
		2.4.1	Dimensions of Lanes and Ramps	
		2.4.2	Dimensions of Parking Bays and Aisles	
		2.4.3	Slope Percentages and Clearance Heights	24
		2.4.4	Other Considerations	
	2.5	Car Pa	ark Layout	27
		2.5.1	Circulation Design Aspects	
		2.5.2	Circulation Layouts	
	2.6	Struct	ural Typologies	
		2.6.1	Deck System	
		2.6.2	Framing System	
		2.6.3	Stability System	
	2.7	Conclu	uding Remarks	

3	Challenges Towards a Sustainable Building Industry 3			
	3.1	Building Sustainable	39	
		3.1.1 Impact of the Building Industry	39	
		3.1.2 Principles for Sustainable Construction	40	
		3.1.3 Barriers of Sustainable Construction	40	
		3.1.4 Drivers of Sustainable Construction	41	
	3.2	Circular Construction	41	
		3.2.1 Resource Loops	42	
		3.2.2 Circular Design Strategies		
	3.3	Incorporating Circular Design Strategies		
		3.3.1 Modularization Prefabrication and Standardization	44	
		3.3.2 Reuse Refurbish and Recvcle	44	
		3.3.3 Building Materials		
		3.3.4 Digital Tools	45	
	34	Sustainability Assessment	46	
	3.5 Concluding remarks		46	
	0.0			
4	Tim	ber Structures	49	
	4.1	Wood Products	49	
		4.1.1 Soft- and Hardwood	49	
		4.1.2 Solid Wood and Solid Wood Products	50	
		4.1.3 Wood-based Products	51	
	4.2	Structural Components	52	
		4.2.1 Beams	52	
		4.2.2 Floors	53	
		4.2.3 Columns	55	
	4.3	Connection Types	56	
		4.3.1 Glued Joints	56	
		4.3.2 Carpentry Joints	56	
		4.3.3 Mechanical Joints	57	
		4.3.4 Comparison	58	
	4.4	Durability	59	
		4.4.1 Risks	59	
		4.4.2 Protective Measures	60	
	4.5	Pavement Surfacing	64	
		4.5.1 Function	64	
		4.5.2 Surfacing Systems With or Without Shear Connection	64	
		4.5.3 Types of pavement surfaces	66	
		4.5.4 Modular and Temporary Structure	67	
		4.5.5 Choice of Pavement System	68	
	4.6	Timber Car Parks	68	
		4.6.1 Timber-built Car Park Studen, Switzerland	68	
		4.6.2 Construction System for Multi-storey Car Parks in BauBuche	71	
	4.7	Concluding remarks	74	
PA	RT I	III: Concept Development	76	
F	Dof	inition Concept Development	77	
5	5 1	Goals for Concept Development	77	
	5.1		····· / / 77	
		5.1.1 vialli Gual	/ / 70	
		5.1.2 Specific Sub-yoars	/ Ø 70	
	50	General Assumptions Concept Development	<i>ور ،</i> 1 ي	
	J.Z	5.2.1 Car Park Design	۲ ں مو	
		U.Z.I DAIFAIN DESIYII	01	

		5.2.2 Car Park Circulation Layout			
	F 0	5.2.3 Building Materials			
	5.3				
		5.3.1 Deck System			
	E /	5.3.2 Framing System	83		
	5.4	Concluding Remarks	00		
6	Stru	ctural Starting Points Conceptual Design	87		
	6.1	General	87		
	6.2	Applied Structural Rules and Norms	87		
		6.2.1 Consequence Class, Service Class, Design Life & Load Factors	87		
		6.2.2 Load Combinations			
	~ ~	6.2.3 Function of Building & Combination Factors			
	6.3				
		6.3.1 Permanent Loads			
	6.4	6.3.2 Variable Loads			
	0.4	Deformations			
		6.4.1 Vertical Deformations			
	e e	0.4.2 Horizontal Displacements			
	0.0				
	0.0	Preise Devenuetore			
	0.7	C 3.4 Eved Decemptore			
		6.7.1 Fixed Parameters			
		6.7.2 Variable Parameters			
7	Modular or Timber Car Park Concepts				
	7.1	Description of Concepts	93		
		7.1.1 ModuPark			
		7.1.2 Car Park Koopman int	94		
		7.1.3 Car Park Morspoort	94		
		7.1.4 BauBuche Concept			
	7.0	7.1.5 B&O-Holzparkhaus			
	7.2				
		7.2.1 Total Deck Height			
		7.2.2 Weight per Level			
	7.3	7.2.3 Load from Standard Column on Foundation			
•	-		404		
ð	FOU		101		
	8.1	Soil Composition			
	8.2	Shallow Foundation			
		8.2.1 Assumptions			
	0.0	8.2.2 Load Carrying Capacity			
	8.3				
		8.3.1 Types of Pile Foundations			
		8.3.2 Assumptions			
	0.4	8.3.3 Load Carrying Capacity			
	ŏ.4	Conclusion			
PA	RT I	V: Variant Analysis	110		
Q	Dec	k System	111		
5		Long open Drefeb Deek System	111		

Deek Oys		
9.1 Long	g-span Prefab Deck System	111
9.1.1	General	111

		9.1.6	LVL Open Rib Panel			
		9.1.7	LVL Semi-open Rib Panel			
		9.1.8	LVL Closed Rib Panel			
		9.1.9	Comparison between Decks			
	9.2	Short-	span Prefab Deck System			
		9.2.1	General			
		9.2.2	CLT Deck			
		9.2.3	LVL Deck			
		9.2.4	CLT vs LVL Deck Comparison			
	9.3	Concl	uding Remarks	144		
		9.3.1	Long-span deck system			
		9.3.2	Short-span deck system			
		9.3.3	Hypotheses Evaluation			
10	Frar	ning S	ystem	147		
	10.1	Desig	n subvariants	147		
		10.1.1	Variant 1			
		10.1.2	2 Variant 2a			
		10.1.3	3 Variant 2b			
		10.1.4	Variant 3			
	10.2	Scher	natisation Structural System			
		10.2.1	Variant 1			
		10.2.2	2 Variant 2a			
		10.2.3	3 Variant 2b			
		10.2.4	Variant 3			
	10.3	Workf	low			
		10.3.1	Parametric Model			
		10.3.2	2 Structural Analysis			
		10.3.3	3 Data Processing			
	10.4	Resul	ts Analysis			
		10.4.1	Total Deck Height			
		10.4.2	? Weight per Level			
		10.4.3	3 Load on Foundation			
		10.4.4	Effect of Alternative Assessment Criteria			
	10.5	Concl	uding Remarks			
		10.5.1	Total Deck Height			
		10.5.2	2 Weight per Level			
	10.5.3 Load on Foundation					
		10.5.4	Choice of Framing System			
PA	RT \	/: Eva	aluation and Conclusions	168		
11	Cas	e Stud	у	169		
	11.1 Reference Projects					
	11 2	Comp	arison	169		
	2	11 2 1	Total Deck Height	170		
		11 2 2	2 Weight per Level	170		
		11.2.3	Load on Foundation	171		

12	Conclusion	175
	12.1 Objective & Main Research Question	
	12.2 Proof of Concept	
	12.3 Variant Study	
	12.3.1 Parking Deck Span	
	12.3.2 Struts	
	12.3.3 Transverse Column Distance	
	12.3.4 Distance between Joists	
	12.3.5 Alternative Assessment Criteria	
	12.4 Case Study	
	12.5 Final Word	177
13	Discussion & Recommendations	179
	13.1 Discussion of Results Literature Study	
	13.1.1 Car Park Design	
	13.1.2 Building Sustainable	
	13.1.3 Building in Timber	
	13.1.4 Suggestions for Future Research	
	13.2 Discussion of Results Design Variants for Proof of Concept	
	13.2.1 Fire-safety	
	13.2.2 Vibrations	
	13.2.3 Deformations	
	13.2.4 Use of Struts	
	13.2.5 Suggestions for Future Research	
	13.3 Discussion of Results Comparison with Alternative Concepts	
	13.3.1 Structural Comparison	
	13.3.2 Sustainability Assessment	
	13.3.3 Suggestions for Future Research	
Bib	liography	185
Ар	pendices	194
Α.	Structural Loads and Limit States	195
	A.1 Load Combinations (ULS/SLS)	
	A.2 Variable Loads	

	A.3 Deflections and Displacements	
В.	Car Park Circulation Layouts	199
	B.1 Flat Decks	
	B.2 Split-level Decks	
	B.3 Sloped Parking Decks	
C.	Timber Joints	207
	C.1 Detailed Overview of Timber Joints	
	C.1.1 Carpentry Joints	
	C.1.2 Mechanical Joints	
	C.2 Comparison of Timber Joints	211
	C.2.1 Glued Joints	
	C.2.2 Carpentry Joints	
	C.2.3 Mechanical Joints	
D.	Calculations Technosoft	217
	D.1 Cone Penetration Tests	217

	D.2 Output Technosoft Shallow Foundation	
	D.3 Output Technosoft Pile Foundation	
E.	Comparison Modular or Timber Car Park Concepts	313
F.	Additional Calculations & Results Deck System	317
	F.1 Long-span Deck Systems	
	F.1.1 Example Calculations	
	F.1.2 Results Minimum Dimensions Long-span Deck Systems	
	F.2 Short-span Deck Systems	
	F.2.1 Example Calculations	
	F.2.2 Results Minimum Dimensions Short-span Deck Systems	
	F.3 Choice of Deck System	410
G.	Additional Calculations & Results Framing System	413
н.	Parametric Model	419
	H.1 General Overview	
	H.2 Inputs	
	H.3 Geometry	
	H.4 Setup of XML File for Structural Analysis in SCIA	
	H.5 Creation of .xml File	

Abstract

Problem definition: In the car park industry, a new trend has emerged, shifting the demand for parking from permanent one-of-a-kind parking structures towards temporary modular structures (Drenth, 2020). Meanwhile, the building industry faces a transition into a more sustainable industry. This requires rethinking of the design process, taking into account possible reuse of a building or its elements as well as potential use of different materials. Timber is a renewable material and as a result of developments in timber engineering, now allows the construction of taller and larger structures. Given these trends and challenges, the objective of this thesis is to provide a structural system proof of concept for a temporary multi-storey car park, using timber as primary structural material.

Approach: A broad literature study provides background information on the various relevant topics and is followed by a refinement of the objective, substantiated by two sub-studies. This has resulted in two sub-goals: the minimization of total deck height (< 700 mm) and minimization of the structural systems weight (column load < 1,000 kN). A proof of concept is developed through a parametric study with structural validation into four main design variants (consisting of a deck and framing system), for which the effect of altering four geometric parameters (parking deck span, use of struts, distance between columns in transverse direction and minimum distance between joists) on these sub-goals is studied. Furthermore, a sensitivity analysis is performed on the effect of altering the requirements concerning fire-safety, vibrations and deflections on the deck height and weight. Finally, the proof of concept's performance is compared to five alternative modular and/or timber car park concepts in a case study.

Evaluation: It is concluded, that design variants using long-span deck systems result in a relatively small deck height, but high self-weight. A CLT rib deck should be applied under "standard" assessment criteria, but significant reductions in deck height (31%) and weight (49%) can be achieved by using LVL rib decks when vibrations are not considered and deformation limits are increased. The total deck height of design variants with short-span decks, which use solid LVL panels, highly depends on the supporting framing system, but its self-weight is roughly halve of the long-span design variants. The effect of alternative assessment criteria on design variants with short-span decks is rather limited. It is concluded that the use of struts significantly reduces the total deck height (17-30%) and structures weight (5-11%) and a transverse column distance of a single parking bay width is preferred. The distance between joists is of less relevance, but a bigger parking deck span increases both deck height and total weight.

The resulting proof of concept consists out of a short-span solid LVL deck system, which spans two times 2.5 m (single parking bay width). It is supported by glulam main girders, spanning the entire parking deck, which are simply supported by a set of columns at a transverse distance of 2.5 m and are additionally supported by slanted struts between the columns and main girders.

The results from the case study show, that the total deck height of the proof of concept (666 mm) is smaller that the sub-goal of 700 mm and comparable to alternatives with decks made of steel, concrete and GFRP. Furthermore it is 40% thinner than the alternative in timber as a result of the use of LVL and struts. In terms of weight, the use of timber is much lighter than using concrete decks (<25%) and only 17% heavier than the extremely lightweight GFRP decks. Furthermore, it results in the smallest loads on the foundation (678 kN which is smaller than the sub-goal of 1,000 kN). A first extremely simplified durability analysis on amounts of embodied carbon in the main structural elements indicates timber has significantly lower amounts of embodied carbon in comparison to the alternative concepts.

Implications: Based on the results presented, it is concluded that timber is applicable for almost all main structural elements of a temporary and modular multi-storey car park. The case study has shown that the proof of concept can compete with alternative modular car park concepts in terms of deck height and outperforms most alternatives in terms of the structures weight. It is also indicated, that timber has the potential to significantly improve the performance of a structure in terms of sustainability.

For future research, it is advised to further develop the proof of concept, focussing on the connections and durability risks and study the incorporation of circular design strategies into the design and construction process of the proof of concept. Furthermore, alternative assessment criteria should be validated. Finally, it is advised to study the potential use of other innovative sustainable building materials like BauBuche and make a detailed LCA study assessing the performance on sustainability.

PART I: Introduction

Introduction

1.1 Background and Relevance

Nowadays, a car park may no longer be seen as a permanent structure, but rather as a construction to facilitate in a temporary demand for parking. This trend has various origins. For example, when large building complexes (like hospitals) are expanded, resulting in less remaining area for parking or temporary demand for alternative parking, a temporary multi-storey car park is required during the construction process (Bekkering, 2017). Similarly, large redevelopment projects in cities can result in a decrease in parking space for multiple years, which can be solved through a temporary car park (E. Jansen, 2013; Mobiliteits Platform, 2013). Also uncertainty in the future need for parking area may result in in a demand for car parks, which are fast and easy to construct and can easily be expanded, scaled down or removed (Drenth, 2020). According to McKinsey (Möller, Padhy, Pinner, & Tschiesner, 2019), four main trends will influence the car industry and this uncertainty: autonomous driving, connectivity, electrification and shared mobility. These trends will have an influence on the role and location of car parks. For example mobility hubs and P+R facilities at the edge of cities are expected, shifting demand for parking area from city centres to its edges (Dekfordt, 2019). All of these trends showcase an increase in future demand of temporary car parks.

A temporary construction has multiple advantages, like shorter processing times for building permits (Drenth, 2020) and less strict building requirements. As a result, a simpler and cheaper construction can be built. However, it can be a waste of resources to demolish a construction already after a short period of time, especially if these materials can no longer be reused or recycling requires a lot of extra steps (Kralj, 2008). Therefore, in the design process of a temporary car park, it is desirable to design for the long term and think about the demountability and possible reuse of the structure.

Thinking about the end of life phase of a building in the design process becomes an important theme as the building industry has to make a transition towards a more sustainable sector. The building sector was responsible for 35% of the global final energy consumption in 2019 and 38% of CO_2 emissions (United Nations Environment Programme, 2020) and therefore has to take action. This transition towards sustainability requires rethinking on how and which building materials are used, how buildings are designed, the construction and fabrication process and the end of life of a building.

Recently, the building industry has shown a lot of interest towards the use of timber as a structural material (Roberts, 2020). There are several reasons to use timber in constructions. First of all, timber has been around in the building industry for ages (Woods, 2016). In 2004, Eurocode 1995: Design of timber structures was released in Europe, which has been updated multiple times since its release, broadening the potential to use timber in larger and higher structures (Nederlands Normalisatie Instituut, 2011). Timber is a renewable material and can be carbon positive. Furthermore it performs well in fires and earthquakes, is aesthetically appealing and allows faster construction with lower labour costs, producing less waste (Wood for Good, 2011). This makes it a material with a high potential.

The fabrication and construction process of the industry is changing. The focus is increasingly turning to aspects such as building speed, the use of prefabrication and implementation of digital tools (Box, 2020). Prefabrication can lead to better working conditions, higher quality products and faster construction on the building site, while producing less waste and using less energy, but resulting in higher productivity (Bertram et al., 2019). Digital tools can help optimize and automate this process.

1.2 Research Definition

1.2.1 Problem Statement

To provide in the quest for a temporary car park designed for reuse, different mostly steel, modular and demountable structural systems with concrete decks have been developed (Ballast Nedam Parking, 2011; Park4all, 2019a). These concepts are designed using a new fabrication and construction process. The result is a car park, which can be dismantled and reused and therefore make full use of the maximum lifespan of its construction elements, which make the structure a more sustainable option. The modularity of the concepts results in larger amounts of prefabrication and shorter building times. Concepts combining steel and concrete in their structural systems have therefore proved to be able to provide a solution to the temporary character of a car park, while also embracing the changes in the fabrication and construction process of the building industry. However, concerning the challenges related to sustainability faced by the industry, it might be possible to further reduce the use of energy and production of emissions. Question is whether timber, considering all of its potential advantages, can fulfil the role of structural material in a car park with a temporary character and help further reduce the footprint of the structure. This results in the following problem statement:

The temporary character of car parks results in only a short required lifespan of its building elements, resulting in a waste of resources. Meanwhile, the building industry faces a sustainability transition, challenging it to rethink which and how building materials are used and how the production and construction process is organized. Question is if and how timber, considering its high potential, can fulfil a roll in both challenges as a structural material.

1.2.2 Objective

The aim of this thesis is to make a preliminary design of a structural system for a temporary multi-storey car park, incorporating the challenges faced by the building industry. Therefore, the technical feasibility of the use of timber as a primary structural material in the design is checked as well as how changing production processes influence the design of building elements. This results in the following objective:

Provide a structural system proof of concept for a temporary multi-storey car park, using timber as primary structural material.

1.2.3 Research Questions

Following the objective, the main research question is defined as:

"To what extent can timber be applied as a structural material in the structural system of a temporary multi-storey car park and how does it compare to alternative concepts?"

The main research question is supported by the following four sub-questions:

- 1) How to design a structural system for a temporary car park using timber as primary structural material?
 - a. What are the design requirements for a multi-storey car park?
 - b. What are the possible typologies for a multi-storey car park?
 - c. How to construct using timber as a structural material?
- 2) How do the challenges faced by the construction industry towards a sustainable industry influence the design of a temporary multi-storey car park?
 - a. Which sustainability building strategies can be applied on the design of a temporary multi-storey car park?
 - b. What is the impact of sustainability building strategies on the design of building elements?

- 3) How do different typologies for a structural system for a temporary multi-storey car park using timber as primary structural material structurally perform?
 - a. What characteristics of a temporary multi-storey car park using timber as primary structural material are relevant to assess its structural performance?
 - b. What level of structural performance is requested from the design for a temporary multi-storey car park?
 - c. What is the impact of altering geometric parameters and limit state requirements on the performance of the different designs for a temporary multi-storey car park using timber as primary structural material?
- 4) How does the proof of concept for a temporary multi-storey car park, using timber as primary structural material compare to alternative modular and/or timber multi-storey car park concepts?

1.2.4 Scope

This research is performed as a proof of concept and therefore, only the preliminary design phase is considered. To be able to answer the research questions within the timeframe of a master thesis, the following scope is defined:

Materials: Timber is considered as the primary building material, but other materials are not excluded. The main elements of the superstructure of the structural system Superstructure: of the multi-storey car park are considered. These include a deck system and a framing system including stability bracings. The substructure of the structural system is briefly considered in Foundation: terms of a range of desired maximum foundation loads. **Connections:** Connections are only considered in terms of the types of connections. They are not modelled into detail. Weather conditions: Effects of temperature, moisture and protection are only considered in terms of their effect on the structure and required protective measures. The effects are not modelled into detail. Possible sustainability strategies are considered. A brief Sustainability: sustainability performance assessment is made, with the goal to show the potential of the proof of concept. No further detailed LCA analysis is conducted. Façades: Only the loads coming from the façades and acting on the structural system are considered. No façade design is made. Finishing: Items like installations, plumbing and electrical equipment are not considered. **Regulations:** The European norms and regulations (Eurocodes) are applied. Fire safety of the car park is only considered briefly. General Fire safety: effects of fire on the deck system are taken into account, however detailed modelling of the effects of fire on the structure and required fire suppression systems are not considered. Vibrations: General requirements for vibrations on structures are taken into account for the deck system of the car park. However a detailed analysis of what specific requirements would be optimal concerning a car park is not made. Vibrational behaviour as a result of earthquakes is not considered.

1.2.5 Methodology

1.2.5.1 Research Design and Process

The methodology used for this research is based on Hall's (1968) methodology for systems engineering. To help describe the methodology, the different phases of Hall's design cycle are projected over the different parts of the report. A visual overview including the methods used can be seen in Figure 1.



Figure 1. Design Methodology

1. Problem definition

This master thesis started with the problem definition. Through exploratory literature research and discussions with engineers, the problem context of temporary car park structures and building in timber was studied. This helped to understand the challenges the parking industry has to deal with and the challenges the entire building industry faces. As a result, a first problem statement was developed, which acted as the basis for this master thesis. Furthermore, a main research question and first set of sub-questions was developed.

2. Value systems design

In the value systems design phase, the question at heart was what the objective is and what the criteria are that a solution has to fulfil. This objective followed from the problem statement defined in the first design phase and was to provide a structural system proof of concept for a temporary multi-storey car park, using timber as primary structural material. In order to come up with the criteria the solution has to fulfil, a rather broad but in-depth literature study was performed into the requirements for such a proof of concept. Based on the exploratory research and objective, three topics were pointed out as being highly relevant to develop a successful proof of concept for the objective. The first relevant topic was related to the general design of car parks, considering aspects like regulations, comfort and economics. A second topic was related to the challenges faced by the building industry to become a sustainable industry, which included aspects like what sustainability strategies exist and how such strategies should be incorporated in the design of a construction. The third topic was related to the field of timber engineering. This includes aspects like how to build a timber structure in terms of different components and connections, but also how to protect it from the durability risks that a timber structure comes with.

The in-depth literature study resulted in a broad overview of relevant aspects for the development of a temporary multi-storey car park primarily in timber. Next, focus needed to be applied to this broad set of aspects, considering what aspects would be further investigated and what not, since it is not possible to consider all aspects in a master thesis project. This was done in the concept development part, which includes two small sub-studies into existing modular and/or timber car parks and possible foundation methods, to further specify what is expected from the proof of concept to deem it successful. As a result, the research questions were further refined and a more specific set of requirements for the proof of concept was presented.

3. Systems synthesis

The systems synthesis is all about generating alternatives. In the literature study, an overview of different typologies for car parks was presented as well as an overview of potential application of different building materials. In the concept development part, a selection of these typologies and application of building materials was made, resulting in an overview of design variants that were further studied. Through the use of parametric modelling, these design variants were further divided into subvariants with slight alterations in geometry, to see if different geometries can result in a more optimal solution for the proof of concept.

4. Systems analysis

In the systems analysis phase, the different design (sub)variants, developed through parametric modelling, were first structurally analysed and optimized to result in a set of structurally validated designs. Next, the performance of these designs on the objectives and criteria stated in the value system design phase was assessed. This allowed for comparison between the different designs on these aspects, to be able to select a best performing design subvariant for the proof of concept.

5. Selecting the best system

Through a case-study, the developed structural system was compared to a selection of five alternative concepts for a modular and/or timber multi-storey car park. The concepts were assessed on their total deck height, weight of the structural system (including load on the foundation) and a first impression was given for their performance in terms of sustainability. Through this case study, one can conclude how the performance of the proof of concept relates to alternative concepts and therefore if it is feasible. This phase therefore ends in a conclusion and discussion on the performance of the developed structural system using timber as primary structural material, answering the main research question.

6. Planning for action

The last phase consists out of the recommendations for future use of timber as a structural material in temporary multi-storey car parks.

1.2.5.2 Data Collection

Exploratory research

Data for the exploratory research was collected through literature research and discussions with engineers. For the exploratory literature research, journals, reviews, news articles and websites were used. This allowed for the author to quickly create an understanding of the relevant topics. Furthermore, discussions with engineers in the field of general structural engineering, timber engineering, digitalization and car park design helped to further understand where challenges are currently found in the various fields of car park engineering. For this exploratory research, a broad scope was applied, looking for information on car park construction and building in timber. On most aspects, a wide variety of literature was available, however the number of studies on car parks build in timber was very limited. Nevertheless, the data found presented a broad overview of relevant information and allowed to create a clear objective.

In-depth literature research

For the in-depth literature study, a lot of information found in the exploratory research could be further investigated. In this phase, the focus was more on scientific journals, handbooks and industry norms. This allowed to create a deeper understanding of the various relevant aspects and find research gaps. As already found in the exploratory research, studies into timber car parks were very limited, but it was found that even less research was conducted into modular and demountable timber car parks.

Concept development

In the concept development part, five different car parks were studied. These car parks were chosen based on the concept of their structural system. These were either modular (and demountable) or (partially) made from timber. These concepts were compared to each other to see how existing timber car parks perform relative to alternative modular car park concepts in different materials. Data on these structures was found from different sources: scientific journals, municipal archives, direct contact with

architects and in-field measurements. Not for every potential interesting car park (the same amount of) data was available. The final selection of car park systems considered, all had sufficient data available for comparison.

Also possible foundation methods were studied and what loads can be carried by different soil buildups in The Netherlands. Information on the various possible methods was found from literature. For the analysis of load carrying capacities of different soils, DINOloket was used to find different soil patterns accompanied by cone penetration tests. Load carrying capacities were calculated based on engineering calculation using Technosoft foundations software.

Case study

The case study included the same car park structures that were studied in the concept development. For all these car parks, the general build-up of the structure and dimensions of main structural elements were available through either architectural drawings or in-field measurements.

1.2.5.3 Data Analysis Methods

Literature

The literature found in the exploratory and in-depth literature studies was grouped into different (sub)categories. The main categories were: car park design, sustainable construction and timber engineering. Furthermore, each item was given a rating and short description. This clustering of literature helped to quickly find relevant information during the different phases of the thesis. The ordering of sources was done using EndNote.

Parametric modelling

Creation of geometry for the design (sub)variants was done through parametric modelling in Rhinoceros in combination with Grasshopper. These software packages allow for fast creation of a large design space for geometry. Furthermore, such a parametric model can easily be expanded in case new insights are requested for alterations on the developed geometry. A parametric model in Grasshopper also allows for coupling with various structural analysis plugins or software packages for further analysis of the developed structure.

Structural validation

Structural analysis and validation was done using the software package SCIA. This package was chosen as it is widely used in the engineering industry and as primary structural analysis package at Sweco. Furthermore, a plugin to connect the software package to Grasshopper is available. Although it was expected this would allow for a fully automated process from generation of geometry to structural analysis, validation and optimization, this turned out not to be the case. Such a fully automated design process requires further development of the software package and plugin for SCIA.

Comparison

Comparison of the performance of the various car par design alternatives as well as different car park concepts in the case study was done through Excel. Python was used to automate creation of the Excel file for comparison based on inputs from the structural analysis and study of structural elements of alternative car park concepts.

1.3 Reading Guide

Following the structure of the methodology, this paragraph describes the structure of the thesis report, see Figure 2. It globally shows what is discussed in the various parts of the report and what research questions are answered in what chapter.

Reading Guide				
Report Part	Chapter	Research question answered	Description	
Part I: Introduction	Ch. 1		Introduction including background of topic and research definition	
Part II: Literature Study	Ch. 2 Ch. 3 Ch. 4	1a 1b 2a 2b 1c	Broad literature study including background information on car park design, the challenges faced by the building industry towards a more sustainable industry and building in timber.	
Part III: Concept Development	Ch. 5 Ch. 6 Ch. 7 Ch. 8	3a 3b	Development of specification of what is expected from the proof-of-concept in terms of perfomance and what design variants are considered.	
Part IV: Variant Analysis	Ch. 9 Ch. 10	3c	Detailed analysis of structural performance of deck and framing system of alternative design variants.	
Part V: Evaluation and Conclusions	Ch. 11 Ch. 12 Ch. 13	4 Main research question	Case-study with comparison of proof-of-concept and alternative modular and/or timber car park concepts followed by conclusion, discussion and recommendations.	

Figure 2. Overview of the structure of the thesis

Part I: Introduction

Chapter 1 forms the introduction of this thesis, describing the context of the problem and presenting the problem statement. Furthermore it presents the research definition, including objective, research questions, scope and methodology.

Part II: Literature Study

The second part of the thesis is the literature research. This literature study is rather broad and describes many different aspects related to the design of a temporary and modular multi-storey car park. It functions as background information for the development of the multi-storey car park. Chapter 2 focusses on the design of car parks and answers sub questions 1a (design requirements) and 1b (typologies). Chapter 3 is all about the challenges in the building industry, looking into sustainable building strategies and its impact on building elements, answering sub questions 2a and 2b. Literature about timber structures is researched in chapter 4, looking into different timber products, connections and durability and answers sub question 1c.

Part III: Concept Development

In part three of the thesis, focus is applied on what specific aspects found in the literature study are considered in the development of the proof of concept and what is expected from the resulting proof of concept. Chapter 5 specifies the main goal and specific sub-goals for the proof of concept, resulting in an answer to sub question 3a. Furthermore, hypotheses are developed for the expected results. Finally, the different design variants are presented, which will be studied in the variant analysis part. In chapter 6, it is defined what are the starting points for the development of the proof of concept concerning structural rules and norms. A first case-study is done in chapter 7. Five alternative concepts for a modular and/or timber multi-storey car park are studied, based on which performance criteria for the proof of concept are developed. This partially answers sub question 3b. Chapter 8 focusses on possible foundation methods and further answers sub question 3b considering performance criteria related to loads on the foundation.

Part IV: Variant Analysis

In part four, the different design (sub)variants are studied and compared to each other. Chapter 9 focusses on the possible deck systems for the structural systems and chapter 10 focusses on the

framing system. These chapters include a parametric study into the use of different geometric parameters and assessment criteria and what effect this has on the performance criteria stated in part three. These chapters result in an answer to sub question 3c.

Part V: Evaluation and Conclusions

Part 4 forms the last part of the thesis. First, in chapter 11, a case-study is conducted, comparing the earlier mentioned five alternative car park concepts with the developed proof of concept, which answers sub question 4. Next, in chapter 12 the final conclusions of this thesis are drawn, answering the main research question. Finally, chapter 13 forms the discussion of this thesis and includes recommendations for future research.

PART II: Literature Study

2 Car Park Design

The design of a multi-storey car park involves integration of many different aspects. First of all, the design has to fulfil the requirements of the owner and users of the car park. These are largely reflected by specific car park design norms. Furthermore, the design has to fulfil the general rules and norms for the structures of buildings. Bearing these prerequisites in mind, different layouts for car parks and different typologies for its construction can be distinguished. These influence each other and goal is to find the most optimal design, which fulfils the clients requirements. This chapter aims to provide an answer through in-depth literature research to what the design requirements are for a multi-storey car park and what the possible typologies are.

2.1 User Requirements

Users of car parks are most concerned with functional requirements. The ease of finding a parking space and doing so without risk at damage to their vehicle is important to them (Van Drooge, 2020). For users it doesn't matter if a car park is temporary. They expect the same level of performance of a temporary car park as from a long term car park. Several characteristics, influencing the structural design of car parks, can be pointed out as being essential for a good user experience: car park category, dimensions of lanes and ramps, dimensions of bays and aisles, circulation through the car park and other considerations like drainage and security. Many of these user requirements are represented by the norm NEN 2443 - design standards and recommendations on parking facilities for passenger cars. This paragraph presents background information on what aspects concerned with user requirements should be considered. Specific dimensions and other recommendations from the parking facilities norm are discussed in chapter 2.4.

2.1.1 Car Park Category

Car parks may suit different types of users, which mostly depends on the location of the car park. One can understand, that supermarket customers have different expectations for a car park compared to business people, who park their car for an entire day. According to Hill (2005), four different types of car park categories can be distinguished: short stay, medium stay, long stay and tidal.

The first category is short stay and is described by intensive usage with high turnaround rates. The average duration of stay per car for short stay is less than 3 hours. This type of car park is usually associated with busy supermarket-type shopping activities.

The second category is medium stay and has less intensive usage and a lower turnaround rate compared to short stay car parks, but still is used quite intensive. The average duration of stay per car is more than 3 hours. This type of car park is mostly used in urban areas, where the car park is used by both business related users as well as visitors who come for shopping.

The third category is long stay, described by an average stay per car of over 12 hours and a light but continuous in- and outflow of vehicles. This category car park is often found at major transport terminals, which can occasionally result in short periods of intensive vehicle movement.

The fourth and last category is tidal and is characterized by its peak moments of in- and outflow. This type of car park serves mostly staff, resulting in a high inflow in the morning and outflow in the evening.

2.1.2 Dimensions of Lanes and Ramps

The dimensions of lanes and ramps are concerned with users of a car park being able to drive through the car park comfortably without the risk of easily incurring damage. Based on the standard design vehicle, minimum widths and radii for corners are known (see chapter 2.4.1), however these minimum values don't give users a comfortable feeling. In turns, users don't like to have full lock when turning. Therefore, it is recommended to provide wider turns at around 150% of the turning circle of the standard design vehicle for corners up to 90° and 200% for corners where 360° turns are anticipated. In general, it is preferred by users to have no turns bigger than 90°, so no spiral ramps and all turns in the same direction to prevent wild steering movements. (Hill et al., 2005)

When turning onto a ramp or coming off a ramp, wide turns are even more important. Also, the slope of the ramp has a big impact on comfort for users. Having to stop on a ramp and drive off again, is difficult if the slope of the ramp is too large. Slopes of ramps are therefore limited for comfort. For steeper ramps, transitional slopes are introduced to prevent grounding of vehicles or hitting the ceiling. When driving in a car park and especially on ramps, clear sight lines are important to prevent accidents between vehicles crossing paths (Van Drooge, 2020). Practical guidelines for the maximum slope percentages of ramps and clearance heights can be found in chapter 2.4.3.

Ramps influence the dynamic capacity of car park, which differs per type of ramp. It is therefore important to consider the required capacity of a car park, which type and how many ramps to use. For short and medium stay car parks, two-way-flow on ramps are not preferred as they result in conflicts between traffic flows (Hill et al., 2005). In other types of car parks, they can be very efficient in boosting static efficiency of a car park.

2.1.3 Dimensions of Bays and Aisles

Dimensions of bays and aisles are concerned with the ease drivers experience in turning into a parking bay and how much room is left for entering and exiting the vehicle by opening the doors. Considering the dimensions of bays, the length should be sufficient for at least the standard design vehicle to fit in the bay (see Figure 3), but the width is more important. Varying the width of the bay can have a significant impact on both the dynamic capacity of the car park as well as the ease of stepping in and out of a vehicle. A wider bay means it is easier to turn into. Considering the gap between adjacent vehicles, the room to get in and out of the car, a small increase in width can have a significant impact on how far a door can open and so how easy it is to get in and out of the vehicle (Hill et al., 2005). Chapter 2.4.2 presents clear guidelines for the dimensions of bays and aisles.

When two vehicles are parked next to each other, the space between both vehicles can be used for getting in and out of the vehicle, while if you are parked next to a wall or column, only the space remaining in the bay of one vehicle is available (see Figure 3). Therefore, it is preferred not to have walls or columns located next to the doors of vehicles. Furthermore, it is preferred not to have columns located at the front of bays as they hinder vehicles from turning into the bay. The best option is to have long column free spans over the parking decks (Van Drooge, 2020).



Figure 3. Examples of problematic dimensions of parking bays and location of columns (Van Drooge, 2020)

Aisles provide space for vehicles to manoeuvre into and out of bays and often for pedestrians to walk through the car park. When two-way-flow aisles are used, they have to be wide enough for two vehicles to pass each other, resulting in enough space to manoeuvre in and out of bays and provide space for pedestrians. If one-way-aisles are used, the aisles can be narrower, which can make the span over the parking deck smaller. Using angled bays can reduce the width even more, at the same time making parking easier and simpler (The Institution of Structural Engineers, 2011). This however leaves less room for pedestrians which should be kept in mind.

Last of all, parking decks can be designed as a slope, which can be an efficient design. This slope however has some disadvantages concerning comfort for users. Shopping cars and wheelchairs can drift off and if the slope becomes too steep, it becomes more difficult to open and close doors as a result of gravity (The Institution of Structural Engineers, 2011).

2.1.4 Circulation

The routing in a car park is very important for users. A clear and logical routing with good sight lines and a minimum number of conflict points, supported with signs, helps users in finding their way in the car park (Van Drooge, 2020). The circulation pattern can also have a significant impact on the dynamic capacity of a car park. Therefore, depending on the type of car park, a suitable circulation pattern should be chosen (Hill et al., 2005).

2.1.5 Miscellaneous User Experiences

As stated earlier, clear sight lines in car parks are important for safety while driving. Users like to have a feeling of security in general (Troup & Cross, 2003). The structure can have a significant impact on this feeling as an open and light structure often gives the feeling of an area being safe, while a closed and dark structure with large walls blocking sight lines gives a feeling of unsafety.

Another general aspect for users of car parks is drainage. Large puddles of water may not only present a risk to the structure, but also hinder stepping in and out of vehicles and when located in the aisles, people can get splashed by vehicles driving by. Therefore, it is required to minimize the possibilities of forming of puddles by proper drainage (Hill et al., 2005).

2.2 Owner Requirements

Owners or developers of car parks, the clients of car park designers, have to make trade-offs between many decisions. They want to provide their customers with the best high level car park, but at the same time keep the costs within budget. Also aspects like sustainability play a role in choices for a structural design. Apart from cost-efficiency and sustainability, design flexibility, the construction process and durability and maintenance have been identified as important aspects for the owner of a car park.

2.2.1 General

Car parks can be developed and owned by different types of clients. It can be a public owner like a municipality or hospital, a public company, a private client or a project developer. Each client has its own goals and needs. A public owner for example might have less flexibility and budget, while a project developer might want to achieve the highest quality (Troup & Cross, 2003).

Regardless of the type of client, for the planning of a car park, several general requirements have to be stated. It should be determined what the capacity of the car park should be and what type of category car park it should be or what level of service should be provided (Hill et al., 2005; Troup & Cross, 2003). These aspects will influence the choice for a certain layout of the car park. Other aspects are the intended location of the car park and its maximum dimensions and number of levels.

2.2.2 Cost-efficiency

For clients, several characteristics of a car park can give a general idea about its performance. These characteristics are concerned with how efficiently the structure is laid out in terms of average area per parking place in m² (Pollak, 2003) and the average costs per parking space (Cross, 2002). These numbers are important as the costs of the construction itself are normally 60 to 70% of the total construction costs (Pollak, 2003). The static efficiency, average area per parking space, generally varies between 20 and 30 m² per car (Hill et al., 2005). For planning purposes, often 28 to 30 m² is used (Simon, 2001). The construction costs for a car park above ground can vary significantly, depending on their construction and design. On average in the US in 2019, according to Cudney & Smith (2019), the construction costs for a new parking structure were \$21,500 per space. Walker (2016) did a similar study in the US in 2015 and found construction costs varying between \$15,000 and \$25,000 per space with an average of \$19,000.

2.2.3 Design Flexibility

In the design, one should take into account several factors. In the future, a client might want to react to a changing market. As a result, one might want to expand or reduce the parking facility (Simon, 2001). Another option would be to converse (part of) the building to another function. To provide in such flexibility requirements, the adaptation possibilities should be incorporated into the design of the original structure.

Many car parks have a similar structural system, but differ significantly in appearance. This is the result of different types of façades. To provide in the possibility for different appearances as well as possible future changes in appearance, the structural system of a car park should provide opportunity for aesthetic expression (Troup & Cross, 2003).

2.2.4 Sustainability

As climate change is more evident than ever before, sustainability in construction projects becomes more important (White, Hardisty, & Habib, 2019). Users expect higher levels of sustainability in all types of services, of which car parks are one. As a result, also developers start to consider sustainability in their projects more often as an important aspect. Sustainability also effects the structural design. Important aspects to keep in mind are the choice for certain building materials; where do they come from, how are they produced and what can be done with them after the building is demolished (Troup & Cross, 2003). Also the construction process is important to keep in mind considering sustainability; transportation and production location of materials for example.

2.2.5 Construction Process

The construction process of a car park is an important step for a developer or owner. During construction, a lot of expenses are made to construct the building. When the car park becomes operational, the owner can start to make a return on its investment, so the sooner it is finished, the better. The speed of construction is therefore an important aspect for a car park and can be influenced by the chosen structural system (Pollak, 2003). A modular steel structure or prefab concrete structure for example is much faster to erect compared to a cast-in-situ construction. Also detailing of connections and the size of structural elements have a significant impact on erection times (Precast/Prestressed Concrete Institute, 1997). In the design, one should also take into account maximum dimensions and weights for transportation and make a trade-off between bigger structural elements for faster erection versus the higher costs as a result of the requirement for heavier equipment like cranes.

2.2.6 Durability and Maintenance

Durability and maintenance of a construction go hand in hand and are massively influenced by the choice for a certain structural system, details and building materials (Troup & Cross, 2003). It influences the life expectancy of the structure and can be a huge cost item (Precast/Prestressed Concrete Institute, 1997). An owner often prefers minimal required maintenance, to minimize downtime of the car park (Simon, 2010). As car parks are open to weather influences and de-icing salts are brought in by cars, protection of the structure by design is essential during development of the car park. After completion, regular inspection and maintenance is required (see Figure 4 for examples of damages to car park structures). It is important to give attention to these aspects during the design process as a good design can minimize maintenance and possible damage. For example a sufficient topping can provide a watertight membrane and by minimizing restraints to movements, tensile stresses which can cause cracking of floors and walls can be minimized (Cross, 2002). A cheaper variant in construction can often proof to be more expensive in the long term.



Figure 4. Damage to car park structures (British Parking Association, 2012; Ducorr, 2018)

2.3 Structural Rules and Norms

A multi-storey car park is a special type of building. It is no regular building like an office or residential complex with a thermal shell nor is it a civil structure for infrastructure like a bridge or tunnel. As a result, in The Netherlands, the Dutch building degree (Ministerie van Binnenlandse Zaken en Koninkrijkrelaties, 2021) and the European Eurocodes for the building sector (NEN, 2021) have to be applied. The Dutch parking norm is not required as it is not part of the building degree, but it consists of elaborate guidelines, compiled from practical examples, and is therefore used in this thesis.

- Dutch Building Degree 2012
- (Ministerie van Binnenlandse Zaken en Koninkrijkrelaties, 2021)
- NEN-EN 1990+NA: Eurocode 0, Basis of structural design
- NEN-EN 1991+NA: Eurocode 1, Actions on structures
- NEN-EN 1993+NA: Eurocode 3, Design of steel structures
- NEN-EN 1995+NA: Eurocode 5, Design of timber structures
- NEN-EN 1997+NA: Eurocode 7, Geotechnical design (NEN, 2021)
- NEN 2443: Design standards and recommendations on parking facilities for passenger cars.

(Nederlands Normalisatie Instituut, 2013)

The focus of this thesis is on the structural design of a car park and the use of timber as a structural material. Therefore, the different rules and norms are discussed only on aspects, which influence the structural design of the main load bearing structure. Other requirements fall out of the scope and are not discussed in this thesis. Relevant norms on the design and calculation for different building materials are also considered. Norms on the design and calculation for earthquakes are not considered.

2.3.1 Preconditions and Assumptions

For the design and calculation of the temporary multi-storey car park, several assumptions are made. Although the aim is to make a modular and demountable design, the structure is not classified as such. The design life of the structure will be similar to a normal fixed building. The only difference is that this structure during its design life will be erected at several places, but it has to withstand the same loads as it would when constructed at a single location.

Type of building:	Public multi-storey car park	
Consequence class:	CC2	
Reliability class:	RC2	k _F = 1.0
Design service life category:	Class 3	50 years

2.3.2 Loads and Limit States

The structure of a multi-storey car park should withstand several permanent and variable loads. These should be combined in the various limit states. An overview of the relevant equations for combination of loads in ultimate limit state (ULS) and serviceability limit state (SLS), including the combination factors can be found in appendix A. Also values for the maximum deflections and displacements are presented in this appendix.

2.3.2.1 Permanent Loads

Permanent loads acting on the structure are self-weight of both structural and non-structural elements. These loads are dependent on the design and building materials used and have to be determined during the design process. For a car park, among others one can think permanent loads following from self-weight of the following elements (Nederlands Normalisatie Instituut, 2019a):

Structural elements

- Floor elements
- Beams
- Columns
- Bracings

Non-structural elements

- Surfacing and coverings
- Partition walls
- Hand rails, safety barriers, parapets and kerbs
- Façades
- Fixed services

2.3.2.2 Variable Loads

Various variable loads can be present on the structure of a multi-storey car park. The imposed vehicle load, specifically concerned with a car park is discussed below. Accidental loads are also relevant for car parks, but for the preliminary design of this thesis are left out of scope. An overview of the other more general variable loads, acting on any building can be found in appendix A. These include snow and wind loads.

Imposed Loads on Garages

Considering the variable loads acting on the structure, a multi-storey car park is classified as category F "Traffic and parking areas for light vehicles (≤ 25 kN gross vehicle weight and ≤ 8 seats not including driver) (Nederlands Normalisatie Instituut, 2019a). A distributed and concentrated load are imposed, see Table 1. The concentrated load has to be applied as an axle load, following the dimensions in Figure 5, in which the square surfaces have sides of 100 mm.

Table 1. Imposed loads on garages and vehicle traffic areas. (Nederlands Normalisatie Instituut, 2019a)



Figure 5. Dimensions of axle load. (Nederlands Normalisatie Instituut, 2019a)

2.3.3 Vibrations

According to the Eurocode (2015), structures of buildings should provide satisfactory vibration behaviour. This behaviour relates to the comfort of the user and the functioning of the construction or its structural elements (no cracking etc.). To assess whether a structure meets the requirements for satisfactory vibration behaviour, the natural frequency of vibrations of the structure or structural members are used under serviceability limit state combinations. For timber floors in residential buildings, Eurocode 5 (Nederlands Normalisatie Instituut, 2011) prescribes additional checks in case the natural frequency of the floor lies beneath 8 Hz, to prevent resonance from persons walking on the floor.

In car parks, vibrations are induced by both people walking, but also by moving vehicles. However, for car parks, no advice is given in the Eurocode concerned with allowed vibrations. Since visitors only walk in a car park for a very short time, they are often not disturbed by floor vibrations (Zanon, 2019). When driving in their car, sufficient damping of vibrations is provided by the suspension of the car. Therefore, there are no formal requirements for car parks for vibrational behaviour. Nevertheless, vibrations in the structure can be annoying. Therefore, the vibrational behaviour for the floors is checked on a less strict design class (Class II, (Schirén & Swahn, 2019)), which requires a frequency criterion of 4.5 Hz and stiffness criterion of 0.500 mm.

2.3.4 Fire Safety

2.3.4.1 Predicting the Structural Behaviour

When assessing the resistance to fire of a load-bearing structure, in general the five steps in the chain of events in case of fire are followed. These are (Breunese & Maljaars, 2015):

1) Ignition This is the first step in the development of a fire. This step requires no explicit task from the engineer, since the assumption is that somewhere in the lifetime of a structure, it is exposed to a fire.

2)	Fire development	As the fire develops, the convection and radiation temperature rise. The speed at which these rise depends on the thermal load and layout of the structure.
3)	Thermal response	Based on the temperatures from the previous step and the thermal properties of the structural elements, the temperature of these structural elements can be determined; the thermal response.
4)	Mechanical properties	Mechanical properties of materials like the material strength and Young's modulus are dependent on the temperature. As the temperature of the structural elements is determined, the mechanical properties of the materials under these conditions can be determined.
5)	Mechanical response	Based on the layout of a structure, the loads under fire conditions and the found mechanical properties under fire, the mechanical response (strength and stability) of the structure can be determined.

To determine the behaviour of a structure under fire conditions, two types of methods can be used: the conventional approach and the fire safety engineering approach (FSE).

The conventional approach is the simplest and is in practice used for relatively standard structures. For such a structure, the mechanical response of each individual component is determined based on the use of a generalized fire curve for the fire load, see Figure 6. With increasing temperature over time, it is determined when a component fails. The time it takes for the first component to fail determines the failure of the entire structure in terms of fire resistance of the structure in minutes. This doesn't mean that in reality the entire structure will have failed, but from a regulatory point of view, this is the case. This method gives a gross approximation of reality in terms of both the approximation of the fire load and of the structural response. However, the method is simple to apply and used in most regulations (Breunese & Maljaars, 2015).





With the more complex FSE approach, a much more accurate temperature of the fire is determined using the natural fire safety concept (NFSC). Instead of using a generalized fire curve to determine the temperatures, characteristics of the fire compartment are considered like: type and amount of combustible materials, floor area and height of compartment, openings in compartment, thermal capacity of the boundary enclosure and active measures such as roof venting, automatic fire alarms and automatic fire extinguishing devices. With this approach, a temperature-time curve is determined for all compartments of a structure. This is done using either relatively simple zone models or for more accurate results with CFD models (Breunese & Maljaars, 2015).

2.3.4.2 Fire Development in a Car Park

The standard fire is based on the development of a fire in a general building (office/residential). Such a building contains a fire load which consists of all kinds of furniture made of different materials. After ignition of the fire, the growth phase starts, in which temperatures in the fire compartment start to rise. At this point, the fire is a local fire. After a certain time, a flashover takes place, resulting in a very fast

increase of temperature. This introduces the next phase of a fully developed fire, in which case all combustible materials in the fire compartment contribute to the fire as a result of the very high temperatures. The entire fire compartment is now on fire. After some time, when most of the combustible materials have been burned, the temperatures start to drop again as the fire comes to the decay phase (Breunese & Maljaars, 2015). The standard fire curve starts at the point of flashover and assumes a continuing growing fire. The decay phase is not represented by the standard fire curve.

A fire in a car park differs from the standard fire as the main fire load in a car park are the cars. A fire starts in a car and after some time, it will spread to an adjacent car. However, a car fire has the characteristic to be a fire with a relatively high rate of heat release, but for a relatively short amount of time (D. Jansen, 2010). As a result, at the moment of fire spread to the adjacent vehicle, the original car is already almost burned out (Van Herpen, 2014). Therefore, it is highly unlikely for a fire in a car park to grow into a fully developed compartment fire. The fire scenario is more that of a local moving fire (Jansze & Van Acker, 2014).

As a result of the different fire scenario, the fire curve of a fire in a car park differs from the standard fire curve. Compared to a standard fire, a car park fire is more intense in the beginning, but much shorter in time as it remains a local moving fire. Figure 7 shows the combined normative fire curve for a car park fire against the standard fire curve. As can be seen in the graph, until 35 minutes, the temperature of a fire is sometimes higher than the standard fire curve. However, after 35 minutes, the temperature is significantly lower compared to the standard fire curve. As a car park in general has to withstand a fire for 90 minutes, the standard fire curve can be seen as a very conservative approach (Goesten, 2022).



Figure 7. Local and global thermal load combined to a single normative car park fire curve (Dijkstra, 2014)

2.3.4.3 Electric Vehicles

In recent years, the number of electric vehicles on the roads has been on a fast rise. Such vehicles with their large batteries present new fire risks. Several studies have been conducted on the fire development of electric vehicles. It has been found that the maximum heat release rate (HRR) of an electric vehicle is comparable with a standard car with a combustion engine as can be seen in Figure 8. Also the evolution of HRR over time is comparable (Sun, Bisschop, Niu, & Huang, 2020). A difference between electric vehicles and cars with a combustion engine is that electric vehicle fires are more difficult to suppress. Their battery packs are difficult to access and they can re-ignite without continuous suppression. The best method of suppression is still the use of water, although continuous large

amounts of water are required (Brzezinska & Bryant, 2022). As the HRR of electric vehicles is comparable with cars with an combustion engine, no different measures have to be taken in terms of fire safety for the main load bearing structure. The use of the standard fire curve for example is still a conservative method. However, the development of new suppression systems aimed at fires in electric vehicles would be very useful.



Figure 8. Evolution of HRR versus time for test vehicles which were suspended over a propane burner of 2 MW: (a) three different pure battery EVs, and (b) a small PHEV and a large PHEV compared with the gas tank and internal combustion engine (ICE) vehicles (Sun et al., 2020)

2.3.4.4 Impact of Fire Safety on Structural Design

According to the Dutch building degree, a fire compartment is 1,000 m² at maximum. Car parks however are often larger. Smaller fire compartments can be created by installing fire screens, which drop down from the ceiling in case of fire (Stöbich, 2022). However, it is allowed for a fire compartment to be bigger than 1,000 m², based on the principle of equivalence, if correct motivation is given why a fire in the larger compartment is as containable as a fire in a compartment smaller than 1,000 m². Such motivation can include installations such as a sprinkler system or smoke and heat disposal system (Kersten, 2011). It is also possible to make detailed zone model calculations, which can provide proof that for the principle of equivalence. As detailed fire safety measures are outside of the scope of this thesis, it is assumed the required motivation if required is available and therefore, fire compartments bigger than 1,000 m² are allowed. Starting point for this thesis is that every floor level is a fire compartment as a whole. However, it is also studied what the effect is if the entire car park is considered to be a single fire compartment, which is also allowed based on the principle of equivalence.

For the main load bearing structure of a multi-storey car park higher than 5 m, in general a fire resistance requirement against collapse of 90 minutes applies. This time can be reduced to 60 minutes in case the permanent fire load is smaller than 500 MJ/m². In case of a timber structure, this reduction is not allowed as timber is a combustible material and adds to the fire load. Therefore, a fire resistance against collapse of 90 minutes is assumed for this thesis as a starting point. If the entire car park is seen as a single fire compartment, a fire resistance against collapse of only 30 minutes is required (Hamerlinck, Breunese, Noordijk, Jansen, & Van Oerle, 2011; Ministerie van Binnenlandse Zaken en Koninkrijkrelaties, 2021).

2.4 Car Park Design Rules and Norms

A car park is a special type of building, for which specific rules and norms have been written, which in The Netherlands is the NEN 2443. This norm is not required, but is extremely useful as many of the user and owner requirements have been translated into this norm. A car park, which fulfils all the requirements of this norm can therefore be seen as a comfortable car park for its users. The norm is therefore used as a basis in this thesis, from which is only deviated if this results in a significant more optimal design. All of the information in this paragraph is based on NEN 2443: Design standards and recommendations on parking facilities for passenger cars (Nederlands Normalisatie Instituut, 2013).

2.4.1 Dimensions of Lanes and Ramps

2.4.1.1 Straight Lanes

The minimum width of lanes in car parks can be found in Table 2. Lanes require a redress lane on both sides of 0.25 m wide. For lanes longer than 30 m, the redress lane should be 0.50 m wide. To provide extra space for pedestrians, extra space parallel to the lane can be reserved with a minimum width of 0.90 m.

Table 2. Lane widths for straight lanes	. (Nederlands Normalisatie	Instituut, 2013)
---	----------------------------	------------------

Type of lane	Width [m]	Width redress lane [m]	Total width lane [m]
One-way-flow	> 2.75 m	0.25 m	> 3.25 m
Two-way-flow	> 5.50 m	0.25 m	> 6.00 m
Space for pedestrians	> 0.90 m		> 0.90 m

2.4.1.2 Curved Lanes

For curved lanes, the lanes have to be wider to accommodate the turning of vehicles. A horizontal slack of 0.50 m between cars and objects has to be incorporated and a horizontal slack of 100 m in case of two-way-flow for upcoming cars. The width of the lanes is dependent on the radius of the lanes, see Figure 9 and Table 3.



Figure 9. Overview of different radii for curved lanes. (Nederlands Normalisatie Instituut, 2013)

One-wa	ay-flow		Two-way-flow	
Obstacle free inner radius rv [m]	Obstacle free outer radius R _v [m]	Obstacle free inner radius rv [m]	Centre radius r _m [m]	Obstacle free outer radius R _v [m]
3.00	7.80	3.00	7.50	11.60
4.00	8.75	4.00	8.45	12.50
5.00	9.70	5.00	9.35	13.40
6.00	10.65	6.00	10.30	14.30
7.00	11.60	7.00	11.25	15.20
8.00	12.55	8.00	12.20	16.10
9.00	13.50	9.00	13.10	17.00
10.00	14.45	10.00	14.05	17.90

Table 3. Relation between different radii for one-way and two-way-flow, including redress lanes and slack (Nederlands Normalisatie Instituut, 2013).

2.4.1.3 Spiral Ramps

Spiral ramps are not popular with motorists, since they give the feeling of being very tight, which could result in damage. Based on the norm vehicle, minimum radii for sprial ramps are given in Table 4. However for comfort, recommended larger radii are also given. For comfort, a minimum cant of 3% is required in spiral ramps.

Table 4. Radii for spiral ramps including redress lanes and obstacle free zone (Nederlands Normalisatie Instituut, 2013).

	Obstacle free inner radius rv [m]	Centre radius r _m [m]	Obstacle free outer radius R _v [m]
One-way-flow minimal	4.50	NA	9.00
One-way-flow recommended	6.00	NA	10.50
Two-way-flow minimal	4.50	9.00	13.30
Two-way-flow recommended	6.00	10.50	14.80

2.4.1.4 S-corner

On locations with a transition between two parallel lanes (or $< 20^{\circ}$ difference), so called S-corners apply. An overview of applicable dimensions can be seen in Figure 10 together with the dimensions from Table 3 for one-way-flow.



Figure 10. Dimensions for S-corners (Nederlands Normalisatie Instituut, 2013).

2.4.2 Dimensions of Parking Bays and Aisles

A standard car park bay for an intensive used public car park has a width of 2.50 m and a length of 5.13 m. To accommodate turning into the bay from the aisle, for 90° parking, an aisle width of 6.00 m is required. Using smaller parking angles, the width of aisles can be reduced, however the length of the parking bay will increase, see Figure 11. Table 5 gives an overview of the different dimensions of parking bays and aisles for different parking angles.

Table 5. Dimensions for different parts of public car parks for different parking angles (Nederlands Normalisatie Instituut, 2013).



Figure 11. Top-left: One-sided parking with varying parking angles. Top-middle: Two-sided parking with varying parking angles. Top-right: Depth double parking strip. Bottom-left: Herringbone setup (parking angle 45°). Bottom-right: Herringbone setup (parking angle 60°). (Nederlands Normalisatie Instituut, 2013)
2.4.2.1 Dead Ends

When dead end lanes are longer than 8 m, two-way-flow aisles are required. In this case, it is also recommended to provide a turning possibility. Furthermore, for dead end lanes longer than 8 m, an extra space at the end of the dead end of 1 m is required.

2.4.2.2 Columns, Side Walls and Other Obstacles

Preferably, car parks have long column free parking decks. However, it is possible to locate columns in between parking bays. The location of columns should not hinder vehicles from turning into the parking bay and columns should not block doors from opening. Therefore, columns at the beginning of a parking bay should not be closer than 0.50 m to the aisle and be at maximum 1.50 m away from the aisle, see Figure 12. Columns at the back of the parking bay should not extend further than 0.60 m from the back of the parking bay should be added. In case a column is positioned on one side of the bay an extra width of 0.15 m should be added to the width of the parking bay. In case on both sides of the bay a column is located, an extra width of 0.35 m should be added. For parking angles < 45°, wider parking bays are also required.



Figure 12. Locations where columns can be placed without requiring wider parking bays (Nederlands Normalisatie Instituut, 2013).

If angled parking is used, some of the dead space in front of the bays may be used to position columns. For public car parks, a space of 0.50 m in length should be left in front of a line perpendicular to the parking bay side line, see Figure 13 left.

Another location where columns can be located without the requirement for wider bays is in the middle of four bays. In this case, columns can extend 0.20 m into the bays, see Figure 13 right.



Figure 13. Left: Space that can be used to locate columns with angled parking. Right: Placing of columns in middle of four parking bays. (Nederlands Normalisatie Instituut, 2013).

2.4.3 Slope Percentages and Clearance Heights

In car parks, different slope percentages are applied, depending on several factors:

- The height difference to be bridged;
- The available horizontal length;
- Whether it is a carriageway or a sloped parking deck;
- The car park category of the facility.

The slope percentage is measured in the centre of the lane for straight ramps. For curved ramps, also the middle of the lane is used to measure the slope, however for two-way-flow ramps, it is measured in the centre of the inner lane.

2.4.3.1 Slopes in Lanes

The maximum slope in a public car park is 14%. An exemption to this requirement is in case of a d'Humy ramp system, which is used in split-deck car parks. In this case a maximum slope of 15% is allowed. The maximum slope also depends on the length of the ramp. These maximum slope percentages can be found in Figure 14.





2.4.3.2 Sloped Parking Decks

For sloped parking decks in public car parks that are used intensively, a maximum slope of 3% is allowed.

2.4.3.3 Compound Ramps

If two or more ramps are used in succession, these ramps have to be separated by a horizontal plane with a minimum length of 10 m.

2.4.3.4 Stopping on Ramps

For ramps, which are prone to vehicles coming to a stop during normal traffic operations, a maximum slope percentage is applied. For ramps with traffic-flow in upwards direction, the maximum slope is 6%. For downward ramps, it is 10%.

2.4.3.5 Clearance Heights

Public car parks should have a clear height of at least 2.30 m in areas intended for moving vehicle traffic. Underneath beams or pipes, it is occasionally allowed to reduce the clear height to 2.20 m. These heights also apply for areas, accessible for pedestrians.

2.4.3.6 Transition Ramps

Steep slopes may result in contact between the bottom or top of vehicles and the structure, see Figure 15. Therefore, in case of ramps with a slope larger than 14%, a transition ramp has to be incorporated with a maximum slope half of the slope of the main ramp. The length of a transition ramp at the bottom of a ramp is 2.77 m and for the top it is 1.385 m.



Figure 15. Possible locations for damage to vehicles (Nederlands Normalisatie Instituut, 2013).

2.4.3.7 Clearance Height at Top and Bottom of Ramp

The clearance height at the top and bottom of a ramp should meet the requirements of Figure 16.

- a) At the bottom of a ramp without a transition ramp: draw a line at a height of 2.20 m starting from both sides of the nod at a distance of 2.77 m from the nod.
- b) At the bottom of a ramp with a transition ramp: draw a line parallel to the transition ramp at a height of 2.30 m until it intersects with the line at a height of 2.20 m parallel to the main ramp and horizontal deck.
- c) At the top of a ramp without transition ramps: draw two lines over a length of 2.77 m at a height of 2.30 m parallel to the ramp and horizontal deck.
- d) At the top of a ramp with transition ramps: the clear height should be 2.20 m for the entire top of the ramp.



Figure 16. Clearance heights at top and bottom of ramps (Nederlands Normalisatie Instituut, 2013).

20

2.4.4 Other Considerations

2.4.4.1 Watertightness

Watertightness of the structure of a multi-storey car park is very important. As it is often an open construction, weather conditions can have a significant impact on the structure. Furthermore, de-icing salts can be introduced on the parking decks, dripping of cars. De-icing salts can affect (rebar) steel, resulting in damage of the structural system. Therefore, the assembly of all structural components of the car park should be impermeable.

Water dripping from the upper parking decks onto cars can result in damage of the paint on cars. Therefore, again measures must be taken to prevent water flowing through and under the decks and elements of the structure.

To collect water coming from cars and rain, sufficient drainage should be applied. A drainage system consists out of both drains and gutters, but also the gradient of the floor elements.

2.4.4.2 Open Structure

A car park that is classified as an open structure has several advantages concerning requirements for example on ventilation and fire safety. To classify as an open structure, at least the following requirements should be met:

- At least two walls opposite one another should be exterior walls and have to be provided with openings, which cannot be closed.
- These two walls are in one direction of the car park at a maximum distance of 54 m from each other.
- The lowest deck of the car park can be at maximum 1.30 m beneath ground level.
- Interior walls should not limit ventilation.
- At least one of the following conditions should be met:
 - The total of all openings in exterior walls, not able to be closed, must together comprise 1/3 of the total area of the outer and interior walls of a compartment.
 - All openings that cannot be closed in two exterior walls, opposite of each other, must per compartment and for each exterior wall be at least 2.5% of the gross area of the parking deck in the compartment.

2.5 Car Park Layout

To achieve the objective of this thesis, which is to provide a structural system proof of concept for a temporary car park, a basic layout of the car park has to be determined. This layout is concerned with aspects like parking bay configurations and traffic flow within the car park. This will influence for example the location of ramps and required dimensions for aisles, which in term will influence the layout of the structural system. Since this thesis focuses on a temporary car park without a specific location, the layout should be applicable in a wide variety of potential building locations.

2.5.1 Circulation Design Aspects

2.5.1.1 Circulation Efficiency

The route drivers must take in their search for a parking bay is determined by the circulation design. It is most efficient, if all of the bays can be searched with just one circuit of the aisles and access-ways. However, this is not always the case, as often drivers must drive through the same aisle more than once. As a result, the circulation design has an impact on the circulation efficiency. This efficiency can be further improved through the use of signage and ability to bypass aisles (Hill et al., 2005). When exiting the car park, it is also desirable to be able to bypass the aisles with bays and drive directly to the exit using only the ramps, improving the circulation efficiency for exiting the car park.

Circulation efficiency can be expressed as a percentage of the shortest travel distance, which is to pass through an aisle with parking bays on both sides, entering from one side and exiting on the other. When ramps and access-ways are used, the circulation efficiency will drop, since extra distance has to be travelled (Hill et al., 2005). Circulation efficiency can be expressed as equation 2.1, in which the travel distance is measured along the centre of the aisles and access-ways.

$$Circulation Efficiency = \frac{Width of parking bays / 2}{Travel distance / Number of bays} 2.1$$

2.5.1.2 Crossovers

For user-friendliness in general, the layout of a car park should not hold any surprises and therefore, drivers should be able to observe other motorists movements. Crossover conditions may result in conflicts and should therefore be avoided (Hill et al., 2005).

2.5.1.3 Dead Ends

In search for a parking bay, it is difficult to see into a dead end (cul-du-sac). Therefore, dead ends should be avoided. If they are used anyhow, the number of bays in length should be limited to six (The Institution of Structural Engineers, 2011).

2.5.1.4 Parking Angle and One-way vs Two-way Aisles

In the circulation design, one-way or two-way aisles can be used. Both have their advantages and disadvantages. When two-way aisles are applied, parking bays will have a 90° angle. In case of one-way aisles, 90° parking bays as well as bays with a smaller angle may be applied.

An advantage of using one-way aisles is the prevention of conflicts between two cars, coming from opposite directions, looking for a vacant bay (Stuart, 2007). Furthermore, it can reduce confusion and congestion. A possible safety hazard appears, when drivers disregard the one-way traffic flow when they see a vacant bay (The Institution of Structural Engineers, 2011).

The advantages for one-way aisles become bigger when smaller parking angles are used. It makes it easier for drivers to enter and exit parking bays, with better visibility when backing out (Stuart, 2007). As a result, parking bays with a smaller angle have a higher dynamic and turnover capacity (The Institution of Structural Engineers, 2011). According to Stuart (2007), using 75° parking results in a more efficient layout, providing more space as a result of smaller aisles. However, according to The Institution of Structural Engineers (2011) reductions in the order of 3% are to be expected for parking angles between 90° and 70° up to 20% for 45°. Hill et al. (2005) and Troup & Cross (2003) also state smaller angle parking to be less efficient. In general, the efficiency of the parking layout is a combination of parking angle, aisle width, location of turns, ramps and barriers. An example of different parking layouts for varying parking bay angles can be seen in Figure 17.



Figure 17. Comparison between different parking layouts for varying parking bay angles (Hill et al., 2005; Stuart, 2007; Troup & Cross, 2003)

2.5.2 Circulation Layouts

Many different circulation layouts for multi-storey car parks exist, each with its own advantages and disadvantages. Choosing the right circulation layout is key in satisfying motorists and clients. Basically it should be simple (Hill et al., 2005). When choosing a circulation system, a critical factor is the volume of vehicles expected to arrive and depart in the peak hour, which has to do with the intended purpose of the car park (short stay or long stay) and its capacity (Stuart, 2007). Another factor influencing the choice for a certain layout is the layout of the site. In general, three different type of layouts can be distinguished: flat decks, split-level decks and sloped parking decks. Within these categories, still many options exist, which mostly differ in the type and location of ramps used and the circulation routing. The general advantages and disadvantages of the three categories are discussed below. A detailed description of the different types within these categories can be found in appendix B. A summary of this appendix in tabular format can be found in paragraph 2.5.2.4, Table 6.

2.5.2.1 Flat Decks

Flat deck car parks are normally built in multiples of a bin width and have an adaptable layout (The Institution of Structural Engineers, 2011). Overall, three different types of flat deck layouts can be distinguished, based on the ramp system used: internal ramps (see Figure 18 left), half external ramps and external ramps (see Figure 18 right). Generally speaking, internal ramps result in low dynamic and static efficiency. Half external ramps cost extra, but reduce the amount of bays used to complete the circulation route, boosting dynamic efficiency. In small car parks, half external ramps function independently form the car park and are often used in large car park facilities, providing high dynamic efficiency and for large car parks reasonable static efficiency. (Hill et al., 2005)



Figure 18. Two examples of flat deck circulation layouts. Left: one-way flow with side-by-side ramps (scissors type). Right: storey height, straight ramps. (Hill et al., 2005)

2.5.2.2 Split-level Decks

Split-level deck car parks generally have a good static and dynamic efficiency and are easy to drive around. They can accommodate large capacities with rapid in- and outflow routes, with most variants using one-way aisles, allowing any angle of parking. These layouts with their internal ramp systems (see Figure 19) are very compact and have a high static efficiency, most varying around 21 m² per car. The shortest variant can be constructed with a length of just 24 metres, resulting in a static efficiency of 26.75 m² per car. Except for the variants which are three or more bins wide, for drivers all turns are in the same direction and no single turn is greater than 90°. (Hill et al., 2005)



Figure 19. Two examples of split-level deck circulation layouts. Left: one-way flow with side-by-side ramps (scissors type). Right: combined one-way-flows, three bins or more wide. (Hill et al., 2005)

2.5.2.3 Sloped Parking Decks

Sloped parking decks switch the ways parking decks and ramps are constructed. They use sloped parking decks with a limited slope and flat ramps, see Figure 20. The slope of the parking deck is limited (maximum 5%) by the effect of gravity on opening and closing of doors. Also for disabled drivers in wheelchairs and the use of shopping cars, sloped parking decks are a disadvantage. These layouts generally have good static efficiency between 20 and 22 m² per car. The dynamic efficiency largely depends on the variant and whether it has a rapid in- and/or outflow route. Sloped parking decks can also be combined with one of the normally sloped decks replaced by a flat deck to compensate some of the disadvantages of a sloped parking deck. This however results in longer parking decks, greater than 72 metres. (Hill et al., 2005)



Figure 20. Two examples of sloped parking deck circulation layouts. Left: two-way-flow single helix. Right: double helix, end connected with one-way-flow on the central access-way. (Hill et al., 2005)

2.5.2.4 Overview of Layouts

The various possible layouts of the three different categories of circulation layouts are presented in detail in appendix B. Aspects such as its efficiency, circulation routing and minimum dimensions are discussed here. An overview of these aspects is presented on the next page in Table 6.

Circulation layout	Average static efficiency (most compact or extra large decks)	Dynamic efficiency	Circulation efficiency	Recir- culation	Outflow route	All turns in same direction	No turn greater than 90°	Conflict points	Dead ends (depth)	Minimum width	One- or two-way aisles	One- or two-way ramps	Maximum capacity
Flat decks													
Internal ramps running across the bins	23.5 m ² /car	Low	Low	Simple	Reasonably	Yes	Yes	Many	Yes	3 bins	One	One or two	
Internal ramps running parallel with the traffic aisles	28 m²/car	Medium	Low	Simple	Reasonably	۶	Yes	Some	Ž	3 bins	One	One or two	
Half external ramps	25 m ² /car	Good	Medium	Simple	Rapid	^S	Yes	Some	<u>ک</u>	2 bins	One	One or two	
Circular external ramps	28 m²/car (24 m²/car)	High	High	Simple	Rapid	£	Variant dependent	Variant dependant	Variant dependant	1 bin	One or two	One or two	
Storey height, straight external ramps	21 m²/car	Good	High	Simple	Rapid	^o Z	Yes	Some	2 N	2 bins	One or two	One or two	
Split-level decks													
One-way-flow with side-by-side ramps (scissors type)	21 m ² /car (27 m ² /car)	Low	Very high	Simple	Slow	Yes	Yes	Many	R	2 bins	One	One	
One-way traffic flow with an excluded rapid outflow route	22 m²/car	Medium	High	Simple	Rapid	Yes	Yes	Little	٩	2 bins	One	One or two	1100 bays
One-way traffic flow with an included rapid outflow route	22 m²/car	Medium	Very high	Simple	Rapid	Yes	Yes	Many	2 2	2 bins	One	One	400 bays
Two-way-flow with 'combined' ramps	21 m ² /car (27 m ² /car)	Low	Very high	Difficult	Slow	٩	No	Many	No.	2 bins	Two	Two	
Three bins or more wide	21 m ² /car	Low	Low	Simple	Slow	Yes	Yes	Many	P N	3 bins	One or two	One	
Sloped parking decks													
Single helix	22 m ² /car (24 m ² /car)	Low	Very high	Difficult	Slow	No	No	Many	N N	2 bins	Two	Two	300 bays
Single helix with half external ramps	22 m ² /car	Medium	Very high	Simple	Rapid	No	Yes	Little	No.	2 bins	One	One	600 bays
Double helix	21 m ² /car (20 m ² /car)	Medium	Very high	Simple	Slow	No	Yes	Many	No.	2 bins	One	One or two	
Combined helix	23m ² /car (21 m ² /car)	Low	Low	Simple	Slow	Depends c	Yes	Many	۶	3 bins	One or two	One	
Combined flat and sloping deck layouts with internal cross-ramps	22 m²/car	Good	Medium	Simple	Reasonably	Yes	Yes	Some	^o Z	2 bins	One	One or two	
						Legend							
	Highest static efficiency				Very prefer	ential charad	cteristic				One	-way	
					Preferen	tial characte	ristic				One or	two-wa	
					Weak	characterist	<u>ici</u>				Two	-way	
	Lowest static efficiency				Problem	atic characte	ristic						
					Varia	nt dependar	Ħ						

Table 6. Combined overview of different characteristics of car park layouts

2.6 Structural Typologies

The structural system of a multi-storey car park consist out of many different elements. In general, one can split up the system into a deck system, a framing system, a stability system and a foundation. Except for the foundation, different typologies for these systems are discussed below.

2.6.1 Deck System

The deck system is one of the most important parts of a car park. According to Cross (2002), the framing system is not the part of a car park that will fail, the deck is. The cause of deck failure is often moisture and chlorides attacking the rebar in concrete or steel in the decks. To protect the deck, a membrane has to be applied or other sufficient measures should be taken to prevent water and chlorides from entering the deck system. Several different types of deck systems exist of which the most common are described below. The different deck systems all have their favourable accompanying framing system, described in the next paragraph for which different building materials can be chosen. Other types of deck systems can be thought of, but the types of deck systems presented below and its way of transferring forces to the rest of the structure can be used as reference material.

2.6.1.1 Long-span Prefab Elements

Long-span prefab elements like double-tees are fabricated in a factory in a controlled environment, resulting in high quality products and minimizing on-site work. They are transported to the building site and hoisted into place (see Figure 21), ready for use, resulting in a very fast erection process. Double-tees are a very economical and durable option (Simon, 2002).

Another long-span prefab element used in car parks that is an economic solution are hollow core slabs, see Figure 21 (Lemieux & Van Kampen, 2018). Opposite to double-tees, hollow core slabs often require an in-situ applied structural screed to be able to drive on the floor. This extra step slows down erection a bit and results in a solid floor slab, which is more difficult to dismantle. Nevertheless, hollow core slabs are still a fast construction technique and relatively light weight. A big advantage compared to double-tees is the ability of hollow core slabs to accommodate free formed floor plans (Soons, Van Raaij, Wagemans, Pasterkamp, & Van Es, 2014). Special care should also be taken to prevent water and de-icing salts from entering the concrete. A membrane and seep holes are possible solutions.



Figure 21. Left: Double T floor (Haitsma Beton, 2018). Right: Hollow core slab floor (Croom Concrete, 2018).

2.6.1.2 Short-Span Prefab Elements

For shorter spans (< 10 m), precast concrete planks can be used as a deck system. It is possible to use these planks as formwork, to create a solid floor slab. However, it is also possible to simply use completely prefabricated concrete deck elements and place them directly on the supporting girders, making construction very fast (Max Boegl, 2020). Through the use of shear connectors on the supporting girders, composite action can also be realized



Figure 22. Prefab concrete floor elements (Max Boegl, 2020).

with this type of floor elements, see Figure 22. Since these elements are created in the factory, a very high quality product can be delivered, able to withstand de-icing salts.

A new development of building materials in the construction industry are composite materials. Shortspan floor elements can also be made using composite glass fibre. A big advantage of this material is that it is very lightweight and durable. It is not affected by chlorides, making it a very interesting product to use in a car park.

2.6.1.3 Composite Floors

Several types of deck systems make use of composite action. Possibly the most known and often used in car parks are corrugated sheet metal sheets combined with in-situ concrete, see Figure 23. The metal sheets act as formwork for the concrete, which speeds up construction as no separate formwork is required. The metal sheets also act as main reinforcement with additional reinforcement to minimize cracking, which could lead to ingress of de-icing salts in a car park deck. This deck system has a low self-weight and can span up to 9.60 m. The depth of the floor can be reduced as the corrugated sheets can be placed on the widened bottom flange of a profile. Trough applying metal studs as shear connectors on the girders supporting the floor, the steel girder will take up tensile stresses and the concrete compression, resulting in a composite beam. (Sarka, 2005; Soons et al., 2014)



Figure 23. Corrugated sheet metal floor under construction in car park (SMD Ltd, 2019).

2.6.1.4 Cast-in-place

A cast-in-place deck has the advantage of giving a high degree of crack control, which is important in car parks. Though post-tensioning, the thickness of the floor can be reduced and cracking can again be controlled. Also free floor plans can be constructed (Kaminker, 2004). Disadvantages of the system are the slower construction and high self-weight. An example of an cast-in-place car park can be found in Figure 24.



Figure 24. Cast-in-place car park (City of Stonnington, 2018).

2.6.2 Framing System

The framing system of a multi-storey car park is build-up of columns and beams supporting the floor elements and a certain stabilizing system, keeping the system upright. The most used framing systems are shown below. More systems and many variations can be thought of, but these systems cover most general principles of the available systems. Different materials can be used for the systems, although not every system might be useful or efficient for each building material. Similarly, not every framing system works with every deck system. Certain combinations of deck system, framing system and chosen building material are used often. Below, different structural typologies for the framing system are presented as potential options for the structural design of the car park.

2.6.2.1 Two-way Carrying Slab

The most basic looking framing system is shown in Figure 25. It is composed of a set of columns, which carry a two-way carrying floor slab. Considering building materials, this type of system is only constructed using concrete. Many variations exist, for example using drop panels around columns to limit floor depth or a waffle slab, minimizing self-weight and resulting in a more efficient structure (ACI Committee 362, 1997). Columns can be located at the end of the parking bays, resulting in a column free bin or can be placed between bays, minimizing floor depth.



Figure 25. Two-way carrying slab framing system (AMC, 2021).

2.6.2.2 Bin Wide Floor Elements

Many (modular) car parks are constructed using double-t floor elements (Englot & Davidson, 2001). This type of floor requires a very simple framing system, comprised out of columns, connected by a girder on top of which the floor elements are placed, which span the entire parking deck, see Figure 26. Other floor elements like hollow core slabs can be thought of, but it is a challenge to find those able to span the parking deck up to more than 16 m.



Figure 26. Bin wide floor elements framing system (Cobouw, 2017)

2.6.2.3 Bin Wide Girders Supporting Perpendicular Floor Elements

This framing system consists out of columns, which carry girders spanning the entire parking deck, see Figure 27. Floor elements are placed perpendicular to the girders, reducing their span. A variant of this system requires less columns, which are replaced by an extra girder spanning in between the columns in the other direction and carry one girder supporting the deck, see Figure 28. This type of framing system can be constructed using many different building materials and types of girders.



Figure 27. Bin wide girders supporting perpendicular floor elements framing system (Zaugg, 2018).





Figure 28. Bin wide girders supporting perpendicular floor elements with extra girder framing system (Bourne Parking Ltd., 2012).

2.6.2.4 Bin Wide Girders with Perpendicular Joists Supporting Floor Elements

Reducing the span of floor elements, this framing system is build-up of columns, girders spanning the entire parking deck and joists in between the girders, see Figure 29. The shorter floor spans presents opportunities to use floor elements with a significantly smaller depth. Considering the different types of spans of beams and floor, different building materials can be used.



Figure 29. Bin wide girders with perpendicular joists supporting floor elements framing system (Park4all, 2019b)

2.6.2.5 Girders Across Parking Bays with Primary and Secondary Joists Supporting Floor Elements This framing system looks rather complicated. The system is normally used featuring columns in between the parking bays, see Figure 30. The columns are connected by girders, which carry a grid of joists on top of which a floor can be placed. This system is useful when very thin floor elements are used, which require a small span. As a lot of connections have to be made, construction can take longer for this type of system.



Figure 30. Girders across parking bays with primary and secondary joists supporting floor elements framing system (Osborne, 2019).

2.6.3 Stability System

2.6.3.1 Vertical Stability

To resist lateral loads, several types of vertical stabilizing systems can be applied: clamped columns, diagonal braces, moment frames, shear walls and cores.

Columns clamped in the foundation can provide stability for relatively low car parks (up to three levels) (Van den Broek, 2019). For higher car parks, the forces on the connection will become too large and another type of stability system has to be incorporated.

Diagonal bracing systems are generally the most economic option (Brinksma, 2004). A double diagonal cross using steel elements is often applied (X-bracing), in which case the members are only loaded in tension. Other types of bracing exist, like K, X or V braces, see Figure 31 (Ülker, Işık, & Ülker, 2017).

Moment frames are a more expensive option and may result in complex details (Troup & Cross, 2003). An additional benefit however is the possible reduction in depth of beams, which may result in a less thick floor.

Shear walls are another solution to provide stability to a building. Such walls can be found in car parks, made out of concrete. A disadvantage of shear walls is that they can block sight lines. Parking garages are preferably open with clear sight lines to provide a feeling of safety (The Institution of Structural Engineers, 2011).

A stiff core like an elevator shaft can also be used to provide stability. Such a single stiff core is more often used in higher buildings and not in relatively low and open structures like car parks. It is however possible to use the stairwells as stabilizing elements.



Figure 31. Different types of diagonal braces (Ülker et al., 2017)

2.6.3.2 Horizontal Stability

Horizontal stability of car parks is often provided by its floors, which can transfer horizontal loads to the stabilizing elements in the building, which can transfer the loads to the foundation. Depending on the type of floor system used, floor elements might have to be coupled together to be able to transfer these loads. Other types of horizontal stabilizing elements often used are diagonal braces. Both types of horizontal stabilizing systems can are shown in Figure 32. (Van den Broek, 2019)





Figure 32. Left & middle: Couplers between deck panels to create horizontal stabilizing element. Right: Diagonal bracing as horizontal stabilizing system (Park4all, 2019a).

2.7 Concluding Remarks

To conclude this chapter, sub-question 1a, "What are the design requirements for a multi-storey car park?", and 1b, "What are the possible typologies for a multi-storey car park?", can be answered.

Considering the design requirement for a multi-storey car park, we can conclude that users of car parks are mostly concerned with functional requirements like the (dynamic) capacity of a car park and how easy the circulation through them is. Such requirements are translated by the parking norm NEN 2443 into a large set of minimum and standard dimensions, rules for layouts of ramps of placement of columns and other general requirements. If these requirements are incorporated in the design of a multi-storey car park, one can say with reasonable certainty, that the design will be appreciated by its users.

The owner's requirements of course also reflect the requirements of the users, since they aim to provide their clients with a pleasurable parking experience. However, apart from such requirements, the owners are also concerned with aspects of the construction that affect their ability to get a return on their investment. This means they require a construction, which is durable, fast to construct and requires minimal maintenance. Furthermore, a high static efficiency and low average cost per parking space should result in a cost-efficient design. Last of all, more recently, owners are also concerned with the sustainability performance of the design as well as its flexibility for future use in other functions.

Different possible typologies for the design of a multi-storey car park have also been presented. These typologies vary between general layouts of car parks, structural typologies for its construction and different types of elements that can be used within them. The different requirements of the users and owners of the car park have to be reflected by the chosen typology. This chapter has presented characteristics of the different typologies, which can be used to assess which car park layouts and structural designs are feasible. As designing is a very creative process with endless possibilities, this chapter functions as reference material for the further design process.

3

Challenges Towards a Sustainable Building Industry

As the building industry is in the middle of a transition towards a more sustainable industry, it has to change the way buildings are designed and constructed. This requires understanding what building sustainable is and how its principles can be translated into the design of a building. The transition can be supported by several developments and innovations that have been introduced into the building industry in recent years. Understanding how such innovations can help to bridge the gap between the traditional building industry and the future sustainable building industry will play an essential role in the speed with which the transition takes place. This chapter aims to provide an answer to what building sustainable is and with what challenges it comes. Furthermore, it aims to provide an answer to how these challenges influence the design of a building.

3.1 Building Sustainable

The building industry is not the only sector which is in a transition towards a more sustainable industry. It is part of a much broader topic, which concerns all people and industries; sustainable development. In 1987, the Brundtland Report (WCED, 1987) published today's most used definition of sustainable development: "Sustainable development is development that meets the needs of the present without compromising the ability of future generations to meet their own needs". The report highlights the inequality between poverty in one part of the world and the non-sustainable consumption and production in the other part. As a result, within sustainability, three key areas can be distinguished (Pitt, Tucker, Riley, & Longden, 2009):

- 1) Environmental responsibility
- 2) Social awareness
- 3) Economic profitability

3.1.1 Impact of the Building Industry

The building industry has a significant impact on sustainability as it affects all three key areas. In terms of environmental aspects, the construction and demolition of a building results in a large use of building materials and waste creation. The production of many building materials also results in carbon emissions and other pollutions (air, noise), which among others affect the bio-diversity. Last of all, during construction, but especially in the use phase, large amounts of energy and water are used (Czarnecki & Kapron, 2010; Pitt et al., 2009).

Buildings also affect social aspects of sustainability. The quality, design and performance of a building directly affects the quality of life in and around the building. Sustainable buildings are healthier to live and work in and can result in greater productivity. Also the location of a building with its access to services and recreation adds value to a society (Pitt et al., 2009). Social aspects also play an important role in the construction phase, as construction work is a physically heavy job in which health and safety is very important (Heller, Hawgood, & Leo, 2007).

Sustainable buildings also have to be funded, so economic aspects always play an important role. In the development of sustainable buildings, the client is a key driver, who is primarily focussed on the aspects of time and costs (Bowen, Cattel, Hall, Edwards, & Pearl, 2012). Sustainable construction

therefore has to be made attractive. Although the initial cost of a sustainable building is higher, they attract higher rents and prices, attract tenants and reduce tenant turnover. Also, in operation and maintenance, sustainable buildings cost less. As the public is more concerned with sustainability, a sustainable building and Value Management (VM), minimizing environmental and social damage, can be attractive in attracting investments (Pitt et al., 2009). By making a cost-effective design, a sustainable building can also become feasible from a financial point of view (Czarnecki & Kapron, 2010).

3.1.2 Principles for Sustainable Construction

Pitt et al. (2009) point out the importance of the design and the design team of a building in sustainable development. Although the client is the key driver and final person to make decisions, the designers have the opportunity to influence and inform the client. They can build on their own reputation and image concerning sustainability. Decisions like the selection of building materials have a significant impact on energy use and greenhouse emissions. Choices made by structural engineers can reduce costs and use of materials. The design team can help demonstrate ways to a client for a faster or better return on investment, to increase flexibility, to reduce costs, to increase market appeal and to improve the clients image (Pitt et al., 2009).

Nine basic principles for sustainable construction can be distinguished (Babalola, 2017; Czarnecki & Kapron, 2010):

- Minimization of consumption of resources;
- Maximization of reuse of resources, components and structures (renovation);
- Use of renewable and recyclable resources;
- Waste management;
- Minimization of emission of pollutants;
- Protection and preservation of natural environment;
- Development of a healthy and non-toxic environment;
- Comfort of use (quality) of built environment;
- Quality control, construction process management and building management in developing the built environment.

3.1.3 Barriers of Sustainable Construction

The building industry is in the middle of a transition towards a more sustainable industry. There are many examples, in which new buildings are designed according to the principles of sustainable construction. However, sometimes projects start off in the feasibility stage of the design aiming for maximum sustainability levels, but scale down this ambition when costs start to rise or when contractors for example do not feel comfortable with new construction methods like timber construction (Fryer, 2012).

Several barriers have been identified as holding back sustainability in the construction industry. Pitt et al. (2009) did research into these barriers and found affordability to be the main barrier. Affordability is associated with the higher initial cost of sustainable buildings and a slow rate of return on the investment of this higher initial cost (Matar, Georgy, & Abou-Zeid, 2010). Another significant barrier is a lack of client demand, which was also found by Matar et al. (2010), who described a lack of interest from major industry stakeholders towards sustainable construction. This might be connected with another barrier, which is lack of client awareness. This stresses the importance of the design team to inform clients on sustainable construction. Other barriers are a lack of business case understanding, lack of proven alternative technology, building regulations, planning policy and lack of labelling or measurement standard. Matar et al. (2010) add to these barriers the fact that the construction industry is a very fragmentated industry, resulting in misperception and misunderstandings, which make implementation of sustainability practices amongst all partners in the value chain a challenge. Furthermore, they point towards a lack of sufficient education and training among industry practitioners, the weak and slow adoption of digital technology and the clatter, confusion and inefficiencies of current tools and approaches to sustainable construction.

3.1.4 Drivers of Sustainable Construction

Sustainable construction is also pushed by several drivers. Pitt et al. (2009) also researched these drivers and found out financial incentives and building regulations to be the most important. Many governments have a sustainability agenda and present for example fiscal incentives for sustainable construction. Through legislation, certain requirements concerning sustainability also end up in standards and building requirements. Other drivers are client awareness and demand, which are becoming more important as corporate and social responsibility practices towards the encouragement of sustainable practices are recognized by the public. Companies want their image and reputation to be connected to sustainability. Less important drivers were planning policy, taxes or levies and investment.

3.2 Circular Construction

Many different approaches, methods, tools and strategies have been developed to contribute to the transition of the building industry towards a sustainable industry. They function as a means to a goal of achieving sustainable development (Anastasiades, Blom, Buyle, & Audenaert, 2020). One of these approaches, which is seen as very promising and which incorporates many of the basic principles for sustainable construction is circular construction or the circular economy (Bocken, de Pauw, Bakker, & van der Grinten, 2016). Guerra & Leite (2021) describe the ultimate goal of the circular economy model as retaining resources circulating at their highest value within the boundaries of the planet, in such a way that no additional resources are required to produce materials, and the discarded materials are not perceived as waste. Products and materials are continuously reused and renewable resources are used where possible, see Figure 33 (Bocken et al., 2016). The focus of the circular economy is wider than just use of materials. It's about better management of the resources by refusing, rethinking and reducing unnecessary consumption (Guerra & Leite, 2021). The European commission also associates the move to a circular economy with opportunities to create jobs and boost economies, contributing to it being a means to true sustainable development (European commission, 2015).



Figure 33. Linear vs circular economy (Petovarga, 2019)

3.2.1 Resource Loops

It's clear that the flow of resources is one of the most important aspects in circular construction. Traditionally, the building industry has always adopted a linear flow of resources, starting with the mining of raw materials, followed by production of building materials and elements, the use phase and finally demolition, in which the materials become waste. The lifecycle of these resources is an open loop and can be referred to as cradle-to-grave. In a circular economy, the start and end-of-life phases of the building cycle are coupled, creating a closed loop, which is referred to as cradle-to-cradle (Guerra & Leite, 2021). Circular construction is concerned with the alteration of these loops, closing them. Bocken et al. (2016) have distinguished three different aspects on how resource loops can be altered: slowing resource flows, closing resource flows and narrowing resource flows. These aspects can be visualized on three axes, resulting in different types of loops, see Figure 34. Apart from linear and circular flows, different combinations with also extended and/or narrowed flows are formed.



Figure 34. Different types of resource flows (Bocken et al., 2016)

3.2.2 Circular Design Strategies

Different circular design strategies can be defined and can be based on alteration of the resource loops as described above. Bocken et al. (2016) conducted research into these strategies and categorized them according to whether they aim at slowing, closing or narrowing resource loops. Guerra & Leite (2021) also did a literature study on circular design strategies, which can also be classified according to the categories of Bocken et al. An overview of the most cited circular design strategies can be found in Table 7. These strategies are discussed in the following paragraphs.

Table 7. Circular design strategies (Bocken et al., 2016; Guerra & Leite, 2021)

Design strategies to slow resource loops	Design strategies to close resource loops
Designing long-life products	Design for a technological cycle
Design for attachment and trust	Design for a biological cycle
Design for reliability and durability	Design for dis- and reassembly
Design for product-life extension	
Design for ease of maintenance and repair	Design strategies to narrow resource loops
Design for upgradability and adaptability	Design out waste
Design for standardization and compatibility	
Design in layers	
Design for dis- and reassembly	

One of the most often cited circular design strategies is to design for dis- and reassembly (Bocken et al., 2016; Dams et al., 2021). It's a strategy, which fits both strategies for closing and slowing loops. Through this strategy, it is possible to replace or repair components of a building, extending the buildings life. When a building is demolished, its components can be disassembled to be reused or recycled in other projects (Anastasiades et al., 2020). Dis- and reassembly require planning during the initial design stages. For example a well-documented disassembly plan in which recyclable and reusable materials are specified improves the possibilities for reuse of elements (Dams et al., 2021; Guerra & Leite, 2021). Also the way connections are designed is very important (Dams et al., 2021).

3.2.2.1 Slowing Loops

Design strategies concerned with slowing loops can be aimed at designing long-life products or at design for product-life extension. Strategies for designing long-life products are to design for attachment and trust and design for reliability and durability (Bocken et al., 2016; Dams et al., 2021). Goal of these strategies is to design building components, which are of high quality, ensuring a long utilization period. Important aspects for these strategies during design are the selection of materials and the appearance of the components. Durmisevic (2019) emphasizes the importance of the way materials are put together as this is essential for ultimate durability and circularity of buildings apart from the durability of its materials.

With design strategies for product-life extension, service loops are introduced, which extend the products life-span. Strategies are to design for ease of maintenance and repair, for upgradability and adaptability, for standardization and compatibility and to design in layers (Bocken et al., 2016; Dams et al., 2021; Guerra & Leite, 2021). Especially designing for adaptability is a very popular strategy as it allows for reconfiguration or conversion of a building, to reflect changes in its purpose or use during its lifespan, minimizing the risk of demolition as a result of changing economic, societal or functional demands (Anastasiades et al., 2020; Dams et al., 2021). Ease of maintenance and repair helps in keeping a building in optimal condition and standardization and compatibility results in parts that fit multiple products, making them interchangeable (Bocken et al., 2016). By designing buildings in layers, different components with different life-spans can be separated from each other, creating opportunities to replace layers or reuse them (Guerra & Leite, 2021).

3.2.2.2 Closing Loops

Design strategies concerned with closing loops, apart from designing for dis- and reassembly, are design for a technological cycle or for a biological cycle. In closing loops, the selection of materials is very important (Guerra & Leite, 2021). The technological cycle is concerned with designing products in such a way, that its materials can continuously and safely be recycled into new materials or products. The properties of the new materials or products should be a least of the same level as the previous materials or products. Downcycling therefore is not permitted as this will only slow down the resource flow, not closing it (Bocken et al., 2016).

The biological cycle is concerned with designing products using safe and healthy materials. These materials are biodegradable, so at the end of their lifespan, a new loop is started (Bocken et al., 2016). One can therefore say this is also a form of recycling.

3.2.2.3 Narrowing Loops

Design strategies concerned with narrowing loops are all about optimizing material use and minimizing the generation of waste (Guerra & Leite, 2021). The use of reused or recycled materials and products also reduces the generation of new waste.

3.3 Incorporating Circular Design Strategies

Several different circular design strategies have been presented. These strategies are often closely related to each other and for optimal and efficient results, all of them should be put into action. The following three key themes combining several circular design strategies are suggested:

- Modularization, prefabrication and standardization;
- Reuse, refurbish and recycle;
- Building materials.

Digitalization and digital tools are another development, which can support the three key themes above.

3.3.1 Modularization, Prefabrication and Standardization

A large group of construction methods, which adopt several circular design strategies is the group of modularization, prefabrication and standardization. Although these are three different concepts, they often go hand-in-hand (Guerra & Leite, 2021).

Architects often like to create unique, complex and aesthetically pleasing buildings, however considering circularity, the resulting buildings are often difficult to adapt and reuse. If more standard materials and components would be used, it would be easier to reuse these materials and components in other buildings and repairs would also be easier, extending the lifespan (Geldermans, 2016). Furthermore, the use of standard regular structural grids with standard dimensions, makes construction and adaptation much simpler (Dams et al., 2021). Standardization therefore facilitates the reuse and repurpose of building components.

Modularization and prefabrication are methods, which can achieve circular construction through standardization. As standard components are used, they can be combined into larger modules, which can be prefabricated in a factory. Through modularization and prefabrication, on-site material waste is reduced, working circumstances and safety are improved, higher quality products are produced and productivity is boosted (Guerra & Leite, 2021; Mischke, 2017). Furthermore, these methods can promote economic viability as project schedules can be reduced, efficiency is improved and costs are reduced (Dams et al., 2021).

3.3.2 Reuse, Refurbish and Recycle

Circular design strategies like designing for dis- and reassembly and designing for upgradability and adaptability are concerned with the theme of reuse, refurbish and recycle. To close resource loops, it is important to first reduce materials used, than reuse them and finally recycle them. To what extend materials and components can easily be reused or recycled depends on whether a building has been designed and composed in a demountable way (Kayaçetin, Verdoodt, Leferve, & Versele, 2022). A building is composed of several layers, which each have a different life expectancy. Often, the "six S" system is used, which distinguishes the site, structure, skin (façade), services (mechanical and electrical), space plan (internal layout) and stuff (occupant possessions) (Brand, 1995), see Figure 35 left. To be able to reuse layers or replace them, extending the lifespan of the entire building, it is important that these layers are accessible and have reversible connections. Important aspects of these layers are material impact, functional independence, technical detachability, physical characteristics and recyclability (Kayaçetin et al., 2022).

The different layers of a building or systems themselves are also composed of different components and elements, see Figure 35 right. The connections between these elements and components also have to be demountable to be able to reuse components and materials to full potential. Dry mechanical connections like bolts, screws, dowels and clamps are preferred. Nails and toothed plates are less suitable for disassembly (E. Durmisevic, 2006; Morgan & Stevenson, 2005). Dry connections should be accessible in its broadest sense as Anastasiades et al. (2020) point out paint can hide screws, hindering disassembly. Durable and reusable connections can be made using stainless steel (Guy & Ciarimboli, 2008) and highly suitable connections for circular construction are interlocking or gravity-connected connections in shear (A. Durmisevic, 2019).



Figure 35. Left: The "six S" diagram (Brand, 1995). Right: Levels of decomposition (Durmisevic, 2006).

Adaptability of a building starts with the design of the original building. Key principles for adaptive reuse are functional performance, structural performance, a buildings ecosystem and its visual appearance (Patil, Patil, & Patil, 2021). Functional flexibility can be created through designing large spans, maximising use of mobile partitions and simple open-plan space (Guy & Ciarimboli, 2008). Structural flexibility can be created through clever over dimensioning certain elements to suit future adaptations like larger foundation capacity or stronger columns, providing the possibility for additional floor levels in the future (Anastasiades et al., 2020). As Strumillo (2016) points out; reusing existing buildings presents the greatest opportunity for the reduction of energy consumption and carbon emissions.

3.3.3 Building Materials

The selection and specification of materials for building construction is a widely discussed topic in literature. It is a very important aspect in closing resource loops through the strategies of designing for a technological and biological cycle. In terms of the technological cycle, to be able to reuse materials, they have to be durable and of high quality as they have to be able to withstand destructive factors such as changes in temperature, humidity, static and dynamic loads and chemical or biological aggressions (Sagan & Sobotka, 2021). Materials can also be recycled, but it is important that the properties of the recycled materials are at least of the same quality as the original material. Furthermore, one should also take into account the process of recycling and how sustainable this process is.

Taking in mind the aspects concerned with demountability in the previous paragraph, one can evaluate main stream building materials like steel and concrete. A steel structure can easily be constructed using dry mechanical connections and its elements are easy to inspect and dismantle, making it easy to reuse them (Densley Tingley & Allwood, 2014). As a final option, steel can be recycled, creating new steel of the same quality. Although this requires 67% less energy compared to using virgin materials, it still does require the consumption of a large amount of energy (Johnson, Reck, Wang, & Graedel, 2008). Concrete is more difficult to reuse or recycle. Its elements are often connected through plastic joints, making disassembly very hard. Furthermore, concrete elements contain rebar, designed for a specific purpose. As the rebar is surrounded by concrete, it is difficult to inspect elements for reuse and if technical drawings are missing, it becomes impossible to see from the outside how an element is reinforced (Anastasiades et al., 2020). In terms of recycling of concrete, over the years, a lot of research has been conducted in this field of research. It is possible to recycle large amounts of concrete, however producing new concrete solely using recycled concrete is very difficult (Le & Bui, 2020). As a result, concrete is still often downcycled and not recycled.

The use of materials in the biological cycle should also be done in a clever way. The use of these materials is only environmentally beneficial if these bio-based resources are managed in a sustainable way. Timber can be such a material, however during its production and transportation, emissions are also produced. Sustainable production and a design life span matching at least the timber rotation period are therefore crucial (Anastasiades et al., 2020).

3.3.4 Digital Tools

The digital era has presented many new ICT possibilities to the world. Many industries have embraced digitalization and as a result have seen a boost in productivity and efficiency (McKinsey & Company, 2017). Digitalization can also provide a supportive role in circular construction. A software database can store all kinds of data of a building and can be coupled to all kinds of design and planning applications. Building materials and elements from a donor site can in this way be registered, creating

an inventory of available construction elements for reuse (Dams et al., 2021). Using for example parametric engineering software, new designs can be optimized or designed using standard elements. It is clear that digital tools provide many possibilities to support circular construction.

3.4 Sustainability Assessment

As presented, many different circular design strategies can be applied to the design of a building. Furthermore, other sustainability strategies aimed at for example minimizing the use of energy in the use phase of a building or generation of green energy can be applied. To be able to say something about the sustainability performance of a building, somehow an assessment should be made.

Many different types of sustainability assessment methods exist. One of the most used type of methods is a Life Cycle Assessment (LCA). Goal of a LCA is to measure the environmental impact of a product over its life cycle from cradle-to-grave (Muralikrishna & Manickam, 2017). LCA's however have several shortcomings if one wants to assess the circularity of a building. LCA's tend to be not practically applicable enough during the design and development phase of a building. To make them useful, they should be simplified. Also, LCA's consider the complete life cycle of a product, but it has been found that in many studies, the end-of-life phase is not or not consistently taken into account (Anastasiades et al., 2020). With circular design strategies, aiming to close resource loops and reuse materials and elements, the end-of-life phase or with it the start of a new cycle is very important. LCA's do not assess the reuse performance of a product enough for a good circularity assessment (Anastasiades et al., 2020).

Today, buildings are often rated for their sustainability performance using a certain certification. Widely used certification systems are LEED and BREEAM (Guerra & Leite, 2021). Again, these frameworks assess the sustainability performance of a building and apply some LCA principles, but they are no good ratings for circularity as their focus is not on the circular design strategies (Anastasiades et al., 2020).

To assess the circularity performance of a building, Anastasiades et al. (2020) suggest to perform a Multi-Cycle Assessment (MCA). Such a method is especially focussed on circular design strategies like designing for disassembly and adaptability. Dams et al. (2021) have developed such a framework, which they call the Circular Construction Evaluation Framework (CCEF). It has specifically been developed to complement LCA assessments on circularity topics and has been developed to be used in the early design and planning stages of a project. The framework evaluates a building as a whole as well as its elements on component level. On whole building level, the main themes are: recorded information, adaptability in design, simplicity in design and health and safety. On component level, the main themes are: durability, material inventory, finishes/treatments, reversibility of connections, reusability and recyclability.

3.5 Concluding remarks

In conclusion of this chapter, an answer can be provided to sub-question 2: "How do the challenges faced by the construction industry towards a sustainable industry influence the design of a temporary multi-storey car park?". Also both its sub-questions can be answered: 2a "Which sustainability building strategies can be applied on the design of a temporary multi-storey car park?" and 2b "What is the impact of sustainability building strategies on the design of building elements?".

It is clear that the building industry can make a significant impact on sustainable development. Construction projects are linked to all three key areas of sustainable development; environmental responsibility, social awareness and economic profitability. Contributing to a sustainable building can be done using the presented nine principles for sustainable construction. Circular construction has been presented as a very promising approach, which incorporates many of these basic principles and gives more guidance on how a sustainable design can be realised. Circular construction aims to make the transition from the linear economy model, which is traditionally used by the building industry, towards a circular economy, in which products and materials and are continuously reused and where renewable materials are used where possible. Three circular design strategies have been identified, which are aimed at altering the resource flows in the building industry, to make the transition from a linear to a

circular economy. These are: slowing resource loops, closing resource loops and narrowing resource loops. For each of these strategies, several measures have been presented.

Incorporation of these circular design strategies has a significant impact on the design of a building. Three key themes have been identified, which incorporates the different circular design strategies and affects the design (process) of a building. Modularization, prefabrication and standardization will fundamentally change the way buildings are designed and produced, as different buildings are designed using similar components. The second theme of reuse, refurbish and recycle, makes use of the first theme as it means standardized building elements can be reused in different building, maximizing its lifespan. The third theme is concerned with the use of different (renewable) building materials, which requires designers and contractors to rethink how they can build with such materials. Digital tools can fulfil a supportive role in this renewed design and construction process.

4 Timber Structures

Timber as a building material has a very long history. Where it started with logs being stacked on top of each other in order to form a structure, over many centuries highly-developed woodworking techniques have been developed. Together with other developments in construction techniques and machinery, this has resulted in an continuing growth of possibilities in the field of timber construction. More recent developments in wood products have paved the way for timber to be used in an even broader range of structures with larger spans, increased loads and higher elevations. This chapter aims to answer the question on how to construct using timber as a constructive material. It gives an overview of the different wood products and components used in construction today and types of connections used. Furthermore, an overview of durability aspects is given and what measures should be taken, as this is inextricably linked to building with timber. Last of all, some reference projects concerning the use of timber in car parks are presented.

4.1 Wood Products

Many types of wood products can be distinguished. Naturally, a division can be made between softand hardwoods and the different species amongst them. However, more recent developments in the timber industry have resulted in so-called engineered wood products. These products have improved characteristics, increasing the potential use of timber in constructions. Each type of product has different characteristics, determining its potential use. Apart from soft- and hardwoods, wood products can be categorized as solid wood and solid wood products or wood-based products. Differences between these products are based on the components they are made of as can be seen in Table 8.

	Wood product	Components	
Solid wood products	Logs Sawn timber Glued laminated timber (glulam) & Cross-laminated timber (CLT)	Stems Squared timber, planks, boards and battens Boards	
Wood- based products	Laminated veneer lumber (LVL) Plywood Parallel strand lumber (PSL) Particleboards Fibreboards	Veneers Veneers or sawn timber Veneer strands Particles (chips) Fibres	•

Table 8. Wood construction products and their components (Blaß & Sandhaas, 2017)

4.1.1 Soft- and Hardwood

Wood can be classified in two different types: softwood and hardwood. Most timber products are made from softwood, about 80% (Arets et al., 2011). Also in structural timber, both soft- and hardwoods are used. Compared to softwood, hardwood is much stronger. Although its strength is much higher, other characteristics like the modulus of elasticity and the stiffness of these structural elements do not increase to the same extent as their strength (Kaufmann, Krötsch, & Winter, 2018). Softwoods are less dense, in general easier to work with and much cheaper than hardwood as it grows much faster (Diffen, n.d.). Softwoods are most popular in construction, but the increasing availability of hardwoods and new technological innovations open up new possibilities (Kaufmann et al., 2018).

4.1.2 Solid Wood and Solid Wood Products

Solid wood and solid wood products are made using sawn timber elements. These can be combined and rearranged to create improved wood products. These solid wood products however remain build-up of wood elements in their original form.

4.1.2.1 Solid Timber

Solid timber are members completely made of wood in its natural structural form. These members are produced through sawing logs longitudinally. The resulting products depending on their cross-sectional dimensions are boards, planks, battens and square timber (Blaß & Sandhaas, 2017), see Figure 36 left. Sawn sections can also be glued together to form products like four-piece beams and duo/trio beams (Herzog, Söffker, & Thrift, 2004).

4.1.2.2 Glued Laminated Timber (Glulam)

Glulam members are produced through laminating boards arranged parallel to the grain together with an adhesive, see Figure 36 middle. As growth-related defects in the wood are partly eliminated, the result is a rigid more homogeneous and therefore stronger beam. Glulam is an engineered wood product and allows for larger cross-sections than sawn timber, but besides simple, straight components, also allows forms with a variable cross-section and/or single or double curvature or twist about the longitudinal axis. It is also possible to mix different strength class timber boards, resulting in a non-homogeneous member. These glulam members are particularly suited to accommodate bending stresses. They are often used as beams and columns. The maximum length of the elements is limited by transportation. (Blaß & Sandhaas, 2017; Herzog et al., 2004)

4.1.2.3 Cross-laminated Timber (CLT)

Cross-laminated timber (CLT) is another engineered wood product. Unlike glulam, which result in linear members, CLT products are plate-shaped, see Figure 36 right. It is made up of multiple cross-wise arranged board layers, which are glued together and can be produced in a wide range of panel thicknesses. The cross-sections can take many shapes as it is possible to make CLT panels partially filled with gaps or with an asymmetrical cross-section. It is also possible to bent a panel into a plane curve. Like with glulam, CLT panels take advantage of being more homogeneous elements, but this time the panels are approximately isotropic in plane. The panels have high strength and stiffness properties and have a high load-bearing capacity relative to their self-weight. They can be loaded perpendicular to the plane (bending parallel or perpendicular to the grain of the external layers) or in plane (bending, compression and tension in plane and parallel or perpendicular to the grain of the external layers). CLT is often used as wall and floor elements, but can also be used for beams or shell structures. (Blaß & Sandhaas, 2017; Föreningen Sveriges Skogsindustrier, 2019)



Figure 36. Left: solid timber products (Herzog et al., 2004; Ilarch, 2017), middle: glulam (Adis, nd; Kaufmann et al., 2018), right: CLT (Föreningen Sveriges Skogsindustrier, 2019; Kaufmann et al., 2018).

4.1.3 Wood-based Products

Instead of solid wood products, wood-based panels are produced by disassembling and subsequently reassembling wood, using adhesives. For different products, different starting materials are used like sawn timber, veneers, particles, wood shavings or fibres. The resulting panels are characterised by their approximated isotropy in plane and low level of variance in properties (Blaß & Sandhaas, 2017).

4.1.3.1 Veneer-based Panels (Plywood, LVL & PSL)

Veneer-based panels consist of multiple layers of veneer bonded to each other using resin. Veneer layers are produced by rotary peeling logs, resulting in 0.5 to 6.0 mm thick strips. The defective portions of the veneer are than removed, so the resulting panels have homogeneous properties. Depending on the configuration in which different layers are placed on top of each other, different products are formed (Blaß & Sandhaas, 2017).

For plywood, the veneer layers are bonded to each other in alternative crosswise layers. By varying the used wood species or thickness of the veneer layer, panels with different properties can be produced. The resulting plywood panels have approximately isotropic properties and are used mainly as sheathing materials in horizontal or vertical diaphragms (floors, roofs or shear walls) (Blaß & Sandhaas, 2017).

In Laminated Veneer Lumber (LVL), either the fibre direction of all the veneers is parallel to the longitudinal direction of the LVL or predominantly parallel and slightly (up to 25%) perpendicular to the longitudinal direction, see Figure 37 left. As a result, LVL is applied mostly for more or less linear construction elements. It can be used as load-bearing sheathing or as a bar-shaped member (Blaß & Sandhaas, 2017; Herzog et al., 2004).

Parallel Strand Lumber (PSL) is a newer invention. It is produced by waxing and pressing strips of veneer on top of each other, resulting in unlimited length beams. There is no general technical approval yet for PSL, but it has high strength values, increased stiffness and dimensional tolerance, making it applicable for beams, structural members, purlins, columns and trusses (Blaß & Sandhaas, 2017).

4.1.3.2 Wood-based Panels made of Chips (LSL, OSB & Particleboard)

These type of panels are produced by pressing multiple layers of wood chips or strands together. Laminated Strand Lumber (LSL) is very similar to LVL and PSL, except for that in the production of LSL, small veneer trim strands are used instead of sheets or long strips of veneer. Depending on the width of the panels, a larger proportion of the trim strands are in a perpendicular board direction for wider panels. The resulting panels are of high strength (Blaß & Sandhaas, 2017).

Oriented Strand Board (OSB) is made using longitudinal strands, which are primarily placed parallel to the board surface, see Figure 37 middle. Especially in the outer layers, the strands are in parallel direction and in the layers in between, it can vary, resulting in different properties. As a result, OSB panels have a high bending strength in the longitudinal direction, but a significantly lower strength in perpendicular direction (Blaß & Sandhaas, 2017; Herzog et al., 2004).

In particleboards, relatively small wood chips are used, which are sprayed with adhesive on a level underlay and then pressed together. The particles are preferably oriented parallel to the panel plane, resulting in advantageous in-plane tensile and compressive strengths, but low tensile strengths for stresses perpendicular to the plane. The boards are often used as structural floor, roof decking or structural wall sheathing (Blaß & Sandhaas, 2017; Herzog et al., 2004).

4.1.3.3 Wood-based Panels, Mineral Bonded

Mineral bonded boards are used as sheathing material in timber frame constructions for dry lining and fire protection. Different types of these panels exist, see Figure 37 right for example, but they all use a certain type of wood particle combined with cement or gypsum as binder (Blaß & Sandhaas, 2017).



Figure 37. Left: LVL (Herzog et al., 2004), Middle: OSB (Herzog et al., 2004; Structural Board Association, 2004), right: mineral board (Herzog et al., 2004)

4.2 Structural Components

A timber structure is build-up of different structural components. In a multi-storey car park, the main structural components are beams, floors and columns. Each variant has its own characteristics, making it optimal for use in a certain span or under certain support conditions. Therefore, an overview of the relevant different types is given.

4.2.1 Beams

Timber beams are linear elements, supporting slabs and roof elements, transferring their loads to the supports. As a result of their function, beams are subjected to bending loads. Depending on parameters like the span of and load on the element, several beam types can be used (Kaufmann et al., 2018).

4.2.1.1 Solid Timber Beams

For limited spans up to around 6 metres, solid timber beams can be used (Soons et al., 2014). Their cross-section and length are limited by the logs that have been felled. Through the use of finger joints, longer beams can be made. Cross-sections can be enlarged by gluing two or three beams on top of each other (Kaufmann et al., 2018).

4.2.1.2 I-beams & Box Beams

I-beams and box beams are light weight, making them easy to handle on the construction site. The stronger flanges take up to tension and compression forces in the element. Box beams as a result of their geometry provide better resistance against lateral buckling compared to I-beams (Anctil, 2021; Kaufmann et al., 2018). Their maximum spans are limited to around 8 metres (Elliott Brothers Ltd., n.d.; Forest & Wood Products Australia Limited, 2008).

4.2.1.3 Glulam, LVL & PSL Beams

Glulam, LVL and PSL beams can be used for larger spans and loads. These wood engineered elements create the possibility for larger and more homogeneous cross-sections, which are stronger and much less likely to warp, twist, bow or shrink than solid wood beams (Anctil, 2021; Kaufmann et al., 2018). They can span up to 30 metres for a beam with a continuous cross-section and by varying the size of the cross-section like with saddle roofs, spans up to 40 metres can be realised (Soons et al., 2014).

4.2.1.4 Trusses

For larger spans, truss beams can be applied. The members of a truss are subjected mainly to normal forces and as a result, it is possible to subject the lower flange by a steel cable, resulting in an under

stressed beam (Kaufmann et al., 2018). Trusses made of solid wood can span up to 25 metres. When engineered wood elements are used, they can span up to 80 metres (Soons et al., 2014).

4.2.1.5 Hybrid Steel Timber Beams

A lot of research is conducted on the use of hybrid steel timber beams. A big advantage of such hybrid elements are their reduced depths as a result of increased load bearing capacity. Such elements still have a relatively low self-weight and as the steel elements can be encased in timber, they are protected against fire loads and stabilized for buckling (Tavoussi, Winter, Pixner, & Kist, 2010; Winter, Tavoussi, Riola-Parada, & Bradley, 2016). Timber beams can also be prestressed using different techniques. The prestressing results in an increased strength, giving the beams a higher load bearing capacity and reducing deformation. Test results show an improvement of up to 885% in SLS and 286% in ULS (de Lima, Costa, & Rodrigues, 2018). Hybrid steel timber beams therefore show significant potential for use in long span elements for which reduced depth is important.

4.2.1.6 Overview

An overview of the approximate possible spans for the different types of timber beams can be found in Table 9



Table 9. Spans for different types of timber beams (Kaufmann et al., 2018)

4.2.2 Floors

Timber floors are panel shaped elements, which directly support the imposed live loads by users on them. Floor elements can carry loads in one or two directions, depending on the type of floor system used. The elements are generally supported by beams to which they transfer their loads. Depending on parameters like the span of and load on the element, different types of floor systems can be used.

4.2.2.1 Dowel Laminated Timber Floor

A dowel laminated timber floor consists out of a number of solid wood planks joined together with the wood fibres running parallel with its span direction, resulting in a load-bearing solid wood structural component. It is a one-way load-bearing component in which each board carries the proportion of load

that is placed on it between its supports. For this reason, it requires linear supports. To provide some cohesion between the planks and spread loads, adjacent boards are joined to each other by nailing, gluing or use of dowels. When dowels are used, the resulting component is not rigid enough to function as a plate, bracing the building. As a solution, the dowel laminated component can be nailed or screwed to a suitable wood-based material panel like OSB. The maximum thickness of this component is limited to around 240 mm as a result of the maximum board width, also limiting its maximum span distance. (Kaufmann et al., 2018)

4.2.2.2 Beam Floor

A beam floor is build-up of a series of beams covering the primary span on which panels or boards span the distance between the beams. They are often a very cost-effective solution for spans of 4 to 5 meters. For both the beams and panels or boards, different types of products can be used. As a result, many different spacings of the beams and spans can be achieved. OSB panels for example can span a smaller distance than CLT or LVL panels of different thicknesses. The materials used, their dimensions and the loads the floor has to carry make for a puzzle for which an optimal configuration has to be determined. The beams and plates can be glued or screwed to each other, but for long spans and large load areas, gluing is required. A rigid plate component can be formed by screwing suitable wood-based material sheeting to the beam floor component. (Kaufmann et al., 2018)

4.2.2.3 Box Floor

Box floors are somewhat the hollow core slabs of timber. These prefabricated lightweight floor slabs consist out of ribs combined with top and bottom planking and edge beams to result in a very high-performance planar support structure spanning in one direction. The planking functions both as load-bearing in composite action while providing dimensional stability at the same time. The planks can be connected to the ribs using glue, screws or carpenters joints. The ribs are braced by the planks and can therefore be very slender. Often a relatively small spacing of 40-70 cm is used. The use of box floors minimizes the thickness of a floor and are used for medium and long spans between 5 and 20 meters, although longer spans are possible. They can be supported by either linear supports or supports at various points and also cantilevering is possible. Downside of these elements is that they are complex and costly to produce. (Kaufmann et al., 2018)

4.2.2.4 CLT Floor

CLT panels can be used as floor elements. Because of their homogenous properties, they can carry load in two directions, can be supported by both linear supports and supports at various points and provide possibilities for cantilevers. CLT floors are limited by transport dimensions and generally, panels of 4 metres wide and up to 22 meters long are used with varying thickness (60 – 400 mm) depending on the span and load. CLT elements are also very rigid plates and therefore suitable to brace buildings. Multiple elements can be joined to each other by butt joints, overlapping, top joints or tongue-and-groove joints. (Kaufmann et al., 2018)

4.2.2.5 LVL Floor

LVL floor elements are homogeneous, although as a result of the direction of the layers fibres, there is a clear main span direction. It is possible to produce LVL with some cross layers, in which case the elements can also accommodate some loads in the other direction. In general, these panels require linear supports, but if cross layers are used, it is also possible to use supports at various points. Depending on the thickness, loads and support conditions, spans of up 4 metres can be achieved with a single LVL panel (Hakkarainen et al., 2019). It is also possible to stack multiple LVL panels on top of each other, resulting in possible spans of up to 19 m (Stora Enso, 2022d). As LVL elements are rigid plates, they are very effective at bracing buildings. (Kaufmann et al., 2018)

4.2.2.6 Composite Timber-concrete Slab

Composite timber-concrete slabs are often used for medium to long spans. Advantages of these elements are their improved structural performance, sound proofing, fire safety characteristics and reducing effect on vibrations as a result of their higher self-weight. The elements are made of a layer of concrete (6 - 12 cm) in compression and a layer of timber in tension which are connected to each other by various means to realise composite action. The type of timber material used can vary from a beam floor to CLT. These slabs are used in single spans up to 10 metres. (Kaufmann et al., 2018)

4.2.2.7 Overview

An overview of the different characteristics of the different types of timber floors can be found in Table 10.

Table 10.	Characteristics	s of different	types	of timber	floors
-----------	-----------------	----------------	-------	-----------	--------



4.2.3 Columns

Columns are linear members, mostly in compression, transferring load in vertical direction. Timber columns can be made of various materials like sawn timber, glulam, CLT and LVL. Three different types of columns in timber can be distinguished: solid wood column, spaced column and box column, see Figure 38 (Fridley, 1999). Advantage of the latter types is their reduced weight, while placing their material around the periphery of the cross-section, making it a more effective cross-section against buckling loads (Harries, Petrou, & Brooks, 2000).



Figure 38. Types of timber columns: a) solid wood, b) spaced column, c) box column (Fridley, 1999)

4.3 Connection Types

Connections in a timber structure are, although they make up only a small part of the structure, one of the most important parts of it. The primary function of a connection is to transfer loads from one element to another. Many other aspects however also play an important role in the choice for a certain method and type of connection. The type of connection used influences the appearance of a structure, how elements are fabricated, connected and erected and with which ease. Furthermore, it determines how cost-efficient a design is and it can have an impact on fire-safety performance (Augustin, Bell, Kuklik, Lokaj, & Premrov, 2008; Herzog et al., 2004). Joints in timber constructions are generally weaker compared to the members which are being connected to each other and often have a semi-rigid character. They can be divided into three different categories: glued joints, carpentry joints and mechanical joints (Blaß & Sandhaas, 2017).

4.3.1 Glued Joints

Like engineered wood products are produced using adhesives, it is also possible to make glued connections between different construction elements. Finger joints are often used in glulam frame construction to form a rigid connection, see Figure 39 left (Herzog et al., 2004). Another glued type of connection used in mostly glulam or LVL constructions are glued-in rods. The rods are placed in or around connections in curved and notched beams to prevent cracks due to tensile stresses perpendicular to the grain or as a way to transmit forces in a construction or part of it, see Figure 39 right. These glued-in rod joints can transmit significant concentrated loads, result in a very rigid joint when exposed to axial load and have effective fire resistance properties (Blaß & Sandhaas, 2017). Glued joints result in rigid connections that are not suitable to dismantle. They are also labour-intensive and as a result a costly option (Herzog et al., 2004).



Figure 39. Left: Finger-jointed frame corner (Herzog et al., 2004). Right: Glued-in rods in a clamped column end joint and in a moment-resisting joint (Blaß & Sandhaas, 2017).

4.3.2 Carpentry Joints

Carpentry joints, also called contact joints, are joints in which forces are transmitted solely by contact and, where applicable, friction. They are semi-rigid, but often have significant moment-resisting capacity (Harte & Dietsch, 2015). Sometimes, wooden or metal fasteners are used to secure the members into position, but these may also contribute in the transmission of force. These joints are used to extend timber elements in a direction parallel to the grain or as transverse joints (Blaß & Sandhaas, 2017). Disadvantage of these type of joints is the weakening of the members. Due to severe shearing and eccentricity effects, only relatively low loads can be carried (Herzog et al., 2004). Advantages of these joints is that no glue or steel is required. The connections are easy to assemble and can also easily be dismantled. Four types of carpentry joints are distinguished (see Figure 40): half-lap joints, step joints, mortise and tenon joints and cogged joints. A detailed description of every type of carpentry joint can be found in appendix C.



Figure 40. Different types of carpentry joints (Harte & Dietsch, 2015).

4.3.3 Mechanical Joints

Mechanical joints are engineered connections that in general consist of steel or metal elements, joining two members together. They can be clustered into two groups: dowel-type fasteners and surface-type fasteners. A characteristic of the used mechanical fasteners and dowel-type fasteners in particular, is they are ductile and the mechanical joints undergo deformations. Therefore, mechanical joints are always semi-rigid joints, although correct configurations of the fasteners can improve the rigidity of the joint. Advantages of these semi-rigid joints is their ability to deform before failure, as plastic deformation or creep may result in load redistribution, relieving a highly stressed area. To prevent a brittle failure of the joint, the timber might require strengthening by a metal plate. (Blaß & Sandhaas, 2017)

Mechanical joints require special attention. The metal elements in mechanical joints can be prone to fire and corrosion and should be protected. To prevent cracks from swelling and shrinkage, fasteners should be clustered on one side of the connected member and if this is not possible, reinforcements should be applied (Blaß & Sandhaas, 2017). Eccentricities which may result from the type of fasteners used, layout of the joint or layout of the structural system, should be prevented as much as possible as this results in bending moments in the members (Herzog et al., 2004). Cracking and splitting of the wood has to be prevented by proper spacing between the fasteners and distance to the edges of the member. If required, predrilling for the fasteners should be applied. Last of all, to prevent overloading due to excessive differences in stiffness, gluing and mechanical fasteners must not be combined (Blaß & Sandhaas, 2017; Herzog et al., 2004).

4.3.3.1 Dowel-type Fasteners

The load transfer using dowel-type fasteners involves both bending and tensile stresses in the fasteners and embedment and shear stresses in the timber along the shank of the dowel (Augustin et al., 2008). Three groups of dowel-type fasteners can be identified: nails and staples, bolts, dowels and threaded rods and screws, see Figure 41. These type of fasteners are often used both in joints directly between two timber members and between two timber members with additional steel plates, see Figure 41 right. Finally, they can also be used to create a connection with cold-formed steel connectors. A detailed overview of the different dowel-type fasteners can be found in appendix C.



Figure 41. Left: Nails & staples. Middle-left: Bolts & dowel. Middle-right: Screws. (Herzog et al., 2004) Right: Dowels and steel plates including dowel hinged connection.

4.3.3.2 Surface-type Fasteners

With surface-type fasteners, forces are spread over the surface area of the timber. Transfer of loads is achieved primarily by a large bearing area at the surface of the members (Augustin et al., 2008). Two categories of surface-type fasteners can be distinguished: connectors and punched metal plate fasteners, see Figure 42. Within the group of connectors, multiple options exist, but all make use of a fastener like a bolt in combination with a connector, which takes up the task of transferring the loads. With punched metal plates, the plate has both the function of transferring loads and fastening the connecting element. A detailed description of the various connectors and punched metal plates can be found in appendix C.



Figure 42. Left: double-sided split ring connector. Middle-left: single-sided shear plate connector. Middle-right: double- and single-sided toothed-plate connector. Right: punched metal plate joint. (Blaß & Sandhaas, 2017; Herzog et al., 2004).

4.3.4 Comparison

All of the different types of connections have their advantages and disadvantages. In Table 11, a comparison between the different connections is made on seven characteristics. A further detailed explanation of the comparison can be found in appendix C. For a temporary car park, the connections have to be reversible. As a result, glued joints cannot be used. Also nails and staples are more difficult to disassemble. Carpentry joints are a good option for joints, which do not require a very stiff connection and have rather small loads. In case higher load bearing capacities or stiffer joints are required, mechanical joints using either dowel-type fasteners or connectors are a better option.

Table 11.	Comparison	of different	types of	connections.
-----------	------------	--------------	----------	--------------

[Connec	tion type	Stiffness	Loadbearing bir capacity	× Types of forces in plane ∧	Ease of assemply	Dismantability for reuse	Poo Doo Doo Doo Doo Doo Doo Doo Doo Doo	Costs High
>		Scarf joint	Low	Low	V	Easy	Easy	Good	Medium
ts t		Lap joint	Low	Low	N & V	Easy	Easy	Good	Medium
oin pe		Step joint	Low	High	N	Easy	Easy	Good	Medium
Ca		Mortise and tenon joint	Medium	Low	M, N & V	Easy	Easy	Good	Medium
		Nails & staples	Low	Medium	N & V	Easy	Medium	Medium	Low
Mechanical joints	type iers	Bolts, dowels & threaded rods	Medium to high	High	M, N & V	Easy	Easy	Medium	Low
	rel-i	Screws	Low	Low	N & V	Easy	Easy	Medium	Low
	Dow fas	Combination with steel plates or connectors	High	High	M, N & V	Easy	Easy	Low to medium	Medium
	face- pe	Connectors	Medium to high	High	M, N & V	Easy	Easy	Medium	Low
	Surf tyj faste	Punched metal plate fasteners	Medium	Medium	M, N & V	Easy	Easy	Low	Low
4.4 Durability

Timber is a natural building material, which can be part of the transition towards a sustainable building industry. However, to be able to fulfil a role in a sustainable and circular economy, the material should also be durable. As timber is a natural material, it can deteriorate very quickly if not protected from the elements (Anastasiades et al., 2020). Three challenges concerning the durability of timber can be distinguished: biological degradation (fungal attack), environmental degradation (weathering) and safety concern (fire hazard), see Figure 43 (Ayanleye, Udele, Nasir, Zhang, & Militz, 2022). Sufficient protective measure should be taken to prevent the degradation of the timber, while also environmental factors are of concern.

4.4.1 Risks

4.4.1.1 Biological Degradation *Fungal attack*



Figure 43. Durability challenges for timber (Ayanleye et al., 2022)

Micro-organisms like bacteria and fungi break down large wood cellulose molecules into smaller and simpler molecules, effectively breaking down the timber to be redirected into the materials cycle, see Figure 44 left. This natural process is part of the life cycle of wood, but should be prevented from occurring in a construction (Blaß & Sandhaas, 2017). For the microbial breakdown to occur, four basic requirements for growth should be fulfilled: wood as a food source, oxygen, water and appropriate temperatures (Kuklik et al., 2008). The optimal range of moisture for most types of fungi is between 30 and 60% and the temperature can vary between -2.5 and +40°C with an optimum between 19 and 31°C. It is not possible for wood-destroying fungi to grow when the wood moisture drops below 20% and they die at temperatures exceeding 60°C (Blaß & Sandhaas, 2017; Kuklik et al., 2008). In general, these conditions are all satisfied in forests, but usually not in buildings. Therefore, in buildings, usually protective measures are taken if required to keep the moisture content of the timber below 20%.

Insect attack

Dry-wood insects can infest semi-dry and dry timber and construction wood. Damage to the timber is mainly the result of the burrowing of larvae by the insects (see Figure 44 right), which can result in a dramatic reduction of the cross-section and load-bearing capacity of the wood. It is possible for larvae to develop with moisture contents of just 8-12%, meaning just keeping the timber dry by constructional measures is not enough. Therefore, it is important to prevent larvae from infesting the timber in the harvesting process by sufficient measures. Furthermore, growth of insects is encouraged by warm conditions, stimulating their development and reproduction (Blaß & Sandhaas, 2017).



Figure 44. Left: fungi rot (a. white partial, b. white full, c. brown), right: common furniture beetle (a & b. adult beetle, c. larvae) (Kuklik et al., 2008)

4.4.1.2 Environmental Degradation

Environmental degradation of timber is the result of weathering effects. Timber ages naturally by exterior factors like UV radiation, air, temperature and moisture changes, together with stress conditions. This results in irreversible changes in the appearance and properties of the timber, see Figure 45 (Kuklik et al., 2008).



Figure 45. Weathering effects on timber specimens (Niklewski, Brischke, Frühwald Hansson, & Meyer-Veltrup, 2018)

Weathering effect scan also result in shrinkage and swelling of timber elements, which can lead to cracks in the timber. Insects can lay eggs in such cracks, resulting in larvae and following degradation by insects attacks. Furthermore, weathering effects can also affect connections between timber elements made using metal fasteners. These fasteners can corrode, which can ultimately result in failure of the connection (Blaß & Sandhaas, 2017).

4.4.1.3 Safety Concerns

Unlike steel and concrete, wood is a combustible material, meaning it can contribute to the fire load of a building. As a result, timber buildings are often considered by the public to be a bigger risk in terms of fire safety. However, the risk of starting a fire in a building is not related to the material it is made of. It is related to aspects like improper electrical installations or human error (Kaufmann et al., 2018).

Wood as a building material is generally seen as a material with an acceptable ignition risk. For solid timber to ignite, surface temperatures exceeding 400 °C or contact with an ignition source of more than 300 °C is required. When wood is ignited, the speed of flame spreading is acceptably low for a combustible material. The characteristics of wood-based materials exposed to a fully developed fire are seen as favourable. The surface of the material, once ignited, will burn vigorously resulting in the formation of a charcoal layer. This layer, even more than wood itself, is a very poor thermal conductor. The result is that the charcoal layer forms an insulating layer around the unburnt wood, protecting it from the heat of the fire. Therefore, wood naturally actually protects itself from fire and thus has favourable characteristics under fire conditions (Blaß & Sandhaas, 2017; Kuklik et al., 2008)

4.4.2 Protective Measures

4.4.2.1 Natural Durability

Wood constitutes of its main molecules cellulose, hemicellulose and lignin, but also incorporates other constituents, which contribute to a natural durability of the wood (Kuklik et al., 2008). These constituents can make the wood more resistant against fungi and insect attacks. They vary significantly between different wood species. Therefore, to describe the natural durability of wood species, durability classes are assigned to the different species. For the natural durability against fungi attack, a number is given to the wood species and for natural durability against insects and wood pests in sea water, a letter is given, see Table 12. Generally speaking, tropical woods are more durable species and also European hardwoods have good durability characteristics. Sapwood is considered to be non-durable against fungi attacks (Blaß & Sandhaas, 2017).

TUDI	able 12. Natarar adrability classes				
Against fungi		Agai	nst insects and wood pests in sea water		
1	Very durable	D	Durable		
2	Durable	М	Moderately durable		
3	Moderately durable	S	Vulnerable		
4	Slightly durable	SH	Where heartwood is also classed as vulnerable		
5	Non-durable				

|--|

4.4.2.2 Service Classes

To determine how vulnerable elements of a timber structure are to durability risks, service classes are used, which describe the environmental conditions to which a timber element is exposed, see Table 13. Based on this service class together with the durability classes, one can determine whether the wood used has sufficient natural durability or if extra protective measures should be taken (Blaß & Sandhaas, 2017; Herzog et al., 2004).

Service		Moisture content /	General condition of use	Threat from				Leaching
Cla	SS	exposure		Insects	Fungi	Soft rot	Wood pests in the sea water	
0		Dry (permanently ≤ 20%), average relative humidity up to 85%	Wood or wood product under roof, not exposed to weathering or moisture, the risk of structural damage caused by insects can be excluded	No	No	No	No	No
1		Dry (permanently ≤ 20%), average relative humidity up to 85%	Wood or wood product under roof, not exposed to weathering or moisture	Yes	No	No	No	No
2		Occasionally humid (> 20%), average relative humidity over 85% or occasional humidification due to condensation	Wood or wood product under roof, not exposed to weathering, high relative humidity can lead to occasional but not permanent humidification	Yes	Yes	No	No	No
3	3.1	Occasionally humid (> 20%), accumulation of water in wood, even on a limited basis, not expected	Wood or wood product not under roof, exposed to weathering, but without permanent contact with soil or water, accumulation of water in wood, even on a limited basis, is not expected due to swift drying	Yes	Yes	No	No	Yes
	3.2	Frequently humid (> 20%), accumulation of water in wood, even on a limited basis, expected	Wood or wood product not under roof, exposed to weathering, but without permanent contact with soil or water, accumulation of water in wood, even on a limited basis, expected	Yes	Yes	No	No	Yes
4		Mainly to permanently humid (> 20%)	Wood or wood product in contact with earth or freshwater and hence mainly to permanently exposed to moisture leading to moderate to severe stress	Yes	Yes	Yes	No	Yes
5		Permanently humid (> 20%)	Wood or wood product permanently exposed to sea water	Yes	Yes	Yes	Yes	Yes

Table	13	Service	classes	for timber	elements	(Rlaß	R	Sandhaas	2017	')
Iable	10.	SEIVICE	6103353		cientento	Diais	x	Sanunaas,	2017	1

4.4.2.3 Constructional Measures

Constructional measures are the most important protective measures to prevent degradation of the wood of a building. By clever design, problematic changes in the moisture content of the timber components can be prevented. Such changes in moisture content can result in conditions suitable for fungi to destroy the timber of can result in swelling and shrinkage, leading to damaging deformations (Herzog et al., 2004).

Furthermore, central heating systems inside buildings result in optimal conditions for insects to grow. Good design (make structural barriers against termites or covering all sides of timber elements for example) can prevent insects from gaining access to concealed components, preventing attacks (Herzog et al., 2004; Kaufmann et al., 2018). Natural durability of the wood and using heartwood combined with regular inspection tends to further improve the protection against insects attacks. Again limiting changes in moisture content prevents crack formation, which can be a location for egg-laying (Blaß & Sandhaas, 2017).

The constructional measures taken result in a certain service class classification. Aim is to have a service class as low as possible (Herzog et al., 2004). The following protective constructional measures should be taken into account (Blaß & Sandhaas, 2017):

- Prevent humidification of wood as far as possible
- Provide swift water drainage and ventilation, if occasional humidification cannot be prevented
- Prevent direct water influx
- Protect end grain surfaces from water penetration
- Take condensation problems into consideration
- Prevent wood from exposure to moisture from contact with ground or areas, where snow can accumulate or water can splash
- Use woods from a sufficient natural durability class, if permanent humidification cannot be prevented
- Correct storage of timber elements during construction

4.4.2.4 Wood Preservatives

Like some wood species naturally have better durability, it is also possible to improve the durability of wood using chemical wood preservatives containing pesticides (Herzog et al., 2004). Timber elements that are completely exposed to continuous changing high moisture contents cannot be protected by chemicals. In such case, the chemicals can only delay the degradation at best. However, chemicals can be used to improve the characteristics of the wood for situations where occasional humidification occurs (Kaufmann et al., 2018). Furthermore, paints and coatings can for example be a means of weathering protection, mitigating precipitation, high humidity, UV rays and mechanical stresses (Blaß & Sandhaas, 2017). Several methods like brushing, spraying, dip-treatment and pressure treatment are used, which should result in a uniform distribution and adequate retention of the preservatives by the timber. Although preservatives in the past have proven to be successful, in recent years, newer preservatives have been developed, which are more environmentally friendly. Also concerns about the health of residents and users of the buildings resulted in development of new preservatives (Ayanleye et al., 2022).

4.4.2.5 Wood Modification

The aim of wood modification is to reduce the wood's affinity for moisture and block the cell walls by penetrating the nanopores in the cell wall by chemical and/or thermal modification. As a result, the wood can absorb less water and thus becomes less vulnerable to biodegrading, which requires moisture (Ayanleye et al., 2022; Blaß & Sandhaas, 2017). Different thermal treatments have been developed like the exposure of wood to high temperatures (160-230°C) under reduced oxygen conditions or using different pressure regimes (vacuum, non-pressurized, pressurized) or shield gasses. As a result of these thermal treatments, the physical and chemical structure of the timber is altered. Such treatments have proven to increase the wood's resistance against weathering influences. The downside however is the reduction of certain strength properties of the timber. Thermally modified wood is also more vulnerable to insect attacks. Chemical treatment has proven to be able to improve the wood's resistance against both biological and environmental degradation, without reducing its strength properties. Downside of chemical treatment is similar with using wood preservatives. Their effects on the environment and health of residents is an important concern (Ayanleye et al., 2022).

4.4.2.6 Corrosion

Wood is normally not attacked by acids and bases. If metallic fasteners are used, these however are vulnerable to corrosion. Different protective measures can be taken like painting or coating or the use of special stainless steel fasteners (Blaß & Sandhaas, 2017).

4.4.2.7 Fire Safety

A timber building should meet the same fire safety regulations as buildings in steel and concrete. This means its structural elements should remain stable (R criterion) and the structural elements enclosing a space should prevent the passage of smoke and gasses (E criterion) and heat (I criterion) (Kaufmann et al., 2018). As wood and especially the charcoal layer in case of fire are poor heat conductors, this favours criterion I. Considering the R and E criterion, timber structural elements should not lose their load-bearing capacity or undergo too large deformations under fire conditions (Kuklik et al., 2008).

To verify the load-bearing capacity of structural elements under fire conditions, two methods can be used: reduced cross-section method (RCSM) or reduced properties method (RPM) (Blaß & Sandhaas, 2017). The RCSM method is a simplified method, in which a remaining unburnt cross-section is considered to depict the fire resistance duration of a structural member. Depending on the applied protective measures and required fire resistance duration, an effective charring depth is calculated. The cross-section of the structural member is reduced with this depth and the remaining effective cross-section should be able to carry the load on the element under fire conditions, see Figure 46. The RPM method is a more accurate method, which works similar like the RCSM method, however this time, the mechanical properties of the remaining cross-section are reduced.



Figure 46. Left: burned wooden element with charcoal layer (Kaufmann et al., 2018), right: reduced cross-section (Blaß & Sandhaas, 2017)

It is also possible to further improve the resistance of timber elements to fire. A coating can be applied to the wood or it can be impregnated with a fire retardant. This does not change the wood into a non-combustible material, however it can make it more difficult to ignite. A downside of such chemicals are often its bad environmental properties (Blaß & Sandhaas, 2017).

Joints with metal fasteners are of concern when exposed to fire. The metal has a much higher thermal conductivity than wood, meaning it can effectively transmit heat to the inside of the members, causing both embedment strength and withdrawal capacity to decline. Furthermore, the yield strength of the metal declines very rapidly with increasing temperatures. Metal fasteners in joints therefore should be protected. Unprotected joints, but with side members made of timber, have a fire resistance duration of 15 minutes. This time can be increased by increasing the member thickness and the end and edge distances to a maximum of 30 minutes. Using the reduced stresses method, it is possible to realise fire resistance durations of up to 40 minutes for unprotected joints and in combination with other methods, this can be increased to 60 minutes. Steel plates with unprotected edges can withstand fire up to 60 minutes, depending on their thickness. It is also possible to protect joints by cladding them with members made of wood (boards or plugs) (Blaß & Sandhaas, 2017).

Like joints, structural elements can also be protected by cladding them with wood or wood-based materials offering resistance to heat, which is also applied with steel structures. Similarly, other general fire safety measures like sprinkler systems, smoke detection and ventilation mechanisms can be applied in timber buildings. These measures are more general fire safety measures and are not further discussed here.

4.5 Pavement Surfacing

Using timber in buildings often has the advantage that finishing of the interior is no longer required, as timber itself already has the required aesthetics. However, in a car park, timber decks are not directly suitable to be driven on and require an appropriate surfacing system. As timber car parks are very new, only a very limited number of car parks are built in timber and research on timber decks in car parks is very limited as well. However, timber bridges have been built for many years and research on timber decks for bridges is much more extensive. Differences can be found between timber bridges and parking decks, but it can be a good starting point as a reference for the timber decks of a car park.

4.5.1 Function

In case timber material is used for the decks of a car park, a suitable surfacing system should be applied to the timber, fulfilling the following functions: 1) provide a smooth and safe surface for vehicle traffic and furthermore improve the poor skid resistance of timber decks (Ritter, 1990; Schiere, Müller, Bueche, & Angst, 2022), 2) protect the deck from vehicle loads (Haynes, Coleri, & Estaji, 2019; Ritter, 1990), and 3) protect the deck from the elements (rain, sun, snow etc.), to prevent biological degradation (Haynes et al., 2019; Ritter, 1990; Schiere et al., 2022).

Timber decks that are made of planks are prone to varying deflections when vehicles drive over them. In such case, the surfacing system should help to minimize such varying deflections and provide a smooth surface (Ritter, 1990). For timber panels such as CLT, these problems are limited to the locations of connection between the various panels. Furthermore, timber can be very slippery, especially when wet. To provide a safe surface, the surfacing system should increase the skid resistance of the decks (Ritter, 1990; Schiere et al., 2022).

The weight of vehicles is transferred to the construction only through the small patches of rubber from the wheels, resting on the parking decks. As a result, relatively large loads are placed on only a small area of the deck. Timber can be vulnerable to such large localized loads due to the anatomy of wood (Blaß & Sandhaas, 2017). The surfacing system can help to spread the loads over a larger area, protecting the timber. Furthermore, at the locations of joints between different panels, again timber is vulnerable for loads from vehicles passing over these joints.

As discussed in paragraph 4.4, building with timber comes with several durability challenges. Timber is very vulnerable for moisture and therefore requires protection from direct exposure to the elements (Haynes et al., 2019). The surfacing system provides this protection. It should seal off the timber surface, preventing water from entering the timber (Scharmacher, Müller, & Brunner, 2014).

4.5.2 Surfacing Systems With or Without Shear Connection

Many different types of pavement surfaces have been developed. In their layup however, one can choose for a layup in which the surfacing system is fully connected to the timber deck (full shear connection) or for a layup in which the surfacing system floats and is bolted to the timber deck (without shear connection) (Scharmacher et al., 2014). Difference between these types of layups is how shear forces are transferred from the surfacing layer to the timber deck. Most pavement surfaces can be used with both types of layups.

In general, starting on top of the timber deck, the surfacing system is build-up out of a surface coating (for systems with shear connection) or separation layer (for systems without shear connection), sealant, protective layer and top layer, see Figure 47. The surface coating acts as the "glue" between the bridge deck and sealant (Scharmacher et al., 2014). In case of a system without shear connection, a separation layer is used (glass-fibre mat or oil-paper), to which the sealant can attach (Schiere et al., 2022). The separation layer is connected to the deck through circular brackets (Haynes et al., 2019). The sealant layer seals of the timber element and provides a watertight barrier, protecting the timber. Furthermore, in case asphalt is used, the sealant layer partially protects the timber from the heat of the pouring process of the asphalt (Haynes et al., 2019). The protective layer acts as an intermediate layer, protecting the sealant from vehicle impacts and can help to spread vehicle loads. Finally, the top layer acts as the driveway with proper skid resistance (Scharmacher et al., 2014).



Figure 47. Left: Sketch of a system with a bonded shear connection between sealant and deck. Right: Sketch of a system without a shear connection between sealant and deck (Scharmacher et al., 2014).

4.5.2.1 With Shear Connection

Advantageous of systems with shear connection, see Figure 48, is that traffic loads are transmitted directly to the deck, including large breaking and acceleration loads. As a result, they can be loaded with heavier traffic and allow for larger slopes (> 6%) (Schiere et al., 2022). If asphalt is used as top layer, its fatigue life is improved with a shear connection system.

The shear capacity of the joint between deck and sealant layer depends on the combination of bonding agent and waterproofing membrane (sealant layer). Research has shown the shear capacity is not influenced by the type of deck and timber performs comparable to concrete and steel decks. The type of failure mechanism (brittle/ductile) is also solely dependent on the type of adhesive used and is not influenced by the material of the deck (Scharmacher et al., 2014; Schiere et al., 2022).



Figure 48. Test specimen of system with shear connection (Haynes et al., 2019)

Blistering

A disadvantage of the use of systems with shear connection is that blistering can occur in case asphalt is used as top layers. Blistering is the result of water vapour, rising from the timber and accumulating directly at the bottom face of the sealant layer, when hot asphalt is poured on the deck. As a result, the accumulated water vapour can cause partial separation of the sealant from the timber deck, see Figure 49, and in some cases it can penetrate the sealant layer, which causes "blisters" in the asphalt (Scharmacher et al., 2014). If the sealant is penetrated, the watertight barrier is no longer intact and water can come in contact with the timber. Furthermore, as a result of the partial separation, the shear capacity of the plane between the sealant and deck is reduced by approximately 10 - 15% (Scharmacher et al., 2014). Since timber contains a relatively large amount of water, timber decks are more prone to blistering compared to steel or concrete decks.



Figure 49. Test specimen of system with shear connection experiencing blistering (Scharmarcher, Müller, & Brunner, 2014)

To prevent or minimize the risk of blistering, several measures should be taken (Scharmacher et al., 2014; Schiere et al., 2022):

- The pouring temperature of asphalt should be below 200 °C;
- The thickness of the protective asphalt layer should not exceed 25 mm;
- The hot asphalt should be placed by hand instead of by a road finishing machine;
- When choosing the type of sealant layer, one should keep its effect on blistering in mind;
- The timber panels should have good surface quality.

4.5.2.2 Without Shear Connection

Advantageous of systems without a shear connection, see Figure 50, is the increased moisture resistance, making blistering no longer a risk. It also eliminates cracking of the pavement at joints between different (glulam) joints (Ritter, 1990). Since no full connection exists between the timber deck and sealant layer, the timber can breathe during the use phase (Haynes et al., 2019).



Figure 50. Test specimen of system without shear connection (Haynes et al., 2019)

Disadvantageous of systems without a shear connection is its lower shear capacity compared to a system with shear connection, although this is still more than sufficient (Haynes et al., 2019). Another disadvantage is the risk of development of surface waves as a result of breaking and acceleration (Scharmacher et al., 2014). Such a system is therefore not applicable in locations were frequent and/or heavy acceleration or breaking occurs. As a result, in Germany, all road bridges with timber decks are required to use a system with shear connection (Scharmacher et al., 2014).

4.5.3 Types of pavement surfaces

4.5.3.1 Asphalt

Asphalt is one of the most used pavement surfaces in the world. It is often used on (timber) bridge decks and current existing timber car parks also use asphalt as pavement surface (HK Architecten, 2021; Zaugg, 2018). Two types of asphalt pavement surfaces exist: mastic asphalt and rolled asphalt. Whereas mastic asphalt is virtually waterproof, rolled asphalt is fairly permeable and as a result not fully waterproof. Furthermore, rolled asphalt requires large compaction energy, which can damage timber decks (Haynes et al., 2019). Based on these two aspects, rolled asphalt is considered not applicable to timber parking decks.

Mastic asphalt has a high heat resistance, flexibility and fatigue strength. As a result of its waterproof properties, it protects the timber decks and its surface is skid resistant. As a result of the thickness of the asphalt layers, it spreads vehicle loads over a larger area, significantly reducing deformations (Haynes et al., 2019). Disadvantageous is its larger self-weight as a result of the thickness of the layer. Its placement requires high temperatures, which can result in blistering (Haynes et al., 2019). With good maintenance, it can service over 20 years (Schiere et al., 2022).

The asphalt pavement is build-up of a surface coating or separation layer, followed by a bitumen membrane, a first asphalt protection layer and finally an asphalt top layer, see Figure 51.



Figure 51. Timber deck with mastic asphalt pavement surface (Haynes et al., 2019)

4.5.3.2 Coating with Aggregate

Many different types of bridges (steel/concrete/timber) are fitted with a wearing layer, which is based on a coating layer into which aggregate is pressed, see Figure 52. First, a primer is applied to the timber deck for proper bonding with the membrane layer, which ensures watertightness. Next, a basecoat layer (epoxy or polyurethane often used) is poured. Aggregate is broadcasted over this layer to add friction to the driving surface. Finally, a topcoat is applied (Haynes et al., 2019).

This type of coating provides a watertight, flexible surface, which is highly resistant against fatigue cracking (Haynes et al., 2019). The system furthermore resists heat, chemicals and de-icing salts. It has a relatively small thickness, resulting in a low self-weight compared to asphalt (approximately 15x lighter) (HIM, 2021). Disadvantageous is the very limited ability to spread loads.



Figure 52. Polyurethane wearing course (HIM, 2021)

4.5.4 Modular and Temporary Structure

The objective of this thesis is concerned with a temporary and modular multi-storey car park. As a result, one of the large challenges for the design of the construction is that all elements should be easy to install and disassemble. Creating a monolithic parking deck by pouring for example an asphalt layer over all different floor panels hinders the disassembly process and is therefore undesirable. Starting point is that the pavement surface should be applied only to a single floor element. The different floor elements are placed against each other, resulting in a joint between them. This joint comes with new challenges for the detailing of the floor elements and the surfacing system.

4.5.4.1 Expansion Joints in Timber Bridge Decks as Reference

Schiere et al. (2022) studied details of timber bridges with asphalt wearing surfaces. Amongst the details were expansion joints, side edges along curbs and details around drains. Expansion joints are considered to be most similar to the joints between the different floor elements in a timber car park. They are highly loaded by traffic, causing impact loads on the edges of the joints. Translating the requirements of expansion joints in timber bridges to the floor joint in a timber car park, they have to fulfil requirements considering sufficient load bearing strength, fatigue resistance, deformation and elasticity, low noise emission and good premises for inspection and repair. Furthermore, the joints in the floors of car parks should be closed, to prevent water from dripping to lower decks.

Schiere et al. (2022) presented a set of recommendations for expansion joints of timber bridges. Relevant recommendations for the joints between the floor elements of a timber car park are as followed:

- The structure on both sides of the joint should have a similar vertical stiffness, to prevent large relative deformations between the two elements when a vehicle crosses the joint.
- The maximum vertical deformation should be limited.

- If timber decks do not have sufficient stiffness, they can be reinforced with steel beams.
- Proper sealing of the timber around the joints should be ensured.
- Ventilation in the vicinity of the joint should be encouraged to improve the drying capacity of the timber elements.

4.5.4.2 Effect of Joints on Type of Surface Pavement

It is clear that the joints between the different floor panels result in important requirements for the detailing of the timber deck panels around this area. However, this also affects the surface pavement which is placed on top of the timber panels. Similar to the timber, the pavement surface should also be able to withstand the impact loads of vehicles, have sufficient fatigue resistance, deformation and elastic capacity. An asphalt layer has a larger thickness and might be vulnerable to impact loads. A system with a coating with aggregate is thinner and might be less vulnerable, but may cause the timber panel to take more impact. Additional research is advised for proper detailing of the joints between floor elements and what the impact of different pavement surfaces is.

4.5.5 Choice of Pavement System

In a car park, vehicles are continuously braking, accelerating and turning to navigate the car park. Although the speeds are low, this results in continuous shear loading of the pavement surface and might lead to development of surface waves. Furthermore, ramps in a car park can exceed a slope of 6%. Considering these aspects, a pavement system with shear connection is chosen.

For the type of surface pavement, both systems of asphalt and a coating with aggregate can be applied, although more research is required considering the joints between the floor elements. Based on the literature research, both systems fulfil the requirements, but the system of a coating with aggregate is much lighter. Therefore, this system is chosen for the development of the temporary multi-storey car park of this thesis.

4.6 Timber Car Parks

Although timber has been used in the construction industry for ages, only recently the material has again become more popular for multi-storey construction and new height records are quickly following each other. The use of timber in car parks however has been quite limited. The world's first significant timber car park has only been constructed very recently in 2018. Recently, several designs for multi-storey car parks in timber have been developed, but many of them are yet to be realised. It seems like the industry is still somewhat sceptical about the realisation of a multi-storey car park in timber. Following are a realised and proposed multi-storey car park project in timber, which function as reference projects.

4.6.1 Timber-built Car Park Studen, Switzerland

The world's first timber-built car park has been realised in Studen, Switzerland. The car park provides 1800 parking stalls for a specialist in automotive transport and logistics, COTRA Autotransport. One of the main reasons to choose for timber as the main structural material were weight issues with the foundation. A solid concrete or hybrid steel-concrete structure proved to be too heavy for the site-conditions. A lightweight three-storey high construction was required and an all-timber construction proved to be feasible and even economical (Zaugg, 2018).

4.6.1.1 Structural Concept

The car park is a modular construction, which has no minimal or maximum size. It has a grid of 5,1 x 15 m and a level height of 2,15 m with bays of 5 x 2,55 m, see Figure 53. The parking deck is column free, allowing for problem-free parking and exit.

Floor system

The inter-storey floor slabs are made of five-ply cross-laminated timber, covered with two layers of mastic asphalt with a separation layer and sealing membrane. These form the carriageway and also act as a fire compartment-forming element. The panels have a thickness of 140 mm and are 2,5 m wide. The floor slab is a multi-span panel with individual spans of 5,1 m.



Figure 53. Left: overview of structure of timber-built car park Studen, right: dimension of grid (Zaugg, 2018)

Supporting beams

The CLT floor slabs are supported by glulam slab joists of 200 x 960 mm at every 5,1 m, see Figure 54. The slab joists are coupled multi-span elements with individual spans of 15 m and coupled by articulated joints. Various strength classes are used for different elements, varying between GL24h, GL28k and GL32k.



Figure 54. Structure of all-timber car park (Zaugg, 2018)

Columns

The slab joists are supported by V-shaped columns, allowing for a smaller span width and enabling optimum preconditions to be created for horizontal force dispersion, see Figure 55. The horizontal member in the columns, supporting the slab joists, is made of BauBuche GL70, to be able to transfer the large loads perpendicular to the grain. The struts supporting this member are made of glulam GL24h and varying cross-sections of up to 200 x 360 mm. In the middle of the column is a vertical glulam GL24h member, finally transferring the loads to the strip foundation.



Figure 55. V-shaped columns (Zaugg, 2018)

Stability

The CLT aisle slabs function as shear panels, transferring horizontal loads. In longitudinal direction, the horizontal loads are transferred to the V-shaped pillars. Horizontal stability in the transverse direction of the structure is realised by steel tie bars forming x-bracings. To take-up the additional vertical forces from the wind bracing, the pillars for the wind loads are made of BauBuche GL70. An extra horizontal coupling between the slab joists is made using small glulam beams.

Joints

Most of the joints used in the design are carpentry joints. The struts in the columns transfer the loads via end-grain contact to the vertical pillars. Step joints are used (see appendix C for a detailed description), in which the members are kept in place by an additional bolt at the bottom, see Figure 55. The horizontal member in the V-column is joined by a half-lap joint with again a bolt securing the members.

The large slab joists are placed on top of the horizontal member of the V-shaped column. They are however also connected to the vertical pillar by a grid of dowels and two bolts, securing the connection, see Figure 55. The slab joists are connected to each other several meters away from the columns at the point of minimal bending moments, see Figure 56. This joint is made up of a single large bolt with steel plates attached to both members.

The joints between the different columns and attachment of the bracing members to the columns is done via slotted in steel plates, see Figure 56. The CLT floor slabs are placed on top of the slab joists, but it remains unclear what specific connection between the floor slab and the joist is made.



Figure 56. Different joints in structure (Zaugg, 2018)

Vibrations

Vibrational behaviour of a multi-storey car park in timber was unknown for a timber structure of this size. Therefore, no reference information or standard practice was available. An existing hybrid construction for a car park with steel support structure and CLT floor panels was used as a reference. Both this construction and the new timber design were modelled and compared to each other to come to an estimation for comfortable behaviour or the all-timber structure.

4.6.1.2 Durability

To protect the structure from weathering and other durability concerns, several measures have been taken. The first are the separation layer and sealing membrane between the mastic asphalt layers and the CLT. This should prevent water coming of vehicles from coming in touch with the wood. The floors have also been placed with a 2% pitch to provide sufficient run-off of the water.

The design choice has been made to use hot-dip galvanised steel façade pillars and stair towers instead of timber variants, see Figure 57. As these elements are located around the façade of the building, they are significantly more exposed to weathering conditions and therefore of more risk to degradation.



Figure 57. Steel elements in façade of car park (Zaugg, 2018)

4.6.1.3 Production and Assembly

The design of the car park has been made modular. The V-shaped column elements were assembled in the factory and transported to the site. At the construction site, assembly of the columns, slab joists and bracings was fast and easy by simple dowels and bolts. The assembly process was vertical, which means every time a segment of three storeys was constructed and afterwards sealed simultaneously. As a result, the entire assembly process was very fast.

The car park has been realised as a permanent structure. Although many connections are made using bolts and dowels, the parking deck is sealed, which makes it very difficult to disassemble without damage. In terms of reuse, the car park is therefore difficult to disassemble.

4.6.2 Construction System for Multi-storey Car Parks in BauBuche

At the Technical University of Munich, a new construction system for a multi-storey car park has been developed, which is made of beech laminated veneer lumber (BauBuche). As beech is much stronger and more stable than softwood and it is highly available in Germany, the use of BauBuche was very interesting to study. The design is a modular system, which can also be disassembled, making it a very interesting reference project. However, the project has not been realised and remains a design on paper, so we cannot draw conclusions on how it performs in practice (Technical University of Munich & Pollmeier Massivholz GmbH & Co.KG, 2015).

4.6.2.1 Structural Concept

The car park is a modular system made up of multiple hybrid parking modules. These parking modules have a column-free span of 16.5 m and are 2.5 m wide, see Figure 58. The bays are 5 by 2.5 m. Car parks of many sizes can be created by combining the parking, ramp and stairwell modules in different configurations, making it a very flexible system. It allows for different types of car park layouts like full storey or split-level and different types of ramps. The maximum height of the construction is in theory only limited by building regulations.

Floor system

The floors are made of 130 mm thick prefabricated reinforced concrete elements, spanning 2.5 m. These elements require no extra layers to drive on, resulting in a floor system that is fast in erection and easy to disassemble. The floors are constructed on an angle of 2% for proper drainage and a gutter is incorporated into the concrete plates. In vertical direction, the floors also act as a fire barrier.

Supporting beams

The concrete floor elements are supported by timber BauBuche GL75 beams of 240 x 600 mm, which are cambered. They span the entire parking deck and have a length of 16.5 m. As the design of the car park is a system of multiple parking modules, the beams are single-span elements.

Columns

The beams are supported by square columns made of BauBuche. They are 240 x 240 mm and again of quality GL75. The columns extend over the entire height of the parking level which is 2.93 m with a clearance height of 2.2 m.



Figure 58. Parking modules and combination of modules (Technical University of Munich & Pollmeier Massivholz GmbH & Co.KG, 2015)

Stability

Stability in longitudinal direction is realised through steel elements in the post planes. In transverse direction, the building is stiffened by reinforced concrete walls of the stairwells.

Joints

The pre-cast concrete deck elements are connected to the supporting timber beams using birdsmouth joints, which are a type of carpentry joint (see appendix C for a detailed description). The result is that the concrete element and timber beam form a rigid structure.

The columns are connected to each other using hollow steel profiles ($90 \times 50 \times 8 \text{ mm}$), to which a steel anchoring plate ($200 \times 200 \times 8 \text{ mm}$) is welded as well as a slotted plate ($150 \times 200 \times 10 \text{ mm}$), see Figure 59. The timber columns are in this way joined by the steel slotted plates using 12 mm steel dowels. The joint between the columns and the timber beams is filled with low-shrinkage expansive mortar.





4.6.2.2 Durability

Beech LVL elements are expected to last for more than 50 years, if proper installation, effective protection from the elements and regular inspections are ensured. Therefore, several measures are taken in the design to protect the timber elements. All the timber beams are coated for extra protection against moisture intake. Furthermore, the timber columns are separated from the concrete deck elements by expansive mortar. This prevents water collecting on the concrete surface to cause damage to the columns. Finally, on the top level, a roof is constructed which protects the rest of the structure against the elements.

4.6.2.3 Production and Assembly

The design incoporates a high degree of prefabrication. This ensures the highest possible quality, while keeping waste to a minimum. Furthermore, it allows for easy maintenance and makes it easy to replace a component of the structure if required. The system is completely standardised, resulting in very fast construction on-site. During construction, the concrete parking decks protect the rest of the timber elements and building, which is under construction, from the weather.

The modular system is also easy to disassemble. It uses reversible steel connections like push-in tubes, slotted plates and steel dowels. The large individual elements are therefore easy to separate and can be re-used or recycled.

A general impression of the design can be seen in Figure 60.



Figure 60. Artist impression of multi-storey car park in BauBuche (Technical University of Munich & Pollmeier Massivholz GmbH & Co.KG, 2015)

4.7 Concluding remarks

Concluding this chapter, sub-question 1a, "How to construct using timber as a structural material?", can be answered.

In recent years, many developments in the timber engineering industry have paved the way for new possibilities for timber construction. Engineered wood products like glulam and CLT make it possible to span larger distances with timber and the improved characteristics (more homogeneous properties for example) of the materials make it possible to use timber in ways that earlier were not possible. Using these engineered wood products, different types of structural components can be made, with varying cross-sections and even composite beams. The results of the developments are visible in newly constructed multi-storey structures build with timber and also in the designs that are made for car parks in timber.

Building with timber requires special attention. Detailing of a timber construction is extremely important. Proper transfer of forces considering the inhomogeneous properties of timber and allowing for fast and cheap erection requires ingenious design. Also the type of connection chosen impacts the load bearing capacity of the joint, but also the reusability of the structural components. Therefore, one should consider what characteristics are required from a connection when choosing between glued, carpentry or mechanical joints. Carpentry joints are a good option for joints with rather small loads, which do not require a very stiff joint. In other cases, mechanical joints using dowel-type fasteners or connectors are a better option.

Also durability of a timber structure requires ingenious design as timber is prone to biological and environmental degradation as well as to fire safety risks. In general, the best protection of the timber is realised through constructional measures. Selecting the most appropriate wood species or type of wood engineered material is another important factor in designing for a durable construction. Also considering fire-safety of timber structures, ingenious structural design is required as not the timber elements themselves are of concern, but the joints between them are. Especially for a multi-storey car park, protection of the timber decks is of concern and requires a proper pavement surfacing.

From a technical point of view, the possibilities to construct a multi-storey car park using timber are endless. Ingenious design incorporating all aspects like transfer of forces, joining of different elements, protection against biological and environmental degradation and fire, allowing fast on-site construction and all of this while keeping the costs competitive is the challenge for the designer.

PART III: Concept Development

5

Definition Concept Development

In the literature study, a large amount of information considering the design of a temporary multi-storey car park in timber has already been discussed. The next step is to use this knowledge to create a proof of concept for the structural design of the car park. This requires filtering of the information found in the literature study and making choices about aspects like the car park layout to be considered, which structural typologies are considered for the design variants and which specific materials are considered. Furthermore, it should be clear what is to be expected from the design for the proof of concept and on what aspects its performance is to be compared to existing variants. This chapter aims to address these aspects and define the concept development by diving into the goals and expected results, the general assumptions and considered design variants.

5.1 Goals for Concept Development

Through the in-depth literature study and interviews with various experts in the AEC industry, a clear overview has been obtained concerning the challenges that arise when building a temporary multistorey car park in timber. Based on this study and the challenges identified, we can more specifically formulate the main goal for the concept development, which incorporates solving various of the main challenges found in the literature study. Apart from this main goal, which is a rather broad description, two specific sub-goals are defined, which help to measure the performance of the design variants that are to be developed and which have an effect on many of the challenges found in the literature study. It is also discussed, what results are expected from the concept development, including a set of fourteen hypotheses. Next, the assumptions for the concept development are described and the different variants for the concept development are presented.

5.1.1 Main Goal

The objective of this thesis has been defined as to "provide a structural system proof of concept for a temporary multi-storey car park, using timber as primary structural material". The literature study has resulted in an overview of aspects which such a structure has to fulfil and what challenges have to be solved. Based on these results, when elaborating on the original objective, we can depict that the main goal for the concept development is threefold.

First of all, the concept for the temporary multi-storey car park should help the AEC industry in its path towards a sustainable industry. CO₂ emissions and energy consumption should be reduced, for which using timber as a primary structural material is a first step. However, in the concept development, more aspects related to making the sustainability transition should be taken into account. Incorporating the circular design strategies, as described in paragraph 3.3, in the concept is important to make the structure truly sustainable. This includes apart from closing the loop through the use of timber, strategies like designing for dis- and reassembly, reliability and durability, ease of maintenance and repair and upgradability and adaptability.

Second, building with timber comes with its challenges as discussed in chapter 4. Creating a durable construction through ingenious design is very important with this material as it comes with several durability risks, see paragraph 4.4. Building with timber comes with challenges concerning moisture, fire, acoustics and vibrations. Furthermore, one has to keep a close eye on the material characteristics to make optimal use of the material in the right places.

Third, the objective is concerned with a 'temporary multi-storey car park'. Considering the tendency for timber structures to be more expensive compared to structures in traditional steel and concrete, it is important to win back costs where possible. Together with the temporary nature of the construction, a fast construction process is an important aspect as this can reduce the building costs. As the structure has to be able to function at various locations and with various user goals in mind, it should be modular, useful in different configurations and function at a high level of quality.

5.1.2 Specific Sub-goals

To make the main goal more concrete, two specific measurable sub-goals are defined. These sub-goals are based on some of the larger challenges found in the literature study, through interviews with various experts as well as analysis of several existing multi-storey car parks with a temporary character or designs in timber. These sub-goals help to later measure the performance of the different developed design variants, but in the process of the design of these variants, the main goal as presented before should always be kept in mind.

5.1.2.1 Minimizing Total Deck Height

In the literature study, it could already be seen from existing car parks in timber (see paragraph 4.6.1), that using glulam timber to span the large spans over the decks of a car park results in a large total deck height. In comparison, a concrete TT-slab can span the same distance with a total deck height halve that of the timber variant discussed in 4.6.1. With such spans, especially the deflection and vibrations of timber beams presents a challenge as the limitations on these deflections result in large required beam heights. Larger floor depths mean the total height of a floor level increases and longer ramps are required to reach the next level. All of this occupies space, which preferably would be used for parking cars instead of being required for the structure. Considering the lower weight of timber, it might also be possible to stack more elements on top of each other for transportation when the elements height is limited. Therefore, a first specific and easy to measure sub-goal is to minimize the total deck height of the concept for the temporary multi-storey car park.

In chapter 7, five different modular and/or timber car park structural concepts are compared to each other. This comparison includes the total deck height of the system and results in a concrete measurable number as a goal for the minimization of the total deck height of the timber variants to be designed.

5.1.2.2 Weight Minimization

The second specific sub-goal is to minimize the weight of the multi-storey car park. Minimizing the weight of the car park has many advantages considering both its temporary nature as well as the sustainability ambitions. A lighter structure requires less transportation as more elements can be transported per truck and requires less heavy machinery on-site for hoisting the elements into place. It requires smaller foundation works and possibly no foundation piles. Also the supporting structure of the floors for example has to carry less heavy loads, which can save on material. All of these aspects speed up the assembly and disassembly process, save costs and make the structure more sustainable, making it a very relevant and measurable goal. No exact numbers on costs will be gathered due to a lack of data and highly fluctuating material prices at the moment of publication of this thesis, due to the aftermath of the COVID-19 pandemic and war in Ukraine. However, minimizing weight in general has a positive influence on many relevant aspects.

Although a lower weight of the structure has many advantages, it also brings some challenges. Floors with a higher weight are for example less prone to vibrations in the structure and better prevent the transmission of sound compared to lighter floor types. Such aspects are important to keep in mind.

In chapter 8, it is studied what foundation types are possible for the car park and what range of loads coming from the columns of the car park could reduce foundation costs and improve the foundation's level of sustainability. This results in another specific measurable sub-goal concerned with minimizing the weight of the structure.

5.1.3 Expected Results

Various alternative designs will be developed and analysed for the above mentioned goals. These can then be compared to each other for the specific sub-goals. The best performing variant for the structural design of the temporary multi-storey car park should provide the objective's proof of concept that fulfils the various requirements.

During the development process of the design variants, several input parameters will be altered to gain insight into what impact changing these parameters has on the structure and the (sub-)goals. Through this analysis, new knowledge is created on what type of design is best for which set of variables, what impact for example changing regulations (such as deformation limits) have on designs and if there are certain limits to designs for car parks in timber. These parameters can be split into two groups: geometric parameters and assessment criteria parameters.

The first set of parameters are related to the geometric design of the structural system of the car park. In paragraph 2.4, already a large set of parameters concerned with the geometry of the car park was given based on the design standard for parking facilities – NEN 2443 (Nederlands Normalisatie Instituut, 2013). However, this still leaves a lot of room for the designer to create the exact geometry of the structure. Furthermore, this standard is not required and can be deviated from. Such design choices can have a significant impact on for example the cross-sections of elements and therefore the specific sub-goals presented above.

The second set of parameters are concerned with the assessment criteria that the construction has to fulfil. As discussed before, due to the lightweight properties of timber, timber elements can be prone to vibrational behaviour. Also deformation limits can determine the minimum cross-section dimensions. The limits of such assessment criteria can be set by the designer and may have a significant effect on the structural system. Therefore, alteration of such assessment criteria can be highly interesting to study.

For the first set of parameters, it is studied what their effect is on both the choice of deck system and frame system (if parameter is applicable at design variant), presented in paragraph 5.3. For the second set, only their effect on the deck system is studied in detail, due to limitations of the used software package. However, the effect of these parameters on the framing system is shortly discussed in paragraph 10.4.4.

Geometric parameters:

- Parking deck span
- Use of struts in main span (yes / no)
- Distance between columns in transverse direction
- Distance between joists

Assessment criteria parameters (deck system only):

- Deflections
- Fire resistance class
- Vibrations

5.1.3.1 Hypotheses

Altering the parameters mentioned above will have an effect on the structural system of the multi-storey car park. Therefore it will also affect the outcomes of the specific sub-goals. For each parameter a hypothesis is drawn concerned with what its expected effect is on each specific sub-goal.

Parking deck span

A larger parking deck span is expected to require thicker floor elements or larger girders, increasing both the total weight of the structure as well as the total deck height. This results in hypotheses H1a and H1b.

H1a: Increasing the parking deck span will result in an increase of the total deck height.

H1b: Increasing the parking deck span will result in an increase of the total construction weight.

Use of struts in main span

Using struts to support the main girder of the structural system shortens the unsupported span of the main girder. Therefore, it is expected that this will reduce the cross-section of this girder, resulting in a decrease of the total deck height. Furthermore, it is expected that the reduction of the cross-section of the main girder will result in a larger decrease of material use than the additional material required for the struts. This results in hypotheses H2a and H2b.

- H2a: The use of struts in the main span of the structural system of the car park will result in a decrease of the total deck height.
- H2b: The use of struts in the main span of the structural system of the car park will result in a decrease of the total construction weight.

Distance between columns in transverse direction

Increasing the distance between the columns in transverse direction requires the floors or girders supporting the parking deck to span a larger distance. Therefore it is expected that this results in larger floor thicknesses and cross-sections, resulting in a larger total deck height and higher construction weight. The reduction of material use as the result of a smaller amount of columns is not expected to cover the additional required material for the larger cross-sections of floors and girders. This results in hypotheses H3a and H3b.

- H3a: Increasing the distance between columns in transverse direction will result in an increase of the total deck height.
- H3b: Increasing the distance between columns in transverse direction will result in an increase of the total construction weight.

Distance between joists

Increasing the distance between the joists results in a longer deck span. Therefore, it is expected this requires a thicker deck and results in bigger loads on the supporting joists, requiring larger cross-sections. This is would result in a larger total deck height and higher construction weight. The lower number of required joists is not expected to compensate for the larger weight of each joist. This results in hypotheses H4a and H4b.

H4a: Increasing the distance between joists will result in an increase of the total deck height.

H4b: Increasing the distance between joists will result in an increase of the total construction weight.

Fire resistance class (decks only)

The fire resistance class applied to a timber structure significantly influences the dimensions of the cross-sections of its structural elements, since for timber every minute of required fire resistance adds a certain thickness to the element (see paragraph 4.4.2.7). A lower fire resistance class can therefore result in smaller cross-sections. This results in hypotheses H5a and H5b.

- H5a: Applying a lower fire resistance class to the structural system of the multi-storey car park results in a decrease of the total deck height.
- H5b: Applying a lower fire resistance class to the structural system of the multi-storey car park results in a decrease of the total construction weight.

Vibrations (decks only)

Whether vibrational behaviour of a construction is checked can affect the dimensions of its structural elements. Heavier elements generally improve the vibrational behaviour of the structure. This of course impacts the total weight of the structure and the total deck height, resulting in hypotheses H6a and H6b.

H6a: Setting lower or no criteria for the vibrational behaviour of the structural system of the multistorey car park will result in a decrease of the total deck height. H6b: Setting lower or no criteria for the vibrational behaviour of the structural system of the multistorey car park will result in a decrease of the total construction weight.

Allowed deflections (decks only)

Allowing larger deflections for the floors and girders of the structural system is expected to result in smaller floor heights and cross-sections, reducing total weight of the construction and total deck height. However, this is only valid if the serviceability limit state is governing. If the ultimate limit state is governing, no reduction of the floor height and cross-sections is possible. In general, hypotheses H7a and H7b are stated.

- H7a: Allowing larger deflections for the structural elements of the car park will result in a decrease of the total deck height.
- H7b: Allowing larger deflections for the structural elements of the car park will result in a decrease of the total construction weight.

5.1.3.2 Deliverables

As final result of this thesis, the goal is to present a proof of concept with design considerations and prerequisites for the design of a temporary multi-storey car park using primarily timber as building material as an addition to existing knowledge on building in timber. These results are related to the following aspects:

- The applicability of timber in a temporary multi-storey car park.
- The number of levels possible in timber.
- Expected dimensions for various elements.
- Expected floor heights for a timber car park.
- Effect of angled parking on the structure of a multi-storey car park.

5.2 General Assumptions Concept Development

For the development of the concept for the structural system of the temporary multi-storey car park, several assumptions have to be made. Already in the literature study, several assumptions have been described concerning aspects like user and owner requirements, structural rules and norms and car park design rules and norms. This paragraph focusses on some of the most important assumptions that influence the design of the structural system.

5.2.1 Car Park Design

A concept for the structural system of a temporary multi-storey car park will be developed, which has to be flexible considering the possible different locations and user functions it has to fulfil. Although it has a temporary character, it has to fulfil the high quality standards of a normal fixed car park both in terms of service to its users as well as structural requirements as described in chapter 2. To fulfil all possible requirements, the concept is developed for the following category: a public multi-storey car park, which is intensively used, see paragraph 2.1.1.

5.2.1.1 Main Span

For the dimensions of bays and aisles, the minimum dimensions from NEN 2443 are used as a starting point, see paragraph 2.1.3 and 2.4.2. The main span is formed by the combination of the dimensions of the length of a parking bay, the width of the aisle, the additional width of the redress lanes and optionally room for a pedestrians walkway. Depending on the parking angle, the length of the aisle and the optional extra room for pedestrians, the main span can vary from a minimum of 12.18 m to a maximum of 17.66 m for the standard dimensions of NEN 2443.

5.2.1.2 Other Dimensions

Starting point for the width of the parking bays is 2.50 m as stated in NEN 2443. A standard clearance height of 2.30 m is used. Furthermore, a column free parking deck is considered required for the concept design. The various dimensions are discussed in detail in paragraph 2.4.

5.2.2 Car Park Circulation Layout

In the literature study in paragraph 2.5 and in appendix B, various car park circulation layouts have been discussed. For the development of the concept design, a choice has to be made which layout(s) is/are considered. For the circulation layout of the car park, the sloped parking deck layouts are ruled out as they contain multiple undesirable characteristics. Considering the user-friendliness of the car park, their sloped parking decks hinder users in getting in and out of their car as doors slam shut or are difficult to close and shopping cars and wheel chairs roll away on the decks. Considering the required modularity of the concept, sloped parking deck layouts result in fixed lengths for the decks as they connect the different floor levels. This hinders the modularity and scalability of the car park concept.

Flat deck circulation layouts are considered as an option. Although their static efficiency is not the best, their other characteristics are generally good. A large advantage of flat deck layouts is that they can be highly modular. With separate ramp systems, car parks of all kinds of sized and layouts can be realised, which is very relevant for the development of a temporary multi-storey car park.

Split-level deck circulation layouts are the preferred option considering their average static efficiency. Compared to the flat deck layouts, their dynamic efficiency is not as good, but at the same time, their circulation efficiency is very high and other characteristics are also good. Downside of the split-level decks is that although they can vary in size, they generally consist out of two bins and require ramps between the split-level decks, making them less flexible for different layouts.

As the structure of both flat decks and split-level decks uses a comparable structure to span the parking decks, it might be possible to develop a structural system, which can function both as a flat deck layout as well as a split-level deck layout.

5.2.3 Building Materials

The primary building material to be used for the design of the multi-storey car park is timber. However, also other building materials may be used if their advantages over the use of timber are very significant and their impact on sustainability is acceptable. As the production of concrete results in high amounts of CO₂ emissions and energy use, is very difficult to fully recycle and adds a large amount of weight to a structure, it is not used for the concept development of this thesis. Steel, although requiring large amounts of energy for its production is allowed as it is fully recyclable, durable, has a high reuse potential and is a very good material to be used in for example connections.

5.3 Design Variants

In the literature study, different structural typologies (frame and deck systems) have been studied, see paragraph 2.6. For the concept development, a choice has to be made concerned with which structural typologies are used to create design variants. Although the framing system and deck system influence each other, they are discussed separately. The deck systems are divided in short and long span systems and for the framing systems, it is stated what type of deck system is required.

5.3.1 Deck System

Possible deck systems have been discussed in paragraph 2.6.1 (deck typologies in general) and 4.2.2 (possible types of timber decks). Considering the modular and temporary aspects of the multi-storey car park to be developed, only two possible type of systems are deemed possible. These are either long-span or short-span prefab elements. Within these type of deck systems, various alternatives exist.

5.3.1.1 Long-span Prefab Deck System

A total of seven different long-span prefab deck systems are considered, which are presented below. These systems use either CLT elements (see Figure 61) or LVL elements (see Figure 62):

- CLT panel;
- CLT open rib panel;
- CLT closed rib panel (box floor);
- LVL panel;
- LVL open rib panel;
- LVL semi-open rib panel;
- LVL closed rib panel.



Figure 62. Long-span prefab deck systems in LVL (Stora Enso, 2022f)

5.3.1.2 Short-span Prefab Deck System

For the short-spans, only two types of prefab elements are considered (see Figure 63), which may be supported by multiple girders, further shortening the (sub)spans:

- CLT panel;
- LVL panel.



Figure 63. CLT panel (left) and LVL panel (right) used for short-span prefab deck systems (Stora Enso, 2022c, 2022d)

5.3.2 Framing System

Following from the literature study (see paragraph 2.6.2), four different main variants for the framing system of the car park are seen as potential solutions for the concept design. These have a column free parking deck and lent themselves to be build using primarily timber materials. The exact design of each variant, like the way the different elements are connected to each other and their dimensions may be altered. The general configuration with the span directions of each element however will remain unchanged.

5.3.2.1 Variant 1

The first variant to be studied consists out of a structural frame from columns and transverse girders, see Figure 64. These girders support bin wide floor elements, for which one of the long-span prefab deck systems is to be chosen. The columns can be positioned every 2.5 m (one parking bay width) or 5.0 m (two parking bay widths). Also the parking deck span can be varied as well as the number of levels. Variant 1 does not contain struts in its structural system.



Figure 64. Variant 1 - 3D overview of design variant

5.3.2.2 Variant 2a

The second variant to be studied consists out of a structural frame using columns, main girders and optional struts, see Figure 65. The main girders support a thin parking deck, for which a short-span prefab deck system has to be chosen. The columns can be positioned every 2.5 m (one parking bay width) or 5.0 m (two parking bay widths). Also the parking deck span can be varied as well as the number of levels.



Figure 65. Variant 2a - 3D overview of design variant

5.3.2.3 Variant 2b

Design variant 2b is an alteration of previous design variant. It contains only halve the number of columns as every subsequent main girder is supported by additional transverse girders instead of columns, see Figure 66. These transverse girders transfer the loads to the remaining columns. The main girders support a short-span prefab deck system that has to be chosen. The columns are positioned at 5.0 m (two parking bay widths) and struts are not used. The parking deck span can be varied as well as the number of levels.



Figure 66. Variant 2b - 3D overview of design variant

5.3.2.4 Variant 3

The last variant to be studied looks similar to variant 2a and uses a framing system consisting out of columns, main girders and optional struts, but additionally contains secondary girders (joists), see Figure 67. These secondary girders span the distance between the main girders and support a short-span prefab deck system over a shorter span. The columns are positioned at 5.0 m (two parking bay widths) and the parking deck span can be varied as well as the number of levels.



Figure 67. Variant 3b - 3D overview of design variant

5.4 Concluding Remarks

In conclusion of this chapter, sub-question 3a, "What characteristics of a temporary multi-storey car park using timber as primary structural material are relevant to assess its structural performance?" can be answered.

Following from the literature study, this chapter has resulted in a narrowed down definition of the concept development. The main goal for the concept development has been stated and is threefold: focus on the lifecycle of the structural system, solve the material challenges related to building with timber and take into account the temporary nature of the concept to be developed, linked to costs and modularity. Two specific concrete and measurable sub-goals have also been stated to be able to assess the performance of the various designs and compare them to alternatives: minimization of total deck height and minimization of the weight of the structural system.

Furthermore, four geometric parameters and three assessment criteria parameters are chosen to study their effect on the performance of the different variants. These are the parking deck span, use of struts in the main span, distance between columns in transverse direction, minimum distance between joists, allowed deflections, fire resistance class and vibrational criteria. For each of these parameters, a hypothesis is stated for its relation to the two specified sub-goals.

Multiple design variants will be studied for their performance. These consist out of a deck and framing system. For the deck system, a total of seven different long-span prefab deck systems are considered and two short-span prefab deck systems. For the framing system, four different typologies are chosen to be studied.

6

Structural Starting Points Conceptual Design

In the design process of the proof of concept for the modular and temporary multi-storey car park, choices have to be made, steering the design process towards the final proof of concept. In chapter 5, already a selection was made from the information found in the literature study concerned with the general design of the car park and its structural typology, resulting in the selection of four different framing systems, seven different long-span deck systems and two different short-span deck systems. This chapter focusses on the structural starting points, which are required for the later calculations of the design variants. Much of the information in this chapter was already discussed in the literature study in paragraph 2.3, but this chapter acts as a summary and clear overview of the structural starting points for the design and calculation of the design variant.

6.1 General

The car park is classified as a public car park according to NEN 2443 (Nederlands Normalisatie Instituut, 2013). This norm is regarded as leading and any deviation from its contents will be explicitly mentioned.

The structural system of the car park consists of a timber frame. The deck is made from CLT or LVL and the beams and columns are made of glulam or glued solid timber. Vertical stability is provided by steel bracing and horizontal stability is provided by in-plane action of the decks.

6.2 Applied Structural Rules and Norms

For the design, in principle the following rules and norms have been applied. Any deviation from this will be explicitly mentioned.

- Dutch Building Degree 2012 (Ministerie van Binnenlandse Zaken en Koninkrijkrelaties, 2021)
- NEN-EN 1990+NA: Eurocode 0, Basis of structural design
- NEN-EN 1991+NA: Eurocode 1, Actions on structures
- NEN-EN 1995+NA: Eurocode 5, Design of timber structures
- NEN-EN 1997+NA: Eurocode 7, Geotechnical design
- (NEN, 2021)

6.2.1 Consequence Class, Service Class, Design Life & Load Factors

The structure of the multi-storey car park is designed according to NEN-EN 1990 (2011) + NA (2019) (NEN, 2021) and NEN-EN 1995-1-1 (2005), assuming the following conditions:

Type of building:	Public multi-storey car park	
Consequence class:	CC2	
Reliability class:	RC2	k _F = 1.0
Design service life category:	Class 3	50 years
Service class (timber):	Class 2	-

Description	Type of load	Factor	6.10a	6.10b
Ultimate Limit State (ULS)	Permanent	$\gamma_{G;unfavourable}$	1.35	1.2
	Permanent	γ _{G;favourable}	0.90	0.9
	Variable	γ _Q	1.5 ψ _{0,1}	1.5
Serviceability Limit State (SLS)	Permanent	$\gamma_{G;unfavourable}$	1.0	1.0
	Permanent	γ _{G:favourable}	1.0	1.0
	Variable	γ _Q	1.0	1.0

Table 14. Load factors ULS and SLS (Nederlands Normalisatie Instituut, 2015)

6.2.2 Load Combinations

Load combinations for the Ultimate Limit States (ULS) are assumed according to art. 6.4.3 of NEN-EN 1990 (Nederlands Normalisatie Instituut, 2015).

Load combinations for the Serviceability Limit States (SLS) are assumed according to art. 6.5.3 of NEN-EN 1990 (Nederlands Normalisatie Instituut, 2015).

Appendix A gives a more detailed overview of the formulas for load combinations.

6.2.3 Function of Building & Combination Factors

All levels of the building are designated as category F: Traffic area, vehicle weight \leq 25 kN, according to NEN-EN 1990 + NA (Nederlands Normalisatie Instituut, 2015). Table 15 gives an overview of the relevant combination factors for the building.

Table 15. Combination factors for buildings. (Nederlands Normalisatie Instituut, 2015)

Action	ψ_0	ψ 1	ψ_2
Imposed loads on buildings:			
Category F: Traffic area, vehicle weight ≤ 25 kN	0.7	0.7	0.6
Snow loads	0	0.2	0
Wind loads	0	0.2	0
Standing water	0	0	0

6.3 Loads

This paragraph describes the characteristic load values for permanent and variable loads on the structure.

6.3.1 Permanent Loads

The following permanent loads act on the structure, see Table 16. As the car park is naturally ventilated and only a simple and lightweight lighting scheme is assumed (a single luminaire per relatively large area), no permanent load for installations is considered.

Table 16. Permanent loads car park

Description	Weight	
Self-weight	(Differs per element)	
Aggregate coating parking deck	0.1 kN/m ²	(HIM, 2020)
Façade & safety barrier	2.0 kN/m ²	(Kaa, 2020)

6.3.2 Variable Loads

6.3.2.1 Imposed Loads

For category F, traffic and parking area for light vehicles, a distributed and concentrated load are imposed, see Table 17. The concentrated load has to be applied as an axle load, following the dimensions in Figure 68, in which the square surfaces have sides of 100 mm.

Table 17. Imposed loads on garages and vehicle traffic areas. (Nederlands Normalisatie Instituut, 2019a)



Figure 68. Dimensions of axle load. (Nederlands Normalisatie Instituut, 2019a)

6.3.2.2 Wind Loads

For the wind load, the following assumptions are made (see Appendix A for detailed formulas):

Wind region:	I
Terrain category:	0 (coastal area)
Height:	21.0 m
CsCd:	1.0

From these assumptions follows: $q_p(z_e) = 1.82 \text{ kN/m}^2$

6.3.2.3 Snow Loads

The top deck of the car park also acts as a roof. Therefore, the roof shape of the car park is flat: $\mu_1 = 0.8$. It is assumed snow cannot accumulate against barriers on the deck. This results in the following snow load (see Appendix A for detailed formulas):

$s = 0.56 \text{ kN/m}^2$

6.4 Deformations

According to the Dutch Building Degree, deformations of a structure do not have to be tested. However, at first, the advised maximum allowed deformations according to the Eurocode NEN-EN 1990 NA (Nederlands Normalisatie Instituut, 2015) are applied. Later, it is studied what the effect is when larger deformations are allowed.

6.4.1 Vertical Deformations

The maximal allowable vertical deflections can be found in Table 18. The different parameters used are displayed in Figure 69.



Table 18. Allowed vertical deflections. (Nederlands Normalisatie Instituut, 2015)

Figure 69. Vertical deflection parameters. (Nederlands Normalisatie Instituut, 2015)

6.4.2 Horizontal Displacements

The total maximum horizontal displacement for buildings with more than one storey under the characteristic combination of actions is:

- h/300 per storey (ui Figure 70)
- h/500 for the entire building (u Figure 70)



Figure 70. Definition of horizontal displacements. (Nederlands Normalisatie Instituut, 2015)

6.5 Vibrations

Vibrations of the decks of the structure are checked for a frequency criterion of 4.5 Hz and a stiffness criterion of 0.500 mm, see paragraph 2.3.3. Later on, it is also studied what the effect on the performance of the structure is when these criteria are not considered.

6.6 Fire Safety

For the main load bearing structure of the multi-storey car park, as a starting point, a fire resistance requirement against collapse of 90 minutes is applied. Later on it is studied what effect reducing the fire resistance requirement has on the dimensions of the structural system. See paragraph 2.3.4 for more information.

6.7 Design Parameters

To limit the range of possible designs to be studied, several design parameters have to be set. This paragraph presents a set of fixed design parameters as well as several variable parameters (as discussed in paragraph 5.1.3), of which their effect on the structural design of the multi-storey car park is studied.

6.7.1 Fixed Parameters

Several parameters for the design of the multi-storey car park are fixed or are related to other parameters of the car park. An overview of these parameters can be found in Table 19.

Parameter	Value
Clearance height	2.30 m
Parking bay width	2.50 m
Vertical height strut	Level height
Horizontal length strut	0.67 m
Dimensions for wind calculation	
Total car park length	60 m
Total car park width	40 m
Nr. of levels	5

Table 19. Fixed design parameters

Material classes				
Glulam timber	GL 28h			
Glued solid timber	C24			
Steel	S235			

6.7.2 Variable Parameters

Some parameters are varied, to study what effect this has on the rest of the structure. These variable parameters were already mentioned in paragraph 5.1.3. Inn Table 20 an overview is given of these parameters, their range of values and for which design variants they are relevant.

Table 20. Variable design parameters

Parameter	Values
Number of floor levels	1 to 5
Parking deck span	12.18 m
	13.70 m
	14.74 m
	15.61 m
	16.26 m
	17.16 m
	17.66 m
Allowed deformations	Allowed additional deflection 1/333 * Irep
	& allowed final deflection 1/250 * I _{rep}
	Allowed additional deflection 1/183 * Irep
	& allowed final deflection 1/150 * Irep
Distance between columns (variants 1 & 2a)	2.5 m
	5.0 m
Distance between joists (variant 3)	Max 2.5 m
	Max 3.75 m
Use of struts (variants 2a & 3)	Yes
	No
Fire resistance class	R 90
	R 30
Vibrations	Criteria: frequency 4.5 Hz & stiffness 0.500 mm
	No check on vibrations

7

Modular or Timber Car Park Concepts

Over the years, many different types of multi-storey car parks have been constructed using various structural concepts. The aim of this thesis is to provide a structural system proof of concept for a temporary multi-storey car park, using timber as primary structural material. Given this objective, one of the sub research questions is to study how such a structural system compares to existing alternative concepts. In this chapter, the basis for the later comparison of the developed structural system is laid by comparing five different modular or timber structural systems. The performance of these alternative concepts is assessed for the specific sub-goals presented in chapter 5. This helps to create a feeling for what level of performance is to be expected from the proof of concept to be able to compete with these alternative concepts.

7.1 Description of Concepts

Since the goal is to design a structural system which uses primarily timber and also is modular, it would be best to compare it to an existing multi-storey car park with such characteristics. However, to the knowledge of the author, such a car park has not been realised yet. Therefore, the focus has been placed on finding either modular car park structures of car parks constructed primarily in timber, to compare the structural system to. This has resulted in five different multi-storey car parks, which have five different types of structural systems. The various concepts are discussed in the following paragraphs.

7.1.1 ModuPark

ModuPark is a modular multi-storey car park concept developed by Ballast Nedam Parking, which is comprised of prefab steel and concrete elements, see Figure 71 (BNParking, 2016). Advantageous of the system is the speed of construction. The car park can be built in just three months. When no longer needed, the entire structure can quickly be disassembled and reused at a new location. Downside of the system is the required heavy foundation as a result of the concrete elements, which often requires concrete foundation piles, which are difficult to remove and reuse.



Figure 71. ModuPark Maarssen (ASK Romein, 2018b)

The concept uses steel columns, transverse girders and bracing, creating a frame, on which concrete TT-plates are placed, spanning the entire parking deck, see Figure 71. These TT-plates do not require any finishing layer and can be directly driven on. A standard TT-plate has a width of 2400 mm (one parking bay) and a length of generally 16.5 m (ASK Romein, 2018a). Columns are placed at a distance of 4.8 m from each other, creating a grid of 16.5 x 4.8 m. Depending on the number of levels, a different thickness of the columns is used to support the larger loads from the upper decks. A total of 6 levels is perfectly possible. Using the standard elements, car parks of many different shapes can be created and by adapting some of the elements, even more shapes are possible. This makes the concept useful in many locations and stimulates reuse.

7.1.2 Car Park Koopman int.

The multi-storey car park for Koopman int., located at the Distelweg in Amsterdam is comprised of prefab steel and fibre-reinforced plastic elements (Park4all, 2020). The system is relatively fast to construct and can easily be disassembled and reused. Since only steel and fibre-reinforced plastics are used, the total structure is extremely lightweight, allowing for a less heavy foundation.

The structural system of the car park uses steel columns, main girders and secondary girders on top of which a fibre-reinforced plastic floor is placed, see Figure 72. This floor uses Fiberline heavy duty planks with a thickness of only 40 mm and a weight of only 8.43 kg/m for a standard width of 500 mm (Fiberline, 2021). A column free parking deck with a span of 15.65 m is realised and columns are placed at every 5.0 m. The car park is realised for two levels, but is designed for addition of two more levels (Park4all, 2020). The car park is designed for a specific client, but can easily be disassembled and reused. As the concept uses all prefabricated elements which are bolted together, it also can be used as the basis for a modular and flexible car park system.



Figure 72. Section of car park Koopman int. Amsterdam (EdJ, 2019)

7.1.3 Car Park Morspoort

The multi-storey car park Morspoort in Leiden is a temporary and demountable structure comprised of prefab steel and steel-concrete composite elements. As it is designed as a temporary structure, ease of disassembly has been taken into account during the design (JVZ Ingenieurs, 2012). The result is a structural concept, which is fast to construct and disassemble and can be reused. The use of steel-concrete composite elements reduces the total weight of the structure and allows lighter foundations.

The car park's structural system is build-up using steel columns at every 6.85 m, supporting the main girders spanning the parking deck with a span of 14.5 m, see Figure 73 (JVZ Ingenieurs, 2012). The main girders support prefabricated composite steel-concrete floor panels, which is the Corus Quantum Deck system. This floor system is a prefabricated steelframeconcretefloor, build-up of cold formed C-profiles, poured into a 51 mm thick concrete plate. This results in a reduced weight of 180 kg/m² (Tata Steel, 2021). The structural system can be adapted for many different shapes and sizes of car parks, including rounded edges.


Figure 73. Car Park Morspoort, Leiden (De Ruiter, 2011)

7.1.4 BauBuche Concept

At the Technical University of Munich, a concept for a modular and demountable multi-storey car park in timber and concrete has been developed (Technical University of Munich & Pollmeier Massivholz GmbH & Co.KG, 2015). This concept was already discussed in paragraph 4.6.2, but is shortly discussed here as well. The structural system is comprised of timber elements in BauBuche (a new development in glulam beams with higher strength properties) and prefabricated concrete floor elements. The entire system is designed to be highly modular, demountable, fast in construction and easy to reuse. The timber elements made from BauBuche allow for smaller dimensions of its cross-sections, but the concrete floor elements significantly add weight to the total system.

As can be seen in Figure 74, the concept uses a standardized modular element made of timber columns at every 2.5 m, supporting timber main girders, spanning the parking deck of 16.5 m. These main girders support 130 mm thick prefab concrete floor panels. The columns and main girders are connected to each other using a ingenious system of steel tubes, slotted plates and dowels, which allow large loads to be transferred from upper decks to the foundation (Technical University of Munich & Pollmeier Massivholz GmbH & Co.KG, 2015). The concept is developed to be used in many different types of car park layouts, such as flat decks, split level and sloped decks.



Figure 74. Modular element of modular car park concept in BauBuche (Technical University of Munich & Pollmeier Massivholz GmbH & Co.KG, 2015)

7.1.5 B&O-Holzparkhaus

The B&O-Holzparkhaus is the only multi-storey car park in this chapter which has a superstructure almost completely made from timber, see Figure 75. It is not designed as a modular, demountable or reusable structure, but is very useful as a concept fully build in timber to compare to the other concepts. The car park is made from different types of wood engineered products, including BauBuche, glulam and CLT. As timber is a relatively lightweight material, the entire structure also has a relative low self-weight (HK Architecten, 2021).

The car park has only one span and two levels. Glulam columns are placed at every 2.60 m, supporting the main girders made from BauBuch GL75, spanning the parking deck of 16.26 m. A CLT parking deck is placed on top of the main girders and is finished with a 60 mm thick layer of asphalt (HK Architecten, 2021). This unfortunately makes disassemble and reuse of the deck very difficult. Between the main girders, additional CLT panels are placed for stability as well as steel wind bracings in the façades.



Figure 75. Timber multi-storey car park Bad Aibling (HK Architecten, 2022)

7.2 Comparison

A comparison is made between the five different modular or timber multi-storey car park concepts presented in previous paragraph. This comparison helps to gain insight into how the use of different concepts and different building materials results in different performance on various aspects: total deck height, weight of the structure and load from a "standard" column on the foundation. It helps to get a feel for what range of values for the various aspects are to be expected and can act as a basis for what performance is requested from the timber car park designs. Later, the timber car park designs can be compared to these values and there performance can be assessed.

As all five car park concepts have different dimensions, they first have to be altered to be able to compare them to each other. The dimensions found in Table 21 are used for all concepts. If necessary, profile dimensions are also in- or decreased to be able to support the loads given the dimensions for the comparison. An overview of the details of the car park concepts like used profile dimensions and their properties can be found in appendix E, Table 55 and Table 56.

The comparison of the three aspects are discussed in the following paragraphs. Table 22 gives an overview of the performance of the five car park concepts on the three aspects. It also shows a comparison of four car park concepts relative to the ModuPark concept, which has the smallest total deck height.

Car park characteristics										
Parking bay width	2.5 m	Span parking deck	16.26 m							
Parking bay length	5.13 m	Clearance height	2.30 m							
Aisle width	6.00 m	Column every 1 or 2 par	king bays							
Parking angle	90°		0							

Table 22. Comparison of various modular or timber car park concepts on floor depth, weight and standard column load

Car park	Floor depth [mm]	Rel. to ModuPark	Weight /level [kg]	Rel. to ModuPark	Load standard column on foundation - 4 levels [kN]	Rel. to ModuPark
ModuPark	550	-	61,261	-	2,472	-
Koopman int.	640	116%	10,477	17%	1,258	51%
Morspoort	651	118%	35,110	57%	1,845	75%
BauBuche concept	730	133%	59,341	97%	1,193	48%
B&O Holzparkhaus	935	170%	17,547	29%	717	29%

7.2.1 Total Deck Height

The first aspect on which the five car park concepts are compared is the total deck height. This height is defined as the total height difference between two car park levels minus the clearance height. It is the height required for the construction of the parking deck to span the large span of the parking deck. Depending on the structural design of the concept the total deck height might only include the thickness of the floor, but it can also include the height of primary or secondary girders.

In Figure 76, the total deck height for the five car park concepts are presented next to each other. The graph clearly shows that the ModuPark concept outperforms the other concepts. This is due to the characteristic of the ModuPark concept that it uses TT-plates which span the entire parking deck and are supported at the sides. The monolithic connection between the ribs of the plate and the deck result in a very efficient element in terms of its length to height ratio.

The other four concepts all use structural systems in which the parking deck panels are supported by primary (and sometimes secondary) girders. These systems result in larger total deck heights as their elements are mainly stacked on top of each other and the elements often do not act together in a way similar to the ribs and plate of the TT-plates. The total deck height of the Koopman int. car park concept is primarily the result of the height of the primary girders (600 mm), while the thickness of the floor is very limited (40 mm). This results in a total deck height of 640 mm, which is 16% larger than the ModuPark concept. The Quantum Deck floor of the Morspoort car park has a total height comparable to that of the Koopman int. car park of 651 mm, which is 18% thicker than the ModuPark concept. It also uses a primary girder of 600 mm which supports only the concrete part of the floor of 51 mm.

The last two concepts are the timber-concrete BauBuche concept and the fully timber build B&O-Holzparkhaus. The BauBuche concept has a total deck height of 730 mm, which is build-up of a concrete plate with a thickness of 130 mm and a primary girder with a height of 600 mm. The height of the primary girder and as a result the total deck height is limited by the use of the stronger BauBuche material. The total deck height is 33% larger compared to the ModuPark concept. The B&O-Holzparkhaus uses a CLT panel for the floor with a thickness of 100 mm and a primary girder with a height of 760 mm. It is the only deck, which cannot directly be driven on, adding another 75 mm of thickness to the total deck height for asphalt and finishing layers. This results in a much larger total deck height of 70% compared to the ModuPark concept.



Figure 76. Comparison of total deck height (floor element + supporting girder (if above parking deck)) for various modular or timber car park concepts

7.2.2 Weight per Level

The second aspect on which the five car park concepts are compared is the weight per level. The weight is calculated for one level with a width of 4 parking bays $(4 \times 2.5 = 10 \text{ m})$ and a length of a parking deck span of 16.26 m (90° parking, 6.00 m aisle). The weight calculation only includes all main load bearing structural elements, present at every grid position of the car park. Local wind bracing, is therefore excluded.

In Figure 77, the weight per level for the five car park concepts is presented next to each other. The graph immediately shows the large differences in weight between the five concepts. The concepts which use concrete floor elements are the heaviest, followed by the concept using a steel-concrete composite floor. The use of timber (CLT) or fibre-reinforced plastics in the floor results in the lightest construction.



Figure 77. Comparison of weight per level (4x2.50 m bay wide, 16.26 m span) for various modular or timber car park concepts

The ModuPark concept with its heavy concrete TT-plates results in the smallest total deck height, but as a downside also the highest weight of the structure at 61,261 kg. The BauBuche concept also uses concrete floor panels, which in combination with the BauBuche elements (heavier than standard glulam elements) result in a structure which is only 3% lighter (59,341 kg) compared to the ModuPark concept. The use of steel-concrete composite floor elements in the Morspoort car park significantly reduces the weight of the structure to 57% (35,110 kg) of the weight of the ModuPark concept. The fact that timber is a very efficient strength to weight material is shown by the B&O-Holzparkhaus. It has a weight of only 29% of that of the ModuPark concept at 17,547 kg. Finally, steel in combination with the extremely lightweight fibre-reinforced plastic floor elements, the concept of the Koopman int. car park, is the lightest at 10,477 kg which is only 17% of the weight of the ModuPark concept.

7.2.3 Load from Standard Column on Foundation

The third aspect on which the five car park concepts are compared is the load on a "standard" column on the foundation for 4 parking levels. This "standard" column is a column in the centre of the car park, which means it carries the load from two halve parking decks over a width of the distance between the columns, see Figure 78. Essentially, it carries the loads of a standard grid from 4 parking levels. The total load includes the self-weight of the structure, the finishing of the parking decks and the imposed loads on the parking decks.





The loads from a "standard" column on the foundation for 4 levels for the various car park concepts are presented in Figure 79 next to each other. This results in a different view on the comparison between the various variants compared to the weight of these variants. This is the result of different distances between the columns (parking deck span is the same for all variants) for the variants. A different distance between the columns affects many aspects of the construction, such as the dimensions of the cross-sections of girders and thickness of the floors. Therefore, it also affects the total deck height as well as the weight of the structure. The reason that it is important to also consider the load from a "standard" column on the foundation is that these loads in the end have to be supported by the foundation.

The heavy ModuPark concept similar to the total weight, also results in the highest column loads on the foundation at 2,472 kN with 5.0 m centres for 4 levels. Due to the smaller centres distances of 2.5 m for the BauBuche concept, its column load is at 48% of that from the ModuPark concept at 1,193 kN. For the more lightweight concepts, as a result of their lower self-weight, the imposed load becomes a bigger part of the total load on the foundation. The concept of the Morspoort car park, with centres of 5.0 m, results in a load of 1,845 kN (75% of ModuPark concept) of which 53% is due to the imposed load on the parking decks. The concept from the Koopman int. car park has a similar load to the BauBuche concept at 1,258 kN (51% of ModuPark concept), but it centres are at twice the distance (5.0 m), compared to the BauBuche concept. In this case, 78% of the total load comes from the imposed loads. Finally, the B&O-Holzparkhaus concept has the smallest loads on the foundation as a result of its materials and small centres distance of 2.5 m. Its load is 717 kN (29% of ModuPark concept) from which 68% is imposed load.



Figure 79. Comparison of load on a standard column on foundation for 4 levels for various modular or timber car park concepts

7.3 Concluding Remarks

In conclusion of this chapter, sub-question 3b can be partially answered: "What level of structural performance is requested from the design for a temporary multi-storey car park?"

It can be stated that the ModuPark concept has the smallest total deck height, but at the same time is the heaviest concept with the largest load of a column on the foundation. The total deck heights of other modular systems lie around 650 mm. Using primarily timber as building material results in a significant increase in total deck height. For structural systems in timber to compete in terms of total deck height with other modular systems, they should approach a total deck height around 650 mm. Therefore, a goal is set to limit the total deck height for the structural systems to be developed in this thesis to **700 mm** for a span of 16.26 m.

Considering the weight of the structure, the concepts discussed in this chapter have shown that the use of concrete in floors of the structure drastically increase the total weight. In order to reduce the weight of the structure and loads on the foundation, concrete should not be used in the floors. It is clear that a timber structure is much lighter compared to concepts in steel and concrete. Only the concept in steel and fibre-reinforced plastics performs better. In chapter 8, possible foundation methods for the car park structure are studied and a range of preferred maximum loads from the columns on the foundation can be presented.

8 Foundation

Like any structure, a car park requires a proper foundation. Since the objective is to design a multistorey car park with a temporary and modular character, this comes with several challenges. It is preferred to create a foundation that can be disassembled and reused. Furthermore, the car park can be located anywhere in The Netherlands, which creates a challenge for the design of the foundation, since it is typically designed for a specific location. In this chapter, first the different soil compositions in The Netherlands are discussed. Next, two types of foundations are discussed: a shallow foundation and a pile foundation. As a result, the advantages and disadvantages of both types of foundations are presented as well as an estimate of what range of loads can be carried by a certain type of foundation. This helps to create feeling for what the performance of the proof of concept should be, concerned with weight minimization as discussed in paragraph 5.1.2.2.

8.1 Soil Composition

The car park is not designed for a specific location, but should be applicable to a wide range of different locations and shapes. For this reason, the structure is designed in a modular and dismountable way. As the type of soil differs for each location, a set of boundary conditions should be established, for which the car park and its foundation can be applied in most locations. To establish these boundary conditions, the soil composition of twelve different locations in The Netherlands are analysed, see Figure 80. These locations are based on the general types of soil structures, found in The Netherlands.



Figure 80. Main soil types in The Netherlands (Florum, 2020)

The main types of soil found near the surface are marine clay, river clay, sand, loess and peat (Florum, 2020). Of these different types of soil, sand is the only one strong enough to build a foundation on. Whether it is possible to use a shallow foundation depends on the depth of the sand layer relative to the ground level. Through soil investigation, the build-up of the different soil layers can be found. Based on the soil data found in DINOloket (2022), the soil composition for the different selected locations from

Figure 80 can be established, see Figure 81. From this figure it can be deduced, that in the areas with sand soils, a shallow foundation can be applied. For the other areas, sometimes a sand layer can be found relatively close to ground level, in which case it is also possible to apply a shallow foundation after excavating the top layer. In locations, where the sand layer lies at larger depths, a pile foundation is required.



Figure 81. Soil composition per location (DINOloket, 2022)

The groundwater level in The Netherlands is several decimetres below ground level in most places. However, in some locations like the Veluwe, it can lie up to several meters below ground level (Ons Water, 2020). The depth of the groundwater level affects the load carrying capacity of a foundation. The higher the groundwater level, the lower the load carrying capacity of the soil, as result of a reduction of the effective stress (J.L. Bijnagte, A.F. Van Tol, & R. Elprama, 2006). Therefore, when establishing the boundary conditions for the load carrying capacity of the foundation, the level of the groundwater should be taken into account. Figure 82 shows the average smallest depth of the groundwater table for The Netherlands.



Figure 82. Groundwater table, average smallest depth 2021(BROloket, 2021)

8.2 Shallow Foundation

A shallow foundation can be applied in case a sand layer is found close to ground level (up to 2.0 m). It normally consist out of a concrete strip footing or isolated footing, which spreads the loads coming from the structure over the sand layer. The top layer of the soil is excavated to place the foundation directly on the sand layer and below the frostline in the soil. A shallow foundation is relatively cheap and causes little nuisance for the surrounding area (Bouwkosten Online, 2022). Furthermore, it can easily be removed when disassembling the construction. Below, a range concerning the load that can be carried by a shallow foundation for a single column of the car park is established.

8.2.1 Assumptions

To determine boundary conditions for the load carrying capacity of a shallow foundation, the following assumptions are made:

Footing

- The loads from a single column are supported by an isolated footing as type of shallow foundation;
- The footing is made from reinforced concrete;
- The footing is prefabricated;
- Dimensions footing: 2000 x 2000 x 200 mm (length x width x height);
- Dimensions concrete column on footing: 300 x 300 x 900 mm (length x width x height); (Column 300 mm above ground level to prevent moisture from deteriorating the timber columns)
- Weight concrete: 25.0 kN/m³;
- Two construction depths for the bottom face of the footing are studied: 800 mm and 2,000 mm below ground level.

Soil Profile

- Load carrying capacity of the shallow foundation is determined for the 12 different locations of Figure 80;
- No soil improvement techniques are considered;
- Groundwater level is at 200 mm below ground level for all locations;
- The footings are covered with loose sand till ground level.

Other

• Only vertical loads are considered.

8.2.2 Load Carrying Capacity

To determine the load carrying capacity of a shallow foundation on the different selected locations, cone penetration tests (CPT's) from the selected location are used, see appendix D.1. Based on the soil properties from the CPT's and given the assumptions discussed above, the load carrying capacity of the soil is calculated using Technosoft, see Table 23 and Table 24 (detailed output calculations in appendix D.2). The characteristic allowable load on a single column of the car park is found after subtracting the weight of the concrete footing, column and soil cover and taking into account the partial load factors. The load carrying capacity of the soil is calculated for two construction depths: 800 mm and 2,000 mm below ground level. Since the soil has different layers, this can result in significantly different load carrying capacities. Deformations are calculated for two different characteristic loads from a single column of the car park: 350 and 1,000 kN.

Location	Undrained	Drained	Punch	Minimum	Weight footing and cover	Allowed load car park	Deformation (for F _k = 350 kN)	Deformation (for F _k = 1,000 kN)	
-0.80 m	R₀ [kN]	R _d [kN]	R _d [kN]	Rd [kN]	Fd [kN]	F _k [kN]	[mm]	[mm]	
1	3848	499	668	499	74.3	353.9	230.4	342.1	
2	3686	306	339	306	74.3	193.1	614.7	931.4	
3	944	412	444	412	74.3	281.4	302.8	426.4	
4		1426	13419	1426	74.3	1126.4	5.4	8.9	
5	3750	520	586	520	74.3	371.4	438.7	647.9	
6		2066		2066	74.3	1659.8	3.3	6.6	
7		1812	6969	1812	74.3	1448.1	4.1	7.9	
8	1492	319	518	319	74.3	203.9	627.0	1073.1	
9	3740	496	587	496	74.3	351.4	569.0	881.6	
10	3686	592	660	592	74.3	431.4	397.3	585.3	
11		1521	4262	<mark>1</mark> 521	74.3	1205.6	4.7	9.2	
12	3686	820	1759	820	74.3	621.4	92.4	138.1	
	= large sand	layer near g	round level		= large loam layer near ground level				

Table 23. Load carrying capacity soil for different locations at construction depth -0.80 m

Table 24. Load carrying capacity soil for different locations at con	nstruction depth	-2.00 m
--	------------------	---------

Location	Undrained	Drained	Punch	Minimum	Weight footing and cover	Allowed load car park	Deformation (for F _k = 350 kN)	Deformation (for F _k = 1,000 kN)
-2.00 m	R _d [kN]	R _d [kN]	R _d [kN]	R _d [kN]	F _d [kN]	F _k [kN]	[mm]	[mm]
1	979	969	1116	969	144.4	687.2	123.3	178.6
2	430	385	462	385	144.4	200.5	490.3	696.2
3	430	870	1017	430	144.4	238.0	142.0	196.7
4		2064	9566	2064	144.4	1599.7	5.1	8.7
5	979	386	475	386	144.4	201.4	490.4	689.4
6		3002		3002	144.4	2381.4	2.7	6.2
7		2429	5010	2429	144.4	1903.9	4.7	8.8
8	979	467	616	467	144.4	268.9	549.1	985.8
9	796	457	591	457	144.4	260.5	648.0	899.3
10	979	498	776	498	144.4	294.7	444.3	630.2
11		1794	2995	1794	144.4	1374.7	6.3	11.5
12	1893	1315	1488	1315	144.4	975.5	67.0	117.0
	= large sand l	layer near g	round level			= large loa	m layer near grou	Ind level

From the results it becomes clear that at locations with sand layers near the surface (locations 4, 6, 7 & 11), the soil has significant load carrying capacity and loads result in only very small deformations (< 10 mm). For shallow foundations constructed at a depth of 0.80 m, the soil can easily support a load of 1,000 kN from a column of a car park and even over 1,600 kN, depending on the location. At a construction depth of 2.00 m, even larger loads can be supported in the range of 1,300 kN upwards to 2,300 kN. In the far south, the soil structure consists of a layer of loam on top of lime stone. These soils can also carry a significant load from a shallow foundation in the range of 600 kN for a construction depth of 0.80 m and close to 1,000 kN for a construction depth of 2.00 m. Deformations in this case are much larger compared to the sand layers, however for the allowable loads they are still smaller than 150 mm and therefore allowable. In terms of horizontal loading, for locations with sand layers near the surface, horizontal loads of up to more than 100 kN can be resisted (see Appendix D3).

The other types of soil structures have much smaller load carrying capacities compared to the sand layers. Furthermore, even smaller loads of 350 kN will result in large deformations (> 150 mm). as a result, these soil structures are unsuitable for a shallow foundation for a car park.

8.3 Pile Foundation

In case a shallow foundation does not generate enough load bearing capacity or results in too large deformations, a pile foundation is used. In this case, one or multiple piles are driven into the ground to reach into a deeper sand layer, which is strong enough to support the load on the pile. The load carrying capacity of a pile foundation comes from a combination of end bearing support and support by skin friction (Vroom Funderingstechnieken, 2019). Pile foundations are in general more expensive compared to a shallow foundation, can give a lot of nuisance for the surroundings and are often difficult to remove and reuse (Bouwkosten Online, 2022).

8.3.1 Types of Pile Foundations

Many different types of pile foundations exist. For this thesis, only timber driven piles and hammer driven prefab concrete piles are considered. The differences between both types in terms of advantages and disadvantages are discussed below.

Timber Foundation Piles

Advantages (Abebe & Smith, 2016; Civil Today, 2019)

- Sustainable material;
- Relatively cheap;
- Installation is easy;
- Low possibility of damage;
- Easy to pull out;
- Easy to trim to required length.

Disadvantages (Abebe & Smith, 2016; Civil Today, 2019)

- Length limited to ~21 m;
- Diameter limited to standard of 160 mm, although bigger diameters are possible;
- Not suitable to be used as end-bearing piles;
- Limited load bearing capacity (up to several hundreds of kN);
- Prone to biological attacks (rot).

Prefab Concrete Foundation Piles

Advantages (Abebe & Smith, 2016; Civil Today, 2019)

- High resistance against chemical and biological cracks;
- High load bearing capacity (up to several thousands of kN);
- Large lengths available (up to 39 m);
- Standard dimensions varying up to 500 x 500 mm.

Disadvantages (Abebe & Smith, 2016; Civil Today, 2019)

- Requires heavy machinery for installation;
- Installation by hammering causes a lot of nuisance for the surroundings;
- Possibility of breakage or damage during handling and driving of piles;
- Difficult to pull out;
- Relatively difficult to cut.

Considering the aim of this thesis to develop a structural system for a temporary and modular multistorey car park, like for the structural material for the superstructure, timber as a sustainable material is also the preferred option for the substructure. Furthermore, the characteristic that they are easy to pull out, makes them even more suited, as after the car park is disassembled, the building site should be returned to its original state. Because of the temporary character of the car park, rot of the timber piles is less of an issue, as long as they are thick enough to withstand their loads after some rot has occurred.

The limitations of timber piles concerning available lengths, diameters and load bearing capacity mean they cannot always be applied or require installation of more piles compared to prefab concrete piles. Whether timber piles are suitable largely depends on the build-up of the soil layers of a building site. Timber piles are most suitable as friction piles and therefore are most suitable to be used in soils which can generate sufficient cohesion between soil and pile (Abebe & Smith, 2016).

To determine a range of loads that can be supported by a timber pile foundation for a column of the car park, the maximum load bearing capacities of a timber foundation pile for the various locations of Figure 80 are calculated. Two different dimensions are used and as reference, also maximum loads for a prefab concrete pile are calculated.

8.3.2 Assumptions

To determine boundary conditions for the load carrying capacity of a pile foundation, the following assumptions are made:

Timber Pile

- Single pile load bearing capacity;
- Circular pile head;
- Diameter 160 mm or 200 mm;
- Maximum length 21 m;
- No concrete extension applied;
- Structural class C16.
 - Prefab Concrete Pile
- Single pile load bearing capacity;
- Square pile head;
- 400 x 400 mm;
- Maximum length 30 m;
- Structural class C35/45.

Soil Profile

- Load carrying capacities of the different pile foundations are determined based on CPT's for the 12 different locations of Figure 80;
- Pile tip at depth generating highest load bearing capacity (within maximum length of pile);
- Negative skin friction trajectory over compressible layers until sand layer in which pile head reaches;
- Positive skin friction trajectory over sand layer in which pile head reaches;
- Groundwater level is at 200 mm below ground level for all locations;
- A distributed load of 10.0 kN/m² is added to the surface to take into account the load from the lower parking level (8 cm paving stones, 10 cm sand, 20 cm rubble, variable load from vehicles).

Other

• Only vertical loads are considered.

8.3.3 Load Carrying Capacity

To determine the load carrying capacities for different pile foundations on the different selected locations, cone penetration tests (CPT's) from the selected location are used (presented in appendix D.1). Based on the soil properties from the CPT's and given the assumptions discussed above, including the skin friction trajectories, the load carrying capacity of the soil is calculated using Technosoft, see Table 25, Table 26, and Table 27 (detailed output calculations in appendix D.3). The geotechnical load bearing capacity as well as the structural load bearing capacity, based on the material assumptions given above, are calculated. The minimum value of both is prescriptive and after taking into account the partial load factors, the allowable load from a column of the car park given an single foundation pile are found.

				Timber	Ø160 mm				
Location	Depth pile tip	Share in total bearing capacity End bearing Friction		in total bearing capacity Pos. skin friction Pos. skin		bearing Reduction bearing Geotechnical Structural Minimun y capacity by neg. bearing bearing bearing ɔs. skin skin friction capacity capacity capacity iction		Minimum bearing capacity	Allowed load car park
	[m]	[%]	[%]	[%]	R _{c;netto;d} [kN]	R _{s;d} [kN]	R _d [kN]	F _k [kN]	
1	-4.0	100%	0%	78%	5	263	5	4	
2	-20.7	65%	35%	67%	91	263	91	76	
3	-5.9	52%	48%	10%	85	263	85	71	
4	-8.0	66%	34%	0%	257	263	257	214	
5	-14.3	37%	63%	5%	457	263	263	219	
6	-19.3	16%	84%	0%	659	263	263	219	
7	-14.5	31%	69%	0%	488	263	263	219	
8	-2.0	100%	0%	262%	-3	263	-3	-3	
9	-15.7	39%	61%	11%	288	263	263	219	
10	-20.1	53%	47%	49%	160	263	160	134	
11	-14.2	26%	74%	0%	326	263	263	219	
12	-9.3	42%	58%	14%	92	263	92	76	
	= soils witl	h large compress	ible layers in fi	rst 20 m		= soils with com enough logd ca	pressible layers rrvina sand laye	, but strong er in first 20 m	

Table 25. Load carrying capacity timber foundation pile (Ø160 mm) for different locations

Table 26. Load carrying capacity timber foundation pile (Ø200 mm) for different locations

				Timber	Ø200 mm					
Location	Depth pile tip	Share in total bearing capacity Pos. skin End bearing friction		Share in total bearing capacity Pos. skin Find bearing Capacity Pos. skin Friction Capacity Capacity by neg. Skin friction Capacity Capacity Capacity Capacity Capacity Capacity Skin friction Capacity Capacity Capacity Capacity Capacity Capacity Capacity Skin Skin friction Capacity		Share in total bearing capacity End bearing Pos. skin friction Reduction bearing capacity by neg. skin friction		Structural bearing capacity	Minimum bearing capacity	Allowed load car park
	[m]	[%]	[%]	[%]	R _{c;netto;d} [kN]	R _{s;d} [kN]	R _d [kN]	F _k [kN]		
1	-4.0	100%	0%	66%	11	411	11	9		
2	-20.6	70%	30%	58%	171	411	171	142		
3	-5.7	57%	43%	10%	111	411	111	93		
4	-7.8	70%	30%	0%	374	411	374	312		
5	-14.1	44%	56%	5%	616	411	411	342		
6	-19.1	20%	80%	0%	850	411	411	342		
7	-14.3	33%	67%	0%	611	411	411	342		
8	-2.0	100%	0%	214%	-4	411	-4	-3		
9	-15.5	45%	55%	10%	385	411	385	321		
10	-19.9	60%	40%	45%	240	411	240	200		
11	-14.0	29%	71%	0%	419	411	411	342		
12	-9.1	48%	52%	13%	122	411	122	102		
	= soils with	h large compress	ible layers in fi	rst 20 m		= soils with com enough load ca	pressible layers, rrying sand laye	, but strong er in first 20 m		

				Concr	ete #400			
Location	Depth pile tip	Share in total bearing capacity End bearing friction		Reduction bearing capacity by neg. skin friction	Geotechnical bearing capacity	Structural bearing capacity	Minimum bearing capacity	Allowed load car park
	[m]	[%]	[%]	[%]	R _{c;netto;d} [kN]	R _{s;d} [kN]	R _d [kN]	F _k [kN]
1	-28.2	61%	39%	58%	341	3728	341	284
2	-27.1	58%	42%	24%	1884	3728	1884	1570
3	-6.1	56%	44%	8%	327	3728	327	273
4	-7.1	83%	17%	0%	1703	3728	1703	1419
5	-13.2	56%	44%	5%	1742	3728	1742	1451
6	-16.0	44%	56%	0%	2597	3728	2597	2164
7	-13.4	37%	63%	0%	1543	3728	1543	1286
8	-26.8	57%	43%	40%	866	3728	866	721
9	-14.6	60%	40%	9%	1182	3728	1182	985
10	-23.7	42%	58%	37%	865	3728	865	721
11	-13.1	43%	57%	0%	1178	3728	1178	981
12	-9.3	51%	49%	12%	355	3728	355	296
	= soils wit	h large compress	ible layers in fi	rst 20 m		= soils with com enough logd ca	pressible layers rrvina sand lay	, but strong er in first 20 m

Table 27. Load carrying capacity concrete foundation pile (#400 mm) for different locations

For the bearing capacity of pile foundations, the locations with compressible soil layers are of relevance (highlighted in tables). From the results it becomes clear that the load bearing capacity of a single timber pile is rather limited. Using prefab concrete piles, which can have a larger cross-section and can be installed at larger depths, can result in much higher load bearing capacities.

Since the range of load bearing capacities of timber piles for the different locations with compressible soil layers is rather broad (0 to 411 kN), it is difficult to give a number for the load coming from a column of the car park that can be supported by a timber pile foundation. To gain insight into the build-up of the bearing capacity of a pile, the percentages of bearing capacity realised by end bearing and positive friction are given. Also, a percentage is given which expresses how much the total bearing capacity of a pile based on end bearing and positive skin bearing capacity is reduced by the negative skin friction. The results show the build-up of the bearing capacity is highly dependent on the structure of the soil layers and based on these limited number of locations, it is difficult to give a well-founded conclusion on how the bearing capacity in general is build-up. What can be concluded is that timber piles are not suitable for soil structures with mainly compressible soil layers within the first 20 metres (light shading in tables).

For soil structures with bigger sand layers within the first 20 metres (dark shading in tables), a minimum load carrying capacity of 140 kN for a timber pile is possible. When a pile group is used, the load carrying capacity might increase as a result of soil compaction. To carry a load from a column of the car park of 1,000 kN would require roughly 7 timber piles. Depending on the soil structure of the location it might be less. For locations, in which timber piles are not suitable, concrete piles should be used.

8.4 Conclusion

In conclusion of this chapter, the remaining part of sub-question 3b, related to the performance on weight minimization can be answered: "What level of structural performance is requested from the design for a temporary multi-storey car park?".

Based on the results presented above, it is concluded that a load of a column from the car park of **1,000 kN** can be supported by the following foundations:

- A shallow foundation in case large sand layers are found near the surface and no large compressible soil layers are found close to this sand layer;
- A timber pile foundation with up to 7 piles in case within the first 20 m, a sand layer with large thickness is found;
- A prefab concrete pile foundation with up to 4 piles in case within the first 20 m, only sand layers of insufficient thickness can be found to support timber piles.

Depending on the exact location of the car park, a smaller or simpler foundation might also be possible. For a shallow foundation, the range of loads that can be carried by the soil varies between roughly 600 up to over 2,300 kN, depending on location and depth of the foundation. At most locations, a load of 1,000 kN can easily be carried by the soil. For pile foundations, at halve of the location studied, a load of 200 kN can be carried by a single timber foundation pile, which would result in a total of 5 piles to carry a load of 1,000 kN. At other locations, more piles or concrete piles are required.

PART IV: Variant Analysis

9 Deck System

In the literature study, various types of timber deck systems have been studied, see paragraph 4.2.2. In this chapter, these deck systems are further studied in detail, to assess their performance on height and weight if to be used to construct the parking decks of the multi-storey car park. Furthermore, the systems are studied on how they perform when different requirements are used considering the ultimate and serviceability limit states. The deck systems are divided into long-span and short-span systems, studied after each other. Finally, this chapter results in a choice for a certain deck system for each design variant as presented in 5.3.2.

9.1 Long-span Prefab Deck System

The only design variant considered to have a long-span deck system is variant 1 (see paragraph 5.3.2.1). Depending on the chosen span of the parking deck, it requires a deck system to span between 12.18 and 17.66 m. To bridge such a long span, several possible prefab timber deck systems can be used, as presented in paragraph 5.3.1.1. These are CLT or LVL decks with sometimes glulam or LVL ribs: CLT panel, CLT open rib panel, CLT closed rib panel, LVL panel, LVL open rib panel, LVL semi-open rib panel and LVL closed rib panel. These seven long-span deck systems are discussed in the following paragraphs, in which the required height of the deck is determined for a base case, in which all standard rules and norms apply and various other cases, for which one or multiple requirements are altered. Finally, the results will provide insight into the performance of the different long-span deck systems considering their weight and height as well as what effect changing parameters and/or requirements have on those aspects.

9.1.1 General

For design variant 1, floor slabs of 2500 mm wide are considered, spanning the entire parking deck and simply supported at both ends, see Figure 83. Concerning the loads on the floors, the top level parking deck is considered as it is normative, which results in the following loads acting on the deck: self-weight, permanent load from parking deck coating, imposed load, snow load and wind load. The values of the loads are mentioned in the starting points in chapter 6 and in Table 28. For the wind load, the downward action is governing in the load combinations, for which the coefficient for downwards wind loading is used for the entire parking deck.

	Wind	
	Snow	
	Imposed	
	Permanent	
	Self-weight	
\mathbf{X}		
111111.		
	Main span [m]	



Load	Value
Self-weight	Differs per type and span
Permanent	0.1 kN/m ²
Imposed	2.0 kN/m ²
Snow	0.56 kN/m ²
Wind	0.36 kN/m ²

All of the considered long-span deck systems are studied for their required thickness for a different number of span lengths. From the required deck thickness also follows the weight of the deck. First, the decks are studied for the basic requirements for ULS, SLS, fire and vibrations as stated in chapter 6. Next, the requirements concerning SLS, fire and vibrations are altered in accordance with the overview of parameters in paragraph 0. Finally, the different deck systems are compared to each other.

For the long-span deck systems, the standard CLT and LVL products of manufacturer Stora Enso are considered (Stora Enso, 2022b, 2022e). The calculations are made using the Calculatis software package by Stora Enso (Stora Enso, 2022a). Examples of calculations can be found in Appendix F.

9.1.2 CLT Panel

The first studied system makes use of a single CLT panel, which forms the deck, see Figure 84. Based on calculations on the standard CLT panels of Stora Enso (Stora Enso, 2022e), it is concluded that for the base case, even for the smallest span of 12.18 m, the largest standard CLT panel (CLT 320 L8s-2) does not fulfil the ULS requirements (see appendix F.1). Only with reduced SLS requirements and no check on vibrations, would the largest panel be sufficient, but only for a span of 12.18 m. It is therefore concluded, that a single solid CLT panel is not an option for the long-span prefab deck system.



Figure 84. CLT panel (Stora Enso, 2022f).

9.1.3 CLT Open Rib Panel

The CLT open rib panel consist out of a CLT panel with several glulam ribs glued to this panel, resulting in a ribbed deck with a full shear connection between rib and panel, see Figure 85. This system is somewhat similar to the TT-slabs in concrete, but in a timber variant.

A CLT open rib panel can be realised in many different configurations. Different CLT panels (thickness and direction of layers) can be used, different crosssections for the ribs as well as its material quality can be chosen and the distance between the ribs can be varied. Since this results in an extremely large number of possible configurations, this design space has to be narrowed down. Based on the product advice of Stora Enso (Stora Enso, 2022f), given a fire resistance class R90, a panel thickness of 120 mm is chosen, a rib width of 200 mm and a rib spacing of 625 mm. The height of the rib follows from a load calculation for the different parking deck spans. The parameters advised by Stora Enso are a trade-off between minimizing the total deck height and minimizing the weight while fulfilling the fire-safety requirements. This has resulted in the advised parameters.



Figure 85. CLT open rib panel (Stora Enso, 2022f).

9.1.3.1 Base Case

Depending on the span distance of the parking deck, the required rib height of the CLT open rib panel varies between 360 and 800 mm and the total deck height between 480 and 920 mm, see Table 29. In this table, for each specific parking deck span, the required thickness of the ribs of the deck for each requirement (ULS, ULS fire, SLS & vibrations, see chapter 6) is presented. Also the self-weight of the panels is shown for the governing requirement. Both the total deck height and self-weight for a 2500 mm wide deck element are plotted against the parking deck span in Figure 86. The graph shows a slightly increasing exponential increase in deck height and weight for increasing spans. Depending on the specific parking deck span, a certain rib height might be just sufficient (UC almost 1) or might have some margin (0.8 < UC < 1.0). As a result, the parking deck spans of 16.26 m and 17.16 m require the same CLT open rib panel.

From Table 29 it can be clearly seen that the vibrational criteria are governing for all spans. Therefore this requirement results in an increase of the deck height and weight compared to the other requirements.

CLT OPEN RIB PANEL BASE CASE Governing rib height L Spacing CLT_{Top panel} $\mathbf{h}_{\mathsf{Rib}}$ h_{tot} WRib **g**0,k **g**_{0,k,tot}(2500mm) Vibrations **JLS Fire** ULS SLS [m] [mm] [mm] [mm] [mm] [mm] [kN/m] [kN/m]<200 <200 300 360 12.18 625 120 200 360 480 1.18 2.94 13.70 200 450 <200 <200 360 450 625 120 570 1.32 3.30 14.56 625 120 200 500 620 1.40 3.50 <200 <200 400 500 15.61 625 120 200 600 720 1.56 3.90 200 200 450 600 120 200 700 240 500 700 16.26 625 820 1.72 4.30 240 17.16 625 120 200 700 820 1.72 4.30 240 240 500 700 200 17.66 625 120 800 920 1.88 4.70 300 300 600 800





Figure 86. Total deck height for CLT open rib panels for various spans.

9.1.3.2 Alternative Requirements

In previous paragraph, it was already stated that the vibrational requirement is governing for all span distances and therefore altering of this requirement can result in significant reduction of the total deck height and weight. Table 29 already showed the different values, from which can be seen that the differences in rib height for the various requirements differ significantly.

Since serviceability limits are not required by the Dutch Building Degree, they can be altered by the engineer (Ministerie van Binnenlandse Zaken en Koninkrijkrelaties, 2021). The fire resistance class can be reduced to R 30 if the entire car park is considered as a single fire-compartment. This is possible based on the equivalence principle, which can be substantiated by special fire-safety calculations like zone-models as was discussed in paragraph 2.3.4. Also deformation limits can be altered, allowing bigger deformation than advised by the Eurocode. This is allowed by the Dutch Building Degree as discussed in paragraph 6.4. Last of all, also setting limitations related to vibrational behaviour of the structure of a car park is up to the designer as discussed in paragraph 2.3.3. In this paragraph, it is studied what are the effects of reduced requirements for fire-safety, SLS and vibrations on the required thickness of the CLT decks. One or more of these requirements are altered, while the other requirements remain the same (equivalent to the base case requirements).

In Figure 87, the effects of different alternative requirements on the total deck height of the CLT open rib panels can be seen for the various parking deck spans. The results that are at the basis of these figures in tabular format as well as an example calculation can be found in Appendix F.

Reduced Fire-safety Requirements

A reduction of the fire-safety requirements from class R90 to R30 does not result in any reduction of total deck height or weight, since the normal ULS strength requirements require the same rib heights as the R90 ULS fire strength requirements.

No check on vibrations

Since the requirements for vibrations are governing, removing the check on vibrations significantly reduces the total deck height and weight. In this case, the SLS deformation limits become governing. This results in a reduction of total deck height of 60 mm (13%) for the smallest span up to 200 mm (24%) for the spans of 16.26 and 17.16 m. The percentual decrease in weight is slightly smaller; 0.24 kN/m (8%) for the span of 12.18 m and 0.80 kN/m for the three largest spans (19% & 17%).

Alternative Serviceability Limit Requirements

As the SLS deformation criteria are governing after the vibrational criteria, altering these requirements can result in a further decrease of total deck height and weight. A further reduction of 60 mm (14%) in height for a span of 12.18 m and 200 mm (28%) for a span of 17.66 m is possible. In terms of weight reduction these numbers are 0.24 kN/m (9%) for a 12.18 m span and 0.80 kN/m (21%) for a 17.66 m span.

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

As already stated before, reduced fire-safety requirements do not result in a decrease in total deck height and weight.

Conclusion

The results presented in this paragraph clearly show the significant effects of alteration of various requirements concerned with serviceability limit states. For all spans, removing the vibrations check already significantly improves the performance of the decks in terms of height and weight. Since rib decks are used, the percentual effects are bigger for the total deck height compared to the weight. Further reducing the SLS requirements can result in a total reduction of height of 120 mm (25%) up to 400 mm (43%) for the largest span and 0.48 kN/m (16%) up to 1.60 kN/m (34%) respectively.



Figure 87. Differences in total deck height for various spans given alternative requirements for CLT open rib panel.

9.1.4 CLT Closed Rib Panel

The CLT closed rib panel consist out of a top CLT panel, bottom CLT panel and several glulam ribs glued to these panels, resulting in a closed rib deck with a full shear connection between ribs and panels, see Figure 88. This system is also often called a box floor.

Closed CLT rib panels can be realised in many different configurations, similar to the open CLT rib panels. Stora Enso does not advice use of specific configurations of the closed rib panels. Therefore, several parameters for the configuration of the closed CLT rib panels are based on the configuration of the open CLT rib panel. For both the top and bottom CLT panel, a thickness of 120 mm is chosen given the R90 fire resistance class. For the ribs, again a spacing of 625 mm is used, but the width of the ribs is reduced to 140 mm, since they are covered by CLT panels on both sides and therefore are protected from fire. The required rib height again follows from calculation for the different parking deck spans.



Figure 88. CLT closed rib panel (Stora Enso, 2022f).

9.1.4.1 Base Case

The required rib heights for the various spans as well as resulting total deck height for the closed CLT rib panels can be found in Table 30. The rib heights vary between 200 mm and 600 mm, or a total deck height of 440 to 840 mm. The table also shows what the required rib heights are for the different requirements for the base case. In Figure 89, the total deck height and self-weight of the deck are plotted against the parking deck span. The graph shows a roughly linear increase in height and weight for larger spans. Both the decks for a span of 17.16 and 17.66 can use the same deck.

Similar to the CLT open rib panels, for all spans, the criterion for vibrations is governing. Table 30 shows the significant potential for reduction of total deck height and weight if the vibrational requirement is removed.

116

Table 30. CLT closed rib panels for various span distances and the required rib height for each requirement.

	CLT CLOS	ED RIB	PANEL		BASE CASE							
L	Spacing	CLT Top panel	CLT Bottom panel	WRib	h _{Rib}	h _{tot}	g 0,k	g 0,k,tot(2500mm)	Gov	erning	; rib he	ight
[m]	[mm]	[mm]		[mm]	[mm]	[mm]	[kN/m]	[kN/m]	NLS	ULS Fire	SLS	Vibrations
12.18	625	120	120	140	200	440	1.42	3.56	<200	<200	<200	<200
13.70	625	120	120	140	300	540	1.54	3.84	<200	<200	240	300
14.56	625	120	120	140	360	600	1.60	4.01	<200	<200	300	360
15.61	625	120	120	140	450	690	1.70	4.26	<200	<200	300	450
16.26	625	120	120	140	500	740	1.76	4.40	<200	<200	360	500
17.16	625	120	120	140	600	840	1.87	4.68	<200	<200	360	600
17.66	625	120	120	140	600	840	1.87	4.68	<200	<200	400	600



Figure 89. Total deck height for CLT closed rib panels for various spans.

9.1.4.2 Alternative Requirements

The impact of altering the requirements on the total deck height and weight can be seen in Figure 90 and are discussed below. The results that are at the basis of Figure 90 in tabular format can be found in Appendix F.

Reduced Fire-safety Requirements

A reduction of the fire-safety requirements from class R90 to R30 does not result in any reduction of total deck height or weight, since the normal ULS strength requirements require the same rib heights as the R90 ULS fire strength requirements.

No check on vibrations

As the vibration requirement is governing for all spans, removal of this requirement results in a decrease of total deck height and weight for all but one span. Only for the shortest span of 12.18 m, no reduction can be seen in the graph. This is due to the minimal applied rib height of 200 mm. For smaller rib heights, the CLT closed rib panel is not seen as an efficient panel. For a span of 13.70 m, a reduction

in total deck height of 60 mm (11%) and weight of 0.17 kN/m (4%) is achieved. The largest reduction in height and weight is found for a span of 17.16 m: a reduction in height of 240 mm (29%) and a reduction in weight of 0.67 kN/m (14%).

Alternative Serviceability Limit Requirements

Altering the deformation limits on top of the removal of the vibration requirement can further reduce the total deck height and weight. A further reduction for a span of 13.70 m of 40 mm (8%) in height and 0.11 kN/m (3%) in weight is possible. The largest reduction is found for a span of 17.16 m: a further reduction of 120 mm (20%) in height and 0.34 kN/m (8%) in weight.

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

As already stated before, reduced fire-safety requirements do not result in a decrease in total deck height and weight.



Figure 90. Differences in total deck height for various spans given alternative requirements for CLT closed rib panel.

Conclusion

From the results in this paragraph it has become clear, that altering the SLS requirements significantly reduces the total deck height and weight. The biggest reduction in height and weight is the result of removing the check for vibrations. Also allowing larger deformations results in a reduction of deck height and weight. A lower fire-safety class has no effect on the deck height and weight. The shortest span excluded, all spans see a significant reduction of height, varying from 100 mm (19%) for the 13.70 m span to 360 mm (43%) for the 17.16 m span. In terms of weight, a reduction of 0.28 kN/m (7%) is found for a span of 13.70 m up to a reduction of 1.01 kN/m (22%) for a span of 17.16 m.

9.1.5 LVL Panel

Thick LVL panels can also act as floor elements. For this, multiple LVL panels are glued on top of each other to create one thick strong LVL panel. A second layer of LVL typically runs across the grain at a 90° angle, to reinforce the veneer (Stora Enso, 2022d), see Figure 91. The result is a panel, which can support perpendicular loading and be used as a floor element.



Figure 91. LVL panel (Stora Enso, 2022f).

9.1.5.1 Base Case

Applying a solid LVL panel to span the large spans of the parking deck results in rather heavy elements. Given the requirements for the base case, it is not possible to span all parking deck spans using the

standard LVL panels of Stora Enso (Stora Enso, 2022e), which have a maximum thickness of 600 mm. Up to a span of 15.61 m, it is possible to use solid LVL panels with a thickness varying from 480 mm for the smallest span of 12.18 m, up to 600 mm for a span of 15.61 m, see Table 31 and Figure 92.

Similar to the CLT deck elements, in all cases for the solid LVL panel, the criteria for vibrations are governing. Alteration of these requirements can therefore result in a decrease of the height and weight of the LVL panel.

	BASE CASE									
L	Name _{LVL}	Nr. LVL panels	h _{LVL}	g 0,k	Normative			e		
[m]	[-]	[-]	[mm]	[kN/m²]	NLS	ULS Fire	SLS	Vibrations		
12.18	LVL G 480 8s	8	480	2.45	168	168	360	480		
13.70	LVL G 540 9s	9	540	2.75	192	180	420	540		
14.56	LVL G 540 9s	9	540	2.75	210	180	480	540		
15.61	LVL G 600 10s	10	600	3.06	216	180	540	600		
16.26	n.a.*	-	-	-	240	210	540	>600		
17.16	n.a.*	-	-	-	252	210	600	>600		
17.66	n.a.*	-	-	-	252	210	600	>600		

Table 31. LVL panels for various span distances and the required height for each requirement.

* none of the standard panels by Stora Enso fulfils the minimum requirements of the base case



Figure 92. Height of LVL panel for various spans.

9.1.5.2 Alternative Requirements

The impact of altering the requirements on the total deck height and weight can be seen in Figure 93 and is discussed below. The results that are at the basis of Figure 93 in tabular format can be found in Appendix F.

Reduced Fire-safety Requirements

Altering the fire-safety class from R90 to R30 does not result in any reduction of the height of the LVL panels, since the normal ULS strength requirements require the same or thicker panels as can be seen in Table 31.

No check on vibrations

Removal of the check on vibrations results in a reduction of the deck height especially for the smaller spans. Furthermore, it allows the larger spans also to be within the range of the standard LVL panels. The biggest decrease in height is for the smallest span of 12.18 m, with a decrease in height and weight of 25% (120 mm and 0.61 kN/m²). For a span of 15.61 m, it is a reduction of 10% (60 mm and 0.31 kN/m²).

Alternative Serviceability Limit Requirements

Altering the serviceability limit requirements, including both vibrations and deformations, results in a further decrease in height and weight of the LVL panels. This reduction can be seen for all spans and lies around 20% for both height and weight. For the smallest span of 12.18 m, this is a reduction 66 mm in height and 0.34 kN/m² in weight, while for all spans larger than 13.70 m, a height reduction of 120 mm and weight reduction of 0.61 kN/m² is realised.

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

As already stated before, reduced fire-safety requirements do not result in a decrease in total deck height and weight.





Conclusion

From this paragraph it can be concluded that altering the serviceability limit requirements again can result in a significant reduction of height and weight of the LVL panels and can bring all parking deck spans into the range of the standard LVL panel dimensions. Similar to the earlier presented deck systems, removing the check on vibrations results in a significant decrease in deck height and weight. Also allowing larger deformations results in a reduction. A lower fire-safety class does not affect the total deck height and weight. The biggest reductions are found for the smaller spans at 39% in total (204 mm in height and 1.04 kN/m² in weight). For the larger spans, these values are slightly smaller.

9.1.6 LVL Open Rib Panel

The LVL open rib panel is similar to the CLT open rib panel, but instead of a CLT panel as deck, a LVL panel is used. For the ribs, again LVL is used instead of glulam ribs. Again, a full shear connection is realised between deck and ribs, resulting in a strong rib panel, able to carry high loads, see Figure 94.

Also for the LVL open rib panel, many different types of configurations can be created. In comparison with the CLT variant, the thickness of the deck and width of the ribs are much smaller as the result of the characteristics of the LVL material. However, this does pose a problem concerning fire-safety of the elements. Whereas with the CLT variant, the firesafety requirements are met through the larger thickness of the elements, this is not possible for the thin LVL elements. Therefore, it is required to add 57 mm of gypsum board, covering the ribs of the panel, to provide a resistance against fire for the class of R90 (Stora Enso, 2023).



Figure 94. LVL open rib panel (Stora Enso, 2022f).

The parameters for the configuration of the LVL open rib panel are chosen based on the advice of Stora Enso (Stora Enso, 2022f). This has resulted in the use of a top panel with a thickness of 37 mm and a rib width of 51 mm. Considering the spacing between the ribs, two options are studied; a spacing of 625 mm and a spacing of 500 mm. The required rib height follows from calculation for the different parking deck spans.

9.1.6.1 Base Case

The results for the base case for both a spacing of 625 mm as well as for a spacing of 500 mm are presented in Table 32. From the results it becomes clear that the LVL open rib panel is not able to span the various parking deck spans, given the thickness of the elements and spacing stated above for the standard maximum height of the ribs of 600 mm. The table however also shows that in all cases, the criteria for vibrations are governing and in some cases also the deformation limits. However, when both criteria are governing, the vibrational criteria are limiting more than the deformation limits. No differences in required rib heights can be seen for the different spacings.

	LVL OPEN	I RIB PANEL		BASE CASE								
L	Spacing		WRib	h _{Rib}	h _{tot}	g 0,k	g 0,k,tot(2500mm)	Go	Governing rib height			
[m]	[mm]	[mm]	[mm]	[mm]	[mm]	[kN/m]	[kN/m]	NLS	SLS	Vibrations		
12.18	625	37	51	n.a.*	-	-	-	300	500	>600		
13.70	625	37	51	n.a.*	-	-	-	360	600	>600		
14.56	625	37	51	n.a.*	-	-	-	400	600	>600		
15.61	625	37	51	n.a.*	-	-	-	450	>600	>600		
16.26	625	37	51	n.a.*	-	-	-	450	>600	>600		
17.16	625	37	51	n.a.*	-	-	-	450	>600	>600		
17.66	625	37	51	n.a.*	-	-	-	500	>600	>600		
12.18	500	37	51	n.a.*	-	-	-	300	500	>600		
13.70	500	37	51	n.a.*	-	-	-	360	600	>600		
14.56	500	37	51	n.a.*	-	-	-	360	600	>600		
15.61	500	37	51	n.a.*	-	-	-	400	>600	>600		
16.26	500	37	51	n.a.*	-	-	-	400	>600	>600		
17.16	500	37	51	n.a.*	-	-	-	450	>600	>600		
17.66	500	37	51	n.a.*	-	-	-	450	>600	>600		

Table 32. LVL open rib panels for various span distances and the required rib height for each requirement.

* none of the standard panels by Stora Enso fulfils the minimum requirements of the base case

9.1.6.2 Alternative Requirements

As mentioned above, altering the requirements can bring the LVL open rib panel into the range of possible deck systems, for its standard dimensions. For the configurations chosen for this deck system, for most spans, LVL open rib panels are an option if alternative requirements are used. A comparison for the height for different spans and situations is presented in Figure 95. Below, it is studied what are the effects of reduced requirements for fire-safety, SLS and vibrations on the required thickness of the LVL decks. One or more of these requirements are altered, while the other requirements remain the same (equivalent to the base case requirements). The results that are at the basis of Figure 95 in tabular format can be found in Appendix F.

Reduced Fire-safety Requirements

Since the fire-safety requirements are met by addition of gypsum board to the panel, reducing the firesafety requirements results in a different required thickness of the gypsum board. For a fire resistance class R30, gypsum boards of 42 mm thick are required (Stora Enso, 2023). This is a reduction of 15 mm in thickness. Logically, this does not have significant results for the total height and weight of the panel. Only reducing the fire-safety requirements therefore does not bring the parking decks into the range of possible standard configurations for the LVL open rib panel.

No check on vibrations

By removal of the check on vibrations, parking deck spans of 12.18 m up to 14.56 m become possible with LVL open rib panels. No difference is found for the various spacings applied.

Alternative Serviceability Limit Requirements

When applying lower deformation limits on top of the removal of the check on vibrations, this brings almost all parking deck spans into the range of the LVL open rib panels for the chosen configurations. This time, a difference is found between applying a spacing of 625 or 500 mm. The spacing of 500 mm performs better, reducing the total height of the panel, while only adding an insignificant amount of weight. Also with a spacing of 500 mm, it is possible to span also the longest span, which is not possible for a spacing of 625 mm. For the spans, which could already be covered by only removal of the vibration requirements, a height reduction between 100 and 150 mm (14-22%) is achieved and a weight reduction of 0.11 to 0.16 kN/m (6-9%).

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

Additionally reducing the fire-safety requirements does not result in a significant decrease in height and weight of the panel as this only results in a small reduction of the height and weight of the gypsum board.

Conclusion

In conclusion of this paragraph, it can be stated that alteration of the requirements is necessary to bring the different parking deck spans into the range of the LVL open rib panels. Only by altering all of the serviceability limit requirements it becomes possible to span also the longest parking deck span. A lower fire-safety class has no significant effect at the total deck height and weight. Removal of the check on vibrations is required to bring the different spans even into the range of available standard panels by Stora Enso. Reduction of the deformation limits results in a reduction of total deck height and weight.



* no standard panel possible for base case scenario and therefore not in graph | no standard panel possible for case without check on vibrations for spans 15.61 m to 17.66 m and therefore not in graph | no standard panel possible with spacing of 625 mm for case with R30 fire-safety class, reduced deformation limits and no check on vibrations for 17.66 m span and therefore not in graph

Figure 95. Differences in total deck height for various spans given alternative requirements for LVL open rib panels.

9.1.7 LVL Semi-open Rib Panel

In addition to the LVL open rib panel, the LVL semi-open rib panel features an extra bottom tension flange to the ribs. The result is an I-shaped section, comprised of a LVL top panel, rib and bottom flange, fully structurally bonded together, see Figure 96. The bottom flange improves the structural performance of the panel, decreasing the height of the panel and making it possible to reach longer spans (Stora Enso, 2022f).

The LVL semi-open rib panels can of course also be created in many different configurations. Similar to the LVL open rib panels, the same parameters are used with the addition of the dimensions for the bottom flange. For this flange, a LVL panel with a thickness of 43 mm is used, which has a width of 300 mm, which is advised by Stora Enso (Stora Enso, 2022f). Also for the semi-open rib panel, addition of 57 mm of gypsum board is required to fulfil the requirements of the R90 fire-safety class. The required rib height is calculated for the different parking deck spans.



Figure 96. LVL semi-open rib panel (Stora Enso, 2022f).

9.1.7.1 Base Case

The results for the base case for both a spacing of 625 mm as well as for a spacing of 500 mm are presented in Table 33. From the results it becomes clear that the LVL semi-open rib panel is not able to span the various parking deck spans, given the thickness of the elements and spacing stated above for the standard maximum height of the ribs of 600 mm. The table however also shows that in all cases, the criteria for vibrations are governing and in one case also the deformation limits. However, when both criteria are governing, the vibrational criteria are limiting more than the deformation limits. Some differences in required rib heights can be seen for the different spacings.

	LVL SEN	1I-OPEN R	IB PANEL		BASE CASE							
L	Spacing	LVL _{Top}	LVL _{Bottom}	WRib	h _{Rib}	h _{tot}	ot g 0,k g 0,k,tot(2500mm)		Governing rib height			
[m]	[mm]	[mm]	[mm]	[mm]	[mm]	[mm]	[kN/m]	[kN/m]	NLS	SLS	Vibrations	
12.18	625	37	43	51	n.a.*	-	-	-	240	360	>600	
13.70	625	37	43	51	n.a.*	-	-	-	240	450	>600	
14.56	625	37	43	51	n.a.*	-	-	-	300	500	>600	
15.61	625	37	43	51	n.a.*	-	-	-	300	600	>600	
16.26	625	37	43	51	n.a.*	-	-	-	300	600	>600	
17.16	625	37	43	51	n.a.*	-	-	-	300	600	>600	
17.66	625	37	43	51	n.a.*	-	-	-	300	>600	>600	
12.18	500	37	43	51	n.a.*	-	-	-	200	360	>600	
13.70	500	37	43	51	n.a.*	-	-	-	200	450	>600	
14.56	500	37	43	51	n.a.*	-	-	-	240	450	>600	
15.61	500	37	43	51	n.a.*	-	-	-	240	500	>600	
16.26	500	37	43	51	n.a.*	-	-	-	240	600	>600	
17.16	500	37	43	51	n.a.*	-	-	-	240	600	>600	
17.66	500	37	43	51	n.a.*	-	-	-	300	600	>600	

Table 33. LVL semi-open rib panels for various span distances and the required rib height for each requirement.

* none of the standard panels by Stora Enso fulfils the minimum requirements of the base case

9.1.7.2 Alternative Requirements

In Table 33 it was already shown, that for alternative requirements, the LVL semi-open rib panel becomes a potential deck system for all spans (depending on the alterations). A comparison for the height for different spans and situations is presented in Figure 97. The results that are at the basis of Figure 97 in tabular format can be found in Appendix F.

Reduced Fire-safety Requirements

Since the fire-safety requirements are met by addition of gypsum board to the panel, reducing the firesafety requirements results in a different required thickness of the gypsum board. For a fire resistance class R30, gypsum boards of 42 mm thick are required (Stora Enso, 2023). This is a reduction of 15 mm in thickness. Logically, this does not have significant results for the total height and weight of the panel. Only reducing the fire-safety requirements therefore does not bring the parking decks into the range of possible standard configurations for the LVL semi-open rib panel.

No check on vibrations

Removal of the check on vibrations brings almost all parking deck spans in the range of the LVL semiopen rib panels given the configuration of these panels. Only the longest span of 17.66 m is not possible when a spacing of 625 mm is used, but is possible for a spacing of 500 mm. Furthermore, a difference between both spacings is found for a span of 14.56 mm, in which case the spacing of 500 mm outperforms the larger spacing in a height difference of 50 mm.

Alternative Serviceability Limit Requirements

Altering the serviceability limit requirements, including both vibrations and deformations, results in a further decrease in height and weight of the LVL semi-open rib panels. A difference between both spacings is only found for a span of 17.16 m, in which case the smaller spacing outperforms the larger one in terms of height by 50 mm. For all spans, a significant reduction in height is found, which is the biggest for the spans in the middle of the set of spans considered. For the smallest span, the reduction

in height is 60 mm (12%), while the weight reduction is rather small at 0.06 kN/m (3%). For a span of 15.61 m, the reduction in height is the biggest at 200 mm (27%), which gives a weight reduction of 0.21 kN/m (10%).

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

Additionally reducing the fire-safety requirements does not result in a significant decrease in height and weight of the panel as this only results in a small reduction of the height and weight of the gypsum board.



* no standard panel possible for base case scenario and therefore not in graph | no standard panel with spacing of 625 mm possible for case without check on vibrations for span 17.66 m and therefore not in graph

Figure 97. Differences in total deck height for various spans given alternative requirements for LVL semi-open rib panels.

Conclusion

The results presented in this paragraph have shown that alternative requirements make it possible to span all parking deck spans with an LVL semi-open rib panel. Altering the fire-safety requirements has no significant effect, while altering the vibrational and deformation limits gives very significant reduction, especially in height.

9.1.8 LVL Closed Rib Panel

The LVL closed rib panel consists out of two LVL panels connected to each other by LVL ribs, see Figure 98. The result is a lightweight box floor system, of which the elements are fully structurally bonded together. Like with the other LVL systems, it

requires additional gypsum boards to make the system pass the fire-safety requirements.

For the configuration of the LVL closed rib panel, the same parameters as for the semi-open rib panel are used, except the bottom flange is now a full panel. The top panel has a thickness of 37 mm, the bottom panel 43 mm and the ribs 51 mm. For the spacing, two situations are considered; a spacing of 625 mm and of 500 mm. The required rib height is calculated for the different parking deck spans.



Figure 98. LVL closed rib panel (Stora Enso, 2022f).

9.1.8.1 Base Case

The results for the rib height and total panel height for the base case for the LVL closed rib panel are shown in Table 34. Given the requirements of the base case, it is possible to span a parking deck span of 12.18 or 13.70 m with the chosen configuration of the LVL closed rib panel for both spacings. Larger spans are not possible given the maximum rib height of 600 mm. For all spans, the criteria for vibrations are governing. Furthermore, some variation can be seen between both spacings for the governing rib heights for ULS requirements.

Table 34. LVL closed rib panels for various span distances and the required rib height for each requirement.

LVL CLOSED RIB PANEL						BASE CASE							
L	Spacing	LVL _{Top}	LVL _{Bottom}	W _{Rib}	h _{Rib}	h _{Rib} h _{tot} g _{0,k} g _{0,k,tot} (2500mm)		Go	Governing rib height				
[m]	[mm]	[mm]		[mm]	[mm]	[mm]	[kN/m]	[kN/m]	NLS	SLS	Vibrations		
12.18	625	37	43	51	600	737	0.98	2.44	300	360	600		
13.70	625	37	43	51	600	737	0.98	2.44	360	400	600		
14.56	625	37	43	51	n.a.*	-	-	-	360	450	>600		
15.61	625	37	43	51	n.a.*	-	-	-	360	500	>600		
16.26	625	37	43	51	n.a.*	-	-	-	400	500	>600		
17.16	625	37	43	51	n.a.*	-	-	-	400	600	>600		
17.66	625	37	43	51	n.a.*	-	-	-	400	600	>600		
12.18	500	37	43	51	600	737	1.04	2.60	240	360	600		
13.70	500	37	43	51	600	737	1.04	2.60	240	400	600		
14.56	500	37	43	51	n.a.*	-	-	-	300	450	>600		
15.61	500	37	43	51	n.a.*	-	-	-	300	500	>600		
16.26	500	37	43	51	n.a.*	-	-	-	300	500	>600		
17.16	500	37	43	51	n.a.*	-	-	-	360	600	>600		
17.66	500	37	43	51	n.a.*	-	-	-	360	600	>600		

* none of the standard panels by Stora Enso fulfils the minimum requirements of the base case

9.1.8.2 Alternative Requirements

With alternative requirements, it is possible to span the entire range of considered parking deck spans with the LVL closed rib panel. A comparison for the height for different spans and situations is presented in Figure 99. The results that are at the basis of Figure 99 in tabular format can be found in Appendix F.

Reduced Fire-safety Requirements

Since the fire-safety requirements are met by addition of gypsum board to the panel, reducing the firesafety requirements results in a different required thickness of the gypsum board. For a fire resistance class R30, gypsum boards of 42 mm thick are required. This is a reduction of 15 mm in thickness. Logically, this does not have significant results for the total height and weight of the panel. Only reducing the fire-safety requirements therefore does not bring the parking decks into the range of possible standard configurations for the LVL semi-open rib panel.

No check on vibrations

When the check on vibrations is removed, it is possible to cover the entire considered range of parking deck spans with the LVL closed rib panels. A significant reduction in height of the panel can also be seen for the two smallest spans, which could already be covered including the check on vibrations. For

the span of 12.18 m, the height of the panel is reduced by 240 mm (33%), which lowers the weight by 0.26 kN/m (11%). No difference can be found between the spacings.

Alternative Serviceability Limit Requirements

Additionally lowering the deformation limits results in a further decrease in height and weight of the panels. For the smallest spans of 12.18 m and 13.70 m, a difference is found between both spacings. For a spacing of 500 mm, the highest reduction in height and weight is found: 160 mm (32%) and 0.21 kN/m (9%). For other spans, most reductions lie around 20% for height (100 - 150 mm) and 8% for weight (0.13-0.20 kN/m).

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

Additionally reducing the fire-safety requirements does not result in a significant decrease in height and weight of the panel as this only results in a small reduction of the height and weight of the gypsum board.



^{*} no standard panel possible for base case scenario and therefore not in graph

Figure 99. Differences in total deck height for various spans given alternative requirements for LVL closed rib panels.

Conclusion

This paragraph has shown that altering the requirements significantly impacts mostly the height of the panels. Without alteration, it is not possible to cover all parking deck spans, which is possible for a different set of requirements for the serviceability limits. Altering the fire-safety class of the structure has no significant effect on the total deck height and weight. By removal of the check on vibrations, it becomes possible to cover the various spans with the LVL closed rib panels. Allowing larger deformations results in a further decrease in total deck height and weight.

9.1.9 Comparison between Decks

A comparison is made between the seven different deck types, first for the base case and second for the alternative requirements. The decks are compared for their total deck height and weight. Following from the comparison, a choice can be made for a preferential deck system given a certain parking deck span and set of requirements.

9.1.9.1 Base Case

Given the requirements of the base case, a comparison on total deck height and weight is made for the seven deck systems. In Figure 100, the total deck heights of the different deck systems are presented next to each other. A similar graph, but for the deck weight is presented in Figure 101. Immediately it can be seen that for not a single span, a bar is found for all seven deck systems. Given the standard wood engineered products by Stora Enso, not a single one of the considered parking deck spans can be covered by a solid CLT panel, LVL open rib panel or LVL semi-open rib panel. These deck systems can therefore be ruled out as an option for the long-span deck systems considered for the structural system of the multi-storey car park. From the remaining four deck systems, the solid LVL panel and LVL closed rib panel are sufficient to span some of the considered parking deck spans. The CLT open rib panel and CLT closed rib panel can span all considered parking deck spans.

When looking at the weight of the deck systems in Figure 101, it stands out that the self-weight of the solid LVL deck for all possible spans with this system is around twice the weight of the different CLT deck systems and even more than twice the weight of the LVL closed rib panel. Despite its relatively small total deck height, the solid LVL panel is ruled out as a potential prefab long-span deck system as a result of its high self-weight.

The LVL closed rib panel, which is possible for the two shortest spans considered (12.18 and 13.70 m), has the smallest self-weight of the remaining potential deck systems. In comparison with the CLT open rib panel, its weight is 11% and 21% smaller for each span respectively. Compared to the CLT closed rib panel, its weight is 27% and 32% smaller. In terms of total deck height however, the systems has the largest total deck height of the remaining deck systems. Compared to the CLT open rib panel, it is 54% and 29% higher for each span respectively. In comparison with the CLT closed rib panel, it is percentages are 68% and 36%. Since the percentual advantage in weight compared to the percentual disadvantage in height for the LVL closed rib panel against the other two systems is in all cases is smaller (for almost all cases by large amount), the LVL closed rib panel is ruled out as a potential prefab long-span deck system.

When comparing the two remaining potential long-span deck systems (CLT open rib panel and CLT closed rib panel), it can be seen that most often, the CLT closed rib panel performs better in terms of total deck height. However, its self-weight is always larger or equal to the weight of the CLT open rib panel. The difference in height between the systems varies between 0% and 10% and the weight difference between 0% and 17%. In the comparison of the two deck systems, if the percentual advantage in terms of height for example of one system is bigger than its disadvantage in terms of weight to the other system, this deck system is given the preferred choice. This results in the deck systems presented per parking deck span in Table 35.

	BASE CASE			
Span	Deck type	\mathbf{h}_{tot}	g 0,k	g 0,k,tot(2500mm)
[m]	[-]	[mm]	[kN/m]	[kN/m]
12.18	Open CLT deck: 120mm CLT - 4 ribs 360x200mm	480	1.18	2.95
13.70	Open CLT deck: 120mm CLT - 4 ribs 450x200mm	570	1.32	3.30
14.56	Open CLT deck: 120mm CLT - 4 ribs 500x200mm	620	1.40	3.50
15.61	Open CLT deck: 120mm CLT - 4 ribs 600x200mm	720	1.56	3.90
16.26	Closed CLT deck: 120mm CLT - 4 ribs 500x140mm - 120mm CLT	740	1.76	4.40
17.16	Open CLT deck: 120mm CLT - 4 ribs 700x200mm	820	1.72	4.30
17.66	Closed CLT deck: 120mm CLT - 4 ribs 600x140mm - 120mm CLT	840	1.87	4.68

Table 35. Choice of prefab long-span deck system for various parking deck spans for base case.



* not all deck systems can span all spans for the base case scenario and therefore can be missing from the graph





* not all deck systems can span all spans for the base case scenario and therefore can be missing from the graph

Figure 101. Total deck weight comparison prefab long-span deck systems - base case.

9.1.9.2 Alternative Requirements

Altering the requirements can result in a change of preferred long-span deck system. As discussed previously in this chapter, altering the requirements can bring other deck systems in range of the considered parking deck spans, which were not possible with the standard requirements of the base case. The changes in preferred deck systems for alternative requirements are shortly discussed.

Reduced Fire-safety Requirements

As was already discussed in the various paragraphs for each deck system, altering the fire-safety requirements does not significantly influence the total deck height and weight of each long-span deck system. Therefore, this also does not result in a change of preferred long-span deck system under these altered requirements.

No check on vibrations

When vibrational behaviour of the long-span deck systems is not considered, this results in a different picture of the performance of the various long-span deck systems. As already discussed in previous paragraphs, this allows the use of different deck systems. What does not change is the much larger self-weight of the solid LVL panels in comparison with all other deck systems, as can be seen in Figure 103. Furthermore, in this graph can be seen that the other LVL deck systems outperform the CLT deck systems in terms of total deck weight. The total deck heights are presented in Figure 102 and present a slightly different picture, in which the LVL decks have a larger total deck height compared to the CLT decks.

Similar to the base case, a trade-off is made between the percentual advantage of a reduction in height and the percentual disadvantage of an increase in self-weight. Based on this trade-off, a choice is made for the preferred long-span deck system, when no check on vibrations is performed. The resulting deck systems are presented in Table 36. What stands out, is that these are all LVL systems, whereas for the base case these were CLT systems. Therefore it can be concluded, that the LVL systems are sensitive to vibrational criteria, but otherwise outperform the CLT systems.



* not all deck systems can span all spans for the scenario without check on vibrations and therefore can be missing from the graph

Figure 102. Total deck height comparison prefab long-span deck systems - no vibrations check.



* not all deck systems can span all spans for the scenario without check on vibrations and therefore can be missing from the graph

Figure 103. Total deck weight comparison prefab long-span deck systems - no vibrations check.

Table 36.	Choice	of prefab	long-span	deck	system fo	or various	parking	deck	spans for	^r alternative	requirement	ts, no
vibrations	check.											

	No vibrations check											
Span	Deck type	\mathbf{h}_{tot}	g 0,k	g 0,k,tot(2500mm)								
[m]	[-]	[mm]	[kN/m]	[kN/m]								
12.18	LVL semi-open rib panel: h _{rib} = 360mm	497	0.82	2.06								
13.70	LVL closed rib panel: h _{rib} = 400mm	537	0.93	2.34								
14.56	LVL semi-open rib panel: h _{rib} = 450mm	587	0.87	2.18								
15.61	LVL closed rib panel: h _{rib} = 500mm	637	0.99	2.47								
16.26	LVL closed rib panel: h _{rib} = 500mm	637	0.99	2.47								
17.16	LVL semi-open rib panel: h _{rib} = 600mm	737	0.95	2.38								
17.66	LVL semi-open rib panel: h _{rib} = 600mm	737	0.95	2.38								
Alternative Serviceability Limit Requirements

When on top of the removal of the check on vibrations, also the deformation limits are altered, some changes can be seen in the required deck heights and weights that follow. The results are presented in Figure 104 and Figure 105. In terms of the best performing deck system, still the LVL closed rib panel and LVL open rib panel perform best. However, as a result of the different deformation limits, depending on the parking deck span, sometimes the other deck system now performs better. An overview of the preferred deck system per parking deck span can be found in Table 37.



Figure 104. Total deck height comparison prefab long-span deck systems - SLS to 1/150 & no vibrations check.



Figure 105. Total deck weight comparison prefab long-span deck systems - SLS to 1/150 & no vibrations check.

	SLS to 1/150 & no vibrations check											
Span	Deck type	\mathbf{h}_{tot}	g 0,k	g 0,k,tot(2500mm)								
[m]	[-]	[mm]	[kN/m]	[kN/m]								
12.18	LVL closed rib panel: h _{rib} = 200mm	337	0.83	2.07								
13.70	LVL closed rib panel: h _{rib} = 300mm	437	0.88	2.20								
14.56	LVL semi-open rib panel: h _{rib} = 360mm	497	0.82	2.06								
15.61	LVL closed rib panel: h _{rib} = 360mm	497	0.91	2.28								
16.26	LVL closed rib panel: h _{rib} = 400mm	537	0.93	2.34								
17.16	LVL semi-open rib panel: h _{rib} = 450mm	587	0.87	2.18								
17.66	LVL closed rib panel: h _{rib} = 450mm	587	0.96	2.40								

Table 37. Choice of prefab long-span deck system for various parking deck spans for alternative requirements, SLS to 1/150 & no vibrations check.

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

Additionally reducing the fire-safety requirements does not result in a significant change in total deck height and weight. Therefore, the deck system types presented in Table 37 also stand for these alternative requirements.

9.2 Short-span Prefab Deck System

In this thesis, (sub)spans between 2.5 and 10 m are considered to be short-spans. Such spans are found in design variants 2 (a/b) and 3 (see paragraphs 5.3.2.2, 5.3.2.3 and 5.3.2.4), in which the deck is supported by different configurations of beam elements, spanning the parking deck. In these design variants, different spans and both single- as well as multi-span configurations can be found. For the short-span decks, two types of prefab deck systems are studied: a CLT deck and a LVL deck. Both types are discussed in the following paragraphs, in which the required height of the deck is determined for a base case, in which all standard rules and norms apply and various other cases, for which one or multiple requirements are altered. Finally, the results will provide insight into the impact of different configurations of short-span deck systems on the height and weight of the CLT or LVL floor as well as what effect changing parameters and/or requirements has on those aspects.

9.2.1 General

In Figure 106, an example of a schematisation of a 2-span short-span deck system is presented with its various loads. Similar to the long-span deck system, the top level parking deck is considered and the same load values as in Table 28 (see also starting points in chapter 6) are used. For the short-span deck system, a section width of 1 m is considered. Considering the vibration analysis, a total deck width of 12.18 m is used as a starting point.



Figure 106. Schematisation short-span deck system (2-span)

Both the CLT deck and LVL deck system are studied for their required deck thickness for a different number of spans and span lengths. From the required deck thickness also follows the weight of the deck. First, the decks are studied for the basic requirements for ULS, SLS, fire and vibrations as stated in chapter 6. Next, the requirements concerning SLS, fire and vibrations are altered in accordance with

the overview of parameters in paragraph 0.. Finally, the deck systems of CLT and LVL are compared to each other.

For the CLT and LVL decks, the standard products of manufacturer Stora Enso are considered (Stora Enso, 2022b, 2022e). The calculations are made using the Calculatis software package by Stora Enso (Stora Enso, 2022a). Examples of calculations can be found in Appendix F.2.

9.2.2 CLT Deck

9.2.2.1 Base Case

Depending on the span distance and number of spans, a CLT deck with a thickness varying between 100 mm and 300 mm is required, see Table 38. This table shows for different configurations of (sub)spans and number of spans the required minimum thickness of the CLT deck for each base requirement (ULS, ULS fire, SLS & vibrations, see Chapter 6). Based on Figure 107, it can be concluded that a 2-span system performs better compared to a single-span system for height or weight vs span ratio. A 3- or more multiple-span system however does not result in a further decrease of the height or weight of the CLT deck.



Figure 107. Comparison between single- and multi-span CLT systems for height vs span distance

Fire-safety requirements are a governing factor for almost all shorter spans up to 6.25 m. The minimum thickness of 100 mm is the result of fire-safety requirements, since a deck thinner than 100 mm will completely burn up within 90 minutes. Spans smaller than 2.5 m will therefore not result in a less thick deck. For decks thicker than 100 mm, bending under fire conditions is in most cases governing for spans up to 6.25 m.

Serviceability limits are governing in almost all configurations. Deformation limits are generally governing for spans longer than 6.25 m and in case of a single span system, also a 5 m span is already problematic. Strict requirements concerning vibrations are problematic for almost all configurations of CLT decks. For short and medium spans (up to ~6.25 m), especially the stiffness criterion is governing, while the frequency criterion is generally easily met. For longer spans (> 7.5 m), the frequency criterion becomes governing.

			BASE CASE							
Name	Nr. sub- spans	L	Name _{cLT}	CLT layers	h _{clt}	g 0,k	Miı per	Minimum he per requirer		ght ent
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	ULS	ULS Fire	SLS	Vibrations
1x2.5m span	1	2.50	CLT 100 L3s	3	100	0.5	60	100	80	90
1x5.0m span	1	5.00	CLT 140 L5s	5	140	0.7	80	140	140	140
1x7.5m span	1	7.50	CLT 220 L7s - 2	7	220	1.1	110	160	220	220
1x10.0m span	1	10.00	CLT 300 L8s	8	300	1.5	140	180	300	300
2x2.5m span	2	2.50	CLT 100 L3s	3	100	0.5	60	100	60	90
2x3.75m span	2	3.75	CLT 120 L5s	5	120	0.6	60	120	90	120
2x5.0m span	2	5.00	CLT 140 L5s	5	140	0.7	80	140	110	140
2x6.25m span	2	6.25	CLT 160 L5s	5	160	0.8	90	160	160	160
2x7.5m span	2	7.50	CLT 180 L5s	5	180	0.9	110	160	180	180
2x8.75m span	2	8.75	CLT 240 L7s - 2	7	240	1.2	140	180	220	240
3x2.5m span	3	2.50	CLT 100 L3s	3	100	0.5	60	100	60	90
3x3.75m span	3	3.75	CLT 120 L3s	3	120	0.6	60	110	90	120
3x5.0m span	3	5.00	CLT 140 L5s	5	140	0.7	80	120	120	140
3x6.25m span	3	6.25	CLT 160 L5s	5	160	0.8	90	160	160	160
4x2.5m span	4	2.50	CLT 100 L3s	3	100	0.5	60	100	60	90
4x3.75m span	4	3.75	CLT 120 L3s	3	120	0.6	60	110	90	120
4x5.0m span	4	5.00	CLT 140 L5s	5	140	0.7	80	120	120	140
5x2.5m span	5	2.50	CLT 100 L3s	3	100	0.5	60	100	60	90
5x3.75m span	5	3.75	CLT 120 L5s	5	120	0.6	60	110	90	120
6x2.5m span	6	2.50	CLT 100 L3s	3	100	0.5	60	100	60	90

Table 38. CLT decks for various span distances and number of spans and required minimum height per requirement.

Normative height

=

9.2.2.2 Alternative Requirements

From previous paragraph it has become clear, that the ULS requirements are not governing for the CLT decks, except in the case of fire (90 minutes). Table 38 already showed that large differences exist between the required minimum height of the CLT deck for the different requirements. Therefore, altering some of these requirements could result in a significant reduction of height and weight of the deck system.

Since serviceability limits are not required by the Dutch Building Degree, they can be altered by the engineer (Ministerie van Binnenlandse Zaken en Koninkrijkrelaties, 2021). The fire resistance class can be reduced to R 30 if the entire car park is considered as a single fire-compartment. This is possible based on the equivalence principle, which can be substantiated by special fire-safety calculations like zone-models as was discussed in paragraph 2.3.4. Also deformation limits can be altered, allowing bigger deformation than advised by the Eurocode. This is allowed by the Dutch Building Degree as discussed in paragraph 6.4. Last of all, also setting limitations related to vibrational behaviour of the structure of a car park is up to the designer as discussed in paragraph 2.3.3. In this paragraph, it is studied what are the effects of reduced requirements for fire-safety, SLS and vibrations on the required thickness of the CLT decks. One or more of these requirements are altered, while the other requirements remain the same (equivalent to the base case requirements).

In Figure 108, the effects of different alternative requirements on the height of CLT decks for a singlespan system can be seen for various span distances. The same, but for multi-span systems can be seen in Figure 109. The results that are at the basis of these figures in tabular format can be found in Appendix F.

Reduced Fire-safety Requirements

Reducing the fire-safety requirements from R 90 to R 30 only results in a slight reduction of the deck height for the smallest span of 2.5 m. The result is a 10 mm (10%) reduction of the deck thickness and 0.05 kN/m² (10%) reduction in self-weight of the deck.

No check on vibrations

Removing the check on vibrations of CLT decks generally speaking has no effect on reducing the height and weight of the deck, since deformation limits or fire-safety requirements are still governing and don't allow reduction of the thickness of the deck.

Alternative Serviceability Limit Requirements

Removing the check on vibrations and allowing larger deflections of the decks results in a significant reduction in height and weight of the CLT decks for larger spans (\geq 7.5 m). The larger the span, the bigger the reduction of height and weight. For a span of 7.5 m, in case of a single-span, a reduction of 40 mm (18%) in height and 0.2 kN/m² (18%) in weight is achieved. In case of a multi-span system, 20 mm (11%) reduction in height and 0.1 kN/m² (11%) in weight are achieved. For longer spans, these number increase to 60 mm (20%) reduction in height for a single 10 m span and 0.3 kN/m² (20%) in weight.

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

When reduced fire-safety requirements are used as well as alternative serviceability limit requirements, a significant reduction of deck thickness and weight can be seen for all span distances. The biggest reduction in height and weight, 40 mm and 0.2 kN/m² (40%), is for the smallest span of 2.5 m. For other spans, the reduction in height and weight is in most cases around 20%.



Figure 108. Differences in CLT deck thickness for various spans given alternative requirements (single-span)





Conclusion

From the results of this paragraph, it follows that reducing the fire-safety requirements and using alternative serviceability limit requirements results in significant height and weight reductions for all spans. For longer spans (≥7.5 m), reduction of fire-safety requirements is not required to reach the same reduction. For smaller spans, significant reductions in height and weight of the CLT deck can only be achieved when both serviceability limits are altered as well as fire safety requirements.

The reduction in height and weight of the short-span deck system can be highly significant. The exact percentage of reduction depends on the exact configuration, but for very short spans (2.5 m), 40% reduction is possible and for other spans it varies between 11 and 25 %, with most configurations around 20%. This highlights the potential of weight and volume savings for deck systems depending on what requirements are stated.

9.2.3 LVL Deck

9.2.3.1 Base Case

The LVL deck is made up of several LVL panels glued together to create a thicker plate, which can span longer distances. For spans between 2.5 and 10 m, the total thickness of the panel varies between 96 and 294 mm, see Table 39. This table shows for different configurations of (sub)spans and number of spans the required minimum thickness of the LVL deck for each base requirement (ULS, ULS fire, SLS & vibrations, see Chapter 6). Based on Figure 110, it can be concluded that a 2-span system performs better compared to a single-span system for height or weight vs span ratio. A 3-span system performs similar to a two-span system, except for a 5 m sub-span, in which case it performs slightly worse than a two-span system, but better compared to a single-span system. This small difference is due to the deflection of the deck in case two fields are loaded by an imposed load and one field is not. In general, it is assumed that a multi-span system with two-spans performs best, since in most cases more sub-spans will not result in a further decrease of height and weight of the LVL deck.

In general, it is concluded that for LVL, often a single requirement is governing. Fire-safety requirements are the governing factor for all spans shorter than 5 m. The minimum thickness of 96 mm is the result of fire-safety requirements, since a deck thinner than 96 mm will completely burn up within 90 minutes. Spans smaller than 2.5m will therefore not result in a less thick deck.

Serviceability limits are governing for all configurations with spans of 5 m or more. Deformation limits are governing for almost all spans of 5 m or longer. Only for a span of 10 m, vibrations become the governing requirement, in which case the acceleration criterion is not met.



Figure 110. Comparison between single- and multi-span LVL systems for height vs span distance

Table	39.	LVL	decks	for	various	span	distances	and	number	of	spans	and	required	minimum	height	per
require	emei	nt.														

			BASE CASE							
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	g 0,k	Mi pei	nimuı r requ	m hei iirem	ght ent
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	ULS Fire	SLS	Vibrations
1x2.5m span	1	2.5	LVL G 96 2s	2	96	0.49	72	96	72	72
1x5.0m span	1	5.0	LVL G 144 3 s	3	144	0.73	72	108	144	72
1x7.5m span	1	7.5	LVL G 210 5s	5	210	1.07	90	120	210	72
1x10.0m span	1	10.0	LVL G 294 7s	7	294	1.50	126	168	288	294
2x2.5m span	2	2.5	LVL G 96 2s	2	96	0.49	72	96	72	72
2x3.75m span	2	3.8	LVL G 108 3s	3	108	0.55	72	108	84	72
2x5.0m span	2	5.0	LVL G 108 3s	3	108	0.55	72	108	108	72
2x6.25m span	2	6.3	LVL G 144 3s	3	144	0.73	72	120	144	72
2x7.5m span	2	7.5	LVL G 168 4s	4	168	0.86	90	126	168	72
2x8.75m span	2	8.8	LVL G 210 5s	5	210	1.07	108	144	210	120
3x2.5m span	3	2.5	LVL G 96 2s	2	96	0.49	72	96	72	72
3x3.75m span	3	3.8	LVL G 108 3s	3	108	0.55	72	108	84	72
3x5.0m span	3	5.0	LVL G 120 2s	2	120	0.61	72	108	120	72
3x6.25m span	3	6.3	LVL G 144 3s	3	144	0.73	72	120	144	72
4x2.5m span	4	2.5	LVL G 96 2s	2	96	0.49	72	96	72	72
4x3.75m span	4	3.8	LVL G 108 3s	3	108	0.55	72	108	84	72
4x5.0m span	4	5.0	LVL G 120 2s	2	120	0.61	72	108	120	72
5x2.5m span	5	2.5	LVL G 96 2s	2	96	0.49	72	96	72	72
5x3.75m span	5	3.8	LVL G 108 3s	3	108	0.55	72	108	84	72
6x2.5m span	6	2.5	LVL G 96 2s	2	96	0.49	72	96	72	72

= Normative height

9.2.3.2 Alternative Requirements

In previous paragraph concerned with CLT floors, it has already been stated that altering the requirements for serviceability limits and fire-safety is allowed and can result in significant reductions of floor thickness and weight. This is also the case for LVL deck systems as one could already see in Table 39. Therefore, in this paragraph, it is studied what are the effects of reduced requirements for fire-safety, SLS and vibrations on the required thickness of the LVL decks. One or more of these requirements are altered, while the other requirements remain the same (equivalent to the base case requirements).

In Figure 111, the effects of different alternative requirements on the height of LVL decks for a singlespan system can be seen for various span distances. The same, but for multi-span systems can be seen in Figure 112. The results that are at the basis of these figures in tabular format can be found in Appendix F.

Reduced Fire-safety Requirements

Reducing the fire-safety requirements from R 90 to R 30 only results in a reduction of the deck height and weight for spans smaller than 5 m. This reduction however is significant. For a span of 2.5 m, a reduction of 24 mm (25%) and 0.12 kN/m² (25%) is achieved and for a 3.75 m span, a reduction of 24 mm (22%) and 0.12 kN/m² (22%) is achieved.

No check on vibrations

For short-span LVL decks, vibrations are much less of a problem, as can be seen in Table 39. Removing the check on vibrations of LVL decks therefore has generally speaking no effect on reducing the height and weight of the deck. Only for a 10 m single-span system, results removing the check on vibrations in a small reduction in height and weight of the deck.

Alternative Serviceability Limit Requirements

Removing the check on vibrations and allowing larger deflections of the decks results in a significant reduction in height and weight of the LVL decks for larger spans (≥ 6.25 m). For a span of 7.5 m, in case of a single-span, a reduction of 42 mm (20%) in height and 0.21 kN/m² (20%) in weight is achieved. In case of a multi-span system, 24 mm (14%) reduction in height and 0.13 kN/m² (15%) in weight are achieved. For longer spans, these numbers are similar with 54 mm (18%) reduction in height for a single 10 m span and 0.28 kN/m² (19%) in weight.

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

When reduced fire-safety requirements are used as well as alternative serviceability limit requirements, a significant reduction of deck thickness and weight can be seen for all span distances. The biggest reduction in height and weight 36 mm and 0.18 kN/m² (33%) is for a multi-span of 3.75 m. For other spans, the reduction in height and weight varies between 14 and 33%.

Conclusion

From the results of this paragraph, it follows that reducing the fire-safety requirements in combination with using alternative serviceability limit requirements results in significant height and weight reductions for all spans. For longer spans (\geq 6.25 m), reduction of fire-safety requirements is not required to reach the same reduction. For smaller spans, significant reductions in height and weight of the LVL deck can only be achieved when both serviceability limits are altered as well as fire safety requirements.

The reduction in height and weight of the short-span deck system can be highly significant. The exact percentage of reduction depends on the exact configuration, but varies between 14 and 33%. This highlights the potential of weight and volume savings for deck systems depending on what requirements are stated.



Figure 111. Differences in LVL deck thickness for various spans given alternative requirements (single-span)





9.2.4 CLT vs LVL Deck Comparison

9.2.4.1 General

The CLT and LVL deck are compared to each other to see which deck system performs best considering the deck height and weight. From the analysis in previous paragraphs, one can already draw several conclusions.

- A 2-span system performs better than a single-span system for both types of decks;
- Generally speaking is the performance of a 3- or more multi-span system no better than a 2span system for both types of decks;
- For CLT decks, often multiple requirements are governing for the minimum thickness of the deck, while for LVL decks, in most cases, a single requirement is governing;
- For both types of decks, fire-safety requirements are governing for spans shorter than ~6.25 m and the minimum thickness as a result of fire-safety requirements lies around 100 mm;
- Serviceability requirements are governing for spans longer than roughly 6.25 m for both types of decks, however in the case of CLT decks, both deformation limits and vibration criteria are governing, while for LVL decks, vibration criteria in general are not governing.

9.2.4.2 Base Case

A further comparison between the CLT and LVL deck is made for the differences in height between both deck types in Figure 113 and Figure 114 as well as for weight in Figure 115 and Figure 116. In these figures, the performance of the decks is compared for different configurations of spans as well as for different situations of alternative requirements. In Appendix F, tables can be found with the exact values and percentages in differences of performance of the CLT and LVL decks.

Focussing on the base case, Figure 113 shows that LVL performs slightly better for most single-spans. For a span of 2.5, 7.5 and 10 m, the LVL deck is between 4 and 11 mm less thick. For a span of 5 m however, the LVL deck is 4 mm thicker. These differences in height vary between -3 and 5%. For multi-span systems, the difference in height between CLT and LVL decks is more significant and always in the advantage of the LVL deck as can be seen in Figure 114. The biggest difference is found for a sub-span of 5 m, in which case the LVL deck is 32 mm thinner, which is a reduction of 23% compared to the CLT deck. For the other spans, the LVL deck is between 4 and 30 mm less thick, which is between 4 and 13% less thick than a CLT deck.

Looking at the weight of the decks, broadly the same picture emerges, see Figure 115 and Figure 116. The LVL deck performs slightly better for a single-span of 2.5 and 7.5 m and slightly worse for 5 m. For a multi-span system, the LVL deck always performs better compared to a CLT deck. In percentage terms, the weight reduction is between 1 and 3 percentage points behind the height reduction. In absolute terms, the maximum weight reduction is 0.15 kN/m^2 for a 5 m sub-span.

Given the better performance of the LVL panels for a multi-span system, it is the preferred choice for the short-span deck system under the base case scenario. Table 40 presents the choice of LVL panel for the various short-spans.

			VARIANT 2A, 2B & 3 - BASE CASE							
	Nr.		Nr.							
Name	sub-	L	Name _{LVL}	LVL	\mathbf{h}_{LVL}	g 0,k				
	spans			panels						
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]				
2x2.5m span	2	2.50	LVL G 96 2s	2	96	0.49				
2x3.75m span	2	3.75	LVL G 108 3s	3	108	0.55				
2x5.0m span	2	5.00	LVL G 108 3s	3	108	0.55				
2x6.25m span	2	6.25	LVL G 144 3s	3	144	0.73				
2x7.5m span	2	7.50	LVL G 168 4s	4	168	0.86				

Table 40. Choice of prefab short-span deck system for various parking deck spans for base case.

9.2.4.3 Alternative Requirements

Figure 113 through Figure 116 also show the difference in performance between the CLT and LVL deck for situations with alternative requirements. The instances for which a alternative set of requirements results in a significant difference in result from the base case are discussed.

Reduced Fire-safety Requirements

When the fire-safety requirements are reduced from 90 to 30 min., the performance of the LVL deck increases significantly compared to the CLT deck for shorter spans (\leq 3.75 m). For a span of 2.5 m, the height of the deck is reduced by 14 mm more compared to the base case and for a span of 3.75 m even 24 mm. In terms of weight reduction, the reduced fire-safety requirements result in a decrease in weight of 0.1 kN/m² for a span of 2.5 or 3.75 m. For other span distances, the reduced requirements do not result in changes compared to the base case as can be seen in Table 41.

			Fire to 30 min.						
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	g 0,k			
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]			
2x2.5m span	2	2.5	LVL G 72 1s	1	72	0.37			
2x3.75m span	2	3.75	LVL G 84 2s	2	84	0.43			
2x5.0m span	2	5	LVL G 108 3s	3	108	0.55			
2x6.25m span	2	6.25	LVL G 144 3s	3	144	0.73			
2x7.5m span	2	7.5	LVL G 168 4s	4	168	0.86			

Table 41. Choice of prefab short-span deck system for various parking deck spans for alternative requirements, fire to 30 min.

No check on vibrations

Removing the check on vibrations only results in changes compared to the base case for spans of 8.75 m or longer. For a multi-span system of 8.75 m, the difference in height and weight between the CLT and LVL deck significantly reduces. In terms of height, the advantage of LVL over CLT is reduced from an advantage of 30 mm to only 10 mm. For a single-span of 10 m however, the advantage of LVL increases slightly from 6 to 12 mm.

Alternative Serviceability Limit Requirements

Removing the check on vibrations and allowing larger deflections of the decks results in two interesting points. For a single-span of 5 m and a multi-span of 6.25 m, the performance of the LVL deck increases by a very large amount compared to the CLT deck. For the 5 m span, a reduction in height of 36 mm is realised compared to the base case. The cause of this large difference is that for the LVL deck, the unity checks for the base case are relatively low (~ 70%), but a smaller type LVL deck is not possible. For the situation with less strict deformation limits and no vibration check however, the unity checks are very high (> 95%). Meanwhile, for the CLT deck, no change in thickness between both situations is realised. Together this results in the very large difference in performance between the CLT and LVL deck for this situation. One other case, in which the performance significantly differs from the base case is a multi-span of 8.75 m. In this case, the difference between CLT and LVL deck is much smaller compared to the base case. Given these alternative requirements, Table 42 presents the preferred choice of deck system for various multi-span distances.

02010 1/100 0/10	o monacionio o	110010.								
			SLS to 1/150 & no vibrations check							
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	g 0,k				
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]				
2x2.5m span	2	2.5	LVL G 96 2s	2	96	0.49				
2x3.75m span	2	3.75	LVL G 108 3s	3	108	0.55				
2x5.0m span	2	5	LVL G 108 3s	3	108	0.55				
2x6.25m span	2	6.25	LVL G 120 2s	2	120	0.61				
2x7.5m span	2	7.5	LVL G 144 3s	3	144	0.73				

 Table 42. Choice of prefab short-span deck system for various parking deck spans for alternative requirements,

 SLS to 1/150 & no vibrations check.

Reduced Fire-safety Requirements & Alternative Serviceability Limit Requirements

Reducing the fire-safety requirements as well as the serviceability limit requirements results in multiple changes in the difference in performance of the CLT and LVL decks. Sometimes, the difference between CLT and LVL deck performance becomes bigger and sometimes smaller. This is highly dependent on the exact configuration of spans. In general, it can be concluded that the LVL deck performs better for all spans larger than 2.5 m, resulting in the choice of panels for the various spans as presented in Table 43.

			Fire to 30 min., SLS to 1/150 & no vibrations check							
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	g 0,k				
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]				
2x2.5m span	2	2.5	CLT 60 L3s	3	60	0.30				
2x3.75m span	2	3.75	LVL G 72 1s	1	72	0.37				
2x5.0m span	2	5	LVL G 90 3s	3	90	0.46				
2x6.25m span	2	6.25	LVL G 120 2s	2	120	0.61				
2x7.5m span	2	7.5	LVL G 144 3s	3	144	0.73				

Table 43. Choice of prefab short-span deck system for various parking deck spans for alternative requirements, fire to 30 min., SLS to 1/150 & no vibrations check.



Figure 113. Comparison in height between CLT and LVL deck for different spans and requirements (single-span).



Figure 114. Comparison in height between CLT and LVL deck for different spans and requirements (multi-span).



Figure 115. Comparison in weight between CLT and LVL deck for different spans and requirements (single-span).



Figure 116. Comparison in weight between CLT and LVL deck for different spans and requirements (multi-span).

9.2.4.4 Conclusion

In conclusion, it can be stated that the LVL deck performs better in almost all situations, both for the base case scenario as well as for different sets of requirements. Especially for multi-span systems, the LVL deck always performs better (except 2.5 m span with different set of requirements). The amount by which the LVL deck outperforms the CLT deck depends highly on the exact configuration of spans and requirements, but it can amount to a difference in height of 36 mm (30%) and weight of 0.17 kN/m² (28%). Therefore, in most cases, the better choice in performance of height and weight of the deck for short-spans is an LVL deck.

9.3 Concluding Remarks

In this chapter, various long- and short-span prefab deck systems in CLT and LVL have been considered. These have been studied for different span lengths and for the short-span systems for different numbers of spans. Furthermore, they have been studied for a base case situation, in which case the standard requirements for ULS and SLS are considered (as mentioned in chapter 6), as well as for alternative situation in which these requirements have been altered. Based on the results, a choice has been made for the different design variants, which deck system to use in which situation. In conclusion of this chapter, sub-question 3c can be partially answered: "What is the impact of altering geometric parameters and limit state requirements on the performance of the different designs for a temporary multi-storey car park using timber as primary structural material?". Furthermore, some of the hypotheses stated in paragraph 5.1.3.1 can be tested.

9.3.1 Long-span deck system

The long-span deck system is used for design variant 1. For the long-span deck systems, in the base case scenario, the CLT open rib panel and CLT closed rib panel systems outperform the other systems. The solid CLT panel and most LVL deck systems cannot even cover most spans considering their standard production dimensions for the base case scenario. Depending on the exact parking deck span, one of these systems is chosen. The chosen type, its total deck height and weight can be found in Table 44.

Considering the alternative requirements, several conclusions can be drawn. For the various decks using either CLT or LVL, vibrational criteria are always governing for long-span deck systems. If these are not considered, a significant reduction in height and weight is found for all spans considered as can be seen in Table 45. A reduction in height of up to 14% (103 mm for 17.66 m span) is possible and up to 49% (0.92 kN/m for 17.66 m span) in weight. After the vibrational criteria, the deformation limits are governing. Again allowing larger deformation results in a significant decrease of total deck height and weight for the various long-span deck systems. Relatively to the base case, a height reduction of up to 31% (253 mm for 17.66 m span) is possible and reduction in weight of up to 49% (0.91 kN/m for 17.66 m span). Reduction of the fire-safety class to R30 has no (significant) effect on the total deck height and weight of the long-span deck systems. Under alternative requirements, the various spans considered can also be covered by the solid CLT panel and various LVL deck systems, which was not possible under the base case scenario. In this case, the CLT systems are no longer the best performing ones. The LVL semi-open rib panel and LVL closed rib panel are now the better performing systems. Again, depending on the parking deck span, one of these systems is chosen, see appendix F.3.

	VARIANT 1 - BASE CASE			
Span	Deck type	\mathbf{h}_{tot}	g 0,k	g 0,k,tot(2500mm)
[m]	[-]	[mm]	[kN/m]	[kN/m]
12.18	Open CLT deck: 120mm CLT - 4 ribs 360x200mm	480	1.18	2.95
13.70	Open CLT deck: 120mm CLT - 4 ribs 450x200mm	570	1.32	3.30
14.56	Open CLT deck: 120mm CLT - 4 ribs 500x200mm	620	1.40	3.50
15.61	Open CLT deck: 120mm CLT - 4 ribs 600x200mm	720	1.56	3.90
16.26	Closed CLT deck: 120mm CLT - 4 ribs 500x140mm - 120mm CLT	740	1.76	4.40
17.16	Open CLT deck: 120mm CLT - 4 ribs 700x200mm	820	1.72	4.30
17.66	Closed CLT deck: 120mm CLT - 4 ribs 600x140mm - 120mm CLT	840	1.87	4.68

Table 44. Choice of deck system for design variant 1.

		No vibra	tions check		SLS to 1/150 & no vibrations chee				
Span	Δh _t	ot	∆g _{0,k}		Δh _{tot}		Δg _{0,k}		
[m]	[mm]	[%]	[kN/m]	[%]	[mm]	[%]	[kN/m]	[%]	
12.18	-17	-4%	0.36	31%	143	30%	0.35	30%	
13.70	33	6%	0.39	30%	133	23%	0.44	33%	
14.56	33	5%	0.53	38%	123	20%	0.58	41%	
15.61	83	12%	0.57	37%	223	31%	0.65	42%	
16.26	103	14%	0.77	44%	203	27%	0.83	47%	
17.16	83	10%	0.77	45%	233	28%	0.85	49%	
17.66	103	12%	0.92	49%	253	30%	0.91	49%	

Table 45. Performance of choice of long-span deck systems under alternative requirements relative to base case.

9.3.2 Short-span deck system

For the design variants 2a, 2b and 3, short-span deck systems are used. For the short-span deck system, only two types of decks were considered: solid CLT panel or solid LVL panel. These can either be single-span or multi-span systems. It was concluded that a multi-span system performs better and that two-spans are sufficient. Therefore, a two-span short-span deck system is chosen for these design variants.

In the base case scenario, the LVL deck outperforms the CLT deck for all span distances (short-spans). The chosen short-span deck systems for the various span configurations can be found in Table 46. Also for most other scenario's does the LVL deck outperform the CLT deck and therefore is chosen as short-span deck system.

Considering the alternative requirements, also several conclusions can be drawn. The results have shown that for the short-span deck systems, for various spans, different criteria are governing. For both CLT and LVL short-span deck systems, in most cases, all criteria have to be altered to result in a reduction of total deck height and weight. For the smallest span, the criteria for fire-safety can be governing and for the longest span often vibrational criteria are governing. In between, multiple criteria including deformations are often governing. Nevertheless, the LVL deck also outperforms the CLT deck under alternative criteria. In Table 47, the preferred choice of short-span deck system is presented for various combinations of alternative requirements. When all requirements are altered, a reduction in deck height and weight of up to 39% is possible. However, in absolute terms, the maximum height reductions for the long-span deck systems. Also in terms of reduction in weight, the possible reductions are much smaller compared to the savings for the long-span deck systems.

			VARIANT 2A, 2B & 3 - BASE CASE							
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	g 0,k				
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]				
2x2.5m span	2	2.50	LVL G 96 2s	2	96	0.49				
2x3.75m span	2	3.75	LVL G 108 3s	3	108	0.55				
2x5.0m span	2	5.00	LVL G 108 3s	3	108	0.55				
2x6.25m span	2	6.25	LVL G 144 3s	3	144	0.73				
2x7.5m span	2	7.50	LVL G 168 4s	4	168	0.86				

Table 46. Choice of deck system for variants 2a, 2b and 3.

15%

Reduced fire-safety SLS to 1/150 & no Fire to 30 min., SLS to 1/150 No vibrations check & no vibrations check requirements vibrations check Δh_{tot} $\Delta g_{0,k}$ ∆h_{tot} ∆h_{tot} $\Delta g_{0,k}$ Span Δh_{tot} ∆g_{0,k} ∆g_{0,k} [mm] [%] [kN/m] [%] [mm] [%] [kN/m] [%] [mm] [%] [kN/m] [%] [%] [kN/m] [%] [m] [mm] 2.50 24 25% 0.12 24% 0 0% 0.00 0% 0 0% 0.00 0% 36 38% 0.19 39% 3.75 24 22% 0.12 0 0% 0.00 0% 0 0% 0.00 0% 36 33% 0.18 22% 33% 5.00 0 0% 0.00 0% 0 0% 0.00 0% 0 0% 0.00 0% 18 17% 0.09 16% 6.25 0 0% 0.00 0% 0 0% 0.00 0% 24 17% 0.12 16% 24 17% 0.12 16% 0 0% 0.00 0% 0 0% 0.00 0% 24 14% 0.13 15% 24 14% 0.13

Table 47. Performance of choice of short-span deck systems under alternative requirements relative to base case.

9.3.3 Hypotheses Evaluation

The various deck systems have also been studied for what the effect of changing the requirements for ULS and SLS are on their total deck height and weight. Given the results of the long- and short-span deck systems, the hypotheses concerned with these aspects can now be tested.

- H5a: Applying a lower fire resistance class to the structural system of the multi-storey car park results in a decrease of the total deck height.
- H5b: Applying a lower fire resistance class to the structural system of the multi-storey car park results in a decrease of the total construction weight.

A reduction in fire resistance class has shown no or very little effect on the total deck height and weight of long-span deck systems. Therefore, hypotheses H5a and H5b cannot be confirmed for long-span deck systems. For short-span systems however, fire-safety requirements are often a governing requirement, although often in combination with other criteria. Especially for the smallest spans, can a reduction in required fire resistance class result in a decrease of total deck height and weight, especially in combination with reduction of SLS criteria. Therefore, for short-span deck systems, hypotheses H5a and H5b can be confirmed, but only for certain spans and combinations of requirements.

- H6a: Setting lower or no criteria for the vibrational behaviour of the structural system of the multistorey car park will result in a decrease of the total deck height.
- H6b: Setting lower or no criteria for the vibrational behaviour of the structural system of the multistorey car park will result in a decrease of the total construction weight.

For long-span deck systems, the criteria for vibrations are almost always governing. Reducing or removal of these requirements therefore can result in a significant reduction of total deck height and weight. This confirms hypotheses H6a and H6b for long-span deck systems. For short-span deck systems however, the criteria for vibrations are not always governing. For CLT decks, multiple criteria are governing and therefore, to reduce total deck height and weight, multiple criteria should be altered. For LVL decks, vibrational criteria are never governing for short-span decks. Therefore, hypotheses H6a and H6b cannot be confirmed for short-span decks.

- Allowing larger deflections for the structural elements of the car park will result in a decrease of H7a: the total deck height.
- Allowing larger deflections for the structural elements of the car park will result in a decrease of H7b: the total construction weight.

After the vibrational criteria, the requirements for deflections are governing for all long-span deck systems. Reducing them results in a significant reduction of total deck height and weight, confirming hypotheses H7a and H7b. Also for short-span deck systems, can reduction of the deflection limits result in a significant decrease in total deck height and weight, both when vibrational criteria are governing and when not. This confirms hypotheses H7a and H7b for short-span deck systems.

7.50

10 Framing System

The framing system is one of the main parts of the structure of the multi-storey car park. Together with the decks, it makes up the structural system of the car park. In paragraph 5.3.2, four different framing systems (design variants) have been presented for the development of the proof of concept for the modular, temporary multi-storey car park in timber. For each of these variants, also several parameters were stated, which could be altered, resulting in multiple subvariants. Examples of such parameters are the parking deck span and column distance. In this chapter, these (sub)variants are first structurally analysed and afterwards compared to each other for their performance on the total deck height, total deck weight and column load on the foundation, as discussed in paragraph 5.1.3.

10.1 Design subvariants

The four different main variants for the framing system were already presented in paragraph 5.3.2. Furthermore, in paragraph 5.1.3, different parameters were presented, which could be altered, resulting in a different structural system that might perform differently. Four of these parameters are used to alter the four main variants for the framing system (if applicable to the considered design variant), creating various subvariants. These parameters are: parking deck span, column distance (perpendicular to parking deck span), use of struts and distance between joists. In combination with the results from the deck systems in chapter 9, it is chosen which parameters are used for each (sub)variant. In total, 77 subvariants are analysed (detailed results in appendix G). The different subvariants are explained below for each main variant.

10.1.1 Variant 1

The first design variant, variant 1, is studied for seven different parking deck spans (12.18, 13.70, 14.56, 15.61, 17.16 and 17.66 m as discussed in paragraph 0). These include various long-span prefab deck systems. Furthermore, different column distances are considered. The columns are either placed at a width of one parking bay (2.5 m) or two parking bays (5.0 m). Larger distances are possible, but will probably result in a significant increase of the cross-section of the transverse main girder, which is not required. Therefore, only these two distances are considered. Using this set of parameters, a total of 14 subvariants are created for design variant 1, see Figure 117.





10.1.2 Variant 2a

For design variant 2a, again seven different parking deck spans are studied. Furthermore, different column distances are considered. The columns are either placed at a width of one parking bay (2.5 m) or two parking bays (5.0 m). For larger distances, the height of the short-span parking deck will significantly increase as was found in paragraph 9.2 and also the height of the main girder is expected to increase. Therefore, only these two distances are considered. Finally, it is also studied what the effect is of the use of struts. Using this set of parameters, a total of 28 subvariants are created for design variant 2a, see Figure 118.



Figure 118. Overview of subvariants design variant 2a.

10.1.3 Variant 2b

In the case of design variant 2b, the only parameter that is varied is the parking deck span. The same seven distances, used for the other variants are also studied for this variant. The columns are placed at a distance of 5 m. A smaller distance would not result in a thinner parking deck as a result of the minimum thickness for fire-safety (see paragraph 9.2) and therefore is not beneficial. For larger distances, the transverse main girders would significantly increase in cross-section. The variation of the parking deck span results in a total of 7 different subvariants, see Figure 119.

10.1.4 Variant 3

For the final main variant, design variant 3, three parameters are varied. First of all, again seven different parking deck spans are considered. Second, the effect of using struts is studied. Third, the number of joists is altered. This number depends on the parking deck span, but the distance between them is either 2.50 m at minimum or 3.75 m at minimum. A shorter span would not result in a decrease of deck thickness and weight and therefore is not beneficial. For a longer span, the deck could also directly span the distance between the main girders, since they are located at a distance of 5.0 m from each other. Using this set of parameters, a total of 28 subvariants are created for design variant 3, see Figure 120.



Figure 119. Overview of subvariants design variant 2b.



Figure 120. Overview of subvariants design variant 3.

10.2 Schematisation Structural System

To be able to structurally analyse the different subvariants, their structural systems have to be schematised. It is chosen to schematise the structural systems in (if required, multiple) 2D frames. The choice has been made to create the model in 2D, since a 3D-model would add a lot of complexity to both the creation of the parametric model as well as during the structural analysis, which is now avoided. The various 2D schematizations of the different main design variants are presented below, each containing all different structural elements present in the variant considered.

10.2.1 Variant 1

10.2.1.1 Structural elements

In Figure 121, at the bottom, the 2D schematisation with the different structural elements of design variant 1 is presented. At the top of the figure, a 3D drawing of the variant has been presented as well. View A contains the most important primary structural elements, used to transfer the loads from the floors to the foundation. In variant 1, the long-span prefab deck elements are used as concluded in paragraph 9.3. The loads coming from the deck elements are transferred to the transverse main girders, which again transfer their loads to the columns. The 2D-model of view A contains an additional girder as the top level railing, which is used as an extra element, to which the façade can be connected. Furthermore, a wind bracing in a single direction is modelled to act as a stability element against wind loading perpendicular to the parking deck span. It is modelled as a tension only element and in reality another set of wind bracings are mounted in the other direction as well.

Most structural elements of view B cannot be seen in the 3D overview. These elements act as the stabilizing structure in the other direction and are located in the façades at every 40 m width of the car park. They contain a set of columns, side girders, top level railings and wind bracings.

All columns are modelled as continuous elements. The transverse main girders and railing top level – x are modelled as simply supported beams in case of a span of 2.5 m and as continuous in case of a span of 5.0 m, to limit the deformations. The side girders and railing top level – y are all modelled as simply supported beams. All these elements are made of rectangular glulam timber sections. The wind bracings are made of steel L-sections, which are connected by hinges to the nodes of the structure. Finally, all columns are pinned supported to the foundations.

10.2.1.2 Loading

In terms of loading, the permanent loads (self-weight and finishing layer) coming from the floors are modelled as line loads on the transverse main girders. On the top transverse main girder, an additional line load for snow is modelled as well as a line load for wind pressure on the top parking deck. The self-weight of the other structural elements is later modelled by SCIA. Wind loading from both directions on the structure is modelled as point loads on the nodes between columns and transverse main girders/top level railings/side girders. These point loads include wind forces coming from pressure and suction on the façades as well as friction on the roof and façades. Finally the self-weight of the façade is modelled as a line load on the transverse main girders, side girders and top level railings, but in this case, these elements carry the loads of only halve a parking deck.



Figure 121. 2D schematisation variant 1.

10.2.2 Variant 2a

10.2.2.1 Structural elements

The schematisation of the structural system of design variant 2a is presented in

Figure 122. For this variant, the schematisation at the bottom of the figure presents the structural elements of the main load bearing structure, transferring the loads from the parking decks to the foundation. The parking decks are made of the short-span prefab deck panels as presented in paragraph 9.3. They transfer their loads to the main girders, spanning the parking deck. They then transfer the loads further to the columns and optionally (dependent on the subvariant) also to the struts, which shorten the span of the main girders.

The two top 2D schematisations are modelled to analyse the stabilizing elements of the structure. At the left, a wind bracing cross at every 40 m stabilizes the structure against wind loading, which acts perpendicular to the parking deck span. Only one wind bracing is modelled, which is a tension only member. Transverse main girders are modelled to support the façade elements, but depending on its weight, it can also be possible to connect the façade elements to the deck system. To stabilize the structure in the direction parallel to the parking deck span, another set of wind bracings are modelled, which are required for every three parking deck spans. The side girders than replace the main girder, supporting the parking deck.

All columns are modelled as continuous elements. The main girders are simply supported in the connection to the columns and are continuous over the struts. The transverse main girders and railing

top level – x are modelled as simply supported beams in case of a span of 2.5 m and as continuous in case of a span of 5.0 m, to limit the deformations. The side girders and railing top level – y are all modelled as simply supported beams. The struts are connected by hinges to the main girder and column. All these elements are made of rectangular glulam timber sections. The wind bracings are made of steel L-sections, which are connected by hinges to the nodes of the structure. Finally, all columns are pinned supported to the foundations.



Figure 122. 2D schematisation variant 2a.

10.2.2.2 Loading

Considering the loads, the main girders are loaded by line loads for the self-weight and permanent loads of the parking deck as well as for the imposed loads. The top level main girder is also loaded by a line load for the snow loading and wind loading (pressure on the top level deck). All other self-weights of the structural elements are created by SCIA.

On the stabilizing structures, wind loading from both directions on the structure is modelled as point loads on the nodes between columns and transverse main girders/top level railings/side girders. These point loads include wind forces coming from pressure and suction on the façades as well as friction on the roof and façades. The stabilizing structures may also support the façade elements, which creates line loads on the transverse main girders, side girders and top level railings. In this case, only halve the loads coming from the parking deck are supported by these elements. Since the side girders also support the deck, here also line loads apply for self-weight, permanent loading and imposed loading of the parking deck. For the top level, also wind and snow loading applies.

10.2.3 Variant 2b

10.2.3.1 Structural elements

The schematisation for the structural system of design variant 2b is presented in Figure 123. In this variant, the short-span decks are supported by main girders, just like in variant 2a. However, not all main girders transfer their loads directly to the columns. Halve the number of girders transfer their loads to a transverse main girder, which then transfers the loads to the columns, trough which they are transferred to the foundation. To stabilize the structure, wind bracings are placed in the façade in a similar way to variant 2a, which are modelled as tension only members.

All columns are modelled as continuous elements. The main girders are simply supported in the connection to the columns. The transverse main girders and railing top level -x are modelled as continuous beams, to limit the deformations. The side girders and railing top level -y are all modelled as simply supported beams. All these elements are made of rectangular glulam timber sections. The wind bracings are made of steel L-sections, which are connected by hinges to the nodes of the structure. Finally, all columns are pinned supported to the foundations.

10.2.3.2 Loading

The loading scheme of variant 2b is somewhat different to that of variant 2a. The parking deck is supported by the main girders, which therefore are subjected to line loads for the self-weight, permanent loads, imposed loads, wind loads and snow loads. The reaction forces of the supports of the main girders are than placed as point loads at midspan of the transverse main girders and the nodes of the main girders with the columns. The point loads for the top level include wind and snow loading, while the others do not. Furthermore, the 2D frame in direction perpendicular to the parking deck span also acts as a stabilizing element with its wind bracings. Point loads on the nodes connecting the transverse main girders and top level railing to the columns include wind forces coming from pressure and suction on the façades as well as friction on the roof and façades. Finally, the transverse main girders (and top level railing) can also support the façade elements, which results in an extra line load, but in which case the girders support only halve a parking deck. For the stabilizing structure in the other direction, also point loads for the wind loads act on the nodes between side girders/top level railing and columns. The side girders support either the full load from a parking deck or halve its load and additionally the self-weight of the façade elements.



Figure 123. 2D schematisation variant 2b.

10.2.4 Variant 3

10.2.4.1 Structural elements

The schematisation for the structural system of design variant 3 looks similar to that of variant 1, but with one large difference. Variant 3 uses short-span deck panels, which are supported by joists. These joists then transfer the loads to the main girders, which are supported by the columns and optionally by struts, see Figure 124. The columns finally transfer the loads to the foundation.

To stabilize the structure, wind bracings are placed in the frame perpendicular to the parking deck span at every 40 m. Also in the other direction, wind bracings are placed in the façades and at every 40 m width of the car park for every three parking deck spans. For the wind bracings, only one element is modelled for simplifying the model, but in reality two bracings are placed in a cross. They are tension only elements.

All columns are modelled as continuous elements. The joists are simply supported beams. The main girders are simply supported in the connection to the columns and are continuous over the struts. The transverse main girders and railing top level -x are modelled as continuous elements, to limit the deformations. The side girders and railing top level -y are all modelled as simply supported beams. The struts are connected by hinges to the main girder and column. All these elements are made of

rectangular glulam timber sections. The wind bracings are made of steel L-sections, which are connected by hinges to the nodes of the structure. Finally, all columns are pinned supported to the foundations.



Figure 124. 2D schematisation variant 3.

10.2.4.2 Loading

The first elements of the loading scheme are the joists, which are loaded by the self-weight, permanent load, imposed load and for the top deck also wind and snow load. These are all placed on the joist as a line load. The reaction forces coming from the joists are than placed at the interval of the joists on the main girders and so include the loads coming from the parking decks. On the stabilizing frames, point loads are placed on the nodes connecting the transverse main girders, top level railings and side girders to the columns and include wind forces coming from pressure and suction on the façades as well as friction on the roof and façades. Furthermore, line loads for the self-weight of the façade are placed on the transverse main girders, side girders and top level railings. The side girders can also support the point loads of the joists, but are halve in case a façade load is applied.

10.3 Workflow

The workflow to analyse and compare the various subvariants consists out of several steps. The first step is the setup of a parametric model, based on the schematisation of the subvariants. This model has been created using Rhinoceros with Grasshopper. The use of the parametric model is twofold: it is first used to create the geometry for the subvariants and second it is used to prepare a structural model for structural analysis. The use of this parametric model speeds up the generation and analysis of the various subvariants. In the next step, a structural analysis is performed on each subvariant using SCIA. The input for this analysis is generated by the parametric model using the plugin Koala, allowing automatic generation of a structural analysis model for SCIA for each subvariant. In SCIA, the models of the subvariants are analysed and all its structural elements are optimized to fulfil the requirements for ULS and SLS as defined in chapter 6. After the structural analysis and optimization is completed, the resulting data considering the required cross-sections for all structural elements of the various subvariants is further processed using Excel. This finally results in an overview of data, concerned with total deck heights, average weight of the structure and loads on the foundation, which can be used to compare the various (sub)variant with each other and later on with alternative existing car park concepts.

10.3.1 Parametric Model

As stated before, a parametric model is created in Grasshopper. In this paragraph, the structure and general workflow of the parametric model is described. A more detailed description of the entire script can be found in appendix H.

An overview of the workflow within the script can be found in Figure 125. The script can be separated into two parts. The first part is concerned with creating the geometry of the structural elements to later be used for the second part, preparing a structural analysis model of the subvariant for SCIA, using the plugin Koala.

The script starts with the inputs. These are either parameterised inputs, like the parking deck span and column distance, but also include the specific parking deck system chosen in chapter 9. A second set of inputs are the so-called standardised inputs, which include car park characteristics like the parking bay width and clearance height, inputs for the wind load calculation and general load input values. These input values are further processed to determine often used input settings for the modelling of the geometry like the total story height.

Next, the geometry for the 2D structural model is created. These lines and nodes are later used to create the model for the structural analysis. For each main design variant, the geometry is created by a separate set of objects.

The preparation for the structural analysis model for SCIA starts with setting up the general settings of the analysis. Next, the different items required for the analysis model are created like the layers for the structural elements, load cases & groups and load combinations. For the cross-sections, self-made libraries for the standard available cross-section dimensions are inserted as well as material types and classes for later selection. The created lines and nodes from the geometry section are then used to create beam elements and support points. Cross-sections are linked to the elements and loads are applied. Finally, hinges are added where needed.

With all the inputs for the generation of the xml file by Koala ready, they are inserted into the xml generation component, which combines all elements into an xml file, which can be opened and analysed in SCIA. However, some settings required for the structural analysis of especially timber structures are not yet available in Koala components. Therefore, the created xml file is further processed by text editing in the Grasshopper model, to add certain settings to the file. This finally results in the xml file, which is used for structural analysis in SCIA.



Figure 125. Workflow parametric model.

10.3.2 Structural Analysis

As stated before, the structural analysis of the structural systems of the various subvariants is conducted in SCIA Engineer. The workflow for the structural analysis can be found in Figure 126.

Before the analysis can be conducted, a template is created in SCIA. This template includes certain settings for the calculation, which could not be added to the xml file. Calculation settings for timber like the service class used and the national annex applied are included in this template.

In the next step, the xml file created by the parametric model can be loaded into the template file. SCIA analyses the structure after which its cross-sections can be optimized to fulfil the unity checks for ultimate and serviceability limit states. This optimization process can only be automized partially as these results do not fulfil all requirements. The final optimization of the cross-sections is done manually by checking the various unity checks for all structural elements.

After the structural system is analysed and checked, it results in a list of all cross-sections applied to the different elements. A list of these cross-sections is exported for further processing.





10.3.3 Data Processing

After finishing the structural analysis, it is known what cross-sections are to be used for the different structural elements of the subvariants. However, the goal is to compare the different (sub)variants on their total deck height, weight of the structure and load on the foundation. To come up with these numbers, the characteristics of the various subvariants and their accompanying list of cross-sections is further processed using Python and Excel. Finally Excel is also used to create graphs for the comparison of the various (sub)variants.

First of all, for each subvariant, the total deck height is calculated. This can be either only the thickness of the deck element or the sum of this thickness and its supporting girders.

Second, for each subvariant, for comparison, the self-weight of the structure is calculated for a single level for a length of 4 parking bays (10 m) and a width of the parking deck span. To come up with this weight, the dimensions and weight of every structural element is determined. Next, it is determined how many items of every structural element are present in the standard length and width for the comparison. The number of elements are multiplied by their weight and summed up, resulting in the total weight of the structure.

Third, the load coming from a "standard" column onto the foundation for 1 to 5 levels is calculated. The description of a "standard" column was given in paragraph 7.2.3. This load includes the self-weight of the structural elements, transferring their loads to the column and all loads acting on the parking deck (self-weight, permanent load & imposed load). The permanent loads are multiplied by the factor 1.20 and the variable loads by 1.50.

Finally, the result is an overview of all relevant numbers required for comparison of the various (sub)variants, which is also presented in appendix G.

10.4 Results Analysis

As stated in paragraph 5.1.2, the various (sub)variants for the structural system of the modular and temporary multi-storey car park will be compared to each other considering their total deck height and weight. Therefore, in this paragraph, the three aspects of total deck height, weight per level and load on foundation are discussed for all subvariants. Apart from the results presented in this paragraph, additional tables which support these findings can be found in appendix G.

10.4.1 Total Deck Height

The total deck height for all subvariants for a span of 16.26 m varies from a minimum of 666 mm to a maximum of 1128 mm as can be seen in Figure 127. For design variant 1, both subvariants have the same deck height, since the same long-span deck panels are used with a thickness of 740 mm. This is the second smallest total deck height of all subvariants analysed.

Within design variant 2a, quite a significant difference in total deck height can be seen between the subvariants. The first conclusions that can be drawn is that the use of struts significantly reduces the total deck height as it reduces the height of the main girder, supporting the deck panels. With a column distance of 1 parking bay, a reduction of 280 mm (30%) is realised and with a column distance of 2 parking bays, it is a reduction of 260 mm (23%). The second conclusion is that also reducing the column distance in the transverse direction to the main parking deck span significantly reduces the total deck height, as the height of the main girder is reduced when it has to carry a smaller area of the deck. Without the use of struts, using a column distance of only one parking bay reduces the total deck height by 182 mm (16%) and with the use of struts, the reduction is 202 mm (23%). In total, the subvariant with a column width of a single parking bay and with the use of struts results in the smallest total deck height amongst all subvariants for a span of 16.26 m.

Design variant 2b results in a total deck height of 986 mm. It uses a column distance of two parking bay widths and has no struts. When compared to design variant 2a, it has a significant smaller height than subvariant 2a with a column distance of two parking bay widths and no struts; a reduction of 142 mm (13%). Compared to the subvariant 2a with a column distance of only a single parking bay width and no struts, its total deck height is only 40 mm (4%) larger.

For design variant 3 it can be seen that the distance between the joists does not have a very big impact on the total deck height. However, the use of struts again significantly reduces the total deck height. With a joist spacing of max 2.5 m, a reduction in total deck height of 190 mm (17%) is achieved. By using struts, the total deck height is slightly smaller than design variant 2b, which uses the same column distance. Without the use of struts, the total deck height is the same as the subvariant of variant 2a without struts and using the same column distance.



Figure 127. Total deck height for various (sub)variants for 16.26 m span.

The final parameter that is studied is the parking deck span. In Figure 128, an overview is presented of all subvariants and their total deck heights for the different parking deck spans. What can be seen is that for all subvariants, the total deck height significantly increases with the parking deck span, although it differs per variant by how much. The total deck height of design variant 1 increases the most: 75% between the smallest an largest span considered. It performs better for spans smaller than 15 m and afterwards increases even faster. The subvariants of design variant 2a perform the best around a span of 16 m. Between the smallest and biggest span, a total deck height difference between 40% and 45% can be seen. Design variant 2b displays a very linear line of the total deck height with a height difference between smallest and biggest span of 41%. The subvariants of design variant 3 also show rather straight lines with height differences around 40% with the use of struts and around 55% without the use of struts.

The different subvariants can also be compared to each other considering their performance relative to each other for different spans. An important finding is that variant 1 is the best performing (sub)variant in terms of total deck height for spans up to 15 m. From 15 m onwards, subvariant 2a with struts and a column distance of a single parking bay is the best performing. Another finding is that subvariant 2a without struts and with a columns distance of two parking bay widths starts to perform better than some subvariants of design variant 3 for longer spans. Otherwise, no real changes in the ranking of the various subvariants can be observed.



Figure 128. Total deck height for various (sub)variants and spans.

10.4.2 Weight per Level

The weight per level for a section of the structural system of the subvariants for a span of 16.26 m is presented in Figure 129. What immediately stands out is the much bigger weight of design variant 1, which is around twice as large compared to the other variants. It can be concluded that using prefab long-span deck panels is not an efficient way to span the parking decks in terms of weight compared to short-span panels in combination with girders supporting the panels. The other (sub)variants are closer to each other in terms of weight. They all weigh around 15,000 kg per level for the section of the car park considered. The difference between the lightest and heaviest of these subvariants is 20%.

Focussing on design variant 2, in terms of weight, a difference can be seen for the use of struts. The biggest reduction in weight by the use of struts is for a column distance of one parking bay: 11% (1970 kg). For a column distance of two parking bays, the reduction is smaller: 5% (824 kg). Furthermore, a smaller column distance results in a higher weight of the structure. Without the use of struts, a column distance of a single parking bay width results in an increase of the weight of 13% (1932 kg) compared to using a column distance of two parking bays. With the use of struts this is an increase of 5% (786 kg).

Design variant 2b is the worst performing variant amongst the subvariants with a column distance of two parking bays in terms of weight (without taking variant 1 into consideration). Compared to subvariant 2a without struts and a two parking bay column distance, it is 12% (1948 kg) heavier.

Amongst the subvariants of design variant 3, it can be seen that increasing the maximum distance between the joists (and therefore reducing the number of joists) results in a reduction of the weight of the structure. If struts are used, increasing the maximum distance between joists from 2.5 m to 3.75 m results in a reduction in weight of 6% (822 kg) and without the use of struts of 5% (796 kg). Furthermore, again the use of struts reduces the total weight of the structure. It reduces the weight for both joist configurations by 8% (1214 & 1188 kg).



Figure 129. Weight per level for various (sub)variants for 16.26 m span.

Last of all, it is also studied what the effects are of different parking deck spans of the weight of the structure. An overview of all subvariants is presented in Figure 130. From this graph it can again be seen that variant 1 is much heavier compared to the other variants and that for longer spans it becomes even relatively more heavier in comparison to the others. The other subvariants have a more linear behaviour and are presented separately in Figure 131 to give a better overview. The exact number differs per subvariant, but all become between 1.6 and 1.8 times heavier between the smallest and largest span considered.

When comparing the different variants relatively to each other for different spans, almost no changes in the ranking of the performance on weight can be seen. The only change that stands out is the performance of subvariant 2a. For the smallest span considered, it is the second best performing subvariant, while its performance drops towards a span of 16.26 m. From this point onwards it is amongst the middle group in terms of performance.



Figure 130. Weight per level for various (sub)variants and spans.



Figure 131. Weight per level for selection of various (sub)variants (not variant 1) and spans.

10.4.3 Load on Foundation

The loads on the foundation of a "standard" column for the different subvariants are compared to each other in Figure 132. The main factor influencing this load is of course the parameter of the column distance in transverse direction of the main parking deck span. The different (sub)variants differ from each other on this parameter which is either the width of a single or two parking bays. Doubling this distance results in doubling of the area that has to be supported by the column. As a result, also the total loads supported by the column for the permanent loads, imposed loads, snow loads and wind loads acting on the parking deck doubles. However, the loads as a result of the self-weight of the structural elements does not simply double, but is dependent on the configuration of the considered (sub)variant. Nevertheless, it can clearly be seen in the graph that doubling the column distance results in (roughly) a doubling of the loads on the foundation.

For variant 1, with its long-span decks, the load on the foundation simply doubles with the doubling of the column distance. For a distance of one parking bay, it is 868 kN for 4 parking levels and for a distance of two parking bays, it is 1735 kN.

Doubling the distance of the columns for variant 2a results in an increase of the load on the foundation by 98% with the use of struts and 95% without the use of struts. For 4 parking levels, the load increases from 678 kN (the smallest load of all subvariants) to 1344 kN with the use of struts and from 702 kN to 1366 kN without the use of struts. The use of struts has only very little influence on the load on the foundations when all other loads acting on the structure are also taken into account. With a column distance of a single parking bay, the use of struts lowers the load on the foundation by just 3% (24 kN for 4 parking levels) and in case of double the distance by just 2% (22 kN for 4 parking levels).

The comparison of design variant 2b for the load on the foundation to the others is similar to the weight comparison. It performs slightly worse than subvariant 2a with also two parking bays column distance and no struts, by 1%.

Similar to the weight comparison of design variant 3, also for the loads on the foundation, a slight increase is found with increasing the distance of the joists and removal of struts. An increase between each subvariant of 1% is found.



Figure 132. Load from standard column on foundation for 4 levels of various (sub)variants for 16.26 m span.

When studying the effect of the parking deck span on the load on the foundation, for each span, the same ranking of the subvariants is found, see Figure 133. Subvariants 2a with a single parking bay column distance perform best, followed by subvariant 1 with the same column distance. However, the gap between these variants increases as the span becomes bigger. At roughly double the load, the subvariants of design variant 2b and 3 are found, followed by subvariant 1 with its double parking bay width column distance. Again, its relative performance to the subvariants of design variant 2b and 3 decreases with the span.



Figure 133. Load from standard column on foundation for 4 levels for various (sub)variants and spans.

10.4.4 Effect of Alternative Assessment Criteria

In the paragraphs above, the effect of changing various geometric parameters of the design variants on its performance has been studied for the specific subgoals of paragraph 5.1.2. Apart from these parameters, for the deck systems, also three parameters related to the assessment criteria for the structural systems were studied. These were the allowable deflections, applied fire-safety class and check on vibrational behaviour. Due to limitations of the software packages used (which make use of automatization), these assessment criteria parameters are not studied in detail for the various design variants including both deck and framing system. However, this does not mean one cannot say anything about what effect is expected in general on the total structural systems when these criteria are altered. Therefore, these expectations are discussed in this paragraph for each design variant.

10.4.4.1 Variant 1

Design variant 1 makes use of the long-span prefab deck system. For these panels, which span the entire parking deck, the effect of alternative assessment criteria on the total deck height as well as total deck weight has already been discussed in paragraph 9.1. Alternative assessment criteria can result in a total reduction in deck height of up to 31% (253 mm for 17.66 m span) as presented in paragraph 9.3.1. In terms of weight, a reduction of up to 49% is possible for the decks under alternative requirements. For this design variant, around 90% of the weight comes from the deck system and only 10% of the rest of the structural system. Therefore, potential reductions in weight of the supporting framing system as a result of alternative assessment criteria is not expected to result in further significant weight reduction of the total structural system of design variant 1. In conclusion, alternative assessment criteria could significantly improve the performance of design variant 1. In terms of total deck height, it would become the best performing variant (other variants under base case scenario) and in terms of weight, it would come into the same range as the other design variants.

10.4.4.2 Variant 2a

Design variant 2a makes use of a short-span prefab deck system, which is supported by a glulam main girder. Since the total deck height for this variant is the summation of the height of the main girder and the thickness of the deck, a different required height of the main girder as a result of alternative assessment criteria can result in a different total deck height. Furthermore, the total weight of the structure can change as a result of different required cross-sections for the elements of the framing system.

Considering the main girders, the ULS strength criterium is governing. Therefore, alternative requirements for deflections, fire-safety and vibrations will not result in a change in cross-section. Also for the other elements of the framing system, the ULS strength criterium is governing. In conclusion, alternative assessment criteria will not result in changing dimensions of the cross-sections of the framing system.

The effect of alternative assessment criteria on the deck system has already been studied in paragraph 9.2. It was concluded in paragraph 9.3.2, that in terms of height and weight, a reduction of up to 39% is possible for a column distance of one parking bay and 16% for a column distance of two parking bays. However, since the total deck is build-up of the short-span deck system in combination with a main girder, in absolute terms, these reductions are less significant. For the height, a maximum reduction of only 36 and 18 mm is achieved for each column distance respectively under alternative assessment criteria. Since no reduction in height of the main girder is possible under alternative requirements, the total height reduction of the deck is rather limited compared to design variant 1 under alternative requirements. Considering the total weight of the structural system, roughly halve of the weight comes from the deck system, the main girders are around 30% and the columns and struts 20% (depending on the specific subvariant). The reduction in weight for the deck under alternative requirements is more significant. In the total structural system, it contributes to a maximum reduction of around 20% in self-weight.

10.4.4.3 Variant 2b

Similar to design variant 2a, its conclusions also stand for design variant 2b. It uses the same shortspan deck system, for which a reduction in total deck height under alternative requirements is rather limited. Also for this variant, the ULS strength criteria are governing for the other structural elements and therefore reductions of cross-sections dimensions under alternative assessment criteria are not possible. Bigger savings can be realised in terms of weight. Again, the deck system is around halve the weight of the total structural system, the main girders are around 40% and the transverse girders and columns the remaining 10%. This again results in a potential reduction of the self-weight of the structural system of 20% under alternative assessment criteria.

10.4.4.4 Variant 3

Design variant 3 again makes use of the short-span deck system, but only has a span of 5 m. Again, the ULS strength criteria are governing for the elements of the framing system and therefore, reduction in total deck height and weight under alternative requirements has to come from the deck system. For a span of 5 m, the possible reduction in height and weight are only 16% for the decks. This is only 18 mm reduction in height, which is not significant. In terms of weight, the deck makes up roughly 60% of the weight of the total structural system, the main girders and joists are around 30% and the columns and struts 10%. As a result, a total reduction in self-weight of the structural system under alternative assessment criteria of up to 10% is possible.

10.4.4.5 Conclusion

In conclusion, it can be stated that alternative assessment criteria have a different effect on the various design variants. For variant 1, it can result in a significant reduction in both height and weight, making it able the best performing variant in terms of height and closer to the other variants in terms of weight compared to the base case scenario. For the other design variants, the reduction in total deck height is not significant, however in terms of weight, a reduction in self-weight of around 20% can be achieved for variants 2a and 2b and 10% for variant 3.

10.5 Concluding Remarks

Based on the results of the structural analysis of the various designs for the frame system and its comparison, several conclusions can be drawn. Hypotheses H1 till H4 a & b can be tested and the best performing design subvariant can be chosen. In conclusion of this chapter, sub-question 3c can further be answered: "What is the impact of altering geometric parameters and limit state requirements on the performance of the different designs for a temporary multi-storey car park using timber as primary structural material?".

10.5.1 Total Deck Height

Considering the total deck height of the different design variants, hypotheses H1a, H2a, H3a and H4a can be tested.

H1a: Increasing the parking deck span will result in an increase of the total deck height.

Focussing on the parking deck span versus the total deck height, as expected a bigger span results in an increase in total deck height, confirming hypothesis H1a.

H2a: The use of struts in the main span of the structural system of the car park will result in a decrease of the total deck height.

The comparison of the deck heights of the various subvariants has clearly shown that the use of struts significantly reduces the total deck height (between 17 and 30% depending on the variant). Already with a small inward distance of the struts of 0.67 m, this reduction is achieved. Using a subvariant with stuts if possible is therefore preferred. This confirms hypothesis H2a.

H3a: Increasing the distance between columns in transverse direction will result in an increase of the total deck height.

The comparison has also proved that using a column distance in the transverse direction of the main parking deck span of one parking bay width results in a lower total deck height compared to a distance of two parking bay widths (reduction of 16 to 23% in total deck height). To minimize the total deck height, a smaller column distance is therefore preferred. This confirms hypothesis H3a.

H4a: Increasing the distance between joists will result in an increase of the total deck height.

For design variant 3, the results have shown that for a larger distance between the joists, the total deck height increases. However, this increase in height is very small. Nevertheless, this confirms hypothesis H4a.

In terms of performance, design variant 1 is the best performing variant for spans up to 15 m, closely followed by subvariant 2a – column every 1 parking bay, struts on. This subvariant performs best for spans larger than 15 m. The other subvariants perform significantly worse in terms of total deck height (>> 100 mm increase in height). Subvariant 2a – column every 2 parking bays, struts off and the subvariants of variant 3 without struts are the worst performers and can be ruled out as potential framing systems.

10.5.2 Weight per Level

With the focus on the weight per level, hypothesis H1b, H2b, H3b and H4b can be tested.

H1b: Increasing the parking deck span will result in an increase of the total construction weight.

Hypothesis H1b can be confirmed as the results have shown that an increase in parking deck span results in an increase of the structures weight.

H2b: The use of struts in the main span of the structural system of the car park will result in a decrease of the total construction weight.

What can be concluded for the comparison of the weight per level of the structural systems is that again, the use of struts is preferred to reduce the structures weight (5-11% reduction). This confirms hypothesis H2b.

H3b: Increasing the distance between columns in transverse direction will result in an increase of the total construction weight.

It was expected that a bigger column distance in the transverse direction to the main parking deck span would result in an increase of the weight, as the larger loads on the elements would increase its size and weight. However, based on the results, hypothesis H3b cannot be confirmed. A column distance of two parking bay widths reduces the weight of the structure between 5 and 13% compared to a distance of a single parking bay width. The reduction in weight however is smaller compared to the increase in total deck height in the case of this parameter.

H4b: Increasing the distance between joists will result in an increase of the total construction weight.

Focussing on design variant 3, the use of a small maximum joist distance is preferred as this reduces the weight of the structure by around 5%. This confirms hypothesis H4b.

For the potential remaining framing systems for the car park design, variant 1 can be ruled out as an option, since its weight is around twice as big as the weight of the other variants.

10.5.3 Load on Foundation

In chapter 8, possible foundation types were studied for the structural foundation of the car park. It was concluded that by limiting the load on the foundation to 1,000 kN, it is possible to use a shallow foundation in most places in The Netherlands, where sand layers are found near the surface. For such a load, it is also often possible to create a pile foundation using multiple timber piles. Therefore, the loads on the foundation of the various subvariants are compared to this number.

For all spans and number of levels, doubling the column distance in the transverse direction of the main parking deck span results in roughly a doubling of the load on the foundation. Therefore, reducing this distance to a single parking bay width is preferred when the loads on the foundation have to be reduced. The subvariants using such a distance are the best performing and within these, subvariant 2a outperforms subvariant 1 by 22% (190 kN for 4 levels). The variation between the subvariants of variant 2b and 3 is not significant and lies around 1%. With subvariant 2a – column every 1 parking bay, struts on, 5 levels can be constructed with a load on the foundation lying still far below the 1,000 kN at 848 kN.

10.5.4 Choice of Framing System

The final choice for the best performing framing system is subvariant 2a - column every 1 parking bay, struts on. It is the best performing subvariant in terms of both total deck height and column load on the foundation. Amongst the weight of the structure, it performs in the same range as most other good performing subvariants.

Under alternative requirements, the biggest shift in performance is expected from design variant 1. It significantly improves in terms of total deck height and furthermore its self-weight is expected to be almost halved. Further study should prove if it is possible to use alternative assessment criteria and if indeed variant 1 in this case is the best performing design alternative.
PART V: Evaluation and Conclusions

11 Case Study

The previous chapters have proven it is possible to design a structural system for a temporary and multi-storey car park primarily in timber. However, it is also very relevant to compare such a design against alternative (existing) concepts and compare how it performs relative to these alternatives. This chapter therefore makes this comparison and answers the fourth sub- research question: "How does the proof of concept for a temporary multi-storey car park, using timber as primary structural material compare to alternative modular and/or timber multi-storey car park concepts?".

11.1 Reference Projects

In chapter 7, a total of five different car park structures have already been compared to each other on the same aspects as the different timber design variants in previous chapter. These car parks have either modular structural systems, are designed for disassembly and reuse or are designed using timber as primary material. These five car park designs are used as reference projects for comparison with the chosen design in timber, developed in this thesis. The five reference car park projects are:

- ModuPark (steel frame with concrete TT-panels as deck);
- Car park Koopman int. (steel frame with fibreglass deck);
- Car park Morspoort (steel frame with steel/concrete composite deck);
- Concept in BauBuche (timber frame with prefab concrete deck panels);
- B&O-Holzparkhaus (timber frame with timber CLT deck).

11.2 Comparison

The comparison between the reference projects and the developed timber concept are made on total deck height and weight, see Table 48. For the weight, the weight for a standard area of the car park for one level is compared as well as the load on the foundation coming from a "standard" column for four levels (see paragraph 7.2.3 for explanation of a "standard" column).

Car park	Floor depth		Weight /level		Load standard		d column on foundation 4 levels
	[mm]	[%]	[kg]	[%]	[kN]	[%]	
ModuPark	550	83%	61,261	402%	2,472	365%	(5.0 m column distance)
Koopman int.	640	96%	10,477	69%	1,258	186%	(5.0 m column distance)
Morspoort	651	98%	35,110	230%	1,845	272%	(5.0 m column distance)
BauBuche concept	730	110%	59,341	389%	1,193	176%	(2.5 m column distance)
B&O Holzparkhaus	935	140%	17,547	115%	717	106%	(2.5 m column distance)
Variant 2a - column every 1 parking bay, struts on	666	100%	15,242	100%	678	100%	

Table 48. Comparison between developed concept in timber vs reference projects including relative performance.

11.2.1 Total Deck Height

The total deck height of the chosen design variant for the timber car park is 666 mm and falls in the middle of the reference projects as can be seen in Figure 134. All values used in the comparison are also presented in Table 48, which also gives the performance of the developed concept relatively to the reference projects. The developed concept performs similar to the concepts with lightweight fibreglass and steel/concrete composite decks of the Koopman int. and Morspoort car parks. Its total deck height is only 4% and 2% higher. Relatively to the ModuPark concept, it is 17% thicker, but relatively to the BauBuche concept it is 10% thinner. In comparison with the realised car park in timber (B&O-Holzparkhaus), it outperforms it in terms of total deck height by 40%. In conclusion, it can be stated that the total deck height of the developed concept is an improvement compared to the existing car park in timber (B&O-Holzparkhaus) and it falls into the same range as the alternative concepts in the comparison.



Figure 134. Comparison of total deck height between developed modular timber concept vs concepts from case study.

11.2.2 Weight per Level

The weight for the given area of a single level of the chosen variant for the timber car park is 15,242 kg. Timber again proves itself to be a very lightweight material. The only structural system of a reference project that results in a lower weight is that of the Koopman int. car park, which uses extremely lightweight fibreglass deck panels, see Figure 135. The developed timber concept is 31% heavier compared to this concept, but is much lighter than the structures of most other reference projects. In comparison with the existing timber car park, it is 15% lighter. In comparison, the Morspoort car park with its steel/concrete composite deck is already more than twice as heavy. The ModuPark concept and BauBuche concept are even around 4 times as heavy. It can therefore only be concluded that the developed timber concept is very lightweight and has a self-weight much lower than most reference projects.



Figure 135. Comparison of weight per level between developed modular timber concept vs concepts from case study.

11.2.3 Load on Foundation

When studying the weight of the structure, also the load of a column of the structure on the foundation is studied. As stated before in the previous chapter, a structural system with a column distance in transverse direction to the main parking deck span of a single parking bay width performs much better than when double this distance is used. Of the reference projects, the ModuPark, Koopman int. car park and Morspoort car park all have a column distance of two parking bays. The BauBuche concept has a distance of a single parking bay. In the overall comparison, the developed timber car park design performs best at 678 kN. The B&O-Holzparkhaus results in a 6% higher load in the foundation. The other projects give much higher loads, see Figure 136. For the BauBuche concept and Kooman int. car park, they result in a load close to two times as large. For the Morspoort car park it is 2.7 times as high and for the ModuPark concept even 3.7 times higher. In conclusion, the developed proof of concept results in the lowest column loads on the foundation in comparison with the reference projects.

In chapter 8, it was concluded that for soils with large sand layers near the surface, a column load of 1,000 kN can be supported by a shallow foundation. The proof of concept with a load of 678 kN for 4 levels can therefore easily be supported by such a soil. For soils with a large sand layer in the first 20 m, a pile load of 678 kN would require between 2 and 5 timber foundation piles. In other cases, 1 up to 3 concrete foundation piles are required.



Figure 136. Comparison of load from standard column between developed modular timber concept vs concepts from case study.

11.2.4 Sustainability

In chapter 3, it was already discussed that the building industry has to make a transition towards a more sustainable industry. Circular construction was discussed as a method to come to a more sustainable industry. Furthermore, three themes were presented, focussing on incorporating circular design strategies into the design and construction of new buildings:

- Modularization, prefabrication and standardization;
- Reuse, refurbish and recycle;
- Building materials.

In the developed proof of concept, all of these themes can be found to some extend. Its design is made modular, with standardized elements that are prefabricated. The entire goal of the design is that it can be dismantled and reused and last of all, it is made of primarily timber instead of steel and concrete.

It is not the goal of this thesis to make a detailed sustainability analysis of the developed proof of concept and alternative car park concepts. However, aim is to be able to say something concerning how the proof of concept performs relative to the alternative concepts, which were discussed in chapter 7. Therefore, in this paragraph a simplified analysis is conducted on the embodied carbon in the various studied car park concepts. For this analysis, only the repetitive main structural elements of the deck and framing system of the car parks are considered. This excludes for example wind bracings, façades, connections, foundations and finishing layers.

The results of the embodied carbon analysis are presented in Table 49. It includes the total amount of embodied carbon for the considered area of the car park concepts (1 level, 10.0 m wide, 16.26 m span) for the main structural elements as discussed above. Furthermore, the embodied carbon intensity is presented, which gives the embodied carbon per unit area. The key figures used for the various materials are presented in Table 50. These key figures consider cradle to gate (A1-A3). The use and end-of-life phase are therefore not considered. For the various timber materials, multiple numbers are presented. Since during the growth of a tree, carbon is stored in the timber, it can have a negative embodied carbon number. In such a case, the end-of-life phase becomes important, as at this point, the stored carbon is again released to the atmosphere. For the comparison of the car park concepts, this negative number is not taken into account, however it is good to keep in mind the further potential of timber as a material that can store carbon. For concrete, different numbers apply considering if the steel used for rebar is recycled or not. For the comparison, it is assumed recycled steel is used. Finally, the amount of materials used in the different car park concepts can be found in appendix E.

In Figure 137 a comparison is made between the various car park concepts. From this graph, it becomes clear that the fully timber car park concepts (B&O and proof of concept) have by far the lowest amount of embodied carbon. The combined timber (BauBuche) and concrete concept already is at double the number of the fully timber concepts. Also the car park concepts, which use mainly steel and concrete have an amount of embodied carbon more than twice as high. Finally, the Koopman int. car park has the highest amount of embodied carbon, almost four times higher than the fully timber concepts. This is mainly due to the used GFRP decks, which have a very high embodied carbon rating.

In conclusion it can be stated that the performance in terms of sustainability, based on the simplified embodied carbon analysis of the proof of concept is far better compared to the other non-fully timber car park concepts. Since this analysis is not very detailed, a lot of assumptions are made, and a lot of structural elements of the car park for example are missing in the analysis, the final results can of course change in a more thorough analysis. However, it can also be expected that the performance of the proof of concept in a detailed analysis further improves. For example the possible smaller foundation works can further increase its performance. Also the potential carbon storage in timber was not considered in this brief analysis and brings a bigger potential. In conclusion, this brief analysis shows the potential of the proof of concept in terms of sustainability performance.

Table 49. Embodied carbon from main structural elements in reference area for various car park concepts.

Car park	Embodied carbon (no carbon storage)	Embodied carbon intensity (no carbon storage)	
	[kgCO₂e]	[kgCO₂e/m²]	
ModuPark	15,643	96.2	
Koopman int.	25,697	158.0	
Morspoort	17,386	106.9	
BauBuche concept	14,250	87.6	
B&O Holzparkhaus	6,900	42.4	
Variant 2a - column every 1 parking bay, struts on	6,852	42.1	

 Table 50. Key figures embodied carbon for various materials (Jones, 2019). *(Crawford, Stephan, & Prideaux, 2019)

	No carbon storage	e With carbon storage		
Motorial	Embodied Carbon	Embodied Carbon	Of which carbon storage	
Material	[kgCO₂e/kg]	[kgCO₂e/kg]	[kgCO ₂ e/kg]	
Timber - Softwood	0.263	-1.29	-1.55	
Timber - CLT	0.437	-1.20	-1.64	
Timber - Glulam	0.512	-0.90	-1.41	
Timber - LVL	0.390	-1.25	-1.64	
Steel - Rebar	1.990	n.a.	n.a.	
Steel - Section	1.550	n.a.	n.a.	
Concrete - Precast (non-recycled steel)	0.249	n.a.	n.a.	
Concrete - Precast (recycled steel)	0.194	n.a.	n.a.	
GFRP	5.0*	n.a.	n.a.	



Figure 137. Comparison of embodied carbon from main structural elements in reference area for various car park concepts.

11.3 Concluding Remarks

Concluding the comparison of the case study, the fourth and final sub-question can be answered: "How does the proof of concept for a temporary multi-storey car park, using timber as primary structural material compare to alternative modular and/or timber multi-storey car park concepts?".

It can be stated that the developed structural system for the modular and temporary car park in timber performs well on the criteria studied in this thesis. In terms of total deck height, it can compete with the various other existing modular and dismantlable concepts. Furthermore, its use of LVL decks results in a smaller deck height compared to other variants in timber. In terms of weight, using timber for almost all structural elements proves to result in a very lightweight car park. It is much lighter compared to other concepts which use concrete. Only the extremely lightweight fibreglass deck can result in a lower self-weight. Furthermore, the reduction in weight creates the potential to use smaller foundation structures. Finally, considering sustainability, it can be stated that a first simplified analysis into the amount of embodied carbon in the structural system of the proof of concept shows that the timber proof of concept is the best performing concept in comparison to the other modular car park concepts. Its amount of embodied carbon is at least halve that of the other non-fully timber build car park concepts, highlighting the potential of timber to bring the building industry further in its transition towards a more sustainable industry.

12 Conclusion

This chapter forms the conclusion of this master thesis. In chapter 1, the objective and research questions were presented. In previous chapters, the various sub-questions have been answered for which the conclusions can be found in the concluding paragraphs of these chapters. Finally, in this chapter, the answer to the main research question is presented.

12.1 Objective & Main Research Question

The objective of this study was to provide a structural system proof of concept for a temporary multistorey car park, using timber as the primary structural material. The proof of concept presented in this thesis results from a literature study into the construction of a multi-storey car park in timber, followed by a parametric study with structural validation on multiple design variants. Finally, it is compared in a case-study against five alternative multi-storey car park concepts. This has resulted in an answer to the following question:

"To what extent can timber be applied as a structural material in the structural system of a temporary multi-storey car park and how does it compare to alternative concepts?"

12.2 Proof of Concept

The developed proof of concept is the result of a parametric study and consists out of a short-span prefab timber deck system, supported by a timber framing system and is stabilized by steel bracing elements. Its design is presented in Figure 138. The deck is made from solid LVL panels with a thickness of 96 mm, coupled together to allow horizontal distribution of wind loads. These panels are 5 m long and are supported every 2.5 m by glulam main girders (570x280 mm for 16.26 m span). The main girders span the entire parking deck span, creating a column-free parking deck. They are simply supported at the edges by glulam columns (440x360 mm for 16.26 m span), transferring the loads to the foundation. Slanted struts (240x140 mm for 16.26 m span) between the columns and main girders reduce the unsupported length of these girders, resulting in a reduction of their cross-section. Steel L-sections are applied in X-bracings in the façades to vertically stabilize the structure. For the foundation, prefab concrete footings are required, since timber is prone to rot, which may or may not require support by foundation piles. In many locations in The Netherlands, timber foundation piles can be used and only in case no sufficient sand layers are found within the first 20 m of soil, concrete piles are to be used.

In the literature study, it was found that building in timber comes with many challenges, which can limit the applicability of timber in a multi-storey car park. These challenges are related to durability risks and proper detailing of the structure. Based on the literature study, it was concluded that for a temporary and demountable car park, carpentry joints are a potential option for joints with limited loads and which require only a low stiffness. For joints with higher loads or which require higher stiffness, dowel-type fasteners like bolts and slotted in steel plates or bolts in combination with connectors are potential connection types. To protect the timber structure against biological and environmental degradation, it was concluded that natural durability and constructional measures are always preferred over wood preservatives or modification. To protect the timber deck against degradation, the literature study concluded that a coating with aggregate is best applicable considering the dismantlability of the car park.



Figure 138. Design for proof of concept temporary multi-storey car park, primarily in timber.

12.3 Variant Study

The proof of concept is the result of a parametric study with structural validation into four different main typologies. Given all relevant aspects and the challenges found in the literature study, it was concluded that the main goal that these designs have to fulfil is threefold: focus on the lifecycle of the structural system, solve the material challenges related to building with timber and take into account the temporary nature of the to be developed concept. To be able to assess the performance of the design variants and compare the resulting proof of concept to alternatives, two specific sub-goals were derived from this main goal: minimization of total deck height and minimization of the weight of the structural system. Given the four main design variants, four geometric parameters were studied, altering the geometry of the main design variants, for their effect on the specific sub-goals. These are: parking deck span, use of struts, distance between columns in transverse direction and minimum distance between joists. Hypotheses were developed in paragraph 5.1.3.1, stating the expected effect of altering these parameters on the outcome of the sub-goals. Furthermore, the effect of altering the assessment criteria for fire resistance, vibrations and deformations on the specific sub-goals was studied. Based on the analysis of all (sub)variants, several conclusions can be drawn.

12.3.1 Parking Deck Span

The results from the parametric study show that an increase in parking deck span results in an increase of both total deck height and total weight of the structural system. For most design variants, this increase in height and weight is roughly linear. Only for the design variants using a long-span deck system, it has more of a logarithmic pattern.

12.3.2 Struts

The parametric study with structural validation has shown that the use of struts significantly reduces the cross-section of the main girder and as a result the total deck height (17-30% depending on design variant) for the design variants that may contain struts. Also a reduction in weight of the total structural system is found, although smaller than the reduction in total deck height (5-11%).

12.3.3 Transverse Column Distance

A transverse column distance of a single parking bay width (2.5 m) results in a smaller total deck height (16-23%) and smaller total weight of the structural system (5-13%) compared to using double the distance. Furthermore, due to the smaller area caried by a column, the load on the foundation is also roughly halved, allowing for smaller foundation works.

12.3.4 Distance between Joists

It is concluded that altering the distance between joists has a negligible effect on the total deck height. In terms of weight reduction, a decrease in maximum joist distance of 50% results in a reduction in weight of 5%.

12.3.5 Alternative Assessment Criteria

The effect of alternative assessment criteria has been studied mainly for different deck systems. The biggest effect is found for long-span deck systems.

The use of a long-span CLT rib panel deck system results in a relatively small total deck height (740 mm for 16.26 m span vs 666 m for proof of concept), but results in a doubling of the self-weight of the structural system in comparison with alternatives that use a short-span solid LVL panel deck system. However, alternative assessment criteria significantly increase the performance of long-span deck systems. When vibrational criteria are not considered and deformations of up to 1/150 x L are allowed, LVL rib panels become possible, resulting in a significant reduction in total deck height (537 mm for 16.26 m span) and almost halving of the self-weight of the deck. This results in a reduction of the weight of the total structural system of more than 40%.

The total deck height for design variants using a solid LVL panel as short-span deck system is highly dependant on the supporting framing system. Regardless of the framing system, the self-weight of the different variants varies within a margin of 20% and is at least halve that of the variant with a long-span deck system. In comparison to the design variants with a long-span deck system, for the design variants with short-span decks, alternative assessment criteria result in smaller reductions in total deck height (3-7%) and total weight of the structural system (10-20%).

12.4 Case Study

The proof of concept is compared to five alternative modular and/or timber multi-storey car park concepts for a span of 16.26 m. These alternatives have different typologies and materials and can be classified as followed: steel frame + concrete TT-plates, steel frame + GFRP deck, steel frame + steel-concrete composite deck, timber frame + concrete deck and fully timber frame and deck.

In terms of total deck height, the proof of concept has a deck height (LVL deck + glulam main girder) smaller than 700 mm (666 mm) and therefore fulfils the goal that was set in paragraph 7.3. It falls into the same range of deck heights as the other modular concepts in steel, concrete and GFRP. It is 17% thicker compared to the smallest deck height of the TT-plates, but 40% thinner compared to the existing alternative in timber.

Considering the self-weight of the structural system, the proof of concept has a self-weight slightly smaller (13% lower) compared to the existing alternative in timber. It cannot compete with the alternative concept, which uses GFRP decks (31% heavier), but is much lighter compared to the other systems which contain (partial) concrete decks. Compared to the steel-concrete composite deck system, the proof of concept has only halve its weight and compared to the fully concrete decks only a quarter.

Also the loads of a column of the structural system on the foundation were studied and with a load of 678 kN, it is lower than the goal set in paragraph 8.4 of 1,000 kN. In comparison with the alternative concepts, this is the lowest load. It is 10% smaller than the existing concept in timber and 43% smaller than the alternative concept in timber and concrete. The other alternatives all have twice the column distance compared to the proof of concept, but all result in a load at least twice that of the proof of concept.

Finally, a simplified sustainability performance analysis on embodied carbon indicates that the performance of the proof of concept is comparable to the existing alternative in timber. All of the other alternatives contain more than twice the amount of embodied carbon compared to the proof of concept and the alternative with GFRP decks almost four times as high. This analysis contains many assumptions, but is a first indication on the sustainability performance of the various concepts.

12.5 Final Word

The results of this thesis have shown through the development of a proof of concept, that a temporary multi-storey car park can be realised using primarily timber as structural material and that such a design can structurally compete with existing alternative modular multi-storey car park concepts in different materials.

13

Discussion & Recommendations

Following the conclusion, in this chapter, the results of this thesis are discussed and recommendations are presented. The discussion focusses on three parts: the results of the literature study, the results of the analysis of the different design variants for the proof of concept and the results of the case study. The discussions include validation and interpretation of the results, limitations of this research and recommendation for future research.

13.1 Discussion of Results Literature Study

In the literature study, three key topics have been addressed, which are concerned with the development of a proof of concept for a temporary and modular multi-storey car park primarily in timber. These are: car park design, challenges towards a sustainable building industry and timber structures. The findings from this literature study have resulted in a broad overview of relevant aspects of these various topics. Based on these results, in the concept development, certain choices have been made and some topics have been elaborated on more in detail, while others have not been further addressed. Therefore, the results from the developed proof of concept do not include all aspects discussed in the literature study. Below, some of these limitations in the results of the proof of concept are discussed.

13.1.1 Car Park Design

The requirements for multi-storey car parks and possible layouts have been extensively discussed in the literature study. For the structural system of the proof of concept, the requirements of the NEN 2443 norm have been considered as mandatory. Deviation of these requirements might result in a further improvement of the performance related to the specified sub-goals. Therefore, it is advised to further study what aspects of the norm limit the performance. Furthermore, the proof of concept is developed for a flat deck or split-level car park. No specific circulation layout has been designed and also the ramps are not included in the proof of concept. It is advised to further develop the proof of concept into different fully developed multi-storey car park designs, based on the findings from the literature study. This allows testing of the modularity and applicability of the concept as a temporary structure.

13.1.2 Building Sustainable

In the literature study, it was studied what principles can be applied to help the building industry in the transition towards a more sustainable industry. It was concluded that many of such principles are found in the approach of a circular economy, for which circular design strategies can be applied. Three key themes have been developed, which combine these circular design strategies and should be applied to the design of a temporary multi-storey car park to result in a sustainable structure:

- Modularization, prefabrication and standardization;
- Reuse, refurbish and recycle;
- Building materials.

These themes have been taken into account in the development of the proof of concept to a certain extend. For the choice of pavement surfacing, modularity has been taken into account and of course timber has been applied as a primary structural material. Additionally, many of the aspects of these themes are to be found in the detailing of the structure; how elements are connected, transported and reused for example. Such details have not been considered in the further development of the proof of concept in this thesis. It is recommended to further develop the proof of concept and apply these principles in the further detailing of the structure.

13.1.3 Building in Timber

Many relevant aspects and challenges of building in timber have been discussed in the literature study. Two important aspects that have not been further developed in the proof of concept and are important to point out are connections and durability.

The connections between different elements of a timber structure can significantly impact the dimensions of those elements. Due to the inhomogeneous material properties of timber, it can be a challenge to transfer loads between elements with different orientations or between elements of different materials. This can require larger cross-sections or use of different materials to be able to transfer the loads, adding complexity to the structure. Furthermore, connections with steel elements can be vulnerable to fire and require proper detailing. All together, the connections of a timber structure are very important and can impact the rest of the structural system. Therefore, it is recommended to further study the detailing of connections of the proof of concept and check what impact they have on the conclusions of this thesis.

Concerning durability of a timber structure, it was concluded that protection by constructional measures is the best solution. Such measures have not been the primary focus in the further development of the proof of concept. Protection can be realised by (slight alteration of) the main structural elements, but also by additional elements like the façade. Therefore, it is advised to further study how the timber elements can be properly protected from durability risks.

13.1.4 Suggestions for Future Research

Based on the discussion of the results of the literature study, the following recommendations for future research are suggested:

- Study what aspects of the NEN 2443 parking norm limit the performance of the proof of concept in terms of the specified sub-goals;
- Further develop the proof of concept into a fully completed design for a multi-storey car park, including ramps and a circulation layout, to study the modularity and performance in terms of user and owner requirements of the concept;
- Incorporate the circular design strategies in the further development of the proof of concept;
- Design the connections between all structural elements of the proof of concept;
- Further develop the design of the proof of concept to incorporate protection of the timber elements from durability risks.

13.2 Discussion of Results Design Variants for Proof of Concept

In total, four main design variants with multiple subvariants have been analysed in the development of the design for the proof of concept for the multi-storey car park. They consist out of a deck and framing system and are compared to each other on total deck height and weight of the structural system. Furthermore, effects of alteration of the assessment criteria for fire-safety, vibrations and deformations on these aspects have been studied. In this paragraph, the results of the comparison of these design alternatives and the effects of alternative assessment criteria are discussed.

13.2.1 Fire-safety

In the literature study, it was discussed that it is possible to allow a fire resistance class of R30 for a multi-storey car park if it is considered to be a single fire compartment. This is allowed based on the principle of equivalence, which has been assumed to be applicable. However, this is yet to be proven and requires further study into the subject. Since timber is often perceived to be a fire risk, requiring approval of a lower fire resistance class for a multi-storey car park in timber might be more difficult compared to one in steel and concrete.

For the proof of concept, only the decks have been analysed in detail for fire-safety. The other structural elements have not been validated for fire-safety. Nevertheless, for the proof of concept it is not expected to have an impact on the results. Verification of fire-safety ULS is expected to impact especially short span elements as was found for the deck elements. Therefore, the long-span main girders of the proof of concept are not expected to change as a result of the check on fire-safety. For other elements, it

might result in a slight increase in cross-section, but this is expected to only have a small effect on the total weight of the structural system and not alter the order of the studied variants.

As was concluded in the variant analysis, the impact of altering fire-safety requirements is rather limited. It impacts the short-span deck systems, but in absolute terms, the effect is very small. Therefore, alteration of the fire-safety requirements is not expected to influence the outcomes of the comparison of the proof of concept.

13.2.2 Vibrations

Vibrational criteria have been taken into account in the development of the structural system of the proof of concept for the deck systems. However, the criteria considered are developed with residential and office buildings in mind and may therefore not be applicable to a multi-storey car park. Since people spend only a short amount of time in a car park partially of which duration they are seated in their car (which has additional suspension), the applied criteria are considered to be a conservative upper limit for vibrational behaviour. According to the Dutch Building Degree, no specific criteria concerned with vibrations apply to a car park, but this does not mean vibrations cannot be unpleasant. Especially since timber structures are lightweight and can be vulnerable to vibrational behaviour, it is advised to further study what limitations in terms of vibrational behaviour should be applied to a multi-storey car park.

Apart from the deck systems, the other structural elements have not been validated for vibrational behaviour. For elements with only a short span, vibrational behaviour is not expected to result in different cross-sections than found in this study, since vibrations only became problematic for deck systems for spans of 10 m and longer. However, for the main girders of the various design variants with long spans, validation of vibrations might result in an increase in cross-section.

As the considered vibrational criteria are an upper limit, it is expected that these can be lowered, which influences the outcomes of the analysis. Altering just the vibrational criteria does not affect the short-span deck systems and therefore does not result in a change of the structural system of the proof of concept. Its conclusions therefore still stand. However, it does significantly impact the performance of the design variants with a long-span deck system. These variants should therefore be studied for the found vibrational criteria and again be compared to the proof of concept for its performance. It is expected that this does result in a better performance in terms of total deck height, but not in terms of total weight.

13.2.3 Deformations

The Eurocode gives advice concerning the maximum allowable deflections. As concluded in the literature study, according to the Dutch Building Degree, the designer can deviate from these values. Therefore, in this study, it has been analysed what the effect is of increasing the allowable deflections from 1/250*L to 1/150*L. However, this new value should be validated for what effect is has on the structural system of a modular car park. It is important to check how different elements fit together, to prevent additional impact loads between ramps and floor elements for example. Especially the effect of live loads on the deflections is important in such cases. Since timber is a lightweight material, the live load presents a bigger percentage in the total load on the structure and therefore, it might be concluded that bigger deformations are actually not possible or only for certain elements.

Different limits for deformations mainly impact the choice of long-span deck system. For short-span deck systems, only in combination with alternative criteria for both fire-safety and vibrations, does it result in a different type of panel in which case the effects are still limited. Therefore it is not expected that allowing bigger deformations has a significant effect on the design of the proof of concept. However, it does significantly impact the performance of the design variants with a long-span deck system if also different vibrational criteria apply. In that case, these variants should therefore be studied for the found alternative criteria and again be compared to the proof of concept for its performance. It is expected that this does result in a better performance in terms of total deck height, but not in terms of total weight.

13.2.4 Use of Struts

It was concluded that the use of struts has a significant effect on the total deck height (17-30%) and structural systems weight (5-11%). The applied inward strut distance is rather limited at just 670 mm. Increasing this distance might result in a further decrease of the deck height and weight of the structure.

Furthermore, for increased angles, the struts might also be able to provide vertical stability of the structure. However, the strut might hinder cars in entering and exiting the parking bay or people entering or exiting their vehicles. Therefore it is advised to study what effect increasing the angle of the struts has on the dimensions of the cross-sections of the various elements and if it can act as stabilizing element. Second, it is advised to study how far and at which height the strut can be located without hindering car parks users.

13.2.5 Suggestions for Future Research

Based on the discussion of the results of the design variants for the proof of concept, the following recommendations for future research are suggested:

- Study if fire-safety requirements can be altered based on the principle of equivalence;
- Analyse the framing system of the proof of concept for fire-safety requirements;
- Study what vibrational criteria are best applicable to a multi-storey car park;
- Analyse the framing system of the proof of concept for vibrational behaviour;
- Study what maximum deformations are allowed for the different structural elements of a multistorey car park, giving special attention to the impact of live loads on the deformations of timber elements;
- Study what effect increasing the angle of the struts has on the dimensions of the cross-sections
 of the various elements and if it can act as stabilizing element;
- Study how far and at which height a strut can be located without hindering car parks users.

13.3 Discussion of Results Comparison with Alternative Concepts

The developed proof of concept has been compared to five alternative modular and/or timber multistorey car park concepts on two themes: total deck height and weight of the structural system. These alternatives and the proof of concept have been compared to each other on four specific measurable aspects: total deck height, weight per level, load on foundation and embodied carbon. In this paragraph, the results of the comparison are discussed.

13.3.1 Structural Comparison

All of the alternatives studied have been designed with different starting points and goals in mind for different locations both in The Netherlands and abroad. This results in different structural systems with different geometric configurations and cross-sectional dimensions. To be able to compare the different concepts with each other, only the parking deck span and grid size of the various concepts have been translated to the same dimensions, to be able to make a relevant comparison on deck height and the structures weight. However, this leaves a lot of unknowns considering how the structure was designed and with what performance goals in mind. This can influence the validity of the comparison between the proof of concept and the alternative concepts. Examples of aspects that could affect the comparison are:

- Applied structural rules and norms;
- Applied loads on the structure;
- Applied deformation limits;
- Applied fire-safety class;
- Whether vibrational behaviour has been checked;
- Whether the structure has been over dimensioned for a possible different function or extension in the future;
- Cross-sections based on different grid size than used for comparison.

As a result of the aspects mentioned above, the various alternative concepts could actually perform both better or worse than was found in the comparison. Except for the concepts, which contain timber, all other alternative concepts have been realised in The Netherlands. Therefore, it is expected that these mainly make use of the same rules and norms and as a result are a good comparison. It is expected that the biggest impact on the comparison is the result of how vibrational criteria are applied and whether the structure has a significantly different grid size than used for the comparison. The alternative in timber has been realised in Germany, but is best to compare to since it is also fully made from timber. Such a comparison can be used to interpret the results of the proof of concept. In terms of total deck height, the proof of concept is 29% thinner compared to the existing alternative in timber, which uses a comparable typology. The deck of the proof of concept is 4 mm thinner (LVL instead of CLT deck), but the biggest difference is found for the height of the main girder. The timber alternative concept has a tapered glulam (BauBuche GL75) beam varying from 600x240 to 760x240 mm and the proof of concept uses a glulam (GL28h) beam of 570x280 mm. Even though the alternative concept uses a stronger material, the use of struts in the proof of concept explains the significantly lower beam height. In comparison, without the use of struts, the beam would require a height of 850 mm, which is somewhat higher than the beam of the timber alternative. This highlights the impact of using struts on the dimensions of the cross-section of the main girder. It also shows that by using BauBuche as a material, which was not considered in the proof of concept is slightly lighter as a result of its smaller cross-sections.

13.3.2 Sustainability Assessment

Focussing on the sustainability analysis of the embodied carbon, the analysis is very simplified and brief. Except for the key figure for the GFRP material, all figures come from the same source. Therefore, they have all been derived in the same way and are therefore comparable. Unfortunately, the source of the key figure for GFRP is different, which can affect the comparison, since possibly different criteria are used to derive this figure. The figure applied is in the lower region of the different values found and therefore expected to be a bottom line for the amount of embodied carbon for GFRP. For all alternatives, the same types of structural elements have been considered and materials used for finishing and connections have not been considered. However, this can have a significant effect on the amount of embodied carbon as could already be seen for the use of GFRP, which has a high amount of embodied carbon. Taking into account all materials used in the structure could therefore result in a different view of the comparison.

Considering the embodied carbon of the various concepts, it is based on the weight of the various materials used in their structural designs. Logically, the amount of embodied carbon in the proof of concept is slightly lower than in the fully timber alternative, since they use the same type of material, but the proof of concept is slightly lighter. As expected, the other concepts have significantly higher amounts of embodied carbon. The use of GFRP results in an amount of embodied carbon almost four times as high as the proof of concept. The use of GRRP as main structural material for the decks therefore has a high impact on the sustainability performance.

For timber materials, the possibility of storing carbon inside the timber is not taken into account in the assessment and is therefore expected to perform even better in a more detailed analysis. The presented number for amount of embodied carbon in the main structural elements is therefore expected to be a upper limit. For steel, in the comparison it is assumed the steel has been recycled. If this would not be the case, the amount of embodied carbon in the main structural elements of the concepts using steel would actually be higher. The presented number is therefore expected to be a bottom limit. The types and amounts of additional materials used for other elements like façades, installations, connections, finishing etc. are not expected to have an impact big enough to change the order of the comparison between the concepts. Foundations may be lighter for the timber and GFRP concepts, resulting in a lower additional amount of embodied carbon compared to the concepts using concrete. At the same time, the timber and GFRP concepts require an additional protective layer of their decks, which may result in an additional amount of embodied carbon. A detailed analysis should point out the exact amount of embodied carbon for each concept, but the order of performance of the concepts is not expected to change.

13.3.3 Suggestions for Future Research

Resulting from the discussion of the results of the comparison between the proof of concept and other alternative modular and/or timber multi-storey car park concepts, two suggestions are given for future research:

- Study use of BauBuche and other innovative sustainable materials;
- Make a detailed LCA study to asses the performance on sustainability of the various concepts.

Bibliography

Abebe, A., & Smith, I. G. N. (2016). *Pile Foundation Design: A Student Guide*. Edinburgh: School of the Built Environment, Napier University.

ACI Committee 362. (1997). Guide for the Design of Durable Parking Structures. Retrieved from

Adis. (nd). IGL3121178169. In.

- AMC. (2021). Industrial Bulky Goods UCPH Carpark Bruce Internal. In.
- Anastasiades, K., Blom, J., Buyle, M., & Audenaert, A. (2020). Translating the circular economy to bridge construction: Lessons learnt from a critical literature review. *Renewable and Sustainable Energy Reviews*, *117*. doi:10.1016/j.rser.2019.109522
- Anctil, D. (2021). Types of wood beams and joint methods. Retrieved from https://usihome.com/en/types-of-wood-beams-and-joints/
- Arets, E. J. M. M., Van der Meer, P. J., Verwer, C. C., Hengeveld, G. M., Tolkamp, G. W., Nabuurs, G. J., & Van Oorschot, M. (2011). Global wood production: Assessment of industrial round wood supply from forest management systems in different global regions (1808). Retrieved from Wageningen:
- ASK Romein. (2018a). ModuPark XXX. In.
- ASK Romein. (2018b). ModuPark Maarssen. In 1900_760modupark_4_1 (Ed.).
- Augustin, M., Bell, K., Kuklik, P., Lokaj, A., & Premrov, M. (2008). *Handbook 2 Design of Timber Structures according to EC 5*. Prague.
- Ayanleye, S., Udele, K., Nasir, V., Zhang, X., & Militz, H. (2022). Durability and protection of mass timber structures: A review. *Journal of Building Engineering*, 46, 103731. doi:https://doi.org/10.1016/j.jobe.2021.103731
- Baaij, R. (n.d.). Standaard brandkromme. In: Baaij Software Engineering.
- Babalola, J. A. (2017). Barriers to Implementation of Sustainable Construction Techniques.
- Ballast Nedam Parking. (2011). About ModuPark. Retrieved from https://www.modupark.com/this-ismodupark/about-modupark/
- Bekkering, P. (2017). Tijdelijke en definitieve parkeergarages bij ziekenhuizen. Mobiliteits Platform.
- Bertram, N., Fuchs, S., Mischke, J., Palter, R., Strube, G., & Woetzel, J. (2019). *Modular construction: From projects to products.* Retrieved from https://www.mckinsey.com/~/media/mckinsey/business%20functions/operations/our%20insig hts/modular%20construction%20from%20projects%20to%20products%20new/modularconstruction-from-projects-to-products-full-report-new.pdf
- Blaß, H. J., & Sandhaas, C. (2017). *Timber Engineering: Principles for Design*. Karlsruhe: Karlsruher Institut für Technologie (KIT).
- BNParking. (2016). Modupark. Retrieved from https://www.bnparking.com/products/modupark/
- Bocken, N. M. P., de Pauw, I., Bakker, C., & van der Grinten, B. (2016). Product design and business model strategies for a circular economy. *Journal of Industrial and Production Engineering*, 33(5), 308-320. doi:10.1080/21681015.2016.1172124

Bourne Parking Ltd. (2012). M1_Fig16. In: Steelconstruction.info.

- Bouwkosten Online. (2022). Bouwkosten Online Utiliteitsbouw Bodem- funderingsvoorzieningen. https://bouwkosten.bouwformatie.nl/bouwkosten/lib-viewer.aspx
- Bowen, P., Cattel, K., Hall, K., Edwards, P., & Pearl, R. (2012). Perceptions of Time, Cost and Quality Management on Building Projects. *Construction Economics and Building*, 2(2), 48-56. doi:https://doi.org/10.5130/AJCEB.v2i2.2900
- Box, A. (2020). Zes grote ontwikkelingen en uitdagingen in de bouwsector. Bouwmachines.
- Brand, S. (1995). How buildings learn: What happens after they're built: Penguin.

Breunese, A. J., & Maljaars, J. (2015). Fire safety design: Course reader CIE5131. Delft.

Brinksma, H. (2004). Jellema 3 Draagstructuur (2 ed.). Utrecht: ThiemeMeulenhoff.

British Parking Association. (2012). Report into Structural Safety and Maintenance of Car Parks in the UK. In.

- BROloket. (2021). Ondergrondmodellen BRO Grondwaterspiegel 2021 Gemiddeld kleinste diepte. https://www.broloket.nl/ondergrondmodellen
- Brzezinska, D., & Bryant, P. (2022). Performance-Based Analysis in Evaluation of Safety in Car Parks under Electric Vehicle Fire Conditions. *Energies*, *15*(2). doi:10.3390/en15020649

City of Stonnington (Writer). (2018). Cato Street car park first slab pour. In.

- Civil Today. (2019). Pile foundation definition types.
- Cobouw. (2017). ab parkeergar 2. In.
- Crawford, R., Stephan, A., & Prideaux, F. (2019). EPiC Database (Publication no. https://doi.org/10.26188/5dc228ef98c5a). from University of Melbourne
- Croom Concrete. (2018). Croom Hollowcore Brochure. In.
- Cross, J. (2002). Steel Parking Structures: A 25-year Perspective. Modern Steel Construction, 41.
- Cudney, G., & Smith, S. (2019). Parking structure cost outlook for 2019. https://wginc.com/wpcontent/uploads/2019/05/2019-Parking-Constr-Cost17x11.pdf
- Czarnecki, L., & Kapron, M. (2010). Sustainable construction as a research area. *International Journal* of the Society of Material Engineering for Resources, 17(2), 99-106. doi:10.5188/ijsmer.17.99
- Dams, B., Maskell, D., Shea, A., Allen, S., Driesser, M., Kretschmann, T., . . . Emmitt, S. (2021). A circular construction evaluation framework to promote designing for disassembly and adaptability. *Journal of Cleaner Production, 316*. doi:10.1016/j.jclepro.2021.128122
- de Lima, L. C. C., Costa, A. A., & Rodrigues, C. F. (2018). On the use of prestress for structural strengthening of timber beams: assessment with numerical support and experimental validation. *International Journal of Architectural Heritage, 12*(4), 710-725. doi:10.1080/15583058.2018.1442526

De Ruiter, P. (2011). Morspoort 4. In.

Dekfordt. (2019). Analysis of the future of the car park business. https://www.dekfordt.nl/wpcontent/uploads/2019/10/Future-of-parking-business-Dekfordt-2019.pdf Densley Tingley, D., & Allwood, J. (2014). Reuse of structural steel: the opportunities and challenges.

- Diffen. (n.d.). Hardwood vs softwood. Retrieved from https://www.diffen.com/difference/Hardwood_vs_Softwood
- Dijkstra, R. (2014). *Brandwerendheid bij een natuurlijke parkeergaragebrand.* Christelijke Hogeschool Windesheim, Zwolle.
- DINOloket. (2022). Ondergrondgegevens. https://www.dinoloket.nl/ondergrondgegevens
- Drenth, J. (2020). 5 Trends in de ontwikkeling van parkeergarages. https://www.carparks.nl/nieuws/5trends-in-de-ontwikkeling-van-parkeergarages
- Ducorr. (2018). TREATING REINFORCED CONCRETE CORROSION IN PARKING STRUCTURES -PART 1. In.
- Durmisevic, A. (2019). Reversible building design. Retrieved from
- Durmisevic, E. (2006). Transformable building structures: design for dissassembly as a way to introduce sustainable engineering to building design & construction.
- EdJ. (2019). Dwarsdoorsneden nieuwe situatie. In: Park4All.

Elliott Brothers Ltd. (n.d.). I-Joist FAQs. Retrieved from https://elliotts.uk/i-joist-faqs

- Englot, J. M., & Davidson, R. I. (2001). Steel Framed Parking Garages Take Off. *Modern Steel Construction, 40*.
- European commission. (2015). First circular economy action plan. Retrieved from https://ec.europa.eu/environment/topics/circular-economy/first-circular-economy-actionplan_en
- Fiberline. (2021). Fact sheet planks. Retrieved from https://fiberline.cloud1.structpim.com/media/tj1p3fbe/fact_sheet_planks.pdf
- Florum. (2020). Bodemsoorten Nederland. In.
- Föreningen Sveriges Skogsindustrier. (2019). *The CLT Handbook* (1st ed.). Stockholm: Skogsindustrierna.
- Forest & Wood Products Australia Limited. (2008). Plywood Box Beam Span Tables for Detached Housing Construction. In F. W. P. A. Limited (Ed.). Melbourne.
- Fridley, K. J. (1999). Timber Structures: Structural Engineering Handbook: CRC Press LLC.
- Fryer, K. (2012). Resistance to change holds timber back. Retrieved from https://www.ttjonline.com/blog/resistance-to-change-holds-timber-back/
- Geldermans, R. J. (2016). Design for Change and Circularity Accommodating Circular Material & Product Flows in Construction. *Energy Procedia*, 96, 301-311. doi:https://doi.org/10.1016/j.egypro.2016.09.153
- Goesten, S. (2022). [Brandveiligheid parkeergarages].
- Guerra, B. C., & Leite, F. (2021). Circular economy in the construction industry: An overview of United States stakeholders' awareness, major challenges, and enablers. *Resources, Conservation and Recycling, 170.* doi:10.1016/j.resconrec.2021.105617

- Guy, B., & Ciarimboli, N. (2008). *DfD: design for disassembly in the built environment: a guide to closed*loop design and building: Hamer Center.
- Haitsma Beton. (2018). Haitsma Beton Parkhaus ASML. In.
- Hakkarainen, J., Linkosalmi, L., Huovinen, A., Vares, S., Häkkinen, T., & Veikkola, M. (2019). *LVL Handbook Europe*. Helsinki: Federation of the Finnish Woodworking Industries.
- Hall, A. D. (1968). A Methodology for Systems Engineering (6th edition ed.). Princeton, NJ.: Van Nostrand.
- Hamerlinck, A. F., Breunese, A., Noordijk, L. M., Jansen, D. W. L., & Van Oerle, N. J. (2011). *Richtlijn: Brandveiligheid Stalen Parkeergarages*. Retrieved from Zoetermeer: https://www.brandveiligmetstaal.nl/upload/File/actueel/Richtlijn_Brandv_Parkeerg_BmS_1e_e ditie_nov2011.pdf
- Harries, K., Petrou, M., & Brooks, G. (2000). Structural Characterization of Built-Up Timber Columns. Journal of Architectural Engineering, 6. doi:10.1061/(ASCE)1076-0431(2000)6:2(58)
- Harte, A. M., & Dietsch, P. (2015). Reinforcement of Timber Structures A state-of-the-art report.
- Haynes, M., Coleri, E., & Estaji, M. (2019). Selection of the most effective pavement surfacing strategy for the Glenwood cross laminated timber parking garage. *Construction and Building Materials*, 226, 162-172. doi:https://doi.org/10.1016/j.conbuildmat.2019.07.220
- Heller, T. S., Hawgood, J. L., & Leo, D. D. (2007). Correlates of Suicide in Building Industry Workers. *Archives of Suicide Research, 11*(1), 105-117. doi:10.1080/13811110600992977
- Herzog, T., Söffker, G. H., & Thrift, P. (2004). Timber construction manual. Basel: Birkhäuser.
- Hill, J., Rhodes, G., Vollar, S., & Whapples, C. (2005). *Car Park Designers' Handbook*. London: Thomas Telford Limited.
- HIM. (2020). Himgrip NT. In. IJmuiden.
- HIM. (2021). EN-Slijtlagen Wearingcourses. In. IJmuiden.
- HK Architecten. (2021). PARKHAUS B&O PARK BAD AIBLING. In. Bad Aibling.
- HK Architecten. (2022). B&O wooden multi-storey car park, Bad Aibling. In.
- llarch. (2017). Beam of whole timber. In.
- J.L. Bijnagte, A.F. Van Tol, & R. Elprama. (2006). Het effect van grondwaterstijging op funderingen op staal. *Geotechniek*.
- JAJA Architects + Open Platform. (2019). Billede2-min. In.
- Jansen, D. (2010). Autobranden in parkeergarages brandscenario's, brandsimulaties en de gevolgen voor constructies. Retrieved from Gent:
- Jansen, E. (2013). Rotterdam krijgt mobiele parkeergarage. Mobiliteits Platform.
- Jansze, W., & Van Acker, A. (2014). *Structural behaviour of prestressed concrete hollow core floors exposed to fire*. Retrieved from 's Hertogenbosch:
- Johnson, J., Reck, B. K., Wang, T., & Graedel, T. E. (2008). The energy benefit of stainless steel recycling. *Energy Policy*, *36*(1), 181-192. doi:https://doi.org/10.1016/j.enpol.2007.08.028

Jones, C. (2019). ICE (Inventory of Carbon & Energy). from Circular Ecology https://circularecology.com/embodied-carbon-footprint-database.html

JVZ Ingenieurs. (2012). Tijdelijke Parkeergarage Morspoort Leiden. In. Deventer.

- Kaa, B. V. d. (2020). Hoofdberekening Garage Distelweg 74. Retrieved from Den Haag:
- Kaminker, A. J. (2004). Lester B. Pearson International Airport Parking Structure. *Practice Periodical* on Structural Design and Construction, 9(4), 9.
- Kaufmann, H., Krötsch, S., & Winter, S. (2018). *Manual of multi-storey timber construction* (First ed.). Munich: Detial Business Information GmbH.
- Kayaçetin, N. C., Verdoodt, S., Leferve, L., & Versele, A. (2022) Evaluation of Circular Construction Works During Design Phase: An Overview of Valuation Tools. In: Vol. 263. Smart Innovation, Systems and Technologies (pp. 89-100).
- Kersten, B. (2011). Brandveiligheid van parkeergarages. Stedebouw & Architectuur, 2011.
- Kralj, D. (2008). Building Materials Reuse and Recycle. WSEAS TRANSACTIONS on ENVIRONMENT and DEVELOPMENT, 4(5).
- Kuklik, P., Augustin, M., Malo, K. A., Angst, V., Materna, A., Lokaj, A., . . . Marynowicz, A. (2008). *Handbook 1 - Timber Structures*. Graz.
- Le, H.-B., & Bui, Q.-B. (2020). Recycled aggregate concretes A state-of-the-art from the microstructure to the structural performance. *Construction and Building Materials*, 257, 119522. doi:https://doi.org/10.1016/j.conbuildmat.2020.119522
- Lemieux, D., & Van Kampen, C. (2018). How to protect hollow core slabs in parking structures from deicing salts. *PCI Journal*.
- Matar, M. M., Georgy, M. E., & Abou-Zeid, A. M. (2010). Developing a BIM-oriented data model to enable sustainable construction in practice. Paper presented at the eWork and eBusiness in Architecture, Engineering and Construction - Proceedings of the European Conference on Product and Process Modelling 2010.
- Max Boegl. (2020). Parkeergarages. Retrieved from https://maxboegl.de/nl/prestatiegebieden/hoogbouw/parkeergarages
- McKinsey & Company. (2017). *Reinventing Construction: A Route To Higher Productivity*. Retrieved from https://www.mckinsey.com/business-functions/operations/our-insights/reinventingconstruction-through-a-productivity-revolution

Ministerie van Binnenlandse Zaken en Koninkrijkrelaties. (2021). Bouwbesluit 2012. In.

- Mischke, J. (2017). Reinventing Construction: A Route to Higher Productivity. Retrieved from
- Mobiliteits Platform. (2013). Tijdelijke parkeergarage Nieuwersluisstraat Den Haag. https://www.mobiliteitsplatform.nl/artikel/tijdelijke-parkeergarage-nieuwersluisstraat-den-haag
- Möller, T., Padhy, A., Pinner, D., & Tschiesner, A. (2019). The future of mobility is at our doorstep. https://www.mckinsey.com/industries/automotive-and-assembly/our-insights/the-future-ofmobility-is-at-our-doorstep#

Morgan, C., & Stevenson, F. J. S. E. D. A. (2005). Design and detailing for deconstruction.

Muralikrishna, I. V., & Manickam, V. (2017). Chapter Five - Life Cycle Assessment. In I. V. Muralikrishna & V. Manickam (Eds.), *Environmental Management* (pp. 57-75): Butterworth-Heinemann.

- Eurocode 5: Ontwerp en berekening van houtconstructies Deel 1-1: Algemeen Gemeenschappelijke regels en regels voor gebouwen, NEN-EN 1995-1-1+C1+A1 (nl) C.F.R. (2011).
- Nederlands Normalisatie Instituut. (2013). NEN 2443 Parkeren en stallen van personenauto's op terreinen en in garages. In. Delft.
- Nederlands Normalisatie Instituut. (2015). NEN-EN 1990+A1+A1/C2 (nl) Eurocode: Grondslagen voor het constructief ontwerp. In. Delft.
- Nederlands Normalisatie Instituut. (2019a). NEN-EN 1991-1-1+C1+C11:2019 nl Eurocode 1: Belastingen op constructies - Deel 1-1: Algemene belastingen - Volumieke gewichten, eigen gewicht en gebruiksbelastingen voor gebouwen. In. Delft.
- Nederlands Normalisatie Instituut. (2019b). NEN-EN 1991-1-3+C1+A1:2019 nl Eurocode 1: Belastingen op constructies Deel 1-3: Algemene belastingen Sneeuwbelasting. In. Delft.
- NEN. (2021). Overzicht Eurocodes. Retrieved from https://www.nen.nl/bouw/constructieveveiligheid/eurocodes/overzicht-eurocodes
- Niklewski, J., Brischke, C., Frühwald Hansson, E., & Meyer-Veltrup, L. (2018). Moisture behavior of weathered wood surfaces during cyclic wetting: measurements and modeling. *Wood Science and Technology*, *52*. doi:10.1007/s00226-018-1044-8

Ons Water. (2020). Grondwater. Retrieved from https://www.onswater.nl/onderwerpen/grondwater

Osborne. (2019). Temporary single deck car park solution didcot station. In.

- Park4all. (2019a). Parking Solutions. Retrieved from https://www.park4all.com/
- Park4all. (2019b). T8A3323. In.
- Park4all. (2020). Uitgangspunten Garage Distelweg 74. Retrieved from Den Haag:
- Patil, D., Patil, A., & Patil, J. (2021). The Decision-Making Criteria for Adaptive Reuse for Sustainable Development. In (pp. 599-607).
- Petovarga. (2019). Circular vs linear economy. In.
- Pitt, M., Tucker, M., Riley, M., & Longden, J. (2009). Towards sustainable construction: promotion and best practices. *Construction Innovation*, *9*(2), 201-224. doi:10.1108/14714170910950830

Pollak, B. S. (2003). Parking Perspective. Modern Steel Construction, 42.

- Precast/Prestressed Concrete Institute. (1997). Parking Structures: Recommended Practice for Design and Construction.
- Ritter, M. A. (1990). *Timber Bridges: Design, Construction, Inspection, and Maintenance*. Washington, DC.
- Roberts, D. (2020). The hottest new thing in sustainable building is, uh, wood. *Vox*. Retrieved from https://www.vox.com/energy-and-environment/2020/1/15/21058051/climate-change-buildingmaterials-mass-timber-cross-laminated-clt
- Sagan, J., & Sobotka, A. (2021). Analysis of factors affecting the circularity of building materials. *Materials, 14*(23). doi:10.3390/ma14237296

Sarka, J. (2005). Prescription for Parking. Modern Steel Construction, 44.

- Scharmacher, F., Müller, A., & Brunner, M. (2014). *Asphalt surfacing on timber bridges*. Paper presented at the COST Timber Bridges Conference.
- Schiere, M., Müller, A., Bueche, N., & Angst, C. (2022). Creating a connection between asphalt wearing surface and timber bridge decks. Paper presented at the 4th International Conference on Timber Bridges, Biel/Bienne, Szwitzerland.
- Schirén, W., & Swahn, T. (2019). Vibrations in residentail timber floors. Retrieved from https://www.divaportal.org/smash/get/diva2:1354527/FULLTEXT01.pdf
- Simon, A. H. (2001). Unique Steel Framed Solution to Parking. Modern Steel Construction, 40.
- Simon, A. H. (2002). Parking with Class. Modern Steel Construction, 41.
- Simon, A. H. (2010). Economy Parking. Modern Steel Construction, 49.
- SMD Ltd. (2019). Kellelea Sheffield Health MSCP. In.
- Soons, F. A. M., Van Raaij, B. P. M., Wagemans, L. A. G., Pasterkamp, S., & Van Es, S. H. J. (2014). Quick Reference. In D. U. o. Technology (Ed.), (2014 ed.). Delft.
- Stöbich. (2022). Brandschermen. Retrieved from https://www.stoebich.com/nl/brandschermen
- Stora Enso. (2022a). Calculatis: a state-of-the-art timber design tool. Retrieved from https://www.storaenso.com/en/products/wood-products/calculatis
- Stora Enso. (2022b). CLT by Stora Enso Technical Brochure. In S. Enso (Ed.).
- Stora Enso. (2022c). Cross-laminated timber (CLT). Retrieved from https://www.storaenso.com/en/products/mass-timber-construction/building-products/clt
- Stora Enso. (2022d). Laminated veneer lumber (LVL). Retrieved from https://www.storaenso.com/en/products/mass-timber-construction/building-products/lvl
- Stora Enso. (2022e). LVL by Stora Enso Technical Brochure. In S. Enso (Ed.).
- Stora Enso. (2022f). Rib panels. Retrieved from https://www.storaenso.com/en/products/mass-timberconstruction/building-products/rib-panels
- Stora Enso. (2023). Calculatis. Retrieved from https://calculatis.storaenso.com/
- Structural Board Association. (2004). OSB Design Manual Construction Sheathing and Design Rated Oriented Strand Board. Ontario.
- Strumiłło, K. (2016) Adaptive reuse of buildings as an important factor of sustainable development. In: Vol. 493. Advances in Intelligent Systems and Computing (pp. 51-59).
- Stuart, M. (2007). Parking Structures. In.
- Sun, P., Bisschop, R., Niu, H., & Huang, X. (2020). A Review of Battery Fires in Electric Vehicles. *Fire Technology*, *56*(4), 1361-1410. doi:10.1007/s10694-019-00944-3
- Tata Steel. (2021). Quantum deck vloer. Retrieved from https://www.tatasteeleurope.com/nl/bouw/produkten/staalplaat-betonvloer/quantum-deckvloer
- Tavoussi, K., Winter, W., Pixner, T., & Kist, M. (2010). *Steel reinforced timber structures for multi storey buildings*. Paper presented at the World Conference on Timber Engineering.

- Technical University of Munich, & Pollmeier Massivholz GmbH & Co.KG. (2015). Development of construction system for multi-storey car parks in BauBuche. https://www.pollmeier.com/dam/jcr:2626b356-5c49-4871-b32b-998316137ea8/Pollmeier%20BauBuche%20-%20Multi-Storey%20Car%20Park.pdf
- The Institution of Structural Engineers. (2011). *Design recommendations for multi-storey and underground car parks*. Retrieved from London:
- Troup, E., & Cross, J. (2003). *Innovative Solutions in Steel: Open-Deck Parking Structures*. Retrieved from https://www.scribd.com/document/62555680/Aisc-Open-Deck-Parking-Solutions
- Ülker, M., Işık, E., & Ülker, M. (2017). The Effect of Centric Steel Braced Frames with High Ductility Level on the Performance of Steel Structures.
- United Nations Environment Programme. (2020). 2020 Global Status Report for Buildings and Construction: Towards a Zero-emission, Efficient and Resilient Buildings and Construction Sector. Retrieved from Nairobi: https://globalabc.org/sites/default/files/inline-files/2020%20Buildings%20GSR_FULL%20REPORT.pdf

Van den Broek, T. (2019). Belasting van parkeergarages.

Van Drooge, J. (2020). Achter de slagboom. Test: 40 parkeergarages. Kampioen, 2020(12), 24-30.

- Van Herpen, R. A. P. (2014). Brandbeveiligingsconcepten voor parkeergarages. Retrieved from Zwolle:
- Vroom Funderingstechnieken. (2019). Betonnen paalfunderingen. Retrieved from https://betonhuis.nl/betonmortel/betonnen-paalfunderingen
- Walker, C. (2016). Mean Construction Costs. Retrieved from http://www.carlwalker.com/wpcontent/uploads/2016/05/2016-Carl-Walker-Cost-Article.pdf
- WCED. (1987). Our Common Future [Brundtland Report]. Retrieved from Oxford:
- White, K., Hardisty, D. J., & Habib, R. (2019). The Elusive Green Consumer. *Harvard Business Review*, 97(4), 124-133.
- Winter, W., Tavoussi, K., Riola-Parada, F., & Bradley, A. (2016). *Timber-Steel Hybrid Beams for Multi-Storey Buildings: Final Report.*
- Wood for Good. (2011). The Timber Industry Manifesto.
- Woods, S. (2016). A History of Wood from the Stone Age to the 21st Century. *The Journal of the American Institute of Architects*.
- Zanon, R. (2019). Floor Vibration Behavior of Car Park Structures Assessment of Different Steel Concrete Solutions. Paper presented at the 7th ECCOMAS Thematic Conference on Computational Methods in Structural Dynamics and Earthquake Engineering, Crete.
- Zaugg, M. (2018). *Dawn of a New Era World First Timber-Built Car Park*. Paper presented at the 8e Forum International Bois Construction FBC, Dijon. http://www.forumboisconstruction.com/conferences/91_FBC2018_Zaugg.pdf

Appendices

A.

Structural Loads and Limit States

A.1 Load Combinations (ULS/SLS)

Ultimate Limit States (ULS)

Table 51. Partial factors on actions (STR/GEO) (group B) (Nederlands Normalisatie Instituut, 2015)

Design situation	Permanent		Leading variable	Accompanying variables	
Formula for combination of actions	Unfavourable	Favourable		Dominant (if present)	Others
(Eq. 6.10a)	1.35 G _{k,j,sup} ª	0.9 G _{k,j,inf}		1.5 ψ _{0,1} Q _{k,1}	1.5 ψ _{0,1} Q _{k,I} (i>1)
(Eq. 6.10b)	1.2 G _{k,j,sup} ^b	0.9 G _{k,j,inf}	1.5 Q _{k,1}		1.5 ψ _{0,1} Q _{k,I} (i>1)
^a In the case of fl ^b This value is ca	uid pressures with a alculated with $\xi=0.89$	a physically limited	d value, the following	shall suffice: 1.2 $G_{k,j,sup}$	

Accompanying equations:

$$\sum_{j\geq 1} \gamma_{Gj} G_{kj} "+" \gamma_P P" + " \gamma_{Q,1} \psi_{0,1} Q_{k,1} "+" \sum_{i>1} \gamma_{Q,i} \psi_{0,i} Q_{k,i} \qquad (Eq.\, 6.10a) \qquad \qquad 0.1$$

$$\sum_{j\geq 1} \xi_{j} \gamma_{Gj} G_{kj} "+" \gamma_{P} P" + " \gamma_{Q,1} Q_{k,1} "+" \sum_{i>1} \gamma_{Q,i} \psi_{0,i} Q_{k,i} \qquad (Eq. \ 6.10b) \qquad 0.2$$

Serviceability Limit States (SLS)

Characteristic

Irreversible serviceability limit states are generally assessed using this combination of actions.

$$\sum_{j\geq 1} G_{kj}"+"P"+"Q_{k,1}"+"\sum_{i>1} \psi_{0,i}Q_{k,i}$$
(Eq. 6.14b) 0.3

Frequent

Reversible serviceability limit states are generally assessed using this combination of actions.

$$\sum_{j\geq 1} G_{kj}"+"P"+"\psi_{1,1}Q_{k,1}"+"\sum_{i>1} \psi_{2,i}Q_{k,i}$$
(Eq. 6.15b) 0.4

Quasi-permanent

Long-term effects and the appearance of the structure are generally assessed using this combination of actions.

$$\sum_{j\geq 1} G_{kj}"+"P"+"\sum_{i>1} \psi_{2,i} Q_{k,i} \tag{Eq. 6.16b} 0.5$$

An overview of the combinations of actions can be found in Table 52.

Table 52. Overview of combinations of actions. (Nederlands Normalisatie Instituut, 2015)

Combination of actions	Permanent (G _d)		Prestressing (P _d)	Variable		
	Unfavourable	Favourable	(-)	Leading $(Q_{d,1})$	Accompanying (Q _{d,j})	
Characteristic	Gk,j,sup	Gk,j,inf	Р	Q k,1	Ψ0,i Q k,i	
Frequent	G _{k,j,sup}	Gk,j,inf	Р	Ψ1,1 Q k,1	$\psi_{2,i} \mathbf{Q}_{k,i}$	
Quasi- permanent	$G_{k,j,sup}$	$G_{k,j,\text{inf}}$	Р	ψ2,1 Qk,1	$\psi_{2,i}\;Q_{k,i}$	

Combination factors

Table 53. Combination factors for buildings. (Nederlands Normalisatie Instituut, 2015)

Action	$\boldsymbol{\psi}_0$	ψ 1	ψ 2
Imposed loads on buildings:			
Category F: Traffic area, vehicle weight ≤ 25 kN	0.7	0.7	0.6
Snow loads	0	0.2	0
Wind loads	0	0.2	0
Temperature (not fire)	0	0.5	0
Standing water	0	0	0

A.2 Variable Loads

Snow

The snow load can be calculated using equation 0.6. In The Netherlands, C_e and C_t are equal to 1. The characteristic value s_k in The Netherlands is equal to 0.7 (Nederlands Normalisatie Instituut, 2019b). The snow load shape coefficient μ_i depends on the shape of the roof and is therefore dependant on the design. Equation 0.6 can now be reduced to equation 0.7.

 $s = 0,7 \cdot \mu_i$

$$s = \mu_i \cdot C_e \cdot C_t \cdot s_k \tag{0.6}$$

0.7

Where:

s

- is the characteristic value of the snow load [kN/m²]
- s_k is the characteristic value of the snow load on the ground [kN/m²]
- μ_i is the snow load shape coefficient [-]
- Ce is the exposure coefficient [-]
- Ct is the thermal coefficient [-]

Wind

The wind load on a structural element can be calculated using equation 0.8. Since the car park will be an open structure, it is assumed the factor c_sc_d is equal to 1. Factor c_f depends on the layout of the building. Factor $q_p(z_e)$ depends on multiple aspects, among which is the height of the building. For now, it is assumed the construction can be located anywhere in The Netherlands. Therefore, the most extreme situation is considered:

- Windregion I
- Coastal area
- Height to be determined

Wind load on structural element:

$$F_w = c_s c_d \cdot c_f \cdot q_p(z_e) \cdot A_{ref}$$
 0.8

Where:

 $\begin{array}{ll} F_w & \text{ is the wind force } [kN] \\ c_s c_d & \text{ is the structural factor } [-] \\ c_f & \text{ is the force coefficient } [-] \\ q_p(z_e) & \text{ is the peak velocity pressure } [kN/m^2] \end{array}$

A_{ref} is the reference area [m²]

A.3 Deflections and Displacements

Vertical Deflections

The maximal allowable vertical deflections can be found in Table 54. The different parameters used are displayed in Figure 139.

Table 54. Allowed vertical deflections. (Nederlands Normalisatie Instituut, 2015)

Туре	Allowed additional deflection (w ₂ + w ₃) [*ℓ _{rep}]	Allowed final deflection (w _{max}) [*ℓ _{rep}]
Floors	0.003	0.004
Floors with crack-sensitive partitions	0.002	0.004
Roof	0.004	0.004
Roof frequently walked on	0.003	0.004
Floor partitions	0.0067	0.0067



Figure 139. Vertical deflection parameters. (Nederlands Normalisatie Instituut, 2015)

Horizontal Displacements

The total maximum horizontal displacement for buildings with more than one storey under the characteristic combination of actions is:

- h/300 per storey (ui Figure 140)
- h/500 for the entire building (u Figure 140)



Figure 140. Definition of horizontal displacements. (Nederlands Normalisatie Instituut, 2015)

Gradient of Floors

To prevent the forming of ponds on water draining surfaces, sufficient gradient and camber has to be incorporated in floors and roofs, draining water towards the available drains, see Figure 141. The characteristic combination of actions should be used in which the imposed load or snow load is leading in the variable actions.



Figure 141. Required gradient as a function of deflection to prevent ponding. (Nederlands Normalisatie Instituut, 2015)

B.

Car Park Circulation Layouts

B.1 Flat Decks

Flat deck car parks are normally built in multiples of a bin width and have an adaptable layout (The Institution of Structural Engineers, 2011). Overall, three different types of flat deck layouts can be distinguished, based on the ramp system used: internal ramps, half external ramps and external ramps. Generally speaking, internal ramps result in low dynamic and static efficiency. Half external ramps cost extra, but reduce the amount of bays used to complete the circulation route, boosting dynamic efficiency. In small car parks, half external ramps may be uneconomic, but with larger car parks, its static efficiency increases. External ramps function independently form the car park and are often used in large car park facilities, providing high dynamic efficiency and for large car parks reasonable static efficiency. (Hill et al., 2005)

The following flat deck layouts can be distinguished:

Internal ramps running across the bins

In a layout with internal ramps running across the bins, all turns are in the same direction with no single turn greater than 90° , recirculation is simple, pedestrians have flat access between adjacent bins and the outflow route is reasonably rapid. Having the internal ramps running across the bins however has a significant impact on the dimensions of the car park, since the ramps have a length around 25 metres, resulting in a minimum width of three bins, see Figure 142. To keep the slope of the ramps acceptable, floor heights should be minimized. Internal ramps also lead to dead ends and the traffic flow pattern has many conflict points, both resulting in a low circulation efficiency. Both static and dynamic efficiency is low, as many bays have to be passed twice to search each deck level. The static efficiency lies around 24 m² per car for a deck with 112 bays. (Hill et al., 2005)



Figure 142. Left: One-way-flow with combined two-way ramps. Right: One-way-flow with side-by-side ramps (scissors type). (Hill et al., 2005)

Internal ramps running parallel with the traffic aisles

Having internal ramps running parallel with the traffic aisles provides simple recirculation capability, flat access for pedestrians between adjacent bins and a reasonably rapid outflow route, like with ramps running across the bins. The different layout of the ramps however eliminates dead ends and allows for any given story height with desired slope of the ramps. This layout can result in potential conflict between traffic and when the ramps are located on the sides of the car park, turns in two directions may follow each other, see Figure 143. The circulation efficiency however is low with a minimum of 44 bays per deck to complete the circulation route. This type of layout should only be used in car parks with large decks, which impacts the static efficiency. For a small layout, the static efficiency drops to around 28 m² per car, which is poor. (Hill et al., 2005)



Figure 143. Left: One-way-flow with two-way-flow ramps. Right: One-way-flow with edge ramps. (Hill et al., 2005)

Half external ramps

Different external ramp types exist, half spiral with one-way-flow and different configurations of straight ramps with one-way-flow. All of these types of ramps can be used with a flat parking deck layout. The choice for a certain ramp is more subjective, with spiral often seen as a more attractive, but more expensive option. Using half external ramps, both flow routes can be rapid or extended and the flow routes can be located side-by-side or separated. This layout provides good recirculation capability and the parking decks are clear of ramp obstructions. The circulation route can be kept extremely short with eight bays when the ramps are located on the side of an aisle, see Figure 144, and just four bays when located at the end of an aisle. In these layouts, both inflow and outflow routes combine with the aisle traffic at each parking level, but this is normally not a serious matter. If desired, an extra aisle as part of the ramp system can be added, resulting in an external ramp type at the cost of an extra 10 metres in width. Dynamic efficiency is good (up to 1100 vehicles per hour or 1500 with the full external option), especially when variable message sign systems are incorporated, eliminating the need to search each floor.



Figure 144. Left: Half spiral with one-way-flow. Right: Straight ramps with one-way-flow (side located) (Hill et al., 2005)

Circular external ramps

Circular external ramps can accommodate minimal width sites, with a minimal width of just 16 metres for the parking deck and 19 metres for the circular ramp (one-way traffic flow), see Figure 145. Generally speaking, they are not popular with the public, which has mostly to do with the minimum diameter. On the in- and outflow routes, decks can be bypassed, resulting in a high dynamic efficiency through the use of variable message signs. The traffic circulation route is vertical rather than horizontal. The static efficiency of the bays is high, but in total it can vary dramatically, depending on the ratio of the size of the parking deck and the size of the ramps. With 56 bays per deck, the static efficiency lies around 28 m² per car, which is average. For 100 bays, this increases to 24 m² per car, which is good for this type of layout. Using two-way traffic flow ramps results in a crossover condition at the entry of each parking deck. Furthermore, the diameter of the circular ramps increases to 30 metres. Having one structure for both flow routes can prove beneficial in site utilisation. (Hill et al., 2005)



Figure 145. Left: One-way-flow between end ramps. Middle: Two-way-flow with a single two-way-flow ramp. Right: Full circular ramp with two-way-flow. (Hill et al., 2005)

Storey height, straight external ramps

Storey height, straight ramps are another form of external ramps, creating a vertical circulation route, unobstructed by other traffic, apart from joining or leaving the ramp system. It has a good dynamic capacity of 1480 vehicles per hour, which can be optimized with a variable message sign system. As with other external ramp systems, just four bays per deck are required to complete the circulation route. A limiting factor in this layout is the length of the car park, since for each level climbed or descended, roughly 14 bays of length are required, see Figure 146. The total length of the car park therefore sets a limit on the number of levels possible. It is possible to combine both traffic flows on one side, saving 5 metres in width, but this will result in conflicts joining and exiting the ramp system. (Hill et al., 2005)



Figure 146. Storey height, straight ramps (Hill et al., 2005)

B.2 Split-level Decks

Split-level deck car parks generally have a good static and dynamic efficiency and are easy to drive around. They can accommodate large capacities with rapid in- and outflow routes, with most variants using one-way aisles, allowing any angle of parking. These layouts with their internal ramp systems are very compact and have a high static efficiency, most varying around 21 m² per car. The shortest variant can be constructed with a length of just 24 metres, resulting in a static efficiency of 26.75 m² per car. Except for the variants which are three or more bins wide, for drivers all turns are in the same direction and no single turn is greater than 90°. (Hill et al., 2005)

The following split-level deck layouts can be distinguished:

One-way-flow with side-by-side ramps (scissors type)

This layout has a full search plan, with its inflow route passing all bays, resulting in a high circulation efficiency. Unfortunately, the outflow route also passes all bays, resulting in a high degree of conflicts, see Figure 147. Recirculation is simple, but narrow width ramps are not popular and reduce the dynamic efficiency. This layout is a very compact option with a minimal length of 24 metres. Its static efficiency amounts to 21.06 m² per car for a length of 28 bays and 26.75 m² per car for its minimal length of 24 metres. (Hill et al., 2005; The Institution of Structural Engineers, 2011)



Figure 147. One-way-flow with side-by-side ramps (scissors type) (Hill et al., 2005)

One-way traffic flow with an excluded rapid outflow route

Passing 80% of the bays on the inflow route, this layout has a reasonable search plan. The location of the ramps results in a rapid outflow route. Inand outflow routes are separated, minimizing conflicts, see Figure 148. Also, recirculation is simple. For smaller car parks (<500 spaces), the internal rams can be combined for compatibility of the structure. This layout can be used for large car parks with up to 1100 spaces (for short to medium stay), if rapid inflow routes are incorporated. The static efficiency amounts to 21.84 m² per car for a total of 48 spaces per split-level deck. (Hill et al., 2005; The Institution of Structural Engineers, 2011)



Figure 148. One-way traffic flow with an excluded rapid outflow route (Hill et al., 2005)
One-way traffic flow with an included rapid outflow route

This layout has a highly efficient inflow circulation, passing all bays, but also a rapid outflow route. The result is a reasonable search plan with simple recirculation options. A disadvantage is the combined use of aisles on the in- and outflow routes, which can result in congestion and conflicts, see Figure 149. This layout is suitable for car parks with up to 400 bays (short or medium stay). The static efficiency amounts to 21.84 m² per car for a total of 48 spaces per split-level deck. (Hill et al., 2005)



Figure 149. One-way traffic flow with an included rapid outflow route (Hill et al., 2005)

Two-way-flow with 'combined' ramps

In this layout, two-way aisles are used. All bays lay directly off the main inflow route. However, in opposite direction, this is also the outflow route. The two-way traffic flow in this layout is dynamically less efficient, compared to one-way flows and makes recirculation very difficult, see Figure 150. This layout can be very compact (as small as 24 metres in length). As a result, this layout is only suitable for small long stay car parks with up to 300 spaces. The static efficiency amounts to 21.06 m² per car for a length of 28 bays and rises to 26.75 m² per car in its most compact form. (Hill et al., 2005)



Figure 150. Two-way-flow with 'combined' ramps (Hill, Rhodes, Vollar, & Whapples, 2005)

Three bins or more wide

Layouts with three or more bins in width can also be used. These layouts generally have a poor search pattern. Two thirds of the bays are passed both on the in- and outflow routes, see Figure 151. Entering and exiting the ramps results in conflicts between drivers. Some variants use two-way traffic aisles, requiring the concerned bin to be wider compared to the others. Recirculation in these layouts is easy and the static efficiency is around 21 m² per car for a length of 28 bays. (Hill et al., 2005)



Figure 151. Combined one-way-flows, three bins or more wide (Hill et al., 2005)

B.3 Sloped Parking Decks

Sloped parking decks switch the ways parking decks and ramps are constructed. They use sloped parking decks with a limited slope and flat ramps. The slope of the parking deck is limited (maximum 5%) by the effect of gravity on opening and closing of doors. Also for disabled drivers in wheelchairs and the use of shopping cars, sloped parking decks are a disadvantage. These layouts generally have good static efficiency between 20 and 22 m² per car. The dynamic efficiency largely depends on the variant and whether it has a rapid in- and/or outflow route. Sloped parking decks can also be combined with one of the normally sloped decks replaced by a flat deck to compensate some of the disadvantages of a sloped parking deck. This however results in longer parking decks, greater than 72 metres. (Hill et al., 2005)

The following sloped parking deck layouts can be distinguished:

Single helix

According to Stuart (2007), a single helix layout is the simplest and cheapest configuration of all different types of layouts. On the inflow route, it passes all of the bays and only twelve bays per deck are required to complete the circulation route. At each end, flat access for pedestrians is provided. By varying deck length and slope, different storey heights can be incorporated. A minimum overall length of roughly 43 metres is required as a result of the maximum parking slope. The two-way traffic flows are less efficient than one-way traffic flows for both static and dynamic efficiency. The single helix layout doesn't have a rapid outflow route nor is there recirculation ability, which can result in congestion during busy periods. This layout is suitable for up to 300 spaces with a good static efficiency of 22.3 m² per car for a deck length of 28 bays. In its shortest configuration (18 bays long), this increases to 23.9 m² per car. (Hill et al., 2005)

Some disadvantages of the single helix layout can be improved by adding a half external ramp as rapid outflow route, see Figure 152. This increases dynamic capacity, reduces congestion and provides full recirculation capabilities, making it suitable up to 600 spaces. The external ramp requires extra space, however one-way-flow reduces required aisle width. As a result, from a length of 52 metres onwards, the extra space required by the ramp is compensated by reduction in aisle width. Static efficiency is good with 22.07 m² per car for a deck length of 28.5 bays. (Hill et al., 2005)



Figure 152. Left: Two-way-flow single helix. Right: One-way-flow single helix rapid outflow route. (Hill et al., 2005)

Double helix

Several types of double helix layouts can be distinguished, but in general the layout consists out of double sloping floor ramps with one-way traffic, providing simple recirculation capability. The inflow route slopes upwards and the outflow route downwards, both passing half of the bays, resulting in a poor search pattern, lacking a rapid in- or outflow route. This together with possible conflicts between in- and outflow traffic entering the central access-way can result in congestion. The circulation route can be completed with 12 to 16 bays per deck. Storey heights are accompanied by varying deck slope and length. The minimal required building length is 72 metres (30 bays), resulting in a good static efficiency of 21.1 m² per car. These type of layouts are generally preferred for situations, which involve everyday users (Stuart, 2007). The static efficiency can even be improved up to 20.06 m² per car by eliminating the central cross-over, see Figure 153, however this aggravates the disadvantages of circulation and therefore makes the layout unusable for short, medium and long stay car parks. (Hill et al., 2005; Stuart, 2007; The Institution of Structural Engineers, 2011)



Figure 153. Left: Double helix, end connected with one-way-flow on the central access-way. Right: Interlocking double helix, with one-way-flows. (Hill et al., 2005)

Combined helix

The combined helix layout is three bins wide and combines two-way-flows in the middle aisle and oneway-flow on the outward aisles, see Figure 154. This results in simple circulation and recirculation ability with all turns in the same direction and not bigger than 90°, but lacks a rapid outflow route. On both the in- and outflow route, two thirds of the bays are passed and the central aisle has to be driven through twice. The two-way-flow in the middle aisle is also less efficient both statically and dynamically and can lead to congestion, making this layout not suitable for short and medium stay. The static efficiency is average with 22.9 m² per car for a deck length of 16 bays (80 bays per deck). By replacing one of the sloped decks with a flat deck, it is possible to create only one-way-flows. This increases the static efficiency to 21.13 m² per car, which is good. However, the long circulation route and driver conflicts remain the same.



Figure 154. Left: Combined helix, side connected with one- and two-way-flows. Right: Combined helix, side connected with one-way-flow. (Hill et al., 2005)

Combined flat and sloping deck layouts with internal cross-ramps

Combining a vertical circulation module with sloped floors with an extended flat deck results in a car park layout with a significant improvement in cross-deck accessibility for pedestrians. It has good static and dynamic efficiency. In the layout, a rapid inflow route can be incorporated and 80% of the bays can be searched efficiently on the inflow circuit. Furthermore, all turns are in the same direction and no bigger than 90 ° and it has simple recirculation ability, see Figure 155. The layout requires a minimum length of 36 metres (15 bays wide). With 96 bays per deck, the static efficiency is good with 21.84 m² per car.



Figure 155. One-way-flow with two one-way-flow ramps. (Hill et al., 2005)

C.1 Detailed Overview of Timber Joints

C.1.1 Carpentry Joints

Scarf Joint

In a scarf joint, two members are connected to each other by removing material from both members to result in a joint with the thickness of the thickest member, see Figure 156. As material is removed from the members, the crosssection is considerably weakened, resulting in a relatively low load-carrying capacity. Nevertheless, scarf joints are considered to be the strongest form of unglued member lengthening (Harte & Dietsch, 2015). Scarf joints are often applied in older historical buildings in roofs and frames (Herzog et al., 2004).

Lap Joint

Two sorts of lapped joints can be distinguished. The first is the full-lap joint, in which two members are placed over each other and connected to each other by a pin. The second type is a half-lap joint with or without cogs, see Figure 157. In this case, material is removed from both members to make them slide into each other, resulting in a joint with the same thickness as the thickest member in the joint (Harte & Dietsch, 2015). These connections are not used often anymore and were used in floor and roof framing (Blaß & Sandhaas, 2017).

Step Joint

Step joints or notched joints are used in timber trusses, transmitting compressive forces of inclined struts to the chords. The joints transfer compressive forces via contact to the front surface of the joint, followed by shear stress in the loaded end to the chord (Blaß & Sandhaas, 2017). Step joints can transfer relatively high compressive forces, however this may require use of a hardwood block, spreading the load, to prevent crushing of the timber perpendicular to the grain (Herzog et al., 2004). Nails, bolts or screws are used to secure the members in their position. Three different types of step joints can be distinguished: single step – notch in front (see Figure 158), notch in heel and double step. Differences between them is the location the forces are transmitted to the chord. A newer multiple step joints has been developed, which is essentially an



Figure 156. Scarf joint (Harte & Dietsch, 2015)



Figure 157. Half-lap joint (Harte & Dietsch, 2015)



Figure 158. Step joint (Harte & Dietsch, 2015)

improved double step joint, but requiring only a third of the notch depth (Blaß & Sandhaas, 2017).

Mortise and Tenon Joint

A mortise and tenon joint consist out of a hole in one element, the mortise, into which a tongue of another element is slot, the tenon, see Figure 159. As material is removed, the cross-section is weakened. These joints are used to connect individual members in floor, wall and roof constructions with angles between 45° and 90° and transfers shear forces. For the strongest result, the tenon should fit snug into the mortise. It is also possible to fasten the joint using dowels. (Blaß & Sandhaas, 2017; Harte & Dietsch, 2015)



Figure 159. Through pinned mortise and tenon joint (Harte & Dietsch, 2015)

C.1.2 Mechanical Joints

Dowel-type Fasteners

Nails & staples

Nails are the most frequently applied type of fasteners in timber construction and used in for example horizontal and vertical diaphragms and trusses. They come in various shapes and sizes and depending on the type of nail and wood, predrilling of holes might be required to prevent wood from splitting. Nails are used in single-shear joints between timber, steel and wood-based materials. They should not be axially loaded as this can result in withdrawal of the nail or pull-through of the nail head. Ringed shank nails can resist more axial loads than smooth shanked nails, but it is still limited. Nails generally exhibit ductile behaviour, which facilitates a balanced load distribution between nails grouped together. (Blaß & Sandhaas, 2017)

Staples are very similar types of fasteners to nails and have very similar load-bearing behaviour. The steel grade used in staples however is much higher and the angle between staple crown and grain direction of the wood should be monitored. Staples can be assumed as two equivalent nails, but the angle between the staple crown and grain direction should be at least 30°. They are most frequently applied in timber frame buildings as they are very fast to install using special fastener tools (Blaß & Sandhaas, 2017).

Bolts, dowels and threaded rods

Bolts are made of steel and include a head and a nut. They are placed in predrilled holes 1 mm larger than the bolt diameter, resulting in a less stiff connection. Dowels, like bolts consist out of a steel rod, but lack the head and bolt and do not require a wider predrilled hole. As a result, they are often preferred because of their visual appearance and more stiff connection. (Blaß & Sandhaas, 2017)

Dowelled joints are economic, easy to set up and excel in transferring large forces. To ensure that the different members in a joint stay together, at least one bolt is required in a dowelled connection. As a result of the shape of the dowel, dowelled connections cannot be exposed to axial loads. Bolted connections, as a result of their bolts and washers, can take up a limited axial force. Bolted and dowelled connections can be used in both timber-to-timber joints, timber and wood-based materials and steel-to-timber joints (Blaß & Sandhaas, 2017; Herzog et al., 2004).

Screws

Screws are self-tapping fasteners, which come in many variations as they are the fastest evolving type of fasteners. They can be subjected to both axial and lateral loads. As a result of the thread, large forces can be transmitted in axial direction. In comparison with dowels and nails of the same nominal diameter however, the allowed lateral loads on screws are smaller, because of the reduced diameter of threaded part. Screwed joints usually have one shear plane as they are ideal for connections between timber and steel plates, steel members or wood-based materials. They can also be used in timber-to-timber joints. Screws are also used to reinforce other type of joints. Depending on the type and diameter of the screw, predrilling might or might not be required. (Blaß & Sandhaas, 2017)



Figure 160. Left: Nails & staples. Middle: Bolts & dowel. Right: Screws. (Herzog et al., 2004)

Joints with multiple shear planes

Dowel-type fasteners are often used in joints in combination with slotted-in steel plates. Slits are sawn into the timber members into which the steel plates are fitted, after which nails, bolts or dowels are pushed through predrilled holes in the timber and steel, see Figure 161. Such joints are categorized as joints with multiple shear planes. The joints are high-performance joints, which can transfer high forces and with correct layout also bending moments. They are often used in heavy trusses and other high-performance joints. (Herzog et al., 2004)



Figure 161. Joints with slotted in steel plates

Cold-formed steel connectors

Traditional carpentry joints have mainly been replaced by coldformed steel connectors, see Figure 162. Such connectors come in all kind of forms and sizes (plates, straps, angles, brackets, anchors, hangers, etc.) and are connected to both members by nails or screws (Blaß & Sandhaas, 2017). The connectors can be used in timber-to-timber connections as well as timber-to-steel or timber-to-masonry/concrete and are fast and easy to assemble (Herzog et al., 2004).



Figure 162. Cold-formed steel joists hangar (Herzog et al., 204).

Surface-type Fasteners **Connectors**

Connectors increase the load-bearing area of a connection, therefore increasing the load that can be accommodated. As a result, instead of using multiple dowel-type fasteners, only a single connector might be required creating a perfect hinged joint. Connectors are secured by fasteners like bolts or screws, pressing the timber members together. These bolts or screws absorb the moments generated in the joints, which press the timber members apart, by tensile forces. Joints using connectors can be prefabricated requiring only the installation of bolts on the construction site, making for a very fast erection method. They can just as easily be dismantled. Typical types of connectors are split ring, shear plates and toothed plates. (Blaß & Sandhaas, 2017)

For split ring and shear plate connectors, see Figure 163 left and middle, depressions are milled around the hole for the bold. The connector is placed in this depression and the timber members are connected to each other and secured by a bolt and nut. It is also possible to use screws instead of bolts. (Blaß & Sandhaas, 2017)

Toothed plates differ with split rings and shear plates as they are simply pressed into the timber members and don't require depressions to be milled, see Figure 163 right. A hydraulic press or special high-strength bolt is used for the pressing-in process after which the final bolt or screw is placed. To prevent a gap between the members as a result of the thickness of the base plate of the toothed plate, a recess of the size and thickness of the base plate can be milled. As these plates have to be pressed into the wood, a maximum density of the wood of 500 kg/m³ is allowed for them to be used. (Blaß & Sandhaas, 2017)

Connector type fasteners transfer shear forces in a joint. Double-sided connectors are used for timberto-timber joints and single-sided connectors are used for both timber-to-timber and steel-to-timber joints. With double-sided connectors, the force from a member is transferred to the connector via embedment stresses, followed by shear resistance of the connector to the other member. In case of one-sided connectors, the force is again transferred by embedment stresses to the connector, but is then transferred to the bolt by embedment stress between the connector and bolt. The bolt transfers the force via shear resistance to the connector of the other member or directly to the steel plate. With these single-sided connectors, a tolerance of the hole diameter is present, resulting in some slip in the connection.



Figure 163. Left: double-sided split ring connector, middle: single-sided shear plate connector, right: double- and single-sided toothed-plate connector (Blaß & Sandhaas, 2017).

Punched metal plate fasteners

Joints between two pieces of wood in the same plane and with the same thickness can be made using punched metal plate fasteners, see Figure 164. These are metal plates with teeth punched in one direction and sticking out of the plate at right angles, which are attached to both sides of the members of a joint. The result is a connection, which for the size of the joint faces can transfer a significantly larger force compared with conventional nailed connections (Herzog et al., 2004). Punched metal plate connections can transfer shear and normal forces and bending moments in the plane of the connection. Forces are transferred from the timber member tot the nails and then to the steel plate. In the same way, these forces are then transferred to the connecting member. They are mainly applied in lightweight timber trusses and in other plane timber structures. For different purposes, different arrangements and shapes of punched metal plates have been developed (Blaß & Sandhaas, 2017).



Figure 164. Punched metal plate joint (Herzog et al., 2004)

C.2 Comparison of Timber Joints

	Glued joints
Stiffness	Glued joints can be considered as moment-resisting joints. The monolithic joint is very stiff.
Loadbearing capacity	The homogeneous properties of glued finger joints result in a high load bearing capacity.
Types of forces in plane	The monolithic joint allows for bending moments, normal forces and shear forces to be transferred.
Ease of assembly	The gluing process is a specialised process requiring skilled workers and machinery. It is a labour intensive type of joint.
Dismantlability for reuse	The gluing results in a permanent joint, which cannot be undone.
Fire resistance	Glued timber finger joints lack the use of steel and therefore, fire safety is a result of the properties of the wood, which can be deemed good. For glued- in rods, the steel is surrounded by wood, protecting the steel elements from fire.
Costs	The labour intensive production process is expensive.

The following tables present the motivation for the comparison of the different types of timber joints.

C_{21}	Glu	ed	loi	nte
0.2.1	. 0เน	cu,	JUI	III S

C.2.2 Carpentry Joints

	Scarf joints
Stiffness	The geometry of a scarf joint in general makes for a hinged joint.
Loadbearing capacity	As material is removed from both members, the cross-section is weakened considerably, limiting the loadbearing capacity of the joint.
Types of forces in plane	Carpentry joints rely on transfer of loads via contact and friction. Scarf joints are mainly suited to transfer shear forces.
Ease of assembly	Assembly of carpentry joints is easy as the members slot into each other. Sometimes an additional steel pin or screw is required.
Dismantlability for reuse	Disassembly of carpentry joints is easy as they are not fixed to each other at all or just by a simple steel pin or screw, which can be removed.
Fire resistance	As these connections rely on contact and friction and not on steel connectors, they have good fire resistant properties.
Costs	The removal of material of member has to be done very precisely to make for a perfect fit between the members. This requires special CNC machinery adding to the costs of these connections.

	Lap joints
Stiffness	The geometry of a lap joint in general makes for a hinged joint.
Loadbearing capacity	As material is removed from both members, the cross-section is weakened considerably, limiting the loadbearing capacity of the joint.
Types of forces in plane	Carpentry joints rely on transfer of loads via contact and friction. Lap joints can mainly transfer shear forces, but adaptation of the geometry like a dovetail crossed lap can result in a connection that is also able to transfer normal forces.
Ease of assembly	Assembly of carpentry joints is easy as the members slot into each other. Sometimes an additional steel pin or screw is required.
Dismantlability for reuse	Disassembly of carpentry joints is easy as they are not fixed to each other at all or just by a simple steel pin or screw, which can be removed.
Fire resistance	As these connections rely on contact and friction and not on steel connectors, they have good fire resistant properties.
Costs	The removal of material of member has to be done very precisely to make for a perfect fit between the members. This requires special CNC machinery adding to the costs of these connections.

	Step joints
Stiffness	The geometry of a step joint in general makes for a hinged joint.
Loadbearing capacity	Step joints can transfer relatively high compressive forces, however this may require use of a hardwood block, spreading the load, to prevent crushing of the timber perpendicular to the grain.
Types of forces in plane	Carpentry joints rely on transfer of loads via contact and friction. The main function of step joints is to transfer normal forces in chords of trusses.
Ease of assembly	Assembly of carpentry joints is easy as the members slot into each other. Sometimes an additional steel pin or screw is required.
Dismantlability for reuse	Disassembly of carpentry joints is easy as they are not fixed to each other at all or just by a simple steel pin or screw, which can be removed.
Fire resistance	As these connections rely on contact and friction and not on steel connectors, they have good fire resistant properties.
Costs	The removal of material of member has to be done very precisely to make for a perfect fit between the members. This requires special CNC machinery adding to the costs of these connections.

	Mortise and tenon joints
Stiffness	Mortise and tenon joints are semi-rigid, but often have significant moment-resisting capacity.
Loadbearing capacity	As material is removed from the members, the cross-section is weakened, limiting the loadbearing capacity of the joint.
Types of forces in plane	Carpentry joints rely on transfer of loads via contact and friction. As the mortise and tenon joints snugly fit into each other, a clamping effect is formed allowing for the transfer of bending moments, normal forces and shear forces.
Ease of assembly	Assembly of carpentry joints is easy as the members slot into each other. Sometimes an additional steel pin or screw is required.
Dismantlability for reuse	Disassembly of carpentry joints is easy as they are not fixed to each other at all or just by a simple steel pin or screw, which can be removed.
Fire resistance	As these connections rely on contact and friction and not on steel connectors, they have good fire resistant properties.
Costs	The removal of material of member has to be done very precisely to make for a perfect fit between the members. This requires special CNC machinery adding to the costs of these connections.

C.2.3 Mechanical Joints

Dowel-type Fasteners

	Nails & staples
Stiffness	Nails and staples exhibit ductile behaviour, facilitating a balanced load distribution between groups of nails or staples. It results in a semi-rigid connection.
Loadbearing capacity	Nails or staples grouped together can transfer medium level forces. It is however important to note they cannot transfer forces axially as this can result in withdrawal of the nail or staple.
Types of forces in plane	Groups of nails and staples are most suited to transfer normal and shear forces.
Ease of assembly	Nails and staples can easily be installed with simple machinery.
Dismantlability for reuse	Nails and staples can be removed, however as they have flat heads, they can be difficult to grasp. When painted over, it can also be difficult to locate them.
Fire resistance	The largest part of the nails or staples is surrounded by wood, protecting them from fire. The heads however are vulnerable to fire and might require protection.
Costs	Applying nails and staples in connections is easy and cheap.

	Bolts, dowels & threaded rods
Stiffness	Bolt or dowel groups result in semi-rigid connections. As dowels do not require wider predrilled holes, they can result in more rigid connections.
Loadbearing capacity	Dowelled connections excel in transferring large forces.
Types of forces in plane	Groups of bolts or dowels can transfer bending moments, normal forces and shear forces.
Ease of assembly	Bolts and dowels are fast and easy to install.
Dismantlability for reuse	Bolts and dowels remain visible and are easy to remove.
Fire resistance	The largest part of the bolts or dowels is surrounded by wood, protecting them from fire. The ends however are vulnerable to fire and might require protection.
Costs	Dowelled joints are economic.

	Screws
Stiffness	Screws exhibit ductile behaviour, facilitating a balanced load distribution between groups of nails or staples. It results in a semi-rigid connection.
Loadbearing capacity	Compared with dowels and nails with the same diameter, screws can transfer smaller forces. As a result of their threads, screws can also transfer axial forces.
Types of forces in plane	Groups of screws are most suited to transfer normal and shear forces.
Ease of assembly	Screws can easily be installed with simple machinery.
Dismantlability for reuse	Screws are easy to remove. Painting however can hide them from sight.
Fire resistance	The largest part of the screws is surrounded by wood, protecting them from fire. The heads however are vulnerable to fire and might require protection.
Costs	Applying screws in connections is easy and cheap.

	Combination with steel plates or connectors
Stiffness	The combination of steel plates with groups of bolts or dowels can result in a semi-rigid connection, but much towards rigid.
Loadbearing capacity	Dowelled connections in combination with steel plates excel in transferring large forces.
Types of forces in plane	Groups of bolts or dowels in combination with steel plates can transfer bending moments, normal forces and shear forces.
Ease of assembly	Bolts and dowels are fast and easy to install.
Dismantlability for reuse	Bolts and dowels remain visible and are easy to remove.
Fire resistance	The steel plates can be exposed to fire conditions. It is also possible to embed the plates into the wood, which can protect the plates from fire conditions.
Costs	The combination of dowelled connections with steel plates can result in economic joints, which can transfer high forces.

Surface-type Fasteners

	Connectors
Stiffness	Connectors with bolt or dowel groups result in semi-rigid connections. The connectors improve the stiffness of the plane between both members.
Loadbearing capacity	The connectors increase the load-bearing area of a connection, increasing the load that can be accommodated.
Types of forces in plane	Groups of connectors with bolts or dowels can transfer bending moments, normal forces and shear forces.
Ease of assembly	Installing connectors might require some additional steps to make recesses for the connectors to fit in. Afterwards, installation is very easy.
Dismantlability for reuse	Bolts and dowels remain visible and are easy to remove.
Fire resistance	The connectors are well protected from fire conditions. The ends of bolts or dowels however might require protection.
Costs	The installation of connectors is easy and cheap.

	Punched metal plate fasteners
Stiffness	Due to the high number of metal pins, the joint can transfer significantly higher loads compared with traditional nailed connections.
Loadbearing capacity	The punched metal plates can transfer medium forces.
Types of forces in plane	The punched metal plates result in a semi-rigid connection, which can transfer bending moments, normal forces and shear forces.
Ease of assembly	The punched metal plates can be easily punched into the timber members.
Dismantlability for reuse	The punched metal plates can simply be peeled of the joints.
Fire resistance	The metal plates are exposed to fire conditions.
Costs	Installation is easy and cheap.

D. Calculations Technosoft

D.1 Cone Penetration Tests

Sweco Nederland BV Blad: 1 Technosoft Funderingen op Staal release 6.70a 7 okt 2022 Project : Shallow Foundation for Timber Columns Onderdeel : Isolated Footing SONDERINGSGEGEVENS ALGEMEEN: Loc01_CPT00000052457 Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maajveld

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc01_CPT00000052457



Sweco Nederland BV						
Technosoft Funderingen o	op Staal release 6.70a	7 okt 2022				
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing					

SONDERINGSGEGEVENS ALGEMEEN: Loc03_CPT00000074906

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc03_CPT00000074906



Onderdeel

SONDERINGSGEGEVENS ALGEMEEN: Loc04_CPT00000055844

: Isolated Footing

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc04_CPT00000055844



Sweco Nederland BV						
Technosoft Funderingen o	p Staal release 6.70a	7 okt 2022				
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing					

SONDERINGSGEGEVENS ALGEMEEN: Loc06_CPT000000170226

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc06_CPT000000170226



SONDERINGSGEGEVENS ALGEMEEN: Loc08_CPT00000062979

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc08_CPT00000062979



Sweco Nederland BV		Blad: 6
Technosoft Funderingen o	p Staal release 6.70a	7 okt 2022
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing	

SONDERINGSGEGEVENS ALGEMEEN: Loc12_CPT00000077437

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc12_CPT00000077437



Onderdeel

SONDERINGSGEGEVENS ALGEMEEN: Loc02_CPT00000003023

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00 Hoogte maaiveld [m]

SONDERINGSGEGEVENS GRAFIEK: Loc02_CPT00000003023



Sweco Nederland BV	Sweco Nederland BV						
Technosoft Funderingen c	p Staal release 6.70a	7 okt 2022					
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing						

SONDERINGSGEGEVENS ALGEMEEN: Loc05_CPT00000161652

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc05_CPT00000161652



Sweco Nederland BV		Blad: 9		
Technosoft Funderingen	op Staal release 6.70a	7 okt 2022		
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing			

SONDERINGSGEGEVENS ALGEMEEN: Loc07CPT00000004530

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc07CPT00000004530



Sweco Nederland BV	Blad: 1		
Technosoft Funderingen c	p Staal release 6.70a	7 okt 2022	
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing		

SONDERINGSGEGEVENS ALGEMEEN: Loc09_CPT00000011161

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc09_CPT00000011161



Onderdeel

SONDERINGSGEGEVENS ALGEMEEN: Loc10_CPT00000004377

: Isolated Footing

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc10_CPT00000004377



Sweco Nederland BV	Blad: 12	
Technosoft Funderingen o	pp Staal release 6.70a	7 okt 2022
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing	

SONDERINGSGEGEVENS ALGEMEEN: Loc11_CPT00000012135

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld Hoogte maaiveld [m] : 0.00

SONDERINGSGEGEVENS GRAFIEK: Loc11_CPT00000012135



D.2 Output Technosoft Shallow Foundation

Sweco Nederl	and BV						Blad: 1
Technosoft Fun	deringen op	Staal	release	6.70a			7 okt 2022
Project	:	Shallo	w Founda	tion for	Timber Co	lumns	
Onderdeel	:	Isolat	ed Footin	ng			
ALGEMENE GEG	EVENS						
Project	:	Shallo	w Foundat	tion for	Timber Co	lumns	
Onderdeel		Isolat	ed Footi	na na	TIMDEL OC	1 dilitio	
Eenheden		[kN][m	l[MPa][g	radenl	tenzii ar	ders vermelo	h
Datum		05-10-	2022		concep an	GOLD FOLMOL	^
Referentienive	au (RN)	Maaive	ld				
Referentieperi	ode :	50 jaa	r				
Restand	•		rs\nlnahi	1\OneDri	ve - Swecc		
Debeana	•	AB\Doc	umenten\;	Afstuder	en\Technos	oft\	
		Draage	anaciteii	t Funder	ing on Sta	al fsw	
		Draago	apaoroor	e ranaer	ING OF DOD		
Toegepaste n	ormen volg	ens E	urocode	met Ne	derlands	e NB	
Beton	NEN-EN 1992	-1-1:2	011(nl)	C2	/A1:2015(r	1) NB:201	5(nl)
Geotechniek	EN 1997-1:2	004		AC	:2009		
	NEN-EN 1997	-1:200	5	C1	+A1:2013	NB:201	5
	NEN 9997-1:	2016		C2	:2017		
PROFIELGEGEV	ENS Isolat	ed Fo	oting 2	000x200	0 x 200		
Туре		:	Poer				
			Links		Rechts		
Breedte	min	[mm] :	1000		1000		
	max	[mm] :	1000		1000		
	stap	[mm] :	100		100		
Lengte	voor	[mm] :	1000				
	achter	[mm] :	1000	De po	er is vier	kant.	
Hoogte		[mm] :	200				
Opstorting	breedte	[mm] :	300				
	lengte	[mm] :	300				
DET ACUTNCCEC	EVENS Lood	250					
Permanent	EVENS LOAD	. 330					
Nr. Omschrijv	ing	Type	Richting	Waarde	AfstandX	AfstandY Afs	standZ
2	2		2	[kN,m]	[m]	[m]	[m]
1 Self-Weig	ht	F/q	Z	-350.00	0.00	0.00	_
Extra lasten t	.g.v. eigeng	ewicht	poer en	opstort	staan bij	de rekenge	gevens.
BELASTINGGEG	EVENS Load	1000					
Permanent		-	D 1 1 1				
Nr. Omschrijv	ıng	'l'ype	Kichting	Waarde	AistandX	AISTANDY AF:	standZ
				[kN,m]	[m]	[m]	[m]
1 Self-Weig	ht	F/q	Z	-1000.00	0.00	0.00	
Extra lasten t	.g.v. eigeng	ewicht	poer en	opstort	staan bij	de rekenge	gevens.

GRONDSOORTEN

Nr	Naam	γ	Ysat	φ'	c'	C ₁₁	C _c /(1+e ₀) C _a	eo	
		[kN/m ³][kN/m³]	[°]	[kPa]	[kPa]	[-]	[-]	[-]	
1	Zand - Schoon - Los	17.0	19.0	30.0	-	-	0.0115	0.0000	0.83	
2	Klei - Schoon - Matig	17.0	17.0	17.5	5.0	50.0	0.1533	0.0061	1.36	
3	Zand - Zwak siltig - K	(18.0	20.0	27.0	-	-	0.0051	0.0000	0.65	
4	Zand - Schoon - Vast	19.0	21.0	35.0	-	-	0.0023	0.0000	0.50	
5	Klei - Schoon - Vast	19.0	19.0	17.5	13.0	100.0	0.0920	0.0037	0.83	
6	Veen - Matig voorbelas	s 12.0	12.0	15.0	2.5	20.0	0.3067	0.0153	4.90	
7	Leem - Zwak zandig - M	1 20.0	20.0	27.5	1.0	100.0	0.0511	0.0020	0.65	
8	Klei - Sterk zandig	18.0	18.0	27.5	0.0	0.0	0.0920	0.0037	1.06	
9	Veen - Niet voorbelast	10.0	10.0	15.0	1.0	10.0	0.4600	0.0230	15.50	
10	Grind - Sterk siltig -	19.0	21.0	32.5	-	-	0.0038	0.0000	0.50	

Blad: 2

7 okt 2022

Sweco Nederland BV

Technosoft Funderingen op Staal release 6.70a

Project Onderdeel : Shallow Foundation for Timber Columns : Isolated Footing

GRONDSOORTEN

Nr	Naam		γ	Ysat	φ'	с'	C ₁₁	$C_{c} / (1 + e_{0})$) Ca	e ₀	
		[kì	J/m ³][kN/m³]	[°]	[kPa]	[kPa]	[-]	[-]	[-]	
11	Grind - Zwak siltig -		19.0	21.0	37.5	-	-	0.0019	0.0000	0.50	
12	Klei - Organisch - Mat	ig	15.0	15.0	15.0	0.0	25.0	0.2300	0.0115	2.30	
13	Zand - Sterk siltig -		18.0	20.0	25.0	-	-	0.0115	0.0000	0.65	
14	Grind - Zwak siltig -		18.0	20.0	35.0	-	-	0.0023	0.0000	0.65	
15	Zand - Schoon - Matig		18.0	20.0	32.5	-	-	0.0038	0.0000	0.65	
16	Grind - Zwak siltig -		17.0	19.0	32.5	-	-	0.0046	0.0000	0.83	
17	Klei - Zwak zandig - S	S	15.0	15.0	22.5	0.0	40.0	0.2300	0.0092	2.30	
18	Leem - Zwak zandig - V	· · ·	21.0	21.0	27.5	2.5	200.0	0.0329	0.0013	0.50	
19	Leem - Sterk zandig		19.0	19.0	27.5	0.0	50.0	0.0511	0.0020	0.83	
20	Klei - Zwak zandig - M	1	18.0	18.0	22.5	5.0	80.0	0.1150	0.0046	1.06	
21	Klei - Zwak zandig - V	· · ·	20.0	20.0	22.5	13.0	120.0	0.0767	0.0031	0.65	
22	Grind - Sterk siltig -	·	18.0	20.0	30.0	-	-	0.0058	0.0000	0.65	

Sweco Nederland BV		Blad: 3
Technosoft Funderingen op	Staal release 6.70a	7 okt 2022
Project : Onderdeel :	Shallow Foundation for Timber Columns Isolated Footing	
REKENGEGEVENS Case 1	(d = -0.80m, F=350)	
Profiel : Belasting : Bodemprofielen : : : : :	Isolated Footing 2000x2000x200 Load 350 Loc01_CPT000000052457, Loc03_CPT000000074906 Loc04_CPT00000055844, Loc06_CPT000000170226 Loc08_CPT000000062979, Loc12_CPT00000007437 Loc02_CPT00000000323, Loc05_CPT000000161652 Loc07_CPT000000004530, Loc09_CPT000000011161 Loc10_CPT000000004377, Loc11_CPT000000012135	
Wapening : Niveau onderkant fnd[m] : Grondwaterniveau [m] : Opstort :	Wapening 1 -0.80 Niveau bovenkant [m] : 0.30 -0.20 0.80 Zand - Schoon - Los	
Materiaalfactoren	gunstig ongunstig	
$\begin{array}{lll} \gamma_{\gamma} & \mbox{gewicht grond} & : \\ \gamma_{\phi}, & \mbox{inwendige wrijving:} \\ \gamma_{c}, & \mbox{cohesie} & : \\ \gamma_{cu} & \mbox{ongedr. schuifst. :} \\ \gamma_{\gamma} & \mbox{gewicht grond BGT :} \end{array}$	1.10 1.00 1.15 1.60 1.35 1.00	
Belastingfactoren o	ngunstig gunstig Ψ	
Permanent : Variabel : Grond :	1.20 0.90 1.50 0.00 0.40 0.90	
Extra belastingen t.g.v. B-tot B-li B-re L H [m] [m] [m] [m] [m] [m]	eigengewicht poer en opstort Omschrijving Type Rich- Waarde AfstX Afs ting [kN] [m] [tY AfstZ m] [m]
2.00 1.00 1.00 2.00 0.20 0.30 0.15 0.15 0.30 0.90 2.00 1.00 1.00 2.00 0.60 0.30 0.15 0.15 0.30 0.60	E.G poer, plaat F Z 20.00 0.00 0.0 + opstorting F Z 2.02 0.00 0.0 E.G opstort (droog) F Z 40.80 0.00 0.0 - tpv opstorting F Z -0.92 0.00 0.0	0 – 0 – 0 – 0 –
REKENGEGEVENS Case 2	(d = -2.00m, F=350)	
Profiel : Belasting : Bodemprofielen : : : : : : : : : : : : : : : : : : :	Isolated Footing 2000x2000x200 Load 350 Loc01_CPT000000052457, Loc03_CPT000000074906 Loc04_CPT000000055844, Loc06_CPT000000170226 Loc08_CPT000000062979, Loc12_CPT000000017437 Loc02_CPT000000003023, Loc05_CPT000000161652 Loc07_CPT000000004530, Loc09_CPT000000011161 Loc10_CPT000000004377, Loc11_CPT000000012135 Wapening 1 -2.00 Niveau bovenkant [m] : 0.30 -0.20	
Opstort :	2.00 Zand - Schoon - Los	
Materiaalfactoren	gunstig ongunstig	
$\begin{array}{llllllllllllllllllllllllllllllllllll$	1.10 1.00 1.15 1.60 1.35 1.00	
Belastingfactoren o	ngunstig gunstig Ψ	
Permanent : Variabel : Grond :	1.20 0.90 1.50 0.00 0.40 0.90	

Sweco	Nederland BV										Bla	ad: 4
Techno	soft Funderingen d	op	Staal r	elease (5.70a					7	okt	2022
Projec Onderd	t eel	:	Shallow Isolate	Foundat d Footir	ion for	Timbe	er Col	Lumns	3			
Extra	belastingen t.g.v		eigengew	icht poe	er en ops	stort						
B-tot	B-li B-re L	Н	Omschr	ijving	Type	e Rich	i- Waa	arde	AfstX	AfstY	Afst	Z
[m]	[m] [m] [m] [r	n]				ting	r ([kN]	[m]	[m]	[n	ι]
2.00	1.00 1.00 2.00 0.2	20	E.G poe	r, plaat		ΕZ	20.	.00	0.00	0.00	-	
0.30	0.15 0.15 0.30 2.3	10		+ opsto	orting	FΖ	4.	72	0.00	0.00	-	
2.00	1.00 1.00 2.00 1.8	30	E.G ops	tort (dı	(poor	FΖ	122.	40	0.00	0.00	-	
0.30	0.15 0.15 0.30 1.8	30	- tpv	opstort	ing	FΖ	-2.	75	0.00	0.00	-	
REKEN	GEGEVENS Case 3		(d = -0)	.80m, 1	F=1000)							
Profie	1	:	Isolate	d Footir	ng 2000x2	2000x2	200					
Belast	ing	:	Load 10	00								
Bodemp	rofielen	:	Loc01_C	PT000000	052457,	Loc03	CPTC	0000	000749	06		
		:	Loc04_C	PT000000	055844,	Loc06	CPTC	0000	001702	26		
		:	Loc08_C	PT000000	062979,	Loc12	CPTC	0000	000774	37		
		:	Loc02_C	PT000000	003023,	Loc05	_CPTC	0000	001616	52		
		:	Loc07_C	PT000000	004530,	Loc09	_CPTC	0000	000111	61		
		:	Loc10_C	PT000000	0004377,	Loc11	_CPTC	0000	000121	35		
Wapeni	ng	:	Wapenin	g 1						0 00		
Niveau	onderkant ind[m]	:	-0.80		Niveau	bover	ikant	[m]	:	0.30		
Grondw	aterniveau [m]	:	-0.20		a 1	-						
Opstor	t	:	0.80	Zand -	Schoon -	- Los						
Materi	aalfactoren		gunstig	ongur	nstig							
γ.,	gewicht grond	:	1.10		1.00							
Yal	inwendige wrijving	a :	1.15									
γ _γ	cohesie	:	1.60									
γ	ongedr. schuifst.	:	1.35									
γ _γ	gewicht grond BGT	:	1.00									
, D-1+	·)T(
Belast	ingractoren	01	ngunstig	gur	istig		Ψ					
Perman	ent	:	1.20		0.90							
Variab	el	:	1.50		0.00	C	.40					
Grond		:			0.90							
Extra	belastingen t.g.v	. e	eigengew	icht poe	er en ops	stort						_

]	B-tot	B-li	B-re	L	Н	Omschrijving	Type	Rich-	Waarde	AfstX	AfstY	AfstZ	
	[m]	[m]	[m]	[m]	[m]			ting	[kN]	[m]	[m]	[m]	
	2.00	1.00	1.00	2.00	0.20	E.G poer, plaat		FΖ	20.00	0.00	0.00	-	
	0.30	0.15	0.15	0.30	0.90	+ opstorti	ng	FΖ	2.02	0.00	0.00	-	
	2.00	1.00	1.00	2.00	0.60	E.G opstort (droog))	FΖ	40.80	0.00	0.00	-	
	0.30	0.15	0.15	0.30	0.60	- tpv opstorting		FΖ	-0.92	0.00	0.00	-	

Sweco Nederland BV	Blad: 5
Technosoft Funderingen op Staal release 6.70a	7 okt 2022
Project : Shallow Foundation for Timber Columns Onderdeel : Isolated Footing	
REKENGEGEVENS Case 4 ($d = -2.00m$, F=1000)	
Profiel : Isolated Footing 2000x2000x200 Belasting : Load 1000 Bodemprofielen : Loc01_CPT000000052457, Loc03_CPT0000000 : Loc04_CPT000000055844, Loc06_CPT0000000 : Loc02_CPT000000062979, Loc12_CPT0000000 : Loc02_CPT00000004530, Loc09_CPT0000000 : Loc01_CCPT00000004530, Loc09_CPT0000000)74906 170226)77437 61652 011161 02135
Wapening : Wapening 1 Niveau onderkant fnd[m]: -2.00 Niveau bovenkant [m]: Grondwaterniveau [m]: -0.20 Opstort : 2.00 Zand - Schoon - Los	0.30
$\begin{array}{llllllllllllllllllllllllllllllllllll$	
Belastingfactoren ongunstig gunstig Ψ	
Permanent : 1.20 0.90 Variabel : 1.50 0.00 0.40 Grond : 0.90 0.90 0.90	
Extra belastingen t.g.v. eigengewicht poer en opstort B-tot B-li B-re L H Omschrijving Type Rich- Waarde Af	stX AfstY AfstZ

DCOC		DIC		11	oncontrol interview	pe niton	maarac	1110 011	TILOCI	111000	
[m]	[m]	[m]	[m]	[m]		ting	[kN]	[m]	[m]	[m]	
2.00	1.00	1.00	2.00	0.20	E.G poer, plaat	FΖ	20.00	0.00	0.00	-	
0.30	0.15	0.15	0.30	2.10	+ opstorting	FΖ	4.72	0.00	0.00	-	
2.00	1.00	1.00	2.00	1.80	E.G opstort (droog)	FΖ	122.40	0.00	0.00	-	
0.30	0.15	0.15	0.30	1.80	- tpv opstorting	FΖ	-2.75	0.00	0.00	-	

 Sweco Nederland BV
 Blad: 6

 Technosoft Funderingen op Staal release 6.70a
 7 okt 2022

 Project
 : Shallow Foundation for Timber Columns

 Onderdeel
 : Isolated Footing





Sweco Nederland BV		Blad: 7
Technosoft Funderingen og	p Staal release 6.70a	7 okt 2022
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing	
INVOER GRAFISCH Case Bodemprofiel: Loc03_CP	1 (d = -0.80m, F=350) (vervolg) T000000074906	

- 1 0 Fgz=-350 Fgz=-35 6 Bel.ref. 1 - - 1 E.g. poer wordt meegenomen. E.g. opstort wordt meegenomen. Fgz:-350 kN 2 3 4 1 3 5 6 7 8 9 10 9 8 Legenda Legenda 1 : Klei - Schoon - Matig 2 : Klei - Zwak zandig - Vast 3 : Klei - Zwak zandig - Slap 4 : Veen - Matig voorbelast - Matig 5 : Leem - Zwak zandig - Vast 6 : Zand - Schoon - Los 7 : Zand - Sterk siltig - Kleiig 8 : Grind - Sterk siltig - Los 9 : Zand - Zwak siltig - Kleiig 10 : Zand - Schoon - Matig 11 : Klei - Schoon - Vast 11 : Klei - Schoon - Vast
12 : Klei - Zwak zandig - Matig

 Sweco Nederland BV
 Blad: 8

 Technosoft Funderingen op Staal release 6.70a
 7 okt 2022

 Project
 : Shallow Foundation for Timber Columns

 Onderdeel
 : Isolated Footing

 INVOER GRAFISCH Case 1 (d = -0.80m, F=350) (vervolg)

 Bodemprofiel:
 Loc04_CPT000000055844





 Sweco Nederland BV
 Blad: 10

 Technosoft Funderingen op Staal release 6.70a
 7 okt 2022

 Project
 : Shallow Foundation for Timber Columns

 Onderdeel
 : Isolated Footing




Sweco Nederland BV		Blad: 11
Technosoft Funderingen o	p Staal release 6.70a	7 okt 2022
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing	
INVOER GRAFISCH Case Bodemprofiel: Loc12_CP	1 (d = -0.80m, F=350) (vervolg) T000000077437	



 Sweco Nederland BV
 Blad: 12

 Technosoft Funderingen op Staal release 6.70a
 7 okt 2022

 Project
 : Shallow Foundation for Timber Columns

 Onderdeel
 : Isolated Footing

 INVOER GRAFISCH Case 1 (d = -0.80m, F=350) (vervolg)

 Bodemprofiel:
 Loc02_CPT00000003023





 Sweco Nederland BV
 Blad: 14

 Technosoft Funderingen op Staal release 6.70a
 7 okt 2022

 Project
 : Shallow Foundation for Timber Columns

 Onderdeel
 : Isolated Footing





243

Sweco Nederland BV		Blad: 15
Technosoft Funderingen op	o Staal release 6.70a	7 okt 2022
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing	
INVOER GRAFISCH Case Bodemprofiel: Loc09_CP	1 (d = -0.80m, F=350) (vervolg)	



 Sweco Nederland BV
 Blad: 16

 Technosoft Funderingen op Staal release 6.70a
 7 okt 2022

 Project
 : Shallow Foundation for Timber Columns

 Onderdeel
 : Isolated Footing

 INVOER GRAFISCH Case 1 (d = -0.80m, F=350) (vervolg)

 Bodemprofiel:
 Loc10_CPT00000004377



Sweco Nederland BV		Blad: 17
Technosoft Funderingen op	Staal release 6.70a	7 okt 2022
Project : Onderdeel :	Shallow Foundation for Timber Columns Isolated Footing	
INVOER GRAFISCH Case Bodemprofiel: Locl1_CPT	1 (d = -0.80m, F=350) (vervolg) 000000012135	



Sweco Nederland BV	Blad: 18
Technosoft Funderingen op Staal release 6.70a	7 okt 2022
Project : Shallow Foundation for Timber Columns Onderdeel : Isolated Footing	
DESULTATEN ONGEDDAINEEDD Case 1 ($d = -0.90m$ E-350)	
RESULTATEN ONGEDRATINEERD Case I $(u = -0.800m, F=550)$ Resultaten ongedraineerd gedrag laag 3 (Redemprofiel Loc01 CPT00)	0000052457)
Er is gerekend volgens art 6 5 2 2 (f) Geval: c pons	0000032437)
B-tot B-li B-re Niv. Berg Leve b' l' A' σ' so i σ'	$\overline{\mathbf{v}}$, $\overline{\mathbf{v}}$
[m]	[kPa] [kN] [kN]
2.00 1.00 1.00 -0.95 2.04 2.04 2.04 2.04 4.17 8.5 1.20 1.000 9	922.6 501 3848
Resultaten ongedraineerd gedrag laag 2 (Bodemprofiel Loc03_CPT000	0000074906)
Er is gerekend volgens art. 6.5.2.2 (f) Geval: b pons	
B-tot B-li B-re Niv. B_{fic} L_{fic} b' l' A' $\sigma'_{v;z;d}$ s _c i _c (m) [m]	σ' _{max;d} V _d ≤ R _d [kPa] [kN] [kN]
2.00 1.00 1.00 -0.80 2.00 2.00 2.00 2.00 4.00 7.5 1.20 1.000 2	236.0 494 944
Posultaton ongodraincord godrag laag 2 (Rodomprofiel Loc08 CPT00)	000062979)
Fr is gerekend volgens art 6.5.2.2 (f) Geval: h pons	00000829797
B-tot B-li B-re Niv B. L. b' l' A' G' s i G	T' V < R
[m]	[kPa] [kN] [kN]
2.00 1.00 1.00 -0.80 2.00 2.00 2.00 2.00 4.00 7.5 1.20 1.000 3	373.1 494 1492
Resultaten ongedraineerd gedrag laag 2 (Bodemprofiel Loc12 CPT00)	0000077437)
Er is gerekend volgens art. 6.5.2.2 (f) Geval: b pons	· · · · · · · · ·
B-tot B-li B-re Niv. B _{fic} L _{fic} b' l' A' $\sigma'_{u.z.d}$ s _c i _c c	$\sigma'_{max,d} V_d \leq R_d$
[m] [m] [m] [m] [m] [m] [m] [m] [m] [m ²][kPá]´´[–̆] [–̆]	[kPa] [kN] [kN]
2.00 1.00 1.00 -0.80 2.00 2.00 2.00 2.00 4.00 7.5 1.20 1.000 9	921.5 494 3686
Resultaten ongedraineerd gedrag laag 2 (Bodemprofiel Loc02 CPT00	000003023)
Er is gerekend volgens art. 6.5.2.2 (f) Geval: b pons	
B-tot B-li B-re Niv. B _{fic} L _{fic} b' l' A' $\sigma'_{v:z:d}$ s _c i _c c	$\sigma'_{max:d} V_d \leq R_d$
[m] [m] [m] [m] [m] [m] [m] [m] [m] [m ²][kPá] [-] [-]	[kPa] [kN] [kN]
2.00 1.00 1.00 -0.80 2.00 2.00 2.00 2.00 4.00 7.5 1.20 1.000 9	921.5 494 3686
Resultaten ongedraineerd gedrag laag 3 (Bodemprofiel Loc05_CPT00	0000161652)
Er is gerekend volgens art. 6.5.2.2 (f) Geval: c pons	
B-tot B-li B-re Niv. B_{fic} L_{fic} b' l' A' $\sigma'_{v,z,d}$ s _c i _c (m) [m]	σ' _{max;d} V _d ≤ R _d [kPa] [kN] [kN]
2.00 1.00 1.00 -0.86 2.02 2.02 2.02 2.02 4.07 7.9 1.20 1.000	922.0 497 3750
Perultaten ongedraineerd gedrag laag 31 (Bodemprofiel Loc07 CPT0)	0000004530)
Er is gerekend volgens art 6522 (f) Geval: c pops	0000004330)
B-tot B-li B-re Niv Ba. La. b' l' $A' \sigma'$. S i	∀ . < B.
[m]	[kPa] [kN] [kN]
2.00 1.00 1.00-11.85 5.11 5.11 5.11 5.11 26.07100.4 1.20 1.0001	014 1.0e3 26e3
Resultaten ongedraineerd gedrag laag 3 (Bodemprofiel Loc09 CPT000	0000011161)
Er is gerekend volgens art. 6.5.2.2 (f) Geval: c pons	-
B-tot B-li B-re Niv. B_{fic} L_{fic} b' l' A' $\sigma'_{v,z,d}$ s_c i_c c	$\sigma'_{max;d} V_d \leq R_d$
$\frac{1}{2} \begin{array}{cccccccccccccccccccccccccccccccccccc$	021 0 406 3740
2.00 1.00 1.00 0.00 2.01 2.01 2.01 2.01	JLII JJU J/40

Swec	o Nec	lerla	nd BV						Blad: 19
Techn	osoft	Funde	eringen	op Staal	release 6.7	0a			7 okt 2022
Proje	ct			: Shallo	w Foundatio	n for Timbe	r Columns		
Onder	deel			: Isolat	ed Footing				
Resul	taten	onged	drainee:	rd gedrag	laag 2 (Bod	emprofiel L	oc10 CPT0	0000000437	7)
Er is	gere	kend v	volgens	art. 6.5.	2.2 (f) Gev	al: b pons	_		
B-tot	B-li	B-re	Niv. 1	B _{fic} L _{fic}	b' l'	A' $\sigma'_{v;z;d}$	s _c i _c	σ' _{max,d} V _c	i ≤ R _d
[m]	1 00	1 0.0	[m]			[m-][kPa]		[KPA] [KN] [KN]
2.00	1.00	1.00	-0.80	2.00 2.00	2.00 2.00	4.00 7.5 1	.20 1.000	921.5 49	4 3686
Resul	taten	onged	drainee:	rd gedrag	laag 15 (Bo	demprofiel	Loc11_CPT	0000000121	35)
Er is	gere	kend v	olgens	art. 6.5.	2.2 (f) Gev	al: c pons			
B-tot	B-li [m]	B-re [m]	Niv. 1	B _{fic} L _{fic}	[m] [m]	A' $\sigma'_{v;z;d}$ $[m^2][kPa]$	s _c i _c [-] [-]	σ' _{max;d} V _c	i ≤ R _d I [kN]
2 00	1 00	1 00	-7 75	3 95 3 95	3 95 3 95 1	5 63 64 8 1	20 1 000	979 0 8e	3 1563
2.00	1.00	1.00	,.,.	5.95 5.95	5.55 5.55 1	5.05 04.0 I	.20 1.000	5,5 0.00	5 1505
RESU	LTATE	IN GEI	DRAINE	ERD Case	1 (d = -0)	0.80m, F=3	50)		
Resul	taten	gedra	ineerd	gedrag al	le lagen (B	odemprofiel	Loc01_CP	r000000052	457)
Er is	gere	kend v	volgens	art. 6.5.	2.2 (h) Gev	al: c			-
B-tot	B-11 [m]	B-re [m]	A' [m ²]	σ' _{max;d;c}	σ' _{max;d;q} [kPa]	σ' _{max;d;γ}	σ' _{max;d} [kPal	V _d ≤ [kN]	R _d [kN]
2 00	1 00	1 00	4 00	40.6	66.2	18 0	124 8	494	499
2.00	1.00	1.00	4.00	40.0	00.2	10.0	124.0	101	100
Resul	taten	gedra	aineerd	gedrag al	le lagen (B	odemprofiel	Loc03_CP	r000000074	906)
Er is	gere	kend v	rolgens	art. 6.5.	2.2 (h) Gev	al: c			_
B-tot	B-11 [m]	B-re [m]	A' [m ²]	σ' _{max;d;c}	σ' _{max;d;q} [kPa]	σ' _{max;d;γ}	σ' _{max;d}	V _d ≤ [kN]	R _d [kN]
2 00	1 00	1 00	4 00	50.0	11 9	[,KE4] 8 2	103.0	191	112
2.00	1.00	1.00	4.00	50.0	44.9	0.2	103.0	494	412
Resul	taten	gedra	aineerd	gedrag al	le lagen (B	odemprofiel	Loc04_CP	r000000055	844)
Er is	gere	kend v	volgens	art. 6.5.	2.2 (h) Gev	al: c			
B-tot	B-11	B-re	A' [m ²]	σ' _{max;d;c}	σ' _{max;d;q}	σ' _{max;d;γ}	σ' _{max;d}	V _d ≤	R _d
2 00	1 00	1 0.0	4 0 0	[Kra]	222 0	[KEA]	256 5	101	1426
2.00	1.00	1.00	4.00	0.0	222.0	100.7	300.0	494	1420
Resul	taten	gedra	ineerd	gedrag al	le lagen (B	odemprofiel	Loc06_CP	r000000170	226)
Er is	gere	kend v	volgens	art. 6.5.	2.2 (h) Gev	al: b	_		
B-tot	B-li	B-re	A'	σ'max;d;c	σ' _{max;d;q}	σ' _{max;d;γ}	σ' _{max;d}	V _d ≤	Rd
[m]	[m]	[m]	[m²]	[KPa]	[KPa]	[KPa]	[kPa]		
2.00	1.00	1.00	4.00	0.0	305.3	211.4	516.6	494	2066
Resul	taten	gedra	ineerd	gedrag al	le lagen (B	odemprofiel	Loc08 CP	T000000062	979)
Er is	gere	kend v	volgens	art. 6.5.	2.2 (ĥ) Gev	al: c	_		
B-tot	B-li	B-re	Α'	σ' _{max;d;c}	σ' _{max;d;q}	σ' _{max;d;γ}	σ' _{max;d}	V _d ≤	R _d
[m]	[m]	[m]	[m²]	[kPa]	[kPa]	[kPa']	[kPa]	[kN]	[kN]
2.00	1.00	1.00	4.00	23.4	48.3	7.9	79.7	494	319
Resul	taten	gedra	ineerd	gedrag al	le lagen (R	odemprofiel	Loc12 CP	F000000077	437)
Er is	gere	kend v	volgens	art. 6.5.	2.2 (h) Gev	al: c			
B-tot	₿−li	B-re	- A'	σ'	σ'	σ'	σ',	V. <	R,

]	3-tot	B-li	B-re	A	σ' _{max:d:c}	σ' _{max:d:g}	σ' _{max;d;v}	σ' _{max;d}	V _d ≤	R _d
	[m]	[m]	[m]	[m ²]	[kPa]	[kPa]	[kPa']	[kPa]	[kN]	[kN]
	2.00	1.00	1.00	4.00	82.1	85.8	37.2	205.1	494	820

Swec	o Nec	lerla	nd BV						Blad: 20
Techn	osoft	Funde	eringen	op Staal n	celease 6.7	0a			7 okt 2022
Proje Onder	ct deel			: Shallow : Isolate	v Foundatio ed Footing	on for Timl	ber Colum	ns	
Resul	taten	gedra	ineerd	gedrag all	Le lagen (B	odemprofi	el Loc02_	CPT0000000	3023)
Er is	gere	kend v	rolgens	art. 6.5.2	2.2 (h) Gev	al: c	_1		P
[m]	B-11 [m]	в-re [m]	[m ²]	σ' _{max;d;c} [kPa]	σ' _{max;d;q} [kPa]	σ' _{max;d;} [kPa	a] [kPa	d V _d ≤ a] [kN]	[kN]
2.00	1.00	1.00	4.00	27.9	42.8	5	.7 76	.4 <u>494</u>	306
Resul Er is	taten	gedra	ineerd	gedrag all	Le lagen (B 2 (h) Gev	odemprofic	el Loc05_	CPT00000016	51652)
B-tot	B-li	B-re	A'	σ'	σ' σ'	σ'	σ'	. V. ≤	R.
[m]	[m]	[m]	[m ²]	[kPa]	[kPa]	max;d; [kPa	a] [kPa]	a] [kN]	[kN]
2.00	1.00	1.00	4.00	68.5	49.6	11	.9 129	.9 494	520
Resul	taten	gedra	ineerd	gedrag all	Le lagen (B	odemprofi	el Loc07_	CPT0000000)4530)
Er is	gere	kend v	olgens	art. 6.5.2	2.2 (h) Gev	ral: c			_
B-tot [m]	B-li [m]	B-re [m]	A' [m ²]	σ' _{max;d;c} [kPa]	σ' _{max;d;q} [kPa]	σ' _{max;d;} [kPa	γ σ' _{max;} a] [kPa	_d V _d ≤ a] [kN]	R _d [kN]
2.00	1.00	1.00	4.00	0.0	273.1	. 179	.9 453	.0 494	1812
Resul Er is	dere	gedra kend v	olgens	gedrag all	Le lagen (B 2.2 (h) Gev	odemprofi al: c	el Loc09_0	CPT00000001	1161)
B-tot	B-li	B-re	A'	σ'	σ' σ'	σ'	σ'	. V. S	R.
[m]	[m]	[m]	[m ²]	[kPa]	max;d;q [kPa]	max;d; [kPa	a] [kPa	a] [kN]	[kN]
2.00	1.00	1.00	4.00	43.8	64.2	15	.8 123	.9 494	496
Resul	taten	gedra	ineerd	gedrag all	Le lagen (B	odemprofi	el Loc10 (CPT00000000)4377)
Er is	gere	kend v	olgens	art. 6.5.2	2.2 (h) Gev	al: c	-		
B-tot	B-li	B-re	Α'	σ' _{max;d;c}	σ' _{max;d;q}	σ' _{max;d;}	γ σ' _{max;}	d V _d ≤	R _d
[m]	[m]	[m]	[m²]	[kPa]	[kPa]	[kPa	a'] [kPa	a] [kN]	[kN]
2.00	1.00	1.00	4.00	72.8	58.8	16	.4 147	.9 494	592
Resul	taten	gedra	ineerd	gedrag all	le lagen (B	odemprofi	el Loc11_	CPT00000001	2135)
Er is	gere	kend v	olgens	art. 6.5.2	2.2 (h) Gev	ral: c			_
B-tot	B-11 [m]	B-re	A'	σ' _{max;d;c}	σ' _{max;d;q}	σ' _{max;d;}	γ σ' _{max} ;	d V _d ≤	R _d
2 00	1 00	1 00	4 0 0	0.0	234 8	145	4 380	1 494	1521
2.00	1.00	1.00	4.00	0.0	201.0	115	.4 500	.1 191	1021
RESU	LTATE	IN GE	DRAINE	ERD PONS	Case 1 (d	d = -0.8	0m, F=35	0)	
Resul	taten	gedra	ineerd	gedrag por	nsberekenin	g (Bodemp:	rofiel Lo	c01_CPT0000	00052457)
B-tot	B-11	B-re	Niv.	Α' σ' _{ma}	(;d;c σ'max	;d;q o'max	$(;d;\gamma \sigma')$	ax;d ^V d	≤ R _d
2 00	1 00	1 00	_1 15	[III-]	[KPA] [65 7	KFA]	[KPA] [KN]	
2.00	1.00	1.00	-1.15	4.40	42.1	95.7	14.0 1	506 8.10	800
Resul	taten	gedra	ineerd	gedrag por	nsberekenin	g (Bodemp	rofiel Lo	CPT0000	00074906)
B-tot	B-li	B-re	Niv.	A' σ' _{ma}	;d;c σ'max	;d;q o'max	$\kappa; d; \gamma \sigma'_n$	ax;d Vd	≤ R _d
[m]	1 00	1 00	_0 00		[KPA] [52 Q	8 2 1	[KPA] [KN]	
2.00	1.00	1.00	-0.00	4.00	50.0	52.9	0.2 1	11.1 <u>494</u>	<u> </u>
Resul	taten	gedra	ineerd	gedrag por	nsberekenin	g (Bodemp	rofiel Lo	c04_CPT0000	00055844)
B-tot	B-li	B-re	Niv.	A' σ' _{ma} ,	(the state of the	;d;q o'max	$_{k;d;\gamma} \sigma'_{n}$	nax;d Vd	≤ R _d
2.00	1.00	1.00	-4.15	8,65	0.0 13	41.5 2	09.3 15	50.8 634	13419

2.00 1.00 1.00 -2.75

6.49

0.0

566.2

90.1

656.3

578

4262

Sweco Ned	lerland BV							Blad: 21
Technosoft	Funderingen	op Staa	al release	6.70a			7	okt 2022
Project Onderdeel		: Sha : Iso	llow Founda lated Foot:	ation fo ing	r Timber Co	lumns		
Resultaten	gedraineerd	gedrag	ponsberek	ening (B	odemprofiel	Loc08_CP		062979)
B-tot B-li [m] [m]	B-re Niv. [m] [m]	Α' σ [m ²]	' _{max;d;c} σ [kPa]	max;d;q [kPa]	σ' _{max;d;γ} [kPa]	σ' _{max;d} [kPa]	V _d ≤ [kN]	R _d [kN]
2.00 1.00	1.00 -1.75	5.14	14.4	80.2	6.2	100.7	517	518
Resultaten	gedraineerd	gedrag	ponsberek	ening (B	odemprofiel	Loc12 CP	7000000	077437)
B-tot B-li [m] [m]	B-re Niv. [m] [m]	Α' σ [m ²]	' _{max;d;c} σ' [kPa]	max;d;q [kPa]	σ' _{max;d;γ} [kPa]	σ' _{max;d} [kPa]	V _d ≤ [kN]	R _d [kN]
2.00 1.00	1.00 -1.75	5.14	94.2	211.2	36.9	342.3	536	1759
Resultaten	gedraineerd	gedrag	ponsberek	ening (B	odemprofiel	Loc02 CP	7000000	003023)
B-tot B-li [m] [m]	B-re Niv. [m] [m]	Α' σ [m ²]	' _{max;d;c} σ' [kPa]	max;d;q [kPa]	σ' _{max;d;γ} [kPa]	σ' _{max;d} [kPa]	V _d ≤ [kN]	R _d [kN]
2.00 1.00	1.00 -0.85	4.06	26.9	51.7	5.0	83.6	496	339
Resultaten	gedraineerd	gedrag	ponsberek	ening (B	odemprofiel	Loc05 CP	T000000	161652)
B-tot B-li	B-re Niv.	Α'σ	'max;d;c σ	max;d;q	$\sigma'_{max;d;\gamma}$	σ' _{max;d}	V _d ≤	R _d
		[m²]	[kPa]	[kPa]	[kPa]	[kPa]	[KN]	[KN]
2.00 1.00	1.00 -1.06	4.30	66.5	62.1	/./	130.3	505	586
Resultaten	gedraineerd	gedrag	ponsberek	ening (B	odemprofiel	Loc07_CP	000000T	004530)
B-tot B-li	B-re Niv.	Α' σ [m ²]	' _{max;d;c} σ' [kPa]	max;d;q	σ' _{max;d;γ} [kPa]	σ' _{max;d} [kPa]	V _d ≤ [kN]	R _d [kN]
2.00 1.00	1.00 -3.15	7.08	0.0	847.7	136.8	984.5	596	6969
Resultaten	gedraineerd	gedrag	ponsberek	ening (B	odemprofiel	Loc09_CP	PT000000	011161)
B-tot B-li [m] [m]	B-re Niv. [m] [m]	Α' σ [m ²]	' _{max;d;c} σ' [kPa]	max;d;q [kPa]	σ' _{max;d;γ} [kPa]	σ' _{max;d} [kPa]	V _d ≤ [kN]	R _d [kN]
2.00 1.00	1.00 -1.05	4.29	40.8	85.2	11.1	137.1	505	587
Resultaten	gedraineerd	gedrag	ponsberek	ening (B	odemprofiel	Loc10_CP		004377)
B-tot B-li [m] [m]	B-re Niv. [m] [m]	Α' σ [m ²]	' _{max;d;c} σ [kPa]	max;d;q [kPa]	σ' _{max;d;γ} [kPa]	σ' _{max;d} [kPa]	V _d ≤ [kN]	R _d [kN]
2.00 1.00	1.00 -1.15	4.40	51.1	86.7	12.1	150.0	509	660
Resultaten	gedraineerd	gedrag	ponsberek	ening (B	odemprofiel	Loc11 CP	T000000	012135)
B-tot B-li	B-re Niv.	Α' σ	' _{max;d;c} σ	max;d;q	σ' _{max;d;v}	σ' _{max;d}	V _d ≤	R _d
[m] [m]	[m] [m]	[m ²]	[kPa]	[kPa]	[kPa]	[kPa]	[kN]	[kN]

Swec	o Nec	lerland	i BV								Blac	d: 22
Techn	osoft	Funder	ingen	op Staal	releas	e 6.70a					7 okt	2022
Proje	ct			: Shallo	ow Foun	dation	for Ti	mber Co	lumns			
Onder	deel			: Isolat	ted Foo	ting						
DECIT	ር.ጥልጥፍ	N 7486	TNC ((d = -1)	0 80m	F-35(าง				
Resul	taten	zakkin	g (Bod	emprofie	Loc01	CPT000	000052	457)				
B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	185.5	44.9	230.4	150.0	2220	
Resul	taten	zakkin	g (Bod	emprofiel	Loc03	CPT000	000074	906)				
B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m]	[m]	[m]		[m]	[KN]	[KPa]	[mm]	[mm]	[mm]	[mm]	[KN/M]	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	248.9	53.9	302.8	150.0	1655	
Resul	taten	zakkin	g (Bod	emprofie	Loc04	_CPT000	000055	844)				
B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	52 [mm]	S	≤ s _{req}	Veerw.	
2 00	1 00	1 00	2 00	2 00	411 Q	103 0	5.4	0.0	5.4	150 0	76831	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	5.4	0.0	5.4	100.0	10031	
Resul	taten	zakkin	g (Bod	emprofiel	Loc06	_CPT000	000170	226)				
B-tot	B-li	B-re	b'	1' [m]	V _d	σ _{gem;d}	s ₁	52 [mm]	S	≤ s _{req}	Veerw.	
2 00	1 00	1 00	2 00	2 00	111 Q	[KPA]	3 3	0.0	2.3	150 0	125545	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	5.5	0.0	5.5	130.0	123343	
Resul	taten	zakkin	g (Bod	emprofiel	Loc08	CPT000	000062	979)				
B-tot	B-li	B-re	b'	1'	Vd	σ _{gem;d}	s ₁	s2	S	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[KN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	427.6	199.4	627.0	150.0	963	
Resul	taten	zakkin	g (Bod	emprofie	l Loc12	CPT000	000077	437)				
B-tot	B-li	B-re	b'	1'	V _d	$\sigma_{\text{gem;d}}$	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	72.9	19.5	92.4	150.0	5651	
Resul	taten	zakkin	g (Bod	emprofie	Loc02	CPT000	000003	023)				
B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s2	S	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[KN]	[KPA]	[mm]	[mm]		[mm]	[KN/M]	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	455.8	158.9	614./	150.0	904	
Resul	taten	zakkin	g (Bod	emprofiel	Loc05	CPT000	000161	652)				
B-tot	B-li	B-re	b'	1'	Vd	$\sigma_{\text{gem;d}}$	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	293.4	145.4	438.7	150.0	1404	
Resul	taten	zakkin	g (Bod	emprofiel	Loc07	CPT000	000004	530)				
B-tot	B-li	B-re	b'	1'	V _d	$\sigma_{\text{gem};d}$	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	4.1	0.0	4.1	150.0	101612	
Resul	taten	zakkin	g (Bod	emprofiel	Loc09	CPT000	000011	161)				
B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	411.9	103.0	366.9	202.1	569.0	150.0	1123	

Technosoft Funderingen op Staal release 6.70a 7 okt 2022 Project : Shallow Foundation for Timber Columns Cnderdeel : Isolated Footing Resultaten zakking (Bodemprofiel Loc1_CPT00000004377) B=tot B=1i B=re b' 1' V _d $\sigma_{gen/d} \leq 1 \leq 2_2 \leq \leq s \leq reg Veerw.$ [m] (m] (m] (m] (m] (m] (kN] [kPa] (mn] (mm] [mm] (kN/m] 2.00 1.00 1.00 2.00 2.00 411.9 103.0 267.8 129.5 397.3 150.0 1538 Resultaten zakking (Bodemprofiel Loc1_CPT00000012135) B=tot B=1i B=re b' 1' V _d $\sigma_{gen/d} \leq 1 \leq 2_2 \leq \leq s \leq reg Veerw.$ [m] (m] (m] (m] (m] (m] (kN] [kPa] (mm] (mm] [mm] [mm] [kN/m] 2.00 1.00 1.00 2.00 2.00 411.9 103.0 4.7 0.0 4.7 150.0 86917 RESULTATEM ZAKKING Case 2 (d = -2.00m, F=350) Resultaten zakking (Bodemprofiel Loc1_CPT000000052457) B=tot B=1i B=re b' 1' V _d $\sigma_{gen/d} \leq 1 \leq 2_2 \leq s \leq r_{reg} Veerw.$ [m] (m] (m] (m] (m) (kN] [kPa] (mm] [mm] [mm] [mm] [kN/m] 2.00 1.00 1.00 2.00 2.00 494.4 123.6 99.8 23.5 123.3 150.0 4954 Resultaten zakking (Bodemprofiel Loc2_CPT000000052457) B=tot B=1i B=re b' 1' V _d $\sigma_{gen/d} \leq 1 \leq s \leq r_{reg} Veerw.$ [[m] (m] (m] (m] (m] (kN] [kPa] (mm] [mm] [mm] [mm] [kN/m] 2.00 1.00 1.00 2.00 2.00 494.4 123.6 99.8 23.5 123.3 150.0 4954 Resultaten zakking (Bodemprofiel Loc2_CPT000000055844) B=tot B=1i B=re b' 1' V _d $\sigma_{gen/d} \leq 1 \leq s \leq s reg Veerw.$ [[m] (m] (m] (m] (m] (m] (kN] [kPa] (mm] (mm] [mm] [mm] [mm] [kN/m] 2.00 1.00 1.00 2.00 2.00 494.4 123.6 5.1 0.0 5.1 150.0 4686 Resultaten zakking (Bodemprofiel Loc2(CPT00000055244) B=tot B=1i B=re b' 1' V _d $\sigma_{gen/d} \leq 1 \leq s \leq s reg Veerw.$ [[m] (m] (m] (m] (m] (m] (kN] [kPa] (mm] (mm] (mm] [mm] [mm] [kN/m] 2.00 1.00 1.00 2.00 2.00 494.4 123.6 5.1 0.0 2.7 150.0 13354 Resultaten zakking (Bodemprofiel Loc2(CPT00000052479) B=tot B=1i B=re b' 1' V _d $\sigma_{gen/d} \leq 1 \leq s \leq s \leq s reg Veerw.$ [[m] (m] (m] (m] (m] (kN] [kPa] (mm] (mm] (mm] [mm] [kN/m] 2.00 1.00 1.00 2.00 2.00 494.4 123.6 5.0 11.9 67.0 150.0 8981 Resultaten zakking (Bodemprofiel Loc2(CPT00000052479) B=tot B=1i B=re b' 1' V _d $\sigma_{gen/d} \leq 1 \leq s \leq s \leq s \leq s \leq s \leq s \le $	Swec	o Nec	lerland	i BV								Blac	d: 23
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Techn	osoft	Funder	ingen d	op Staal	releas	e 6.70a					7 okt	2022
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Proje	ct			: Shallo	ow Foun	dation :	for Tin	nber Co	lumns			
$\begin{array}{l l l l l l l l l l l l l l l l l l l $	Onder	deel			: Isolat	ed Foo	ting						
$ B-tot B-1i B-re b' 1' V_d \sigma_{gen;d} s_1 s_2 s \leq s_{req} Veerw. \\ $	Resul	taten	zakkin	g (Bode	emprofiel	L Loc10	_CPT000	000004	377)				
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	B-tot	B-li	B-re	b'	1' [m]	V _d	σ _{gem;d}	s ₁	s ₂	S [mm]	≤ s _{req}	Veerw.	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2.00	1.00	1.00	2.00	2.00	411.9	103.0	267.8	129.5	397.3	150.0	1538	
$\begin{array}{l c c c c c c c c c c c c c c c c c c c$	_												
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Resul	B-li	zakkin Baro	g (Bode	emprofiel	L Loc11	_CPT000	000012:	135)		< ~	Voorw	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	_ Sreq [mm]	[kN/m]	
$ \begin{array}{l c c c c c c c c c c c c c c c c c c c$	2.00	1.00	1.00	2.00	2.00	411.9	103.0	4.7	0.0	4.7	150.0	86917	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	RESU	T. TA TF	N ZAKI	KING C	ase 2 (d = -3	2.00m.	F=35())				
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Resul	taten	zakkin	g (Bode	emprofiel	L Loc01	_CPT000	000052	457)				
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S [mm]	≤ s _{req}	Veerw.	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	2.00	1.00	1.00	2.00	2.00	494.4	123.6	99.8	23.5	123.3	150.0	4954	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $													
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Resul	B-li	zakkin B-re	g (Bode	emprofiel	L Loc03	_CPT000	0000749	906)	9	< 9	Veerw	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	_ Sreq [mm]	[kN/m]	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2.00	1.00	1.00	2.00	2.00	494.4	123.6	118.9	23.1	142.0	150.0	4158	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Resul	taten	zakkin	g (Bode	emprofiel	L Loc04	CPT000	000055	844)				
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	s	≤ s _{req}	Veerw.	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	2 00	[m]	[m]	[m]	[m]	[KN]	[KPa]	[mm]	[mm]	[mm]	[mm]	[KN/m]	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2.00	1.00	1.00	2.00	2.00	1,1,1	125.0	0.1	0.0	5.1	100.0	20000	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Resul	taten B-li	zakkin	g (Bode	emprofiel		_CPT000	000170	226)	5	< ~	Voorw	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	[m]	[m]	[m]	[m]	[m]	[kN]	gem;d [kPa]	[mm]	[mm]	[mm]	⊃ ^S req [mm]	[kN/m]	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	2.00	1.00	1.00	2.00	2.00	494.4	123.6	2.7	0.0	2.7	150.0	183534	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Resul	taten	zakkin	a (Bode	mprofiel		CPT000	000062	979)				
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2.00	1.00	1.00	2.00	2.00	494.4	123.6	370.2	178.9	549.1	150.0	1336	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Resul	taten	zakkin	g (Bode	emprofiel	L Loc12	_CPT000	000077	437)				
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	B-tot	B-li	B-re	b'	1' [m]	V _d	σ _{gem;d}	s ₁	5 ₂	S [mm]	≤ s _{req}	Veerw.	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	2.00	1.00	1.00	2.00	2.00	494.4	123.6	55.0	11.9	67.0	150.0	8981	
$ \begin{array}{c c} Resultaten zakking (Bodemprofile Locuz CProvou00000000000000000000000000000000000$				(5.1									
[m] [m] [m] [m] [m] [kN] [kPa] [mm] [mm] [mm] [kN/m]	B-tot	B-li	zakkin B-re	g (воає b'	mprories l'	L LOCUZ	_CPT000	51 S1	JZ3) So	s	≤ s	Veerw.	
	[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00 1.00 2.00 2.00 494.4 123.6 359.6 130.7 490.3 150.0 1375	2.00	1.00	1.00	2.00	2.00	494.4	123.6	359.6	130.7	490.3	150.0	1375	
Resultaten zakking (Bodemprofiel Loc05_CPT000000161652)	Resul	taten	zakkin	g (Bode	emprofiel	L Loc05	_CPT000	000161	652)				
B-tot B-li B-re b' l' $V_d \sigma_{gem;d} s_1 s_2 s \le s_{req}$ Veerw.	B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S [mm]	≤ s _{req}	Veerw.	
	2.00	1.00	1.00	2.00	2.00	494.4	123.6	362.8	127.6	490.4	150.0	1.363	
	2.00	2.00	2.00	2.00	2.00							2000	

Sweco Ned	lerland	вv								Blac	d: 24
Technosoft	Funderi	ngen	op Staal	releas	e 6.70a					7 okt	2022
Project			: Shall	ow Foun	dation :	for Ti	mber C	olumns			
Onderdeel			: Isola	ted Foc	ting						
Pogultaton	zakkino	(Pod	omorofio	1 1 0 0 0 7	CDTOOO	000004	530)				
B-tot B-li	B-re	, (воц b'	l'	V _d		S1	530) So	S	≤ srog	Veerw.	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	494.4	123.6	4.7	0.0	4.7	150.0	104231	
Resultaten	zakking	g (Bod	emprofie	1 Loc09	CPT000	000011	161)				
B-tot B-li	B-re	b'	1'	Vd	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	494.4	123.6	441.2	206.9	648.0	150.0	1121	
Resultaten	zakking	g (Bod	emprofie	l Loc10	_CPT000	000004	377)				
B-tot B-li	B-re	b'	1'	V _d	$\sigma_{\text{gem;d}}$	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	494.4	123.6	331.8	112.5	444.3	150.0	1490	
Resultaten	zakking	g (Bod	emprofie	l Loc11	_CPT000	000012	135)				
B-tot B-li	B-re	b'	1'	Vd	$\sigma_{gem;d}$	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	494.4	123.6	6.3	0.0	6.3	150.0	78638	
RESULTATE	N ZAKK	ING C	Case 3	(d = -	0.80m,	F=100	00)				
Resultaten	zakking	g (Bod	emprofie	1 Loc01	_CPT000	000052	457)				
B-tot B-li	B-re	b'	1'	Vd	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	1061.9	265.5	289.6	52.5	342.1	150.0	3667	
Resultaten	zakking	g (Bod	emprofie	1 Loc03	_CPT000	000074	906)				
B-tot B-li	B-re	b'	1'	V _d	$\sigma_{\text{gem;d}}$	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]		[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	1001.9	203.3	370.9	55.5	420.4	150.0	2003	
Resultaten	zakking	J (Bod	emprofie	l Loc04	_CPT000	000055	844)				
B-tot B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
2 00 1 00	[m]	2 00	2 00	[KN]	265 5	2 Q		[mm]	[mm]	[KN/M]	
2.00 1.00	1.00	2.00	2.00	1001.9	203.5	0.9	0.0	0.9	10.0	110003	
Resultaten	zakking	g (Bod	emprofie	1 Loc06	_CPT000	000170	226)				
B-tot B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s2	S	≤ s _{req}	Veerw.	
[m] [m]	[m]	[m]	[m]	[KN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	1061.9	265.5	6.6	0.0	6.6	150.0	161359	
Resultaten	zakking	g (Bod	emprofie	1 Loc08	_CPT000	000062	979)				
B-tot B-li	B-re	b'	1'	V _d	$\sigma_{\text{gem;d}}$	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	1061.9	265.5	758.3	314.9	1073.1	150.0	1400	
Resultaten	zakking	g (Bod	emprofie	l Loc12	_CPT000	000077	437)				
B-tot B-li	B-re	b'	1'	Vd	σ _{gem;d}	s ₁	s2	S	≤ s _{req}	Veerw.	
	[m]	[m]	[m]	[KN]	[KPA]	[mm]	[mm]	[mm]	[mm]	[KN/M]	
2.00 1.00	1.00	2.00	2.00	1061.9	265.5	11/.5	20.6	138.1	150.0	9035	

Sweco Ne	ederlan	d BV								Blac	d: 25
Technosof	t Funder	ringen c	op Staa	l releas	e 6.70a					7 okt	2022
Project Onderdeel			: Shal : Isol	low Foun ated Foo	dation : ting	for Tin	mber Co	lumns			
Resultate	n zakkin	ng (Bode	mprofi	el Loc02	CPT0000	000003	023)				
B-tot B-l	i B-re	b'	1'	Vd	σ _{αem:d}	s ₁	S2	S	≤ s _{rea}	Veerw.	
[m] [m] [m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.0	0 1.00	2.00	2.00	1061.9	265.5	741.5	189.9	931.4	150.0	1432	
Resultate	n zakkir	na (Bode	mprofi	el Loc05	CPT0000	000161	652)				
B-tot B-1	i B-re	b'	1'	Vd	σ _{σom} .d	S ₁	s.	S	≤ s _{rog}	Veerw.	
[m] [m] [m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.0	0 1.00	2.00	2.00	1061.9	265.5	502.6	145.4	647.9	150.0	2113	
Resultate	n zakkir	ng (Bode	mprofi	el Loc07	_CPT0000	000004	530)				
B-tot B-l	i B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m] [m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.0	0 1.00	2.00	2.00	1061.9	265.5	7.9	0.0	7.9	150.0	135238	
Resultate	n zakkir	ng (Bode	mprofi	el Loc09	CPT0000	000011	161)				
B-tot B-l	i B-re	b'	1'	Vd	σ _{αem;d}	s ₁	s ₂	S	≤ s _{rea}	Veerw.	
[m] [m] [m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.0	0 1.00	2.00	2.00	1061.9	265.5	649.8	231.8	881.6	150.0	1634	
Resultate	n zakkir	ng (Bode	mprofi	el Loc10	CPT0000	000004	377)				
B-tot B-1	i B-re	b'	1'	Vd	σ _{σom} .d	S ₁	S ₂	S	≤ srog	Veerw.	
[m] [m] [m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.0	0 1.00	2.00	2.00	1061.9	265.5	455.8	129.5	585.3	150.0	2330	
Resultate	n zakkir	ng (Bode	mprofi	el Loc11	_CPT0000	000012	135)				
B-tot B-l	i B-re	b'	1'	V _d	$\sigma_{gem;d}$	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m] [m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.0	0 1.00	2.00	2.00	1061.9	265.5	9.2	0.0	9.2	150.0	115841	
RESULTAT	EN ZAK	KING C	ase 4	(d = -:	2.00m,	F=100	00)				
Resultate	n zakkir	ig (Bode	mprofi	el Loc01	_CPT0000	000052	457)				
B-tot B-1	i B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	S	≤ s _{req}	Veerw.	
[m] [m] [m]	[m]	[m]	[KN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[KN/m]	
2.00 1.0	0 1.00	2.00	2.00	1144.4	286.1	150.4	28.2	178.6	150.0	7609	
Resultate	n zakkir	ng (Bode	mprofi	el Loc03	CPT0000	000074	906)				
B-tot B-l	i B-re	b'	1'	Vd	σ _{gem;d}	s ₁	s ₂	S	≤ s _{rea}	Veerw.	
[m] [m] [m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	

2.00 1.00 1.00 2.00 2.00 1144.4 286.1 171.0 25.7 196.7 150.0 6693

Resultaten zakking (Bodemprofiel Loc04_CPT00000055844)

B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	s	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	1144.4	286.1	8.7	0.0	8.7	150.0	132186	

Resultaten zakking (Bodemprofiel Loc06_CPT000000170226)

B-tot	B-li	B-re	b'	1'	V _d	σ _{gem;d}	s ₁	s ₂	s	≤ s _{req}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	1144.4	286.1	6.2	0.0	6.2	150.0	184120	

Sweco Nec	lerland	BV								Blac	d: 26
Technosoft	Funderi	ngen op	Staal	l releas	e 6.70a					7 okt	2022
Project Onderdeel		:	Shall Isola	low Found ated Foo	dation : ting	for Tin	mber Co	lumns			
Resultaten	zakking	(Boder	nprofie	el Loc08	CPT0000	000062	979)				
B-tot B-li [m] [m]	B-re [m]	b' [m]	 [m]	V _d [kN]	σ _{gem;d} [kPa]	s ₁ [mm]	s ₂ [mm]	s [mm]	≤ s _{req} [mm]	Veerw. [kN/m]	
2.00 1.00	1.00	2.00	2.00	1144.4	286.1	670.0	315.8	985.8	150.0	1708	
Resultaten	zakking	(Boder	nrofie	1 1.0012	CPT0000	000077	437)				
P-tot P-li	Baro		11				-377	0	< .	Vooru	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	Sreq [mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	1144.4	286.1	91.4	25.6	117.0	150.0	12520	
Dogultaton		(Dede	mmafia	1 1 0 0 0 2	CDE0000		0221				
Resultaten	zakking	(Bodel	uprorre	T TOGOS		000003	023)		/		
[m] [m]	B-re [m]	[m]	[m]	V _d [kN]	σ _{gem;d} [kPa]	s ₁ [mm]	s ₂ [mm]	s [mm]	≤ s _{req} [mm]	veerw. [kN/m]	
2.00 1.00	1.00	2.00	2.00	1144.4	286.1	554.9	141.3	696.2	150.0	2062	
Pegultaton	zakking	(Boder	nrofie	1 1.0005	CDT0000	000161	652)				
	Zakking	(Boden	uprorre	T TOCOD		000101	052)		< ~	1700001	
[m] [m]	[m]	[m]	[m]	[kN]	O _{gem;d} [kPa]	[mm]	[mm]	[mm]	≤ S _{req} [mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	1144.4	286.1	561.8	127.6	689.4	150.0	2037	
Resultaten	zakking	(Boder	nrofie	1 1.0007	CPT000	00004	530)				
B-tot B-li	Baro	h!		<u>, 10007</u>				c .	< ~	Voorw	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	_ Sreq [mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	1144.4	286.1	8.8	0.0	8.8	150.0	130573	
Resultaten	zakking	(Boder	profie	1 1.0009	CPT0000	000011	161)				
Resultaten	Baro	h!						c .	< ~	Voorw	
	D-IG	[m]		V d	gem;d	5 ₁	52 [mm]	ت 1 سسا	- Sreq	Veerw.	
	[III]	[[[]	[III]	[K IN]	[KPA]	[mm]	[mm]	ן וונננו ן	[11111]		
2.00 1.00	1.00	2.00	2.00	1144.4	286.1	692.4	206.9	899.3	150.0	1653	
Resultaten	zakking	(Boder	mprofie	el Loc10	CPT0000	000004:	377)				
B-tot B-li	B-re	b'	1'	Vd	σ	S ₁	So	S	≤ sre~	Veerw.	
[m] [m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00 1.00	1.00	2.00	2.00	1144.4	286.1	510.2	120.0	630.2	150.0	2243	

Resultaten zakking (Bodemprofiel Loc11_CPT00000012135)

B-tot	B-li	B-re	b'	1'	V _d	σ _{gemid}	s ₁	s ₂	S	≤ s _{rea}	Veerw.	
[m]	[m]	[m]	[m]	[m]	[kN]	[kPa]	[mm]	[mm]	[mm]	[mm]	[kN/m]	
2.00	1.00	1.00	2.00	2.00	1144.4	286.1	11.5	0.0	11.5	150.0	99765	

Horizontal Loading

Sweco Nederland BV	Blad: 1
Technosoft Funderingen op Staal release 6.70a	29 mrt 2023
Project : Shallow Foundation for Timber Columns Onderdeel : Isolated Footing	
ALGEMENE GEGEVENS	
Project : Shallow Foundation for Timber Columns Onderdeel : Isolated Footing Zenheden : [kN][m][MPa][graden] tenzij anders vermeld Datum : 05-10-2022 Referentieniveau (RN) : Maaiveld Referentieperiode : 50 jaar Bestand : C:\Users\nlpabu\OneDrive - Sweco AB\Documenten\Afstuderen\Technosoft\ Draagcapaciteit Fundering op Staal (incl. hor).fsw	
Toegepaste normen volgens Eurocode met Nederlandse NB	
Beton NEN-EN 1992-1-1:2011(nl) C2/A1:2015(nl) NB:2016 Geotechniek EN 1997-1:2004 AC:2009 NEN-EN 1997-1:2005 C1+A1:2013 NB:2016 NEN 997-1:2016 C2:2017	(nl)
PROFIELGEGEVENS Isolated Footing 2000x2000x200	
Type : Poer	
Links Rechts Breedte min [mm]: 1000 1000 max [mm]: 1000 1000 stap [mm]: 100 100	
Lengte voor [mm] : 1000 achter [mm] : 1000 De poer is vierkant. Hoogte [mm] : 200 Dpstorting breedte [mm] : 300 lengte [mm] : 300	
BELASTINGGEGEVENS Load hor	
Nr. Omschrijving Type Richting Waarde AfstandX AfstandY Afst [kN,m] [m] [m]	candZ [m]
1 Wind F/q X 63.00 (2 Self-Weight F/q Z -344.00 0.00 0.00	0.00 _
Extra lasten t.g.v. eigengewicht poer en opstort staan bij de rekengege	evens.
RESULTATEN ONGEDRAINEERDE AFSCHUIVING Case 1 (d = -0.80m, FzResultaten ongedraineerde afschuiving (Bodemprofiel Loc12_CPT000000077B-tot B-li B-re Niv. B_{fic} L_{fic} V_d b' $1'$ A' $c_{u,d}$ F [m] [m] [m][m][m][m][m][m][m][m] $[m]$ $[m]$	2 =344, Fh=63 437) H _d ≤ R _d N] [kN]
2.00 1.00 1.00 -0.80 2.00 2.00 365.3 2.00 2.00 4.00 148.1 75	.6 592.6
Resultaten ongedraineerde afschuiving (Bodemprofiel Loc07_CPT000000045 B-tot B-li B-re Niv. B _{fic} L _{fic} V _d b' l' A' c _{u;d} F [m] [m] [m] [m] [m] [m] [kN] [m] [m] [m ²] [kPa] [kN	530) H _d ≤ R _d N] [kN]
2.00 1.00 1.00-11.85 5.11 5.11 773.0 2.94 5.11 15.03 148.1 75	.6 2227.4

Sweco Ned	lerland BV					Blad: 2
Technosoft	Funderingen	op Staal rel	ease 6.70a			29 mrt 2023
Project		: Shallow F	oundation f	or Timber Co	Lumns	
Ollderdeer		. ISOIALEU	FOOLING			
Resultaten	ongedrainee	rde afschuivi	ng (Bodempr	ofiel Loc11 (CPT00000001213	5)
B-tot B-li	B-re Niv.	B _{fic} L _{fic}	V _d b'	l' A'	c _{u;d} H _d	\leq R _d
[m] [m]	[m] [m]	[m] [m]	[kN] [m]	[m] [m²]	[kPa] [kN]	[kN]
2.00 1.00	1.00 -7.75	3.95 3.95	617.5 2.25	3.95 8.90	148.1 75.6	1318.9
RESULTATE	N GEDRAINE	ERDE AFSCHU	JIVING Cas	e 1 (d = -0).80m, Fz=34	4, Fh=63)
Resultaten	gedraineerde	e afschuiving	(Bodemprof	iel Loc04_CP	000000055844)	
B-TOT B-TI	B-re $\phi_{cv;d}$	od V'd	H _d S	K _d FN1		
2.00 1.00	1.00 33.7	22.5 365.3	75.6 15	1.1		
Resultaten	gedraineerde	e afschuiving	(Bodemprof	iel Loc06 CP	000000170226	
B-tot B-li	B-re o'	δ. V'.	H. S	R.		
[m] [m]	[m] [°]	[°] [kN]	[kN] [kN]		
2.00 1.00	1.00 33.7	22.5 365.3	75.6 15	1.1		
Resultaten	gedraineerde	e afschuiving	(Bodemprof	iel Loc12_CP	000000077437)	
B-tot B-li	B-re $\phi'_{cv;d}$	δ_{d} V' _d	H _d ≤	R _d		
[m] [m]	[m] [°]	[°] [kN]	[kN] [kN]		
2.00 1.00	1.00 24.4	16.2 365.3	75.6 10	6.4		
Resultaten	gedraineerde	e afschuiving	(Bodemprof	iel Loc07 CP	000000004530)	
B-tot B-li	B-re $\phi'_{cv;d}$	δ_{d} V' _d	H _d ≤	R _d		
[m] [m]	[m] [°]	[°] [kN]	[kN] [kN]		
2.00 1.00	1.00 33.7	22.5 365.3	75.6 15	1.1		
Resultaten	gedraineerde	e afschuiving	(Bodemprof	iel Loc11 CP	000000012135)	
B-tot B-li	B-re $\phi'_{cv;d}$	δ_{d} V' _d	H _d ≤	R _d		
[m] [m]	[m] [°]	[°] [kN]	[kN] [kN]		
2.00 1.00	1.00 33.7	22.5 365.3	75.6 15	1.1		
RESULTATE	N ONGEDRAI	NEERDE AFSC	HUIVING C	ase 2 (d =	-2.00m, Fz=	344, Fh=63)
Resultaten	ongedrainee	rde afschuivi	ng (Bodempr	ofiel Loc12_(CPT00000007743	7)
R-LOT R-II	B-re NiV.	B _{fic} L _{fic}	V _d D'	A.	C _{u;d} H _d	S R _d
2 00 1 00	1 00 -2 00	2 00 2 00	120 E 2 00	2 0 0 4 0 0	74 1 75 C	206.3
2.00 1.00	1.00 -2.00	2.00 2.00	439.3 2.00	2.00 4.00	/4.1 /5.6	290.3
Resultaten	ongedrainee	rde afschuivi	ng (Bodempr	ofiel Loc07_(CPT00000000453	0)
B-tot B-11	B-re NiV.	B _{fic} L _{fic}	V _d D'	A.	C _{u;d} H _d	S R _d
2 00 1 00	1 00-11 95		700 7 2 01	1 77 12 0C	140 1 75 6	2052.2
2.00 1.00	1.00-11.85	4.// 4.//	/99./ 2.91	4.// 13.00	140.1 /5.0	2033.2
Resultaten	ongedrainee	rde afschuivi	ng (Bodempr	ofiel Loc11_(CPT00000001213	5)
B-tot B-li	B-re Niv.	B _{fic} L _{fic}	V _d b'	1' A'	C _{u;d} H _d	≤ R _d
2.00 1.00	1.00 -7.75	3.62 3.62	644.2 2.27	3.62 8.20	148.1 75.6	1214.3
						4
RESULTATE	N GEDRAINE	ERDE AFSCHU	IVING Cas	$e^{2}(a = -2)$	2.00m, $Fz=34$	4, Fn=63)
Resultaten	georaineerde	e arscnuiving δ. ν'	(Bodemprof	R.	100000055844)	
[m] [m]	[m] [¹]	[°] [kN]	[kN] [kN]		
2.00 1.00	1.00 29.0	19.3 439.5	75.6 15	4.1		

Sweco Neo	derland BV	Blad: 3
Technosoft	Funderingen op Staal release 6.70a	29 mrt 2023
Project Onderdeel	: Shallow Foundation for Timber Columns : Isolated Footing	
Resultaten B-tot B-li [m] [m]	$\begin{array}{llllllllllllllllllllllllllllllllllll$	
2.00 1.00	1.00 33.7 22.5 439.5 75.6 181.8	
Resultaten B-tot B-li [m] [m]	$\begin{array}{llllllllllllllllllllllllllllllllllll$	
2.00 1.00	1.00 15.3 10.2 439.5 75.6 79.3	
Resultaten B-tot B-li [m] [m]	$\begin{array}{llllllllllllllllllllllllllllllllllll$	
2.00 1.00	1.00 33.7 22.5 439.5 75.6 181.8	
Resultaten B-tot B-li [m] [m] 2.00 1.00	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	

D.3 Output Technosoft Pile Foundation

Example Calculation Pile Foundation

Sweco Nederland BV		Blad: 1
Technosoft Paalfundering	en release 6.70a	14 okt 2022
Project Onderdeel	: Pile Foundation for Timber Columns : Timber Pile	
DETAIL BER. DRAAGVERN Maaiveld-4.00	OGEN Case 1; Loc01_CPT000000052457;	
Uitgangspunten		
- gehanteerde sondering	: Loc01_CPT00000052457	
- gehanteerde paal	: Timber pile d=160	
- paalpuntniveau	: Maaiveld-4.00 m	
- traject positieve klee:	t : Maaiveld-13.50 m	
	tot: Maaiveld -4.00 m	
Maximale draagkracht van	de paalpunt	
De maximale puntweerstand	d volgens art. 7.6.2.3 (e) bedraagt :	
$q_{b:max} = 0.5 * \alpha_{p} *$	β * s * ((q _{c:I:gem} + q _{c:II:gem})/2 + q _{c:III:g}	em)

	= 1.758 MPa		0,111,90
waarin :		ir	n dit geval :
q _{c;I;gem}	= de gemiddelde waarde van de conusweer- standen over traject I	=	3.60 MPa
q _{c;II;gem}	= de gemiddelde waarde van de conusweer- standen over traject II	=	2.75 MPa
q _c ;III;gem	= de gemiddelde waarde van de conusweer- standen over traject III	=	1.85 MPa
α_{n}	= paalklassefactor	=	0.70 -
β	= factor voor de paalvoetvorm	=	1.00 -
φ	= hoek van de inwendige wrijving	=	27.5 -
r	= verhouding b/a	=	1.00 -
S	= factor voor de vorm van de voet	=	1.00 -

Voor een uitgebreide beschrijving van het bepalen van de gemiddelde conusweerstanden in de gebieden I, II en III wordt verwezen naar art. 7.6.2.3 (e) in de norm.

De maximale draagkracht van de paalpunt volgens art. 7.6.2.3 (c) bedraagt:

```
R_{b;cal;max;i} = A_{b} * q_{b;max;i}
= 35 kN
```

waarin : A_b in dit geval : = 0.0201 m²

Maximale paalschachtwrijving

De maximale paalschachtwrijving volgens art. 7.6.2.3 (i) bedraagt:

= oppervlak van de paalvoet

 $q_{s;max;z} = \alpha_s * q_{c;z;a}$

De maximale schachtwrijvingskracht volgens art. 7.6.2.3 (c) bedraagt: $R_{s;cal;max;i} = \underset{0}{\overset{0}{\text{s}}; \Delta_{l;gem}} * \Sigma q_{s;max;z;i} * d_{z}$

Sweco Neder	land BV							Blad: 2
Technosoft Pa	alfunderingen relea	ase 6.70a					1	4 okt 2022
Project Onderdeel	: Pile I : Timbe:	Foundatior r Pile	n for	Timber	Colu	mns		
Per laag Alle niveaus/I	noogtes/peilmaten :	zijn t.o.v	7.: Ma	aiveld				_
Nr Laag		Nivo C [m])s;gen [m ¹]	$\alpha_{s} Pe$	erc. [%]	q _{c;z;a} [MPa]	q _{s;max} d _z [MPa] [m	R _{c;cal}] [kN]
 1 Leem - Zwai	 k zandig - Vast	-13.50 -13.70	 0.50	 0.0000	0	 5.94	0.000 0.2	0 0.0
totaal			0.50	0.0000		0.00	0.000-9.5	0 0.0
Maximale draa	gkracht							
De maximale d	raagkracht van de j	paal volge	ens ai	st. 7.6	.2.3	(c) bea	draagt:	
R _{c;cal;} i	= R _{b;cal;max;i} = 35 kN (= 35	- R _{s;cal;m} + 0)	ax;i					
De karakteris volgens art.	tieke waarde van de 7.6.2.3 (b) bedraad	e maximale gt:	e draa	igkrach	t van	de paa	al	
R _{c;k}	$= R_{c;cal} / \xi_{3}$ (n=	1)						
waarin : $\xi_{3 (n=1)} = fa$	actor volgens art.	A.3.3.3 k	oij 1	sonder	ing =	in di 1.39	t geval : -	
Voor de rekenv volgens art. 2	waarde van de maxin 2.4.7.3.3 worden aa	nale draag angehouder	gkrach n :	nt van o	de pa	al kan		
R _{c;d}	$= R_{C;k} / \gamma_{R}$ $= 21 kN$							

		= 21 KN			
waarin	:		in dit o	geval :	
γ _R		= partiële weerstandsfactor volgens art.	A.3.3.2		
		tabel A.6, A.7 of A.8	= 1.20 -	-	

Sweco Nederland BV Blad: 3 14 okt 2022 Technosoft Paalfunderingen release 6.70a Project : Pile Foundation for Timber Columns Onderdeel : Timber Pile DETAIL BER. NEGATIEVE KLEEF Case 1; Loc01_CPT00000052457;

Maaiveld-4.00 Uitgangspunten

-	gehanteerde sondering	:	Loc01 CPT00000052457
-	gehanteerde paal	:	Timber pile d=160
-	paalpuntniveau	:	Maaiveld -4.00 m
-	paalkopniveau	:	Maaiveld 0.30 m
-	traject negatieve kleef	:	Maaiveld 0.00 m
	tot	:	Maaiveld-13.50 m
-	P _{sur;k}	:	10.00 kN/m ²

Berekening negatieve kleef

De karakteristieke waarde van de maximale negatieve kleefbelasting v.e. alleenstaande paal volgens art. 7.3.2.2 (d) bedraagt:

F _{nk;k}	= $O_{s;gem} * \sum d_j * K_{0;j;k} * \tan \delta_{j;k} * (\sigma'_{v;j-1;k} + \sigma'_{v;j;k})/2.0$ = -16.6 kN
waarin :	
O _s ;gem	= omtrek van de dwarsdoorsnede van de paalschacht
di	= de dikte van de grondlaag i
K ₀ ;j;k	= de karakteristieke waarde van de neutrale gronddrukfactor in laag i
δ _{i;k}	= de karakteristieke waarde van de wrijvingshoek
σ' _{v;j;k}	<pre>= de karakteristieke waarde van de effectieve verticale spanning onder in laag j</pre>

Per laag

Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld

AL.	le niveaus/hoogtes/peilmaten zij	jn t.o.v	.: Maaıv	reid		
Nr	Laag	Nivo [m]	Hoogte [m]	O _{s;gem} [m ¹]	$K_{0;j}$ *tan(δ_i)	σ' _{v;j;k} [kN/m ²]
		0.00				10.00
1	Grind - Zwak siltig - Vast	-0.20	0.20	0.50	0.25	14.00
2	Grind - Zwak siltig - Vast	-0.32	0.12	0.50	0.25	15.44
3	Zand - Schoon - Vast	-0.46	0.14	0.50	0.25	17.12
4	Zand - Zwak siltig - Kleiig	-0.60	0.14	0.50	0.25	18.66
5	Zand - Sterk siltig - Kleiig	-0.74	0.14	0.50	0.25	20.20
6	Zand - Schoon - Los	-0.88	0.14	0.50	0.25	21.60
7	Leem - Zwak zandig - Vast	-1.08	0.20	0.50	0.27	24.00
8	Klei - Zwak zandig - Vast	-1.22	0.14	0.50	0.25	25.54
9	Klei - Zwak zandig - Matig	-1.34	0.12	0.50	0.25	26.74
10	Klei - Zwak zandig - Slap	-1.48	0.14	0.50	0.25	27.86
11	Klei - Schoon - Vast	-1.60	0.12	0.50	0.25	29.06
12	Klei - Schoon - Matig	-1.74	0.14	0.50	0.25	30.32
13	Klei - Zwak zandig - Slap	-1.88	0.14	0.50	0.25	31.44
14	Klei - Schoon - Matig	-2.00	0.12	0.50	0.25	32.52
15	Klei – Zwak zandig – Slap	-2.12	0.12	0.50	0.25	33.48
16	Klei - Zwak zandig - Matig	-2.26	0.14	0.50	0.25	34.88
17	Klei - Schoon - Matig	-2.40	0.14	0.50	0.25	36.14
18	Klei - Zwak zandig - Matig	-2.52	0.12	0.50	0.25	37.34
19	Klei - Schoon - Matig	-2.64	0.12	0.50	0.25	38.42

Sweco Nederland BV					Blad: 4			
Technosoft Paalfunderingen	14 okt 2022							
Project : P Onderdeel : T								
Per laag Alle niveaus/hoogtes/peilma Nr Laag	ten zijn t.o.v Nivo [m]	.: Maaiv Hoogte [m]	eld O _{s;gem} [m ¹]	K _{0;j} *tan(δ _i)	σ' _{v;j;k} [kN/m ²]			
20 Klei - Zwak zandig - Mat 21 Leem - Zwak zandig - Vas 22 Zand - Schoon - Los 23 Leem - Zwak zandig - Vas 24 Zand - Schoon - Los 25 Zand - Sterk siltig - Kl 26 Zand - Schoon - Los 27 Leem - Zwak zandig - Vas	ig -2.78 t -2.90 -3.22 t -3.36 -3.48 eiig -3.60 -3.72 t -4.00	0.14 0.12 0.32 0.14 0.12 0.12 0.12 0.28	0.50 0.50 0.50 0.50 0.50 0.50 0.50	0.25 0.27 0.25 0.27 0.25 0.25 0.25 0.25 0.25	39.82 41.26 44.46 46.14 47.34 48.66 49.86 53.22			
Rekenwaarde De rekenwaarde van de maximale negatieve kleefbelasting van een alleenstaande paal bedraagt : $F_{nk;d} = F_{nk;k} * \gamma_{f;nk} = -16.6 \text{ kN}$								
waarin : $\gamma_{f;nk}$ = belastingf (art. 7.3.	actor voor de 2.2 (b))	negatiev	e kleef 1	in dit geva 0 -	1 :			

Results Pile Foundation

Sweco Nederland BV	7					В	lad: 1
Technosoft Paalfunder	ingen release	6.70a				14 ok	t 2022
Project Onderdeel	: Pile Foun : Timber Pi	dation : le	for Timbe	er Columi	ns		
ALGEMENE GEGEVENS							
Project	: Pile Four	dation :	for Timbe	er Columi	ns		
Onderdeel	: Timber Pi	le					
Datum	: 05-10-202	2 nlnahu)(DreDrive	Curo do			
Destand	AB\Docume	enten\Af:	studeren	Technos	oft\		
	Draagcapa	citeit 1	Houten Pa	alfunde:	ring d160	0.pvw	
Berekeningstype	: Verticaal	. belast	e paal				
Alle niveaus/hoogtes/	peilmaten zijn	n t.o.v.	: Maaive	Ld			
Toegepaste normen	volgens Euro	ocode m	et Nede	rlandse	e NB		
Geotechniek EN 199	7-1:2004		AC:20	09			
NEN-EN	1 1997-1:2005		C1+A1	L:2013	NB:2	2016	
CRONDCOORTEN	97-1:2016		02:20)1/			
Nr. Omschrijving		γ_{k-1}	Yestikii	Φ ' _k , 1	Y	Yestikia	φ'
		[kN/m ³]	[kN/m ³]	[°]΄	[kN/m ³]	[kN/m ³]	<u>[°]</u>
1 Grind - Zwak silt	ig - Los	17.00	19.00	32.50	18.00	20.00	35.00
2 Grind - Zwak silt	ig - Matig	18.00	20.00	35.00	19.00	21.00	37.50
4 Grind - Sterk sil	tig - Los	18.00	20.00	30.00	19.00	22.00	32.50
5 Grind - Sterk sil	tig - Matig	19.00	21.00	32.50	20.00	22.00	35.00
6 Zand - Schoon - I	JOS	17.00	19.00	30.00	18.00	20.00	32.50
7 Zand - Schoon - M	latig Vast	18.00	20.00	32.50	19.00	21.00	35.00
9 Zand - Zwak silti	ast a - Kleija	18.00	20.00	27.00	20.00	22.00	32.50
10 Zand - Sterk silt	ig - Kleiig	18.00	20.00	25.00	19.00	21.00	30.00
11 Leem - Zwak zandi	.g - Matig	20.00	20.00	27.50	21.00	21.00	32.50
12 Leem - Zwak zandi	.g – Vast	21.00	21.00	27.50	22.00	22.00	35.00
13 Leem - Sterk zand 14 Klei - Schoon - M	lig Matio	17 00	19.00	27.50	20.00	20.00	35.00
15 Klei - Schoon - V	Vast	19.00	19.00	17.50	20.00	20.00	25.00
16 Klei - Zwak zandi	.g - Slap	15.00	15.00	22.50	18.00	18.00	22.50
17 Klei - Zwak zandi	.g - Matig	18.00	18.00	22.50	20.00	20.00	22.50
19 Klei - Sterk zandi	.g – vast lig	20.00	20.00	22.50	21.00	20.00	27.50
20 Klei - Organisch	- Slap	13.00	13.00	15.00	15.00	15.00	15.00
21 Klei – Organisch	- Matig	15.00	15.00	15.00	16.00	16.00	15.00
22 Veen - Niet voorb	elast - Slap	10.00	10.00	15.00	12.00	12.00	15.00
23 veen - Matig voor	belast - Matig	g 12.00	12.00	15.00	13.00	13.00	15.00
PAALGEGEVENS Timbe	er pile d=160	D					
Туре	: Hou	iten paal	l (consta	ant)			
Wijze van installeren	1 : Hei	.en					
Diameter	[m] : 0.1	.60					
Elasticiteitsmodulus Factor α (tabel 7.0	EC 7.1 : 0.0)))10 (zai	ndlagen:	voor kle	eilagen :	zie tabe [:]	17.d)
Factor α_{t} (tabel 7.c	EC 7.1) : 0.0	070 (zai	ndlagen;	voor kle	eilagen :	zie tabe	1 7.d)
Paalklassefactor $lpha_{ m p}$: 0.7	0			-		
Paalvoetvormfactor β	: 1.0	0	ingende -				
Verm.factor * m '	i Gro • 0 7	naverar: '5	ingenae p	Jadi			
торук торук		-					

Sweco Nederland BV Blad: 2 Technosoft Paalfunderingen release 6.70a 14 okt 2022 Project : Pile Foundation for Timber Columns Onderdeel : Timber Pile PAALGEGEVENS Timber pile d=200 Туре : Houten paal (constant) Wijze van installeren : Heien Diameter [m] : 0.200 Elasticiteitsmodulus [N/mm²] : 3600 (zandlagen; voor kleilagen zie tabel 7.d)

Factor α_s (tabel 7.c EC 7.1) : 0.010 (zandlagen; voor kleilagen zie tabel 7.d) Factor α_t (tabel 7.c EC 7.1) : 0.0070 (zandlagen; voor kleilagen zie tabel 7.d) Paalklassefactor α_p : 0.70 Paalvoetvormfactor β : 1.00 Type lastzakkingsdiagram : Grondverdringende paal Verm.factor * $\varphi'_{j;k}$: 0.75

PAALGEGEVENS Concrete pile #400

Sweco Nederland BV Blad: 3 Technosoft Paalfunderingen release 6.70a 14 okt 2022 Project : Pile Foundation for Timber Columns Onderdeel : Timber Pile REKENGEGEVENS Case 1 : Ontwerpend : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc01_CPT000000052457 Berekening Rekenmethode Sondering(en) Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1): 1.39Factor ξ_4 (min): 1.39Factor ξ_4 (min): 1.39Weerstandsfactor γ_R : 1.20 $\gamma_{f;nk}$: 1.0 $\gamma_{f;nk}$: 0.75 : NEE Paal : Timber pile d=160 Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -4.00 Bovenbel. [kN/m²] : -10.00

Sweco Nederland	i bv							Blad: 4
Technosoft Paalfu	nderinge	n releas	e 6.70a				14	okt 2022
Project Onderdeel	:	Pile Fo Timber	undation Pile	for Tim	ber Colum	ns		
SAMENVATTINGST	ABEL Ca	se 1 (n	=1)					
Uitgangspunten								
- paal		: Т	imber pi	le d=160				
- paaltype		: H	outen pa	al (cons	tant)			
- schachtafmeting		: 1	60 mm					
Paalklassefactor	$\alpha_{\rm p}$: 0	.70					
Factor $lpha_{ m s}$ (tabel	7.c ⁻ EC 7	.1) : 0	.010 (z	andlagen	; voor kl	eilagen	zie tabe	1 7.d)
Correlatiefactor	ξ3(n=1)	: 1	.39					
Alle niveaus/hoog	tes/peil	maten zi	jn t.o.v	.: Maaiv	eld			
ma	aiveld p	aalpunt	Bezwi	jkdraagv	ermogen	Rekenv	waarden	
sondering	niveau	niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R _c [kN]	;netto;d [kN]
Loc01_CPT0000000	0.00	-4.00	35.4	0.0	35.4	21.2	-16.6	4.6

REKENGEGEVENS Case 2

Berekening Rekenmethode Sondering(en)	: Ontwerpend : Drukpalen volgens NEN-EN : Loc02_CPT000000003023	1997-1, art. 7.6.2
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 op 0.75 * R _{b;cal;max;i} : er negatieve kleef :	NEE NEE
Paal Niveau paalkop [m] Paalpuntniveau Bovenbel. [kN/m²]	: Timber pile d=160 : Maaiveld 0.30 : Maaiveld -20.70 : -10.00	

 Sweco Nederland BV
 Blad: 5

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

SAMENVATTINGSTABEL Case 2 (n=1)
Uitgangspunten

- paal : Timber pile d=160 - paaltype : Houten paal (constant) - schachtafmeting : 160 mm Paalklassefactor α_p : 0.70 Factor α_s (tabel 7.c EC 7.1) : 0.010 (zandlagen; voor kleilagen zie tabel 7.d) Correlatiefactor $\xi_{3(n=1)}$: 1.39 Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden sondering niveau niveau $R_{b;cal}$ $R_{c;cal}$ $R_{c;cal}$ $R_{c;d}$ $F_{nk;d}$ $R_{c;netto;d}$

			[kN]						
Loc02_CPT0000000	0.00	-20.70	301.6	164.4	466.0	279.4	-188.1	91.2	

REKENGEGEVENS Case 3

: Ontwerpend Berekening Rekenmethode : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc03_CPT000000074906 Sondering(en) : NEE : NEE Stijf bouwwerk Paalgroep Aantal sonderingen : 1 $\begin{array}{c|c} \mbox{Aantal songeringen} & \cdot \\ \mbox{Factor} & \mbox{\xi}_3 & (n=1) \\ \mbox{Factor} & \mbox{\xi}_3 & (gem) \\ \mbox{Factor} & \mbox{\xi}_4 & (min) \\ \mbox{Weerstandsfactor} & \mbox{\gamma}_R & : \\ \mbox{\gamma}_{f\cdot nk} & : \end{array}$ 1.39 1.39 1.39 1.20 1.20 $\begin{array}{rll} \gamma_{\text{f,nk}} & : & 1.0 \\ R_{\text{s,cal,max,i}} & \text{begrenzen op } 0.75 \ ^{\text{K}}\text{R}_{\text{b,cal,max,i}} & : \text{NEE} \\ \text{UGT draagvermogen zonder negatieve kleef} & : \text{NEE} \end{array}$: Timber pile d=160 Paal Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -5.90 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

Technosoft Paalfunderingen release 6.70a 14 okt Project : Pile Foundation for Timber Columns Onderdeel : Timber Pile SAMENVATTINGSTABEL Case 3 (n=1) Uitgangspunten - paal : Timber pile d=160 - paaltype : Houten paal (constant) - schachtafmeting : 160 mm Paalklassefactor α_p : 0.70 Factor α_s (tabel 7.c EC 7.1) : 0.010 (zandlagen; voor kleilagen zie tabel 7.c Correlatiefactor $\xi_{3(n=1)}$: 1.39 Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden sondering niveau niveau $R_{b, cal} R_{c; cal} R_{c; cal} R_{c; d} F_{nk; d} R_{c; net}$ [kN] [kN] [kN] [kN] [kN] [kN] [kN] [kN]	Sweco Nederland BV	Blad:
$\begin{array}{cccc} \mbox{Project} & : \mbox{Pile Foundation for Timber Columns} \\ \mbox{Onderdeel} & : \mbox{Timber Pile} \\ \mbox{SAMENVATTINGSTABEL Case 3 (n=1)} \\ \mbox{Uitgangsputen} & & & & & & & & & & & & & & & & & & &$	Technosoft Paalfunderingen release 6.70a	14 okt 2022
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Project : Pile Foundation for Timber Co Onderdeel : Timber Pile	blumns
$\label{eq:spin} \begin{array}{cccc} \textbf{Uitgangspunten} & & & & \\ \textbf{-} paal & & : \end{tabular} Timber pile d=160 & & \\ \textbf{-} paaltype & : \end{tabular} Houten paal (constant) & \\ \textbf{-} schachtafmeting & : 160 mm & & \\ \texttt{Paalklassefactor} & \textbf{a}_p & : \end{tabular} 0.70 & & \\ \texttt{Factor} & \textbf{a}_s & (tabel \end{tabular} 1.c \end{tabular} C \end{tabular} 1.39 & & \\ \texttt{Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld} & & \\ & \end{tabular} \begin{array}{c} \texttt{maaiveld paalpunt} & \texttt{Bezwijkdraagvermogen} & \end{tabular} \end{tabular} \begin{array}{c} \texttt{Rekenwaarden} & & \\ \texttt{sondering} & & \texttt{niveau} & \texttt{niveau} & \texttt{R}_{b}, \texttt{cal} & \texttt{R}_{c}, \texttt{cal} & cal$	SAMENVATTINGSTABEL Case 3 (n=1)	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Uitgangspunten	
Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden sondering niveau niveau R _{b;cal} R _{s;cal} R _{c;cal} R _{c;d} F _{nk;d} R _{c;net} [kN] [kN] [kN] [kN] [kN] [kN] [kN] [kN]	$\begin{array}{llllllllllllllllllllllllllllllllllll$	c kleilagen zie tabel 7.d)
sondering niveau niveau $R_{b,cal} R_{s,cal} R_{c,cal} R_{c,cal} R_{c,d} F_{nk,d} R_{c,net}$ [kN] [kN] [kN] [kN] [kN] [kN] [kN] [kN]	Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermoge	en Rekenwaarden
	sondering niveau niveau R _{b;cal} R _{s;cal} R _{c;ca} R _{c;ca} [kN] [kN] [kN] [kN] [kN] [kN] [kN] [kN]	l R _{c;d} F _{nk;d} R _{c;netto;d} N] [kN] [kN] [kN]
Loc03_CPT0000000 0.00 -5.90 81.6 76.8 158.3 94.9 -9.5 85	Loc03_CPT0000000 0.00 -5.90 81.6 76.8 158.	.3 94.9 -9.5 85.4

Berekening: OntwerpendRekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Loc04_CPT000000055844Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1):1.39Factor ξ_3 (gem):Factor ξ_4 (min):1.39Weerstandsfactor γ_R : 1.20 $\gamma_{f;nk}$: 1.0 $R_{s;cal;max;i}$ begrenzen op 0.75 * $R_{b;cal;max;i}$: NEEUGT draagvermogen zonder negatieve kleef: NEEPaal: Timber pile d=160Niveau paalkop[m] : MaaiveldNiveau: MaaiveldBovenbel.[kN/m²] : -10.00

 Sweco Nederland BV
 Blad: 7

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

 SAMENVATTINGSTABEL Case 4 (n=1)

 Uitgangspunten

			נעזען	נעוען	[עוא]	נעזען	[\[\[\]	[עדע]	
Loc04_CPT0000000	0.00	-8.00	281.5	147.7	429.2	257.3	0.0	257.3	

REKENGEGEVENS Case 5

Berekening: OntwerpendRekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Loc05_CPT000000161652Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (gem): ξ_3 (gem):1.39Factor ξ_4 (min): ξ_4 (min):1.39Weerstandsfactor γ_R : 1.20 $\gamma_{f;nk}$: 1.0 $R_{s;cal;max;i}$ begrenzen op 0.75 * $R_{b;cal;max;i}$: NEEUGT draagvermogen zonder negatieve kleef: NEEPaal: Timber pile d=160Niveau paalkop [m]: Maaiveld 0.30Paalpuntniveau: Maaiveld -14.30Bovenbel.[kN/m²]:-10.00

Sweco Nederland H	BV						Blad: 8
Technosoft Paalfunde	eringen relea	se 6.70a				14	l okt 2022
Project Onderdeel	: Pile F : Timber	oundation Pile	for Tim	ber Colu	mns		
SAMENVATTINGSTABL	EL Case 5 (n=1)					
Uitgangspunten - paal - paaltype - schachtafmeting Paalklassefactor Factor α_s (tabel 7.0 Correlatiefactor ξ_3	: x _p :: cEC 7.1): (n=1):	Timber pi Houten pa 160 mm 0.70 0.010 (z 1.39	le d=160 al (cons andlagen	tant) ; voor k	leilagen	zie tabe	el 7.d)
Alle niveaus/hoogtes maair	s/peilmaten z veld paalpunt	ijn t.o.v Bezwi	.: Maaiv jkdraagv	eld ermogen	Reken	waarden	
sondering niv	veau niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R [kN]	c;netto;d [kN]
Loc05 CPT0000001 (0.00 -14.30	301.6	503.5	805.1	482.7	-25.9	456.8
REKENGEGEVENS Cas	se 6						

Berekening: OntwerpendRekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Loc06_CPT000000170226Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1):1.39Factor ξ_3 (gem):Factor ξ_4 (min):1.39Weerstandsfactor γ_R : 1.20 $\gamma_{f,nk}$: 1.0 $R_{s,cal,max,i}$ begrenzen op 0.75 * $R_{b,cal,max,i}$: NEEUGT draagvermogen zonder negatieve kleef: NEEPaal: Timber pile d=160Niveau paalkop[m] : Maaiveld0.30PaalpuntniveauBovenbel.[kN/m²] : -10.00

 Sweco Nederland BV
 Blad: 9

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

 SAMENVATTINGSTABEL Case 6 (n=1)
 Itigangspunten

 - paal
 : Timber pile d=160

 - paaltype
 : Houten paal (constant)

sondering	niveau	niveau	^K b;cal [kN]	^K s;cal [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	^r nk;d [kN]	^K c;netto;d [kN]
Loc06_CPT0000001	0.00	-19.30	178.6	919.9	1098.5	658.6	0.0	658.6

REKENGEGEVENS Case 7

Berekening: OntwerpendRekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Loc07_CPT000000004530Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1):1.39Factor ξ_3 (gem):Factor ξ_4 (min):1.39Weerstandsfactor γ_R : 1.20 $\gamma_{f,nk}$: 1.0 $R_{s,cal,max,i}$ begrenzen op 0.75 * $R_{b,cal,max,i}$: NEEUGT draagvermogen zonder negatieve kleef: NEEPaal: Timber pile d=160Niveau paalkop[m] : MaaiveldNiveau: MaaiveldBovenbel.[kN/m²] : -10.00

Sweco Nederland	BV						Blad: 10
Technosoft Paalfund	eringen releas	e 6.70a				14	okt 2022
Project Onderdeel	: Pile Fo : Timber	undation Pile	for Tim	ber Colu	mns		
SAMENVATTINGSTABEL Case 7 (n=1)							
Uitgangspunten - paal - paaltype - schachtafmeting Paalklassefactor Factor α_s (tabel 7. Correlatiefactor ξ_3	: T : H ap : 0 c EC 7.1) : 0 (n=1) : 1	imber pi outen pa 60 mm .70 .010 (z .39	le d=160 al (cons andlagen	tant) ; voor k	leilagen	zie tabe	1 7.d)
Alle niveaus/hoogte maai	s/peilmaten zi veld paalpunt	jn t.o.v Bezwi	.: Maaiv jkdraagv	eld ermogen	Rekenv	waarden	
sondering ni	veau niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R [kN]	;netto;d [kN]
Loc07_CPT0000000	0.00 -14.50	251.1	562.7	813.9	487.9	0.0	487.9

REKENGEGEVENS Case 8

Berekening	: Ontwerpend
Rekenmethode	: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2
Sondering(en)	: Loc08_CPT000000062979
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 a op 0.75 * R _{b;cal;max;i} : NEE Her negatieve kleef : NEE
Paal	: Timber pile d=160
Niveau paalkop [m]	: Maaiveld 0.30
Paalpuntniveau	: Maaiveld -2.00
Bovenbel. [kN/m²]	: -10.00

 Sweco Nederland BV
 Blad: 11

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

 SAMENVATTINGSTABEL Case 8 (n=1)

Uitgangspunten - paal : Timber pile d=160 - paaltype : Houten paal (constant) - schachtafmeting : 160 mm (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld aaiveld paalpunt ______ niveau niveau R_{b;cal} R_{s;cal} ______ [kN] [kN] maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden sondering R_{c;cal} [kN] F_{nk;d} R_{c;netto;d} [kN] [kN] R_{c;d} [kN] 0.00 Loc08 CPT0000000 -2.00 3.5 0.0 3.5 2.1 -5.5 -3.4

REKENGEGEVENS Case 9

: Ontwerpend Berekening Rekenmethode : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc09_CPT000000011161 Sondering(en) : NEE Stijf bouwwerk Paalgroep : NEE Aantal sonderingen : 1 $\begin{array}{c|c} \mbox{Aantal songeringen} & \cdot \\ \mbox{Factor} & \mbox{\xi}_3 & (n=1) \\ \mbox{Factor} & \mbox{\xi}_3 & (gem) \\ \mbox{Factor} & \mbox{\xi}_4 & (min) \\ \mbox{Weerstandsfactor} & \mbox{\gamma}_R & \\ \mbox{\gamma}_{\text{f.nk}} & \mbox{i} \end{array}$ 1.39 1.39 1.39 1.39 1.20 $\begin{array}{rll} \gamma_{\text{f,nk}} & : & 1.0 \\ R_{\text{s,cal,max,i}} & \text{begrenzen op } 0.75 \ ^{\text{K}}\text{R}_{\text{b,cal,max,i}} & : \text{NEE} \\ \text{UGT draagvermogen zonder negatieve kleef} & : \text{NEE} \end{array}$ Paal: Timber pile d=160Niveau paalkop[m]: Maaiveld 0.30Paalpuntniveau: Maaiveld -15.70 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

Sweco Nederlar	nd BV							Blad:	12
Technosoft Paalfunderingen release 6.70a 14					okt 2	022			
Project Onderdeel		: Pile Fo : Timber	undatior Pile	n for Tim	ber Colu	mns			
SAMENVATTINGSTABEL Case 9 (n=1)									
Uitgangspunten									
- paal		: T	imber pi	le d=160)				
- paaltype		: H	outen pa	al (cons	stant)				
- schachtaimetin	g	: 1	60 mm						
Paalklassefactor	$\alpha_{ m p}$: 0	. 70						
Factor $lpha_{ m s}$ (tabel	7.c EC	7.1) : 0	.010 (z	andlagen	ı; voor k	leilagen	zie tabe	1 7.d)	
Correlatiefactor	ξ _{3 (n=1)}	: 1	.39						
Alle niveaus/hoo	gtes/pei	lmaten zi	jn t.o.v	.: Maaiv	reld				
m	aaiveld	paalpunt	Bezwi	jkdraagv	vermogen	Reken	waarden		
sondering	niveau	niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R _c [kN]	;netto [kN]	o;d
Loc09_CPT0000000	0.00	-15.70	212.6	327.0	539.6	323.5	-35.9	287.7	

REKENGEGEVENS Case 10

Berekening Rekenmethode Sondering(en)	: Ontwerpend : Drukpalen volgens NEN-EN 1997-1, art. : Loc10_CPT00000004377	7.6.2
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 a op 0.75 * R _{b;cal;max;i} : NEE Her negatieve kleef : NEE	
Paal Niveau paalkop [m] Paalpuntniveau Bovenbel. [kN/m²]	: Timber pile d=160 : Maaiveld 0.30 : Maaiveld -20.10 : -10.00	

 Sweco Nederland BV
 Blad: 13

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

 SAMENVATTINGSTABEL Case 10 (n=1)

Uitgangspunten - paal : Timber pile d=160 - paaltype : Houten paal (constant) - schachtafmeting : 160 mm (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden niveau niveau R_{b;cal} R_{s;cal} R_{c;cal} [kN] [kN] [kN] sondering R_{c;d} F_{nk;d} R_{c;netto;d} [kN] [kN] [kN]

249.0

528.8

317.0

-156.8

160.2

279.8

REKENGEGEVENS Case 11

Loc10_CPT0000000 0.00

: Ontwerpend Berekening Rekenmethode : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc11_CPT000000012135 Sondering(en) : NEE : NEE Stijf bouwwerk Paalgroep Aantal sonderingen : 1 $\begin{array}{c|c} \text{Aantal songeringen} \\ \text{Factor} & \xi_3 & (n=1) \\ \text{Factor} & \xi_3 & (gem) \\ \text{Factor} & \xi_4 & (min) \\ \text{Weerstandsfactor} & \gamma_R \\ \gamma_{\text{f.nk}} & \vdots \end{array}$ 1.39 1.39 1.39 1.20 1.20 $\begin{array}{rll} \gamma_{\text{f,nk}} & : & 1.0 \\ R_{\text{s,cal,max,i}} & \text{begrenzen op } 0.75 \ ^{\text{K}}\text{R}_{\text{b,cal,max,i}} & : \text{NEE} \\ \text{UGT draagvermogen zonder negatieve kleef} & : \text{NEE} \end{array}$ Paal: Timber pile d=160Niveau paalkop[m]: Maaiveld 0.30Paalpuntniveau: Maaiveld -14.20 Paalpuntniveau : Maaiveld Bovenbel. [kN/m²] : -10.00

-20.10
Sweco Nederlar	nd BV							Blac	d: 14
Technosoft Paalfunderingen release 6.70a 14 d						okt	2022		
Project Onderdeel		: Pile Fo : Timber	oundatior Pile	n for Tim	ber Colu	mns			
SAMENVATTINGS	TABEL C	ase 11 ((n=1)						
Uitgangspunten									
- paal		: 1	'imber pi	le d=160					
- paaltype		: H	louten pa	aal (cons	tant)				
 schachtafmetin 	g	: 1	.60 mm						
Paalklassefactor	$lpha_{ m p}$: 0).70						
Factor $\alpha_{ m s}$ (tabel	7.c [°] EC	7.1) : 0).010 (2	andlagen	; voor k	leilagen	zie tabe	l 7.d	l)
Correlatiefactor	ξ _{3 (n=1)}	: 1	.39						
Alle niveaus/hoo	gtes/pei	lmaten zi	.jn t.o.v Bezwi	v.: Maaiv	eld	Rekent	waarden		
sondering	niveau	niveau	R	R .	R	R	F B		
	veau	niveau	'`b;cal [kN]	'`s;cal [kN]	'`c;cal [kN]	'`c;d [kN]	'nk;d 'o [kN]	;net [kN	to;d]]
Loc11_CPT0000000	0.00	-14.20	138.6	404.5	543.1	325.6	0.0	325.	6

Berekening	: Ontwerpend
Rekenmethode	: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2
Sondering(en)	: Loc12_CPT000000077437
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 op 0.75 * R _{b;cal;max;i} : NEE er negatieve kleef : NEE
Paal	: Timber pile d=160
Niveau paalkop [m]	: Maaiveld 0.30
Paalpuntniveau	: Maaiveld -9.30
Bovenbel. [kN/m²]	: -10.00

Sweco Nederland BV							Blac	a: 15
Technosoft Paalfunderin	gen releas	lease 6.70a 14						2022
Project Onderdeel	: Pile Foundation for Timber Columns : Timber Pile							
SAMENVATTINGSTABEL (Case 12 (n=1)						
<pre>Uitgangspunten - paal - paaltype - schachtafmeting Paalklassefactor α_p Factor α_s (tabel 7.c EC Correlatiefactor ξ_{2/c-1})</pre>	: 1 : H : 1 : C 7.1) : C	imber pi louten pa 60 mm .70 .010 (z .39	le d=160 aal (cons andlagen	tant) ; voor ki	leilagen	zie tabe	1 7.c	1)
Alle niveaus/hoogtes/pe maaiveld sondering niveau	' ilmaten zi paalpunt niveau	jn t.o.v Bezwi R _{b;cal} [kN]	v.: Maaiv jkdraagv R _{s;cal} [kN]	eld ermogen ^R c;cal [kN]	Reken [,] R _{c;d} [kN]	waarden F _{nk;d} R _o [kN]	;net	to;d]]
Loc12 CPT000000 0.00	-9 30	75 2	102 7	178 0	106 7	-14 9	91	8

Sweco Nederlan	d BV						Blad: 16
Technosoft Paalf	Technosoft Paalfunderingen release 6.70a						
Project Onderdeel	oject : Pile Foundation for Timber Columns derdeel : Timber Pile						
OVERZICHT NETT	O DRAAGV	ERMOGEN	DRUKPALEN				
Netto paaldraagv Alle niveaus/hoo	ermogen(s) gtes/peilm	zijn naa maten zijn	r beneden toe t.o.v.: Maai	e afgerond veld	op: 1.0	kN nauwk	eurig
sondering	niveau	niveau	^R c;netto;d Case 1	Case 2	Case 3	Case 4	Case !
Loc01_CPT0000000	0.00	-4.00	4				
Loc03_CPT0000000	0.00	-5.90			85		
Loc04_CPT0000000	0.00	-8.00				257	
Loc02_CPT0000000	0.00	-20.70		91			
Loc05_CPT0000001	0.00	-14.30					45

Sweco Nederlan	В	lad: 17					
Technosoft Paalfunderingen release 6.70a							kt 2022
Project Onderdeel	:	Pile Foun Timber Pi	dation for Ti le	mber Colu	umns		
Netto paaldraagve Alle niveaus/hoog	ermogen(s gtes/peil maaiveld) zijn naa maten zijn paalpunt	r beneden toe t.o.v.: Maai R	e afgerond veld [kN]	l op: 1.0	kN nauwke	urig
sondering	niveau	niveau	Case 6	Case 7	Case 8	Case 9	Case 10
Loc06_CPT0000001	0.00	-19.30	658				
Loc08_CPT0000000	0.00	-2.00			-4		
Loc07_CPT0000000	0.00	-14.50		487			
Loc09_CPT0000000	0.00	-15.70				287	
Loc10_CPT0000000	0.00	-20.10					160

Sweco Nederlan	d BV			Blad: 18		
Technosoft Paalfu	Technosoft Paalfunderingen release 6.70a					
Project Onderdeel	oject : Pile Foundation for Timber Columns derdeel : Timber Pile					
Netto paaldraagve Alle niveaus/hoog	kN nauwkeurig					
sondering	niveau	niveau	Case 11 Case 12			
Loc12_CPT000000	0.00	-9.30	91			
Loc11_CPT0000000	0.00	-14.20	325			

Sweco Nederland BV Blad: 21 Technosoft Paalfunderingen release 6.70a 14 okt 2022 Project : Pile Foundation for Timber Columns Onderdeel : Timber Pile REKENGEGEVENS Case 1 : Ontwerpend : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc01_CPT000000052457 Berekening Rekenmethode Sondering(en) Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1): 1.39Factor ξ_4 (min): 1.39Factor ξ_4 (min): 1.39Weerstandsfactor γ_R : 1.20 $\gamma_{f;nk}$: 1.0 $\gamma_{f;nk}$: 0.75 : NEE Paal : Timber pile d=200 Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -4.00 Bovenbel. [kN/m²] : -10.00

Sweco Nederland	l BV							Blad	1: 22
Technosoft Paalfu	nderinge	n releas	ease 6.70a					okt	2022
Project Onderdeel	:	Pile For Timber 1	undation Pile	for Tim	ber Colu	umns			
SAMENVATTINGST	BEL Ca	se 1 (n:	=1)						
Uitgangspunten									
- paal		: T:	imber pi	le d=200					
- paaltype		: He	outen pa	al (cons	tant)				
 schachtafmeting 		: 21	DO mm						
Paalklassefactor	α_{p}	: 0	.70						
Factor $lpha_{ m s}$ (tabel '	7.c [°] EC 7	.1) : 0	.010 (z	andlagen	; voor }	kleilagen	zie tabe	l 7.d	l)
Correlatiefactor &	53 (n=1)	: 1	.39						
Alle niveaus/hoog	tes/peil	maten zi	jn t.o.v	.: Maaiv	reld				
maa	aiveld p	aalpunt	Bezwi	jkdraagv	rermogen	Rekenv	waarden		
sondering	niveau	niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R _c [kN]	;net [kN	to;d []
Loc01 CPT0000000	0.00	-4.00	52.3	0.0	52.3	31.4	-20.7	10.	6

Berekening	: Ontwerpend
Rekenmethode	: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2
Sondering(en)	: Loc02_CPT00000003023
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 : op 0.75 * R _{b;cal;max;i} : NEE Her negatieve kleef : NEE
Paal	: Timber pile d=200
Niveau paalkop [m]	: Maaiveld 0.30
Paalpuntniveau	: Maaiveld -20.60
Bovenbel. [kN/m²]	: -10.00

 Sweco Nederland BV
 Blad: 23

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

SAMENVATTINGSTABEL Case 2 (n=1)

Uitgangspunten - paal : Timber pile d=200 - paaltype : Houten paal (constant) - schachtafmeting : 200 mm (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden aaiveld paarpunt Bears, and R_{c;cal} R_{c;cal} R_{c;cal} R_{kN}] [kN] [kN] [kN] sondering F_{nk;d} R_{c;netto;d} [kN] [kN] R_{c;d} [kN] Loc02_CPT0000000 0.00 -20.60 471.2 205.5 676.7 405.7 -235.1 170.6

REKENGEGEVENS Case 3

: Ontwerpend Berekening : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc03_CPT000000074906 Rekenmethode Sondering(en) : NEE : NEE Stijf bouwwerk Paalgroep Aantal sonderingen : 1 $\begin{array}{c} \text{Aantal songeringen} \\ \text{Factor} & \xi_3 & (n=1) \\ \text{Factor} & \xi_3 & (gem) \\ \text{Factor} & \xi_4 & (min) \\ \text{Weerstandsfactor} & \gamma_R & \\ \gamma_{\text{f.nk}} & \end{array}$ 1.39 1.39 1.39 1.39 1.20 $\begin{array}{rrrr} & & & 1.0 \\ R_{s;cal;max;i} & begrenzen op 0.75 * R_{b;cal;max;i} & : NEE \\ UGT & draagvermogen & zonder & negatieve & kleef & : NEE \end{array}$: Timber pile d=200 Paal Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -5.70 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

Sweco Nederland BV		Blad: 24
Technosoft Paalfunderi	ngen release 6.70a 14	okt 2022
Project Onderdeel	: Pile Foundation for Timber Columns : Timber Pile	
SAMENVATTINGSTABEL	Case 3 (n=1)	
Uitgangspunten		
- paal	: Timber pile d=200	
- paaltype	: Houten paal (constant)	
 schachtafmeting 	: 200 mm	
Paalklassefactor $lpha_{ m p}$: 0.70	
Factor $\alpha_{\rm s}$ (tabel 7.c E	C 7.1) : 0.010 (zandlagen; voor kleilagen zie tabel	. 7.d)
Correlatiefactor $\xi_{3(n=1)}$: 1.39	
Alle niveaus/hoogtes/p	eilmaten zijn t.o.v.: Maaiveld	
maaivel	d paalpunt Bezwijkdraagvermogen Rekenwaarden	
sondering nivea	ı niveau R _{b;cal} R _{s;cal} R _{c;cal} R _{c;d} F _{nk;d} R _c [kN] [kN] [kN] [kN] [kN] [kN]	;netto;d [kN]
Loc03_CPT0000000 0.0	0 -5.70 116.0 89.0 205.1 122.9 -11.9	111.0
REKENGEGEVENS Case	4	
Berekening	: Ontwerpend	

Rekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Locod_CPT00000055844Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1):1.39Factor ξ_3 (gem):1.39Factor ξ_4 (min):1.39Weerstandsfactor γ_R 1.00 $R_{s,cal,max,i}$ begrenzen op 0.75 * $R_{b,cal,max,i}$: NEEUGT draagvermogen zonder negatieve kleefPaal:Timber pile d=200Niveau paalkop[m] : MaaiveldPaalpuntniveau:Bovenbel.[kN/m²] : -10.00

 Sweco Nederland BV
 Blad: 25

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

 SAMENVATTINGSTABEL Case 4 (n=1)

Uitgangspunten - paal : Timber pile d=200 - paaltype : Houten paal (constant) - schachtafmeting : 200 mm (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden aaiveld paarpunt Bearlynne R_{s;cal} R_{c;cal} R_{c;cal} R_{c;cal} [kN] [kN] [kN] sondering F_{nk;d} R_{c;netto;d} [kN] [kN] R_{c;d} [kN]

184.6

624.4

374.4

0.0

374.4

439.8

REKENGEGEVENS Case 5

Loc04_CPT0000000 0.00

: Ontwerpend Berekening Rekenmethode : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc05_CPT000000161652 Sondering(en) : NEE : NEE Stijf bouwwerk Paalgroep Aantal sonderingen : 1 $\begin{array}{c|c} \mbox{Aantal songeringen} & \cdot \\ \mbox{Factor} & \mbox{\xi}_3 & (n=1) \\ \mbox{Factor} & \mbox{\xi}_3 & (gem) \\ \mbox{Factor} & \mbox{\xi}_4 & (min) \\ \mbox{Weerstandsfactor} & \mbox{\gamma}_R & : \\ \mbox{\gamma}_{f\cdot n\,k} & : \end{array}$ 1.39 1.39 1.39 1.39 1.20 $\begin{array}{rrrr} & & & 1.0 \\ R_{s;cal;max;i} & begrenzen op 0.75 * R_{b;cal;max;i} & : NEE \\ UGT & draagvermogen & zonder & negatieve & kleef & : NEE \end{array}$ Paal: Timber pile d=200Niveau paalkop[m]: Maaiveld 0.30Paalpuntniveau: Maaiveld -14.10 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

-7.80

Sweco Nederland	BV						Blad: 26
Technosoft Paalfunderingen release 6.70a							okt 2022
Project Onderdeel	: Pile Fo : Timber	undation Pile	for Tim	ber Col	umns		
SAMENVATTINGSTAB	EL Case 5 (n	=1)					
$\begin{array}{l} \textbf{Uitgangspunten} \\ - \text{ paal} \\ - \text{ paaltype} \\ - \text{ schachtafmeting} \\ \text{Paalklassefactor} \\ \text{Factor } \alpha_{s} \text{ (tabel 7.} \\ \text{Correlatiefactor } \xi_{3} \end{array}$: T : H 2 2 c EC 7.1) : 0 (n=1) : 1	imber pi outen pa 00 mm .70 .010 (z .39	le d=200 al (cons andlagen	tant) ; voor	kleilagen	zie tabe	1 7.d)
Alle niveaus/hoogte maai sondering ni	s/peilmaten zi veld paalpunt veau niveau	jn t.o.v Bezwi R _{b;cal}	.: Maaiv jkdraagv R _{s.cal}	eld ermogen ^R c;cal	Rekenv R _{c;d}	vaarden F _{nk;d} R	c;netto;d
Loc05 CPT0000001	0.00 -14.10	[kN] 471.2	[kN] 610.5	[kN] 1081.8	[kN] 648.5	[kN] -32.4	[kN] 616.2

Berekening	: Ontwerpend
Rekenmethode	: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2
Sondering(en)	: Loc06_CPT000000170226
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 op 0.75 * R _{b;cal;max;i} : NEE er negatieve kleef : NEE
Paal	: Timber pile d=200
Niveau paalkop [m]	: Maaiveld 0.30
Paalpuntniveau	: Maaiveld -19.10
Bovenbel. [kN/m²]	: -10.00

 Sweco Nederland BV
 Blad: 27

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

 SAMENVATTINGSTABEL Case 6 (n=1)

 Uitgangspunten

			[KN]	[KN]	[KN]	[KN]	[KN]	[KN]	
Loc06_CPT0000001	0.00	-19.10	283.1	1134.6	1417.8	850.0	0.0	850.0	

REKENGEGEVENS Case 7

Berekening: OntwerpendRekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Loc07_CPT000000004530Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (gem): ξ_3 (gem):1.39Factor ξ_4 (min): f_{nn} : 1.0 $\chi_{f,nk}$: 1.0UGT draagvermogen zonder negatieve kleef: NEEPaal: Timber pile d=200Niveau paalkop[m] : MaaiveldNiveau: MaaiveldBovenbel.[kN/m²] : -10.00

Sweco Nederlan	d BV							Blad: 28
Technosoft Paalfu	underinge	n releas	e 6.70a				14	okt 2022
Project Onderdeel	:	Pile Fo Timber	undation Pile	for Tim	ber Colu	mns		
SAMENVATTINGST	ABEL Ca	se 7 (n	=1)					
Uitgangspunten								
- paal		: T	imber pi	le d=200				
- paaltype		: Н	outen pa	al (cons	tant)			
- schachtafmeting	ſ	: 2	00 mm					
Paalklassefactor	$\alpha_{\rm p}$: 0	.70					
Factor α_{\circ} (tabel	7.c EC 7	.1) : 0	.010 (z	andlagen	; voor k	leilagen	zie tabe	1 7.d)
Correlatiefactor	ξ _{3 (n=1)}	: 1	.39	2		2		
Alle niveaus/hoog	gtes/peil	maten zi	jn t.o.v	.: Maaiv	eld			
ma	aiveld p	aalpunt	Bezwi	jkdraagv	ermogen	Rekenw	vaarden	
sondering	niveau	niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R [kN]	c;netto;d [kN]
Loc07_CPT0000000	0.00	-14.30	334.4	684.6	1019.0	610.9	0.0	610.9

Berekening Rekenmethode Sondering(en)	: Ontwerpend : Drukpalen volgens NEN-EN 1997-1, art. 7.6 : Loc08_CPT000000062979	.2
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 a op 0.75 * R _{b;cal;max;i} : NEE der negatieve kleef : NEE	
Paal Niveau paalkop [m] Paalpuntniveau Bovenbel. [kN/m²]	: Timber pile d=200 : Maaiveld 0.30 : Maaiveld -2.00 : -10.00	

 Sweco Nederland BV
 Blad: 29

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

SAMENVATTINGSTABEL Case 8 (n=1)

Uitgangspunten - paal : Timber pile d=200 - paaltype : Houten paal (constant) - schachtafmeting : 200 mm (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld aaiveld paalpunt ______ niveau niveau R_{b;cal} R_{s;cal} ______ [kN] [kN] maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden sondering R_{c;cal} [kN] F_{nk;d} R_{c;netto;d} [kN] [kN] R_{c;d} [kN] 0.00 Loc08 CPT0000000 -2.00 5.3 0.0 5.3 3.2 -6.8 -3.6

REKENGEGEVENS Case 9

: Ontwerpend Berekening Rekenmethode : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc09_CPT000000011161 Sondering(en) : NEE Stijf bouwwerk Paalgroep : NEE Aantal sonderingen : 1 $\begin{array}{c|c} \mbox{Aantal songeringen} & \cdot \\ \mbox{Factor} & \mbox{\xi}_3 & (n=1) \\ \mbox{Factor} & \mbox{\xi}_3 & (gem) \\ \mbox{Factor} & \mbox{\xi}_4 & (min) \\ \mbox{Weerstandsfactor} & \mbox{\gamma}_R & \\ \mbox{\gamma}_{\text{f.nk}} & \mbox{i} \end{array}$ 1.39 1.39 1.39 1.39 1.20 $\begin{array}{rrrr} & & & 1.0 \\ R_{s;cal;max;i} & begrenzen op 0.75 * R_{b;cal;max;i} & : NEE \\ UGT & draagvermogen & zonder & negatieve & kleef & : NEE \end{array}$ Paal: Timber pile d=200Niveau paalkop[m]: Maaiveld 0.30Paalpuntniveau: Maaiveld -15.50 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

Sweco Nederla	nd BV						Blad:	30
Technosoft Paalf	lunderingen r	elease 6.	.70a			14	okt 20	022
Project Onderdeel	: Pi : Ti	le Founda mber Pile	ation for T e	Cimber Col	umns			
SAMENVATTINGS	TABEL Case	9 (n=1)						
Uitgangspunten								
- paal		: Timbe	er pile d=2	200				
- paaltype		: Houte	en paal (co	onstant)				
- schachtafmetir	ıg	: 200 r	nm					
Paalklassefactor	α_{n}	: 0.70						
Factor α_{c} (tabel	. 7.c [°] EC 7.1)	: 0.010) (zandlad	en; voor	kleilagen	zie tabe	1 7.d)	
Correlatiefactor	ξ _{3(n=1)}	: 1.39		, -				
Alle niveaus/hoo	ogtes/peilmat	en zijn t		aiveld				
n	naaiveld paal	punt H	3ezwijkdraa	agvermogen	Reken	waarden		
sondering	niveau ni	veau R _b ;	[kN] ^R s;ca] R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R _c [kN]	;netto [kN]	;d
LOCOS CETODOOOO	0 00 -1	5 50 33	24 0 392	7 716 6	429 6	-44 8	384 8	

Berekening Rekenmethode Sondering(en)	: Ontwerpend : Drukpalen volgens NEN-EN : Loc10_CPT000000004377	1997-1,	art.	7.6.2
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 : 0p 0.75 * R _{b;cal;max;i} : her negatieve kleef :	NEE NEE		
Paal Niveau paalkop [m] Paalpuntniveau Bovenbel. [kN/m²]	: Timber pile d=200 : Maaiveld 0.30 : Maaiveld -19.90 : -10.00			

 Sweco Nederland BV
 Blad: 31

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Timber Columns

 Onderdeel
 : Timber Pile

 SAMENVATTINGSTABEL Case 10 (n=1)

Uitgangspunten - paal : Timber pile d=200 - paaltype : Houten paal (constant) - schachtafmeting : 200 mm (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden aaiveld paarpunt Beans, and R_{s;cal} R_{c;cal} niveau niveau R_{b;cal} R_{s;cal} R_{c;cal} [kN] [kN] [kN] sondering F_{nk;d} R_{c;netto;d} [kN] [kN] R_{c;d} [kN]

292.4

726.5

435.5

-196.0

239.5

434.1

REKENGEGEVENS Case 11

Loc10_CPT0000000 0.00

: Ontwerpend Berekening Rekenmethode : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc11_CPT000000012135 Sondering(en) : NEE : NEE Stijf bouwwerk Paalgroep Aantal sonderingen : 1 1.39 1.39 1.39 1.20 1.20 $\begin{array}{rll} \gamma_{\text{f,nk}} & : & 1.0 \\ R_{\text{s,cal,max,i}} & \text{begrenzen op } 0.75 \ ^{\text{K}}\text{R}_{\text{b,cal,max,i}} & : \text{NEE} \\ \text{UGT draagvermogen zonder negatieve kleef} & : \text{NEE} \end{array}$ Paal: Timber pile d=200Niveau paalkop[m]: Maaiveld 0.30Paalpuntniveau: Maaiveld -14.00 Paalpuntniveau : Maaiveld Bovenbel. [kN/m²] : -10.00

-19.90

				Blac	a: 32
Technosoft Paalfunderingen release 6.70a			14	okt	2022
Project : Pile Foundation for Time Onderdeel : Timber Pile	per Coli	umns			
SAMENVATTINGSTABEL Case 11 (n=1)					
Uitgangspunten					
- paal : Timber pile d=200					
- paaltype : Houten paal (const	tant)				
- schachtafmeting : 200 mm					
Paalklassefactor $\alpha_{\rm p}$: 0.70					
Factor α_{c} (tabel 7.c ^F EC 7.1) : 0.010 (zandlagen;	voor 1	kleilagen	zie tabe	1 7.d	1)
Correlatiefactor $\xi_{3(n=1)}$: 1.39		5			
Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaive	eld				
maaiveld paalpunt Bezwijkdraagve	ermogen	Reken	waarden		
sondering niveau niveau R _{b;cal} R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R [kN]	c;net [kN	to;d []
Loc11_CPT0000000 0.00 -14.00 204.1 494.0	698.2	418.6	0.0	418.	6

Berekening	: Ontwerpend
Rekenmethode	: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2
Sondering(en)	: Loc12_CPT000000077437
$\begin{array}{llllllllllllllllllllllllllllllllllll$: NEE : NEE : 1 : 1.39 : 1.39 : 1.39 : 1.20 : 1.0 : op 0.75 * R _{b;cal;max;i} : NEE Her negatieve kleef : NEE
Paal	: Timber pile d=200
Niveau paalkop [m]	: Maaiveld 0.30
Paalpuntniveau	: Maaiveld -9.10
Bovenbel. [kN/m²]	: -10.00

Sweco Nederland BV							Blad	: 33
Technosoft Paalfunderi	ngen release	6.70a				14	okt	2022
Project Onderdeel	: Pile Found : Timber Pil	dation f le	or Timb	ber Colum	nns			
SAMENVATTINGSTABEL	Case 12 (n=	1)						
Uitgangspunten - paal - paaltype - schachtafmeting Paalklassefactor α_p Paator α_c (tabel 7 or F	: Timl : Hou : 200 : 0.7 : 7 1) : 0 0	per pile ten paal mm)	d=200 (const	ant)	oilagon	zio tabo	174	N
Correlatiefactor $\xi_{3(n=)}$: 1.39	9	urayen,	VOOL KI	erragen	ZIE Labe	_ /.u)
Alle niveaus/hoogtes/p maaivel	eilmaten zijn d paalpunt	t.o.v.: Bezwijk	Maaive draagve	eld ermogen	Rekenv	vaarden		
sondering nivea	ı niveau R _l	c;cal R [kN]	s;cal [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R [kN]	;net [kN	to;d]
Loc12 CPT000000 0.0) -9 10	113 7	120 8	234 5	140 6	-18 6	122	0

Sweco Nederlan	d BV					Bl	ad: 34
Technosoft Paalfu	underinger	n release	6.70a			14 ok	t 2022
Project Onderdeel	:	Pile Foun Timber Pi	dation for Ti le	mber Colu	mns		
OVERZICHT NETT	O DRAAGV	ERMOGEN	DRUKPALEN				
Netto paaldraagve Alle niveaus/hoog	ermogen(s) gtes/peilm	zijn naa maten zijn	r beneden toe t.o.v.: Maai	afgerond veld	op: 1.0	kN nauwkeu	rig
sondering	maaiveld niveau	paalpunt niveau	R _{c;netto;d} Case 1	[kN] Case 2	Case 3	Case 4	Case 5
Loc01_CPT0000000	0.00	-4.00	10				
Loc03_CPT0000000	0.00	-5.70			111		
Loc04_CPT0000000	0.00	-7.80				374	
Loc02_CPT0000000	0.00	-20.60		170			
Loc05_CPT0000001	0.00	-14.10					616

Sweco Nederlan	В	Blad: 35					
Technosoft Paalfu	underinge	n release	6.70a			14 o	kt 2022
Project Onderdeel	:	Pile Foun Timber Pi	dation for Ti le	mber Colu	umns		
Netto paaldraagve Alle niveaus/hoog	ermogen(s gtes/peiln maaiveld) zijn naa maten zijn paalpunt	r beneden toe t.o.v.: Maai ^R c;netto;d	e afgerond veld [kN]	l op: 1.0	kN nauwke	urig
sondering	niveau	niveau	Case 6	Case 7	Case 8	Case 9	Case 10
Loc06_CPT0000001	0.00	-19.10	849				
Loc08_CPT0000000	0.00	-2.00			-4		
Loc07_CPT0000000	0.00	-14.30		610			
Loc09_CPT0000000	0.00	-15.50				384	
Loc10_CPT0000000	0.00	-19.90					239

Sweco Nederlan	Sweco Nederland BV									
Technosoft Paalfu	Inderinger	n release	6.70a	14 okt 2022						
Project Onderdeel										
Netto paaldraagve Alle niveaus/hoog	etto paaldraagvermogen(s) zijn naar beneden toe afgerond op: 1.0 kN nauwkeurig lle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt R _{c;netto;d} [kN]									
sondering	niveau	niveau	Case 11 Case 12							
Loc12_CPT0000000	0.00	-9.10	122							
Loc11_CPT0000000	0.00	-14.00	418							

Sweco Nederland BV Blad: 37 Technosoft Paalfunderingen release 6.70a 14 okt 2022 Project : Pile Foundation for Concrete Columns Onderdeel : Concrete Pile REKENGEGEVENS Case 1 : Ontwerpend : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc01_CPT000000052457 Berekening Rekenmethode Sondering(en) $\begin{array}{cccc} \text{Aantal sonderingen} & 1 \\ \text{Factor} & \xi_3 & (n=1) \\ \text{Factor} & \xi_3 & (gem) \\ \text{Factor} & \xi_4 & (min) \\ \text{Weerstandsfactor} & \gamma_R \\ \gamma_{\text{f},nk} & 1.0 \\ \end{array}$ Paal : Concrete Pile 400x400 Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -28.20 Bovenbel. [kN/m²] : -10.00

Sweco Nederland	BV							Blac	1: 38
Technosoft Paalfur	ndering	en releas	e 6.70a				1	4 okt	2022
Project Onderdeel		: Pile Fo : Concret	undation e Pile	for Con	icrete Co	lumns			
SAMENVATTINGSTA	BEL Ca	ase 1 (n	=1)						
Uitgangspunten - paal - paaltype - schachtafmeting Paalklassefactor Factor α_s (tabel 7 Correlatiefactor 8 Alle niveaus/hoogt	α _p 7.c EC 3(n=1) ces/pei	: C : G : 4 : 0 7.1) : 0 : 1 lmaten zi	oncrete eheide p 00 x 400 .70 .010 (z .39 jn t.o.v	Pile 400 aal (bet andlagen .: Maaiv	ex400 con) veld	leilagen	zie tab	el 7.c	1)
maa	aiveld	paalpunt	Bezwi	jkdraagv	vermogen	Reken	waarden		
sondering r	niveau	niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} [kN]	R _{c;net} [k1	to;d]]
Loc01_CPT0000000	0.00	-28.20	818.4	522.6	1341.0	804.0	-462.9	341.	0
REKENGEGEVENS C	ase 2								
Berekening Rekenmethode	:	Ontwerpen Drukpalen	d volgens	NEN-EN	1997-1, a	art. 7.6	.2		

Rekenmethode: Drukpalen volgens NEN-EN 1997-1Sondering(en): Loc02_CPT000000003023Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (gem):factor ξ_3 (gem):factor ξ_4 (min):1.39Factor ξ_4 (min):factor ξ_4 (min):1.00 $\chi_{f,nk}$: 1.00 $R_{s;cal;max;i}$ begrenzen op 0.75 * $R_{b;cal;max;i}$: NEEUGT draagvermogen zonder negatieve kleef: NEEPaal: Concrete Pile 400x400Niveau paalkop[m]:Maaiveld0.30Paalpuntniveau: Maaiveld -27.10Bovenbel.[kN/m²]: -10.00

Sweco Nederland BV Blad: 39 14 okt 2022 Technosoft Paalfunderingen release 6.70a : Pile Foundation for Concrete Columns Project Onderdeel : Concrete Pile SAMENVATTINGSTABEL Case 2 (n=1) Uitgangspunten - paal : Concrete Pile 400x400 - paaltype : Geheide paal (beton) - schachtafmeting : 400 x 400 (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen niveau niveau R_{b;cal} R_{s;cal} R_{c;cal} Rekenwaarden R_{c;d} F_{nk;d} R_{c;netto;d} sondering

Loc02_CPT0000000 0.00 -27.10 2400.0 1740.9 4140.9 2482.6 -598.8 1883.8	-			[kN]	[kN]	[kN]	[kN]	[kN]	[kN]
	Loc02_CPT0000000	0.00	-27.10	2400.0	1740.9	4140.9	2482.6	-598.8	1883.8

REKENGEGEVENS Case 3

: Ontwerpend Berekening Rekenmethode : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc03_CPT000000074906 Sondering(en) : NEE : NEE Stijf bouwwerk Paalgroep Aantal sonderingen : 1 1 1.39 1.39 1.39 1.20 1.0 $\begin{array}{c|c} \text{Aantal songeringen} \\ \text{Factor} & \xi_3 & (n=1) \\ \text{Factor} & \xi_3 & (gem) \\ \text{Factor} & \xi_4 & (min) \\ \text{Weerstandsfactor} & \gamma_R \\ \gamma_{\text{f.nk}} & \vdots \end{array}$ $\begin{array}{rll} \gamma_{\text{f,nk}} & : & 1.0 \\ R_{\text{s,cal,max,i}} & \text{begrenzen op } 0.75 \ ^{\text{K}}\text{R}_{\text{b,cal,max,i}} & : \text{NEE} \\ \text{UGT draagvermogen zonder negatieve kleef} & : \text{NEE} \end{array}$ 1.0 Paal : Concrete Pile 400x400 Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -6.10 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

Sweco Nederland B	v						Blad: 40
Technosoft Paalfunder	ringen releas	e 6.70a				1	4 okt 2022
Project Onderdeel	: Pile Fo : Concret	undation e Pile	for Con	crete Co	lumns		
SAMENVATTINGSTABE	L Case 3 (n	=1)					
Uitgangspunten							
- paal - paaltype - schachtafmeting Paalklassefactor α_{g} Factor α_{s} (tabel 7.c Correlatiefactor $\xi_{3(r}$ Alle niveaus/hoogtes,	: C : G : 4 : 0 EC 7.1) : 0 =1) : 1 /peilmaten zi	oncrete 1 eheide pa 00 x 400 .70 .010 (za .39 jn t.o.v	Pile 400: aal (beto andlagen) .: Maaivo	x400 on) ; voor k. eld	leilagen	zie tab	el 7.d)
sondering nive	eld paalpunt eau niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	ermogen R _{c;cal} [kN]	Reken R _{c;d} [kN]	F _{nk;d} [kN]	R _{c;netto;d} [kN]
Loc03_CPT0000000 0.	.00 -6.10	331.7	264.4	596.1	357.4	-30.3	327.1
REKENGEGEVENS Case	e 4						
Berekening Rekenmethode Sondering (en)	: Ontwerpen : Drukpalen : Loc04_CPT	d volgens 00000005	NEN-EN 3 5844	1997-1,	art. 7.6	.2	

Sweco Nederland BV Blad: 41 14 okt 2022 Technosoft Paalfunderingen release 6.70a : Pile Foundation for Concrete Columns Project Onderdeel : Concrete Pile SAMENVATTINGSTABEL Case 4 (n=1) Uitgangspunten - paal : Concrete Pile 400x400 - paaltype : Geheide paal (beton) - schachtafmeting : 400 x 400 (zandlagen; voor kleilagen zie tabel 7.d)

REKENGEGEVENS Case 5

: Ontwerpend Berekening : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc05_CPT000000161652 Rekenmethode Sondering(en) : NEE : NEE Stijf bouwwerk Paalgroep Aantal sonderingen : 1 $\begin{array}{c|c} \mbox{Aantal songeringen} & \cdot \\ \mbox{Factor} & \mbox{\xi}_3 & (n=1) \\ \mbox{Factor} & \mbox{\xi}_3 & (gem) \\ \mbox{Factor} & \mbox{\xi}_4 & (min) \\ \mbox{Weerstandsfactor} & \mbox{\gamma}_R & : \\ \mbox{\gamma}_{f\cdot nk} & : \end{array}$ 1.39 1.39 1.39 1.39 1.20 $\begin{array}{rrrr} \gamma_{\text{f;nk}} & : & 1.0 \\ R_{\text{s;cal;max;i}} & \text{begrenzen op } 0.75 \ ^{\text{K}}\text{R}_{\text{b;cal;max;i}} & : \text{NEE} \\ \text{UGT draagvermogen zonder negatieve kleef} & : \text{NEE} \end{array}$ Paal : Concrete Pile 400x400 Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -13.20 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

Sweco Nederland B	v	Blad: 42
Technosoft Paalfunde	ringen release 6.70a 1	4 okt 2022
Project Onderdeel	: Pile Foundation for Concrete Columns : Concrete Pile	
SAMENVATTINGSTABE	L Case 5 (n=1)	
	: Concrete Pile 400x400 : Geheide paal (beton) : 400 x 400 p : 0.70 EC 7.1) : 0.010 (zandlagen; voor kleilagen zie tabe n=1) : 1.39 /peilmaten zijn t.o.v.: Maaiveld eld paalpunt Bezwijkdraagvermogen Rekenwaarden eau niveau R _{b;cal} R _{s;cal} R _{c;cal} R _{c;d} F _{nk;d} F [kN] [kN] [kN] [kN] [kN] [kN]	el 7.d) ^R c;netto;d [kN]
Loc05_CPT0000001 0	.00 -13.20 1703.7 1338.8 3042.5 1824.1 -82.4	1741.7
REKENGEGEVENS Cas Berekening Rekenmethode Sondering (en)	<pre>e 6 : Ontwerpend : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc06_CPT000000170226</pre>	

Stijf bouwwerk : NEE Paalgroep : NEE Aantal sonderingen : 1 Factor ξ_3 (n=1): 1.39 Factor ξ_3 (gem): 1.39 Factor ξ_4 (min): 1.39 Weerstandsfactor γ_R : 1.20 $\gamma_{f;nk}$: 1.0 $\gamma_{f;nk}$: 1.0 $\gamma_{f;nk}$: 1.0 Paal : Concrete Pile 400x400 Niveau paalkop [m]: Maaiveld 0.30 Paalpuntniveau : Maaiveld -16.00 Bovenbel. [kN/m²]: -10.00

Sweco Nederland BV Blad: 43 14 okt 2022 Technosoft Paalfunderingen release 6.70a : Pile Foundation for Concrete Columns Project Onderdeel : Concrete Pile SAMENVATTINGSTABEL Case 6 (n=1) Uitgangspunten - paal : Concrete Pile 400x400 - paaltype : Geheide paal (beton) - schachtafmeting : 400 x 400 (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld aaiveld paalpunt Decart R₅;cal R_c;cal R_c;cal R_c;cal [kN] [kN] [kN] [kN] maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden sondering F_{nk;d} R_{c;netto;d} [kN] [kN] R_{c;d} [kN] Loc06_CPT0000001 0.00 -16.00 1916.2 2415.0 4331.2 2596.6 0.0 2596.6

REKENGEGEVENS Case 7

: Ontwerpend Berekening : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc07_CPT00000004530 Rekenmethode Sondering(en) Stijf bouwwerk : NEE : NEE Paalgroep Aantal sonderingen : 1 $\begin{array}{c|c} \mbox{Aantal songeringen} & \cdot \\ \mbox{Factor} & \mbox{\xi}_3 & (n=1) \\ \mbox{Factor} & \mbox{\xi}_3 & (gem) \\ \mbox{Factor} & \mbox{\xi}_4 & (min) \\ \mbox{Weerstandsfactor} & \mbox{\gamma}_R & : \\ \mbox{\gamma}_{f\cdot nk} & : \end{array}$ 1.39 1.39 1.39 1.39 1.20 $\begin{array}{rrrr} \gamma_{\text{f;nk}} & : & 1.0 \\ R_{\text{s;cal;max;i}} & \text{begrenzen op } 0.75 \ ^{\text{K}}\text{R}_{\text{b;cal;max;i}} & : \text{NEE} \\ \text{UGT draagvermogen zonder negatieve kleef} & : \text{NEE} \end{array}$ Paal : Concrete Pile 400x400 Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -13.40 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

Sweco Nederland H	3V	Blad: 44
Technosoft Paalfunde	eringen release 6.70a 14	okt 2022
Project Onderdeel	: Pile Foundation for Concrete Columns : Concrete Pile	
SAMENVATTINGSTAB	3L Case 7 (n=1)	
Uitgangspunten - paal - paaltype - schachtafmeting Paalklassefactor G Factor α_s (tabel 7. Correlatiefactor ξ_3 Alle niveaus/hoogtes	: Concrete Pile 400x400 : Geheide paal (beton) : 400 x 400 x _p : 0.70 c EC 7.1) : 0.010 (zandlagen; voor kleilagen zie tabel (n=1) : 1.39 s/peilmaten zijn t.o.v.: Maaiveld	7.d)
maai	veld paalpunt Bezwijkdraagvermogen Rekenwaarden	
sondering niv	yeau niveau R _{b;cal} R _{s;cal} R _{c;cal} R _{c;d} F _{nk;d} R _{c;} [kN] [kN] [kN] [kN] [kN] [kN]	;netto;d [kN]
Loc07_CPT0000000	0.00 -13.40 962.1 1612.4 2574.5 1543.5 0.0 1	543.5
REKENGEGEVENS Ca	se 8	
Berekening Rekenmethode	: Ontwerpend : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2	

Rekenmethode: Drukpalen volgens NEN-EN 1997-1Sondering(en): Loc08_CPT000000062979Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (gem):factor ξ_3 (gem):factor ξ_4 (min):i.39Factor ξ_4 (min):factor ξ_4 (min):i.30Weerstandsfactor γ_R : 1.20 $\gamma_{f,nk}$: 1.0R_s;cal;max;ibegrenzen op 0.75 * Rb;cal;max;iUGT draagvermogen zonder negatieve kleef: NEEPaal: Concrete Pile 400x400Niveau paalkop[m]:Maaiveld0.30Paalpuntniveau: MaaiveldBovenbel.[kN/m²]:-10.00

Sweco Nederland BV Blad: 45 14 okt 2022 Technosoft Paalfunderingen release 6.70a : Pile Foundation for Concrete Columns Project Onderdeel : Concrete Pile SAMENVATTINGSTABEL Case 8 (n=1) Uitgangspunten paalpaaltype : Concrete Pile 400x400 : Geheide paal (beton) - schachtafmeting : 400 x 400 (zandlagen; voor kleilagen zie tabel 7.d) Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden aandaning

sondering	IIIveau	IIIveau	^r b;cal [kN]	rs;cal [kN]	「c;cal [kN]	[kN]	^r nk;d [kN]	ⁿ c;netto;d [kN]
Loc08_CPT0000000	0.00	-26.80	1377.5	1041.9	2419.4	1450.5	-584.9	865.5

REKENGEGEVENS Case 9

Berekening: OntwerpendRekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Loc09_CPT000000011161Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1):1.39Factor ξ_3 (gem):Factor ξ_4 (min):1.39Weerstandsfactor γ_R : 1.20 $\gamma_{f;nk}$: 1.0 $R_{s;cal;max;i}$ begrenzen op 0.75 * $R_{b;cal;max;i}$: NEEUGT draagvermogen zonder negatieve kleef: NEEPaal: Concrete Pile 400x400Niveau paalkop[m] : MaaiveldNiveau: Maaiveld -14.60Bovenbel.[kN/m²] : -10.00

Sweco Nederland BV						Blad: 46
Technosoft Paalfunderingen re	lease 6.70a				14	okt 2022
Project : Pil Onderdeel : Cor	e Foundatior crete Pile	n for Con	crete Co	lumns		
SAMENVATTINGSTABEL Case	9 (n=1)					
$\begin{array}{l} \textbf{Uitgangspunten} \\ \textbf{-} paal \\ \textbf{-} paaltype \\ \textbf{-} schachtafmeting \\ Paalklassefactor \alpha_p \\ Factor \alpha_s (tabel \ 7.c \ EC \ 7.1) \\ Correlatiefactor \xi_{3 \ (n=1)} \end{array}$: Concrete : Geheide p : 400 x 400 : 0.70 : 0.010 (z : 1.39	Pile 400 baal (bet andlagen	x400 on) ; voor k	leilagen	zie tabe	1 7.d)
Alle niveaus/hoogtes/peilmate maaiveld paalp	n zijn t.o.v unt Bezwi	v.: Maaiv jkdraagv	eld ermogen	Reken	waarden	
sondering niveau niv	eau R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R [kN]	c;netto;d [kN]
Loc09_CPT0000000 0.00 -14	.60 1290.1	871.9	2162.0	1296.2	-114.1	1182.1
REKENGEGEVENS Case 10						

Berekening: OntwerpendRekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Locl0_CPT00000004377Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1):1.39Factor ξ_3 (gem):1.39Factor ξ_4 (min):1.39Weerstandsfactor γ_R : 1.20 γ_{fink} : 1.0 $R_{s;cal;max;i}$ begrenzen op 0.75 * $R_{b;cal;max;i}$: NEEUGT draagvermogen zonder negatieve kleef: NEEPaal: Concrete Pile 400x400Niveau paalkop[m] : MaaiveldNiveau paalkop[m] : Maaiveld0.30PaalpuntniveauEvenbel.[kN/m²] : -10.00

 Sweco Nederland BV
 Blad: 47

 Technosoft Paalfunderingen release 6.70a
 14 okt 2022

 Project
 : Pile Foundation for Concrete Columns

 Onderdeel
 : Concrete Pile

 SAMENVATTINGSTABEL Case 10 (n=1)

 Uitgangspunten

 - paal
 : Concrete Pile 400x400

 - paaltype
 : Geheide paal (beton)

- schachtafmeting : 400 x 400 Paalklassefactor α_p : 0.70 Factor α_s (tabel 7.c EC 7.1) : 0.010 (zandlagen; voor kleilagen zie tabel 7.d) Correlatiefactor $\xi_{3(n=1)}$: 1.39 Alle niveaus/hoogtes/peilmaten zijn t.o.v.: Maaiveld maaiveld paalpunt Bezwijkdraagvermogen Rekenwaarden sondering niveau niveau $R_{b,cal}$ $R_{c,cal}$ $R_$

			[kN]	[kN]	[kN]	[kN]	[kN]	[kN]	
Loc10_CPT0000000	0.00	-23.70	960.7	1314.5	2275.2	1364.0	-499.2	864.8	

REKENGEGEVENS Case 11

: Ontwerpend Berekening Rekenmethode : Drukpalen volgens NEN-EN 1997-1, art. 7.6.2 : Loc11_CPT000000012135 Sondering(en) : NEE Stijf bouwwerk Paalgroep : NEE Aantal sonderingen : 1 $\begin{array}{c|c} \mbox{Aantal songeringen} & \cdot \\ \mbox{Factor} & \mbox{\xi}_3 & (n=1) \\ \mbox{Factor} & \mbox{\xi}_3 & (gem) \\ \mbox{Factor} & \mbox{\xi}_4 & (min) \\ \mbox{Weerstandsfactor} & \mbox{\gamma}_R & : \\ \mbox{\gamma}_{f\cdot nk} & : \end{array}$ 1.39 1.39 1.39 1.20 $\begin{array}{rll} \gamma_{\text{f,nk}} & : & 1.0 \\ R_{\text{s,cal,max,i}} & \text{begrenzen op } 0.75 \ ^{\text{K}}\text{R}_{\text{b,cal,max,i}} & : \text{NEE} \\ \text{UGT draagvermogen zonder negatieve kleef} & : \text{NEE} \end{array}$ Paal : Concrete Pile 400x400 Niveau paalkop [m] : Maaiveld 0.30 Paalpuntniveau : Maaiveld -13.10 Paalpuntniveau : Maaivelo Bovenbel. [kN/m²] : -10.00

Sweco Nederlan	d BV							Blad: 48
Technosoft Paalf	undering	en releas	e 6.70a				14	okt 2022
Project Onderdeel		: Pile Fo : Concret	undation e Pile	for Cor	crete Co	lumns		
SAMENVATTINGSI	ABEL C	ase 11 (n=1)					
Uitgangspunten								
- paal - paaltype - schachtafmeting Paalklassefactor Factor α_s (tabel Correlatiefactor	g α _p 7.c EC ξ _{3 (n=1)}	: C : G : 4 : 0 7.1) : 0 : 1	oncrete eheide p 00 x 400 .70 .010 (z .39	Pile 400 aal (bet andlagen	x400 .on) ; voor k	leilagen	zie tabe	l 7.d)
Alle niveaus/hoo	gtes/pei	lmaten zi	jn t.o.v Bezwi	.: Maaiv	reld	Reken	waarden	
sondering	niveau	niveau	R _{b;cal} [kN]	R _{s;cal} [kN]	R _{c;cal} [kN]	R _{c;d} [kN]	F _{nk;d} R _c [kN]	;netto;d [kN]
Loc11_CPT000000	0.00	-13.10	839.7	1124.6	1964.3	1177.6	0.0	1177.6
REKENGEGEVENS	Case 1	2						

Berekening: OntwerpendRekenmethode: Drukpalen volgens NEN-EN 1997-1, art. 7.6.2Sondering(en): Locl2_CPT000000077437Stijf bouwwerk: NEEPaalgroep: NEEAantal sonderingen: 1Factor ξ_3 (n=1):factor ξ_3 (gem):factor ξ_4 (min):factor ξ_6 (nequence)Vf;nk: 1.0Rs,cal;max;ibegrenzen op 0.75 * Rb;cal;max;iVGTdraagvermogen zonder negatieve kleefVaeal: Concrete Pile 400x400Niveau paalkop[m] : MaaiveldNaiveld-9.30Bovenbel.[kN/m²] : -10.00

Sweco Nederland B	J						Blac	d: 49
Technosoft Paalfunder	ringen relea	se 6.70a				14	okt	2022
Project Onderdeel	: Pile Fo : Concre	oundatio te Pile	n for Con	crete Co	lumns			
SAMENVATTINGSTABE	L Case 12	(n=1)						
Uitgangspunten - paal - paaltype - schachtafmeting Paalklassefactor α_s Factor α_s (tabel 7.c Correlatiefactor ξ .	: (: (: (:) : (:) : (Concrete Geheide (400 x 40 0.70 0.010 (1 39	Pile 400 paal (bet 0 zandlagen	x400 on) ; voor ki	leilagen	zie tabe	1 7.0	1)
Alle niveaus/hoogtes, maaive sondering nive	/peilmaten z. eld paalpunt eau niveau	ijn t.o. Bezw R _{b;cal} [kN]	v.: Maaiv ijkdraagv ^R s;cal [kN]	eld ermogen ^R c;cal [kN]	Rekenv R _{c;d} [kN]	vaarden F _{nk;d} R _c [kN]	;net [kN	to;d 1]
Loc12 CPT000000 0	00 -9 30	344 5	327 0	671 4	402 5	-47 3	355	3

Sweco Nederlan	d BV					Bl	Lad: 50
Technosoft Paalfu	underingen	n release	6.70a			14 ok	kt 2022
Project Onderdeel	:	Pile Found Concrete 1	dation for Co Pile	oncrete Co	lumns		
OVERZICHT NETT	O DRAAG	/ERMOGEN	DRUKPALEN				
Netto paaldraagve Alle niveaus/hoog	ermogen(s) gtes/peilr	zijn naa: naten zijn	r beneden toe t.o.v.: Maai	e afgerond veld	op: 1.0	kN nauwkeu	urig
sondering	maaiveld niveau	paalpunt niveau	R _{c;netto;d} Case 1	[kN] Case 2	Case 3	Case 4	Case 5
Loc01_CPT0000000	0.00	-28.20	341				
Loc03_CPT0000000	0.00	-6.10			327		
Loc04_CPT0000000	0.00	-7.10				1702	
Loc02_CPT0000000	0.00	-27.10		1883			
Loc05_CPT0000001	0.00	-13.20					1741

Sweco Nederlan	d BV					В	lad: 51
Technosoft Paalfu	underingen	n release	6.70a			14 o	kt 2022
Project Onderdeel	:	Pile Foun Concrete	dation for Co Pile	oncrete Co	olumns		
Netto paaldraagve Alle niveaus/hoog	ermogen(s) gtes/peilm) zijn naa maten zijn	r beneden toe t.o.v.: Maai	e afgerond veld	d op: 1.0	kN nauwke	urig
sondering	maaıveld niveau	paalpunt niveau	R _{c;netto;d} Case 6	[KN] Case 7	Case 8	Case 9	Case 10
Loc06_CPT0000001	0.00	-16.00	2596				
Loc08_CPT0000000	0.00	-26.80			865		
Loc07_CPT0000000	0.00	-13.40		1543			
Loc09_CPT0000000	0.00	-14.60				1182	
Loc10_CPT0000000	0.00	-23.70					864
Sweco Nederlan	d BV			Blad: 52			
--	--------------------------------------	------------------------------------	---	-------------------			
Technosoft Paalfu	Inderinger	n release	6.70a	14 okt 2022			
Project Onderdeel	:	Pile Foun Concrete	dation for Concrete Columns Pile	3			
Netto paaldraagve Alle niveaus/hoog	ermogen(s) gtes/peilm maaiveld	zijn naa maten zijn paalpunt	r beneden toe afgerond op: t.o.v.: Maaiveld Re.netto.d [kN]	1.0 kN nauwkeurig			
sondering	niveau	niveau	Case 11 Case 12				
Loc12_CPT0000000	0.00	-9.30	355				
Loc11_CPT0000000	0.00	-13.10	1177				

Comparison Modular or Timber Car Park Concepts

<u>ModuPark</u>															
IDXX	Modupark, 16.50m (16.26	m), column	every 2 p	arking ba	ys (correc	ted for compa	rable din	nensions)							
Car Park Characteristics															
Parking bay width	2.5 m	Span park	ing deck		16.26 m	Floor depth	550	mm	Load sta	ndard column	- 1 level	618.1 kN			
Parking bay length	5.13 m	Column fr	ee parking	g deck	Yes	Total weight	61,261	kg	Load sta	ndard column	- 2 levels	1,236.1 kN			
Aisle width	6.00 m	Clearance	height		2.3 m				Load sta	ndard column	- 3 levels	1,854.2 kN			
Parking angle	90°	Number o	f parking	evels	4				Load sta	ndard column	- 4 levels	2,472.3 kN			
Description	Profile	Material	Length	Width	Height	Cross-area	A	Volume		Density	Weight	Weight Nr.	of elements 1	Weight structure	Volume
			[mm]	[mm]	[m m]	[mm ²]	[m ²]	[mm ³]	[m_]	[kg/m ³]	[kg]	[kN]		ر [kg]	[m³]
TT-slab	TT-550+0	C50/60	16,260	2,500	550	374,702	0.37	6,092,654,520	6.09	2400	14,622.4	146.22	4	58,489.5	24.4
Transverse main girder	IPE 360	S355	5,000	170	360	7,273	0.01	36,365,000	0.04	7850	291.0	2.91	4	1,164.0	0.1
Column	SHS-HF 250x250x12.5(8-20)	S355	2,850	250	250	17,971	0.02	51,217,350	0.05	7850	401.9	4.02	4	1,607.4	0.2
Koopman int., Ams	sterdam														
XXDI	Koopman int. (steel + fibre	glass), 15.6	5m (16.26	inlo), colui	nn every	2 parking bays	(correcte	ed for compare	able dim	ensions)					
Car Park Characteristics															
Parking bay width	2.5 m	Span park	ing deck		16.26 m	Floor depth	640	mm	Load sta	ndard column	- 1 level	314.6 kN			
Parking bay length	5.13 m	Column fr	ee parking	g deck	Yes	Total weight	10,477	kg	Load sta	ndard column	- 2 levels	629.2 kN			
Aisle width	6.00 m	Clearance	height		2.3 m				Load sta	ndard column	- 3 levels	943.7 kN			
Parking angle	90°	Number o	f parking	evels	4				Load sta	ndard column	- 4 levels	1,258.3 kN			
									Load sta	ndard column	- 5 levels	1,572.9 kN			
Description	Profile	Material	Length	Width	Height	Cross-area	A	Volume	>	Density	Weight	Weight Nr.	of elements \	Veight structure	Volume
			[mm]	[mm]	[mm]	[mm ²]	[m ²]	[mm³]	[m³]	[kg/m ³]	[kg]	[kN]		[kg]	[m³]
Deck	Fiberline Plank 40HD	GFRP	16,260	500	40	4,783	0.00	77,771,580	0.08		137.1	1.37	20	2,741.4	1.6
Secondary girder	HEA 140	S355	5,000	140	133	3,142	0.00	15,710,000	0.02	7850	125.5	1.26	24	3,012.0	0.4
Main	IPE 600	S355	16,260	220	600	15,598	0.02	253,623,480	0.25	7850	2,032.5	20.33	2	4,065.0	0.5
Column	K 260x260x7.1	S355	2,940	260	260	7,128	0.01	20,956,320	0.02	7850	164.6	1.65	4	658.6	0.1
Morspoort, Leiden															
IDXX	Morspoort (steel + compo	site), 14.50	(16.26m),	column (every 2 pa	rking bays (co	rected fo	or comparable	dimensi	ons)					
Car Park Characteristics															
Parking bay width	2.5 m	Span park	ing deck		16.26 m	Floor depth	651	mm	Load sta	ndard column	- 1 level	461.1 kN			
Parking bay length	5.13 m	Column fr	ee parking	g deck	Yes	Total weight	35,110	kg	Load sta	ndard column	- 2 levels	922.2 kN			
Aisle width	6.00 m	Clearance	height		2.3 m				Load sta	ndard column	- 3 levels	1,383.4 kN			
Parking angle	90°	Number o	f parking	evels	4				Load sta	ndard column	- 4 levels	1,844.5 kN			
									Load sta	ndard column	- 5 levels	2,305.6 kN			
Description	Profile	Material	Length	Width	Height	Cross-area	۷	Volume	>	Density	Weight	Weight Nr.	of elements V	Veight structure	Volume
			[mm]	[mm]	[mm]	[mm ²]	[m ²]	[mm³]	[m³]	[kg/m ³]	[kg]	[kN]		[kg]	[m ³]
Deck	Quantum Deck Floor	e (Concret	5,000	16,260	250		0.00	0	0.00		14,634.0	146.34	2	29,268.0	0.0
Joist	K 120x120x5	S275	5,000	120	120	2,273	0.00	11,365,000	0.01	7850	89.0	0.89	8	712.0	0.1
Main	IPE600	S355	16,260	220	600	15,598	0.02	253,623,480	0.25	7850	2,032.5	20.33	2	4,065.0	0.5
Column	K 300x300x10	S275	2,951	300	300	11,493	0.01	33,915,843	0.03	7850	266.2	2.66	4	1,064.7	0.1

Table 55. Data modular or timber car park concepts for comparison

Construction syste	m for multi-storey car g	oark in Ba	auBuche												
IDXX	BauBuche concept (timber	+ concrete	e), 16.50m	(16.26m)	column e	very 1 parkin	g bay (co	rrected for con	nparable	e dimensions)					
Car Park Characteristics															
Parking bay width	2.5 m	Span parl	king deck		16.26 m	⁻ loor depth	730	mm l	-oad sta	ndard column	- 1 level	298.3 kN			
Parking bay length	5.13 m	Column f	ree parking	g deck	res -	Fotal weight	59,341	kg I	-oad sta	ndard column	- 2 levels	596.6 kN			
Aisle width	6.00 m	Clearance	e height		2.3 m			_	-oad sta	ndard column	- 3 levels	894.9 kN			
Parking angle	90°	Number o	of parking l	evels	. 4				-oad sta	ndard column	- 4 levels	1,193.2 kN			
								1	-oad sta	ndard column	- 5 levels	1,491.5 kN			
Description	Profile	Material	Length	Width	Height	Cross-area	A	Volume	^	Density	Weight	Weight Nr. o	f elements \	Veight structure	Volume
			[mm]	[mm]	[mm]	[mm ²]	[m²]	[mm³]	[m ³]	[kg/m ³]	[kg]	[kN]		[kg]	[m ³]
Deck	Precast concrete 130 mm	C50/60	2,500	16,260	130	2,113,800	2.11	5,284,500,000	5.28	2400	12,682.8	126.83	4	50,731.2	21.1
Main	240x600, cambered 40 mm	BB GL75	16,260	240	600	144,000	0.14	2,341,440,000	2.34	800	1,873.2	18.73	4	7,492.6	9.4
Column	240x240	BB GL75	3,030	240	240	57,600	0.06	174,528,000	0.17	800	139.6	1.40	∞	1,117.0	1.4
B&O-Holzparkhaus	s, Bad Aibling														
IDXX	B&O-Holzparkhaus (timber	-), 16.24m	(16.26m),	column ev	ery 1 par	cing bay (corn	ected for	comparable di	mensio	us)					
Car Park Characteristics															
Parking bay width	2.5 m	Span park	ting deck		16.26 m	-loor depth	935	mm	-oad sta	ndard column	- 1 level	179.2 kN			
Parking bay length	5.13 m	Column f	ree parking	g deck	res -	Fotal weight	17,547	kg I	-oad sta	ndard column	- 2 levels	358.3 kN			
Aisle width	6.00 m	Clearance	e height		2.3 m			_	-oad sta	ndard column	- 3 levels	537.5 kN			
Parking angle	90°	Number o	of parking l	evels	4			_	-oad sta	ndard column	- 4 levels	716.7 kN			
								1	-oad sta	ndard column	- 5 levels	895.8 kN			
Description	Profile	Material	Length	Width	Height	Cross-area	٩	Volume	>	Density	Weight	Weight Nr. o	f elements \	Neight structure	Volume
			[mm]	[mm]	[mm]	[mm²]	[m²]	[mm³]	[m³]	[kg/m ³]	[kg]	[kN]		[kg]	[m³]
Deck	BSP 100 3s 40/20/40	C24	2,500	16,260	100	1,626,000	1.63	4,065,000,000	4.07	500	2,032.5	20.33	4	8,130.0	16.3
Joist	400×60	C24	2,500	60	400	24,000	0.02	60,000,000	0.06	500	30.0	0.30	∞	240.0	0.5
Main	240x600-760	BB GL75	16,260	240	600-760	182,400	0.18	2,653,632,000	2.65	800	2,122.9	21.23	4	8,491.6	10.6
Column	240x240	GL24h	3,235	240	240	57,600	0.06	186,336,000	0.19	460	85.7	0.86	8	685.7	1.5

Table 56. Data modular or timber car park concepts for comparison

D19	Variant 2a. 16.26m. struts On. cc	olumn every 1 parkin.	e bav(s)												
Car Bark Characteristics			1-11-0										Т		
Cal Fair Clial acteristics															
Parking bay width	2.5 m	Span parking dec	Y'	16.26 n	1 Floor	depth	666 mr	٤	Lo	ad standard colum	1 - 1 level	169.6 kN			
Parking bay length	5.13 m	Column free par.	king deck	Yes	Total	weight	15,242 kg		Lo	ad standard columi	1 - 2 levels	339.2 kN			
Aisle width	6.00 m	Clearance height	t	2.3 m					Lo	ad standard columi	1 - 3 levels	508.8 kN			
Parking angle	00°	Number of parki	ing levels	4					Lo	ad standard columi	1 - 4 levels	678.4 kN			
									Lo	ad standard columi	1 - 5 levels	848.0 kN			
Description	Profile	Material Lengt	th [mm] Wic	tth [mm] Height	[mm] Cross-	-area [mm²] 4	V [m ²] Vo	lume [mm³]	/ [m ³] De	insity [kg/m ³] We	ight [kg] \	Weight [kN] Nr. of element	s Weight structure	kg] Volume [m ³	_
Multi-span LVL panel	LVL G 96 2s	C24	2,500	16,260	96	1,560,960	1.56	3,902,400,000	3.90	500	1,951.2	19.51	4 7,	804.8	15.6
Transverse main girder	440x140	GL28h	2,500	140	440	61,600	0.06	154,000,000	0.15	460	70.8	0.71	00	566.7	1.2
Column	440x360	GL28h	2,966	360	440	158,400	0.16	469,814,400	0.47	460	216.1	2.16	8 1,	728.9	3.8
Main	570x280	GL28h	16,260	280	570	159,600	0.16	2,595,096,000	2.60	460	1,193.7	11.94	4 4,	75.0	10.4
Strut	240×140	GL28h	2,966	140	240	33,600	0.03	99.661.395	0.10	460	45.8	0.46	8	366.8	0.8

Table 57. Data proof of concept: variant 2a, columns every 1 parking bay, struts on.

Additional Calculations & Results Deck System

F.1 Long-span Deck Systems

F.1.1 Example Calculations

Example Calculation Solid CLT Panel

	Multi-storey ca 1x12.18m span	r park_v2						Page 1/8 02.02.2023
storgenso	Paul Brink				Desi	gner PB		Checker
Stordenso								
System								
	q,=0.56 [kN/m]		LC4: NL - si	now				
*	0.00 (101/1							
•	q₁=0.36 [krv/m]		LC5: NL - W	/ind				
	q _x =2.00 [kN/m]		LC3: NL - liv	ve load ca	t. F: traffic a	rea (vehicl	es <= 25 kN)	
	q⊾=0.10 [kN/m]		I C2: NL - d					
•								
	q ⊾=1.60 [kN/m]		LC1:self-we	ight struct	ure			
Â.	Field 1		Ż					
Ϊ ∙−−−−	12.180 [m]							
Global utilization	ratio							178 %
ULS	38 % ULS Fire	27 % SLS	14	42 % S	LS Vibrati	on	178 % Supp	ort -1 %
Section: CLT 220	190 2							
Section. CET 320	LU3 - 2	Ŧ	Layer		Thick	ness	Orientation	Material
		320	1		40.0	mm	0°	C24 spruce ETA (2019)
		mm	2		40.0	mm	0°	C24 spruce ETA (2019)
4	1000 mm	<u>+</u>	3		40.0	mm	90°	C24 spruce ETA (2019)
			4		40.0	mm	0°	C24 spruce ETA (2019)
			5		40.0	mm	0°	C24 spruce ETA (2019)
			6		40.0	mm	90°	C24 spruce ETA (2019)
			7		40.0	mm	0°	C24 spruce ETA (2019)
			8		40.0	mm	0°	C24 spruce ETA (2019)
			tclt.		320.0	mm		
Section Fire: CLT	320 L8s - 2		1		Thisk		Orientation	Motorial
		24	1		40.0	mm	0°	C24 spruce
		mm	2		40.0	mm	0°	C24 spruce
4	*	-	3		40.0	mm	90°	C24 spruce
	1000 mm		4		40.0	mm	0°	C24 spruce
			5		40.0	mm	0°	C24 spruce
			6		40.0	mm	90°	C24 spruce
Fire resistance clas	ss:R 90		tclt Time		240.0	mm min		ETA (2019)
Fire protection laye	ering : no additional fire pr	otection	k ₀	d ₀	d _{char,0,h}	d _{ef,h}	d _{char,0,v} d _{ef,v}	
			[-] 1	[mm] 7	[mm] 71.0	[mm] 78.0	[mm] [mn 0.0 0.0	1J

ALL	N	lulti-stor	ey car pa	ark_v2					Pag	e 2/8	
	1>	x12.18m spa	n						02.0	2.202	3
storage	Pa	aul Brink				De	esigner PB		Che	cker	
Material v											
Material	f _{m,l}	k f _{t,0,k}	f _{t,90,k}	f _{c,0,k}	f _{c,90,k}	f _{v,k}	f _{r,k min}	E _{0,mean}	G _{mean}	G	r,mean
C24 spruc	[N/m	m²] [N/mm² 00 14.00	²] [N/mm ²] 0.12	[N/mm ²] 21.00	[N/mm²] 2.50	[N/mm²] 4.00	[N/mm²] 1.25	[N/mm ²] 12.000.00	[N/mm ²] 690.00] [N/ 5	/mm²] 0.00
ETA (2019))							,			
Load											
Load case	e groups										
Lo	ad case cat	egory		Туре	Duration	Kmod	Yinf	Ysup	Ψ ₀ ν	₽ı	Ψ_2
LC1 se	If-weight str	ucture		G	permanent	0.6	1	1.35	1	1	1
LC3 NL	- live load	cat. F: traffic a	rea (vehicles	<= Q	medium	0.0	Ó	1.5	0.7	0.7	0.6
25	kN)			0	term	0.0	0	1 5	0	0.2	c
LC4 NL	- snow			Q	term	0.9	0	1.5	0	0.2	U
LC5 NL	- Wind			Q	short term	0.9	0	1.5	0	0.2	0
LC1:self-v	veight stru	cture									
continue											
Continuo	us loau		-								
Field		Load at sta	Irt								
1		1.60									
LC2: NL -	deadload										
continuo	us load										
Field		Load at sta	irt								
		[kN/m]									
1		0.10									
LC3: NL -	live load ca	at. F: traffic a	rea (vehicles	<= 25 kN)							
continuo											
Continuo	us loau										
Field		Load at sta	Irt								
1		2.00									
LC4: NL -	snow										
continuo	us load										
Field		Load at sta	irt								
1		[kN/m] 0.56									
LC5: NL -	Wind										
continuo	ous load										
Field		Load at sta	irt								
		flch1/ma1									
		[KIN/m]									



	Mu 1x1:	I lti-storey 2.18m span	′ car par	k_v2						Page 4/8 02.02.2023
storaenso	Pau	l Brink					Design	ier PE	3	Checker
Flexural stress a	nalysi	S								
$\begin{array}{l} M_{y,d} = \\ M_{z,d} = \\ N_{t,d} = \end{array}$	98.19 0.00 0.00	kNm kNm kN		$\begin{array}{l} f_{m,k} = \\ f_{m,k,z} = \\ \gamma_m = \\ k_{mod} = \\ k_{sys,y} = \\ k_{h,m,y} = \\ k_{h,m,z} = \\ k_{l} = \end{array}$	24.00 24.00 1.25 0.80 1.10 1.00 1.00 1.00	N/mm ² N/mm ² - - - -				
$\sigma_{t,d} = \sigma_{m,y,d} = \sigma_{m,z,d} =$	0.00 -6.46 0.00	N/mm² N/mm² N/mm²	<	$\begin{array}{l} f_{t,0,d} = \\ f_{m,y,d} = \\ f_{m,z,d} = \end{array}$	8.96 16.90 0.00	N/mm² N/mm² N/mm²		~		
Utilization ratio							38 %			
Shear stress and	alvsis									
V _d =	-32.25	kN		$\begin{array}{l} f_{v,k} = \\ \gamma_m = \\ k_{mod} = \\ k_{h,v} = \end{array}$	4.00 1.25 0.80 0.00	N/mm² - -				
T _{v,d} =	0.14	N/mm²	<	t _{v,d} =	2.56	N/mm²	5 %	~		
Rolling shear an	alysis									
V _d =	-32.25	kN		$f_{r,k} = \gamma_m = k_{mod} =$	1.05 1.25 0.80	N/mm² - -				
T _{r,d} =	0.13	N/mm²	<	f _{r,d} =	0.67	N/mm²		✓		
Utilization ratio							19 %			

Ultimate limit state (ULS) fire design - results



all	10	Μι	ulti-st	ore	y car p	ark_v2					Page	5/8
		1x1	2.18m	spar	ı						02.02	.2023
		Pau	ıl Brink						Designer Pl	В	Check	ker
stora	ENSO											
ULS Fi	re Flexu	ral desig	gn									
Field	Dist.	f _{m,k}	γn 21 Γ	n 1	k _{mod}	k _{sys,y}	k _{fi}	f _{m,y,d}	M _{y,d}	σ _{m,y,d}	Ratio	
1	6.09	24.0	J [00	1.00	1.10	1.15	30.36	53.78	-8.13	27 %	LCO12
ULS Fi	re Sheai	r analysi	is									
Field	Dist.	f _{v,k}	Υn	n	k _{mod}	k _{fi}	f _{v,d}	Vd	T _{v,d}	Ratio		
	[m]	[N/mm ²	2] [-]]	[-]	[-]	[N/mm ²]	[kN]	[N/mm ²]			
1	12.18	4.0	00 1.	00	1.00	1.15	4.60	-17.66	0.1	3 3%	LCO12	
ULS Fi	re Rollin	ng shear	•									
Field	Dist.	f _{r,k}	γn	n	k _{mod}	k _{fi}	f _{r,d}	Vd	Tr,d	Ratio		
	[m]	[N/mm ²	²] [-]]	[-]	[-]	[N/mm ²]	[kN]	[N/mm ²]	0 44.0/	1.0040	
1	12.18	1.(J5 1.	00	1.00	1.15	1.21	-17.66	0.1	3 11%	LC012	
Stress	diagram	ı										
		Flexural st	ress				Shear stres	3		Rolling	shear stress	
	-8.13	[N/mm·	-]				[N/mm-j			[P	ı/mm•j	
								\sum				
								0.13			0.13	
			0.13									
Flexura	al stress	analysi	s Fire									
N	1 _{y,d} = 1 _{2 d} =	53.78 0.00	kNm kNm			f _{m,k} = f _{m,k} =	24.00 24.00	N/mm ² N/mm ²				
1	$V_{t,d} =$	0.00	kN			γ _m =	1.00	-				
						k _{mod} = k _{svs v} =	1.00	-				
						k _{h,m,y} =	1.00	-				
						$\kappa_{h,m,z} = k_l =$	1.00	-				
		0.00	NI/ma ma	2		k _{fi} =	1.15	- N1/mama2				
σm	Jt,d =	-8.13	N/mm	2		f _{m,y,d} =	30.36	N/mm ²				
σm	n,z,d =	0.00	N/mm	2	<	f _{m,z,d} =	0.00	N/mm ²	✓			
Utilizat	tion ratio)							27 %			
Shear s	stress a	nalysis l	Fire									
	V _d =	-17.66	kN			f _{v,k} =	4.00	N/mm ²				
						γ _m = k _{mod} =	1.00	-				
						k _{h,v} =	0.00	-				
1	T _{v,d} =	0.13	N/mm	2	<	$\kappa_{fi} = f_{v,d} =$	4.60	- N/mm²	×			
Utilizat	tion ratio)							3 %			
Della		mahari	Fire									
Rolling	jsneara V₄=	-17 66	rire kN			f	1.05	N/mm ²				
	• a	-17.00	1114			γ _m =	1.00	-				
						k _{mod} =	1.00	-				
	T _{r,d} =	0.13	N/mm	2	<	f _{r,d} =	1.21	N/mm²	✓			
Utilizat	tion ratio	>							11 %			



	Multi-stor 1x12.18m spa	ey car par ^{an}	k_v	2				Page 7/8 02.02.2023
storaenso	Paul Brink					Designe	er PB	Checker
Vibration analys	is							
Analysis								
Criterion Frequency criter Frequency criter Acceleration crite Stiffness criterion	ion min ion erion n	Calc. 4.346 [Hz] 4.346 [Hz] 0.178 [m/s²] 0.197 [mm]	Clas 4.5 [6.0 [0.1 [0.5 [s II Hz] Hz] m/s²] mm]	Class II 104 % 138 % 178 % 39 %	CI. II × ✓		
Support reaction	ı							
Load case catego	ory	k _{mod}	A _V	Bv				
self-weight struct	ure	0.6	9.74 9.74	9.74				
NL - deadload		0.6	0.61	0.61				
NL - live load cat. <= 25 kN)	F: traffic area (vel	hicles 0.8	12.18	12.18				
NL - snow		0.9	3.41	3.41				
NL - Wind		0.9	2.19	2.19 0.00				
Reference docu	ments for this an	alysis						
English title			D	escription				
EN 338 EN 1995-1-1			E E C	N 338 - S N 1995-1- ommon ri	tructural timber 1 1 - Eurocode 5: lles and rules fo	? Strength Design o r building	n classes f timber structures -	Part 1-1: General -
ETA-14/0349 Expertise Rolling EN 1995-1-2	shear - no edge g	luing, H.J. Blass	Ē	uropean 1 xpertise o N 1995-1-	echnical Assess n Rolling shear f 2 - Eurocode 5	sment ET. for CLT — Design	A-14/0349 of 02.10. of timber structures	2014 s — Part 1-2: Genera
Technical expertis bearing capacity a	se 122/2011/02: and separation per	nalysis of load rformance of CL ⁻	V F st	erification ructures v	of the load bear with Stora Enso	ring capao CLT	city and the insulatio	n criterion of CLT
Technical expertis gypsum fire board EN 1990 ÖNorm B 1995-1-	se 2434/2012 - BB ds (GKF) according -1 NA	3: failure time tf o g to ON B 3410	f E: ar E Ö pa	xpertise o nd gypsur N 1990 - I NORM El arameters	n failure time tf o n wall boards typ Eurocode ? Basi N 1995-1-1 - Aus – Eurocode 5: [of gypsum be DF acc is of struc stria - Nat Design of	n wall fire boards act cording to EN 520 tural design ional Annex – Natio timber structures –	cording to ON B3410 nally determined Part 1-1: General-
ÖNorm B 1995-1-	-2 NA		C Ö tir sp na	ommon ru NORM El nber struc pecificatio ational su	iles and rules fo N 1995-1-2 - Aus ctures ? Part 1-2 ns concerning C oplements	r buildings stria - Nat :: General NORM E	s ional Annex - Euroc ? Structural fire des N 1995-1-2, nationa	ode 5: Design of sign ? National I comments and
Fire safety in time Europe National specifica 1-2, national com chapter 12	per buildings - tech ations concerning (ments and nationa	nical guildeline f ÖNORM EN 199 al supplements,	or Fi S 5- Ö 1-	re safety P Technic NORM El 2, nationa	in timber buildin al Research Ins N 1995-1-2 - Nat al comments and	gs - techn titute of S tional spe d national	ical guideline for Eu weden cifications concernir supplements, chapt	rope; publishes by ng ÖNORM EN 1995 ter 12
Expertise Rolling ÖNORM EN 1995	shear, H.J. Blass 5-1-1_NA, chapter	7.3	E Ö pa	xpertise o NORM El arameters	n rolling shear s N 1995-1-1 - Aus – Eurocode 5: I	trength ar stria - Nat Design of	nd rolling shear mod ional Annex – Natio timber structures –	lulus of CLT panels nally determined Part 1-1: General-

Disclaimer

The software was created to assist engineers in their daily business. The software is an engineering software that is dealing with a very complex matter of structural analysis and building physics analysis. Therefore, this software shall only be operated by skilled, experienced engineers, with a deep understanding of structural engineering and building physics related to timber structures. The user of the software is obliged to check all input values, no matter if they were given by the user or given by default by the software and all results for plausibility. The use of the software should not be relied upon as the basis for any decision or action. Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. The user has the possibility to make print-outs from the software. Any modification of those are not allowed. Stora Enso Wood Products GmbH does not assume any warranty regarding the software. The software has been developed with utmost diligence, nevertheless Stora Enso Wood Products GmbH does also not assume any warranty for the general usability of the software, its suitability for a special purpose or for the compatibility of the software. Stora Enso Wood Products GmbH does also not assume any warranty for the general usability of the software, its suitability for a special purpose or for the compatibility of the software. Stora Enso Wood Products GmbH does also not assume any warranty for the general usability of the software is suitability for a special purpose or for the compatibility of the software. Stora Enso Wood Products GmbH does also not assume any warranty for the general usability of the software. Its suitability for a special purpose or for the compatibility of the software. Stora Enso Wood Products GmbH does also not assume any warranty for the general usability of the software is usability for a special purpose or for the compatibility of the software. Stora Ens

Example Calculation CLT Open Rib Panel

AN MARK		/lulti	i-store	y car	park_v2	2				Pa	age 1/13
		2.18m	n - 4 ribs	- 200mr	n - CL112	0				02	2.02.2023
stora	F ISO	aul B	rink					Designe	er PB	CI	hecker
Storder											
System											
			a,=0.56 [kN/m]								
						LC4: N	L - snow				
			q_=0.36 [kN/m]	1		LC5: N	L - Wind				
			q_=2.00 [kN/m]]			L Buo lood a	at E: traffic area	(vahislas <= 2E	LND	
<u> </u>						LC3: N	L - live load (cat. F: tramic area	(venicies <= 25	KIN)	
			q x=0.10 [kN/m]	1		LC2: N	L - deadload				
			q_=1.18 [kN/m]]			lf-weight stru	icture			
			Field 1			→ Lo 1.50 ス	n noight sud				
<i>7777.</i>			T IEIG T		Ź	B					
I4			12.180 [m]			-1					
Global ut	ilization ra	io									100 %
ULS	30	% U	LS Fire	2	7 % SLS		68 %	SLS Vibration	76 %	Support	9 %
Section:	CI T rih nai	al hv	Stora Eng	so: CLT	120 56 - 2	0/36					
4	625 mr	1				Lay	rer Th	nickness	Width	Orientation	Material
				120		1		[mm] 30.0	[mm] 606.0	0°	C24 spruce
		<u> </u>	¥777A			2		20.0	606.0	90°	ETA (2019) C24 spruce
			ļ	36		3		20.0	606.0	0°	ETA (2019) C24 spruce
İ			İ	mm		4		20.0	606.0	90°	ETA (2019) C24 spruce
						5		30.0	606.0	0°	ETA (2019) C24 spruce
I	4 200 mr	•	1 -			6		360.0	200.0	0°	ETA (2019) GL 28h
Ribs wid	th are redu	ced by	/ 20 mm fe	or the de	sign			490.0			
						LCLT		480.0 mm			
111 S T-0											
0201-0											
Field	Range 1	rom	Until	Width	Moment of	of inertia	Area	G*A	Kappa	Ely,netto)
1	Start	[m] 0	[m] 0.625	[cm] 51.70	[mn 1,937,	n⁴] .479.000	[mm²] 93,568	[kN] 61097	.93 0.5527	[kNm ²] 23249	.75
1 1	Center End 1	0.625 1.555	11.555 12.18	60.60 51.70	2,075, 1.937.	708,000	99,264 93,568	65170 61097	0.56 0.5197 0.93 0.5527	2490 23249	8.5 .75
Lover	Thickness		\\/idth	Tur	.,,	Mata	rial		0		
Layer	[mm]		[mm]	тур	e	wate	IIdl	E [N/mm²]	[N/mm ²]		
1 2	30.0 m 20.0 m	im im	606.0 m 606.0 m	im im	L C	C24 spruc C24 spruc	e ETA (20 e ETA (20	19) 9,600 19) 0	552 40		
3 4	20.0 m	im Im	606.0 m	im im	L C	C24 spruc	e ETA (20	19) 9,600 19) 0	552 40		
5	30.0 m	im	606.0 m	im Im	Ĺ	C24 spruc	ce ETA (20	19) 9,600 28b 10.080	552 520		
			200.0 11						020		
ULS T=∞											

		Multi 12.18m	-storey	car 200mm	park_v2 n - CLT120				Page 02.02	2/13 2.2023
storae	inso	Paul Br	ink				Designer P	В	Chec	ker
Field 1 1	Range Start Center	from [m] 0.625	Until [m] 0.625 11.555	Width [cm] 51.70 60.60	Moment of inertia [mm ⁴] 1,530,403,000 1,640,238,000	Area [mm²] 74,227 78,608 74,227	G*A [kN] 48391.95 51524.75	Kappa 0.5527 0.5197	Ely,netto [kNm ²] 18364.84 19682.86	
Layer 1 2 3 4 5 6	Thickne [mm] 30.0 20.0 20.0 20.0 30.0 360.0	PT.555 PSS D mm D mm D mm D mm D mm D mm D mm	Width [mm] 606.0 mr 606.0 mr 606.0 mr 606.0 mr 606.0 mr 200.0 mr	Typ n n n n n n	e Matu L C24 spru C C24 spru L C24 spru C C24 spru L C24 spru L C24 spru L	rial ice ETA (2019) ice ETA (2019) ice ETA (2019) ice ETA (2019) ice ETA (2019) ice ETA (2019) GL 28h	E C [N/mm ²] [N/m 7,385 0 7,385 0 7,385 8,129	6.5327 6 425 31 425 31 425 31 425 419	10304.04	
SLS T=0)									
Field	Range	from [m]	Until [m]	Width [cm]	Moment of inertia [mm ⁴]	Area [mm²]	G*A [kN]	Kappa	Ely,netto [kNm²]	
1 1 1	Start Center End	0 0 12.18	0 12.18 12.18	51.70 60.60 51.70	2,421,849,000 2,594,635,000 2,421,849,000	116,960 124,080 116,960	76372.41 81463.2 76372.41	0.5527 0.5197 0.5527	29062.19 31135.62 29062.19	
Layer	Thickne [mm]	ess]	Width [mm]	Тур	e Mate	erial	E (N/mm²) [N/n	G nm²]		
1 2 3 4 5 6	30.0 20.0 20.0 20.0 30.0 360.0) mm) mm) mm) mm) mm	606.0 mr 606.0 mr 606.0 mr 606.0 mr 606.0 mr 200.0 mr	ກ ກ ກ ກ ກ	L C24 spru C C24 spru L C24 spru C C24 spru L C24 spru L C24 spru L	ice ETA (2019) ice ETA (2019) ice ETA (2019) ice ETA (2019) ice ETA (2019) ice ETA (2019) GL 28h	12,000 0 12,000 0 12,000 12,600	690 50 690 50 690 650		
SLS T=•	0									
Field	Range	from	Until	Width	Moment of inertia	Area	G*A	Kanna	Elv netto	

Field	Range	from	Until	Width	Moment of inertia	Area	G*A	Kappa	Ely,netto
		[m]	[m]	[cm]	[mm ⁴]	[mm ²]	[kN]		[kNm ²]
1	Start	0	0	51.70	2,743,166,000	135,860	88072.41	0.5527	32917.99
1	Center	0	12.18	60.60	2,942,978,000	142,980	93163.2	0.5197	35315.73
1	End	12.18	12.18	51.70	2,743,166,000	135,860	88072.41	0.5527	32917.99
Lovor	Thickne	200	\\/idth	Tune	Moto	riol	E (

Layer	THICKNESS	width	Type	Materia		G
	[mm]	[mm]			[N/mm ²]	[N/mm ²]
1	30.0 mm	606.0 mm	L	C24 spruce ETA (2019)	12,000	690
2	20.0 mm	606.0 mm	С	C24 spruce ETA (2019)	0	50
3	20.0 mm	606.0 mm	L	C24 spruce ETA (2019)	12,000	690
4	20.0 mm	606.0 mm	С	C24 spruce ETA (2019)	0	50
5	30.0 mm	606.0 mm	L	C24 spruce ETA (2019)	12,000	690
6	360.0 mm	200.0 mm	L	GL 28h	15,750	813

Section Fire: CLT rib panel by Stora Enso: CLT 120 L5s - 20/36

4 625 mm	ω	Layer	Thickness	Width	Orientation	Material
	_°		[truth]	funui		
	. ₹ <u></u>]	1	30.0	606.0	0°	C24 spruce ETA (2019)
	Ŧ	2	9.0	606.0	90°	C24 spruce
	N					ETA (2019)
	97	3	11.0	74.0	90°	C24 spruce
						ETA (2019)
	2	4	20.0	74.0	0°	C24 spruce
	<u>*</u>					ETA (2019)
		5	20.0	74.0	90°	C24 spruce
· <u> </u>						ETA (2019)
74 mm		6	30.0	74.0	0°	C24 spruce
Ribs width are reduced by 20 mm	n for the design					ETA (2019)
		7	297.0	74.0	0°	GL 28h



Multi-storey car park_v2

12.18m - 4 ribs - 200mm - CLT120

Designer PB

Paul Brink

Checker

Page 3/13

02.02.2023

Section Fire: CLT rib panel by Stora Enso: CLT 120 L5s - 20/36

	tсьт	417.0 mm		
Fire protection layering : no additional fire protection	k ₀	d ₀	d _{char,0,h}	d _{ef,h}
	[-]	[mm]	[mm]	[mm]
	0	0	0.0	0.0

Field	Range	from	Until	Width	Moment of inertia	Area	G*A	Kappa	Ely,netto
		[m]	[m]	[cm]	[mm⁴]	[mm ²]	[kN]		[kNm ²]
1	Start	0	0.625	51.70	790,286,500	42,287	27887.95	0.2137	9483.438
1	Center	0.625	11.555	60.60	843,028,100	44,957	29770.3	0.1408	10116.34
1	End	11.555	12.18	51.70	790,286,500	42,287	27887.95	0.2137	9483.438

Layer	Thickness	Width	Туре	Material	Е	G
	[mm]	[mm]			[N/mm ²]	[N/mm ²]
1	30.0 mm	606.0 mm	L	C24 spruce ETA (2019)	12,000	690
2	9.0 mm	606.0 mm	С	C24 spruce ETA (2019)	0	50
3	11.0 mm	74.0 mm	С	C24 spruce ETA (2019)	0	50
4	20.0 mm	74.0 mm	L	C24 spruce ETA (2019)	12,000	690
5	20.0 mm	74.0 mm	С	C24 spruce ETA (2019)	0	50
6	30.0 mm	74.0 mm	L	C24 spruce ETA (2019)	12,000	690
7	297.0 mm	74.0 mm	L	GL 28h	12,600	650

Material values

Material	f _{m,k}	$f_{t,0,k}$	f _{t,90,k}	f _{c,0,k}	f _{c,90,k}	$f_{v,k}$	f _{r,k min}	E _{0,mean}	G _{mean}	G _{r,mean}
	[N/mm ²]	[N/mm ²]	[N/mm ²]	[N/mm ²]	[N/mm²]	[N/mm ²]	[N/mm ²]	[N/mm ²]	[N/mm ²]	[N/mm ²]
C24 spruce ETA (2019)	24.00	14.00	0.12	21.00	2.50	4.00	1.25	12,000.00	690.00	50.00
GL 28h	28.00	22.30	0.50	28.00	2.50	2.50	1.20	12,600.00	650.00	65.00

Load

Load o	case groups								
	Load case category	Туре	Duration	Kmod	Yinf	Ysup	Ψ_0	Ψ_1	Ψ_2
LC1	self-weight structure	G	permanent	0.6	1	1.35	1	1	1
LC2	NL - deadload	G	permanent	0.6	1	1.35	1	1	1
LC3	NL - live load cat. F: traffic area (vehicles <= 25 kN)	Q	medium term	0.8	0	1.5	0.7	0.7	0.6
LC4	NL - snow	Q	short term	0.9	0	1.5	0	0.2	0
LC5	NL - Wind	Q	short term	0.9	0	1.5	0	0.2	0

LC1:self-weight structure

continuous load	
Field	Load at start [kN/m]
1	1.18

LC2: NL - deadload

continuous load	
Field	Load at start
	[kN/m]
1	0.10

	Multi-storey car 12.18m - 4 ribs - 200n	Multi-storey car park_v2 12.18m - 4 ribs - 200mm - CLT120									
storaspeo	Paul Brink		Designer PB	Checker							
SIOIDENSO											
LC3: NL - live load	d cat. F: traffic area (veh	icles <= 25 kN)									
continuous load											
Field	Load at start										
1	2.00										
LC4: NL - snow											
continuous load											
Field	Load at start										
1	[kN/m] 0.56										
LC5: NL - Wind											
continuous load											
Field	Load at start										
-	[kN/m]										
1	0.36										
Ultimate limit stat	e (ULS) - design results	T=0									
		Moments (kNm)									
-50.00				min M=0.00 [kNm] max M=54 74 [kNm]							
0.00	+		+								
	= 4.86/17.98 [kN]			V = 4.86/17.98 [kN]							
50.00-		***************************************									
100.00											
0.00		Axial forces [kN]									
				max N=0.00 [kN]							
ſ				5							
1.00											

© 2023 - Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.



	Mu	Multi-storey car park_v2								Page 6/13
	12.1	8m - 4 rib	s - 200mm -	CLT120						02.02.2023
storaenso	Pau	l Brink					Designer PB		Checker	
lexural stress	analysi	s T=0								
M _{y,d} =	54.74	kNm		f _{m,k} =	28.00	N/mm²				
$M_{z,d} =$	0.00	kNm		$f_{m,k,z} =$	28.00	N/mm ²				
N _{t,d} =	0.00	KIN		γ _m =	1.25	-				
				keve v =	1.10	-				
				k _{h.m.y} =	1.05	-				
				$k_{h,m,z} =$	1.10	-				
				ki =	1.00	-				
$\sigma_{t,d} =$	0.00	N/mm ²		f _{t,0,d} =	15.70	N/mm ²				
$\sigma_{m,y,d} =$	0.00	N/mm²	-	f f	20.75	N/mm ²		1		
Um,z,d -	0.00	IN/11111		Im,z,d —	0.00	IN/IIIII	20.9/			
							23 /0			
Shear stress an	alysis 1	= 0								
V _d =	17.98	kN		f _{v,k} =	2.50	N/mm ²				
				γ _m =	1.25	-				
				k _{mod} =	0.80	-				
T =	0.27	N/mm ²	-	$K_{h,v} = f_{h,v} =$	0.00	- N/mm²		1		
Itilization ratio	0.27	IN/11111		Iv,d —	1.00	IN/IIIII	17 %			
							17 /0			
olling shear a	nalysis	Т=0								
V _d =	17.98	kN		f _{r,k} =	1.25	N/mm ²				
				γ _m =	1.25	-				
	0.40	N1/ma ma 2		k _{mod} =	0.80	-		1		
T _{r,d} =	0.13	N/mm*	~	T _{r,d} =	0.80	N/mm-	4= 04	•		
tilization ratio							17 %			
uckling analys	sis T=0									
M _{y,d} =	54.74	kNm		f _{m,k} =	28.00	N/mm ²				
$M_{z,d} =$	0.00	kNm								
N _{c,d} =	0.00	kN		γ _m =	1.25	-				
				k _{mod} =	0.80	-				
				K _{sys,y} =	1.00	-				
				K _{sys,z} –	1.00	-				
				$k_{h,m,z} =$	1.00	-				
$\sigma_{c,d} =$	0.00	N/mm²		f _{c,0,d} =	17.92	N/mm ²				
σ _{m,y,d} =	4.27	N/mm²		f _{m,y,d} =	17.92	N/mm²				
a . =	0.00	N/mm²	<	$f_{m,z,d} =$	0.00	N/mm²		~		
Om,z,d -										





Alle	Mu	ılti-sto	rey car pa		Page 8/13				
	12.1	18m - 4 ri	ibs - 200mm -	CLT120					02.02.2023
	Pau	l Brink					Desig	ner P	PB Checker
storaenso									
Stress diagram	T=∞								
F	lexural str [N/mm ²	ress]		S	hear stress [N/mm²]				Rolling shear stress [N/mm²]
-4.2	29				\sim				
7	$\overline{}$					7			0.13
						27 -			
		$\langle \rangle$							
		$\langle \rangle$							
		6.13							
Flexural stress a	analysi	s T=∞							
M _{y,d} =	54.74	kNm		f _{m,k} =	28.00	N/mm ²			
N _{t,d} =	0.00	kinm kN		$\tau_{m,k,z} = \gamma_m =$	28.00	N/mm² -			
				k _{mod} = k _{eve v} =	0.80 1.10	2			
				k _{h,m,y} =	1.05	-			
				$k_{h,m,z} = k_i =$	1.10 1.00	-			
$\sigma_{t,d} =$	0.00	N/mm²		f _{t,0,d} =	15.70	N/mm²			
$\sigma_{m,y,d} = \sigma_{m,z,d} =$	6.13 0.00	N/mm² N/mm²	<	f _{m,y,d} = f _{m,z,d} =	20.75	N/mm² N/mm²		~	
Utilization ratio							30 %		
Shoar stross and	alveie 1	[=							
	17.98	kN		f _{v k} =	2.50	N/mm²			
				γ _m =	1.25	-			
				k _{mod} = k _{b v} =	0.80	-			
T _{v,d} =	0.27	N/mm²	<	f _{v,d} =	1.60	N/mm²		~	
Utilization ratio							17 %		
Rolling shear an	alysis	T=∞							
V _d =	17.98	kN		f _{r,k} =	1.25	N/mm²			
				γ _m = k _{mod} =	1.25	-			
Tr,d =	0.13	N/mm²	<	f _{r,d} =	0.80	N/mm²		~	
Utilization ratio							17 %		
Buckling analys	is T=∞								
M _{y,d} =	54.74	kNm		f _{m,k} =	28.00	N/mm²			
M _{z,d} =	0.00	kNm kN		V =	1 25	_			
rec,a —	0.00			k _{mod} =	0.80	-			
				k _{sys,y} = k _{sys,y} =	1.00	-			
				k _{h,m,y} =	1.00	-			
σ _{od} =	0.00	N/mm²		k _{h,m,z} = fccd =	1.00 17.92	- N/mm²			
σ _{m.y,d} =	4.36	N/mm ²		f _{m,y,d} =	17.92	N/mm ²			
σ _{m,z,d} =	0.00	N/mm²	<	t _{m,z,d} =	0.00	N/mm²	24 %	~	
Junzation ratio							2-7 /0		

^{© 2023 -} Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.



	Multi-st 12.18m - 4	orey car pai ribs - 200mm - 0	'k_v2 CLT120					Page 10/13 02.02.2023
storaenso	Paul Brink					Designer	PB	Checker
ULS Fire Rollin	a shear							
Field Dist. [m] 1 0.0	f _{r,k} γ _m [N/mm ²] [-] 1.25 1.0	k _{mod} [-] 0 1.00	k _{fi} [-] [N. 1.15	f _{r,d} /mm²] 1.44	V _d [kN] 9.42	T _{r,d} [N/mm²] 0	Ratio .39 27 %	LCO12
Stress diagram								
~~~~	Flexural stress [N/mm ² ] -5.14		s 	hear stress [N/mm ² ]	••••)		Rolling [	shear stress Wmm ⁺ ]
Flexural stress	analysis Fire							
$\begin{array}{l} M_{y,d} = \\ M_{z,d} = \\ N_{t,d} = \end{array}$	28.70 kNm 0.00 kNm 0.00 kN		$\begin{array}{l} f_{m,k} = \\ f_{m,k,z} = \\ \gamma_m = \\ k_{mod} = \\ k_{sys,y} = \\ k_{h,m,y} = \\ k_{h,m,z} = \\ k_{h,m,z} = \\ k_{l} = \\ k_{l} = \\ k_{l} = \end{array}$	28.00 28.00 1.00 1.10 1.07 1.10 1.00 1.15	N/mm ² N/mm ² - - - - -			
$\sigma_{t,d} = \\ \sigma_{m,y,d} = \\ \sigma_{m,z,d} =$	0.00 N/mm² 9.51 N/mm² 0.00 N/mm²	<	$f_{t,0,d} = f_{m,y,d} = f_{m,z,d} =$	28.21 38.00 0.00	- N/mm² N/mm² N/mm²	~		
Utilization ratio						25 %		
Shear stress ar	nalysis Fire							
V _d =	9.42 kN		$\begin{array}{l} f_{v,k} = \\ \gamma_m = \\ k_{mod} = \\ k_{h,v} = \\ k_{fi} = \end{array}$	2.50 1.00 1.00 0.00 1.15	N/mm² - - -			
T _{v,d} =	0.41 N/mm ²	<	f _{v,d} =	2.88	N/mm²	√ 4.4.9/		
utilization ratio						14 %		
Rolling shear a	nalysis Fire							
V _d =	9.42 kN		$\begin{array}{l} f_{r,k} = \\ \gamma_m = \\ k_{mod} = \\ k_{fi} = \end{array}$	1.25 1.00 1.00 1.15	N/mm² - -			
T _{r,d} =	0.39 N/mm ²	<	f _{r,d} =	1.44	N/mm²	×		
Utilization ratio						27%		





		<b>Multi-sto</b>	orey ca	r par	<b>k_v</b> 2	<b>2</b>					Page 12/13
	D.	Paul Brink	153 - 2001	iiiii - O		0			Designer PB		Checker
storaer	150										
w _{inst} = w[c	har]										
Field	$K_{\text{def,top}}$	$K_{\text{def},\text{Rib}}$	Limit [-]	W _{limit} [mm]		w _{calc.} [mm]	Ratio				
1	1	0.8	L/1	12180	0.0	19.9	0 %				
w _{fin} = w[cl	har] + w[	a.p.∞]									
Field	K _{def,top}	K _{def,Rib}	Limit	Wlimit		Wcalc.	Ratio				
1	1	0.8	[-] L/250	[mm] 48	3.7	[mm] 33.2	68 %				
W _{net,fin} = W	[q.p.] + v	v[q.p.∞]	Limit				Datia				
Field	K _{def,top}	K _{def,Rib}	Limit [-]	Wlimit [mm]		w _{calc.} [mm]	Ratio				
1	1	0.8	L/250	48	8.7	28.3	58 %				
Vibration	analysis										
General											
Total mas	SS width						4	7.53	[t] [m]	1	
Stiffness	Longitudi	inal direction					311	35.6	[kNm ² ]		
Stiffness Modal da	Cross dir mping	rection					2	208.0	[kNm²] [%]		
α Man weig	. e						-	0.1	[-]		
Modal ma	ass						25	505.0	[kg]		
Analysis											
Criterion			Calc.		Class	s II	Class II		CI. II		
Frequence	y criterio v criterio	n min n	6.554 [ 6.554 [	Hz] Hz]	4.5 [l 6.0 [l	Hz] Hz]		69 % 92 %	<b>√</b>		
Accelerat	tion criter	ion	0.203	m/s²]	0.1 [i	m/s²] mml	2	03 %	1		
					[.						
Support r	eaction										
Load case	category	1		k _{mod}	Av	Bv					
self-weigh	t structur	e		0.6	ا] 4.48	kNJ 4.48					
NL - deadl	oad			0.6	4.48 0.38	4.48 0.38					
NL - live lo	ad cat. F	: traffic area (v	/ehicles	0.8	0.38 7.61	0.38 7.61					
<= 25 kN)					0.00	0.00					
INL - Show				0.9	0.00	0.00					
NL - Wind				0.9	1.37 0.00	1.37 0.00					
Deferre	decur	anto for this	nalucia								
English titl	e aocum	ents for this a	marysis		D	escription	1				
EN 338	0				E	N 338 - S	' tructural tim	ber?	Strength class	es	
EN 1995-1	-1				El Ci	N 1995-1 ommon r	<ul> <li>-1 - Eurocolules and rul</li> </ul>	de 5: es for	Design of timbe buildings	er structures - Pa	art 1-1: General -
ETA-14/03	349 Rolling st	near - no edge	aluina H	l Blace	E	uropean ·	Technical A	ssess	ment ETA-14/0	349 of 02.10.20	14
EN 1995-1	-2	icai - no euge	giuniy, 11.c	. 01055	E	N 1995-1	-2 - Euroco	de 5 -	– Design of tim	ber structures –	– Part 1-2: General
EN 14080					E	- Structur N 14080	ai fire desig - Timber Sti	n ructur	es - Glued lami	nated timber an	d glued solid
Technical	expertise	122/2011/02:	analysis o	load	tir Ve	nber - Re erificatior	equirements of the load	bear	ing capacity and	d the insulation	criterion of CLT
bearing ca	pacity an	d separation p	performanc	e of CL	T st	ructures	with Stora E	inso (	CLT		
Technical	expertise	2434/2012 - 1	BB: failure f	ime tf o	f Ex	xpertise o	on failure tin	ne tf c	of gypsum wall f	ire boards accor	rding to ON B3410
gypsum fir	e boards	(GKF) accord	ing to ON E	5 3410	ar	ia gypsui	n wall boar	us typ	e DF according	10 EN 520	

## Example Calculation CLT Closed Rib Panel

		<b>Mu</b> Clos	lti-store ed 12.18m	<b>y car</b> - 4 ribs	<b>park_</b> - 200m	_ <b>v2</b> m - CL	.T120 c	Pa 02	Page 1/14 02.02.2023			
		Paul	Brink						Designe	r PB	CI	necker
store	JENSO											
Syster	m											
			q.=0.56 [kN/m q.=0.36 [kN/m q.=2.00 [kN/m q.=0.10 [kN/m				LC4: NL - LC5: NL - LC3: NL - LC2: NL -	- snow - Wind - live load - deadloa	i cat. F: traffic area ( d	vehicles <= 25	kN)	
			q _k =1.42 [kN/m				LC1:self-	weight str	ructure			
7	<del>∕∕.</del>   <del>•</del>		12.180 [m]									
Globa	l utilization	ratio										76 %
ULS		28 %	ULS Fire	1	5 % SL	.s		76 %	SLS Vibration	73 %	Support	-1 %
Sectio	on: CLT rib j	panel I	oy Stora En	so: CLT	120 L5s	- 14/20	0 - CLT	120 L5:	5			
	62	25 mm					Laye	r 1	[hickness	Width	Orientation	Material
		N / / / / N / / / / / / / / / / / / / /	\\\////К \\\////К \\\///К	120 mm			1		30.0	607.0	0°	C24 spruce ETA (2019)
				200			2		20.0	607.0	90	ETA (2019)
				mm			3		20.0	607.0	0°	C24 spruce ETA (2019)
	¥7774			120			4		20.0	607.0	90°	C24 spruce ETA (2019)
		NZZZA	¥///K	<u>+</u>			5		30.0	607.0	0°	C24 spruce
	14	10 mm					6		200.0	140.0	0°	GL 28h
Ribs v	vidth are re	duced	by 20 mm f	or the de	sign		7		30.0	607.0	0°	C24 spruce
							8		20.0	607.0	90°	C24 spruce
							9		20.0	607.0	0°	C24 spruce
							10		20.0	607.0	90°	C24 spruce
							11		30.0	607.0	0°	C24 spruce ETA (2019)
							tсьт		440.0 mm			(,
ULS T	=0											
Field	Range	from	Until	Width	Mome	nt of in	ertia	Area	G*A	Kappa	Ely,netto	

Field	Range	from	Until	Width	Moment of inertia	Area	G*A	Kappa	Ely,netto
		[m]	[m]	[cm]	[mm ⁴ ]	[mm ² ]	[kN]		[kNm ² ]
1	Start	0	0.625	48.10	1,752,024,000	85,088	58581.12	0.3739	21024.29
1	Center	0.625	11.555	60.70	2,190,437,000	101,216	70112.65	0.2926	26285.24
1	End	11.555	12.18	48.10	1,752,024,000	85,088	58581.12	0.3739	21024.29

Layer	Thickness	Width	Туре	Material	E	G
	[mm]	[mm]			[N/mm ² ]	[N/mm ² ]
1	30.0 mm	607.0 mm	L	C24 spruce ETA (2019)	9,600	552
2	20.0 mm	607.0 mm	С	C24 spruce ETA (2019)	0	40
3	20.0 mm	607.0 mm	L	C24 spruce ETA (2019)	9,600	552
4	20.0 mm	607.0 mm	С	C24 spruce ETA (2019)	0	40

all		Multi	-storey	car p	bai	'k_v2				Page	2/14
		Closed	12.18m -	4 ribs -	20		сору			02.02	.2023
	W.	Paul Br	ink					Designe	r PB	Chec	ker
storae	INSO										
Layer	Thickne [mm]	ess 1	Width [mm]	Туре	Э	Mate	rial	E [N/mm²] [[	G N/mm²]		
5	30.0	) mm	607.0 mr	n	L	C24 sprud	ce ETA (2019)	9,600	552		
6	200.0	0 mm	140.0 mr	n	L	004	GL 28h	10,080	520		
8	20.0	) mm	607.0 mr	n n	L C	C24 sprud C24 sprud	ce ETA (2019)	9,600	552 40		
9	20.0	0 mm	607.0 mr	n	Ľ	C24 sprud	ce ETA (2019)	9,600	552		
10	20.0	) mm	607.0 mr	n	С	C24 sprud	ce ETA (2019)	0	40		
11	30.0	) mm	607.0 mr	n	L	C24 sprud	ce ETA (2019)	9,600	552		
шет-	•										
ULS 1=•	×										
The lat	Denes	6	11-41	14/: -141-		and a firm attac	A	0*4	Kanna	Elemente	
Field	Range	from	Until	width	IVIC	frame#1	Area	G A	карра	Ely,netto	
1	Ctort	[m]	[m]	[cm]		[mm*]	[mm²]	[KN]	24 0.2720	[KNm ² ]	
1	Center	0.625	0.020	48.10		1,350,629,000	00,328 78 734	45604.	34 0.3739 74 0.2926	20254 43	
1	End	11.555	12.18	48.10		1,350,629,000	66,328	45604.3	34 0.3739	16207.55	
			14/2 141	-				-	0		
Layer	Thickne	ess	Width	Туре	Э	Mate	rial	E	G		
	[mm]	]	[mm]					[N/mm ² ] [I	N/mm²]		
1	30.0	) mm	607.0 mr	n	L	C24 sprud	CE ETA (2019)	7,385	425		
3	20.0	) mm	607.0 mr	n	ĩ	C24 sprud	ce ETA (2019)	7 385	425		
4	20.0	0 mm	607.0 mr	n	c	C24 sprud	ce ETA (2019)	0	31		
5	30.0	) mm	607.0 mr	n	L	C24 sprud	ce ETA (2019)	7,385	425		
6	200.0	0 mm	140.0 mr	n	L	001	GL 28h	8,129	419		
7	30.0	) mm	607.0 mr	n	L	C24 spruc	CE ETA (2019)	7,385	425		
0 9	20.0	) mm	607.0 mr	n	ĩ	C24 sprud	ce FTA (2019)	7 385	425		
10	20.0	) mm	607.0 mr	n	č	C24 sprud	ce ETA (2019)	0,000	31		
11	30.0	0 mm	607.0 mr	n	L	C24 sprud	ce ETA (2019)	7,385	425		
SLS T=0	)										
Field	Range	from	Until	Width	Mo	oment of inertia	Area	G*A	Kappa	Ely,netto	
		[m]	[m]	[cm]		[mm ⁴ ]	[mm²]	[kN]		[kNm ² ]	
1	Start	0	0	48.10		2,190,030,000	106,360	73226	0.4 0.3739	26280.36	
1	Center	12 18	12.18	60.70 48.10		2,738,046,000	126,520	87640	0.8 0.2926	32856.55	
	Liid	12.10	12.10	40.10		2,100,000,000	100,000	10220	.+ 0.0700	20200.00	
Layer	Thickne	ess	Width	Туре	э	Mate	rial	E	G		
	[mm]	]	[mm]					[N/mm²] [I	N/mm²]		
1	30.0	) mm	607.0 mr	n	L	C24 sprud	ce ETA (2019)	12,000	690		
2	20.0	) mm	607.0 mr	n	С	C24 sprud	ce ETA (2019)	0	50		
3	20.0	0 mm	607.0 mr	n	L	C24 sprud	ce ETA (2019)	12,000	690		
4	20.0	) mm	607.0 mr	n	C I	C24 sprud	CE ETA (2019)	12 000	50		
6	200.0	) mm	140.0 mr	n	Ľ	OZ4 Sprut	GL 28h	12,600	650		
7	30.0	) mm	607.0 mr	n	Ē	C24 sprud	ce ETA (2019)	12,000	690		
8	20.0	) mm	607.0 mr	n	С	C24 sprud	ce ETA (2019)	0	50		
9	20.0	0 mm	607.0 mr	n	L	C24 sprud	ce ETA (2019)	12,000	690		
10	20.0	) mm ) mm	607.0 mr	n n	C I	C24 sprue	Ce ETA (2019)	12 000	50 690		
	00.0	- (1111	001.0111		-	524 Sprut	20 217 (2010)	.2,000			
SLS T=•	ø										
Field	Range	from	Until	Width	Mo	oment of inertia	Area	G*A	Kappa	Ely,netto	
	.9-	[m]	[m]	[cm]		[mm ⁴ ]	[mm²]	[kN]		[kNm ² ]	
1	Start	0	0	48.10		2,214,530.000	113,710	77776	.4 0.3739	26574.36	
1	Center	Ő	12.18	60.70		2,762,546,000	133,870	92190	.8 0.2926	33150.55	
1	End	12.18	12.18	48.10		2,214,530,000	113,710	77776	0.3739	26574.36	

		<b>Mul</b> f Close	<b>ti-storey c</b> d 12.18m - 4	<b>car park</b> ribs - 200n	_ <b>v2</b> 1m - CLT	[120 cop	ру			P 02	age 3/14 2.02.2023
storae	Inso	Paul I	Brink					Desigr	ier PB	С	hecker
Layer	Thickne [mm]	ess	Width [mm]	Туре	r	Material		E [N/mm²]	G [N/mm²]		
1 2 3 4 5 6 7 8 9 10 11	2000 2000 2000 2000 2000 2000 2000 200	9 mm 9 mm 9 mm 9 mm 9 mm 9 mm 9 mm 9 mm	607.0 mm 607.0 mm 607.0 mm 607.0 mm 607.0 mm 607.0 mm 607.0 mm 607.0 mm 607.0 mm 607.0 mm	L C L C L C L C L	C24 s C24 s	spruce E spruce E spruce E spruce E spruce E spruce E spruce E spruce E spruce E spruce E	TA (2019) TA (2019) TA (2019) TA (2019) TA (2019) GL 28h TA (2019) TA (2019) TA (2019) TA (2019) TA (2019)	$\begin{array}{c} 12,000\\ 0\\ 12,000\\ 0\\ 12,000\\ 15,750\\ 12,000\\ 0\\ 12,000\\ 0\\ 12,000\\ 0\\ 12,000\\ \end{array}$	690 50 690 50 690 813 690 50 690 50 690		
Section	Fire: CLT	rib par	nel by Stora Er ►	150: CLT 12	0 L5s - 1	4/20 - CL Layer	T 120 L5s. Thickn	ess	Width	Orientation	Material
111				120 n		1	[r 3	nm] 30.0	[mm] 607.0	0°	C24 spruce
				1m 20		2	2	20.0	607.0	90°	C24 spruce ETA (2019)
				0 mm		3	2	20.0	607.0	0°	C24 spruce ETA (2019)
				39 m		4	2	20.0	607.0	90°	C24 spruce ETA (2019)
	4					5	3	30.0	607.0	0°	C24 spruce ETA (2019)
Ribs wic	14 Ith are red	0 mm ¹ luced b	by 20 mm for t	he design		6 7	20 3	00.0 80.0	140.0 607.0	0° 0°	GL 28h C24 spruce ETA (2019)

140 mm	6	200.0	14	10.0	0°	GL 28h
Ribs width are reduced by 20 mm for the design	7	30.0	60	07.0	0°	C24 spruce
······································						ETA (2019)
	8	9.0	60	07.0	90°	C24 spruce
						ETA (2019)
	9	11.0	14	10.0	90°	C24 spruce
						ETA (2019)
	t _{CLT}	370.0				
		mm				
Fire resistance class: enter minutes	Time	80 min				
Fire protection layering : no additional fire protection	k ₀	d ₀	d _{char,0,h}	d _{ef,h}		
	[-]	[mm]	[mm]	[mm]		
	0	0	0.0	0.0		

Field	Range	from [m]	Until [m]	Width [cm]	Moment of inertia [mm ⁴ ]	Area [mm²]	G*A [kN]	Kappa	Ely,netto [kNm²]
1	Start	0	0.625	48.10	1,090,202,000	82,310	55963.35	0.5368	13082.43
1	Center	0.625	11.555	60.70	1,325,280,000	96,170	65835.45	0.4843	15903.36
1	End	11.555	12.18	48.10	1,090,202,000	82,310	55963.35	0.5368	13082.43

Layer	Thickness	Width	Туре	Material	Е	G
	[mm]	[mm]			[N/mm ² ]	[N/mm ² ]
1	30.0 mm	607.0 mm	L	C24 spruce ETA (2019)	12,000	690
2	20.0 mm	607.0 mm	С	C24 spruce ETA (2019)	0	50
3	20.0 mm	607.0 mm	L	C24 spruce ETA (2019)	12,000	690
4	20.0 mm	607.0 mm	С	C24 spruce ETA (2019)	0	50
5	30.0 mm	607.0 mm	L	C24 spruce ETA (2019)	12,000	690
6	200.0 mm	140.0 mm	L	GL 28h	12,600	650
7	30.0 mm	607.0 mm	L	C24 spruce ETA (2019)	12,000	690
8	9.0 mm	607.0 mm	С	C24 spruce ETA (2019)	0	50
9	11.0 mm	140.0 mm	С	C24 spruce ETA (2019)	0	50

Material values	5									
Material	f _{m,k}	f _{t,0,k}	<b>f</b> _{t,90,k}	f _{c,0,k}	f _{c,90,k}	f _{v,k}	f _{r,k min}	E _{0,mean}	G _{mean}	G _{r,mean}
	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]	[N/mm²]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]
C24 spruce ETA (2019)	24.00	14.00	0.12	21.00	2.50	4.00	1.25	12,000.00	690.00	50.00
GL 28h	28.00	22.30	0.50	28.00	2.50	2.50	1.20	12,600.00	650.00	65.00

### Load

 $\Psi_2$ 

1

0

0

0.6

Page 4/14 Multi-storey car park_v2 Closed 12.18m - 4 ribs - 200mm - CLT120 copy 02.02.2023 Paul Brink Designer PB Checker storaenso Load case groups Load case category Duration Kmod  $\Psi_0$  $\Psi_1$ Туре Yinf Ysup NL - deadload cat. F: traffic area (vehicles <= 25 kN) NL - snow 0.6 0.6 0.8 LC1 1.35 G permanent 1 1 1 LC2 LC3 G Q permanent 1 0 1.35 1.5 0.7 medium 0.7 term Q LC4 0.9 0 0.2 short 0 1.5 term LC5 NL - Wind Q 0.9 0 0 0.2 1.5 short term LC1:self-weight structure continuous load Field Load at start [kN/m] 1 1.42 LC2: NL - deadload continuous load Field Load at start [kN/m] 1 0.10 LC3: NL - live load cat. F: traffic area (vehicles <= 25 kN) continuous load Field Load at start [kN/m] 2.00 1 LC4: NL - snow continuous load Field Load at start [kN/m] 1 -0.56 LC5: NL - Wind continuous load Field Load at start [kN/m] 1 0.36

^{© 2023 -} Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.



### ULS Flexural design T=0

		-							
Field	Dist.	γm	k _{mod}	k _{sys,y}	k _{h,m,y}	k _{h,m,z}	$\mathbf{f}_{m,k}$	$\mathbf{f}_{m,y,d}$	$\mathbf{f}_{t,0,d}$
	[m]	[-]	[-]	[-]	[-]	[-]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]
1	5.47	1.25	0.80	1.10	1.00	1.00	24.00	16.90	8.96
Field	f _{c,0,d}	M _{y,d}	N _{c,d}	N _{t,d}	$\sigma_{\text{m,y,d}}$	$\sigma_{c,d}$	$\sigma_{t,d}$	Ratio	
	[N/mm ² ]	[kNm]	[kN]	[kN]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]		
1	13.44	58.62	0.00	0.00	4.71	0.00	0.00	28 %	LCO2

#### ULS Axial force design T=0

Field	Dist.	$f_{t,0,k}$	f _{c,0,k}	Υm	k _{mod}	k _{sys,y}	k _{h,m,y}	k _{h,m,z}	k,	f _{m,y,d}
	[m]	[N/mm ² ]	[N/mm ² ]	[-]	[-]	[-]	[-]	[-]	[-]	[N/mm ² ]
1	5.47	14.00	21.00	1.25	0.80	1.10	1.00	1.00	1.00	16.90
Field	Dist.	$M_{y,d}$	f _{c,0,d}	N _{c,d}	f _{t,0,d}	N _{t,d}	$\sigma_{c,d}$	$\sigma_{t,d}$	Utilization	
	[m]	[kNm]	[N/mm ² ]	[kN]	[N/mm ² ]	[kN]	[N/mm ² ]	[N/mm ² ]		
1	5.47	58.62	13.44	0.00	8.96	0.00	0.00	4.39	49 %	LCO2

Page 6/14 Multi-storey car park_v2 Closed 12.18m - 4 ribs - 200mm - CLT120 copy 02.02.2023 Paul Brink Designer PB Checker storaenso ULS Shear analysis T=0 Field Dist.  $\mathbf{f}_{\mathbf{v},\mathbf{k}}$  $V_{d}$ Ratio k_{mod} k_{cr} f_{v,d} T_{v,d} γm [N/mm²] . [-] 1.25 [-] [-] [kN] [N/mm²] [m] [N/mm²] 0.0 0.80 1.00 1.60 19.25 0.43 27 % LCO2 2.50 1 ULS Rolling shear T=0 Field Dist.  $V_{d}$ Ratio f_{r,k} ٧m k_{mod} f_{r,d} T_{r.d} [-] 0.80 [-] 1.25 [kN] [N/mm²] [N/mm²] [m] [N/mm²] 1.25 0.80 0.20 0.0 19.25 25 % LCO2 1 Stress diagram T=0 Shear stress [N/mm²] Flexural stress [N/mm²] Rolling shear stress [N/mm²] -4.71 ---0.43 20 4.71 Flexural stress analysis T=0  $f_{m,k} = f_{m,k,z} = \gamma_m = k_{mod} = k_{sys,y} = k_{tsm,y} = k_$ 58.62 kNm 0.00 kNm 0.00 kN  $M_{y,d} = M_{z,d} =$ 24.00 N/mm² 24.00 N/mm² 24.00 N/mm² 1.25 -0.80 -1.10 -1.00 -1.00 -8.96 N/mm² N_{t,d} = k_{h,m,y} = k_{h,m,z} = k_l =  $\begin{aligned} & f_{t,0,d} = \\ & f_{m,y,d} = \\ & f_{m,z,d} = \end{aligned}$  $\sigma_{t,d} = \sigma_{m,y,d} =$ 0.00 N/mm² 4.71 N/mm² 0.00 N/mm² 16.90 N/mm² 0.00 N/mm² 16.90  $\sigma_{m,z,d} =$ < Utilization ratio 28 % Shear stress analysis T=0 2.50 N/mm² 1.25 -0.80 -0.00  $f_{v,k} =$  $\gamma_m =$  $k_{mod} =$  $k_{h,v} =$  $f_{mod} =$ V_d = 19.25 kN 0.00 -1.60 N/mm² T_{v.d} = 0.43 N/mm²  $f_{vd} =$ Utilization ratio 27 % Rolling shear analysis T=0  $\begin{array}{l} f_{r,k} = \\ \gamma_m = \\ k_{mod} = \\ f_{r,d} = \end{array}$ 1.25 N/mm² 1.25 -V_d = 19.25 kN 2 0.80 0.20 N/mm² 0.80 N/mm² T_{r.d} = Utilization ratio 25 %

^{© 2023 -} Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.





Ultimate limit state (ULS) - design results  $\mathsf{T}{=}\infty$ 

all	10	Mul	ti-store	Page 8/1	4							
		Close	02.02.20	23								
		Paul Brink De							r PB	Checker		
ULS Flexural design T=∞												
Field	Dist. [m]	Υm [-]	K _{mod}	K _{sys,y} [-]	K _{h,m,y} [-]	K _{h,m,z}	t _{m,k} [N/mm²]	f _{m,y,d} [N/mm²]	f _{t,0,d} [N/mm²]			
1	5.47	1.25	0.80	1.10	1.00	1.00	24.00	16.90	8.96			
Field	f _{c,0,d}	M _{y,d}	N _{c,d}	N _{t,d}	σ _{m,y,d}	σ _{c,d}	σ _{t,d}	Ratio				
1	[IN/mm ² ] 13.44	[KINM] 58.62	[KIN] 0.00	[KN] 0.00	[N/mm²] -4.70	[IN/mm²] 0.00	[N/mm²] 0.00	28 %	LCO30			
			_									
ULS A	xial force	design '	f=∞ £		le .	k	k	Ŀ	k	f		
Field	Iml	^T t,0,k [N/mm ² ]	I _{c,0,k} [N/mm ² ]	Υm [-]	Kmod	K _{sys,y} [-]	Kh,m,y	K _{h,m,z}	к _і [-]	I _{m,y,d}		
1	5.47	14.00	21.00	1.25	0.80	1.10	1.00	1.00	1.00	16.90		
Field	Dist.	M _{y,d}	f _{c,0,d}	N _{c,d}	f _{t,0,d}	N _{t,d}	σ _{c,d}	σ _{t,d}	Utilization			
1	[m] 5.47	[KNM] 58.62	[N/mm ² ] 13.44	[KN] 0.00	[N/mm²] 8.96	[KN] 0.00	[N/mm²] 0.00	[N/mm²] 4.38	49 %	LCO30		
ULS S	hear anal	ysis T=∞										
Field	Dist.	f _{v,k} [N/mm²]	Υm [_]	K _{mod}	K _{cr}	f _{v,d} [N/mm	ר א 1 גע	/d T _v	_{,d} Rat	tio		
1	0.0	2.50	1.25	0.80	1.00	[IN/IIII	1.60	19.25	0.43 2	27 % LCO30		
ULS R	Diet	eari=∞ f.	V	k		F .	ν.	τ.	Ra	tio		
i leiu	[m]	νr,κ [N/mm	²] [-]	[-]	a [N/i	mm²]	[kN]	[N/mm ² ]	i ka			
1	0.0	-	1.25 1.2	5 (	).80	0.80	19.25		0.20 2	25 % LCO30		
Stress	Stress diagram T=∞											
		Elevural etre				Shoar etr				alling choor stress		
	-4.70	[N/mm ² ]	55			[N/mm ²			P.	[N/mm ² ]		
	-4.70					N				$\mathbb{N}$		
	7						$\searrow$					
			<u> </u>							0.20		
			4.70			V				V		
Flexur	al stress	analysis	T=∞									
1	И _{у.d} = И _{z.d} =	58.62 0.00	kNm kNm		f _{m,k} f _{m,k,z}	= 24.0 = 24.0	00 N/mm² 00 N/mm²					
	N _{t,d} =	0.00	kN		Ym	= 1.2	25 - 80 -					
					K _{sys,y}	= 1.	10 -					
					K _{h,m,y} K _{h,m,z}	= 1.0 = 1.0	00 - 00 -					
	a=	0.00	N/mm ²		ki fi o i	= 1.0 = 8.0	00 - 96 N/mm²					
σr	n,y,d =	-4.70	N/mm²		f _{m,y,d}	= 16.9	90 N/mm ²					
σ, Utiliza	_{n,z,d} = tion ratio	0.00	N/mm²	<	f _{m,z,d}	= 0.0	UU N/mm²	28 %				
20 /0												
Shear stress analysis T=∞												
	V _d =	19.25	kN		f _{v,k}	= 2.5	50 N/mm²					
					γ _m k _{mod}	- 1.2 = 0.8	20 - 80 -					
	Tvd =	0.43	N/mm²	<	k _{h,v} f. a	= 0.0 = 1.6	00 - 60 N/mm²	~				
Utiliza	tion ratio				••,0	1.		27 %				

	Multi-storey car park_v2 Closed 12.18m - 4 ribs - 200mm - CLT120 copy									9/14 2.2023
storgenso	Pau	l Brink			Designer	РВ	Chec	ker		
								_		
Rolling shear an	alysis	T=∞								
V _d =	19.25	kN		$f_{r,k} = \gamma_m = k_{mod} =$	1.25 1.25 0.80	N/mm² - -				
τ _{r,d} =	0.20	N/mm²	<	f _{r,d} =	0.80	N/mm²	✓			
Utilization ratio							25 %			
Buckling analys	is T=∞									
$M_{y,d} = M_{z,d} = N_{c,d} = N_{c,d}$	58.62 0.00 0.00	kNm kNm kN		$f_{m,k} =$ $Y_m =$ $k_{mod} =$ $k_{sys,y} =$ $k_{sys,z} =$ $k_{h,m,y} =$ $k_{h,m,z} =$	28.00 1.25 0.80 1.00 1.00 1.00 1.00	N/mm ²				
$\sigma_{m,y,d} = \sigma_{m,z,d} =$	2.94 0.00	N/mm ² N/mm ²	<	$f_{m,y,d} = f_{m,z,d} =$	17.92 17.92 0.00	N/mm ² N/mm ²	$\checkmark$			
Utilization ratio							16 %			

Utilization ratio





© 2023 - Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.




Service limit state design (SLS) - design results





	Multi-storey		Page 13/14							
		4 1105 - 2001	- C	,LI 12	о сору	D		02.02.2023		
storaenso	Paul Brink					Design	er PB	Checker		
Vibration analysi	s									
Analysis										
Criterion Frequency criteri Frequency criteri Acceleration crite Stiffness criterior	C. on min 6. on 6. prion 0. n 0.	alc. 161 [Hz] 161 [Hz] 202 [m/s²] 367 [mm]	Class 4.5 [Hz 6.0 [Hz 0.1 [m 0.5 [m	 :] :s²] m]	Class II 73 % 97 % 202 % 73 %	CI. II ✓				
Support reaction										
Load case catego	ry	k _{mod} A	۸ _۷ ۲۳۷	B∨						
self-weight structu	ire	0.6	5.42 5.42	5.42 5.42						
NL - deadload		0.6	0.38	0.38						
NL - live load cat. <= 25 kN)	F: traffic area (vehicle	es 0.8	7.61	7.61						
NL - snow		0.9	0.00	0.00						
NL - Wind		0.9	0.00 1.37 0.00	0.00 1.37 0.00						
Reference docum	nents for this analys	is								
English title			Des	criptio	n					
EN 338 EN 1995-1-1 ETA-14/0349			EN EN Cor Eur	338 - 8 1995-1 nmon i opean	Structural timber I-1 - Eurocode 5: rules and rules fo Technical Asses	? Strengt Design o br building sment E1	h classes of timber structure js FA-14/0349 of 02	es - Part 1-1: General - .10.2014		
Expertise Rolling s EN 1995-1-2	shear - no edge gluin	g, H.J. Blass	Exp EN — S	Expertise on Rolling shear for CLT EN 1995-1-2 - Eurocode 5 — Design of timber structures — Part 1-2: Genera — Structural fire design						
EN 14080 Technical expertis	e 122/2011/02: analy	sis of load	EN timt Ver	EN 14080 - Timber Structures - Glued laminated timber and glued solid timber - Requirements Verification of the load bearing capacity and the insulation criterion of CI T						
bearing capacity a elements	and separation perform	mance of CLT	stru	ctures	with Stora Enso	CLT				
Technical expertis gypsum fire board	e 2434/2012 - BB: fa s (GKF) according to	ilure time tf of ON B 3410	Exp and	ertise gypsu	on failure time tf m wall boards ty	of gypsur pe DF ac	n wall fire boards cording to EN 52	according to ON B3410		
EN 1990 ÖNorm B 1995-1-	1 NA		EN ÖN	1990 - ORM E	Eurocode ? Bas EN 1995-1-1 - Au	stria - Na	ctural design tional Annex – N f timbor structure	ationally determined		
ÖNorm B 1995-1-	2 NA		Cor ÖN	nmon i DRM F	ules and rules fo	or building stria - Na	is IS IS	s - Part 1-1: General-		
			timt	er stru cificatio	ons concerning C	2: Genera	Il ? Structural fire EN 1995-1-2, nati	design ? National ional comments and		
Fire safety in timb	er buildings - technica	al guildeline fo	nati r Fire	onal su safety	upplements in timber buildin	igs - tech	nical guideline fo	r Europe; publishes by		
Europe National specificat 1-2, national comr	tions concerning ÖNC ments and national su	DRM EN 1995 Ipplements,	- ÖN 1-2,	DRM E DRM E natior	ical Research Ins N 1995-1-2 - Na nal comments an	stitute of s itional spe d nationa	Sweden ecifications conce I supplements, ch	erning ÖNORM EN 1995- napter 12		
EN 1992-1-1			EN	1992-1	I-1 - Eurocode 2:	Design o	of concrete struct	ures - Part 1-1: General		
ETA-11/0030			ETA	-11/00 use in 1	30 European Te	chnical A	pproval; Rothobl	aas; Self-tapping screws		
Z-9.1-472			DIB 6,5,	t techn WT-T	ical approval Z-9 -6,5, WR-T-9,0 a	9.1-472; S nd WR-T	SFS intec GmbH; -13 as connector	SFS connectors WT-S- s for timber construction		
Expertise Rolling s Deriviation of the t elements for the e focus sts 2.2.3 1	shear, H.J. Blass tributary width in CLT ngineering practice; ⁻	-rib deck ſU-Graz,	Exp Der prac	ertise viatior ctice; T	on rolling shear s of the tributary v U-Graz, focus_s	strength a width in C ts 2.2.3_	nd rolling shear r CLT-rib deck elem 1	modulus of CLT panels nents for the engineering		
ÖNORM EN 1995	-1-1_NA, chapter 7.3		ÖN para Cor	ORM E ameter	EN 1995-1-1 - Au s – Eurocode 5: rules and rules fo	stria - Na Design o or building	tional Annex – N f timber structure js; chapter 7.3	ationally determined s – Part 1-1: General-		
			Sell	aping						

	Multi-storey car par 1x12.18m span	Page 1/7 02.02.2023				
storacinso	Paul Brink			Designer PB		Checker
Storderiso						
System						
	q⊾=0.56 [kN/m]	LC4: NI				
•		L04. NE -	show			
•	q x=0.36 [kN/m]	LC5: NL -	Wind			
	q_=2.00 [kN/m]	LC3: NL -	live load ca	t. F: traffic area (vehicl	es <= 25 kN)	
	a.=0.10.lkN/ml	¥				
•	dr-or to fermini	LC2: NL -	deadload			
	q_k=2.45 [kN/m]	LC1:self-v	eight struct	ure		
tin.	Field 1	Â				
A	12.180 [m]	▶				
						70.0/
ULS 19 %	ULS Fire 9 % SLS	48 % SLS Vibra	ition	70 % Support	-1 % Voi	id -1 %
Section: LVL G	-X flatwise 100/48	Laver		Thickness	Orientation	Material
		1		60.0 mm	90°	LVL G-X
	80 mm	2		60.0 mm	90°	LVL G-X
+	÷	3		60.0 mm	90°	LVL G-X
. 100	o mm	4		60.0 mm	90°	LVL G-X
		5		60.0 mm	90°	LVL G-X
		6		60.0 mm	90°	LVL G-X
		7		60.0 mm	90°	LVL G-X
		8		60.0 mm	90°	LVL G-X
		tclt		480.0 mm		natwise
Section Fire: LV	/L G-X flatwise 100/48					
	Ť	Layer 1		Thickness 60.0 mm	Orientation 90°	Material LVL G-X
	414 mm	2		60.0 mm	90°	flatwise LVL G-X
		3		60.0 mm	90°	flatwise LVL G-X
100	I0 mm	4		60.0 mm	90°	flatwise LVL G-X
		5		60.0 mm	90°	flatwise LVL G-X
		6		60.0 mm	90°	flatwise LVL G-X
		7		54.0 mm	90°	flatwise LVL G-X
		tсьт		414.0 mm		flatwise
Fire resistance of Fire protection la	lass:R 90 ayering : no additional fire protectior	Time	de	90 min	debar 0. v d-r	
		[-]	[mm]	[mm] [mm] 59.0 66.0	[mm] [mm] 58.5 0.0	
					50.0 0.0	

## Example Calculation Solid LVL Panel



		<b>Mult</b> 1x12.1	<b>i-storey</b> 18m span	[,] car parl	k_v2		Page 2/7 02.02.2023						
stora	IENSO	Paul E	Brink				De	esigner PB		Checker			
Materia	al values	5											
Materia	al	f _{m,k} [N/mm²]	f _{t,0,k} [N/mm²]	f _{t,90,k} [N/mm²]	f _{c,0,k} [N/mm²]	f _{c,90,k} [N/mm²]	f _{v,k} [N/mm²]	f _{r,k min} [N/mm²]	E _{0,mean} [N/mm²]	G _{mean} [N/mm²	'] [N	E _{0,5} /mm²]	
LVL G-	X	31.00	26.00	0.00	26.00	2.20	1.30	0.60	10,100.00	120.00	8,8	300.00	
LVL G- edgewi	X	32.00	26.00	6.00	26.00	9.00	4.50	0.00	10,500.00	600.00	8,8	300.00	
Load													
Load c	ase gro	ups											
	G-X 32.00 26.00 6.00 wise d d case groups Load case category self-weight structure NL - deadload NL - live load cat. F: traffic area (vehicles - 25 kN)			Туре	Duration	Kmod	Yinf	Ysup	$\Psi_0$	Ψı	$\Psi_2$		
LC1 LC2 LC3	LC1 self-weight structure LC2 NL - deadload LC3 NL - live load cat. F: traffic area (vehicles <=					permanent permanent medium term	0.6 0.6 0.8	1 1 0	1.35 1.35 1.5	1 1 0.7	1 1 0.7	1 1 0.6	
LC4	NL - sno	w			Q	short	0.9	0	1.5	0	0.2	0	
LC5	NL - Wi	nd			Q	short term	0.9	0	1.5	0	0.2	0	
1.01	lf wai-h	4 of											
LU1:SE	eit-weigr	it structure											

continuous load											
Field	Load at start										
	[kN/m]										
1	2.45										

LC2: NL - deadload

continuous load	
Field	Load at start
	[kN/m]
1	0.10

## LC3: NL - live load cat. F: traffic area (vehicles <= 25 kN)

continuous load	
Field	Load at start
	[kN/m]
1	2.00

## LC4: NL - snow

## LC5: NL - Wind

continuous load	
Field	Load at start [kN/m]
1	0.36



				$k_{h,m,y} =$	0.80	-	
				$k_{h,m,z} =$	0.87	-	
				k _i =	0.92	-	
$\sigma_{t,d} =$	0.00	N/mm ²		f _{t,0,d} =	15.94	N/mm ²	
$\sigma_{m,y,d} =$	3.11	N/mm ²		$f_{m,y,d} =$	16.62	N/mm ²	
$\sigma_{m,z,d} =$	0.00	N/mm ²	<	$f_{m,z,d} =$	18.46	N/mm ²	
Utilization ratio							19 %

^{© 2023 -} Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.







### ULS Fire Flexural design

40.00-

Field	Dist.	f _{m,k}	$\mathbf{f}_{c,0,k}$	f _{t,0,k}	γm	$\mathbf{k}_{mod}$	k _{fi}	k _{sys,z}	k _{h,m,y}	$\mathbf{k}_{h,m,z}$	kı	f _{m,y,d}	f _{m,z,d}	f _{t,0,d}	f _{c,0,d}
	[m]	[N/mm ² ]	[N/mm²]	[N/mm ² ]	[-]	[-]	[-]	[-]	[-]	[-]	[-]	[N/mm ² ] [	N/mm²] [N	/mm²]	[N/mm ² ]
1	6.09	31.00	26.00	26.00	1.20	0.80	1.00	1.00	0.80	0.87	0.92	16.62	18.46	15.94	17.33
Field	Field M _{y,d} M _{z,d}		d.	N _{c,d}	$N_{t,d}$	ν _{t,d} σ _{m,y}		$\sigma_{\text{m,z,d}}$	σ _{m,z,d} σ _{c,d}		$\sigma_{t,d}$		Ratio		
	[kNm]	[kNr	n]	[kN]	[kN]	٩]	V/mm²]	[N/mm ² ]	[	N/mm²]	[N/	mm²]			
1	119.4	42	0.00	0.00	0.0	00	3.11	0.0	0	0.00		0.00	19 %	LCC	)2
ULS Fi	re Shear	analysis													
Field	Dist.	$f_{v,k}$	γm	k _{mod}	k	kh,v	k _{fi}	f	,d	Vd		T _{v,d}	Ratio		
	[m]	[N/mm²]	[-]	[-]		[-]	[-]	[N/n	[N/mm²] [kN]		[N/mm ² ]				
1	0.41	1.30	1.00	1.00	)	0.82	1.	10	1.43	21.27	7	0.08	5	% L(	012

	<b>Mu</b> 1x12	Multi-storey car park_v2 1x12.18m span											
storaenso	Paul	Brink			Desigr	ner PB		Checker					
Flexural stress a	inalysis	Fire											
$M_{y,d} = M_{z,d} = N_{t,d} = \sigma_{m,y,d} $	69.50 0.00 0.00 2.43 0.00	kNm kNm kN N/mm ² N/mm ²	<	$\begin{array}{l} f_{m,k} = \\ f_{m,k,z} = \\ Y_m = \\ k_{mod} = \\ k_{sys,y} = \\ k_{h,m,y} = \\ k_{h,m,z} = \\ k_{l} = \\ k_{l} = \\ f_{t,0,d} = \\ f_{m,y,d} = \end{array}$	31.00 31.00 1.00 1.00 0.82 0.87 0.92 1.10 26.29 27.96 30.46	N/mm ² N/mm ² - - - - N/mm ² N/mm ²		~					
Utilization ratio							9 %						
Shear stress and	alvsis F	ire											
V _d =	21.27	kN		$\begin{array}{l} f_{v,k} = \\ \gamma_m = \\ k_{mod} = \\ k_{h,v} = \\ k_{fi} = \end{array}$	1.30 1.00 1.00 0.82 1.10	N/mm ² - - -							
T _{v,d} =	0.08	N/mm ²	<	f _{v,d} =	1.43	N/mm ²	5 %	~					

### Service limit state design (SLS) - design results



0.8

1

© 2023 - Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.

14.0 0%

		Multi-storey car park_v2 1x12.18m span											Page 6/7 02.02.2023		
storaen	so	Paul Brink						Design	er PB			Check	er		
w _{fin} = w[ch	ar] + w[	a.p.1*kdef													
Field	K _{def}	Limit	Wlimit	Wcalc.	Ra	atio									
1	0.8	[-] L/250	[mm] 48.7	[mm] 23	3.2 48	%									
w = wi	a n 1 + v	vla p 1*kdof													
Field	K _{def}	Limit	Wlimit	Wcalc.	Ra	atio									
	0.0	[-]	[mm]	[mm]		0/									
1	0.8	L/250	48.7	20	J.8 43	%									
Vibration a	analysis														
General															
Total mas Tributary	is width						38.53 4.5	[t] [m]							
Stiffness I	Longitud Cross dir	nal direction					95581.6 2500.0	[kNm²] [kNm²]							
Modal dar	mping	001011					2.0	[%]							
α Man weig	ht						700.0	[-] [N]							
Modal ma	ISS						7036.3	[kg]							
Analysis															
Criterion Frequenc	y criterio	n min	Calc. 6.425	[Hz]	Class 4.5 [ł	s I Hz]	Class II 4.5 [Hz]	Class I	70 %	Class II	70 %	CI. I ✓	CI. II ✓		
Frequenc	y criterio	n	6.425	[Hz] [m/s²]	8.0 [H	lz] [m/s²]	6.0 [Hz]		125 %		93 % 76 %	×			
Stiffness	criterion		0.088	[mm]	0.05	[mm]	0.5 [mm]		35 %		18 %	✓ ✓	✓		
Support re	eaction														
Load case	category	,		$\mathbf{k}_{mod}$	A _V	Bv									
self-weight	structur	Э		0.6	14.91	14.91									
NL - deadlo	oad			0.6	0.61	0.61									
NL - live lo	ad cat. F	: traffic area (	vehicles	0.8	12.18	12.18									
<= 25 KN)					0.00	0.00									
NL - snow				0.9	3.41 0.00	3.41 0.00									
NL - Wind				0.9	2.19 0.00	2.19 0.00									
Poforonco	docum	onte for thie	analveie												
English title	euocum		anaiysis		De	escriptior	1								
EN 338	1				E	1 338 - S	tructural timber ?	Strengt	h classe	es	n De	4 4 4. 0			
EN 1995-1	-1				Er	1995-1 mmon r	<ul> <li>-1 - Eurocode 5: ules and rules for</li> </ul>	Design c building	s s	rstructure	es - Par	t 1-1: G	Seneral -		
EN 1995-1	-2				E1	V 1995-1 Structur	-2 - Eurocode 5 - al fire design	– Desigr	n of timt	per structi	ures —	Part 1-	2: Genera		
EN 14080					El tin	14080 1ber - Re	- Timber Structur	es - Glue	ed lamir	ated timb	er and	glued s	solid		
EN 1990	1005 1 1	NA			El	1990 -	Eurocode ? Basi	s of struc	tural de	esign	ational	v dete-	minod		
	1995-1-1	NA			pa	rameters	s – Eurocode 5: E	Design of	timber	structure:	s – Par	t 1-1: G	ieneral-		
ÖNorm B 1	1995-1-2	NA			Co Öl	ommon r NORM E	ules and rules for N 1995-1-2 - Aus	· building tria - Na	s tional A	nnex - Eu	rocode	5: Des	ign of		
	_				tin	nber stru	ctures ? Part 1-2	Genera	I ? Strue	ctural fire	design	? Natio	onal s and		
ÖN0211-	14005		7.0		na	tional su	plements			- 1-2, IIdli					
UNORM E	N 1995-'	I-1_NA, chapt	ter 7.3		OI pa	NORM E	N 1995-1-1 - Aus s – Eurocode 5: E	itria - Na Design of	tional A timber	nnex – Na structure:	ationall s – Par	y deterr t 1-1: G	mined ieneral-		
CERTIFICA	ATE NO	EUFI29-2000	)0564-C		Co	ommon ri oduct ce	ules and rules for rtificate	building	s; chap	ter 7.3					
SERTINO/		201 129-2000	,		1.1		Tinodio								

		<b>Multi</b> 12.18m	-storey	<b>car parl</b> LVL37 - 51	k_v2					F	² age 1/7 )2.02.20	7 )23
stora		Paul B	rink				De	signer PB	3	(	Checker	
Diord	CIIDO											
System	n											
Ē			a. =0.56 [kN/m]									
t t			di-erea farming			LC4: NL - snow						
]			q .=0.36 [kN/m]			LC5: NL - Wind						
			a. =2.00 [kN/m]									
t t	,		41-2.00 (a and			LC3: NL - live lo	ad cat. F: traffi	c area (vehicl	es <= 25 k	N)		
]			q .=0.10 [kN/m]			LC2: NL - deadl	oad					
			a. =0.51 [kN/m]									
	,		dr-c.or [anni]			LC1:self-weight	structure					
- <del>M</del>	×		Field 1			Ż						
ł	•		12.180 [m]									
Global	utilizatio	on ratio										10/ %
ULS	utilizatit	28 % U	LS Fire	0 %	SLS	52.9	6 SLS Vibr	ation	194 %	Support		-1 %
010		20 /0 0	201110	0,0	020	,		ulon	101 70	oupport		
Section	n: LVL ri	b panel by	Stora Enso	: 1/60								
1.	500 n	nm '	37			Layer	Thickness [mm]	V\ [!	ndth C mm]	rientation	Mater	rial
			Ŧ			1	37.0	5	00.0	0°	LVL >	K se
						2	600.0	1	51.0	0°	LVL S	S wise
			600 mm									
			Ţ									
	51 m	im										
						tc⊥⊤	637.0 mm					
Materia	al values	fmk	ftor	ft oo k	faak	foook	fue	fr k min	Famo	n G	2000	Gemaan
matoria		[N/mm²]	[N/mm²]	[N/mm²]	[N/mm ² ]	] [N/mm²]	[N/mm²]	[N/mm²]	[N/mr	n²] [N/r	nm²]	[N/mm ² ]
LVL S edgewis	se	44.00	35.00	0.80	35.00	6.00	4.20	0.00	13,800	.00 600	).00	
LVL X flatwise	•	36.00	26.00	6.00	26.00	2.20	1.30	0.60	10,500	.00 120	).00	
hattitoo												
Load												
Load c	ase grou	ips				_						
LC1	Load ca self-weig	se category	)		Гуре	<ul> <li>Duration</li> <li>permanent</li> </ul>	Kmod 0.6	Yinf 1	Ysup 1	Ψ ₀ .35	Ψ ₁	Ψ ₂ 1 1
LC2	NL - dea	idload		(	Ğ	permanent	0.6	1	1	.35	1	1 1
LC3	NL - live 25 kN)	load cat. F	traffic area	(vehicles <=	Q	medium term	0.8	0		1.5 0.	/ 0.7	0.6
LC4	NL - sho	w			Q	short term	0.9	0		1.5	0 0.2	2 0

## Example Calculation LVL Open Rib Panel

		<b>Multi-storey car</b> 12.18m - 4 ribs - LVL3	<b>r park_</b> 37 - 51	v2									Pag 02.0	e 2/7 )2.202	23	
store		Paul Brink					De	esigner	PB				Che	cker		
Load	case groups	5														
LC5	Load case NL - Wind	category		Type Q	Duration short term	Kmod	0.9	Yinf	0	Ysup	1.5	$\Psi_0$	0	₽ ₁ 0.2	$\Psi_2$	0
LC1:s	elf-weight s	tructure														
conti	nuous load															
Field		Load at start [kN/m]														
1		0.51														
LC2: N	NL - deadloa	ad														
conti	nuous load															
Field		Load at start [kN/m]														
1		0.10														
LC3: N	NL - live loa	d cat. F: traffic area (veh	icles <= 2	5 kN)												
conti	nuous load															
Field		Load at start [kN/m]														
1		2.00														
LC4: N	NL - snow															
conti	nuous load															
Field		Load at start														
1		[kN/m] 0.56														
LC5: N	NL - Wind															
conti	nuous load															
Field		Load at start														
1		0.36														
Ultima	ate limit stat	e (ULS) - design results	T=0													
	20.00			M	oments [kNm	]										
	-20.00											min max	M=0 < M=3	.00 [kN 35.46 [l	m] (Nm]	
	0.00										-					
	- fii	= 1.86/11.65 [kN]									<b>~</b> _	1.86/1	1.65 [	kN1		
									,, ,,		T					
	20.00-						سر پ									
		<u> </u>	********													
	40.00															



Utilization ratio

	<b>Mu</b> 12.1	8m - 4 ribs -	<b>car par</b> LVL37 - 51	k_v2						Page 4/7 02.02.2023
storaenso	Pau	l Brink					Designe	er PB	}	Checker
Shear stress ana	lysis 1	Γ=0								
V _d =	11.65	kN		$\begin{array}{l} f_{v,k} = \\ \gamma_m = \\ k_{mod} = \\ k_{h,v} = \end{array}$	4.20 1.20 0.80 0.00	N/mm² - -				
T _{v,d} =	0.49	N/mm²	<	f _{v,d} =	2.80	N/mm ²	19.0/			
otilization ratio							10 /0			
Buckling analysi	s T=0									
M _{y.d} = M _{z,d} = N _{c,d} =	35.46 0.00 0.00	kNm kNm kN		$f_{m,k} =$ $Y_m =$ $k_{mod} =$ $k_{sys,y} =$ $k_{sys,z} =$ $k_{h,m,y} =$ $k_{h,m,z} =$	44.00 1.20 0.80 1.00 1.00 1.00 1.00	N/mm ²				
$\sigma_{m,y,d} = \sigma_{m,z,d} = \sigma_{m$	0.00 5.61 0.00	N/mm ² N/mm ²	<	$f_{m,y,d} =$ $f_{m,z,d} =$	23.33 29.33 0.00	N/mm ² N/mm ²	,			
Utilization ratio							19 %			

Ultimate limit state (ULS) - design results  $T=\infty$ 



## ULS Flexural design T=∞

Field	Dist.	γm	k _{mod}	k _{sys,y}	f _{m,k}	f _{m,y,d}	f _{t,0,d}	f _{c,0,d}
	[m]	[-]	[-]	[-]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]
1	6.09	1.20	0.80	1.00	44.00	26.99	21.01	23.33
Field	$M_{y,d}$	N _{c,d}	N _{t,d}	$\sigma_{m,y,d}$	$\sigma_{c,d}$	$\sigma_{t,d}$	Ratio	
	[kNm]	[kN]	[kN]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]		
1	35.46	0.00	0.00	7.55	0.00	0.00	28 %	LCO20

all	10	Mu	lti-sto	orey	car pa	ark_v2						Page	5/7
		12.1	8m - 4	ribs - L	VL37 -	51						02.02	.2023
stora	επεο	Pau	l Brink						Desigr	ner PB		Chec	ker
ULS SP	near ana	lvsis T=	20										
Field	Dist. [m]	f _{v,k}	n²]	Υm [-]	k _{mod} [-]	k _{cr} [-]	f _{v,c} [N/m	m²l	V _d [kN]	T _{v,d} [N/mr	m²]	Ratio	
1	12.18	<b>L</b>	4.20	1.20	0.8	0 1	00	2.80	-11.65	<b>.</b>	0.49	18 %	LCO20
Stress	diagram	<b>T=</b> ∞											
		Flexural str	ess				Shear stress						
	2	- <del>3.86</del>	,				[]	_					
		$\backslash$						$\langle \rangle$					
								9.49 -					
								/					
			$\langle \rangle$										
			7.55				V						
Flexura	al stress	analysis	s T=∞										
M	l _{y,d} = l _{z,d} =	35.46 0.00	kNm kNm			f _{m,k} = f _{m,k,z} =	44.00 44.00	N/mm² N/mm²					
Ν	l _{t,d} =	0.00	kN			γ _m = k _{mod} =	1.20 1.80	:					
						k _{sys,y} = k _{h m v} =	1.00 0.92	-					
						k _{h,m,z} =	1.00	-					
с 	σ _{t,d} =	0.00	N/mm² N/mm²			f _{t,0,d} =	21.01	N/mm² N/mm²					
σ _m	,y,d = ,z,d =	0.00	N/mm ²		<	f _{m,z,d} =	0.00	N/mm ²		×			
Utilizat	ion ratio								28 %				
Shear s	stress ar	nalysis T	[≡∞										
	V _d =	-11.65	kN			f _{v,k} =	4.20	N/mm²					
						k _{mod} =	0.80	-					
т	r _{v,d} =	0.49	N/mm²		<	f _{v,d} =	2.80	N/mm²		~			
Utilizat	ion ratio								18 %				
Bucklin	ng analy	sis T=∞											
M	l _{y,d} =	35.46	kNm kNm			f _{m,k} =	44.00	N/mm²					
N	$I_{c,d} =$	0.00	kN			γ _m =	1.20	-					
						k _{mod} = k _{sys,y} =	0.80 0.80 ·	-					
						k _{sys,z} = k _{h.m.v} =	1.00 1.00	-					
	=	0.00	N/mm²			k _{h,m,z} =	1.00	- N/mm²					
n		0.00				rc,u,d =	20.00						
σ ,	,y,d =	5.70	N/mm ²		2	f _{m,y,d} =	29.33	N/mm ²		1			

^{© 2023 -} Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.



	Multi-store	<b>ey car</b> s - LVL37	<b>park_v</b> 2 - 51	2					Page 7/7 02.02.2023	3
storaenso	Paul Brink					Desigr	ner PB		Checker	
Vibration analysis	6									
General										
Total mass Tributary width Stiffness Longitud Stiffness Cross di Modal damping α Man weight Modal mass	linal direction rection				22.7 1. 26187. 8. 2. 0. 700. 562.	6 [t] 5 [m] 2 [kNm ² ] 4 [kNm ² ] 0 [%] 0 [-] 0 [N] 2 [kg]	]			
Analysis										
Criterion Frequency criteric Frequency criteric Acceleration criter Stiffness criterion	on min on rion	Calc. 9.712 [Hz 9.712 [Hz 0.256 [m/ 0.969 [mr	Clas [] 4.5 [ [] 6.0 [ s ² ] 0.1 [ n] 0.5 [	s II Hz] Hz] m/s²] mm]	Class II 46 9 256 9 194 9	Cl. II ∕₀ ∕₀ ∕₀ ×				
Support reaction				_						
Load case categor	у	Km	od Av I	B∨ kN1						
self-weight structur	е	0.6	6 1.56 1.56	5 1.56 5 1.56						
NL - deadload		0.6	6 0.30	0.30						
NL - live load cat. F <= 25 kN)	F: traffic area (veh	icles 0.8	3 6.09	6.09						
NL - snow		0.9	0.00 9 1.71	0.00						
NL - Wind		0.9	9 1.10 0.00	0.00 0 1.10 0 0.00						
Poforonco docum	onto for this and	hucic								
English title	ionto for this ana	iyala		escriptio	n					
EN 1995-1-1			E	N 1995-	1-1 - Eurocode 5	: Desian	of timber	structures -	Part 1-1: Gener	ral -
EN 1990 ÖNorm B 1995-1-1 ETA-12/0063	NA	C E Ö P C S	ommon N 1990 NORM I arameter ommon FS intec	rules and rules f Eurocode ? Bas EN 1995-1-1 - Au rs – Eurocode 5: rules and rules f AG; Self-tappin	or building sis of stru ustria - Na Design o or building g screws	gs ctural de ational A of timber gs for use in	esign nnex – Nation structures – I n timber cons	nally determine Part 1-1: Gener	d 'al-	
ETA-12/0062 ETA-11/0086 ETA-11/0190			S C( G S(	⊢A intec onstructi H Variou elftaping	AG; ETA-12/00 ons is Angle Bracket screw by Würth	oz; selftaj ts	oping sci	rews for use i	in timber	
Disclaimer										

The software was created to assist engineers in their daily business. The software is an engineering software that is dealing with a very complex matter of structural analysis and building physics analysis. Therefore, this software shall only be operated by skilled, experienced engineers, with a deep understanding of structural engineering and building physics related to timber structures. The user of the software is obliged to check all input values, no matter if they were given by the user or given by default by the software and all results for noussibility.

Dulliding trypace any set increments in the software is obliged to check all input values, no matter if they were given by the user or given by detault by the software is not an encoded of the results of the software should not be relied upon as the basis for any decision or action. Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. The user thas the possibility to make print-outs from the software. Any modification of those are not allowed. Stora Enso Wood Products GmbH does not assume any warranty regarding the software. The software has been developed with utmost diligence, nevertheless Stora Enso Wood Products GmbH, neither expressily nor implicitly, provides any warranty for the general usability of the software, its suitability for a special purpose or for the compatibility of the software. Stora Enso Wood Products GmbH does also not assume any warranty for the general usability of the software, its suitability for a special purpose or for the compatibility of the software is down and data created by the software. Stora Enso Wood Products GmbH is only liable for damages caused by gross negligence or intent through Stora Enso Wood Products GmbH is as well not liable for operational failures or the loss of programs and/or data of the user's data processing system. Applicable Law: These terms of use shall be governed by the laws of Austria excluding however any conflict of laws rules and any laws regarding the Convention of the International Sale of Goods (CISG).

# Example Calculation LVL Semi-open Rib Panel

		<b>Mul</b> Semi-	ti-storey -open 12.18	<b>car parl</b> m - 4 ribs -	<b>k_v2</b> LVL37 - I	LVL43 - 51					Pag 02.0	e 1/8 )2.202:	3
store		Paul	Brink				De	signer PB			Che	cker	
Stord	IEIISO												
System	n												
Ī			q_=0.56 [kN/m]			_C4: NL - snow							
, , ,			a. =0.36 [kN/m]										
			dr-eree (wand			.C5: NL - Wind							
	,		q _k =2.00 [kN/m]		, i	C3: NL - live lo	ad cat. F: traffi	c area (vehicl	es <= 25	kN)			
Ì			q_=0.10 [kN/m]		ı	.C2: NL - deadle	bad						
, i			g⊾=0.65 [kN/m]										
, i i i i i i i i i i i i i i i i i i i			Field 1		<u> </u>	LC 1:self-weight	structure						
	7. •				B								
			12.180 [m]		-1								
Global	utilizatio	on ratio											127 %
ULS		24 %	ULS Fire	0 %	SLS	34 %	6 SLS Vibr	ation	127 %	6 Sup	port		-1 %
Section	n: LVL ri	b panel b	y Stora Enso	: 1/60									
•	500 mr	n M	3			Layer	Thickness [mm]	۷۷ ۱]	/idth mm]	Orienta	ation 1	√aterial	I
						1	37.0	50	0.00	0°	f L	_VL X latwise	
						2	600.0	1	51.0	0°	Ĺ	.VL S ∋dgewis	se
	4+  51 mm  4	1 1 1	600 mm 43 mm			3	43.0	2	18.2	0°	f	-VL X latwise	
						tclt	680.0 mm						
Materia	al values	;											
Materia	al	f _{m,k}	f _{t,0,k}	f _{t,90,k}	f _{c,0,k}	f _{c,90,k}	f _{v,k}	f _{r,k min}	E _{0,r}	mean	G _{mean}	G	r,mean
LVL S		[N/mm ² ] 44.00	[N/mm ² ] 35.00	[N/mm ² ] 0.80	[N/mm ² ] 35.00	[N/mm²] 6.00	[N/mm²] 4.20	[N/mm²] 0.00	[N/n 13,80	nm²] 00.00	[N/mm ² ] 600.00	] [N/	'mm²]
edgewi LVL X	se	36.00	26.00	6.00	26.00	2.20	1.30	0.60	10,50	00.00	120.00		
flatwise LVL X	•	36.00	26.00	6.00	26.00	2.20	1.30	0.60	10,50	00.00	120.00		
flatwise	9												
Load													
Load c	ase grou	se catego	rv.		Type	Duration	Kmod	Vinf	Verm		Ψ, ι	Ψ.	$\Psi_2$
LC1 LC2 LC3	self-weig NL - dea NL - live 25 kN)	ght structu adload e load cat.	re F: traffic area	(vehicles <=	G G Q	permanent permanent medium term	0.6 0.6 0.8	1 1 0	Ysup	1.35 1.35 1.5	1 1 0.7	1 1 0.7	1 1 0.6

0

0

Page 2/8 Multi-storey car park_v2 Semi-open 12.18m - 4 ribs - LVL37 - LVL43 - 51 02.02.2023 Paul Brink Designer PB Checker storaenso Load case groups Type Duration Q short Load case category  $\Psi_0$  $\Psi_1$ Kmod  $\Psi_2$  $\gamma_{\text{inf}}$  $\gamma_{\text{sup}}$ 0.9 0 1.5 0.2 LC4 NL - snow 0 term Q LC5 NL - Wind 0.9 0 1.5 0 0.2 short term LC1:self-weight structure continuous load Field Load at start [kN/m] 1 0.65 LC2: NL - deadload continuous load Field Load at start [kN/m] 1 0.10 LC3: NL - live load cat. F: traffic area (vehicles <= 25 kN) continuous load Field Load at start [kN/m] 2.00 1 LC4: NL - snow continuous load Field Load at start [kN/m] 1 0.56 LC5: NL - Wind continuous load Load at start Field

[kN/m]

0.36

1

^{© 2023 -} Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.



Alle	Multi	i-storey	car park	<_v2						Page 4/8
	Semi-c	pen 12.18	m - 4 ribs -	LVL37 - LVI	_43 - 5	1				02.02.2023
	Paul B	rink					Desigr	ner PB		Checker
storaenso										
Flexural stress	analysis T	=0								
	37.19 kM 0.00 kM 0.00 kM	Vm Vm V		$\begin{array}{l} f_{m,k} = \\ f_{m,k,z} = \\ \gamma_m = \\ k_{mod} = \\ k_{sys,y} = \\ k_{h,m,y} = \\ k_{h,m,z} = \\ k_l = \end{array}$	44.00 44.00 1.20 0.80 1.00 0.92 1.00 0.90	N/mm ² N/mm ² - - - -				
$\sigma_{t,d} = \sigma_{mu,d} =$	0.00 N/ 4.25 N/	/mm² /mm²		$f_{t,0,d} =$ $f_{mu,d} =$	21.01	N/mm ² N/mm ²				
$\sigma_{m,z,d} =$	0.00 N/	/mm²	<	$f_{m,z,d} =$	0.00	N/mm²		✓		
Utilization ratio							16 %			
<b>a</b> <i>i</i>										
Shear stress an	12 21 KN	J		f	1 30	N/mm²				
v _d –	12.21 KI	N (1-1-1-2		$\gamma_{m} = $ $\gamma_{mod} = $ $k_{h,v} = $	1.20 0.80 0.00	- - -				
T _{v,d} =	0.21 N/	/mm²	<	t _{v,d} =	0.87	N/mm²	24.9/	~		
Ullization ratio							24 /0			
Buckling analy	sis T=0									
$\begin{split} M_{y,d} &= \\ M_{z,d} &= \\ N_{c,d} &= \\ \sigma_{c,d} &= \\ \sigma_{m,y,d} &= \\ \sigma_{m,z,d} &= \end{split}$	37.19 kM 0.00 kM 0.00 kM 0.00 kM 3.73 N/ 0.00 N/	Vm Vm V /mm² /mm²	<	$\begin{array}{l} f_{m,k} = \\ & \\ Y_m = \\ k_{mod} = \\ k_{sys,y} = \\ k_{sys,z} = \\ k_{h,m,y} = \\ k_{h,m,y} = \\ k_{h,m,z} = \\ f_{e,0,d} = \\ f_{m,y,d} = \\ f_{m,z,d} = \end{array}$	44.00 1.20 0.80 1.00 1.00 1.00 23.33 29.33 0.00	N/mm ² - - - - N/mm ² N/mm ² N/mm ²		*		
Utilization ratio							13 %			
Ultimate limit s	tate (ULS)	- design re	sults T=∞							
-20.00 —				Mom	ents [kN	m]			mi ma	n M=0.00 [kNm] ax M=37.19 [kNm]
0.00	÷.		_							
Í	V = 2.28/12.21	[kN]							V = 2.28	/12.21 [kN]
20.00 —		· · · · · · · · · · · · · · · · · · ·					مر ا	a se se se se se se se se se se se se se		
		1111	· · · ·							
			· · · · · · · · · · · · · · · · · · ·	-						

40.00-



	<b>Mu</b> Sen	ni-open 12.18	<b>/ car par</b> 8m - 4 ribs -	<b>k_v2</b> LVL37 - LV	′L43 - 5	51				Page 6/8 02.02.2023	
	Pau	l Brink					Desigr	ner P	'B	Checker	
storaenso											
Shear stress and	alysis 1	Γ=∞									
V _d =	0.20	kN N/mm²	<	$f_{v,k} = \gamma_m = k_{mod} = k_{h,v} = f_{v,d} = $	1.30 1.20 0.80 0.00 0.87	N/mm ² - - - N/mm ²		~			
Utilization ratio							24 %				
Buckling analys	is T=∞										
$\begin{array}{l} M_{y,d} = \\ M_{z,d} = \\ N_{c,d} = \end{array}$	37.19 0.00 0.00	kNm kNm kN		$f_{m,k} =$ $\gamma_m =$ $k_{mod} =$	44.00 1.20 0.80	N/mm² - -					
				k _{sys,y} = k _{sys,z} = k _{h,m,y} = k _{h,m,z} =	1.00 1.00 1.00 1.00	- - -					
$\sigma_{c,d} = \sigma_{m,y,d} = \sigma_{m,z,d} = \sigma_{m,z$	0.00 3.85 0.00	N/mm² N/mm² N/mm²	<	$f_{c,0,d} =$ $f_{m,y,d} =$ $f_{m,z,d} =$	23.33 29.33 0.00	N/mm ² N/mm ² N/mm ²		~			
Utilization ratio	2.00			*****	2.00		13 %				

Service limit state design (SLS) - design results



© 2023 - Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.



	Multi-storey ca Semi-open 12.18m -	<b>ar pa</b> 4 ribs	r <b>k_v2</b> - LVL37	- LVL4	13 - 51		Page 8/8 02.02.2023
storaenso	Paul Brink				[	Designer PB	Checker
Support reaction							
Load case category	/	k _{mod}	A _V [ki	B∨ N]			
NL - Wind		0.9	1.10 0.00	1.10 0.00			
Reference docum	ents for this analysis						
English title			Des	scription	l .		
EN 1995-1-1 EN 1990 ÖNorm B 1995-1-1	NA		EN Cor EN ÖN par Cor	1995-1- mmon ru 1990 - I ORM El ameters	-1 - Eurocode 5: D ules and rules for I Eurocode ? Basis N 1995-1-1 - Aust - Eurocode 5: De ules and rules for	Design of timber str buildings of structural desig ria - National Anne esign of timber stru buildings	ructures - Part 1-1: General - in ex – Nationally determined uctures – Part 1-1: General-
ETA-12/0063 ETA-12/0062			SF SF/	S intec A A intec A Istruction	AG; Self-tapping s AG; ETA-12/0062; ns	crews for use in tir selftapping screw	mber constructions rs for use in timber
ETA-11/0086 ETA-11/0190			GH self	Various taping s	s Angle Brackets screw by Würth		

#### Disclaimer

The software was created to assist engineers in their daily business. The software is an engineering software that is dealing with a very complex matter of structural analysis and building physics analysis. Therefore, this software shall only be operated by skilled, experienced engineers, with a deep understanding of structural engineering and building physics related to timber structures. The user of the software is obliged to check all input values, no matter if they were given by the user or given by default by the software and all results for plausibility. The use of the results of the software should not be relied upon as the basis for any decision or action. Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. The user has the possibility to make print-outs from the software. Any modification of those are not allowed. Stora Enso Wood Products GmbH neither expressly nor implicitly, provides any warranty in terms of accuracy, validity, timeliness and completeness of information and data created by the software with the ones of third party producers or providers. Stora Enso Wood Products GmbH is on to assume any warranty for the general usability of the software is usual pulse for damages caused by gross negligence or intent through Stora Enso Wood Products GmbH is only liable for damages caused by gross negligence or intent through Stora Enso Wood Products GmbH is as well not liable for operational injury. Under the aforementioned conditions Stora Enso Wood Products GmbH is as well not liable for damages caused by gross negligence or intent through Stora Enso Wood Products GmbH is as well not liable for operational injury. Under the aforementioned conditions Stora Enso Wood Products GmbH is as well not liable for operational failures or the loss of programs and/or data of the user's data processing system. Applicable Law: These terms with a software is a software and an law

# Example Calculation LVL Closed Rib Panel

		<b>Mult</b> Close	t <b>i-storey</b> d 12.18m -	<b>car par</b> 4 ribs - LVL	<b>k_v2</b> .37 - LVL	_43 - 51					Pag 02.0	e 1/8 )2.2023	
store		Paul E	Brink				De	signer PE	3		Che	cker	
Store	IEIISO												
System	n												
	,		q⊾=0.56 [kN/m]			LC4: NL - snow							
			q_=0.36 [kN/m]			LC5: NL - Wind							
					*								
	,		q ₁ =2.00 [kN/m]			LC3: NL - live lo	ad cat. F: traffi	c area (vehic	les <= 2	5 kN)			
			q_=0.10 [kN/m]			LC2: NL - deadl	bad						
			0.74 (0.01/cm)										
	,		q _k =0.74 [kiv/m]			LC1:self-weight	structure						
- fi	2		Field 1										
ŀ	•		12.180 [m]		Ĭ								
Clabal													07.0/
Global	utilizati		II S Fire	0.%	515	30.9	SI S Vibr	ation	87.9	% Sur	port		<b>07 %</b>
OLO		30 %	5LOT II C	0 70	010	50 /		auon	07		pon		-1 /0
Sectio	n: LVL ri	ib panel by	/ Stora Enso	o: 1/60									
	500 n	nm 1	a			Layer	Thickness [mm]	V [	Vidth [mm]	Orient	ation I	Material	
			₹mm			1	37.0	5	00.0	0	°	_VL X ilatwise	
i						2	600.0		51.0	0	°	_VL S edaewise	÷
ļ		i	600			3	43.0	5	00.0	0	°	_VL X	-
			mm								·	latineo	
			±43										
			<b>*</b>										
	51 m	m				tau a	680.0						
						CLI	mm						
Materia	al values	5											
Materia	al	f _{m,k}	f _{t,0,k}	<b>f</b> _{t,90,k}	$f_{c,0,k}$	<b>f</b> _{c,90,k}	$\mathbf{f}_{v,k}$	f _{r,k min}	E ₀	,mean	G _{mean}	G _{r,r}	mean
IVIS		[N/mm ² ] 44 00	[N/mm ² ] 35.00	[N/mm ² ]	[N/mm ² ]	[N/mm ² ]	[N/mm ² ] 4 20	[N/mm ² ]	[N/r 13.8	mm²]	[N/mm ² 600.00	] [N/n	nm²]
edgewi	ise	36.00	26.00	6.00	26.00	2.20	1.20	0.00	10,0	00.00	120.00		
flatwise	Э	30.00	20.00	0.00	20.00	2.20	1.50	0.00	10,5	00.00	120.00		
flatwise	e	36.00	26.00	6.00	26.00	2.20	1.30	0.60	10,5	00.00	120.00		
Load													
Load d	Load ca	ups ase categor	v		Type	Duration	Kmod	Vinf	Voun		$\Psi_0$	Ψ1 (	μ2
LC1	self-wei	ght structur	re		G	permanent	0.6	1	Lonh	1.35	1	1	1
LC2 LC3	NL - de NL - live 25 kN)	adioad e load cat. F	■: traffic area	(vehicles <=	G Q	permanent medium term	0.6 0.8	1	)	1.35 1.5	1 0.7	1 0.7	1 0.6

0

0

Page 2/8 Multi-storey car park_v2 Closed 12.18m - 4 ribs - LVL37 - LVL43 - 51 02.02.2023 Paul Brink Checker Designer PB storaenso Load case groups Type Duration Q short Load case category  $\Psi_0$  $\Psi_1$ Kmod  $\Psi_2$  $\gamma_{\text{inf}}$  $\gamma_{\text{sup}}$ LC4 0.9 0 1.5 0.2 NL - snow 0 term Q LC5 NL - Wind 0.9 0 1.5 0 0.2 short term LC1:self-weight structure continuous load Field Load at start [kN/m] 1 0.74 LC2: NL - deadload continuous load Field Load at start [kN/m] 1 0.10 LC3: NL - live load cat. F: traffic area (vehicles <= 25 kN) continuous load Field Load at start [kN/m] 2.00 1 LC4: NL - snow continuous load Field Load at start [kN/m] 1 0.56 LC5: NL - Wind continuous load Load at start Field

[kN/m]

0.36

1

^{© 2023 -} Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.



	Mu	ılti-stor	ey car pa	r <b>k_v2</b>						Page 4/8	
	Clos	sed 12.18	m - 4 ribs - LV	L37 - LVL4	3 - 51					02.02.2023	
	Pau	l Brink					Design	er PB		Checker	
storaenso											
Flexural stress	analysi	s T=0									
M _{y,d} =	38.34	kNm		f _{m,k} =	44.00	N/mm ²					
$M_{z,d} =$	0.00	kNm		$f_{m,k,z} =$	44.00	N/mm²					
Nt,d -	0.00	KIN		γm – kmad =	0.80	-					
				k _{sys.y} =	1.00	-					
				k _{h,m,y} =	0.92	-					
				k _{h,m,z} =	1.00	-					
				k, =	0.90	-					
$\sigma_{t,d} =$	0.00	N/mm ²		f _{t,0,d} =	21.01	N/mm ²					
O _{m,y,d} =	-2.97	N/mm ²	~	fm,y,d =	26.99	N/mm ²		1			
Litilization ratio	0.00			4m,2,0	0.00		11 %				
otilization ratio							11 70				
Shear stress an	alysis 1	Г=0									
V _d =	12.59	kN		f _{v.k} =	1.30	N/mm²					
				Ym =	1.20	-					
				k _{mod} =	0.80	-					
	0.04	N1/ma ma 2		k _{h,v} =	0.00	- N1/ma ma 2		1			
T _{v,d} =	0.31	N/mm-	~	I _{v,d} =	0.87	N/mm-	26.9/	•			
ounzation ratio							30 %				
Buckling analys	sis T=0										
M _{v.d} =	38.34	kNm		f _{m.k} =	44.00	N/mm²					
$M_{z,d} =$	0.00	kNm									
N _{c,d} =	0.00	kN		γ _m =	1.20	-					
				k _{mod} =	0.80	-					
				K _{sys,y} =	1.00	-					
				k _{sys,z} =	1.00	-					
				$k_{h,m,r} =$	1.00	-					
$\sigma_{c,d} =$	0.00	N/mm²		f _{c.0.d} =	23.33	N/mm ²					
σ _{m,y,d} =	2.85	N/mm²		$f_{m,y,d} =$	29.33	N/mm ²					
$\sigma_{m,z,d} =$	0.00	N/mm²	<	f _{m,z,d} =	0.00	N/mm ²		✓			
Utilization ratio							10 %				
Ultimate limit st	tate (UL	S) - desig	n results T=∞								
	•			Mc	ments [kh	m					
0.00	~			MIC	Anema (KN	]			<u> </u>	in M=0.00 [kNm]	
	V = 2 56/1	2 59 [kN]				_			<b>W</b> -25	ax M=38.34 [kNn	1]

20.00-

40.00-



	Mu Clos	Iti-storey sed 12.18m ·	<b>/ car pa</b> - 4 ribs - LV	r <b>k_v2</b> ′L37 - LVL4:	3 - 51					Page 6/8 02.02.202	3
	Pau	l Brink					Desig	ner P	В	Checker	
storaenso										 	
Shear stress and	alysis 1	<b>「=</b> ∞									
V _d =	-12.59	kN N/mm²	<	$f_{v,k} = \gamma_m = k_{mod} = k_{h,v} = f_{v,d} = $	1.30 1.20 0.80 0.00 0.87	N/mm ² - - N/mm ²		~			
Utilization ratio	0.01			10,0	0.01		35 %				
Buckling analys	is T=∞										
	38.34 0.00 0.00	kNm kNm kN		$f_{m,k} =$ $\gamma_m =$ $k_{mod} =$ $k_{sys,y} =$ $k_{sys,z} =$ $k_{b,y,z} =$	44.00 1.20 0.80 1.00 1.00 1.00	N/mm ² - - -					
$\sigma_{c,d} = \\ \sigma_{m,y,d} = \\ \sigma_{m,z,d} =$	0.00 2.96 0.00	N/mm² N/mm² N/mm²	<	$k_{h,m,z} = k_{h,m,z} = f_{c,0,d} = f_{m,y,d} = f_{m,z,d} = f_{m,z,d} = f_{m,z,d}$	1.00 23.33 29.33 0.00	- N/mm² N/mm² N/mm²		✓			
Utilization ratio							10 %				

Service limit state design (SLS) - design results



© 2023 - Calculatis by Stora Enso - Version 5.04.0 Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. For more information see the Terms of Use.



Paul Brink     Designer PB     Checke       Support reaction     Kmod     Av     Bv	r
Support reaction         kmod         Av         Bv           Load case category         kmod         Av         Bv	
Load case category $k_{mod} = A_V = B_V$	
[kN]	
NL - Wind 0.9 1.10 1.10 0.00 0.00	
Reference documents for this analysis	
English title Description	
EN 1995-1-1       EN 1995-1-1 - Eurocode 5: Design of timber structures - Part 1-1: Ge Common rules and rules for buildings         EN 1990       EN 1990 - Eurocode ? Basis of structural design         ÖNorm B 1995-1-1 NA       ÖNORM EN 1995-1-1 - Austria - National Annex – Nationally determ parameters – Eurocode 5: Design of timber structures – Part 1-1: Ge Common rules and rules for buildings	ineral - ined neral-
ETA-12/0063 SFS intec AG; Self-tapping screws for use in timber constructions ETA-12/0062 SFA intec AG; ETA-12/0062; selftapping screws for use in timber constructions	
ETA-11/0086     GH Various Angle Brackets       ETA-11/0190     selftaping screw by Würth	

#### Disclaimer

The software was created to assist engineers in their daily business. The software is an engineering software that is dealing with a very complex matter of structural analysis and building physics analysis. Therefore, this software shall only be operated by skilled, experienced engineers, with a deep understanding of structural engineering and building physics related to timber structures. The user of the software is obliged to check all input values, no matter if they were given by the user or given by default by the software and all results for plausibility. The use of the results of the software should not be relied upon as the basis for any decision or action. Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/building physics engineer. The user has the possibility to make print-outs from the software. Any modification of those are not allowed. Stora Enso Wood Products GmbH nees not assume any warranty in terms of accuracy, validity, timeliness and completeness of information and data created by the software with the ones of third party producers or providers. Stora Enso Wood Products GmbH is on to assume any warranty for the general usability of the software is usuitability for specied purpose or for the compatibility of the software with the ones of third party producers or providers. Stora Enso Wood Products GmbH is only liable for damages caused by gross negligence or intent through Stora Enso Wood Products GmbH is as well not liable for operational injury. Under the aforementioned conditions Stora Enso Wood Products GmbH is as well not liable for operational failures or the loss of programs and/or data of the user's data processing system. Applicable Law: These terms warranty by the adverted and the user's data processing system. Applicable Law: These terms during of the law of use shall be governed by the laws of Austria excluding however any conflict of laws rules and any laws regarding the

F.1.2 Results Minimum Dimensions Long-span Deck Systems

Table 58. Long-span deck systems with required minimum thickness for various span with governing requirement (base case requirements).



												VIDIALIONS							
											'ng	575							
											overn	012 116							
										<u>–</u>	Ğ	510							
										o Pan	(m	5							
										ed Ril	ot(2500r	[m/m]							
										L Clos	g0,k,te	ž							
										Ξ	go,k	[m/n]							
											ţ	<u>*</u> ۳							
											ے م	<u>ٿ</u> -							
		Vibrations	U	U	U	U	U	U	U		h	<u></u>							
	ning	<b>S</b> 1S									<u>م</u>	Vibrations							
	Gover	<b>ULS Fire</b>									vernir	STS							
anel	-	SIU								le	ĝ	ULS Fire							
Rib Pa	00mm)	Ē	6	4	Ч	G	0	~	8	tib Pa	Ē	5111							
osed	,k,tot(25	[kn/r	3.5(	3.8	4.0	4.2(	4.4(	4.68	4.68	pen R	đ(2500m	[m/n]							
CLT CI	ğ	Ē	5	4	0	0	9	7	7	emi-o	go,k,to	ž							
	8	[kN/	1.4	1.5	1.6	1.7	1.7	1.8	1.8	LVLS	80,k	[m/m]							
	$\mathbf{h}_{\mathrm{tot}}$	[mm]	440	540	600	690	740	840	840		ţo	¥] [u							
	$\mathbf{h}_{\mathrm{Rib}}$	[m m]	200	300	360	450	500	600	600		ي و	<u>ا</u> آد							
		Vibrations	U	U	U	U	U	U	U		ŗ	Ē							
	rning	<b>S</b> 1S									å	Vibrations							
	Gove	<b>ULS Fire</b>									verni	315							
Inel		SJU								_	ອິ	013							
Rib Pa	(mm00	Ē	4	0	0	0	0	0	0	Pane	Ĵ.	5111							
Dpen	),k,tot(2	[kN/	2.9	3.3	3.5	3.9	4.3	4.3	4.7	en Rib	ot(2500n	[m/m]							
CLT	50 	[u]	18	32	<del>1</del> 0	26	72	72	88	VI Ob	Bo,k,t	÷							
	ый 1	[kN	i,	Ŀ.	ij.	ij.	÷.	÷	Ļ.		80,k	kN/m							
	h		480	570	620	720	820	820	920		tot	] [wu							
	$h_{Rib}$	um]	360	450	500	600	700	700	800		Rib	ت ۳							
		Vibrations	U							-	-		(7)	(7)	(7)	(7)	(7)	(7)	(7
	erning	S1S	σ								ing	212	Ŭ	Ŭ	Ŭ	Ŭ	Ĩ	Ŭ	Ĩ
	Gov	<b>ULS Fire</b>									overn	ULS Fire							
	-	SIU									G	SIU							
	2500mm	[m]									(mm	-							
anel	g0,k,tot(	[kN								le le	,tot(2500	kN/m	6.13	6.88	6.88	7.65			
CLT P	0'K	/m ² ]								VLPar	g _{0,k}	] [							
Solid	50 L	j [kn								olid	g _{0,k}	kN/m	2.45	2.75	2.75	3.06			
	hcu	mm]								S	ľ	] [uu	08t	540	540	200			
	layers	Ξ									es F	5 [	7	u)	u 1	3			
	CLT	_									Nr. L	Ľ	8	6	6	5 1(			
	ecut	_									ie _{LVL}	_	80 8s	40 9s	40 9s	00 105			
	Nam	E	×								Nam	Ľ	/L G 4;	/L G 5-	/LG 5-	/r g 6(	×	×	Ş
			× ∞	.0	9	,,	9	9	9				8	<u></u>	<u>د</u> و	1	× 9	× 9	x 9
	-	[ <u>u</u> ]	12.1	13.7	14.5	15.6	16.2	17.1	17.6		-	[Ľ	12.1	13.7	14.5	15.6	16.2	17.1	17.6
											l		I						

Table 59. Long-span deck systems with required minimum thickness for various span with governing requirement (reduced fire-safety requirements).

												Vibrations							
											rning	SIS	ს	U	U	U	U	U	U
											Gove	ULS Fire							
										anel		SUU							
										closed Rib P	0,k,tot(2500mm)	[kN/m]	2.28	2.34	2.40	2.47	2.47	2.60	2.60
										LVL	80,k	[kN/m]	0.91	0.93	0.96	0.99	0.99	1.04	1.04
											h	] [we	497	537	587	637	637	737	737
											l _{Rib}	] [uu	360	400	150	200	200	200	200
	50	Vibrations									-	Vibrations 		-	-	-,		-	-
	vernin	STS	U	U	U	U	U	U	ŋ		ning	STS	IJ	U	U	U	U	U	U
	ĝ	ULS Fire									Gover	ULS Fire							
Pane	Ē	SIN								Panel		SIU							
T Closed Rib	<b>B</b> 0,k,tot(2500m	[kN/m]	3.56	3.67	3.84	3.84	4.01	4.01	4.12	ni-open Rib	,k,tot(2500mm)	[kN/m]	2.06	2.18	2.18	2.38	2.38	2.38	2.38
5	g _{0,k}	kN/m]	1.42	1.47	1.54	1.54	1.60	1.60	1.65	VL Ser	aa ≍	E	32	37	37	95	95	95	95
	ltot	] [uu	140	180	540	540	200	005	540		20	[kn	0	ö		ö	ö	ö	0
	Rib	ר <u>י</u> [ע	700	40 4	80	00	60	60	00		h tot		497	587	587	737	737	737	737
+	-	Vibrations -	2	7	m	m	m	m	4		h _{Rib}	<u></u>	360	450	450	600	600	600	600
	ing	SIS	IJ		50	Vibrations													
	ioverr	ULS Fire									erning	STS	Ⴊ	ט	U	U	ט	U	U
le	G	SJU									gov	ULS Fire							
ib Par	(mm0	Ē			_	_	_	_		Panel	Ē	SJU							
CLT Open R	<b>B</b> 0,k,tot(250	l [kN/n	2.70	2.94	3.10	3.30	3.50	3.50	3.90	Open Rib I	30,k,tot(2500mn	[kN/m]	1.91	2.04	2.04				
	B0,k	] [kN/n	1.08	1.18	1.24	1.32	1.40	1.40	1.56	Ľ	Bo,k	kN/m]	0.76	0.82	0.82				
	$\mathbf{h}_{\text{tot}}$		420	480	520	570	620	620	720		to L	] [mr	594	594	594				
	$\mathbf{h}_{\mathrm{Rib}}$	E L	300	360	400	450	500	500	600		Rib	<u>ן</u> ב	8	8	8	×	×	×	×
	<b>F</b>	Vibrations								$\vdash$	-	vibrations =	5 S	9	9	×	×	×	×
	ernin	S1S	U								tive	SIS	IJ	U	IJ	U	U	<del>ن</del>	IJ
	Gov	ULS Fire									lorma	ULS Fire							
	Ê	SUU										SIU							
Panel	<b>B</b> 0,k,tot(2500mr	[kN/m]								nel	,k,tot(2500mm)	[kN/m]	4.60	5.35	6.13	6.88	6.88	7.65	7.65
lid CLT	80,k	kN/m²]								1 LVL Pa	o,k Bo	/m ² ]	84	14	45	75	75	06	06
S	JCLT	[ [ [ [ [ [ []]]								Solic	60	-] [kn	t t	2.	2.	2.	2.	č.	ē.
	ers	5									P [V	E E	360	420	480	540	540	600	600
	CLT lay	Ξ									Nr. LVI panels	Ξ	9	7	∞	6	6	10	10
	Name _{cur}	Ξ	XXX								Name _{LVL}	Ξ	LVL G 360 6s	LVL G 420 7s	LVL G 480 8s	LVL G 540 9s	LVL G 540 9s	LVL G 600 10s	LVL G 600 10s
	_	[m]	12.18	13.70	14.56	15.61	16.26	17.16	17.66		_	Ē	12.18	13.70	14.56	15.61	16.26	17.16	17.66

Table 60. Long-span deck systems with required minimum thickness for various span with governing requirement (no vibrations requirements).

												Vibrations							
											ning	STS	U	U	U	U	თ	<del>ن</del>	U
											ioveri	ULS Fire		_					
										lei	0	SIU	U						
										ed Rib Paı	ot(2500mm)	[m/N]	2.07	2.20	2.28	2.28	2.34	2.40	2.40
										LVL Clos	0,k 80,k,ti	(m) [k	83	88	91	91	93	96	96
											50	<u>لا</u>	0	0	0	0	0	0	0
											ية م	آ آ	33	) 43	0 49	0 49	53	58	58
		Vibrations									h _{Ril}	Ē	200	300	360	360	40	450	450
	ning	STS	თ	U	U	U	U	U	U		60	Vibrations							
	Gover	ULS Fire									vernir	STS	U	U	U	U	U	U	G
ane	-	SJU								lei	ĝ	ULS Fire							
Rib Pa	00mm)	Ē	5				~	~		ib Paı	Ê	SIN							
.T Closed I	<b>g</b> 0,k,tot(25	] [kn/r	3.56	3.56	3.56	3.56	3.67	3.67	3.87	ni-open R	0,k,tot(2500mr	[kN/m]	1.98	2.06	2.06	2.11	2.18	2.18	2.24
σ	go,k	[kN/m	1.42	1.42	1.42	1.42	1.47	1.47	1.54	VL Sei	0'K	[m/	79	82	82	84	87	87	6
	$\mathbf{h}_{\mathrm{tot}}$	[uu	440	440	440	440	480	480	540		50	[kv	0.	0.	0	0	0.	0.	0.
	Rib	_] [w	00	00	00	00	40	40	00		Å,		43.	49	49	23.	28	28	63.
	-		2	~	2	2	2	2	m		h _{Rib}	E E	300	360	360	400	450	450	500
	Governing	STS	<del>ن</del>	U	IJ	U	U	U	U			Vibrations							
e		ULS Fire									erning	STS	U	U	Ⴊ	Ⴊ	Ⴊ	თ	თ
		SJU									Gove	ULS Fire							
ib Par	(mm)	2								anel	e	SJU							
.T Open R	<b>g</b> 0,k,tot(250	] [kN/m	2.46	2.70	2.70	2.94	2.94	3.10	3.10	Dpen Rib F	,k,tot(2500mn	[kN/m]	1.77	1.84	1.91	1.91	2.04	2.04	2.04
J	80,k	kN/m]	0.98	1.08	1.08	1.18	1.18	1.24	1.24	LVL 0	ый ×	[m]	71	74	76	76	82	82	82
	tot	_] [we	360	120	120	180	180	520	520		20	] [kn	0	0	0	0	0. 0	0. 0	0.0
	dib T	<u>ב</u> ב	40 3	00	00	50 4	50 4	8	8		$\mathbf{h}_{\mathrm{tot}}$	<u>س</u>	494	544	594	594	694	694	694
_	ב		2,	ñ	Ж	ñ	ñ	4	4		$\boldsymbol{h}_{\text{Rib}}$	[mm]	400	450	500	500	600	600	600
	erning	SIS	<del>ں</del>								ive	SLS Vibrations	U	U	U	U	U	IJ	U
	Gov	ULS Fire									orma	ULS Fire							
	ē	SJU									z	SIU							
anel	30,k,tot(2500mn	[kN/m]	3.75							e	tot(2500mm)	kN/m]	3.75	4.28	4.60	5.35	5.35	6.13	6.13
CLT P;	0,k £	/m ² ]	.5							/L Pan	<b>B</b> 0,k,	] [							
Solid	500 -	i] [kn	1							olid L	<b>8</b> 0,k	kN/m	1.50	1.71	1.84	2.14	2.14	2.45	2.45
	h _{cu}	шш ]	300							S	h _{LVL}	] [mr	294	336	360	420	420	480	480
	CLT layers	Ξ	8								Nr. LVL panels	Ē	7	7	9	, 7	7	~	~
	Name _{ct}	Ξ	LT 300 L8s-2	xx							Name _{LvL}	Ξ	VL G 294 7s	VL G 336 7s	VL G 360 6s	VL G 420 7s	VL G 420 7s	VL G 480 8s	VL G 480 8s
	-	[Ľ	12.18 C	13.70 ×	14.56	15.61	16.26	17.16	17.66		_	٦ س	12.18 L	13.70 L	14.56 L	15.61 L	16.26 L	17.16 L	17.66 L

Table 61. Long-span deck systems with required minimum thickness for various span with governing requirement (reduced serviceability limit requirements).
												vibrations							
											Bu	ere	(5	(5	(5	(5	(5	(5	(5)
											verni	313	0	0	0	0	0	0	0
										_	ß	111 S Fire							
										Pane	÷	510	U						
										d Rib	2500mr	Ξ	69	83	91	91	96	03	03
										Close	0,k,tot(	N [KN	÷	÷	÷	÷	÷	5	5
										Ľ	¥'	۲ س	88	73	76	76	78	81	81
											20	] [ku	ö	Ö	0	Ö	Ö	ö	ö
											$\mathbf{h}_{\mathrm{tot}}$	ľu ľ	322	422	482	482	522	572	572
											h _{Rib}	[mm	200	300	360	360	400	450	450
	50	Vibrations										Vibrations							_
	ernin	STS	U	Ⴊ	U	U	U	U	U		ing	575	ڻ	U	U	U	ڻ	ڻ	IJ
	Gov	<b>ULS Fire</b>									overr	OLS FILE							
ane		SIU								anel	Ū	510							
RibP	(um0)	Ē	9	9	9	9	~	2	4	Sib Pa	(E	5111							
osed	k,tot(25	[kn/	3.5	3.5	3.5	3.5	3.6	3.6	3.8	pen F	t(2500m	[m]	60	.68	.68	.73	80	80	.87
	<b>.</b>	Ē	5	2	2	2	~	2	4	emi-o	go,k,to	X	-	-	-	-	-	-	-1
	<b>8</b> 0,k	[kN/r	1.4	1.42	1.42	1.42	1.47	1.4.	1.5	LVL S	30,k	[m/]	.64	.67	.67	.69	.72	.72	.75
	h _{tot}	[w w	440	440	440	440	480	480	540			ח] [k	2	2	2	2	0	2	2
	Rib	_ [	8	8	8	8	40	40	8		h Lo		42.	48.	48.	52	57.	57.	62
$\vdash$	ء		2	2	2	2	2	2	e		$\mathbf{h}_{\mathrm{Rib}}$	[m m]	300	360	360	400	450	450	500
	gu	cac anoiterdiV					(5					Vibrations							
	vernii	515	0	0	0	0	0	0	0		ning	SIS	თ	U	U	U	U	U	IJ
	<u>6</u>	ULS Fire									Gover	<b>ULS Fire</b>							
anel	ē	SJU								e l	Ũ	SIU							
Rib	2500mr	[m/	46	70	70	94	94	10	10	b Par	(mm	-							
Oper	30,k, tot	[kn	5.	2.	2.	2.	5	ς. Έ	m.	oen Ri	tot(250	kN/m	1.40	1.46	1.53	1.53	1.66	1.66	1.66
9	ž	<u>ت</u>	98	08	08	18	18	24	24	VL OF	80,k	-							
	00	] [kn	o	Ļ	Ļ	Ļ	Ļ	ų	Ļ	-	80,k	kN/m	0.56	0.59	0.61	0.61	0.67	0.67	0.67
	$\mathbf{h}_{\mathrm{tot}}$	<u>س</u>	360	420	420	480	480	520	520		tot	] [	67.	29	79	79	79	79	79
	h _{Rib}	[	240	300	300	360	360	400	400		ي و	<u>ה</u>	0	0	0 5	0	9 0	9 0	0
		Vibrations									ř	Ē	4	45	50	50	09	99	60
	ning	STS	U								e	Vibrations							
	iover	ULS Fire									mativ	STS	U	U	U	U	U	U	G
	0	SIU									Nor	ULS Fire							
	(mm	-									_	SJU							
	tot(2500	«N/m	3.75								500mm	Ē	5	8	0	5	5	E.	3
Pane	80,k,i	-								ane	),k,tot(2	[k N/	3.7	4.2	4.6	5.3	5.3	6.1	6.1
Ę	go,k	۷/m ² ]	1.5							LVL P.	20	n²]	_	_	.+	-	-	10	
Solic	F	n] [ki								Solid	80,4	[kN/n	1.5(	1.7	$1.8^{\circ}$	2.1	2.1	2.4	2.4
	s h _{CL}	um]	30								h LvL	[u L	294	336	360	420	420	480	480
	layer	Ξ	∞								es L	-	~	~				~	~
	CLT										Nr.	<u>ک</u>			Ψ			80	8
	cur		L8s-2								I'VL		4 7s	6 7s	0 6s	0 7s	0 7s	0 8s	0 8s
	Name	Ξ	T 300								Name	Ξ	. G 29.	. G 33	. G 36	. G 42	. G 42	. G 48	. G 48
Ц	_		5	XXX									۲	Ľ	Ľ	۲	Ľ	Ľ	Ľ
	-	Ē	12.18	13.70	14.56	15.61	16.26	17.16	17.66		-	Ξ	12.18	13.70	14.56	15.61	16.26	17.16	17.66
				-	-	-	-	-					-	-	-	-	-	-	"

Table 62. Long-span deck systems with required minimum thickness for various span with governing requirement (reduced fire-safety and serviceability limit requirements).

# F.2 Short-span Deck Systems

## F.2.1 Example Calculations

Example Calculation CLT Deck

	Multi-storey 2x5.0m span	F	Page 1/8 28.11.2022				
storaspso	Paul Brink			Designer PB	(	Checker	
SIGILIENSO							
System							
g, =0.5	56 [kN/m]	g,=0.56 [kN/m]					
			LC4: NL - snow				
q _x =0.3	36 [kN/m]	q.=0.36 [kN/m]	LC5: NL - Wind				
q _x =2.0	00 [kN/m]	qx=2.00 [kN/m]	LC3: NL - live load cat	F: traffic area (vehicl	es <= 25 kN)		
a -0.1	10. (kN/m)	a =0.10 [kN/m]					
q _k =0.1		40.10 [Kivii]	LC2: NL - deadload				
q _x =0.7	70 [kN/m]	q,=0.70 [kN/m]	LC1:self-weight struct	ure			
Fie	eld 1	Field 2					
A 5.0	B ▶ 4	5.000 (m)					
Global utilization	ratio	77.44 01.0	54.00	0.) <i>(</i> 7)		84 %	
ULS	24 % ULS Fire	77 % SLS	54 % SL	S Vibration	84 % Support	-1 %	
Section: CLT 140	L5s						
		Ťå	Layer 1	Thickness 40.0 mm	Orientation 0°	Material C24 spruce	
4			2	20.0 mm	90°	ETA (2019) C24 spruce	
• ***	1000 mm		3	20.0 mm	0°	ETA (2019) C24 spruce	
			4	20.0 mm	90°	ETA (2019) C24 spruce	
			5	40.0 mm	0°	ETA (2019) C24 spruce ETA (2019)	
			tour	140.0 mm	Ū		
				140.0 1111			
Section Fire: CLT	140 L5s		Laver	Thickness	Orientation	Material	
			1	40.0 mm	0°	C24 spruce	
•	1000 mm	+	2	16.0 mm	90°	C24 spruce	
			t _{CLT}	56.0 mm		E177 (2013)	
Fire resistance clas	ss:R 90 pring : po additional :	fire protection	Time	90 min			
The protection laye	anny . no additionai	life protection	k₀ d₀ [-] [mm]	d _{char,0,h} d _{ef,h} [mm] [mm]	d _{char,0,v} d _{ef,v} [mm] [mm]		
			1 7	77.0 84.0	0.0 0.0		
Material values							
Material	f _{m,k} f _{t,0,k}	ft,90,k fc,0,k	fc,90,k 1	fv,k fr,kmin	Eo,mean Gr	mean Gr,mean	
C24 spruce	1/mm²] [N/mm²] 24.00 14.00	[N/mm²] [N/mm²] 0.12 21.00	2.50 4	mm²] [N/mm²] .00 1.25	[N/mm ² ] [N/m 12,000.00 690	nm²j [N/mm²] 0.00 50.00	
ETA (2019)							
Load							
Load case groups	3						
Load case LC1 self-weight	category structure	Type G	Duration Kmo permanent	od γ _{inf} 0.6 1	γ _{sup} Ψ ₀ 1.35		

1	2
AMA A	M

Multi-storey car park_v2 2x5.0m span

Designer PB

Page 2/8

28.11.2022

Checker

stor									
Load	case groups								
	Load case category	Туре	Duration	Kmod	Yinf	Ysup	$\Psi_0$	$\Psi_1$	$\Psi_2$
LC2	NL - deadload	G	permanent	0.6	1	1.35	1	1	1
LC3	NL - live load cat. F: traffic area (vehicles <= 25 kN)	Q	medium term	0.8	0	1.5	0.7	0.7	0.6
LC4	NL - snow	Q	short term	0.9	0	1.5	0	0.2	0
LC5	NL - Wind	Q	short term	0.9	0	1.5	0	0.2	0

LC1:self-weight structure

I
Load at start
[kN/m]
0.70
0.70

Paul Brink

#### LC2: NL - deadload

continuous loa	d
Field	Load at start [kN/m]
1 2	0.10 0.10

### LC3: NL - live load cat. F: traffic area (vehicles <= 25 kN)

continuous load	t
Field	Load at start
	[kN/m]
1	2.00
2	2.00

### LC4: NL - snow

## continuous load

Load at start
[kN/m]
0.56
0.56

## LC5: NL - Wind

continuous loa	ad
Field	Load at start
	[kN/m]
1	0.36
2	0.36













		Multi-ste		Page 7/8 28.11.2022						
Storger	150	Paul Brink					Design	er PB		Checker
Winst = w[c	har]									
Field	Kdef	Limit [-]	Wlimit [mm]	W _{calc.} [mm]	Ratio					
1 2	1	U1 U1	5000.0	6.5	0%					
w _{fin} = w[cl	har] + w[	q.p.]*kdef								
Field	K _{def}	Limit [-]	W _{limit} [mm]	W _{calc.} [mm]	Ratio					
1 2	1 1	L/250 L/250	20.0 20.0	10.8 10.9	54 % 54 %					
w = w	(a n 1 + v	via n 1*kdef								
Field	Kdef	Limit	Wlimit	Wcalc	Ratio					
		[-]	[mm]	[mm]						
1 2	1 1	L/250 L/250	20.0 20.0	8.8 8.8	44 % 44 %					
Vibration	analysis									
General										
Tributary Stiffness Stiffness Modal da α Man weig Modal ma	width Longitudi Cross dir mping ght ass	inal direction ection				2.4 2536.0 2.0 0.0 700.0 990.2	4 [m] ) [kNm²] ) [kNm²] ) [%] ) [-] ) [N] 2 [kg]			
Analysis										
Criterion Frequence Frequence Accelerate Stiffness	cy criterio cy criterio tion criter criterion	n min n ion	Calc. 11.083 11.083 0.084 [r 0.422 [r	(Hz] 4 [Hz] 6 n/s²] ( nm] (	Class II 4.5 [Hz] 5.0 [Hz] 0.1 [m/s²] 0.5 [mm]	Class II 41 % 54 % 84 %				
Reference	e docum	ents for this	analysis							
English titl	е		-		Description	n				
EN 338 EN 1995-1 ETA-14/03	I-1 349				EN 338 - 5 EN 1995-1 Common r European	Structural timber I-1 - Eurocode 5 rules and rules fo Technical Asses	? Strengt Design c br building sment ET	h classes of timber str s A-14/0349	uctures - F of 02.10.2	Part 1-1: General - 014
Expertise I EN 1995-1	Rolling sh I-2	near - no edge	e gluing, H.J	. Blass	Expertise EN 1995-1 — Structur	on Rolling shear I-2 - Eurocode 5 ral fire design	for CLT — Desigr	n of timber	structures	— Part 1-2: Genera
Technical bearing ca elements	expertise pacity ar	122/2011/02 d separation	: analysis of performance	load e of CLT	Verification structures	n of the load bea with Stora Enso	ring capa CLT	city and the	e insulation	criterion of CLT
Technical gypsum fir EN 1990 ÖNorm B	expertise e boards 1995-1-1	2434/2012 - (GKF) accord	BB: failure t ling to ON E	ime tf of 3 3410	Expertise and gypsu EN 1990 - ÖNORM F	on failure time tf im wall boards ty Eurocode ? Bas N 1995-1-1 - Au	of gypsun pe DF ac is of struc istria - Na	n wall fire b cording to l ctural desig tional Anne	oards acc EN 520 n ex – Nation	ording to ON B3410
ÖNorm B 1995-1-1 NA					parameter Common r ÖNORM E timber stru specificatio	s – Eurocode 5: ules and rules fo EN 1995-1-2 - Au ictures ? Part 1-2 ons concerning (	Design of or building Istria - Na 2: Genera ÖNORM E	f timber stru is tional Anne I ? Structur EN 1995-1-	uctures – P ex - Euroco al fire desi 2, national	art 1-1: General- de 5: Design of gn ? National comments and
Fire safety Europe National sj 1-2, nation	r in timber pecificational comm	r buildings - te ons concernin ents and natio	echnical guil g ÖNORM I onal suppler	deline for EN 1995- nents.	Fire safety SP Techni ÖNORM E 1-2, nation	in timber buildir cal Research Ins N 1995-1-2 - Na al comments an	ngs - techr stitute of S ational spe d national	nical guidel Sweden ecifications I suppleme	ine for Eur concerning nts, chapte	ope; publishes by g ÖNORM EN 1995 er 12
chapter 12 Expertise I	2 Rolling st	near, H.J. Blas	SS	-	Expertise	on rolling shear	strength a	nd rolling s	hear modu	lus of CLT panels

## Example Calculation LVL Deck

State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State State		Mult 2x5.0r	Multi-storey car park_v2 2x5.0m span												
1		Paul B	Brink				De	signer PB		Ch	ecker				
store	IENSO														
Suctor	<b>n</b>														
Syster															
	q	, =0.56 [kN/m]		q.=0.56 [kN/m	1	C4: NL - snow									
	P	x=0.36 [KN/m]	<b>↓</b>	q1=0.36 [KN/m	" <b> </b> 1	LC5: NL - Wind									
	q	x=2.00 [kN/m]		q_=2.00 [kN/m	1	I C3: NL - live load cat F: traffic area (vehicles <= 25 kN)									
	q	x=0.10 [kN/m]		q.=0.10 [kN/m		LC2: NL - dead									
	q	=0.55 [kN/m]		q.=0.55 [kN/m	1	C1:self-weight									
	1	Field 1	<u></u>	Field 2											
77	7.		В	11610 2	TTT .										
I	4	5.000 [m]	► 4	5.000 [m]	P										
Globa	utilizati	on ratio										98 %			
ULS	31 %	ULS Fire	98 %	6 SLS	98 % SI	S Vibration	35 %	5 Support	-1 %	Void		-1 %			
				1				1							
Sectio	n: LVL G	G-X flatwise	100/10.8			1	<b>T</b> L :		Orienteti		Madaziat				
					108 m	Layer 1	36	CKNESS 6.0 mm	90°	on	LVL G-X				
4		1000 n	im	*	<u>+</u> 5	2	36	6.0 mm	90°		flatwise LVL G-X				
						3	36	3.0 mm	90°		flatwise				
						tor 108.0 mm				flatwise					
						ICLT	108	5.0 mm							
Sectio	n Fire: L	VL G-X flat	wise 100/10	).8											
					30 mm	Layer 1	Thi 30	ckness ).0 mm	Orientati 90°	Material LVL G-X					
•		1000 m	m	*		tсіт	30	).0 mm		flatwise					
Fire re-	sistance	oloco: D 00				Time		00 min							
Fire pr	otection I	ayering : no	additional f	ire protectio	n	k ₀ d	d _{char,0,t}	h d _{ef,h}	d _{char,0,v}	d _{ef,v}					
						[-] [r	nm] [mm]	[mm]	[mm]	mm]					
						1 7	71.0	78.0	56.5	0.0					
Materi	al values	5													
Materia	al	f _{m,k}	ft,0,k	ft,90,k	f _{c,0,k}	fc,90,k	f _{v,k}	fr _{,k} min	E0,mean	G _{mean}	1 E	0,5			
LVL G	X	31.00	26.00	0.00	26.00	2.20	1.30	0.60	10,100.00	120.0	0 8,80	00.00			
flatwise	e X	32.00	26.00	6.00	26.00	9.00	4.50	0.00	10,500.00	600.0	0 8,80	00.00			
edgew	ise														
heol															
Luau															
Load	ase aro	ups													
	Load ca	ise category	/		Туре	Duration	Kmod	Yinf	Ysup	Ψ0	Ψ1 0	Ψ ₂			
LC1 LC2	self-wei NL - de	ght structur adload	e		G G	permanent permanent	0.6 0.6	1	1.35 1.35	1 1	1 1	1 1			

		Multi-storey car 2x5.0m span	r park_v						Page 2/7 28.11.2022					
stor		Paul Brink		Design								Cł	necker	
Load	case droup	e												
Loau	L oad case	category	-	Type	Duration	Kmod		Viet		Vaura		W _o	W.	W ₂
LC3	NL - live lo	ad cat. F: traffic area (veh	icles <= (	Q	medium	Rinou	0.8	Y Int	0	Ysup	1.5	0.7	0.7	0.6
LC4	25 kN) NL - snow	ue rependental en la consensa en acentralizada en esperante en successo en la consensa.	(	Q	term short		0.9		0		1.5	0	0.2	0
LC5	NL - Wind		(	Q	short term		0.9		0		1.5	0	0.2	0
LC1:s	self-weight s	tructure												
cont	inuous load													
Field	E	Load at start												
1 1010		[kN/m]												
1		0.55												
2		0.55												
LC2:	NL - deadloa	ad												
cont	inuous load													
Field	Ľ	Load at start												
		[kN/m]												
1		0.10												
-		0.10												
LC3:	NL - live loa	d cat. F: traffic area (veh	icles <= 25	kN)										
cont	inuous load													
Field	I	Load at start												
		[kN/m]												
1		2.00 2.00												
LC4:	NL - snow													
cont	inuous load													
Field	I	Load at start												
1		0.56												
2		0.56												
LC5:	NL - Wind													
cont	inuous load													
Field	Ĕ	Load at start [kN/m]												
1		0.36												
2		0.36												





ULS Fi	re Flexura	I design														
Field	Dist.	f _{m,k}	f _{c,0,k}	f _{t,0,k}	Υm	k _{mod}	k _{fi}	k _{sys,z}	k _{h,m,y}	k _{h,m,z}	k,	f _{m,y,d}	f _{m,z,d}	f _{t,0,0}	t	f _{c,0,d}
	[m]	[N/mm ² ] [	N/mm²]	[N/mm ² ]	[-]	[-]	[-]	[-]	[-]	[-]	[-]	[N/mm²]	[N/mm²] [	N/mr	n²] [N	l/mm²]
1 2	5.0 0.0	31.00 31.00	26.00 26.00	26.00 26.00	1.20 1.20	0.80 0.80	1.00 1.00	1.00 1.00	0.98 0.98	0.87 0.87	0.93 0.93	20.18 20.18	18.46 18.46	16. 16.	13 13	17.33 17.33
Field	M _{y,d}	M _{z,c}	1	N _{c,d}	N _{t,d}	(	Jm,y,d	$\sigma_{m,z,d}$		σ _{c,d}		σ _{t,d}	Ratio			
	[kNm]	[kNn	n]	[kN]	[kN]	[N	/mm²]	[N/mm ²	1] [1	l/mm²]	[N/	/mm²]				
1	-12.12	2 0	00.00	0.00	0.	00	6.23	0.0	00	0.00		0.00	31 9	6 L	CO2	
2	-12.12	2 0	0.00	0.00	0.	00	6.23	0.0	00	0.00		0.00	31 9	6 L	CO2	
ULS Fi	re Shear a	analysis														
Field	Dist.	f _{v,k}	Υm	k _{mod}		k _{h,v}	k _{fi}	1	v,d	Vd		T _{v,d}	Rati	О		
	[m]	[N/mm ² ]	[-]	[-]		[-]	[-]	[N/I	mm²]	[kN]		[N/mm²]				
1	4.97	1.30	1.00	1.0	00	1.00	1.	10	1.43	5.73	3	0.29	9 20	) %	LCO	12
2	0.03	1.30	1.00	1.0	00	1.00	1.	.10	1.43	5.73	3	0.29	9 20	) %	LCO	12

	<b>Mu</b> 2x5	I <b>lti-Sto</b> .0m span	rey car p	oark_v2				Page 5/7 28.11.2022
	Pau	l Brink				Desig	ner PB	Checker
storaenso	-							
Flexural stress	analysi	s Fire						
$\begin{split} M_{y,d} &= \\ M_{z,d} &= \\ N_{t,d} &= \\ \sigma_{t,d} &= \\ \sigma_{m,y,d} &= \end{split}$	-5.78 0.00 0.00 0.00 38.56	kNm kNm kN		$\begin{array}{l} f_{m,k} = \\ f_{m,k,z} = \\ Y_m = \\ k_{mod} = \\ k_{sys,y} = \\ k_{n,my} = \\ k_{n,my} = \\ k_{n,m,z} = \\ k_{l} = \\ k_{l} = \\ k_{fl} = \\ f_{tl,0,d} = \\ f_{m,y,d} = \end{array}$	31.00 N/mm ² 31.00 N/mm ² 1.00 - 1.00 - 1.15 - 0.87 - 0.93 - 1.10 - 26.61 N/mm ² 39.34 N/mm ²			
om,z,d =	0.00	N/mm-		Tm,z,d =	30.46 N/mm*	98 %	~	
Shear stress an	alysis I	Fire						
Vd =	5.73 0.29	kN N/mm²	<	$f_{V,k} =  \gamma_m =  k_{mod} =  k_{h,V} =  k_{f_i} =  f_{V,d} = $	1.30 N/mm ² 1.00 - 1.00 - 1.00 - 1.10 - 1.43 N/mm ²		~	
Utilization ratio	0.20			iv,d —		20 %		

Utilization ratio





Field	Kdef	Limit	Wlimit	Wcalc.	Ratio
		[-]	[mm]	[mm]	
1	0.8	L/1	5000.0	12.8	0 %
2	0.8	L/1	5000.0	12.8	0 %

Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Service Servic		Multi-sto 2x5.0m spa	orey car ^{an}	park	_v2						Page 6 28.11.	6/7 2022
storger	150	Paul Brink					Designe	er PB			Check	er
	horl + w	a n 1*k dof										
Wfin = W[C	narj + w	[q.p.] "kaer			Datia							
Field	Ndef	[-]	Wiimit [mm]	Wcalc. [mm]	Ralio							
1 2	0.8	L/250 L/250	20.0	19.6	98 % 98 %							
w _{net,fin} = w	(q.p.] + v	w[q.p.]*kdef										
Field	K _{def}	Limit [-]	Wlimit [mm]	W _{calc.}	Ratio							
1 2	0.8 0.8	L/250 L/250	20.0 20.0	15.3 15.3	77 % 77 %							
Vibration	analysis	3										
General												
Total ma: Tributary Stiffness Stiffness Modal da α Man weig Modal m	ss width Longitud Cross di imping ght	linal direction rection				8.08 4.2 3560.3 2500.0 2.0 0.0 700.0 1277.0	[t] [m] [kNm²] [kNm²] [%] [-] [N]					
WOdarm	455					1311.9	[Ng]					
Analysis			Calc	(		Class	Class		Class II		CLI	CLI
Frequence Frequence Accelerate	cy criteric cy criteric tion criter	on min on rion	18.995 18.995 0.003 [r	[Hz] 2 [Hz] 8 n/s²] (	4.5 [Hz] 3.0 [Hz] 3.05 [m/s ² ]	4.5 [Hz] 6.0 [Hz] 0.1 [m/s ² ]	010001	24 % 42 % 5 %	oluss II	24 % 32 % 3 %	~	✓ ✓
Sumess	criterion		0.170 [	ining (	).25 [mm]	0.5 [mm]		70 %		33 70		
Reference	e docum	ents for this	analysis									
English titl	е				Descriptio	n						
EN 338 EN 1995-1	1-1				EN 338 - 3 EN 1995-	Structural timber ? 1-1 - Eurocode 5: rulos and rulos for	Strength Design o	classe timber	s structure	es - Par	t 1-1: G	eneral -
EN 1995-1	1-2				EN 1995- — Structu	1-2 - Eurocode 5 - ral fire design	– Design	of timb	er structu	ıres —	Part 1-	2: General
EN 14080					EN 14080 timber - R	- Timber Structur equirements	es - Glue	d lamin	ated timb	er and	glued s	olid
EN 1990 ÖNorm B	1995-1-1	NA			EN 1990 - ÖNORM E parameter	Eurocode ? Basis EN 1995-1-1 - Aus s – Eurocode 5: E	s of struc stria - Nat Design of	tural de ional Ar timber	sign nnex – Na structures	ationally s – Par	y deterr t 1-1: G	nined eneral-
ÖNorm B	1995-1-2	2 NA			ÖNORM E timber stru specificati national s	EN 1995-1-2 - Aus actures ? Part 1-2 ons concerning Ö upplements	stria - Nat General NORM E	, ional Ar ? Struc N 1995	nnex - Eu tural fire -1-2, nati	rocode design onal co	5: Des ? Natic mment	ign of mal s and
ÖNORM E	EN 1995-	1-1_NA, chap	ter 7.3		ÖNORM E parameter Common	EN 1995-1-1 - Aus s – Eurocode 5: E rules and rules for	stria - Nat Design of buildings	ional Ar timber s; chapt	nnex – Na structures er 7.3	ationally s – Par	y deterr t 1-1: G	nined eneral-
CERTIFIC LVL G by column&b	ATE NO Stora En eam_ V0	. EUFI29-200 so_Structural )1	00564-C design man	ual	Product ce Design ma	ertificate anual	5					
ETA 20_0	291 LVL	G by Stora Er	nso		ETA							

#### Disclaimer

The software was created to assist engineers in their daily business. The software is an engineering software that is dealing with a very complex matter of structural analysis and building physics analysis. Therefore, this software shall only be operated by skilled, experienced engineers, with a deep understanding of structural engineering and building physics related to timber structures. The user of the software is obliged to check all input values, no matter if they were given by the user or given by default by the software and all results for plausibility. The use of the results of the software should not be relied upon as the basis for any decision or action. Any use of results of the software is only allowed, if the results have been verified and approved regarding completeness and correctness by a project structural/builing physics engineer. The user has the possibility to make print-tubic from the software. Any modification of those are not allowed. Stora Enso Wood Products GmbH does not assume any warranty regarding the software. The software has been developed with utmost diligence, nevertheless Stora Enso Wood Products GmbH, neither expressly nor implicitly, provides any warranty in terms of accuracy, validity, timeliness and completeness of information and data created by the software.

## F.2.2 Results Minimum Dimensions Short-span Deck Systems

Minimum Thickness LVL Decks for Different Requirements

Table 63. CLT decks with required minimum thickness for various span distances and number of spans with governing requirement (base case requirements).

			BASE CASE								
Name	Nr. sub- spans	L	Name _{cLT}	CLT layers	h _{clt}	<b>g</b> 0,k	(	Gove	rning	S	
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	<b>ULS Fire</b>	SLS	Vibrations	
1x2.5m span	1	2.50	CLT 100 L3s	3	100	0.5		G			
1x5.0m span	1	5.00	CLT 140 L5s	5	140	0.7		G	G	G	
1x7.5m span	1	7.50	CLT 220 L7s - 2	7	220	1.1			G	G	
1x10.0m span	1	10.00	CLT 300 L8s	8	300	1.5			G	G	
2x2.5m span	2	2.50	CLT 100 L3s	3	100	0.5		G			
2x3.75m span	2	3.75	CLT 120 L5s	5	120	0.6		G		G	
2x5.0m span	2	5.00	CLT 140 L5s	5	140	0.7		G		G	
2x6.25m span	2	6.25	CLT 160 L5s	5	160	0.8		G	G	G	
2x7.5m span	2	7.50	CLT 180 L5s	5	180	0.9			G	G	
2x8.75m span	2	8.75	CLT 240 L7s - 2	7	240	1.2				G	
3x2.5m span	3	2.50	CLT 100 L3s	3	100	0.5		G	I		
3x3.75m span	3	3.75	CLT 120 L3s	3	120	0.6				G	
3x5.0m span	3	5.00	CLT 140 L5s	5	140	0.7				G	
3x6.25m span	3	6.25	CLT 160 L5s	5	160	0.8		G	G	G	
4x2.5m span	4	2.50	CLT 100 L3s	3	100	0.5		G	I		
4x3.75m span	4	3.75	CLT 120 L3s	3	120	0.6				G	
4x5.0m span	4	5.00	CLT 140 L5s	5	140	0.7				G	
5x2.5m span	5	2.50	CLT 100 L3s	3	100	0.5		G			
5x3.75m span	5	3.75	CLT 120 L5s	5	120	0.6				G	
6x2.5m span	6	2.50	CLT 100 L3s	3	100	0.5		G			

Fire to 30 min. Nr. CLT Name sub-L Nameclt  $\mathbf{h}_{\text{CLT}}$ Governing **g**0,k layers spans Vibrations **ULS Fire** ULS SLS [-] [-] [-] [-] [mm] [kN/m²] [m] 1x2.5m span 1 2.50 CLT 90 L3s 3 90 0.45 G 1x5.0m span 1 5.00 CLT 140 L5s 5 140 0.70 G G 7 7.50 CLT 220 L7s - 2 G 1x7.5m span 1 220 1.10 G 1x10.0m span 1 10.00 CLT 300 L8s 8 300 1.50 G 2x2.5m span 2 2.50 CLT 90 L3s 3 90 0.45 G 5 G 2x3.75m span 2 3.75 CLT 120 L5s 120 0.60 2x5.0m span 2 5.00 CLT 140 L5s 5 140 0.70 G 2 6.25 5 0.80 G G 2x6.25m span CLT 160 L5s 160 2 7.50 5 0.90 G G 2x7.5m span CLT 180 L5s 180 7 2 G 8.75 240 1.20 2x8.75m span CLT 240 L7s - 2 3 3 G 3x2.5m span 2.50 CLT 90 L3s 90 0.45 3x3.75m span 3 3.75 CLT 120 L3s 3 120 0.60 G 3 5.00 5 0.70 G 3x5.0m span CLT 140 L5s 140 3x6.25m span 3 6.25 CLT 160 L5s 5 160 0.80 G G 4x2.5m span 4 2.50 CLT 90 L3s 3 90 0.45 G 3 G 4 3.75 0.60 4x3.75m span CLT 120 L3s 120 5 G 4x5.0m span 4 5.00 CLT 140 L5s 140 0.70 3 G 5 2.50 90 0.45 5x2.5m span CLT 90 L3s 5 5 G 5x3.75m span 3.75 CLT 120 L5s 120 0.60 6x2.5m span 6 2.50 CLT 90 L3s 3 90 0.45 G

Table 64. CLT decks with required minimum thickness for various span distances and number of spans with governing requirement (reduced fire-safety requirements).

Table 65. CLT decks with required minimum thickness for various span distances and number of spans with governing requirement (no vibrations requirements).

			No vibrations check									
Name	Nr. sub- spans	L	Name _{CLT}	CLT layers	h _{clt}	<b>g</b> 0,k	(	Gove	rning	S		
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	<b>ULS Fire</b>	SLS	Vibrations		
1x2.5m span	1	2.50	CLT 100 L3s	3	100	0.50		G				
1x5.0m span	1	5.00	CLT 140 L5s	5	140	0.70		G	G			
1x7.5m span	1	7.50	CLT 220 L7s - 2	7	220	1.10			G			
1x10.0m span	1	10.00	CLT 300 L8s	8	300	1.50			G			
2x2.5m span	2	2.50	CLT 100 L3s	3	100	0.50		G				
2x3.75m span	2	3.75	CLT 120 L5s	5	120	0.60		G				
2x5.0m span	2	5.00	CLT 140 L5s	5	140	0.70		G				
2x6.25m span	2	6.25	CLT 160 L5s	5	160	0.80		G	G			
2x7.5m span	2	7.50	CLT 180 L5s	5	180	0.90			G			
2x8.75m span	2	8.75	CLT 220 L7s - 2	7	220	1.10			G			
3x2.5m span	3	2.50	CLT 100 L3s	3	100	0.50		G				
3x3.75m span	3	3.75	CLT 110 L3s	3	110	0.55		G				
3x5.0m span	3	5.00	CLT 120 L3s	3	120	0.60		G	G			
3x6.25m span	3	6.25	CLT 160 L5s	5	160	0.80		G	G			
4x2.5m span	4	2.50	CLT 100 L3s	3	100	0.50		G				
4x3.75m span	4	3.75	CLT 110 L3s	3	110	0.55		G				
4x5.0m span	4	5.00	CLT 120 L3s	3	120	0.60		G	G			
5x2.5m span	5	2.50	CLT 100 L3s	3	100	0.50		G				
5x3.75m span	5	3.75	CLT 110 L3s	3	110	0.55		G				
6x2.5m span	6	2.50	CLT 100 L3s	3	100	0.50		G				

			SLS to 1/150 & no vibrations check												
Name	Nr. sub- spans	L	Name _{cLT}	CLT Name _{cLT} dayers h _{cLT} g _{0,k} Governing ျားကို											
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	<b>ULS Fire</b>	SLS	Vibrations					
1x2.5m span	1	2.50	CLT 100 L3s	3	100	0.50		G							
1x5.0m span	1	5.00	CLT 140 L5s	5	140	0.70		G	G						
1x7.5m span	1	7.50	CLT 180 L7s	7	180	0.90			G						
1x10.0m span	1	10.00	CLT 240 L7s - 2	7	240	1.20			G						
2x2.5m span	2	2.50	CLT 100 L3s	3	100	0.50		G							
2x3.75m span	2	3.75	CLT 120 L5s	5	120	0.60		G							
2x5.0m span	2	5.00	CLT 140 L5s	5	140	0.70		G							
2x6.25m span	2	6.25	CLT 160 L5s	5	160	0.80		G	G						
2x7.5m span	2	7.50	CLT 160 L5s - 2	5	160	0.80		G	G						
2x8.75m span	2	8.75	CLT 180 L7s	7	180	0.90		G	G						
3x2.5m span	3	2.50	CLT 100 L3s	3	100	0.50		G							
3x3.75m span	3	3.75	CLT 110 L3s	3	110	0.55		G							
3x5.0m span	3	5.00	CLT 120 L3s	3	120	0.60		G							
3x6.25m span	3	6.25	CLT 160 L5s	5	160	0.80		G	G						
4x2.5m span	4	2.50	CLT 100 L3s	3	100	0.50		G							
4x3.75m span	4	3.75	CLT 110 L3s	3	110	0.55		G							
4x5.0m span	4	5.00	CLT 120 L3s	3	120	0.60		G							
5x2.5m span	5	2.50	CLT 100 L3s	3	100	0.50		G							
5x3.75m span	5	3.75	CLT 110 L3s	3	110	0.55		G							
6x2.5m span	6	2.50	CLT 100 L3s	3	100	0.50		G							

Table 66. CLT decks with required minimum thickness for various span distances and number of spans with governing requirement (reduced serviceability limit requirements).

			Fire to 30 min., SLS to 1/150 & no vibrations check									
Name	Nr. sub- spans	L	Name _{cLT}	CLT layers	h _{clt}	<b>g</b> 0,k		Gove	rning	3		
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	<b>ULS Fire</b>	SLS	Vibrations		
1x2.5m span	1	2.50	CLT 60 L3s	3	60	0.30						
1x5.0m span	1	5.00	CLT 110 L3s	3	110	0.55		G	G			
1x7.5m span	1	7.50	CLT 180 L7s	7	180	0.90			G			
1x10.0m span	1	10.00	CLT 240 L7s - 2	7	240	1.20			G			
2x2.5m span	2	2.50	CLT 60 L3s	3	60	0.30						
2x3.75m span	2	3.75	CLT 90 L3s	3	90	0.45		G				
2x5.0m span	2	5.00	CLT 110 L3s	3	110	0.55		G				
2x6.25m span	2	6.25	CLT 120 L5s	5	120	0.60			G			
2x7.5m span	2	7.50	CLT 160 L5s - 2	5	160	0.80		G	G			
2x8.75m span	2	8.75	CLT 180 L7s	7	180	0.90		G	G			
3x2.5m span	3	2.50	CLT 60 L3s	3	60	0.30						
3x3.75m span	3	3.75	CLT 90 L3s	3	90	0.45		G				
3x5.0m span	3	5.00	CLT 110 L3s	3	110	0.55		G				
3x6.25m span	3	6.25	CLT 120 L3s	3	120	0.60			G			
4x2.5m span	4	2.50	CLT 60 L3s	3	60	0.30						
4x3.75m span	4	3.75	CLT 90 L3s	3	90	0.45		G				
4x5.0m span	4	5.00	CLT 110 L3s	3	110	0.55		G				
5x2.5m span	5	2.50	CLT 60 L3s	3	60	0.30						
5x3.75m span	5	3.75	CLT 90 L3s	3	90	0.45		G				
6x2.5m span	6	2.50	CLT 60 L3s	3	60	0.30						

Table 67. CLT decks with required minimum thickness for various span distances and number of spans with governing requirement (reduced fire-safety and serviceability limit requirements).

## Minimum Thickness LVL Decks for Different Requirements

Table 68. LVL decks with required minimum thickness for various span distances and number of spans with governing requirement (base case requirements).

					BASE C	ASE				
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	<b>g</b> 0,k	ſ	Norm	nativo	9
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	<b>ULS Fire</b>	SLS	Vibrations
1x2.5m span	1	2.50	LVL G 96 2s	2	96	0.49		G		
1x5.0m span	1	5.00	LVL G 144 3 s	3	144	0.73			G	
1x7.5m span	1	7.50	LVL G 210 5s	5	210	1.07			G	
1x10.0m span	1	10.00	LVL G 294 7s	7	294	1.50				G
2x2.5m span	2	2.50	LVL G 96 2s	2	96	0.49		G		
2x3.75m span	2	3.75	LVL G 108 3s	3	108	0.55		G		
2x5.0m span	2	5.00	LVL G 108 3s	3	108	0.55		G	G	
2x6.25m span	2	6.25	LVL G 144 3s	3	144	0.73			G	
2x7.5m span	2	7.50	LVL G 168 4s	4	168	0.86			G	
2x8.75m span	2	8.75	LVL G 210 5s	5	210	1.07			G	
3x2.5m span	3	2.50	LVL G 96 2s	2	96	0.49		G		
3x3.75m span	3	3.75	LVL G 108 3s	3	108	0.55		G		
3x5.0m span	3	5.00	LVL G 120 2s	2	120	0.61			G	
3x6.25m span	3	6.25	LVL G 144 3s	3	144	0.73			G	
4x2.5m span	4	2.50	LVL G 96 2s	2	96	0.49		G		
4x3.75m span	4	3.75	LVL G 108 3s	3	108	0.55		G		
4x5.0m span	4	5.00	LVL G 120 2s	2	120	0.61			G	
5x2.5m span	5	2.50	LVL G 96 2s	2	96	0.49		G		
5x3.75m span	5	3.75	LVL G 108 3s	3	108	0.55		G		
6x2.5m span	6	2.50	LVL G 96 2s	2	96	0.49		G		

			Fire to 30 min.									
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	<b>g</b> 0,k	ſ	Norm	native	9		
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	<b>ULS Fire</b>	SLS	Vibrations		
1x2.5m span	1	2.50	LVL G 72 1s	1	72	0.37						
1x5.0m span	1	5.00	LVL G 144 3 s	3	144	0.73			G			
1x7.5m span	1	7.50	LVL G 210 5s	5	210	1.07			G			
1x10.0m span	1	10.00	LVL G 294 7s	7	294	1.50				G		
2x2.5m span	2	2.50	LVL G 72 1s	1	72	0.37						
2x3.75m span	2	3.75	LVL G 84 2s	2	84	0.43			G			
2x5.0m span	2	5.00	LVL G 108 3s	3	108	0.55		G	G			
2x6.25m span	2	6.25	LVL G 144 3s	3	144	0.73			G			
2x7.5m span	2	7.50	LVL G 168 4s	4	168	0.86			G			
2x8.75m span	2	8.75	LVL G 210 5s	5	210	1.07			G			
3x2.5m span	3	2.50	LVL G 72 1s	1	72	0.37						
3x3.75m span	3	3.75	LVL G 84 2s	2	84	0.43			G			
3x5.0m span	3	5.00	LVL G 120 2s	2	120	0.61			G			
3x6.25m span	3	6.25	LVL G 144 3s	3	144	0.73			G			
4x2.5m span	4	2.50	LVL G 72 1s	1	72	0.37						
4x3.75m span	4	3.75	LVL G 84 2s	2	84	0.43			G			
4x5.0m span	4	5.00	LVL G 120 2s	2	120	0.61			G			
5x2.5m span	5	2.50	LVL G 72 1s	1	72	0.37						
5x3.75m span	5	3.75	LVL G 84 2s	2	84	0.43			G			
6x2.5m span	6	2.50	LVL G 72 1s	1	72	0.37						

Table 69. LVL decks with required minimum thickness for various span distances and number of spans with governing requirement (reduced fire-safety requirements).

No vibrations check Nr. Nr. LVL Name sub-L NameLVL  $\mathbf{h}_{\mathsf{LVL}}$ Normative **g**0,k panels spans Vibrations **ULS Fire** ULS SLS [mm] [kN/m²] [-] [-] [-] [-] [m] 1x2.5m span 1 2.50 LVL G 96 2s 2 96 G 0.49 1x5.0m span 1 5.00 LVL G 144 3 s 3 0.73 144 G 7.50 5 G 1x7.5m span 1 LVL G 210 5s 210 1.07 G 1x10.0m span 1 10.00 LVL G 288 6s 6 288 1.47 2x2.5m span 2 2.50 LVL G 96 2s 2 96 0.49 G 2x3.75m span 2 3.75 LVL G 108 3s 3 108 0.55 G 2x5.0m span 2 5.00 LVL G 108 3s 3 108 0.55 G G 2 6.25 LVL G 144 3s 3 0.73 G 2x6.25m span 144 2 4 G 2x7.5m span 7.50 LVL G 168 4s 168 0.86 2 5 G 8.75 LVL G 210 5s 2x8.75m span 210 1.07 3 2 3x2.5m span 2.50 LVL G 96 2s 96 0.49 G 3x3.75m span 3 3.75 LVL G 108 3s 3 108 0.55 G 3 5.00 2 3x5.0m span LVL G 120 2s 120 0.61 G 3x6.25m span 3 6.25 LVL G 144 3s 3 144 0.73 G 4x2.5m span 4 2.50 LVL G 96 2s 2 96 0.49 G 4 3.75 3 G 4x3.75m span LVL G 108 3s 108 0.55 2 G 4x5.0m span 4 5.00 LVL G 120 2s 120 0.61 5 2 5x2.5m span 2.50 LVL G 96 2s 96 0.49 G 5 3 G 5x3.75m span 3.75 LVL G 108 3s 108 0.55 6x2.5m span 6 2.50 LVL G 96 2s 2 96 0.49 G

Table 70. LVL decks with required minimum thickness for various span distances and number of spans with governing requirement (no vibrations requirements).

			SLS to 1/150 & no vibrations check									
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	<b>g</b> 0,k	1	Norm	native	9		
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	<b>ULS Fire</b>	SLS	Vibrations		
1x2.5m span	1	2.50	LVL G 96 2s	2	96	0.49		G				
1x5.0m span	1	5.00	LVL G 108 3s	3	108	0.55		G	G			
1x7.5m span	1	7.50	LVL G 168 4s	4	168	0.86			G			
1x10.0m span	1	10.00	LVL G 240 4s	4	240	1.22			G			
2x2.5m span	2	2.50	LVL G 96 2s	2	96	0.49		G				
2x3.75m span	2	3.75	LVL G 108 3s	3	108	0.55		G				
2x5.0m span	2	5.00	LVL G 108 3s	3	108	0.55		G				
2x6.25m span	2	6.25	LVL G 120 2s	2	120	0.61		G	G			
2x7.5m span	2	7.50	LVL G 144 3s	3	144	0.73			G			
2x8.75m span	2	8.75	LVL G 168 4s	4	168	0.86			G			
3x2.5m span	3	2.50	LVL G 96 2s	2	96	0.49		G				
3x3.75m span	3	3.75	LVL G 108 3s	3	108	0.55		G				
3x5.0m span	3	5.00	LVL G 108 3s	3	108	0.55		G				
3x6.25m span	3	6.25	LVL G 120 2s	2	120	0.61		G	G			
4x2.5m span	4	2.50	LVL G 96 2s	2	96	0.49		G				
4x3.75m span	4	3.75	LVL G 108 3s	3	108	0.55		G				
4x5.0m span	4	5.00	LVL G 108 3s	3	108	0.55		G				
5x2.5m span	5	2.50	LVL G 96 2s	2	96	0.46		G				
5x3.75m span	5	3.75	LVL G 108 3s	3	108	0.55		G				
6x2.5m span	6	2.50	LVL G 96 2s	2	96	0.49		G				

Table 71. LVL decks with required minimum thickness for various span distances and number of spans with governing requirement (reduced serviceability limit requirements).

			Fire to 30 min., SLS to 1/150 & no vibrations check										
Name	Nr. sub- spans	L	Name _{LVL}	Nr. LVL panels	h _{LVL}	<b>g</b> 0,k	ſ	Norm	ative	9			
[-]	[-]	[m]	[-]	[-]	[mm]	[kN/m²]	NLS	<b>ULS Fire</b>	SLS	Vibrations			
1x2.5m span	1	2.50	LVL G 72 1s	1	72	0.37							
1x5.0m span	1	5.00	LVL G 108 3s	3	108	0.55			G				
1x7.5m span	1	7.50	LVL G 168 4s	4	168	0.86			G				
1x10.0m span	1	10.00	LVL G 240 4s	4	240	1.22			G				
2x2.5m span	2	2.50	LVL G 72 1s	1	72	0.37							
2x3.75m span	2	3.75	LVL G 72 1s	1	72	0.37							
2x5.0m span	2	5.00	LVL G 90 3s	3	90	0.46			G				
2x6.25m span	2	6.25	LVL G 120 2s	2	120	0.61			G				
2x7.5m span	2	7.50	LVL G 144 3s	3	144	0.73			G				
2x8.75m span	2	8.75	LVL G 168 4s	4	168	0.86			G				
3x2.5m span	3	2.50	LVL G 72 1s	1	72	0.37							
3x3.75m span	3	3.75	LVL G 72 1s	1	72	0.37							
3x5.0m span	3	5.00	LVL G 96 2s	2	96	0.49			G				
3x6.25m span	3	6.25	LVL G 120 2s	2	120	0.61			G				
4x2.5m span	4	2.50	LVL G 72 1s	1	72	0.37							
4x3.75m span	4	3.75	LVL G 72 1s	1	72	0.37							
4x5.0m span	4	5.00	LVL G 96 2s	2	96	0.49			G				
5x2.5m span	5	2.50	LVL G 72 1s	1	72	0.37							
5x3.75m span	5	3.75	LVL G 72 1s	1	72	0.37							
6x2.5m span	6	2.50	LVL G 72 1s	1	72	0.37							

Table 72. LVL decks with required minimum thickness for various span distances and number of spans with governing requirement (reduced fire-safety and serviceability limit requirements).

				B	ASE CA	SE CLT			1		BA	SE CAS	E LVL						COMPA	RISON	
Name	Nr. sub- spans		Name _{cLT}	CLT layers	h _{сцт}	g _{o,k}	60	verni	ß	Name _{LVL}	Nr. LVL panels	h _{LVL}	go,k	z	orma	tive	Ĭ	eight diffe	rence	Weight dif	ference
Ξ	E	[m]	E	E	[mm]	[kN/m ² ]	S10	212 סבט דוונפ	Vibrations	E	[-]	[mm]	[kN/m ² ]	SIU	ULS Fire	SIS		[mm]	[%]	kN/m ² ]	[%]
1x2.5m span	1	2.50	CLT 100 L3s	ŝ	100	0.5		(5		LVL G 96 2s	2	96	0.49		თ			4	4%	0.01	2%
1x5.0m span	1	5.00	CLT 140 L5s	S	140	0.7		(J	U	LVL G 144 3 s	£	144	0.73			IJ		-4	-3%	-0.03	-4%
1x7.5m span	1	7.50	CLT 220 L7s - 2	7	220	1.1		G	U	LVL G 210 5s	5	210	1.07			ڻ	_	10	5%	0.03	3%
1x10.0m span	1	10.00	CLT 300 L8s	∞	300	1.5		U	U	LVL G 294 7s	7	294	1.50			Ŭ	(7)	9	2%	0.00	%0
2x2.5m span	2	2.50	CLT 100 L3s	m	100	0.5		(5		LVL G 96 2s	2	96	0.49		Ⴊ			4	4%	0.01	2%
2x3.75m span	2	3.75	CLT 120 L5s	S	120	0.6		(5	U	LVL G 108 3s	ŝ	108	0.55		U			12	10%	0.05	8%
2x5.0m span	2	5.00	CLT 140 L5s	5	140	0.7		(5	U	LVL G 108 3s	ŝ	108	0.55		თ	IJ		32	23%	0.15	21%
2x6.25m span	2	6.25	CLT 160 L5s	ß	160	0.8	~	U (T	U	LVL G 144 3s	£	144	0.73			U		16	10%	0.07	%6
2x7.5m span	2	7.50	CLT 180 L5s	ß	180	0.9		G	U	LVL G 168 4s	4	168	0.86			IJ		12	7%	0.04	4%
2x8.75m span	2	8.75	CLT 240 L7s - 2	7	240	1.2			U	LVL G 210 5s	5	210	1.07			IJ		30	13%	0.13	11%
										-							-				
				Fire	e to 30 i	min. CLT					Fire t	o 30 m	in. LVL						Compa	rison	
Name	Nr. sub- spans		<b>Name</b> _{cLT}	CLT layers	h _{ctT}	go,k	6	verni	ß	Name _{LvL}	Nr. LVL panels	h _{LvL}	go,k	z	orma	tive	Ĭ	eight diffe	rence	Weight dif	ference
Ξ	Ξ	[ <u>æ</u> ]	Ξ	Ξ	[mm]	[kN/m ² ]	S10		Vibrations	Ξ	Ξ	[mm]	[kN/m ² ]	SJU	ULS Fire	SIS		[mm]	[%]	kN/m²]	[%]
1x2.5m span	1	2.50	CLT 90 L3s	æ	60	0.45			U	LVL G 72 1s	1	72	0.37					18	20%	0.08	18%
1x5.0m span	1	5.00	CLT 140 L5s	ß	140	0.70		G	U	LVL G 144 3 s	ŝ	144	0.73			IJ		-4	-3%	-0.03	-4%
1x7.5m span	1	7.50	CLT 220 L7s - 2	7	220	1.10		G		LVL G 210 5s	5	210	1.07			IJ		10	5%	0.03	3%
1x10.0m span	1	10.00	CLT 300 L8s	8	300	1.50		G	U	LVL G 294 7s	7	294	1.50			Ŭ	(7)	9	2%	0.00	%0
2x2.5m span	2	2.50	CLT 90 L3s	m	6	0.45			U	LVL G 72 1s	1	72	0.37					18	20%	0.08	18%
2x3.75m span	2	3.75	CLT 120 L5s	5	120	0.60			U	LVL G 84 2s	2	84	0.43			U		36	30%	0.17	28%
2x5.0m span	2	5.00	CLT 140 L5s	ß	140	0.70			U	LVL G 108 3s	£	108	0.55		Ⴊ	IJ		32	23%	0.15	21%
2x6.25m span	2	6.25	CLT 160 L5s	5	160	0.80		G	U	LVL G 144 3s	ŝ	144	0.73			IJ		16	10%	0.07	%6
2x7.5m span	2	7.50	CLT 180 L5s	5	180	06.0		G	U	LVL G 168 4s	4	168	0.86			IJ		12	7%	0.04	4%
2x8.75m span	2	8.75	CLT 240 L7s - 2	7	240	1.20			G	LVL G 210 5s	5	210	1.07			IJ		30	13%	0.13	11%

Results Comparison Short-span Deck Systems Table 73. Comparison short-span CLT and LVL deck system; base case and fire to 30 min.

	e,		1										ΙΓ		e.		1									
	fference	[%]	2%	-4%	3%	2%	2%	8%	21%	%6	4%	3%			fference	[%]	2%	21%	4%	-2%	2%	8%	21%	24%	%6	
rison	Weight di	[kN/m ² ]	0.01	-0.03	0.03	0.03	0.01	0.05	0.15	0.07	0.04	0.03		rison	Weight di	[kN/m ² ]	0.01	0.15	0.04	-0.02	0.01	0.05	0.15	0.19	0.07	
Compa	ference	[%]	4%	-3%	5%	4%	4%	10%	23%	10%	7%	5%		Compa	ference	[%]	4%	23%	7%	%0	4%	10%	23%	25%	10%	, , ,
	Jeight diff	[mm]	4	-4	10	12	4	12	32	16	12	10			Jeight diff	[ <b>mm</b> ]	4	32	12	0	4	12	32	40	16	,
	-	vibrations											-		-	Vibrations										-
	tive	575		U	IJ	U			IJ	IJ	IJ	IJ			tive	STS		U	U	U				U	U	,
	orma	ULS Fire	Ⴊ				IJ	IJ	ڻ					۲۲	orma	ULS Fire	U	U			U	IJ	IJ	U		
	Ż	SIU												eck L	Ż	SIU									I	
(LVL	¥	m²]	6	č	2	5	ō.	S	5	ξ	9	2		ıs ch	¥	m ² ]	6	ŝ	9	2	6	Ŋ	S	1	g	,
check	8	[kN/	0.4	0.7	1.0	1.4	0.4	0.5	0.5	0.7	0.8	1.0		ratior	<b>B</b> 0,	[kN/	0.4	0.5	0.8	1.2	0.4	0.5	0.5	0.6	0.7	i
rations o	h _{Lvt}	[ <b>m</b> m]	96	144	210	288	96	108	108	144	168	210		د no vibı	h _{LVL}	[mm]	96	108	168	240	96	108	108	120	144	
No vib	Nr. LVL panels	Ξ	2	e	S	9	2	£	m	m	4	S		1/150 8	Nr. LVL panels	Ξ	2	m	4	4	2	£	£	7	m	,
	Name _{LVL}	Ξ	VL G 96 2s	VL G 144 3 s	VL G 210 5s	VL G 288 6s	VL G 96 2s	VL G 108 3s	VL G 108 3s	VL G 144 3s	VL G 168 4s	VL G 210 5s		SLS to	Name _{LvL}	Ð	VL G 96 2s	VL G 108 3s	VL G 168 4s	VL G 240 4s	VL G 96 2s	VL G 108 3s	VL G 108 3s	VL G 120 2s	VL G 144 3s	
		vibrations	-				_					_	-			Vibrations	-				_	_				
	ning	SIS		U	Ⴊ	U				თ	Ⴊ	U			ning	STS		თ	U	U				თ	U	)
	iover	ULS Fire	Ⴊ	Ⴊ			თ	Ⴊ	Ⴊ	Ⴊ				СĽТ	iover	ULS Fire	G	Ⴊ			Ⴊ	U	Ⴊ	Ⴊ	U	)
_	0	SIU												heck	0	SIU										
check CL ⁻	<b>g</b> o,k	[kN/m ² ]	0.50	0.70	1.10	1.50	0.50	0.60	0.70	0.80	06.0	1.10		ations ch	go,k	[kN/m ² ]	0.50	0.70	06.0	1.20	0.50	0.60	0.70	0.80	0.80	
No vibrations	h _{ctT}	[ [ [ [ [	100	140	220	300	100	120	140	160	180	220		k no vibı	h _{ciT}	[ [mm]	100	140	180	240	100	120	140	160	160	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
	CLT layers	Ξ	е	ß	7	8	ε	S	Ŋ	S	Ŋ	7		1/150 8	CLT layers	Ξ	ŝ	Ŋ	2	2	e	ß	S	Ŋ	ы	,
	Name _{сгт}	Ξ	CLT 100 L3s	CLT 140 L5s	CLT 220 L7s - 2	CLT 300 L8s	CLT 100 L3s	CLT 120 L5s	CLT 140 L5s	CLT 160 L5s	CLT 180 L5s	CLT 220 L7s - 2		SLS to	Name _{ct}	Ξ	CLT 100 L3s	CLT 140 L5s	CLT 180 L7s	CLT 240 L7s - 2	CLT 100 L3s	CLT 120 L5s	CLT 140 L5s	CLT 160 L5s	CLT 160 L5s - 2	
	_	<u>[</u>	2.50	5.00	7.50	10.00	2.50	3.75	5.00	6.25	7.50	8.75			L	[u]	2.50	5.00	7.50	10.00	2.50	3.75	5.00	6.25	7.50	, , ,
	Nr. sub- spans	Ξ	1	H	H	-	2	2	2	2	2	2			Nr. sub- spans	Ξ	1	Ч	Ч	с,	2	2	2	2	2	1
	Name	Ξ	1x2.5m span	1x5.0m span	1x7.5m span	1x10.0m span	2x2.5m span	2x3.75m span	2x5.0m span	2x6.25m span	2x7.5m span	2x8.75m span			Name	Ξ	1x2.5m span	1x5.0m span	1x7.5m span	1x10.0m span	2x2.5m span	2x3.75m span	2x5.0m span	2x6.25m span	2x7.5m span	

Table 74. Comparison short-span CLT and LVL deck system; no vibrations and SLS to 1/150 & no vibrations

			Fire to 30 min.,	SLS to	1/150 &	k no vibrati	ons ch	neck CL	F.	Fire to 30 min.	, SLS to 1	l/150 &	no vibrati	ons ch	ieck L	۲L		Compai	rison	
Name	Nr. sub- spans		Name _{cu} la	CLT ayers	<b>h</b> _{сгт}	<b>g</b> o,k	Gov	rerning		Name _{LVL}	Nr. LVL panels	h _{LVL}	go,k	Nor	mativ	a	Height diff	erence	Weight diff	erence
Ξ	Ξ	<u>[</u>	ī	Ξ	[ [ [ [ [ [ ]	[kN/m²]	ULS Fire	STS	Vibrations	Ξ	Ξ	[mm]	kN/m²] [		STS	Vibrations	[mm]	] [%]	[kN/m ² ]	[%]
1x2.5m span	ч	2.50	CLT 60 L3s	m	60	0.30			_	LVL G 72 1s	1	22	0.37				-12	-20%	-0.07	-23%
1x5.0m span	Ч	5.00	CLT 110 L3s	ŝ	110	0.55	U	U	_	LVL G 108 3s	ŝ	108	0.55		U		2	2%	0.00	%0
1x7.5m span	1	7.50	CLT 180 L7s	7	180	0.90		U		LVL G 168 4s	4	168	0.86		U		12	7%	0.04	4%
1x10.0m span	1	10.00	CLT 240 L7s - 2	7	240	1.20		თ		LVL G 240 4s	4	240	1.22		U		0	%0	-0.02	-2%
2x2.5m span	2	2.50	CLT 60 L3s	ŝ	60	0.30			-	LVL G 72 1s	1	72	0.37				-12	-20%	-0.07	-23%
2x3.75m span	2	3.75	CLT 90 L3s	ŝ	6	0.45	G		-	LVL G 72 1s	1	72	0.37				18	20%	0.08	18%
2x5.0m span	2	5.00	CLT 110 L3s	m	110	0.55	G		-	LVL G 90 3s	m	06	0.46		ש		20	18%	0.09	16%
2x6.25m span	2	6.25	CLT 120 L5s	ß	120	0.60		U	-	LVL G 120 2s	2	120	0.61		U		0	%0	-0.01	-2%
2x7.5m span	2	7.50	CLT 160 L5s - 2	ß	160	0.80	U	G	-	LVL G 144 3s	m	144	0.73		ש		16	10%	0.07	%6
2x8.75m span	2	8.75	CLT 180 L7s	7	180	0.90	G	U	-	LVL G 168 4s	4	168	0.86		U		12	7%	0.04	4%

Table 75. Comparison short-span CLT and LVL deck system; fire to 30 min., no vibrations and SLS to 1/150 & no vibrations

	BASE CASE				Fire to 30 min.				No vibrations of	check		
Span	Deck type	h _{tot}	go,k g	0,k,tot(2500mm)	Deck type	h _{tot}	go,k	<b>G</b> 0,k,tot(2500mm)	Deck type	h _{tot}	So,k Sc	,k,tot(2500mm)
Ē	[-]	[mm]	[kN/m]	[kN/m]		[mm]	[kN/m]	[kN/m]	[-]	[mm]	[kN/m]	[kN/m]
12.18	Open CLT deck: 120mm CLT - 4 ribs 360x200mm	480	1.18	2.95	Open CLT deck: 120mm CLT - 4 ribs 360x200mm	480	1.18	2.95	VL semi-open rib panel: h _{rib} = 360mm	497	0.82	2.06
13.70	Open CLT deck: 120mm CLT - 4 ribs 450x200mm	570	1.32	3.30	Open CLT deck: 120mm CLT - 4 ribs 450x200mm	570	1.32	3.30	-VL closed rib panel: h _{rib} = 400mm	537	0.93	2.34
14.56	Open CLT deck: 120mm CLT - 4 ribs 500x200mm	620	1.40	3.50	Open CLT deck: 120mm CLT - 4 ribs 500x200mm	620	1.40	3.50	VL semi-open rib panel: h _{rib} = 450mm	587	0.87	2.18
15.61	Open CLT deck: 120mm CLT - 4 ribs 600x200mm	720	1.56	3.90	Open CLT deck: 120mm CLT - 4 ribs 600x200mm	720	1.56	3.90	:VL closed rib panel: h _{rib} = 500mm	637	0.99	2.47
16.26	Closed CLT deck: 120mm CLT - 4 ribs 500x140mm - 120mm CLT	740	1.76	4.40	Closed CLT deck: 120mm CLT - 4 ribs 500x140mm - 120mm CLT	740	1.76	4.40	:VL closed rib panel: h _{rib} = 500mm	637	0.99	2.47
17.16	Open CLT deck: 120mm CLT - 4 ribs 700x200mm	820	1.72	4.30	Open CLT deck: 120mm CLT - 4 ribs 700x200mm	820	1.72	4.30	:VL semi-open rib panel: h _{rib} = 600mm	737	0.95	2.38
17.66	Closed CLT deck: 120mm CLT - 4 ribs 600x140mm - 120mm CLT	840	1.87	4.68	Closed CLT deck: 120mm CLT - 4 ribs 600x140mm - 120mm CLT	840	1.87	4.68	'VL semi-open rib panel: h _{rib} = 600mm	737	0.95	2.38
						1000						
	SLS to 1/150 & no vibrations che	÷			Fire to 30 min., SLS to 1/150 & no vibratic	ons cneck						
Span	Deck type	h _{tot}	go,k B	0,k,tot(2500mm)	Deck type	$\mathbf{h}_{\text{tot}}$	go,k	g0,k,tot(2500mm)				
<u></u>	[-]	[mm]	[kN/m]	[kN/m]	[-]	[mm]	[kN/m]	[kN/m]				
12.18	LVL closed rib panel: $h_{rib}$ = 200mm	337	0.83	2.07	LVL closed rib panel: h _{rib} = 200mm	337	0.83	2.07				
13.70	LVL closed rib panel: $h_{rib} = 300 mm$	437	0.88	2.20	LVL closed rib panel: $h_{rib} = 300 \text{mm}$	437	0.88	2.20				
14.56	LVL semi-open rib panel: $h_{rib} = 360mm$	497	0.82	2.06	LVL semi-open rib panel: h _{rib} = 360mm	497	0.82	2.06				
15.61	LVL closed rib panel: $h_{rib} = 360 mm$	497	0.91	2.28	¹ LVL closed rib panel: h _{rib} = 360mm	497	0.91	2.28				
16.26	LVL closed rib panel: $h_{rib}$ = 400mm	537	0.93	2.34	¹ LVL closed rib panel: h _{rib} = 400mm	537	0.93	2.34				
17.16	LVL semi-open rib panel: h _{rib} = 450mm	587	0.87	2.18	¹ LVL semi-open rib panel: h _{rib} = 450mm	587	0.87	2.18				
17.66	LVL closed rib panel: $h_{rib} = 450 mm$	587	0.96	2.40	LVL closed rib panel: h _{rib} = 450mm	587	0.96	2.40				

F.3 Choice of Deck System Table 76. Choice of long-span deck system for various scenario's.

# **G**. Additional Calculations & Results Framing System

## All Subvariants Total Deck Height, Weight and Column Load

Table 77. Total deck height, weight per level and loads on foundation for all subvariants.

					Load	standar	d column [kN]	on foun	dation
ID	Car park (sub)variant	Span [m]	Floor depth [mm]	Weight /level [kg]	1 level	2 levels	3 levels	4 levels	5 levels
1	Variant 1, 12.18m, column every 1 parking bay(s)	12.18	480	16,098	140.8	281.6	422.3	563.1	703.9
2	Variant 1, 13.7m, column every 1 parking bay(s)	13.70	570	20,267	164.4	328.8	493.2	657.5	821.9
3	Variant 1, 14.56m, column every 1 parking bay(s)	14.56	620	22,734	178.2	356.5	534.7	713.0	891.2
4	Variant 1, 15.61m, column every 1 parking bay(s)	15.61	720	26,979	198.8	397.5	596.3	795.0	993.8
5	Variant 1, 16.26m, column every 1 parking bay(s)	16.26	740	31,507	217.0	434.0	651.0	868.1	1,085.1
6	Variant 1, 17.16m, column every 1 parking bay(s)	17.16	820	32,595	227.0	454.0	681.0	908.1	1,135.1
7	Variant 1, 17.66m, column every 1 parking bay(s)	17.66	840	36,213	241.6	483.1	724.7	966.2	1,207.8
8	Variant 1, 12.18m, column every 2 parking bay(s)	12.18	480	16,238	282.0	564.0	846.0	1,127.9	1,409.9
9	Variant 1, 13.7m, column every 2 parking bay(s)	13.70	570	20,226	328.6	657.3	985.9	1,314.6	1,643.2
10	Variant 1, 14.56m, column every 2 parking bay(s)	14.56	620	22,747	356.5	713.1	1,069.6	1,426.1	1,782.7
11	Variant 1, 15.61m, column every 2 parking bay(s)	15.61	720	26,880	397.2	794.4	1,191.6	1,588.8	1,986.1
12	Variant 1, 16.26m, column every 2 parking bay(s)	16.26	740	31,439	433.8	867.7	1,301.5	1,735.3	2,169.1
13	Variant 1, 17.16m, column every 2 parking bay(s)	17.16	820	32,577	454.0	907.9	1,361.9	1,815.9	2,269.9
14	Variant 1, 17.66m, column every 2 parking bay(s)	17.66	840	36,211	483.1	966.2	1,449.3	1,932.4	2,415.5
15	Variant 2a, 12.18m, struts On, column every 1 parking bay(s)	12.18	516	9,529	122.1	244.2	366.4	488.5	610.6
16	Variant 2a, 13.7m, struts On, column every 1 parking bay(s)	13.70	586	11,466	139.1	278.3	417.4	556.5	695.7
17	Variant 2a, 14.56m, struts On, column every 1 parking bay(s)	14.56	636	12,679	149.2	298.3	447.5	596.7	745.8
18	Variant 2a, 15.61m, struts On, column every 1 parking bay(s)	15.61	676	13,890	160.8	321.5	482.3	643.0	803.8
19	Variant 2a, 16.26m, struts On, column every 1 parking bay(s)	16.26	666	15,242	169.6	339.2	508.8	678.4	848.0
20	Variant 2a, 17.16m, struts On, column every 1 parking bay(s)	17.16	716	16,479	180.2	360.4	540.6	720.8	901.0

21	Variant 2a, 17.66m, struts On, column every 1 parking bay(s)	17.66	746	17,305	186.4	372.7	559.1	745.4	931.8
22	Variant 2a, 12.18m, struts Off, column every 1 parking bay(s)	12.18	726	11,352	127.1	254.2	381.3	508.4	635.5
23	Variant 2a, 13.7m, struts Off, column every 1 parking bay(s)	13.70	806	13,368	144.7	289.4	434.1	578.8	723.5
24	Variant 2a, 14.56m, struts Off, column every 1 parking bay(s)	14.56	846	14,630	154.9	309.7	464.6	619.5	774.4
25	Variant 2a, 15.61m, struts Off, column every 1 parking bay(s)	15.61	906	16,221	167.6	335.1	502.7	670.3	837.8
26	Variant 2a, 16.26m, struts Off, column every 1 parking bay(s)	16.26	946	17,212	175.5	351.0	526.6	702.1	877.6
27	Variant 2a, 17.16m, struts Off, column every 1 parking bay(s)	17.16	996	18,733	186.8	373.5	560.3	747.0	933.8
28	Variant 2a, 17.66m, struts Off, column every 1 parking bay(s)	17.66	1016	19,626	193.0	386.0	579.0	771.9	964.9
29	Variant 2a, 12.18m, struts On, column every 2 parking bay(s)	12.18	668	9,769	245.5	490.9	736.4	981.9	1,227.3
30	Variant 2a, 13.7m, struts On, column every 2 parking bay(s)	13.70	748	11,292	277.9	555.9	833.8	1,111.7	1,389.6
31	Variant 2a, 14.56m, struts On, column every 2 parking bay(s)	14.56	798	12,291	296.9	593.8	890.7	1,187.6	1,484.5
32	Variant 2a, 15.61m, struts On, column every 2 parking bay(s)	15.61	838	13,739	321.8	643.5	965.3	1,287.1	1,608.8
33	Variant 2a, 16.26m, struts On, column every 2 parking bay(s)	16.26	868	14,456	336.0	672.0	1,008.0	1,344.0	1,680.1
34	Variant 2a, 17.16m, struts On, column every 2 parking bay(s)	17.16	928	15,539	356.3	712.7	1,069.0	1,425.4	1,781.7
35	Variant 2a, 17.66m, struts On, column every 2 parking bay(s)	17.66	968	16,180	368.0	736.0	1,103.9	1,471.9	1,839.9
36	Variant 2a, 12.18m, struts Off, column every 2 parking bay(s)	12.18	858	10,671	250.6	501.1	751.7	1,002.2	1,252.8
37	Variant 2a, 13.7m, struts Off, column every 2 parking bay(s)	13.70	958	12,250	283.8	567.5	851.3	1,135.1	1,418.9
38	Variant 2a, 14.56m, struts Off, column every 2 parking bay(s)	14.56	1008	13,312	303.1	606.2	909.3	1,212.4	1,515.5
39	Variant 2a, 15.61m, struts Off, column every 2 parking bay(s)	15.61	1078	14,494	326.6	653.2	979.8	1,306.4	1,633.0
40	Variant 2a, 16.26m, struts Off, column every 2 parking bay(s)	16.26	1128	15,280	341.5	682.9	1,024.4	1,365.8	1,707.3
41	Variant 2a, 17.16m, struts Off, column every 2 parking bay(s)	17.16	1178	16,392	361.9	723.8	1,085.8	1,447.7	1,809.6
42	Variant 2a, 17.66m, struts Off, column every 2 parking bay(s)	17.66	1218	17,118	373.8	747.7	1,121.5	1,495.3	1,869.2
43	Variant 2b, 12.18m, column every 2 parking bay(s)	12.18	756	11,001	251.9	503.9	755.8	1,007.7	1,259.7

44	Variant 2b, 13.7m, column every 2 parking bay(s)	13.70	836	12,831	286.2	572.4	858.6	1,144.8	1,431.0
45	Variant 2b, 14.56m, column every 2 parking bay(s)	14.56	886	14,016	306.3	612.5	918.8	1,225.1	1,531.3
46	Variant 2b, 15.61m, column every 2 parking bay(s)	15.61	946	15,476	330.9	661.9	992.8	1,323.7	1,654.7
47	Variant 2b, 16.26m, column every 2 parking bay(s)	16.26	986	16,404	346.4	692.9	1,039.3	1,385.7	1,732.2
48	Variant 2b, 17.16m, column every 2 parking bay(s)	17.16	1036	17,745	368.0	736.1	1,104.1	1,472.2	1,840.2
49	Variant 2b, 17.66m, column every 2 parking bay(s)	17.66	1066	18,437	380.0	760.0	1,140.0	1,520.0	1,899.9
50	Variant 3, 12.18m, struts On, column every 2 parking bay(s)	12.18	656	9,319	245.1	490.2	735.3	980.4	1,225.5
51	Variant 3, 13.7m, struts On, column every 2 parking bay(s)	13.70	756	10,986	279.4	558.8	838.2	1,117.6	1,397.0
52	Variant 3, 14.56m, struts On, column every 2 parking bay(s)	14.56	816	11,881	298.0	596.0	893.9	1,191.9	1,489.9
53	Variant 3, 15.61m, struts On, column every 2 parking bay(s)	15.61	886	13,043	321.0	642.0	963.0	1,284.0	1,605.0
54	Variant 3, 16.26m, struts On, column every 2 parking bay(s)	16.26	936	13,808	335.5	671.0	1,006.5	1,342.0	1,677.5
55	Variant 3, 17.16m, struts On, column every 2 parking bay(s)	17.16	996	14,879	356.3	712.6	1,068.9	1,425.2	1,781.5
56	Variant 3, 17.66m, struts On, column every 2 parking bay(s)	17.66	1026	15,380	367.1	734.2	1,101.3	1,468.5	1,835.6
57	Variant 3, 12.18m, struts On, column every 2 parking bay(s)	12.18	678	10,031	249.0	498.1	747.1	996.1	1,245.2
58	Variant 3, 13.7m, struts On, column every 2 parking bay(s)	13.70	768	11,643	282.3	564.7	847.0	1,129.3	1,411.6
59	Variant 3, 14.56m, struts On, column every 2 parking bay(s)	14.56	828	12,590	301.2	602.4	903.7	1,204.9	1,506.1
60	Variant 3, 15.61m, struts On, column every 2 parking bay(s)	15.61	898	13,916	325.8	651.7	977.5	1,303.3	1,629.1
61	Variant 3, 16.26m, struts On, column every 2 parking bay(s)	16.26	948	14,630	339.6	679.3	1,018.9	1,358.6	1,698.2
62	Variant 3, 17.16m, struts On, column every 2 parking bay(s)	17.16	1008	15,757	360.6	721.3	1,081.9	1,442.5	1,803.2
63	Variant 3, 17.66m, struts On, column every 2 parking bay(s)	17.66	1038	16,288	371.6	743.3	1,114.9	1,486.5	1,858.2
64	Variant 3, 12.18m, struts Off, column every 2 parking bay(s)	12.18	856	10,181	250.2	500.4	750.5	1,000.7	1,250.9
65	Variant 3, 13.7m, struts Off, column every 2 parking bay(s)	13.70	956	11,983	285.3	570.5	855.8	1,141.0	1,426.3
66	Variant 3, 14.56m, struts Off, column every 2 parking bay(s)	14.56	1006	12,911	304.0	608.1	912.1	1,216.2	1,520.2
67	Variant 3, 15.61m, struts Off, column every 2 parking bay(s)	15.61	1076	14,199	327.8	655.6	983.4	1,311.2	1,639.0

68	Variant 3, 16.26m, struts Off,	16.26	1176	15 022	2127	685.3	1 028 0	1 270 7	1 712 /
00	column every 2 parking bay(s)	10.20	1120	13,022	542.7	005.5	1,028.0	1,570.7	1,713.4
69	column every 2 parking bay(s)	17.16	1176	16,152	363.8	727.6	1,091.5	1,455.3	1,819.1
70	Variant 3, 17.66m, struts Off, column every 2 parking bay(s)	17.66	1216	16,767	375.3	750.6	1,126.0	1,501.3	1,876.6
71	Variant 3, 12.18m, struts Off, column every 2 parking bay(s)	12.18	868	10,859	253.9	507.8	761.7	1,015.6	1,269.5
72	Variant 3, 13.7m, struts Off, column every 2 parking bay(s)	13.70	958	12,598	287.9	575.9	863.8	1,151.8	1,439.7
73	Variant 3, 14.56m, struts Off, column every 2 parking bay(s)	14.56	1018	13,619	307.3	614.6	921.9	1,229.1	1,536.4
74	Variant 3, 15.61m, struts Off, column every 2 parking bay(s)	15.61	1088	15,072	332.6	665.3	997.9	1,330.6	1,663.2
75	Variant 3, 16.26m, struts Off, column every 2 parking bay(s)	16.26	1128	15,818	346.7	693.3	1,040.0	1,386.6	1,733.3
76	Variant 3, 17.16m, struts Off, column every 2 parking bay(s)	17.16	1188	17,029	368.2	736.3	1,104.5	1,472.6	1,840.8
77	Variant 3, 17.66m, struts Off, column every 2 parking bay(s)	17.66	1218	17,621	379.5	759.0	1,138.5	1,518.1	1,897.6
xx	Modupark, 16.50m (16.26m), column every 2 parking bays (corrected for comparable dimensions)	16.26	550	61,261	618.1	1,236.1	1,854.2	2,472.3	3,090.3
xx	Koopman int. (steel + fibreglass), 15.65m (16.26m), column every 2 parking bays (corrected for comparable dimensions)	16.26	640	10,477	314.6	629.2	943.7	1,258.3	1,572.9
xx	Morspoort (steel + composite), 14.50 (16.26m), column every 2 parking bays (corrected for comparable dimensions)	16.26	651	35,110	461.1	922.2	1,383.4	1,844.5	2,305.6
xx	BauBuche concept (timber + concrete), 16.50m (16.26m), column every 1 parking bay (corrected for comparable dimensions)	16.26	730	59,341	298.3	596.6	894.9	1,193.2	1,491.5
xx	B&O-Holzparkhaus (timber), 16.24m (16.26m), column every 1 parking bay (corrected for comparable dimensions)	16.26	935	17,547	179.2	358.3	537.5	716.7	895.8
# Parametric Model

This appendix describes the script of the parametric model using Rhinoceros and Grasshopper to create a .xml file for structural analysis in SCIA using the plugin Koala. First, a general overview of the parametric script is given after which the various components are discussed.

## H.1 General Overview

An overview of the workflow within the script can be found in Figure 165 and the total script is presented in Figure 166. The script can be separated into two parts. The first part is concerned with creating the geometry of the structural elements to later be used for the second part, preparing a structural analysis model of the subvariant for SCIA, using the plugin Koala.

The script starts with the inputs. These are either parameterised inputs, like the parking deck span and column distance, but also include the specific parking deck system chosen in chapter 9. A second set of inputs are the so-called standardised inputs, which include car park characteristics like the parking bay width and clearance height, inputs for the wind load calculation and general load input values. These input values are further processed to determine often used input settings for the modelling of the geometry like the total story height.

Next, the geometry for the 2D structural model is created. These lines and nodes are later used to create the model for the structural analysis. For each main design variant, the geometry is created by a separate set of objects.

The preparation for the structural analysis model for SCIA starts with setting up the general settings of the analysis. Next, the different items required for the analysis model are created like the layers for the structural elements, load cases & groups and load combinations. For the cross-sections, self-made libraries for the standard available cross-section dimensions are inserted as well as material types and classes for later selection. The created lines and nodes from the geometry section are then used to create beam elements and support points. Cross-sections are linked to the elements and loads are applied. Finally, hinges are added where needed.

With all the inputs for the generation of the xml file by Koala ready, they are inserted into the xml generation component, which combines all elements into an xml file, which can be opened and analysed in SCIA. However, some settings required for the structural analysis of especially timber structures are not yet available in Koala components. Therefore, the created xml file is further processed by text editing in the Grasshopper model, to add certain settings to the file. This finally results in the xml file, which is used for structural analysis in SCIA.



Figure 165. Overview of workflow in parametric script.



Figure 166. Grasshopper script parametric model.

To further structure the Grasshopper script, various colours are used to indicate the type of logic.



# H.2 Inputs

## User Inputs

## Parameterized Inputs

The parameterized input section reads various input parameters from an Excel file. These include for example the parking deck span and weight of the deck. These are then extracted and stored in separate containers for further use, see Figure 167



Figure 167. Parameterized inputs.

#### Standardized inputs

The standardized input section includes three types of inputs, which do not change for the various considered design (sub)variants, see Figure 168. The first type are general settings for the geometry of the structure: parking bay width, clearance height, height railing top deck and inward distance struts. The second type of inputs are related to the inputs required for the wind calculations. These include for example the wind peak force to be considered and wind zone coefficients. The third set of inputs is related to the various loads considered in the structural validation of the structure: dead load, imposed load, snow load, wind load and self-weight of the façade.



Figure 168. Standardized inputs.

#### Settings Grasshopper/Rhino

The last section under inputs include various general settings for Grasshopper and Rhinoceros, like the colour of the grasshopper scripts and drawing distance between different sections of the structure, see Figure 169.

Settings Grasshopper/Rhino							
Colours Grasshopper script	Settings Grasshopper/Rhino geometry default						
Toggle True T Swatch B Swatch W	Nr. of spans for calculation						
С н	Edge or middle frame Two-sided V Edge or middle frame						

Figure 169. Settings Grasshopper/Rhino.

## Processing for Often Used Inputs

#### Create Filename

Using the various inputs by the user, a filename is created, by combining various inputs and a unique ID into the filename, see Figure 170.



Figure 170. Create filename.

#### Specification of Dimensions Car Park

For the creation of the geometry of the car park structure, various parameters are required, which depend on multiple input parameters. Some of these parameters, which are often used in the script are therefore specified at this point, see Figure 171.



Figure 171. Specification of dimensions car park.

#### Loads

In the standardized input section, the various loads were already inserted in  $kN/m^2$ . These are now converted to kN/m, see Figure 172.



Figure 172. Loads.

#### Wind Loads

In this section, the various point loads on the structure as a result of wind loading are calculated, see Figure 173. The calculation consist out of various steps. First, the façades are distributed into various sections, which transfer their wind loads to a supporting elements. The heights of these different sections are calculated. Next, the various types of wind loads acting on the structure are calculated. These are the friction load on the roof in x and y direction and the pressure and suction loads on the façades in x and y direction. All of these loads act on a certain point on the structure, where they are supported. Finally, all the wind point loads acting on the structure.



Figure 173. Wind loads.

# H.3 Geometry

The second part of the script consists out of the creation of the geometry of the structural system of the design variant. In this paragraph, the creation of the geometry of design variant 1 is described. This results in a set of lines and points as presented in Figure 174.



Figure 174. Geometry of design variant 1.

## Columns

The main columns following the repetitive pattern in the transverse direction are created in this section, see Figure 175 (left vertical elements in Figure 174).



Figure 175. Columns.

## Transverse Main Girders & Top Level Railing

In this section, the transverse main girders and top level railing are created, see Figure 176. These are the horizontal elements in Figure 174. The transverse main girders can also be the floor panels in the final design, which can later be selected when choosing the cross-section applied to this element.



Figure 176. Transverse main girders & top level railing.

## Wind Bracing X-direction

For the wind bracings, only slanted elements in one direction are created, see Figure 177. These elements function as tension only elements and in the model, only horizontal loads coming from the left side are considered. In reality, cross bracings will be applied, but for modelling purposes, elements in only one direction are sufficient.



Figure 177. Wind bracing x-direction.

**Ground Support Points** 

Points for the ground supports are created based on the geometry of the columns, see Figure 178.



Figure 178. Ground support points.

## Columns Sides

The columns at the sides are created to provide a stabilizing frame to the main structure. These columns are found in the right part of Figure 174 and are created by the logic of Figure 179.



Figure 179. Columns sides.

Side Girders & Top Level Railing

The stabilizing frame also contains side girders and a top level railing, which are created in the logic of Figure 180.



Figure 180. Side girders & top level railing.

## Wind Bracing Y-direction

The last elements of the stabilizing side frame are the wind bracings in y-direction, see Figure 181. Again, only slanted elements in one direction are crated, while in reality elements in both directions are used.



Figure 181. Wind bracing y-direction.

# H.4 Setup of XML File for Structural Analysis in SCIA

In this part of the script, using the plugin Koala, a .xml file is to be later used as input for the structural analysis of the system in SCIA. This file contains amongst others geometry including materials and cross-sections, loads and settings for the structural analysis.

## Project Setup

The project setup section contains general settings for start up of a SCIA project, see Figure 182.



Figure 182. Project setup.

## Create .xml Output File

This section includes the logic required for the creation of the .xml output file. This includes a button for creation of the file as well as names and locations of the .xml and xml.def files, see Figure 183.



Figure 183. Create .xml output file (name).

#### Layers

This section creates the different layers of the model in SCIA. This includes a layer for the geometry (frame) and one for the loads, see Figure 184.



Figure 184. Layers.

## **Cross-sections**

The creation of the cross-sections and applying them to various elements consists out of various sections presented below.

#### Cross-section Library

In Excel, a library is created, which contains the cross-sections dimensions for various types of timber and steel elements. For timber, different sections are found for glulam, glued solid timber and solid timber beams. For steel, different sections are included like RMS, CMS, L, RD and FL sections. In this section, the library from Excel is imported and split out for the various types of sections, see Figure 185.



Figure 185. Cross-section library.

## Material Types & Classes Library

The different material types and classes for the timber and steel are manually created to form a library to later choose from, see Figure 186.



Figure 186. Material types & classes library.

#### **Element Names**

The names of all the different elements are manually added to the script and sorted into a tree data structure, see Figure 187.



Figure 187. Element names.

#### Cross-section Selection

For each group of elements, a cross-section is applied for analysis in SCIA. For these elements, the material, material type, material class, height of the cross-section and finally to be applied cross-section has to be selected from a list of inputs as presented in Figure 188. The lists from which can be chosen are updated based on the choice in a previous aspect. Updating these lists is done in the logic from Figure 189, which includes all the libraries presented above and imports the choices from the drop-down menus. This also creates the name of the cross-section in the format required by the Koala plugin.



Figure 188. Input for selection of cross-section for main girders.



Figure 189. Logic for selection correct libraries to be applied and creation of section name.

#### **Output Sections**

Finally, all the different sections are combined and inserted into the Koala sections component (see Figure 190), which then contains a list of all sections to be applied to the various elements.



Figure 190. Output with all cross-sections.

#### Beams

To create the beam elements for the .xml file, first the active geometry for the considered design (sub)variant is fetched (see green block Figure 191). For these elements, the settings for the beam creation can than be selected in the pink block.

	( Main (/ Secondary Girders)	Structural Type Beam	FEM Type Standard	Member System Line Centre
	Transverse Main Girders (/ Joists)	Structural Type Beam	FEM Type Standard 💎	Member System Line Centre 👿
Element / Greek section Names	All Columns	Structural Type Column	FEM Type Standard	Member System Line Centre 👿
(0)	Struts	Structural Type Secondary Column	FEM Type Standard	Member System Line Centre
0 Main (/ Secondary Girder) 1 Transverse Main Girder (/ Joist)	Primary Joists x-direction	Structural Type Beam	FEM Type Standard	Member System Line Centre
2 Column 3 Strut	Secondary Joists y-direction	Structural Type Beam	FEM Type Standard	Member System Line Centre
4 Primary Joist x-direction 5 Secondary Joist y-direction	Primary Girders / Joists y-direction	Structural Type Beam	FEM Type Standard	Member System Line Centre 💎
6 Frimary Girder / Joist y-direction 7 Secondary Joist x-direction	Secondary Joists x-direction	Structural Type Beam	FEM Type Standard	Member System Line Centre
0 Side Girder 9 Wind Bracing - x	Gerein Side Girders	Structural Type Beam	FEM Type Standard	Member System Line Centre
10 Railing top level - x 11 Railing top level - y	Wind Bracing - x	Structural Type Wall Bracing	FEM Type Axial force only	Member System Line Centre
12 Wind Bracing - y	Q Railing top level - x	Structural Type Beam	FEM Type Standard	Member System Line Centre
	Railing top level - y	Structural Type Beam	FEM Type Standard	Member System Line Centre
	Wind Bracing - y	Structural Type Wall Bracing	FEM Type Axial force only	Member System Line Centre 👿

Figure 191. Input geometry and select settings for beam creation.

A datatree of all beam elements is created as well as a tree for each type of setting, containing the setting for each beam element, see Figure 192.



Figure 192. Merging all settings for beam elements.



Now the beam elements can be created (see Figure 193), which requires the input of the various datatrees created before.

Figure 193. Creation of beam elements.

Finally, the list of beams is sorted according to the data structure that was applied to the creation of geometry, see Figure 194. This later makes it easier to find the correct beam name according to a line selected from the geometry.



Figure 194. Sorting beams to data structure of geometry.

#### Supports

Supports are created for the structure. This is done by selection of the support points in the geometry and finding the nodes from the beam and column elements that apply to these nodes. Furthermore, the degrees of freedom for the support points are selected, see Figure 195. Finally, since 2D is considered, Rx and Rz are removed as these otherwise create problems with importing the .xml file in SCIA.



Figure 195. Supports.

## Hinges

For the various beam elements, hinges are created. From the list of beams, the beams to be considered are selected based on the name of the elements. Next, the settings for the hinges on one or both sides of the beam can be selected, after which the hinges are created, see Figure 196. All hinges for all elements are finally added into a single data tree.



Figure 196. Hinges.

## Load Cases & Groups

The different load cases and groups are created in the logic of Figure 197.



Figure 197. Load cases & groups.

## Load Combinations

Different types of linear load combinations are created, see Figure 198. These include load combinations for the Ultimate Limit State and various Serviceability Limit States (characteristic, frequent and quasi).



Figure 198. Linear load combinations.

#### Loads

The various loads are created by the Koala plugin for the .xml file. For the creation of the self-weight of the floors, dead load and imposed load, the logic of Figure 199 is used. This requires input of the value of the load, correct load case and elements the load should be applied to.



Figure 199. Creation of loads for self-weight floors (same logic for dead load and imposed load).

The snow load and wind line load (for floors) only act on the top level floor. This requires a slight alteration of the logic, see Figure 200.



Figure 200. Creation of snow load on top floor level (same logic for wind on top floor level).

The creation of the loads for the self-weight of the façade requires using the input of the different height sections of the façade. These loads were calculated earlier and are applied to the correct load case, see Figure 201.



Figure 201. Creation of load self-weight façade.

For the creation of point loads for wind loading, first the correct nodes to which the loads should be applied have to be found. This is done by finding the selection of these point from the geometry and finding the corresponding nodes from the beam (column) elements generated by the Koala element, see Figure 202. Next, for both the x- and y-direction, the point wind loads that were calculated earlier are applied and the wind point loads are created by Koala. Again, for the 2D model, Rx and Rz are removed.



Figure 202. Create point loads wind.

# H.5 Creation of .xml File

## Create .xml File by Koala

Using all the input created in the previous section, the Koala CreatXML component combines all the input and creates a .xml file for structural analysis in SCIA, see Figure 203.

Crea	ate XML -	> :	SCIA	١
(	StructureType			ן ו
4	UlLanguage			
4	Materials			
4	MeshSize			
۷.	Sections			
€ 🖌	Nodes			L
۷.	Beams			
¢	Surfaces			
4	Openings			
¢	Nodesupports			
¢	Edgesupports			
d	Loadcases			
4	Loadgroups			
4	Lineloads			
4	Surfaceloads			
4	FreePointloads			
4	FreelLineloads			
٢	FreeSurfaceloads			L
٩	Hinges			
4	FileName			
4	Scale			
٩	RemDupINodes			
4	OnDemand			
4	Edgeloads			
4	PointloadsonPoints			
4	PointloadsonBeams			
٩	LinCombinations	B	filename	h
d	NonLinCombinations	9	and a second second	r I

Figure 203. Koala component for creation of .xml file.

## Editing of Koala .xml File

The .xml file created by the Koala plugin is not complete. The Koala plugin does not yet contain all possible inputs that are available in SCIA. Some of these inputs are required for proper analysis of the structural models. Such inputs are for example the load group and load duration. Therefore, these are manually added to the .xml file by copying the file generated by Koala and saving a newer version.

#### Prepare .xml File

The .xml file created by the Koala plugin is read and transferred into a .txt file, see Figure 204. Each .xml file is supported by an additional xml.def file, which also needs to be updated. In the newer .xml file, this updated xml.def file should be used.



Figure 204. Preparation of .xml file for editing and updating .def file.

#### Add New Section

Adding a new section to the .xml file is always done in four steps. The first step is to find the section which needs to be edited or location where the new section should be inserted. This is done by finding the unique ID string in the .txt file and snipping the section considered from the .txt file. In Figure 205, this step can be seen for finding the load group section.



Figure 205. Find load group section.

The second step is to retrieve the input that has to be inserted into the .txt file. See Figure 206 for the retrieval of input for the load groups and types.



Figure 206. Retrieve input for load groups & types.

The third step is to create the section that has to be inserted into the .txt file. The retrieved input is correctly formatted and a code block is created. Finally as the fourth step, the created section is inserted into the .txt file, replacing the original section, see Figure 207.



Figure 207. Add new code lines for load groups to file.

## Load Case Section

Similar to the steps for adding the section for load groups, the same procedure is followed for adding the load case section, see Figure 208 trough Figure 210.



Figure 208. Find LoadCase section.



Figure 209. Retrieve input for load case.



Figure 210. Add new code lines for load cases to file.

#### **ProjectData Section**

Also a section for project data needs to be inserted. This time, the input is not retrieved, but directly specified in a panel, see Figure 211 and Figure 212.



Figure 211. Find ProjectData section.



Figure 212. Add new code lines for project data.

#### Write Updated .xml File

The last step is to create the updated .xml file. This is simply done by transforming the updated .txt file into a .xml file, see Figure 213.



Figure 213. Write updated .xml file.

## Editing of Koala .xml.def File

The .xml.def file contains all the building blocks that can be used in the .xml file, which uses this .xml.def file. Within these building blocks, elements are presented, for which input values can be given. However, in the .xml.def file created by the Koala plugin, not all the elements, which are available in SCIA are present in the blocks. Therefore, they need to be added manually, so they can be used.

## Prepare .xml.def File

The .xml.def file created by the Koala plugin is read and transferred into a .txt file, see Figure 214.



Figure 214. Preparation of .xml.def file for editing.

## Add Element to Block

Adding an element to a block is done in three steps. The first step is to find the load group block, see Figure 215.



Figure 215. Find load group block.

The second step is to increase the size of the block with the number of elements that are added, see Figure 216.



Figure 216. Change size LoadGroup.

The third step is to add the new code lines for the additional element to the building block. This block is then inserted into the .txt file, replacing the old building block, see Figure 217.



Figure 217. Add new code lines for additional element to building block and replace building block in .txt file.

#### Load Case Block

Similar to the steps for adding the elements to the block for load groups, the same procedure is followed for adding the load case elements, see Figure 218 through Figure 220.



Figure 218. Find load case block.



Figure 219. Change size load case block.



Figure 220. Add new code lines for additional element to building block and replace building block in .txt file.

## Write Updated .xml.def File

The last step is to create the updated .xml.def file. This is simply done by transforming the updated .txt file into a .xml.def file, see Figure 221.



Figure 221. Write updated .xml.def file.